



Leibniz Institute for the Social Sciences

VARIABLE Reports

2013 34

European Values Study

EVS 2008 - Variable Report Austria

Documentation Release 2013/09/30
Related to the national dataset
GESIS Study-No. ZA4754, Version: 1.1.0, doi:10.4232/1.10152

European Values Study and GESIS Data Archive for the Social Sciences

GESIS-Variable Reports No. 2013 | 34

EVS 2008 – Variable Report Austria

Documentation release 2013/09/30

Related to the national dataset GESIS Study-No. ZA4754, Version: 1.1.0, doi:10.4232/1.10152

European Values Study and GESIS Data Archive for the Social Sciences

GESIS – Leibniz Institute for the Social Sciences 50667 Köln Unter Sachsenhausen 6–8 Germany

Phone: +49/(0)221/47694-467 Fax: +49/(0)221/47694-199

E-Mail: evelyn.brislinger@gesis.org

ISSN 2190-6742 (Online)

Publisher: GESIS – Leibniz Institute for the Social Sciences

Unter Sachsenhausen 6-8, 50667 Köln

info@gesis.org, www.gesis.org

Acknowledgements

The fieldwork of the 2008 European Values Study (EVS) was financially supported by universities and research institutes, national science foundations, charitable trusts and foundations, companies and church organizations in the EVS member countries.

A major sponsor of the surveys in several Central and Eastern European countries was Renovabis.



Renovabis - Solidarity initiative of the German Catholics with the people in Central and Eastern Europe: Project No. MOE016847 http://www.renovabis.de/.

An overview of all national sponsors of the 2008 survey is provided in the "EVS 2008 Guidelines and Recommendations", the "EVS 2008 Method Report", and on the website of the European Values Study http://www.europeanvaluesstudy.eu/evs/sponsoring.html.

The project would not have been possible without the National Program Directors in the EVS member countries and their local teams.

Gallup Europe developed a special questionnaire translation system WebTrans, which appeared to be very valuable and enhanced the quality of the project.

Special thanks also go to the teams at Tilburg University, CEPS/INSTEAD Luxembourg, and GESIS Data Archive for the Social Sciences Cologne.

Contents

In		ion and preliminary remarks	
1	Eur	opean Values Study 1981-2008	5
2	Acc	ess to data and documentation	7
3	Stu	dy Description	
	3.1	Bibliographic citation	
	3.2	Study scope	
	3.3	Methodology and processing	12
4	Dat	aset structure and variable definition	13
	4.1	Standards of dataset structure	13
	4.2	Administrative and ID variables	
	4.3	Data quality checks	16
	4.4	International standards and EVS coding frames	17
5	Var	able documentation and types of notes added	23
	5.1	Notes in English variable documentation according to Master Questionnaire	23
	5.2	Notes in the original language documentation of the dataset	23
6	Exp	lanation of the variable documentation	24
	Append	lix 1: Overview of flag variables to identify inconsistencies	26
	Variabl	e Documentation	30
		Archive and ID variables	
		Variables of content	
		Demographics and interview characteristics	
		List of contents of variable documentation	
Lis	st of Fig	ures and Tables	
Fig	gure 1:	Bilingual variable reports for all languages fielded in EVS 2008	3
Та	ble 1:	Participating countries in European Values Study 1981-2008	5
Та	ble 2:	Distinctions between Integrated Dataset and national datasets	13
Та	ble 3:	Overview of administrative variables	14
Та	ble 4:	Overview of harmonized variables	17
Та	ble 5:	Currency and date of exchange rate	20
Та	ble 6:	Country codes (ISO 3166-1)	21

Introduction and preliminary remarks

The variable documentation provided for every national dataset of the 2008 wave of the European Values Study consists of two main parts. The English Variable Report offers a description of the data on study and variable level and contains frequency counts for almost all variables. The supplementary bilingual variable documentation (English/original language) supports direct comparisons of English master questions with original questions as used in the field questionnaires.

English Variable Report

The Variable Report at hand is structured into six sections

- Section 1 provides brief information on the European Values Study.
- Section 2 gives an overview of all data and documentation available and information on how to access.
- Section 3 contains methodological information on origin and content of the data.
- Section 4 refers to the structure of the dataset and standardization efforts made.
- Section 5 provides information on variable documentation and notes added.
- The last section contains the Variable Documentation. It describes variables in the data file in terms of relevant metadata and frequency distribution.

Bilingual variable documentation

The figure below gives an overview of all languages fielded in the EVS 2008 member countries. For each language, a bilingual variable documentation is provided including the original question wording.

Figure 1: Languages fielded in EVS 2008*

	ted Dataset er Questionnaire						
Albania Albanian	Armenia Armenian	Austria German	Azerbaijan Azerbaijani Russian	Belarus Russian	Belgium Flemish/Dutch French	Bosnia- Herzegovina Bosnian	<i>Bulgaria</i> Bulgarian
Croatia Croatian	Cyprus Greek	Northern Cyprus Turkish	Czech Republic Czech	Denmark Danish	Estonia Estonian Russian	Finland Finnish	France French
Georgia Georgian Russian	Germany German	Great Britain English	Greece Greek	Hungary Hungarian	Iceland Icelandic	Ireland English	Italy Italian
Kosovo Albanian Serbian	<i>Latvia</i> Latvian Russian	Lithuania Lithuanian	Luxembourg Luxembourgish German	Netherlands Dutch	Northern Ireland English	<i>Norway</i> Norwegian	Macedonia Macedonia Albanian
Malta Maltese English	Moldova Moldovan Russian	Montenegro Montenegrin	French Portuguese English	Poland Polish	Portugal Portuguese	Romania Romanian Hungarian	Russia Russian
Serbia Serbian	Slovak Republic Slovak Hungarian	Slovenia Slovenian	<i>Spain</i> Spanish	Sweden Swedish	Switzerland German French Italian	Turkey Turkish	Ukraine Ukrainian Russian

^{*} The blue boxes indicate countries sharing one or more languages, shared languages are marked in red.

Data access

Primary data for statistical analysis and related documentation (English Master Questionnaire, national field questionnaires, variable reports, and method reports) are made available online by the GESIS Data Catalogue.

Anonymized data

According to data regulations in participating countries, only anonymized data are made available to users. Each national team was responsible for checking their data confidentiality before depositing data.

Citation requirements

Publications based on EVS data should acknowledge this by means of bibliographic citations as listed in the Study Description under item "Citation". To ensure that such source attributions are captured for social science bibliographic utilities, citations must appear in the footnotes or in the reference section of publications.

Usage requirement

To provide funding agencies with essential information about the use of EVS data and to facilitate the exchange of information about the EVS, users of EVS data are required to send bibliographic citations and/or electronic copies of each completed report, article, conference paper or thesis abstract using EVS data to the EVS team. Then, these will be included in the EVS repository. For more information, see http://www.europeanvaluesstudy.eu/evs/publications/.

Disclaimer

EVS, GESIS, and the producers bear no responsibility for the use of EVS data, or for interpretations or inferences based on these uses. EVS, GESIS, and the producers accept no liability for indirect, consequential or incidental damages or losses arising from use of the data collection, or from the unavailability of, or break in access to the service for whatever reason.

1 European Values Study 1981-2008

The European Values Study is the most comprehensive research project on human values in Europe. It is a large-scale, cross-national, and longitudinal survey research program on how Europeans think about family, work, religion, politics and society. Repeated every nine years in an increasing number of countries, the survey provides insights into the ideas, beliefs, preferences, attitudes, values, and opinions of citizens all over Europe.

The research program was initiated by the European Value System Study Group (EVSSG) in the late 1970s and has emerged as a well-established network of social and political scientists aiming at high standards in data creation and processing. From 1981 to 2008 four waves of survey were conducted in European and other countries. These surveys explore value differences, similarities, and value changes among citizens of the EVS member countries.

Survey 2008

The fourth wave has a persistent focus on a broad range of values. Questions with respect to family, work, religious, political and societal values are highly comparable with those in earlier waves (1981, 1990 and 1999). This longitudinal scope of the study makes it possible to study trends in time. The EVS has an increasing international and regional coverage. In 2008 it covers 47 countries/regions of Europe. Table 1 gives an overview of countries participating in EVS waves and year of fieldwork.

Table 1:	Participating	i countries in Euro,	pean Values Stud	y 1981-2008

Country/Region	1981	1990	1999	2008
USA	1982	1990		
Canada	1982	1990		
Belgium	1981	1990	1999	2009
Denmark	1981	1990	1999	2008
France	1981	1990	1999	2008
Germany*	1981	1990	1999	2008/2009
Great Britain	1981	1990	1999	2009/2010
Iceland	1984	1990	1999	2009/2010
Ireland	1981	1990	1999/2000	2008
Italy	1981	1990	1999	2009
Malta	1983/1984	1991	1999	2008
Netherlands	1981	1990	1999	2008
Northern Ireland	1981	1990	1999	2008
Spain	1981	1990	1999	2008
Sweden	1982	1990	1999/2000	2009/2010
Norway	1982	1990		2008
Austria		1990	1999	2008
Bulgaria		1991	1999	2008
Czech Republic (-1991Czechoslovakia)		1991	1999	2008
Estonia		1990	1999	2008
Finland		1990	2000	2009
Hungary		1991	1999	2008/2009
Latvia		1990	1999	2008
Lithuania		1990	1999	2008
Poland		1990	1999	2008
Portugal		1990	1999	2008

Romania	1993	1999	2008
Slovak Republic	1991	1999	2008
Slovenia	1992	1999	2008
Belarus		2000	2008
Croatia		1999	2008
Greece		1999	2008
Luxembourg		1999	2008
Russian Federation		1999	2008
Turkey		2001	2008/2009
Ukraine		1999	2008
Albania			2008
Armenia			2008
Azerbaijan			2008
Bosnia and Herzegovina			2008
Cyprus			2008
Northern Cyprus			2008
Georgia			2008
Kosovo			2008
Macedonia, Republic of			2008
Moldova, Republic of			2008
Montenegro, Republic of			2008
Serbia			2008
Switzerland			2008

^{*1981} only Germany (West)

Large efforts were taken to guarantee high scientific standards in developing and translating the Master Questionnaire and the field questionnaires, in high quality fieldwork, and standardized data processing and documentation.

A set of guidelines and recommendations was set up and the whole process of data creation and processing was guided and monitored by the EVS advisory groups. Data processing and documentation was implemented in a close cooperation with the EVS member countries. Further information and documentation is given at http://www.europeanvaluesstudy.eu/evs/surveys/survey-2008.html.

2 Access to data and documentation

The data and documentation of the four EVS waves is publicly available on the EVS and GESIS websites

The documentation created for the fourth EVS wave includes both the guidelines and standards developed to produce comparable data and detailed information on peculiarities of the national datasets at study as well as at variable level.

Data: Beside the Integrated Dataset the national datasets, including additional country-specific variables, are provided for national rather than multinational use.

Study Description: Study description offers methodological information on origin and content of the data and overviews on dataset versions and errata.

Reports: Standards and recommendations designed by the EVS advisory groups for all countries are compiled in the report "EVS 2008 Guidelines and Recommendations". The "EVS 2008 Method Report" includes both summarized information on the Integrated Dataset and country reports with country-specific information on the origin of the national datasets. It is based on the methodological questionnaires submitted by all EVS member countries. The report "EVS 2008 Project and Data Management" provides insights into the origin of the data and documentation, the project workflow, and the tools used.

Variable documentation: Variable reports cover the documentation of the national dataset with question texts of the English Master Questionnaire as well as frequency counts for almost all variables. The report is supplemented by bilingual (English/original language) variable documentation for each language fielded in a country.

Questionnaires: Field questionnaires in all languages fielded are downloadable as well.

Access to EVS data and documentation

General study information

The EVS website (www.europeanvaluesstudy.eu) covers comprehensive information on the origin of the project and provides access to all data and documentation available for EVS surveys.

Data and documentation download

Data and documentation can be obtained at the GESIS Data Archive for the Social Sciences in Cologne through their online download facilities GESIS Data Catalogue (http://gesis.org/datacatalogue/) and ZACAT-Online Study Catalogue (http://zacat.gesis.org/).

GESIS Data Catalogue contains study descriptions for all datasets providing information on the origin of data, version history, and errata. It supports direct access to data and documentation for download.

ZACAT provides data and documentation for retrieval purposes and data exploration. It assists users in identifying trend variables of all EVS waves. Furthermore, it enables comparisons of original questions in survey languages of the third and fourth wave. ¹

¹ GESIS Data Catalogue: https://dbk.gesis.org/dbksearch/GDESC2.asp?no=0009&DB=E ZACAT: http://zacat.gesis.org/webview/index.jsp?object=http://zacat.gesis.org/obj/fCatalog/Catalog5

Data analyses and data downloads are possible after registration.

Online variable information

Extended Study Description (http://info1.gesis.org/EVS/Studies): The extended Study Description for the EVS 2008 provides country-specific information on the origin and outcomes of the national surveys.

Online Variable Overview (http://info1.gesis.org/EVS/Variables): The variable overview allows for comparisons of trend variables of the four EVS waves 1981, 1990, 1999, and 2008. In addition, comparisons of original guestion texts across the waves 1999 and 2008 are supported.

Full-text retrieval

Question text retrieval is a facility for word/phrase searches in EVS text documents. It supports full-text searches in the master questionnaire and variable reports on the EVS integrated datasets and national datasets.²

Publication

EVS Repository (www.europeanvaluesstudy.eu/evs/publications/) contains direct links to publications based on the data of the European Values Study. A subset of the publications is enhanced by additional information on the data, interview questions, and descriptions of variable construction.

Atlas of European Values "Trends and Traditions at the turn of the Century"

The Atlas of European Values (www.atlasofeuropeanvalues.eu) unlocks the results of the European Values Study project for the general public. It presents values, beliefs, attitudes, and opinions through graphs, charts, and maps.

Secondary education

For secondary education, an interactive website offers courses and modules in different languages to provide a better understanding of European values in the classroom (http://www.europeanvaluesstudy.eu/evs/education.html).

² Question text retrieval: http://www.gesis.org/en/services/data-analysis/survey-data/european-values-study/question-text-retrieval-evs-1981-2008/

3 Study Description

Study description included in the GESIS Data Catalogue gives information about the origin of the data, as well as data releases, errors detected between data releases, and error corrections.

3.1 Bibliographic citation

Title: European Values Study 2008 - Austria

GESIS Study No.: 4754

Current version: 1.1.0, 2010-11-30, doi:10.4232/1.10152

Date of collection: 2008-07-21 to 2008-10-22

Citation

Publications based on EVS data should acknowledge this by means of bibliographic citations. To ensure that such source attributions are captured for social science bibliographic utilities, citations must appear in the footnotes or in the reference section of publications.

How to cite the data

EVS (2010): European Values Study 2008 - Austria. GESIS Data Archive, Cologne. ZA4754 Data File Version 1.1.0, doi:10.4232/1.10152.

How to cite this publication

EVS, GESIS (2013): European Values Study 2008 - Variable Report Austria. GESIS-Variable Report 2013/34.

Principal investigators

- Zulehner, Paul M., University of Vienna, Austria

Data collector

- GFK Austria GmbH, Vienna

3.2 Study scope

Abstract

Moral, religious, societal, political, work, and family values of Europeans.

Topics

1. Perceptions of life: importance of work, family, friends and acquaintances, leisure time, politics and religion; frequency of political discussions with friends; happiness; self-assessment of own health;

memberships and unpaid work (volunteering) in: social welfare services, religious or church organisations, education, or cultural activities, labour unions, political parties, local political actions, human rights, environmental or peace movement, professional associations, youth work, sports clubs, women's groups, voluntary associations concerned with health or other groups; tolerance towards minorities (people with a criminal record, of a different race, left/right wing extremists, alcohol addicts, large families, emotionally unstable people, Muslims, immigrants, AIDS sufferers, drug addicts, homosexuals, Jews, gypsies and Christians - social distance); trust in people; estimation of people's fair and helpful behaviour; internal control: satisfaction life. or external with 2. Work: reasons for people to live in need; importance of selected aspects of occupational work; employment status; general work satisfaction; freedom of decision-taking in the job; importance of work (work ethics, scale); important aspects of leisure time; attitude towards following instructions at work without criticism (obedience work); give priority to nationals over foreigners as well as men over women in jobs.

- 3. Religion: Individual or general clear guidelines for good and evil; religious denomination; current and former religious denomination; current frequency of church attendance and at the age of 12; importance of religious celebration at birth, marriage, and funeral; self-assessment of religiousness; churches give adequate answers to moral questions, problems of family life, spiritual needs and social problems of the country; belief in God, life after death, hell, heaven, sin and re-incarnation; personal God versus spirit or life force; own way of connecting with the divine; interest in the sacred or the supernatural; attitude towards the existence of one true religion; importance of God in one's life (10-point-scale); experience of comfort and strength from religion and belief; moments of prayer and meditation; frequency of prayers; belief in lucky charms or a talisman (10-point-scale); attitude towards the separation of church and state.
- 4. Family and marriage: most important criteria for a successful marriage (scale); attitude towards childcare (a child needs a home with father and mother, a woman has to have children to be fulfilled, marriage is an out-dated institution, woman as a single-parent); attitude towards marriage, children, and traditional family structure (scale); attitude towards traditional understanding of one's role of man and woman in occupation and family (scale); attitude towards: respect and love for parents, parent's responsibilities for their children and the responsibility of adult children for their parents when they are in need of long-term care; importance of educational goals; attitude towards abortion. 5. Politics and society: political interest; political participation; preference for individual freedom or social equality; self-assessment on a left-right continuum (10-point-scale); self-responsibility or governmental provision; free decision of job-taking of the unemployed or no permission to refuse a job; advantage or harmfulness of competition; liberty of firms or governmental control; equal incomes or incentives for individual efforts; attitude concerning capitalism versus government ownership; postmaterialism (scale); expectation of future development (less emphasis on money and material possessions, greater respect for authority); trust in institutions; satisfaction with democracy; assessment of the political system of the country as good or bad (10-point-scale); preferred type of political system (strong leader, expert decisions, army should rule the country, or democracy); attitude towards democracy (scale).
- 6. Moral attitudes (scale: claiming state benefits without entitlement, cheating on taxes, joyriding, taking soft drugs, lying, adultery, bribe money, homosexuality, abortion, divorce, euthanasia, suicide, corruption, paying cash, casual sex, avoiding fare on public transport, prostitution, experiments with human embryos, genetic manipulation of food, insemination or in-vitro fertilization and death penalty).
- 7. National identity: geographical group the respondent feels belonging to (town, region of country, country, Europe, the world); citizenship; national pride; fears associated with the European Union (the loss of social security and national identity, growing expenditure of the own country, the loss of power in the world for one's own country and the loss of jobs); attitude towards the enlargement of the European Union (10-point-scale); voting intensions in the next election and party preference; party

that appeals most; preferred immigrant policy; opinion on terrorism; attitude towards immigrants and their customs and traditions (take jobs away, undermine a country's cultural life, make crime problems worse, strain on country's welfare system, threat to society, maintain distinct customs and traditions); feeling like a stranger in one's own country; too many immigrants; important aspects of national identity (being born in the country, to respect country's political institutions and laws, to have country's ancestry, to speak the national language, to have lived for a long time in the country); interest in politics in the media; give authorities information to help justice versus stick to own affairs; closeness to family, neighbourhood, the people in the region, countrymen, Europeans and mankind; concerned about the living conditions of elderly people, unemployed, immigrants and sick or disabled people.

8. Environment: attitude towards the environment (scale: readiness to give part of own income for the environment, overpopulation, disastrous consequences from human interference with nature, human ingenuity remains earth fit to live in, the balance of nature is strong enough to cope with the impacts of modern industrial nations, humans were meant to rule over the rest of nature, an ecological catastrophe is inevitable).

Demography: sex; age (year of birth); born in the country of interview; country of birth; year of immigration into the country; father and mother born in the country; country of birth of father and mother; current legal marital status; living together with the partner before marriage or before the registration of partnership; living together with a partner and living with a partner before; steady relationship; married to previous partner; living together with previous partner before marriage; end of relationship; number of children; year of birth of the first child; size and composition of household; experienced events: the death of a child, of father or mother, the divorce of a child, of the parents or of another relative; age of respondent when these events took place; age at completion of education; highest educational level attained; employment status; employed or self-employed in the last job; profession (ISCO-88) and occupational position; supervising function and span of control; size of company.

Social origin and partner: respondent's partner or spouse: partner was born in the country and partner's country of birth; highest educational level; employment status of the partner; employment or self-employment of the partner in his/her last job; partner's profession (ISCO-88) and occupational position; supervising function of the partner and span of control; unemployment and dependence on social-security of the respondent and his partner longer than three months in the last five years; scale of household income; living together with parents when the respondent was 14 years old; highest educational level of father/mother; employment status of father/mother when the respondent was 14 years old; profession of father/mother (ISCO-88) and kind of work; number of employees (size of business); supervising function and span of control of father and mother; characterization of the parents when respondent was 14 years old (scale: liked to read books, discussed politics at home with their child, liked to follow the news, had problems making ends meet, had problems replacing broken things); region the respondent lived at the age of 14, present place of residence (postal code); size of town; region.

Interviewer rating: respondent's interest in the interview.

Additionally encoded: interviewer number; date of the interview; total length of the interview; time of the interview (start hour and start minute, end hour and end minute); language in which the interview was conducted.

Additional country specific variables are included in this national dataset.

Universe

Persons 18 years or older who are resident within private households, regardless of nationality and citizenship or language.

3.3 Methodology and processing

Selection method

A representative multi-stage or stratified random sample was used in EVS member countries.

For more country-specific information about the sampling procedure, please go to the EVS website http://www.europeanvaluesstudy.eu/evs/surveys/survey-2008/participatingcountries/ and see the method reports provided for all participating countries.

Mode of data collection

Face-to-face interview with standardized questionnaire - PAPI (paper)

Fieldwork was conducted on the basis of detailed and uniform instructions prepared by the EVS advisory groups. The English basic questionnaire was translated into other languages by means of the questionnaire translation system WebTrans, a web-based translation platform designed by Gallup Europe. The whole translation process was closely monitored and quasi-automated documented.

Number of units: 1510

Number of variables: 525

4 Dataset structure and variable definition

The structure of the national datasets and the Integrated Dataset are guided by all-over harmonization and standardization based on the Master Questionnaire, EVS 2008 Guidelines and Recommendations³, set up syntax files for general variables and contact forms, respectively.

4.1 Standards of dataset structure

- Distinctions between Integrated Dataset and national datasets

The Integrated Dataset differs from the national datasets in a number of aspects (see table 2).

rabie 2:	Distinctions between integrated Dataset and national datasets
A +	

Aspect	Integrated Dataset	National datasets
N of variables	468	at least 441
Country-specific variables	-	in at hr cy ncy cz DK fi fr De GB GR hu is it kos me pl Pt ro sk es se ch tr ua
StudyNo	GESIS Study Number	GESIS Study Number
StudyNoC	GESIS Study Number (national datasets)	-
Version	GESIS Archive Version	GESIS Archive Version
VersionC	GESIS Archive Version (national datasets)	-
V353_ppp	Monthly household income in Euros (corrected for purchasing power parity)	-
weight	General weight	General weight
weight_c	Weight for country-specific characteristics DE and BE	weight for country-specific characteristics DE and BE
country-specific weight	-	in at HR FR KOS RO SK SE CH

- Naming conventions

The naming of variables follows several standards differentiated by type of variables.

Standard variables

In the national datasets and the Integrated Dataset all standard variables are build in the scheme "vNNN" (e.g. v301), except for harmonized variables with respective suffixes and administrative variables (see below "3.3 Administrative variables"). The harmonized variables are compiled in the Integrated Dataset and generally have an intuitive suffix (e.g. v339SIOPS for "occupational status respondent" measured in "Standard International Occupational Prestige Scale"). For an overview of harmonized/standardized variables see table 4.

Country-specific variables

In 25 national datasets country-specific variables are included. For 24 countries (see table 2 above) additional questions/items were included in the field questionnaires. These questions/items were in-

EVS (2010): EVS 2008 Guidelines and Recommendations. GESIS-Technical Reports 2010/16. Retrieved from http://dbk.gesis.org/dbksearch/file.asp?file=ZA4800_standards.pdf

cluded as additional country-specific variables which are named by the alpha-numerical country code (ISO 3166-1 alpha-2) plus a continuous two (/three)-digit number (e.g. ATO1). Their positions in the datasets mirror their positions in the field questionnaires. In addition, for eight countries country-specific weights were provided by the Program Directors (see table 2). These variables are named by the same scheme as the other country-specific variables and positioned at the end of the national datasets.

- Missing value codes

For all variables a set of missing value codes with negative values was defined and applies throughout.

- 1 "dk" (i.e. "don't know")
- 2 "na" (i.e. "no answer")
- 3 "nap" (i.e. "not applicable")
- 4 "question not asked" (i.e. question not included in country's field questionnaire or not provided for by Program Director)
- 5 "other missing" (includes: system missing, wild codes, illogic answer-patterns in e.g. filters)

- Dummy variables

For questions that allow for multiple responses a dummy variable is created for each answer category. E.g., a question with four answer categories in an item battery has four dummy variables as final data. In general, a value "1" means that the respondent chose the particular category as an answer (mentioned, yes, ...). In case of a value "0" the respondent did not choose the particular category as answer (not mentioned, no, ...). Dummy variables are also created for "None of these", "Don't know" or other residual responses. Inconsistencies between category 'none of these' and other mentioned items of a battery are flagged by flag variables. (see section 4.3: Flag variables indicating inconsistencies).

4.2 Administrative and ID variables

Administrative variables are used mainly for identifying datasets and respondents (see table 3 below). They include different kinds of variables:

- Technical variables for identifying the dataset or cases
- Weighting variables to help adjust samples to the universe

Table 3: Overview of administrative variables

Variable name	Variable label
StudyNo	GESIS Study Number
StudyNoC*	GESIS Study Number (National datasets)
Version	GESIS Archive Version
VersionC*	GESIS Archive Version (National datasets)
id_cocas	unified respondent number
caseno	original respondent number
intno	interviewer number
wave	wave
year	survey year
country	country code
country1	country code [with split ups]
c_abrv	country abbreviation
c_abrv1	country abbreviation [with split ups]

weight	weight
weight_c	weight for country-specific characteristics
cntry_y	country_year
cntry1_y	country_[with split ups]- year

^{*} Variable not included in national datasets

- Version of dataset

In order to uniquely identify a dataset, each dataset is assigned a GESIS Study No., a GESIS archive version no., and a Digital Object Identifier (DOI). The DOI name points to the Study Description in the GESIS Data Catalogue including an overview of errata and dataset versions.

Variable "version" includes the GESIS archive version number of a dataset and the corresponding release date. It is composed of a sequence of three numbers (alphanumeric string).

- The major number is incremented when there are changes in the composition of the data set (e.g. additional variables or cases).
- The minor or second number is incremented when significant errors have been fixed (e.g. coding errors, misleading value labels).
- The third or revision number is incremented when minor bugs are fixed (e.g. spelling errors in variable or value labels).

- Country variables with split ups

Country variables country, c_abrv and cntry_y each have a pendant of a country variable "with split ups". In the country variables country1, c_abrv1 and cntry1_y Germany is represented by two different codes (ISO 3166-1) for East and West Germany (for further description of country variables see below "4.4 International standards and EVS coding frames" - sub-bullet "Nationality: ISO 3166-1").

Variables: country; country1; c_abrv; c_abrv1; cntry_y; cntry1_y

- Identification variables

Each record in the dataset contains two identification variables:

- Caseno: 'Original respondent number'
- Id_cocas: 'Unified respondent number' structured as a 12 digit code: YYYY (year of wave), CCCC (country ISO 3166-1), NNNN (new continuous case number)

- Weight: gender by age

The variable "weight" was computed by the EVS for all national datasets on the basis of information and population statistics provided by the EVS countries. The weight is constructed on the basis of gender and age categories (-24; 25-34; ...; 65-74; 75 and over). Value '0' implies that year of birth information was missing in the data.

The weight adjusts the socio-structural characteristic in the samples to the distribution of gender and age of the universe-population. The current weight variable should be used with caution. Especially when the weights are "big", say outside the 0.50-2.00 range.

Variable: weight

For some national datasets country specific original weight variables are provided additionally as delivered by countries (Austria, Croatia, France, Kosovo, Romania, Slovak Republic, Sweden, and Switzerland).

4.3 Data quality checks

The data of the EVS 2008 national datasets was checked for wild codes, system missing, filter inconsistency, and contradictions between variables of item batteries.

- Flag variables indicating inconsistencies

For the most common non-conformed interviewer instructions and illogic answer-patterns across national datasets flag variables were created which mark such non-conformities and leave correction-decisions up to the researcher. Information about both flag and flagged variables is provided by notes in the Variable Documentation.

Example:

Flagged variable: v67 'why do people live in need-most important (Q12)'

Note added: Please see flag variable f67 which indicates inconsistencies for variables v67/v68

Flag variable f67 'flag variable: why do people live in need - most/second important (Q12/Q13)'

Indicates inconsistencies for the variables v67/v68.

If most important reason is identical with next most important reason.

if v67 = v68 then f67=1

(0 consistent; 1 inconsistent)

There are three types of non-conformities which were flagged.

1. "first/second choice"-variables which were flagged in case the value of the variable for the first choice equaled to the value of the variable for the second choice (see Appendix 1).

```
Variables: v67/v68 - f67; v201/v202 - f201; v253/v254 - f253
```

2. "multiple response question"-variables which were flagged in case respondents chose more than the 5 maximally permitted items (see Appendix 1).

Variables: v170 to v180 - f180

3. "'none'-item battery"-variables which were flagged in case the answer-pattern of the "none"-item variable and the other variables of the item battery of a question was illogic. As several illogic answer-patterns are possible the flag variables have nine categories (see Appendix 1).

In the national datasets some countries included additional items within these item batteries which had to be taken into account as base for the flagging of answer-patterns as well. As these country-specific variables are not included in the Integrated Dataset the bases for flagging changed minimally, i.e. the sum of flagged cases in the national datasets does not equal to the total of flagged cases in the Integrated Dataset (see Appendix 1).

Variables: v10 to v25 - f25; v28 to v43 - f43; v69 to v86 - f86; v170 to v181 - f181

- Notes on deviating filtering

In case filtering in countries' field questionnaires deviated from filtering in the Master Questionnaire (either by lacking filter or by adding country specific filter) the Master Questionnaire's filtering was restored by recoding. Only systematic filter deviations in national datasets were remarked by variable notes, not by flag variables. The data in general was cleaned according to the skip pattern of the Master Questionnaire.

4.4 International standards and EVS coding frames

International standards and EVS coding frames are used for harmonization/standardization of EVS datasets concerning variables on religion, political party preference, occupation, education, income, region, nationality, and language. Harmonization of these variables enables comparison between different countries.

Table 4: Overview of harmonized variables

/ariable name	Variable label
/106	which religious denomination do you belong to (Q23a) (standardized)
v106_cs	which religious denomination do you belong to (Q23a)
v108	did you belong to a religious denomination (Q24a) (standardized)
v108_cs	did you belong to a religious denomination (Ω24a)
v264	which political party would you vote for (Q75a)
v264_LR	which political party would you vote for - left/right scale (Q75a)
v265	which political party appeals to you most (Q75b)
v265_LR	which political party appeals to you most - left/right scale (Q75b)
v336	Educational level respondent: ISCED-code one digit (Q110)
v336_2	Educational level respondent: ISCED-code two digits (Q110)
v336_3	Educational level respondent: ISCED-code three digits (Q110)
v336_cs	Country-specific: ISCED code education respondent (Q110)
v339ISCO	kind of job respondent - ISCO88 code (Q112)
v339SIOPS	occupational status respondent - SIOPS (Q112)
v339ISEI	occupational status respondent - ISEI (Q112)
v339ESeC	occupational status respondent - European ESeC (Q112)
v339egp	occupational status respondent - egp11 (Q112)
v344	Educational level partner: ISCED-code one digit (Q116)
v344_2	Educational level partner: ISCED-code two digits (Q116)
v344_3	Educational level partner: ISCED-code three digits (Q116)
v344_cs	Country-specific: ISCED code education partner (Q116)
v346ISCO	kind of job spouse/partner - ISCO88 code (Q118)
v346SIOPS	occupational status spouse/partner - SIOPS (Q118)
v346ISEI	occupational status spouse/partner - ISEI (Q118)
v346ESeC	occupational status spouse/partner - European ESeC (Q118)
v346egp	occupational status spouse/partner - egp11 (Q118)
v353WK	Weekly household income (Q125)
v353MM	monthly household income (Q125)
v353YR	annual household income (Q125)
v353W_cs	Country-specific: Weekly household income (Q125)
v353M_cs	Country-specific: Monthly household income (Q125)
v353Y_cs	Country-specific: Annual household income (Q125)
v355	Educational level father [mother]: ISCED-code one digit (Q127)
v355_2	Educational level father [mother]: ISCED-code two digits (Q127)
v355_3	Educational level father [mother]: ISCED-code three digits (Q127)
v355_cs	Country-specific: ISCED code education father (Q127)
v357ISCO	kind of job father/mother - ISCO88 code (Q129a)
v357SIOPS	occupational status father/mother - SIOPS (Q129a)
v357ISEI	occupational status father/mother - ISEI (Q129a)
v357ESeC	occupational status father/mother - European ESeC (Q129a)
v357egp	occupational status father/mother - egp11 (Q129a)
v368b_CC	Region at age 14: country (Q133)
v368b_N3	Region at age 14: NUTS-3 code (Q133)
v368b_N2	Region at age 14: NUTS-2 code (Q133)

v368b_N1	Region at age 14: NUTS-1 code (Q133)
v371b_N3	Region: NUTS-3 code (Q136)
v371b_N2	Region: NUTS-2 code (Q136)
v371b_N1	Region: NUTS-1 code (Q136)

- Religion: EVS-categories

The categories of the compiled variables v106_cs and v108_cs consist of all religious denominations included in the field questionnaires. The values are a combination of the country identifier (ISO 3166-1) and the given order of religious denominations in the field questionnaires.

A harmonized version of the religious denominations can be found in the variables v106 and v108. All categories of religious denominations used in the field questionnaires were assigned (based on the assignments made for EVS 1999 and/or given by Program Directors) to these variables harmonized categories.

The categories for religion are:

- 1 "Roman Catholic"
- 2 "Protestant"
- 3 "Free church/ non-conformist/ evangelical"
- 4 "Jew"
- 5 "Muslim"
- 6 "Hindu"
- 7 "Buddhist"
- 8 "Orthodox"
- 9 "Other"

Variables: v106; v106_cs; v108; v108_cs

- Party preference: EVS-categories

The categories of the compiled variables v264 and v265 consist of all political parties included in the field questionnaires. The values are a combination of the country identifier (ISO 3166-1) and the given order of political parties in the field questionnaires.

Variables v264_LR and v265_LR are based on the positioning of political parties on a left/right scale given in the methodological reports.

Variables: v264; v264_LR; v265; v265_LR

- Occupation

ISCO88 4 digits

International Standard Classification of Occupations

Source(s): http://www.ilo.org/public/english/bureau/stat/isco/isco88/index.htm

http://www.ilo.org/public/english/bureau/stat/isco/isco88/major.htm

Date: 18-09-2004 (Last update according to site ILO)

See also: EVS 2008 Guidelines and Recommendations 2010/16; Appendix C Official Classifications

http://info1.gesis.org/dbksearch/file.asp?file=ZA4800_standards.pdf

Variables: v339ISCO, v346ISCO, v357ISCO

SIOPS

Standard International Occupational Prestige Scale (Treiman, 1977)

Source classification: Ganzeboom, H. B. G. & Treiman, D. J. (1996), Internationally Comparable Measures of Occupational Status for the 1988 International Standard Classification of Occupations. Social Science Research, 25, 201-239.

Source syntax to compute classification: http://home.fsw.vu.nl/hbg.ganzeboom/pisa/index.htm Variables: v339SIOPS, v346SIOPS, v357SIOPS

ISFI

International Socio-Economic Index of Occupational Status (Ganzeboom et al., 1992)

Source classification: Ganzeboom, H. B. G. & Treiman, D. J. (1996), Internationally Comparable Measures of Occupational Status for the 1988 International Standard Classification of Occupations. *Social Science Research*, *25*, 201-239.

Source syntax to compute classification: http://home.fsw.vu.nl/hbg.ganzeboom/pisa/index.htm

Variables: v339ISEI, v346ISEI, v357ISEI

FGP

EGP class scheme (Erikson/Goldthorpe/Portocarero, 1979)

Source classification: Ganzeboom, H. B. G. & Treiman, D. J. (1996), Internationally Comparable Measures of Occupational Status for the 1988 International Standard Classification of Occupations. Social Science Research, 25, 201-239.

Source syntax to compute classification: http://home.fsw.vu.nl/hbg.ganzeboom/pisa/index.htm

Variables: v339egp, v346egp, v357egp

EseC

European Socio-economic Classification

Source: http://www.iser.essex.ac.uk/research/esec/matrices-and-syntax

Date: 27-09-2006 (last update syntax). Variables: v339ESeC, v346ESeC, v357ESeC

- Education: ISCED-97

International Standard Classification of Education

Source: OECD (1999). Classifying Educational Programmes: Manual for ISCED-97 Implementation in

OECD Countries, 1999 edition.

Date: 1999

Note: The educational classifications of countries were based on the information about the ISCED-97 levels in the manual. However, not the country-specific classifications mentioned in the manual were used, but the national Program Directors and the Methodology group created a classification in which all national educational programmes were included.

Variables: v336; v336_2; v336_3; v336_cs; v344; v344_2; v344_3; v344_cs; v355_2; v355_3; v355_cs

- Income: EVS-categories

Source: Draft Fieldwork Guidelines 05-11-2007.pdf

Date: 05-11-2007

Note: Countries were asked to convert the twelve (weekly, monthly and annual) EVS income categories in their national currency. Countries were allowed to further split up the categories if necessary. In some cases a lower cut-off point was allowed. The following table informs about the currency and the date of the exchange rates used for Euro conversion. Since the monetary values are asked in Euro for those countries having the Euro, a conversion for those countries was not necessary.

Variables: v353WK, v353MM, v353YR, v353W_cs, v353M_cs, v353Y_cs

Table 5: Currency and date of exchange rate

Country/Region	Currency	Date of exchange rate
Albania	Lek	29.02.2008
Armenia	Dram	21.01.2008
Austria	Euro	nap
Azerbaijan	Manat	23.01.2008
Belarus	Belarusian Ruble	06.01.2008
Belgium	Euro	nap
Bosnia Herzegovina	Convertible Mark	10.03.2008
Bulgaria	Lev	25.01.2008
Croatia	Kuna	22.01.2008
Cyprus	Euro	nap
Northern Cyprus	Turkish Lira	29.02.2008
Czech Republic	Czech Koruna	21.01.2008
Denmark	Danish Krone	21.01.2008
Estonia	Estonian Kroon	26.03.2008
Finland	Euro	nap
France	Euro	nap
Georgia	Lari	11.03.2009
Germany	Euro	nap
Great Britain	Pound Sterling	10.01.2008
Greece	Euro	nap
Hungary	Forint	04.12.2007
Iceland	Icelandic Króna	05.06.2009
Ireland	Euro	nap
Northern Ireland	Pound Sterling	10.01.2008
Italy	Euro	nap
Kosovo	Euro	nap
Latvia	Lats	25.06.2008
Lithuania	Lithuanian Litas	30.06.2008
Luxembourg	Euro	nap
Macedonia, Republic of	Macedonian Denar	11.03.2008
Malta	Euro	nap
Moldova, Republic of	Moldovan Leu	09.01.2008
Montenegro, Republic of	Euro	nap
Netherlands, The	Euro	nap
Vorway	Norwegian Krone	28.03.2008
Poland	Złoty	25.02.2008
Portugal	Euro	nap
Romania	Romanian Leu	05.12.2007
Russian Federation	Ruble	06.02.2008
Serbia	Serbian Dinar	01.12.2007
Slovak Republic	Slovanska Koruna	31.01.2008
Slovenia	Euro	nap
Spain	Euro	nap
Sweden	Swedish Krona	unknown
Switzerland	Swiss Francs	20.03.2008
Turkey	Turkish Lira	unknown
Ukraine	Hryvnia	07.01.2008

- Regional codes: NUTS-3-2-1

Nomenclature of Units for Territorial Statistics

Source(s): http://epp.eurostat.ec.europa.eu/portal/page/portal/nuts_nomenclature/introduction

Date: 28-07-2010

See also: EVS 2008 Guidelines and Recommendations 2010/16; Appendix C Official Classifications http://info1.gesis.org/dbksearch/file.asp?file=ZA4800_standards.pdf

Note: For countries for which no official NUTS classification was available, a regional classification was created by the national Program Directors and the Methodology group that resembled the NUTS classification as much as possible.

Variables: v305b, v307b, v310b, v312b, v343b, v368b_CC, v368b_N3, v368b_N2, v368b_N1, v371b_N3, v371b_N2, v371b_N1

- Language: ISO 639-1

International Organization for Standardization - Codes for the representation of names of languages

Source(s): http://en.wikipedia.org/wiki/ISO_639 http://en.wikipedia.org/wiki/List_of_ISO_639-1_codes

Version/date: 2002

See also: EVS 2008 Guidelines and Recommendations 2010/16; Appendix C Official Classifications

http://info1.gesis.org/dbksearch/file.asp?file=ZA4800_standards.pdf

Variable: v376

- Nationality: ISO 3166-1

International Organization for Standardization - Codes for the representation of names of countries and their subdivisions

Source: http://www.iso.org/iso/english_country_names_and_code_elements

Date: 21-07-2007

Note: We did add some (former) countries

See also: EVS 2008 Guidelines and Recommendations 2010/16; Appendix C Official Classifications

http://info1.gesis.org/dbksearch/file.asp?file=ZA4800_standards.pdf

Variables:

- id_cocas "unified respondent number" (structured as a 12 digit code: YYYY (year of wave), CCCC (country ISO 3166-1), NNNN (new continuous case number)),
- c_abrv and c_abrv1 (ISO 3166-1-alpha-2)
- v305b, v307b, v310b, v312b, v343b, v368b_CC (country ISO 3166-1)

Table 6: Country codes (ISO 3166-1)

Country/Region	Country abbreviation ISO 3166-1 alpha-2	Study No.	ISO 3166-1	EVS: ISO 3166-1
Albania	AL	4783	008	0080
Armenia	AM	4784	051	0510
Austria	AT	4754	040	0400
Azerbaijan	AZ	4785	031	0310
Belarus	BY	4782	112	1120
Belgium	BE	4759	056	0560
Bosnia-Herzegovina	ВА	4786	070	0700
Bulgaria	BG	4774	100	1000
Croatia	HR	4775	191	1910
Cyprus	CY	4787	196	1961
Northern Cyprus	NCY*	4788	_*	1962
Czech Republic	CZ	4770	203	2030
Denmark	DK	4760	208	2080
Estonia	EE	4766	233	2330
Finland	FI	4762	246	2460
France	FR	4751	250	2500
Georgia	GE	4789	268	2680
Germany West	DE	4753	276	2761
Germany East	DE	4753	276	2762
Great Britain	GB	4752	826	8261
Greece	GR	4776	300	3000
Hungary	HU	4772	348	3480
Iceland	IS	4763	352	3520
Ireland	IE	4765	372	3720
Northern Ireland	NIR*	4764	-*	8262
Italy	IT	4755	380	3800
Kosovo	KOS*	4790	_*	9150
Latvia	LV	4790	428	4280
Lithuania	LT	4768	440	4400
Luxembourg	LU	4779	442	4420
Macedonia, Republic of	MK	4779	807	8070
Malta	MT	4778	470	4700
	MD	4778	470	4700
Moldova, Republic of		4793	498	
Montenegro, Republic of	ME NL	4794	528	4990 5280
Netherlands, The				
Norway	NO	4795	578	5780
Poland	PL	4769	616	6160
Portugal	PT	4757	620	6200
Romania	RO	4773	642	6420
Russian Federation	RU	4777	643	6430
Serbia	RS	4796	688	6880
Slovak Republic	SK	4771	703	7030
Slovenia	SI	4780	705	7050
Spain	ES	4756	724	7240
Sweden	SE	4761	752	7520
Switzerland	CH	4797	756	7560
Turkey	TR	4791	792	7920
Ukraine	UA	4781	804	8040

^{*}No official ISO 3166-1 and ISO 3166-1 alpha-2 code.

5 Variable documentation and types of notes added

The variable documentation is set up by means of the Dataset Documentation Manager (DSDM) based on DDI/XML v.2. The documentation combines the exact wording and sequence of question components (question text, interviewer instruction, response categories, etc.) and show cards from a questionnaire with the variable description (variable name, values, and labels) of a dataset.

For the national datasets, two types of variable documentation are available. Beside the present English documentation, bilingual (English/original languages) variable documentation facilitates a comparison of the English master and original question wording as in field questionnaire.

General remarks and comments referring to a certain variable are defined as variable notes. Notes could be references to trend/comparability information, international standards applied or EVS coding frames, problems in questionnaire translation, but also references to data problems or specific characteristics. A distinction between two types of variable notes can be made:

- Notes referring to the English documentation of national datasets,
- Notes that refer to original language documentation set up for on basis of the field questionnaires.

5.1 Notes in English variable documentation according to Master Questionnaire

- Harmonized variables (religion, left/right political preference, education, income, region coded NUTS 2 and NUTS 1): reference to international standards or EVS coding frames used for harmonization; mention of the respective standardized country-specific variable ("_cs" variables).
- Standardized country-specific variables (religion, political party preference, education, income) containing country identifier (ISO 3166-1) and the given order of country-specific categories in the field questionnaires; reference to respective harmonized variable.
- Standardized variables with reference to international standards (nationality ISO 3166-1 alpha-2 code, language - ISO 639-2 code, region - NUTS 3 code, and occupation - ISCO88 4 digits code).
- Information on coding of open-ended questions and multiple response questions.
- Notes referring deviations between the master and the field questionnaire.
- Notes documenting translation problems.
- Information on data processing.

5.2 Notes in the original language documentation of the dataset

- Standard note added to the original language documentation referring to the recoding and documentation of missing values.
- Notes on harmonized variables with reference to the original country-specific variables.
- Notes on errors in field questionnaires (e.g. typing error).

6 Explanation of the variable documentation

The variable documentation part describes each variable in the analysis data file in terms of relevant metadata and frequency distributions. It is set up by means of the GESIS Dataset Documentation Manager (DSDM) and in accordance with the international metadata specification for the social and behavioral sciences, established by the Data Documentation Initiative (DDI-Codebook/XML).

The documentation combines the variable description (variable name, values, and labels) of the dataset with the exact wording and sequence of question components (question text, interviewer instruction, response categories, etc.) from the questionnaire.

The example explained below reproduces the information appearing in the variable documentation part for a typical substantive variable. In the case of variables with long lists of coded "real values" (e.g. age), results are only presented for the corresponding categorized variable, if available. The content and construction of technical, administrative, or other generated variables is described as appropriate.

If an adequate weight is provided in the dataset, the frequencies are calculated on the base of weighted data, as indicated with each table. Due to rounding, users may find slightly different figures if using other statistical packages, especially if the number of cases is very low.

The complete question and answer texts are taken from the English Master Questionnaire as provided by the European Values Study.

Digital Object Identifier (doi):

Header with doi linked to the Study Description in the GESIS Data Catalogue.

VARIABLE NAME (serial number from questionnaire or assigned by the archive) and VARIABLE LABEL as defined in the dataset, including question number from questionnaire. The numbers or letters that may appear together with the variable name reflect the item order (multiple item or multiple response questions) or indicate if a question was recoded (R), harmonized, or if asked only in a single country (country abbreviation).

QUESTION NUMBER and full QUESTION TEXT from the questionnaire, including interviewer instructions.

CODE VALUES occurring in the data for this variable and complete ANSWER TEXT from the questionnaire. Code values may differ from the questionnaire specification due to standardization and harmonization.

EVS 2008 - France

GESIS StudyNo 4751, v2.0.0, http://dx.doi.org/doi:10.4232/1.10198

v13 - do you belong to: trade unions (Q5aD)

Q5a <SHOW CARD 5>

Please look carefully at the following list of voluntary organisations and activities and say ... a) which, if any, do you belong to?

<CODE ALL MENTIONED UNDER (a)>

Q5a.D Trade unions

- -5 other missing
- 4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

v13, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.1	
-1	dk	М	3	0.1	
1	mentioned		146	4.8	4.8
2	not mentioned		2919	95.1	95.2
	Sum		3071	100.0	100.0
	Valid Cases		3064		

NOTES delineate additional explanatory text subsuming archive remarks on data processing, deviations between released data in dataset, and other issues relevant for the understanding and application of the variable. Further information refers to the construction principle of indices or other derived variables (not in the example).

FREQUENCY COUNTS: The absolute frequencies of the occurrence of values (count) refer to the number of cases in the dataset, which is not weighted in some countries, in this example weighted by weight. Relative frequencies are calculated including (percent) and excluding (valid percent) missing data (M), i.e. percent on the basis of all cases (sum) and valid percent on the basis of the valid cases.

Appendix 1: Overview of flag variables to identify inconsistencies

Variable	Label	N	total N	Values	Value Labels	Description
f180	learn children at home (Q52)	0	1510	0	Limitation followed Limitation ignored	If respondent named more than the allowed 5 qualities. if (v170 to v180=1) > 5 then f180=1
f67	why do people live in need - most/second important (Q12/Q13)	32	1510	0 1	consistent inconsistent	If most important aim is identical with next most important aim. if v67 = v68 then f67=1
f201	aims of this country - most/second important (Q60/Q61)	62	1510	0	consistent inconsistent	If most important reason is identical with second most important reason. if v201 = v202 then f201=1
f253	belong to geo- graphic group - most/second (Q69/Q70)	36	1510	0 1	consistent inconsistent	If geographical group that respondent states to belong to first of all is identical with secondly stated group. if v253 = v254 then f253=1
f25*	do you belong to: none (Q5a)	1510	1510	0	consistent	
	(253)	0	1510	1	inconsistent1	If respondent mentions at least one organisation and "none". if v25=1 and any of v10 to v24 and AT01=1 then f25=1
		0	1510	2	inconsistent2	If respondent does not know for at least one organization whether s/he belongs to it and mentions "none". if v25=1 and none of v10 to v24 and AT01=1 and any of v10 to v24 and AT01=8 then f25=2
		0	1510	3	inconsistent3	If respondent does not know for at least one organization whether s/he belongs to it and does not mention "none". if v25=2 and none of v10 to v24 and AT01=1 and any of v10 to v24 and AT01=8 then f25=3
		0	1510	4	inconsistent4	If respondent does not mention any organisation and does not mention "none". if v25=2 and all of v10 to v24 and AT01=2 then f25=4
		0	1510	5	inconsistent5	If respondent mentions at least one organization and does not know whether s/he belongs to "none". if v25=8 and any of v10 to v24 and AT01=1 then f25=5
		0	1510	6	inconsistent6	If respondent does not mention any organization and does not know whether s/he belongs to "none". if v25=8 and all of v10 to v24 and AT01=2 then f25=6
		0	1510	7	inconsistent7	If respondent mentions at least one organization and does not answer whether s/he belongs to "none". if v25=9 and any of v10 to v24 and AT01=1 then f25=7

		0	1510	8	inconsistent8	If respondent does not mention any organization and does not answer whether s/he belongs to "none". if v25=9 and all of v10 to v24 and AT01=2 then f25=8
f43*	do you work unpaid for: none (Q5b)	1510	1510	0	consistent	
		0	1510	1	inconsistent1	If respondent mentions at least one organisation and "none". if v43=1 and any of v28 to v42 and ATO2=1 then f43=1
		0	1510	2	inconsistent2	If respondent does not know for at least one organization whether s/he works for it and mentions "none". if v43=1 and none of v28 to v42 and AT02=1 and any of v28 to v42 and AT02=8 then f43=2
		0	1510	3	inconsistent3	If respondent does not know for at least one organization whether s/he works for it and does not mention "none". if v43=2 and none of v28 to v42 and ATO2=1 and any of v28 to v42
		0	1510	4	inconsistent4	and AT02=8 then f43=3 If respondent does not mention any organisation and does not mention "none". if v43=2 and all of v28 to v42 and AT02=2 then f43=4
		0	1510	5	inconsistent5	If respondent mentions at least one organization and does not know whether s/he works for "none". if v43=8 and any of v28 to v42 and ATO2=1 then f43=5
		0	1510	6	inconsistent6	If respondent does not mention any organization and does not know whether s/he works for "none". if v43=8 and all of v28 to v42 and ATO2=2 then f43=6
		0	1510	7	inconsistent7	If respondent mentions at least one organization and does not answer whether s/he works for "none". if v43=9 and any of v28 to v42 and ATO2=1 then f43=7
		0	1510	8	inconsistent8	If respondent does not mention any organization and does not answer whether s/he works for "none". if v43=9 and all of v28 to v42 and ATO2=2 then f43=8
f86**	important in a job: none of these (Q14)	1510	1510	0	consistent	
	(2)	0	1510	1	inconsistent1	If respondent mentions at least one aspect and "none". if v86=1 and any of v69 to v85 and AT08 to AT10=1 then f86=1
		0	1510	2	inconsistent2	If respondent does not know for at least one aspect whether it is important and mentions "none". if v86=1 and none of v69 to v85 and AT08 to AT10=1 and any of v69

						to v85 and AT08 to AT10=8 then f86=2
		0	1510	3	inconsistent3	If respondent does not to know for at least one aspect whether it is important and does not mention "none". if v86=2 and none of v69 to v85 and AT08 to AT10=1 and any of v69 to v85 and AT08 to AT10=8 then f86=3
		0	1510	4	inconsistent4	If respondent does not mention any aspect and does not mention "none". if v86=2 and all of v69 to v85 and AT08 to AT10=2 then f86=4
		0	1510	5	inconsistent5	If respondent mentions at least one aspect and does not know whether "none" is important. if v86=8 and any of v69 to v85 and AT08 to AT10=1 then f86=5
		0	1510	6	inconsistent6	If respondent does not mention any aspect and does not know whether "none" is important. if v86=8 and all of v69 to v85 and AT08 to AT10=2 then f86=6
		0	1510	7	inconsistent7	If respondent mentions at least one aspect and does not answer whether "none" is important. If v86=9 and any of v69 to v85 and AT08 to AT10=1 then f86=7
		0	1510	8	inconsistent8	If respondent does not mention any aspect and does not answer whether "none" is important. if v86=9 and all of v69 to v85 and ATO8 to AT10=2 then f86=8
f181	learn children at home: none (Q52)	1510	1510	0	consistent	
	nome. Home (202)	0	1510	1	inconsistent1	If respondent mentions at least one quality and "none". if v181=1 and any of v170 to v180=1 then f181=1
		0	1510	2	inconsistent2	If respondent does not know for at least one quality whether it is important and mentions "none". if v181=1 and none of v170 to v180=1 and any of v170 to v180=8 then f181=2
		0	1510	3	inconsistent3	If respondent does not know for at least one quality whether it is important and does not mention "none". If v181=2 and none of v170 to v180=1 and any of v170 to v180=8 then f181=3
		0	1510	4	inconsistent4	If respondent does not mention any quality and does not mention "none". If v181=2 and all of v170 to v180=2 then f181=4
		0	1510	5	inconsistent5	If respondent mentions at least one quality and does not know whether "none" is important.

					if v181=8 and any of v170 to v180=1 then f181=5
	0	1510	6	inconsistent6	If respondent does not mention any quality and does not know whether "none" is important. if v181=8 and all of v170 to v180=2 then f181=6
	0	1510	7	inconsistent7	If respondent mentions at least one quality and does not answer whether "none" is important. if v181=9 and any of v170 to v180=1 then f181=7
	0	1510	8	inconsistent8	If respondent does not mention any quality and does not answer whether "none" is important. if v181=9 and all of v170 to v180=2 then f181=8

Variable Documentation

GESIS StudyNo 4754, v1.1.0, http://dx.doi.org/doi:10.4232/1.10152

Variable, Label

Question Text (English Language)

StudyNo - GESIS Study Number

GESIS Study Number:

ZA4754

StudyNo

Value Label		Label	Missing Count	Percent	Valid Percent
	4754		1510	100.0	100.0
		Sum	1510	100.0	100.0
		Valid Cases	1510		

GESIS StudyNo 4754, v1.1.0, http://dx.doi.org/doi:10.4232/1.10152

version - GESIS Archive Version

GESIS Archive Version:

1.1.0 '2010-11-30'

version

Value	Label	Missing Count	Percent	Valid Percent
1.1.0 "2010-11-30"		1510	100.0	100.0
	Sum	1510	100.0	100.0
	Valid Cases	1510		

GESIS StudyNo 4754, v1.1.0, http://dx.doi.org/doi:10.4232/1.10152

id_cocas - unified respondent number

Unified respondent number:

Code comprises 12 digits: YYYY (year of wave), CCCC (ISO-Code for country), and NNNN (casenumber)

caseno - original respondent number

Original respondent number

intno - interviewer number

Q141

Interviewer number:

wave - wave

4th EVS Wave

wave

Value	Label	Missing Count	Percent	Valid Percent
2008	wave 2008	1510	100.0	100.0
	Sum	1510	100.0	100.0
	Valid Cases	1510		

year - survey year

Survey year: 2008

year

Value	Label	Missing Count	Percent	Valid Percent
2008	2008	1510	100.0	100.0
	Sum	1510	100.0	100.0
	Valid Cases	1510		

country - country code

Country/Sample

400 Austria

country

Value	Label	Missing Count	Percent	Valid Percent
400	Austria	1510	100.0	100.0
	Sum	1510	100.0	100.0
	Valid Cases	1510		

country1 - country code [with split ups]

Country/Sample:

With split-up for Germany (West and East Germany)

400 Austria

country1

Valu	e Label	Missing Count	Percent	Valid Percent
40	00 Austria	1510	100.0	100.0
	Sum	1510	100.0	100.0
	Valid Cases	1510		

c_abrv - country abbreviation

Country abbreviation

AT Austria

c_abrv

Value	Label	Missing Count	Percent	Valid Percent
AT		1510	100.0	100.0
	Sum	1510	100.0	100.0
	Valid Cases	1510		

c_abrv1 - country abbreviation [with split ups]

Country abbreviation with split-up for Germany (West and East Germany)

AT Austria

c_abrv1

Value	Label	Missing Count	Percent	Valid Percent
AT		1510	100.0	100.0
	Sum	1510	100.0	100.0
	Valid Cases	1510		

weight - weight

Weight by gender and age.

The weight is constructed on bases of sex and age categories. It adjusts the socio-structural characteristics in the sample to the distribution of gender and age of the universe–population.

Note:

For detailed information, see description of weight variable in the Variable Report.

cntry_y - country_year

Country/Year

Austria (2008)

cntry_y

Value	Label	Missing Count	Percent	Valid Percent
Austria (2008)		1510	100.0	100.0
	Sum	1510	100.0	100.0
	Valid Cases	1510		

cntry1_y - country_[with split ups]- year

Country/Year with split-up for Germany (West and East Germany)

Austria (2008)

cntry1_y

Value	Label	Missing Count	Percent	Valid Percent
Austria (2008) [with split ups]		1510	100.0	100.0
	Sum	1510	100.0	100.0
	Valid Cases	1510		

v1 - how important in your life: work (Q1A)

WE START WITH SOME QUESTIONS ABOUT LIFE IN GENERAL, LEISURE TIME ACTIVITIES AND WORK

Q1

<SHOW CARD 1>

Please say, for each of the following, how important it is in your life.

Q1.A Work

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very important
- 2 quite important
- 3 not important
- 4 not at all important

v1, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	1	0.1	
1	very important		810	53.6	53.7
2	quite important		533	35.3	35.3
3	not important		119	7.9	7.9
4	not at all important		46	3.0	3.1
	Sum		1510	100.0	100.0
	Valid Cases		1508		

v2 - how important in your life: family (Q1B)

WE START WITH SOME QUESTIONS ABOUT LIFE IN GENERAL, LEISURE TIME ACTIVITIES AND WORK

Q1

<SHOW CARD 1>

Please say, for each of the following, how important it is in your life.

Q1.B Family

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very important
- 2 quite important
- 3 not important
- 4 not at all important

v2, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	1	0.1	
1	very important		1187	78.6	78.6
2	quite important		266	17.6	17.6
3	not important		45	3.0	3.0
4	not at all important		12	0.8	0.8
	Sum		1510	100.1	100.0
	Valid Cases		1509		

v3 - how important in your life: friends and acquaintances (Q1C)

WE START WITH SOME QUESTIONS ABOUT LIFE IN GENERAL, LEISURE TIME ACTIVITIES AND WORK

Q1

<SHOW CARD 1>

Please say, for each of the following, how important it is in your life.

Q1.C Friends and acquaintances

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very important
- 2 quite important
- 3 not important
- 4 not at all important

v3, weighted by weight

Value	Label	Missing Count	Percent	Valid Percent
1	very important	843	55.8	55.8
2	quite important	564	37.4	37.4
3	not important	97	6.4	6.4
4	not at all important	6	0.4	0.4
	Sum	1510	100.0	100.0
	Valid Cases	1510		

v4 - how important in your life: leisure time (Q1D)

WE START WITH SOME QUESTIONS ABOUT LIFE IN GENERAL, LEISURE TIME ACTIVITIES AND WORK

Q1

<SHOW CARD 1>

Please say, for each of the following, how important it is in your life.

Q1.D Leisure time

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very important
- 2 quite important
- 3 not important
- 4 not at all important

v4, weighted by weight

Value	Lahal	Missing Count	Doroont	Valid Darsont
Value	Label	Missing Count	Percent	valid Percent
1	very important	665	44.0	44.0
2	quite important	673	44.6	44.6
3	not important	152	10.1	10.1
4	not at all important	20	1.3	1.3
	Sum	1510	100.0	100.0
	Valid Cases	1510		

v5 - how important in your life: politics (Q1E)

WE START WITH SOME QUESTIONS ABOUT LIFE IN GENERAL, LEISURE TIME ACTIVITIES AND WORK

Q1

<SHOW CARD 1>

Please say, for each of the following, how important it is in your life.

Q1.E Politics

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very important
- 2 quite important
- 3 not important
- 4 not at all important

v5, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	5	0.3	
1	very important		168	11.1	11.2
2	quite important		455	30.1	30.2
3	not important		632	41.9	42.0
4	not at all important		250	16.6	16.6
	Sum		1510	100.1	100.0
	Valid Cases		1504		

v6 - how important in your life: religion (Q1F)

WE START WITH SOME QUESTIONS ABOUT LIFE IN GENERAL, LEISURE TIME ACTIVITIES AND WORK

Q1

<SHOW CARD 1>

Please say, for each of the following, how important it is in your life.

Q1.F Religion

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very important
- 2 quite important
- 3 not important
- 4 not at all important

v6, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	11	0.7	
1	very important		275	18.2	18.3
2	quite important		434	28.7	28.9
3	not important		507	33.6	33.8
4	not at all important		284	18.8	18.9
	Sum		1510	100.1	100.0
	Valid Cases		1499		

v7 - how often discuss politics with friends (Q2)

Ω2

When you get together with your friends, would you say you discuss political matters frequently, occasionally or never?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 frequently
- 2 occasionally
- 3 never

v7, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	5	0.3	
1	frequently		264	17.5	17.5
2	occasionally		945	62.6	62.8
3	never		296	19.6	19.7
	Sum		1510	100.0	100.0
	Valid Cases		1505		

v8 - taking all things together how happy are you (Q3)

Q3

<SHOW CARD 3>

Taking all things together, would you say you are:

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very happy
- 2 quite happy
- 3 not very happy
- 4 not at all happy

v8, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	13	0.9	
1	very happy		434	28.7	29.0
2	quite happy		894	59.2	59.7
3	not very happy		146	9.7	9.7
4	not at all happy		24	1.6	1.6
	Sum		1510	100.1	100.0
	Valid Cases		1497		

v9 - describe your state of health these days (Q4)

Q4

<SHOW CARD 4>

All in all, how would you describe your state of health these days? Would you say it is

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very good
- 2 good
- 3 fair
- 4 poor
- 5 very poor

v9, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	1	0.1	
1	very good		436	28.9	28.9
2	good		669	44.3	44.4
3	fair		303	20.1	20.1
4	poor		83	5.5	5.5
5	very poor		17	1.1	1.1
	Sum		1510	99.9	100.0
	Valid Cases		1509		

v10 - do you belong to: welfare organisation (Q5aA)

Q5a

<SHOW CARD 5>

Please look carefully at the following list of voluntary organisations and activities and say ...

a) which, if any, do you belong to?

<CODE ALL MENTIONED UNDER (a)>

Q5a.A Social welfare services for elderly, handicapped or deprived people

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v10, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	7	0.5	
-1	dk	М	4	0.3	
1	mentioned		109	7.2	7.3
2	not mentioned		1390	92.1	92.7
	Sum		1510	100.0	100.0
	Valid Cases		1499		

v11 - do you belong to: religious organisation (Q5aB)

Q5a

<SHOW CARD 5>

Please look carefully at the following list of voluntary organisations and activities and say ...

a) which, if any, do you belong to?

<CODE ALL MENTIONED UNDER (a)>

Q5a.B Religious or church organisations

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v11, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	7	0.5	
-1	dk	М	7	0.5	
1	mentioned		201	13.3	13.4
2	not mentioned		1296	85.8	86.6
	Sum		1510	100.1	100.0
	Valid Cases		1496		

v12 - do you belong to: cultural activities (Q5aC)

Q5a

<SHOW CARD 5>

Please look carefully at the following list of voluntary organisations and activities and say ...

a) which, if any, do you belong to?

<CODE ALL MENTIONED UNDER (a)>

Q5a.C Education, arts, music or cultural activities

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v12, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	7	0.5	
-1	dk	М	6	0.4	
1	mentioned		130	8.6	8.7
2	not mentioned		1367	90.5	91.3
	Sum		1510	100.0	100.0
	Valid Cases		1498		

v13 - do you belong to: trade unions (Q5aD)

Q5a

<SHOW CARD 5>

Please look carefully at the following list of voluntary organisations and activities and say ...

a) which, if any, do you belong to?

<CODE ALL MENTIONED UNDER (a)>

Q5a.D Trade unions

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v13, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	8	0.5	
-1	dk	М	8	0.5	
1	mentioned		118	7.8	7.9
2	not mentioned		1377	91.2	92.1
	Sum		1510	100.1	100.0
	Valid Cases		1495		

v14 - do you belong to: political parties/groups (Q5aE)

Q5a

<SHOW CARD 5>

Please look carefully at the following list of voluntary organisations and activities and say ...

a) which, if any, do you belong to?

<CODE ALL MENTIONED UNDER (a)>

Q5a.E Political parties or groups

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v14, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	7	0.5	
-1	dk	М	8	0.5	
1	mentioned		98	6.5	6.6
2	not mentioned		1397	92.5	93.4
	Sum		1510	100.0	100.0
	Valid Cases		1495		

v15 - do you belong to: local community action (Q5aF)

Q5a

<SHOW CARD 5>

Please look carefully at the following list of voluntary organisations and activities and say ...

a) which, if any, do you belong to?

<CODE ALL MENTIONED UNDER (a)>

Q5a.F Local community action on issues like poverty, employment, housing, racial equality

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v15, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	8	0.5	
-1	dk	М	7	0.5	
1	mentioned		35	2.3	2.3
2	not mentioned		1461	96.8	97.7
	Sum		1510	100.1	100.0
	Valid Cases		1496		

v16 - do you belong to: 3w-development/human rights (Q5aG)

Q5a

<SHOW CARD 5>

Please look carefully at the following list of voluntary organisations and activities and say ...

a) which, if any, do you belong to?

<CODE ALL MENTIONED UNDER (a)>

Q5a.G Third world development or human rights

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v16, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	8	0.5	
-1	dk	М	7	0.5	
1	mentioned		48	3.2	3.2
2	not mentioned		1447	95.8	96.8
	Sum		1510	100.0	100.0
	Valid Cases		1495		

v17 - do you belong to: environment (Q5aH)

Q5a

<SHOW CARD 5>

Please look carefully at the following list of voluntary organisations and activities and say ...

a) which, if any, do you belong to?

<CODE ALL MENTIONED UNDER (a)>

Q5a.H Conservation, the environment, ecology, animal rights

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v17, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	8	0.5	
-1	dk	М	4	0.3	
1	mentioned		123	8.1	8.2
2	not mentioned		1375	91.1	91.8
	Sum		1510	100.0	100.0
	Valid Cases		1499		

v18 - do you belong to: professional associations (Q5al)

Q5a

<SHOW CARD 5>

Please look carefully at the following list of voluntary organisations and activities and say ...

a) which, if any, do you belong to?

<CODE ALL MENTIONED UNDER (a)>

Q5a.I Professional associations

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v18, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	8	0.5	
-1	dk	М	9	0.6	
1	mentioned		64	4.2	4.3
2	not mentioned		1429	94.6	95.7
	Sum		1510	100.0	100.0
	Valid Cases		1494		

v19 - do you belong to: youth work (Q5aJ)

Q5a

<SHOW CARD 5>

Please look carefully at the following list of voluntary organisations and activities and say ...

a) which, if any, do you belong to?

<CODE ALL MENTIONED UNDER (a)>

Q5a.J Youth work (e.g. scouts, guides, youth clubs etc.)

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v19, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	6	0.4	
-1	dk	М	7	0.5	
1	mentioned		48	3.2	3.2
2	not mentioned		1449	96.0	96.8
	Sum		1510	100.0	100.0
	Valid Cases		1497		

v20 - do you belong to: sports/recreation (Q5aK)

Q5a

<SHOW CARD 5>

Please look carefully at the following list of voluntary organisations and activities and say ...

a) which, if any, do you belong to?

<CODE ALL MENTIONED UNDER (a)>

Q5a.K Sports or recreation

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v20, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	5	0.3	
-1	dk	М	7	0.5	
1	mentioned		199	13.2	13.3
2	not mentioned		1299	86.0	86.7
	Sum		1510	100.0	100.0
	Valid Cases		1498		

v21 - do you belong to: womens groups (Q5aL)

Q5a

<SHOW CARD 5>

Please look carefully at the following list of voluntary organisations and activities and say ...

a) which, if any, do you belong to?

<CODE ALL MENTIONED UNDER (a)>

Q5a.L Women's groups

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v21, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	7	0.5	
-1	dk	М	6	0.4	
1	mentioned		65	4.3	4.3
2	not mentioned		1432	94.8	95.7
	Sum		1510	100.0	100.0
	Valid Cases		1497		

v22 - do you belong to: peace movement (Q5aM)

Q5a

<SHOW CARD 5>

Please look carefully at the following list of voluntary organisations and activities and say ...

a) which, if any, do you belong to?

<CODE ALL MENTIONED UNDER (a)>

Q5a.M Peace movement

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v22, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	8	0.5	
-1	dk	М	7	0.5	
1	mentioned		26	1.7	1.7
2	not mentioned		1469	97.3	98.3
	Sum		1510	100.0	100.0
	Valid Cases		1495		

v23 - do you belong to: voluntary health organisations (Q5aN)

Q5a

<SHOW CARD 5>

Please look carefully at the following list of voluntary organisations and activities and say ...

a) which, if any, do you belong to?

<CODE ALL MENTIONED UNDER (a)>

Q5a.N Voluntary organisations concerned with health

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v23, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	8	0.5	
-1	dk	М	7	0.5	
1	mentioned		65	4.3	4.3
2	not mentioned		1430	94.7	95.7
	Sum		1510	100.0	100.0
	Valid Cases		1496		

AT01 - do you belong to: fire fighters (Q5a_O)

Q5a

<SHOW CARD 5>

Please look carefully at the following list of voluntary organisations and activities and say ...

a) which, if any, do you belong to?

<CODE ALL MENTIONED UNDER (a)>

Q5aO_at Fire fighters

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT01, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	6	0.4	
-1	dk	М	7	0.5	
1	mentioned		77	5.1	5.1
2	not mentioned		1420	94.0	94.9
	Sum		1510	100.0	100.0
	Valid Cases		1497		

v24 - do you belong to: other groups (Q5aO)

Q5a

<SHOW CARD 5>

Please look carefully at the following list of voluntary organisations and activities and say ...

a) which, if any, do you belong to?

<CODE ALL MENTIONED UNDER (a)>

Q5a.O Other groups

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item) and item numbering (P instead of O).

v24, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	9	0.6	
-1	dk	М	12	0.8	
1	mentioned		93	6.2	6.3
2	not mentioned		1395	92.4	93.8
	Sum		1510	99.9	100.0
	Valid Cases		1488		

f25 - flag variable: do you belong to: none (Q5a)

Flag variable:

Indicates inconsistencies for the variables v10-v25.

Inconsistent 1

If respondent mentiones at least one organisation and "none".

if v25=1 and any of v10 to v24=1 then f25=1

Inconsistent 2

If respondent does not know for at least one organization whether s/he belongs to it and mentiones "none".

if v25=1 and none of v10 to v24=1 and any of v10 to v24=8 then f25=2

Inconsistent 3

If respondent does not know for at least one organization whether s/he belongs to it and does not mention "none".

if v25=2 and none of v10 to v24=1 and any of v10 to v24=8 then f25=3

Inconsistent 4

If respondent does not mention any organisation and does not mention "none".

if v25=2 and all of v10 to v24=2 then f25=4

Inconsistent 5

If respondent mentions at least one organization and does not know whether s/he belongs to "none".

if v25=8 and any of v10 to v24=1 then f25=5

Inconsistent 6

If respondent does not mention any organization and does not know whether s/he belongs to "none".

if v25=8 and all of v10 to v24=2 then f25=6

Inconsistent 7

If respondent mentions at least one organization and does not answer whether s/he belongs to "none".

if v25=9 and any of v10 to v24=1 then f25=7

Inconsistent 8

If respondent does not mention any organization and does not answer whether s/he belongs to "none".

if v25=9 and all of v10 to v24=2 then f25=8

Note:

For detailed information, see overview flag variables.

f25

Value	Label	Missing Count	Percent	Valid Percent
0	consistent	1510	100.0	100.0
	Sum	1510	100.0	100.0
	Valid Cases	1510		

v25 - do you belong to: none (Q5a)

Q5a

<SHOW CARD 5>

Please look carefully at the following list of voluntary organisations and activities and say ...

a) which, if any, do you belong to?

<CODE ALL MENTIONED UNDER (a)>

None (spontaneous)

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v25, weighted by weight

Value Label	Missing Count	Percent	Valid Percent
1 mentioned	880	58.3	58.3
2 not mentioned	630	41.7	41.7
Sum	1510	100.0	100.0
Valid Cases	1510		

v28 - do you work unpaid for: welfare organisation (Q5bA)

Q5b

<SHOW CARD 5>

Please look carefully at the following list of voluntary organisations and activities and say

b) which, if any, are you currently doing unpaid voluntary work for?

<CODE ALL MENTIONED UNDER (b)>

Q5b.A Social welfare services for elderly, handicapped or deprived people

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v28, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	6	0.4	
1	mentioned		80	5.3	5.3
2	not mentioned		1420	94.0	94.7
	Sum		1510	100.0	100.0
	Valid Cases		1500		

v29 - do you work unpaid for: religious organisation (Q5bB)

Q5b

<SHOW CARD 5>

Please look carefully at the following list of voluntary organisations and activities and say

b) which, if any, are you currently doing unpaid voluntary work for?

<CODE ALL MENTIONED UNDER (b)>

Q5b.B Religious or church organisations

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v29, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	6	0.4	
1	mentioned		115	7.6	7.7
2	not mentioned		1385	91.7	92.3
	Sum		1510	99.9	100.0
	Valid Cases		1500		

v30 - do you work unpaid for: cultural activities (Q5bC)

Q5b

<SHOW CARD 5>

Please look carefully at the following list of voluntary organisations and activities and say

b) which, if any, are you currently doing unpaid voluntary work for?

<CODE ALL MENTIONED UNDER (b)>

Q5b.C Education, arts, music or cultural activities

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v30, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	6	0.4	
1	mentioned		84	5.6	5.6
2	not mentioned		1416	93.8	94.4
	Sum		1510	99.9	100.0
	Valid Cases		1500		

v31 - do you work unpaid for: trade unions (Q5bD)

Q5b

<SHOW CARD 5>

Please look carefully at the following list of voluntary organisations and activities and say

b) which, if any, are you currently doing unpaid voluntary work for?

<CODE ALL MENTIONED UNDER (b)>

Q5b.D Trade unions

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v31, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	8	0.5	
1	mentioned		27	1.8	1.8
2	not mentioned		1471	97.4	98.2
	Sum		1510	100.0	100.0
	Valid Cases		1498		

v32 - do you work unpaid for: political parties/groups (Q5bE)

Q5b

<SHOW CARD 5>

Please look carefully at the following list of voluntary organisations and activities and say

b) which, if any, are you currently doing unpaid voluntary work for?

<CODE ALL MENTIONED UNDER (b)>

Q5b.E Political parties or groups

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v32, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	6	0.4	
1	mentioned		41	2.7	2.7
2	not mentioned		1459	96.6	97.3
	Sum		1510	100.0	100.0
	Valid Cases		1499		

v33 - do you work unpaid for: local community action (Q5bF)

Q5b

<SHOW CARD 5>

Please look carefully at the following list of voluntary organisations and activities and say

b) which, if any, are you currently doing unpaid voluntary work for?

<CODE ALL MENTIONED UNDER (b)>

Q5b.F Local community action on issues like poverty, employment, housing, racial equality

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v33, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	6	0.4	
1	mentioned		17	1.1	1.1
2	not mentioned		1482	98.1	98.9
	Sum		1510	99.9	100.0
	Valid Cases		1499		

v34 - do you work unpaid for: 3w-development/human rights (Q5bG)

Q5b

<SHOW CARD 5>

Please look carefully at the following list of voluntary organisations and activities and say

b) which, if any, are you currently doing unpaid voluntary work for?

<CODE ALL MENTIONED UNDER (b)>

Q5b.G Third world development or human rights

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v34, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	6	0.4	
1	mentioned		18	1.2	1.2
2	not mentioned		1482	98.1	98.8
	Sum		1510	100.0	100.0
	Valid Cases		1499		

v35 - do you work unpaid for: environment (Q5bH)

Q5b

<SHOW CARD 5>

Please look carefully at the following list of voluntary organisations and activities and say

b) which, if any, are you currently doing unpaid voluntary work for?

<CODE ALL MENTIONED UNDER (b)>

Q5b.H Conservation, the environment, ecology, animal rights

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v35, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	6	0.4	
1	mentioned		43	2.8	2.9
2	not mentioned		1458	96.6	97.1
	Sum		1510	100.1	100.0
	Valid Cases		1500		

v36 - do you work unpaid for: professional associations (Q5bI)

Q5b

<SHOW CARD 5>

Please look carefully at the following list of voluntary organisations and activities and say

b) which, if any, are you currently doing unpaid voluntary work for?

<CODE ALL MENTIONED UNDER (b)>

Q5b.I Professional associations

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v36, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	8	0.5	
1	mentioned		19	1.3	1.3
2	not mentioned		1479	97.9	98.7
	Sum		1510	100.0	100.0
	Valid Cases		1498		

v37 - do you work unpaid for: youth work (Q5bJ)

Q5b

<SHOW CARD 5>

Please look carefully at the following list of voluntary organisations and activities and say

b) which, if any, are you currently doing unpaid voluntary work for?

<CODE ALL MENTIONED UNDER (b)>

Q5b.J Youth work (e.g. scouts, guides, youth clubs etc.)

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v37, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	6	0.4	
1	mentioned		30	2.0	2.0
2	not mentioned		1470	97.4	98.0
	Sum		1510	100.0	100.0
	Valid Cases		1500		

v38 - do you work unpaid for: sports/recreation (Q5bK)

Q5b

<SHOW CARD 5>

Please look carefully at the following list of voluntary organisations and activities and say

b) which, if any, are you currently doing unpaid voluntary work for?

<CODE ALL MENTIONED UNDER (b)>

Q5b.K Sports or recreation

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v38, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	5	0.3	
1	mentioned		99	6.6	6.6
2	not mentioned		1402	92.8	93.4
	Sum		1510	99.9	100.0
	Valid Cases		1501		

v39 - do you work unpaid for: womens groups (Q5bL)

Q5b

<SHOW CARD 5>

Please look carefully at the following list of voluntary organisations and activities and say

b) which, if any, are you currently doing unpaid voluntary work for?

<CODE ALL MENTIONED UNDER (b)>

Q5b.L Women's groups

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v39, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	6	0.4	
1	mentioned		39	2.6	2.6
2	not mentioned		1461	96.8	97.4
	Sum		1510	100.0	100.0
	Valid Cases		1500		

v40 - do you work unpaid for: peace movement (Q5bM)

Q5b

<SHOW CARD 5>

Please look carefully at the following list of voluntary organisations and activities and say

b) which, if any, are you currently doing unpaid voluntary work for?

<CODE ALL MENTIONED UNDER (b)>

Q5b.M Peace movement

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v40, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	8	0.5	
1	mentioned		4	0.3	0.3
2	not mentioned		1494	98.9	99.7
	Sum		1510	100.0	100.0
	Valid Cases		1498		

v41 - do you work unpaid for: voluntary health organisations (Q5bN)

Q5b

<SHOW CARD 5>

Please look carefully at the following list of voluntary organisations and activities and say

b) which, if any, are you currently doing unpaid voluntary work for?

<CODE ALL MENTIONED UNDER (b)>

Q5b.N Voluntary organisations concerned with health

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v41, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	5	0.3	
1	mentioned		32	2.1	2.1
2	not mentioned		1468	97.2	97.9
	Sum		1510	99.9	100.0
	Valid Cases		1500		

AT02 - do you work unpaid for: fire fighters (Q5b_O)

Q5b

<SHOW CARD 5>

Please look carefully at the following list of voluntary organisations and activities and say

b) which, if any, are you currently doing unpaid voluntary work for?

<CODE ALL MENTIONED UNDER (b)>

Q5bO_at Fire fighters

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT02, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	6	0.4	
1	mentioned		57	3.8	3.8
2	not mentioned		1443	95.6	96.2
	Sum		1510	99.9	100.0
	Valid Cases		1500		

v42 - do you work unpaid for: other groups (Q5bO)

Q5b

<SHOW CARD 5>

Please look carefully at the following list of voluntary organisations and activities and say

b) which, if any, are you currently doing unpaid voluntary work for?

<CODE ALL MENTIONED UNDER (b)>

Q5b.O Other groups

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item) and item numbering (P instead of O).

v42, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	6	0.4	
-1	dk	М	7	0.5	
1	mentioned		62	4.1	4.1
2	not mentioned		1436	95.1	95.9
	Sum		1510	100.1	100.0
	Valid Cases		1497		

f43 - flag variable: do you work unpaid for: none (Q5b)

Flag variable:

Indicates inconsistencies for the variables v28-v43.

Inconsistent 1

If respondent mentiones at least one organisation and "none".

if v43=1 and any of v28 to v42=1 then f43=1

Inconsistent 2

If respondent does not know for at least one organization whether s/he works for it and mentiones "none".

if v43=1 and none of v28 to v42=1 and any of v28 to v42=8 then f43=2

Inconsistent 3

If respondent does not know for at least one organization whether s/he works for it and does not mention "none".

if v43=2 and none of v28 to v42=1 and any of v28 to v42=8 then f43=3

Inconsistent 4

If respondent does not mention any organisation and does not mention "none".

if v43=2 and all of v28 to v42=2 then f43=4

Inconsistent 5

If respondent mentions at least one organization and does not know whether s/he works for "none".

if v43=8 and any of v28 to v42=1 then f43=5

Inconsistent 6

If respondent does not mention any organization and does not know whether s/he works for "none".

if v43=8 and all of v28 to v42=2 then f43=6

Inconsistent 7

If respondent mentions at least one organization and does not answer whether s/he works for "none".

if v43=9 and any of v28 to v42=1 then f43=7

Inconsistent 8

If respondent does not mention any organization and does not answer whether s/he works for "none".

if v43=9 and all of v28 to v42=2 then f43=8

Note:

For detailed information, see overview flag variables.

f43

Value	Label	Missing Count	Percent	Valid Percent
0	consistent	1510	100.0	100.0
	Sum	1510	100.0	100.0
	Valid Cases	1510		

v43 - do you work unpaid for: none (Q5b)

Q5b

<SHOW CARD 5>

Please look carefully at the following list of voluntary organisations and activities and say

b) which, if any, are you currently doing unpaid voluntary work for?

<CODE ALL MENTIONED UNDER (b)>

None (spontaneous)

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v43, weighted by weight

Value Label	Missing Count	Percent	Valid Percent
1 mentioned	1090	72.2	72.2
2 not mentioned	420	27.8	27.8
Sum	1510	100.0	100.0
Valid Cases	1510		

AT03 - reasons to work as a volunteer: sense of duty, moral obligation (Q5_at)

Q5_at

<SHOW CARD 5_AT>

There are various reasons to participate in voluntary organisations. I would like to read you a couple of different statements concerning this topic. Can you please tell me, how important these motivations are for you personally or whether these motivations would be a reason for you to work as a volunteer

<CODE ONE ANSWER PER QUESTION>

Q5A_at Sens of duty, moral obligation

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 unimportant
- 2
- 3
- 4
- 5 very important

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT03, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	19	1.3	
-1	dk	М	80	5.3	
1	unimportant		206	13.6	14.6
2			213	14.1	15.1
3			324	21.5	23.0
4			368	24.4	26.1
5	very important		300	19.9	21.3
	Sum		1510	100.0	100.0
	Valid Cases		1411		

AT04 - reasons to work as a volunteer: religious conviction (Q5_at)

Q5_at

<SHOW CARD 5_AT>

There are various reasons to participate in voluntary organisations. I would like to read you a couple of different statements concerning this topic. Can you please tell me, how important these motivations are for you personally or whether these motivations would be a reason for you to work as a volunteer

<CODE ONE ANSWER PER QUESTION>

Q5B_at Religious conviction

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 unimportant
- 2
- 3
- 4
- 5 very important

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT04, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	20	1.3	
-1	dk	М	82	5.4	
1	unimportant		442	29.3	31.3
2			316	20.9	22.4
3			295	19.5	20.9
4			194	12.8	13.8
5	very important		163	10.8	11.6
	Sum		1510	100.1	100.0
	Valid Cases		1409		

AT05 - reasons to work as a volunteer: to cause social or political changes (Q5_at)

Q5_at

<SHOW CARD 5_AT>

There are various reasons to participate in voluntary organisations. I would like to read you a couple of different statements concerning this topic. Can you please tell me, how important these motivations are for you personally or whether these motivations would be a reason for you to work as a volunteer

<CODE ONE ANSWER PER QUESTION>

Q5C_at To cause social or political changes

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 unimportant
- 2
- 3
- 4
- 5 very important

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT05, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	21	1.4	
-1	dk	М	78	5.2	
1	unimportant		268	17.7	19.0
2			211	14.0	15.0
3			421	27.9	29.8
4			322	21.3	22.8
5	very important		189	12.5	13.4
	Sum		1510	100.0	100.0
	Valid Cases		1412		

AT06 - reasons to work as a volunteer: to learn something new and make useful experiences (Q5_at)

Q5_at

<SHOW CARD 5_AT>

There are various reasons to participate in voluntary organisations. I would like to read you a couple of different statements concerning this topic. Can you please tell me, how important these motivations are for you personally or whether these motivations would be a reason for you to work as a volunteer

<CODE ONE ANSWER PER QUESTION>

Q5D_at To learn something new and make useful experiences

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 unimportant
- 2
- 3
- 4
- 5 very important

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT06, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	20	1.3	
-1	dk	М	73	4.8	
1	unimportant		182	12.1	12.8
2			195	12.9	13.8
3			347	23.0	24.5
4			422	27.9	29.8
5	very important		271	17.9	19.1
	Sum		1510	100.0	100.0
	Valid Cases		1418		

AT07 - reasons to work as a volunteer: because I want to do something for others (Q5_at)

Q5_at

<SHOW CARD 5_AT>

There are various reasons to participate in voluntary organisations. I would like to read you a couple of different statements concerning this topic. Can you please tell me, how important these motivations are for you personally or whether these motivations would be a reason for you to work as a volunteer

<CODE ONE ANSWER PER QUESTION>

Q5E_at Because I want to do something for others

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 unimportant
- 2
- 3
- 4
- 5 very important

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT07, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	19	1.3	
-1	dk	М	70	4.6	
1	unimportant		128	8.5	9.0
2			166	11.0	11.7
3			284	18.8	20.0
4			396	26.2	27.8
5	very important		448	29.7	31.5
	Sum		1510	100.1	100.0
	Valid Cases		1422		

v46 - dont like as neighbours: people with criminal record (Q6A)

06

<SHOW CARD 6 - CODE AN ANSWER FOR EACH>

On this list are various groups of people. Could you please sort out any that you would not like to have as neighbours?

Q6.A People with a criminal record

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

v46, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	5	0.3	
-1	dk	М	53	3.5	
1	mentioned		662	43.8	45.6
2	not mentioned		791	52.4	54.4
	Sum		1510	100.1	100.0
	Valid Cases		1452		

v47 - dont like as neighbours: people of different race (Q6B)

OF

<SHOW CARD 6 - CODE AN ANSWER FOR EACH>

On this list are various groups of people. Could you please sort out any that you would not like to have as neighbours?

Q6.B People of a different race

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

v47, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	7	0.5	
-1	dk	М	28	1.9	
1	mentioned		263	17.4	17.8
2	not mentioned		1212	80.3	82.2
	Sum		1510	100.0	100.0
	Valid Cases		1475		

v48 - dont like as neighbours: left wing extremists (Q6C)

06

<SHOW CARD 6 - CODE AN ANSWER FOR EACH>

On this list are various groups of people. Could you please sort out any that you would not like to have as neighbours?

Q6.C Left wing extremists

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

v48, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	5	0.3	
-1	dk	М	46	3.0	
1	mentioned		725	48.0	49.7
2	not mentioned		734	48.6	50.3
	Sum		1510	100.0	100.0
	Valid Cases		1459		

v49 - dont like as neighbours: heavy drinkers (Q6D)

Q6

<SHOW CARD 6 - CODE AN ANSWER FOR EACH>

On this list are various groups of people. Could you please sort out any that you would not like to have as neighbours?

Q6.D Heavy drinkers

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

v49, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	5	0.3	
-1	dk	М	16	1.1	
1	mentioned		901	59.7	60.5
2	not mentioned		589	39.0	39.5
	Sum		1510	100.1	100.0
	Valid Cases		1490		

v50 - dont like as neighbours: right wing extremists (Q6E)

OF

<SHOW CARD 6 - CODE AN ANSWER FOR EACH>

On this list are various groups of people. Could you please sort out any that you would not like to have as neighbours?

Q6.E Right wing extremists

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

v50, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	5	0.3	
-1	dk	М	31	2.1	
1	mentioned		929	61.5	63.0
2	not mentioned		545	36.1	37.0
	Sum		1510	100.0	100.0
	Valid Cases		1474		

v51 - dont like as neighbours: large families (Q6F)

06

<SHOW CARD 6 - CODE AN ANSWER FOR EACH>

On this list are various groups of people. Could you please sort out any that you would not like to have as neighbours?

Q6.F People with large families

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

v51, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	7	0.5	
-1	dk	М	24	1.6	
1	mentioned		233	15.4	15.8
2	not mentioned		1246	82.5	84.2
	Sum		1510	100.0	100.0
	Valid Cases		1479		

v52 - dont like as neighbours: emotionally unstable people (Q6G)

OF

<SHOW CARD 6 - CODE AN ANSWER FOR EACH>

On this list are various groups of people. Could you please sort out any that you would not like to have as neighbours?

Q6.G Emotionally unstable people

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

v52, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	5	0.3	
-1	dk	М	43	2.8	
1	mentioned		539	35.7	36.9
2	not mentioned		923	61.1	63.1
	Sum		1510	100.0	100.0
	Valid Cases		1462		

v53 - dont like as neighbours: muslims (Q6H)

Q6

<SHOW CARD 6 - CODE AN ANSWER FOR EACH>

On this list are various groups of people. Could you please sort out any that you would not like to have as neighbours?

Q6.H Muslims

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

v53, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	7	0.5	
-1	dk	М	37	2.5	
1	mentioned		459	30.4	31.3
2	not mentioned		1007	66.7	68.7
	Sum		1510	100.0	100.0
	Valid Cases		1466		

v54 - dont like as neighbours: immigrants/foreign workers (Q6I)

OF

<SHOW CARD 6 - CODE AN ANSWER FOR EACH>

On this list are various groups of people. Could you please sort out any that you would not like to have as neighbours?

Q6.I Immigrants/foreign workers

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

v54, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	6	0.4	
-1	dk	М	39	2.6	
1	mentioned		345	22.8	23.5
2	not mentioned		1121	74.2	76.5
	Sum		1510	100.1	100.0
	Valid Cases		1466		

v55 - dont like as neighbours: people with AIDS (Q6J)

OF

<SHOW CARD 6 - CODE AN ANSWER FOR EACH>

On this list are various groups of people. Could you please sort out any that you would not like to have as neighbours?

Q6.J People who have AIDS

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

v55, weighted by weight

١	Value	Label	Missing	Count	Percent	Valid Percent
	-2	na	М	5	0.3	
	-1	dk	М	29	1.9	
	1	mentioned		385	25.5	26.1
	2	not mentioned		1091	72.3	73.9
		Sum		1510	100.0	100.0
		Valid Cases		1476		

v56 - dont like as neighbours: drug addicts (Q6K)

06

<SHOW CARD 6 - CODE AN ANSWER FOR EACH>

On this list are various groups of people. Could you please sort out any that you would not like to have as neighbours?

Q6.K Drug addicts

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

v56, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	M	19	1.3	
1	mentioned		960	63.6	64.6
2	not mentioned		527	34.9	35.4
	Sum		1510	100.0	100.0
	Valid Cases		1487		

v57 - dont like as neighbours: homosexuals (Q6L)

06

<SHOW CARD 6 - CODE AN ANSWER FOR EACH>

On this list are various groups of people. Could you please sort out any that you would not like to have as neighbours?

Q6.L Homosexuals

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

v57, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	7	0.5	
-1	dk	М	21	1.4	
1	mentioned		360	23.8	24.3
2	not mentioned		1122	74.3	75.7
	Sum		1510	100.0	100.0
	Valid Cases		1482		

v58 - dont like as neighbours: jews (Q6M)

OF

<SHOW CARD 6 - CODE AN ANSWER FOR EACH>

On this list are various groups of people. Could you please sort out any that you would not like to have as neighbours?

Q6.M Jews

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

v58, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	9	0.6	
-1	dk	M	30	2.0	
1	mentioned		255	16.9	17.3
2	not mentioned		1216	80.5	82.7
	Sum		1510	100.0	100.0
	Valid Cases		1471		

v59 - dont like as neighbours: gypsies (Q6N)

06

<SHOW CARD 6 - CODE AN ANSWER FOR EACH>

On this list are various groups of people. Could you please sort out any that you would not like to have as neighbours?

Q6.N Gypsies

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

v59, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	6	0.4	
-1	dk	М	54	3.6	
1	mentioned		464	30.7	32.0
2	not mentioned		986	65.3	68.0
	Sum		1510	100.0	100.0
	Valid Cases		1449		

v60 - dont like as neighbours: christians (Q6O)

Derivation:

Question not implemented in Field Questionnaire. Computed and coded as -4 'question not asked'.

v60, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-4	question not asked	М	1510	100.0	
	Sum		1510	100.0	0.0
	Valid Cases		0		

v62 - people can be trusted/cant be too careful (Q7)

Ω7

Generally speaking, would you say that most people can be trusted or that you can't be too careful in dealing with people?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 most people can be trusted
- 2 can't be too careful

v62, weighted by weight

Value Label		Missing	Count	Percent	Valid Percent
-2 na		М	1	0.1	
-1 dk		М	57	3.8	
1 most people can b	e trusted		528	35.0	36.4
2 cannot be too care	ful		924	61.2	63.6
Sum			1510	100.0	100.0
Valid Cases			1452		

v63 - most people try to take advantage of you or try to be fair (Q8)

Q8

<SHOW CARD 8>

Using this card, do you think that most people would try to take advantage of you if they got the chance, or would they try to be fair? How would you place your view on this scale?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 most people would try to take advantage of me

2

3 4

5

6

7

8 9

10 most people would try to be fair

v63, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	25	1.7	
1	most people try to take advantage of me		94	6.2	6.3
2			20	1.3	1.3
3			117	7.7	7.9
4			75	5.0	5.1
5			331	21.9	22.3
6			138	9.1	9.3
7			209	13.8	14.1
8			278	18.4	18.7
9			78	5.2	5.3
10	most people try to be fair		145	9.6	9.8
	Sum		1510	100.1	100.0
	Valid Cases		1485		

v64 - most of the time people try to be helpful or mostly looking out for themselves (Q9)

Q9

<SHOW CARD 9>

Would you say that most of the time people try to be helpful or that they are mostly looking out for themselves? Please use this card.

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 people mostly look out for themselves

2

4

5

6

7 8

9

10 people mostly try to be helpful

v64, weighted by weight

Value	Label	Missing Count	Percent	Valid Percent
-1	dk	M 8	0.5	
1	people mostly look out for themselves	227	15.0	15.1
2		54	3.6	3.6
3		215	14.2	14.3
4		125	8.3	8.3
5		331	21.9	22.0
6		155	10.3	10.3
7		171	11.3	11.4
8		129	8.5	8.6
9		24	1.6	1.6
10	people mostly try to be helpful	72	4.8	4.8
	Sum	1510	100.1	100.0
	Valid Cases	1502		

v65 - how much control over your life (Q10)

Q10

<SHOW CARD 10>

Some people feel they have completely free choice and control over their lives, and other people feel that what they do has no real effect on what happens to them. Please use the scale to indicate how much freedom of choice and control you feel you have over the way your life turns out?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 none at all

2

3

4

5

6

7 8

9

10 a great deal

v65, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	17	1.1	
1	none at all		35	2.3	2.3
2			10	0.7	0.7
3			62	4.1	4.1
4			69	4.6	4.6
5			263	17.4	17.6
6			155	10.3	10.4
7			217	14.4	14.5
8			287	19.0	19.2
9			109	7.2	7.3
10	a great deal		287	19.0	19.2
	Sum		1510	100.1	100.0
	Valid Cases		1493		

v66 - how satisfied are you with your life (Q11)

Q11

<SHOW CARD 11>

All things considered, how satisfied are you with your life as a whole these days? Please use this card to help with your answer.

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 dissatisfied

2

3

4

5

6

7 8

9

10 satisfied

v66, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	1	0.1	
1	dissatisfied		30	2.0	2.0
2			12	0.8	0.8
3			42	2.8	2.8
4			47	3.1	3.1
5			138	9.1	9.2
6			113	7.5	7.5
7			204	13.5	13.5
8			392	26.0	26.0
9			209	13.8	13.9
10	satisfied		321	21.3	21.3
	Sum		1510	99.9	100.0
	Valid Cases		1509		

f67 - flag variable: why do people live in need - most/second important (Q12/Q13)

Flag variable:

Indicates inconsistencies for the variables v67/v68.

If most important aim is identical with next most important aim.

if v67 = v68 then f67=1

0 consistent

1 inconsistent

Note:

For detailed information, see overview flag variables.

f67

١	Value	Label	Missing Count	Percent	Valid Percent
	0	consistent	1478	97.9	97.9
	1	inconsistent	32	2.1	2.1
		Sum	1510	100.0	100.0
		Valid Cases	1510		

v67 - why do people live in need-most important (Q12)

Q12

<SHOW CARD 12>

Why are there people in this country who live in need? Here are four possible reasons. Which one reason do you consider to be most important?

<CODE ONE UNDER (Q12) BELOW>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 because they are unlucky
- 2 because of laziness and lack of willpower
- 3 because of injustice in our society
- 4 it's an inevitable part of modern progress
- 5 none of these (spontaneous)

v67, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	38	2.5	
1	unlucky		176	11.7	11.9
2	laziness or lack of willpower		417	27.6	28.3
3	injustice in society		522	34.6	35.4
4	part modern progress		306	20.3	20.8
5	none of these		52	3.4	3.5
	Sum		1510	100.1	100.0
	Valid Cases		1472		

v68 - why do people live in need-2nd most important (Q13)

[Q12: Why are there people in this country who live in need? Which of four possible reasons is most important?]

Q13

<SHOW CARD 12>

And which reason do you consider to be the second most important?

<CODE ONE UNDER (Q13) BELOW>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 because they are unlucky
- 2 because of laziness and lack of willpower
- 3 because of injustice in our society
- 4 it's an inevitable part of modern progress
- 5 none of these (spontaneous)

v68, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	2	0.1	
-1	dk	М	41	2.7	
1	unlucky		348	23.0	23.7
2	laziness or lack of willpower		336	22.3	22.9
3	injustice in society		370	24.5	25.2
4	part modern progress		343	22.7	23.4
5	none of these		70	4.6	4.8
	Sum		1510	100.0	100.0
	Valid Cases		1467		

v69 - important in a job: good pay (Q14A)

Q14

<SHOW CARD 14 - CODE ALL MENTIONED>

Here are some aspects of a job that people say are important. Please look at them and tell me which ones you personally think are important in a job?

Q14.A Good pay

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific items). Error in show card: Items S and T have the same label. Item T should be "Ein Beruf, der anerkannt und geachtet wird".

v69, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	2	0.1	
-1	dk	М	2	0.1	
1	mentioned		1018	67.4	67.6
2	not mentioned		488	32.3	32.4
	Sum		1510	100.0	100.0
	Valid Cases		1506		

v70 - important in a job: pleasant people (Q14B)

Q14

<SHOW CARD 14 - CODE ALL MENTIONED>

Here are some aspects of a job that people say are important. Please look at them and tell me which ones you personally think are important in a job?

Q14.B Pleasant people to work with

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific items). Error in show card: Items S and T have the same label. Item T should be "Ein Beruf, der anerkannt und geachtet wird".

v70, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	1	0.1	
1	mentioned		1012	67.0	67.2
2	not mentioned		493	32.6	32.8
	Sum		1510	100.0	100.0
	Valid Cases		1505		

v71 - important in a job: not too much pressure (Q14C)

Q14

<SHOW CARD 14 - CODE ALL MENTIONED>

Here are some aspects of a job that people say are important. Please look at them and tell me which ones you personally think are important in a job?

Q14.C Not too much pressure

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific items). Error in show card: Items S and T have the same label. Item T should be "Ein Beruf, der anerkannt und geachtet wird".

v71, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	5	0.3	
-1	dk	М	9	0.6	
1	mentioned		449	29.7	30.0
2	not mentioned		1047	69.3	70.0
	Sum		1510	100.0	100.0
	Valid Cases		1496		

v72 - important in a job: job security (Q14D)

Q14

<SHOW CARD 14 - CODE ALL MENTIONED>

Here are some aspects of a job that people say are important. Please look at them and tell me which ones you personally think are important in a job?

Q14.D Good job security

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific items). Error in show card: Items S and T have the same label. Item T should be "Ein Beruf, der anerkannt und geachtet wird".

v72, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	5	0.3	
1	mentioned		1080	71.5	72.0
2	not mentioned		421	27.9	28.0
	Sum		1510	100.0	100.0
	Valid Cases		1501		

v73 - important in a job: good hours (Q14E)

Q14

<SHOW CARD 14 - CODE ALL MENTIONED>

Here are some aspects of a job that people say are important. Please look at them and tell me which ones you personally think are important in a job?

Q14.E Good hours

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific items). Error in show card: Items S and T have the same label. Item T should be "Ein Beruf, der anerkannt und geachtet wird".

v73, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	5	0.3	
1	mentioned		667	44.2	44.4
2	not mentioned		834	55.2	55.6
	Sum		1510	100.0	100.0
	Valid Cases		1501		

v74 - important in a job: use initiative (Q14F)

Q14

<SHOW CARD 14 - CODE ALL MENTIONED>

Here are some aspects of a job that people say are important. Please look at them and tell me which ones you personally think are important in a job?

Q14.F An opportunity to use initiative

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific items). Error in show card: Items S and T have the same label. Item T should be "Ein Beruf, der anerkannt und geachtet wird".

v74, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	5	0.3	
-1	dk	М	13	0.9	
1	mentioned		662	43.8	44.4
2	not mentioned		830	55.0	55.6
	Sum		1510	100.0	100.0
	Valid Cases		1492		

v75 - important in a job: useful for society (Q14G)

Q14

<SHOW CARD 14 - CODE ALL MENTIONED>

Here are some aspects of a job that people say are important. Please look at them and tell me which ones you personally think are important in a job?

Q14.G A useful job for society

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific items). Error in show card: Items S and T have the same label. Item T should be "Ein Beruf, der anerkannt und geachtet wird".

v75, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	11	0.7	
1	mentioned		535	35.4	35.8
2	not mentioned		960	63.6	64.2
	Sum		1510	100.0	100.0
	Valid Cases		1495		

v76 - important in a job: generous holidays (Q14H)

Q14

<SHOW CARD 14 - CODE ALL MENTIONED>

Here are some aspects of a job that people say are important. Please look at them and tell me which ones you personally think are important in a job?

Q14.H Generous holidays

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific items). Error in show card: Items S and T have the same label. Item T should be "Ein Beruf, der anerkannt und geachtet wird".

v76, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	13	0.9	
1	mentioned		380	25.2	25.4
2	not mentioned		1114	73.8	74.6
	Sum		1510	100.0	100.0
	Valid Cases		1494		

v77 - important in a job: meeting people (Q14I)

Q14

<SHOW CARD 14 - CODE ALL MENTIONED>

Here are some aspects of a job that people say are important. Please look at them and tell me which ones you personally think are important in a job?

Q14.I Meeting people

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific items). Error in show card: Items S and T have the same label. Item T should be "Ein Beruf, der anerkannt und geachtet wird".

v77, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	5	0.3	
-1	dk	М	7	0.5	
1	mentioned		612	40.5	40.9
2	not mentioned		885	58.6	59.1
	Sum		1510	99.9	100.0
	Valid Cases		1497		

v78 - important in a job: achieving something (Q14J)

Q14

<SHOW CARD 14 - CODE ALL MENTIONED>

Here are some aspects of a job that people say are important. Please look at them and tell me which ones you personally think are important in a job?

Q14.J A job in which you feel you can achieve something

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific items). Error in show card: Items S and T have the same label. Item T should be "Ein Beruf, der anerkannt und geachtet wird".

v78, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	7	0.5	
1	mentioned		844	55.9	56.3
2	not mentioned		656	43.4	43.7
	Sum		1510	100.0	100.0
	Valid Cases		1500		

v79 - important in a job: responsible job (Q14K)

Q14

<SHOW CARD 14 - CODE ALL MENTIONED>

Here are some aspects of a job that people say are important. Please look at them and tell me which ones you personally think are important in a job?

Q14.K A responsible job

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific items). Error in show card: Items S and T have the same label. Item T should be "Ein Beruf, der anerkannt und geachtet wird".

v79, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	10	0.7	
1	mentioned		715	47.4	47.8
2	not mentioned		781	51.7	52.2
	Sum		1510	100.0	100.0
	Valid Cases		1496		

v80 - important in a job: interesting job (Q14L)

Q14

<SHOW CARD 14 - CODE ALL MENTIONED>

Here are some aspects of a job that people say are important. Please look at them and tell me which ones you personally think are important in a job?

Q14.L A job that is interesting

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific items). Error in show card: Items S and T have the same label. Item T should be "Ein Beruf, der anerkannt und geachtet wird".

v80, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	5	0.3	
1	mentioned		1036	68.6	68.9
2	not mentioned		467	30.9	31.1
	Sum		1510	100.1	100.0
	Valid Cases		1502		

v81 - important in a job: meeting abilities (Q14M)

Q14

<SHOW CARD 14 - CODE ALL MENTIONED>

Here are some aspects of a job that people say are important. Please look at them and tell me which ones you personally think are important in a job?

Q14.M A job that meets one's abilities

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific items). Error in show card: Items S and T have the same label. Item T should be "Ein Beruf, der anerkannt und geachtet wird".

v81, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	9	0.6	
1	mentioned		897	59.4	59.9
2	not mentioned		601	39.8	40.1
	Sum		1510	100.0	100.0
	Valid Cases		1498		

v82 - important in a job: learning new skills (Q14N)

Q14

<SHOW CARD 14 - CODE ALL MENTIONED>

Here are some aspects of a job that people say are important. Please look at them and tell me which ones you personally think are important in a job?

Q14.N Learning new skills

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific items). Error in show card: Items S and T have the same label. Item T should be "Ein Beruf, der anerkannt und geachtet wird".

v82, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	9	0.6	
1	mentioned		694	46.0	46.4
2	not mentioned		803	53.2	53.6
	Sum		1510	100.0	100.0
	Valid Cases		1497		

v83 - important in a job: family friendly (Q14O)

Q14

<SHOW CARD 14 - CODE ALL MENTIONED>

Here are some aspects of a job that people say are important. Please look at them and tell me which ones you personally think are important in a job?

Q14.0 Family friendly

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific items). Error in show card: Items S and T have the same label. Item T should be "Ein Beruf, der anerkannt und geachtet wird".

v83, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	5	0.3	
1	mentioned		672	44.5	44.8
2	not mentioned		829	54.9	55.2
	Sum		1510	100.0	100.0
	Valid Cases		1501		

v84 - important in a job: have a say (Q14P)

Q14

<SHOW CARD 14 - CODE ALL MENTIONED>

Here are some aspects of a job that people say are important. Please look at them and tell me which ones you personally think are important in a job?

Q14.P Have a say in important decisions

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific items). Error in show card: Items S and T have the same label. Item T should be "Ein Beruf, der anerkannt und geachtet wird".

v84, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	16	1.1	
1	mentioned		563	37.3	37.8
2	not mentioned		927	61.4	62.2
	Sum		1510	100.0	100.0
	Valid Cases		1490		

v85 - important in a job: people treated equally (Q14Q)

Q14

<SHOW CARD 14 - CODE ALL MENTIONED>

Here are some aspects of a job that people say are important. Please look at them and tell me which ones you personally think are important in a job?

Q14.Q People treated equally at the workplace

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific items). Error in show card: Items S and T have the same label. Item T should be "Ein Beruf, der anerkannt und geachtet wird".

v85, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	2	0.1	
-1	dk	М	7	0.5	
1	mentioned		775	51.3	51.6
2	not mentioned		726	48.1	48.4
	Sum		1510	100.0	100.0
	Valid Cases		1501		

AT08 - important in a job: good chances for promotion (Q14R)

Q14

<SHOW CARD 14 - CODE ALL MENTIONED>

Here are some aspects of a job that people say are important. Please look at them and tell me which ones you personally think are important in a job?

Q14R_at Good chances for promotion

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

Error in show card: Items S and T have the same label. Item T should be "Ein Beruf, der anerkannt und geachtet wird".

AT08, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	5	0.3	
-1	dk	М	10	0.7	
1	mentioned		643	42.6	43.0
2	not mentioned		853	56.5	57.0
	Sum		1510	100.1	100.0
	Valid Cases		1495		

AT09 - important in a job: to have time off at the weekends (Q14S)

Q14

<SHOW CARD 14 - CODE ALL MENTIONED>

Here are some aspects of a job that people say are important. Please look at them and tell me which ones you personally think are important in a job?

Q14S_at To have time off at the weekends

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

Error in show card: Items S and T have the same label. Item T should be "Ein Beruf, der anerkannt und geachtet wird".

AT09, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	5	0.3	
-1	dk	М	7	0.5	
1	mentioned		586	38.8	39.1
2	not mentioned		911	60.3	60.9
	Sum		1510	99.9	100.0
	Valid Cases		1498		

AT10 - important in a job: a job respected by people in general (Q14T)

Q14

<SHOW CARD 14 - CODE ALL MENTIONED>

Here are some aspects of a job that people say are important. Please look at them and tell me which ones you personally think are important in a job?

Q14T_at A job respected by people in general

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

Error in show card: Items S and T have the same label. Item T should be "Ein Beruf, der anerkannt und geachtet wird".

AT10, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	5	0.3	
-1	dk	М	11	0.7	
1	mentioned		686	45.4	45.9
2	not mentioned		808	53.5	54.1
	Sum		1510	100.0	100.0
	Valid Cases		1494		

f86 - flag variable: important in a job: none of these (Q14)

Flag variable:

Indicates inconsistencies for the variables v69-v86.

Inconsistent 1

If respondent mentiones at least one aspect and "none".

if v86=1 and any of v69 to v85=1 then f86=1

Inconsistent 2

If respondent does not know for at least one aspect whether it is important and mentiones "none".

if v86=1 and none of v69 to v85=1 and any of v69 to v85=8 then f86=2

Inconsistent 3

If respondent does not to know for at least one aspect whether it is important and does not mentione "none".

if v86=2 and none of v69 to v85=1 and any of v69 to v85=8 then f86=3

Inconsistent 4

If respondent does not mention any aspect and does not mention "none".

if v86=2 and all of v69 to v85=2 then f86=4

Inconsistent 5

If respondent mentions at least one aspect and does not know whether "none" is important.

if v86=8 and any of v69 to v85=1 then f86=5

Inconsistent 6

If respondent does not mention any aspect and does not know whether "none" is important.

if v86=8 and all of v69 to v85=2 then f86=6

Inconsistent 7

If respondent mentions at least one aspect and does not answer whether "none" is important.

if v86=9 and any of v69 to v85=1 then f86=7

Inconsistent 8

If respondent does not mention any aspect and does not answer whether "none" is important.

if v86=9 and all of v69 to v85=2 then f86=8

Note:

For detailed information, see overview flag variables.

f86

Value	Label	Missing Count	Percent	Valid Percent
0	consistent	1510	100.0	100.0
	Sum	1510	100.0	100.0
	Valid Cases	1510		

v86 - important in a job: none of these (Q14)

Q14

<SHOW CARD 14 - CODE ALL MENTIONED>

Here are some aspects of a job that people say are important. Please look at them and tell me which ones you personally think are important in a job?

None of these (spontaneous)

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific items).

v86, weighted by weight

Value Label		Missing Count	Percent	Valid Percent
1 no job	aspect mentioned	150	9.9	9.9
2 job asp	ect mentioned	1360	90.1	90.1
Sum		1510	100.0	100.0
Valid C	Cases	1510		

v89 - are you employed yes/no (Q15)

015

Are you yourself employed or not?

< INCLUDING SELF EMPLOYED!>

<IF CODE 2, -1, -2: GO TO Q18>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

Derivation:

Data cleaning of potentially plausible inconsistencies of v89 and v337 (Q111: paid employment/no paid employment) deviates from general procedure (i.e. no recoding). V89 was recoded for 66 cases into 2 "not employed" and for 24 cases into 1 "employed" based on information and inferred from v337.

v89, weighted by weight

Value	Label	Missing Count	Percent	Valid Percent
1	yes	820	54.3	54.3
2	no	690	45.7	45.7
	Sum	1510	100.0	100.0
	Valid Cases	1510		

v90 - how satisfied are you with your job (Q16)

Q16

<SHOW CARD 16>

Overall, how satisfied or dissatisfied are you with your job?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 dissatisfied

2

3

4

5

6

7 8

9

10 satisfied

v90, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	690	45.7	
-2	na	М	24	1.6	
1	dissatisfied		14	0.9	1.8
2			8	0.5	1.0
3			22	1.5	2.8
4			17	1.1	2.1
5			70	4.6	8.8
6			70	4.6	8.8
7			120	7.9	15.1
8			182	12.1	22.9
9			97	6.4	12.2
10	satisfied		196	13.0	24.6
	Sum		1510	100.0	100.0
	Valid Cases		797		

v91 - how free are you to make decisions in job (Q17)

Q17

<SHOW CARD 17>

How free are you to make decisions in your job? Please use this card to indicate how much decision-making freedom you feel you have.

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 none at all
- 2
- 3
- 4
- 5
- 6
- 7
- 8
- 9
- 10 a great deal

v91, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	690	45.7	
-2	na	М	24	1.6	
-1	dk	М	2	0.1	
1	none at all		34	2.3	4.3
2			16	1.1	2.0
3			45	3.0	5.7
4			32	2.1	4.0
5			111	7.4	14.0
6			98	6.5	12.3
7			142	9.4	17.9
8			146	9.7	18.4
9			69	4.6	8.7
10	a great deal		101	6.7	12.7
	Sum		1510	100.0	100.0
	Valid Cases		795		

AT11 - and how satisfied or dissatisfied are you with your job security? (Q17_at)

Q17_at

<SHOW CARD 17_at>

And how satisfied or dissatisfied are you with your job security. Please tell me according to this scale, where 1 means you are completely dissatisfied and 10 means you are completely satisfied

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 completely dissatisfied

2

3

4

5

6

7

8 9

10 completely satisfied

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country

AT11, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	690	45.7	
-2	na	М	24	1.6	
-1	dk	М	5	0.3	
1	completely dissatisfied		33	2.2	4.2
2			9	0.6	1.1
3			29	1.9	3.7
4			16	1.1	2.0
5			68	4.5	8.6
6			51	3.4	6.4
7			98	6.5	12.4
8			137	9.1	17.3
9			99	6.6	12.5
10	completely satisfied		251	16.6	31.7
	Sum		1510	100.0	100.0
	Valid Cases		791		

v92 - job needed to develop talents (Q18A)

Q18

<ASK ALL>

<SHOW CARD 18>

Do you agree or disagree with the following statements?

<READ OUT AND CODE AN ANSWER FOR EACH>

Q18.A To fully develop your talents, you need to have a job

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree strongly

Remarks on original question in the field questionnaire:

Deviation in interviewer instruction ("ask all" is missing).

v92, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	3	0.2	
1	agree strongly		580	38.4	38.5
2	agree		668	44.2	44.3
3	neither agree nor disagree		159	10.5	10.6
4	disagree		75	5.0	5.0
5	disagree strongly		25	1.7	1.7
	Sum		1510	100.1	100.0
	Valid Cases		1507		

v93 - humiliating receiving money without working (Q18B)

Q18

<ASK ALL>

<SHOW CARD 18>

Do you agree or disagree with the following statements?

<READ OUT AND CODE AN ANSWER FOR EACH>

Q18.B It is humiliating to receive money without having to work for it

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree strongly

Remarks on original question in the field questionnaire:

Deviation in interviewer instruction ("ask all" is missing).

v93, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	18	1.2	
1	agree strongly		293	19.4	19.7
2	agree		455	30.1	30.5
3	neither agree nor disagree		377	25.0	25.3
4	disagree		242	16.0	16.2
5	disagree strongly		123	8.1	8.3
	Sum		1510	99.9	100.0
	Valid Cases		1491		

v94 - people turn lazy not working (Q18C)

Q18

<ASK ALL>

<SHOW CARD 18>

Do you agree or disagree with the following statements?

<READ OUT AND CODE AN ANSWER FOR EACH>

Q18.C People who don't work turn lazy

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree strongly

Remarks on original question in the field questionnaire:

Deviation in interviewer instruction ("ask all" is missing).

v94, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	15	1.0	
1	agree strongly		512	33.9	34.2
2	agree		506	33.5	33.8
3	neither agree nor disagree		272	18.0	18.2
4	disagree		142	9.4	9.5
5	disagree strongly		63	4.2	4.2
	Sum		1510	100.0	100.0
	Valid Cases		1495		

v95 - work is a duty towards society (Q18D)

Q18

<ASK ALL>

<SHOW CARD 18>

Do you agree or disagree with the following statements?

<READ OUT AND CODE AN ANSWER FOR EACH>

Q18.D Work is a duty towards society

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree strongly

Remarks on original question in the field questionnaire:

Deviation in interviewer instruction ("ask all" is missing).

v95, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	9	0.6	
1	agree strongly		508	33.6	33.8
2	agree		576	38.1	38.4
3	neither agree nor disagree		254	16.8	16.9
4	disagree		99	6.6	6.6
5	disagree strongly		64	4.2	4.3
	Sum		1510	100.1	100.0
	Valid Cases		1500		

v96 - work comes always first (Q18E)

Q18

<ASK ALL>

<SHOW CARD 18>

Do you agree or disagree with the following statements?

<READ OUT AND CODE AN ANSWER FOR EACH>

Q18.E Work should always come first, even if it means less spare time

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree strongly

Remarks on original question in the field questionnaire:

Deviation in interviewer instruction ("ask all" is missing).

v96, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	9	0.6	
1	agree strongly		273	18.1	18.2
2	agree		542	35.9	36.2
3	neither agree nor disagree		352	23.3	23.5
4	disagree		234	15.5	15.6
5	disagree strongly		98	6.5	6.5
	Sum		1510	99.9	100.0
	Valid Cases		1500		

v97 - leisure time: meeting nice people (Q19A)

Q19

<SHOW CARD 19 - READ OUT EACH ITEM AND CODE AN ANSWER FOR EACH>

I am going to read to you some aspects of leisure time that some people say are important. Please tell me for each how important they are for you personally.

<DEFINE LEISURE TIME: TIME NOT SPENT IN PAID JOB OR HOUSEWORK>

Q19.A Meeting nice people

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very important
- 2 quite important
- 3 not important
- 4 not important at all

v97, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	5	0.3	
1	very important		666	44.1	44.3
2	quite important		673	44.6	44.7
3	not important		147	9.7	9.8
4	not important at all		18	1.2	1.2
	Sum		1510	99.9	100.0
	Valid Cases		1505		

v98 - leisure time: relaxing (Q19B)

Q19

<SHOW CARD 19 - READ OUT EACH ITEM AND CODE AN ANSWER FOR EACH>

I am going to read to you some aspects of leisure time that some people say are important. Please tell me for each how important they are for you personally.

<DEFINE LEISURE TIME: TIME NOT SPENT IN PAID JOB OR HOUSEWORK>

Q19.B Relaxing

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very important
- 2 quite important
- 3 not important
- 4 not important at all

v98, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	3	0.2	
1	very important		917	60.7	60.9
2	quite important		516	34.2	34.3
3	not important		68	4.5	4.5
4	not important at all		5	0.3	0.3
	Sum		1510	99.9	100.0
	Valid Cases		1507		

v99 - leisure time: doing as I want (Q19C)

Q19

<SHOW CARD 19 - READ OUT EACH ITEM AND CODE AN ANSWER FOR EACH>

I am going to read to you some aspects of leisure time that some people say are important. Please tell me for each how important they are for you personally.

<DEFINE LEISURE TIME: TIME NOT SPENT IN PAID JOB OR HOUSEWORK>

Q19.C Doing as I want

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very important
- 2 quite important
- 3 not important
- 4 not important at all

v99, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	3	0.2	
1	very important		687	45.5	45.7
2	quite important		632	41.9	42.0
3	not important		167	11.1	11.1
4	not important at all		18	1.2	1.2
	Sum		1510	99.9	100.0
	Valid Cases		1506		

v100 - leisure time: learning something new (Q19D)

Q19

<SHOW CARD 19 - READ OUT EACH ITEM AND CODE AN ANSWER FOR EACH>

I am going to read to you some aspects of leisure time that some people say are important. Please tell me for each how important they are for you personally.

<DEFINE LEISURE TIME: TIME NOT SPENT IN PAID JOB OR HOUSEWORK>

Q19.D Learning something new

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very important
- 2 quite important
- 3 not important
- 4 not important at all

v100, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	7	0.5	
1	very important		462	30.6	30.7
2	quite important		610	40.4	40.6
3	not important		356	23.6	23.7
4	not important at all		76	5.0	5.1
	Sum		1510	100.1	100.0
	Valid Cases		1503		

v101 - follow instructions at work/must be convinced first (Q20)

Q20

<SHOW CARD 20>

People have different ideas about following instruction at work. Some say that one should follow instructions of one's superiors even when one does not fully agree with them. Others say that one should follow one's superiors' instructions only when one is convinced that they are right. Which of these two opinions do you agree with?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 should follow instructions
- 2 must be convinced first
- 3 depends

v101, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	2	0.1	
-1	dk	М	11	0.7	
1	follow instructions		345	22.8	23.1
2	be convinced first		597	39.5	39.9
3	depends		554	36.7	37.0
	Sum		1510	99.9	100.0
	Valid Cases		1497		

v102 - jobs are scarce: giving...(nation) priority (Q21A)

Q21

<SHOW CARD 21 - READ OUT AND CODE ONE ANSWER PER LINE>

Do you agree or disagree with the following statements?

Q21.A When jobs are scarce, employers should give priority to [NATIONALITY] people over immigrants

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree
- 2 disagree
- 3 neither (spontaneous)

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v102, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	9	0.6	
1	agree		985	65.2	65.7
2	disagree		365	24.2	24.3
3	neither		149	9.9	9.9
	Sum		1510	100.1	100.0
	Valid Cases		1498		

v103 - jobs are scarce: giving men priority (Q21B)

Q21

<SHOW CARD 21 - READ OUT AND CODE ONE ANSWER PER LINE>

Do you agree or disagree with the following statements?

Q21.B When jobs are scarce, men have more right to a job than women

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree
- 2 disagree
- 3 neither (spontaneous)

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v103, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	20	1.3	
1	agree		310	20.5	20.8
2	disagree		975	64.6	65.6
3	neither		202	13.4	13.6
	Sum		1510	100.1	100.0
	Valid Cases		1486		

AT12 - jobs are scarce: giving younger people priority (Q21C)

Q21

<SHOW CARD 21 - READ OUT AND CODE ONE ANSWER PER LINE>

Do you agree or disagree with the following statements?

Q21C_at When there are few jobs, younger people have more right to a job than older people

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree
- 2 disagree
- 3 neither (spontaneous)

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT12, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	8	0.5	
-1	dk	М	25	1.7	
1	agree		383	25.4	25.9
2	disagree		853	56.5	57.8
3	neither (spontanous)		241	16.0	16.3
	Sum		1510	100.0	100.0
	Valid Cases		1478		

v104 - good and evil: clear/no clear guidelines (Q22)

NOW WE ARE GOING TO ASK YOU SOME QUESTIONS ABOUT THE MEANING AND PURPOSE OF LIFE Q22

<SHOW CARD 22 - READ OUT AND CODE ONE ANSWER ONLY>

Here are three statements which people sometimes make when discussing good and evil. Which one comes closest to your own point of view?

- A: There are absolutely clear guidelines about what is good and evil. These always apply to everyone, whatever the circumstances.
- B: There are absolutely clear guidelines about what is good and evil. However, deviating from these guidelines is sometimes justified by special circumstances.
- C: There can never be absolutely clear guidelines about what is good and evil. What is good and evil depends entirely upon the circumstances at the time.
- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 Agree with statement A
- 2 Agree with statement B
- 3 Agree with statement C

v104, weighted by weight

Value Label		Missing	Count	Percent	Valid Percent
-2 na		М	1	0.1	
-1 dk		М	43	2.8	
1 clear	guidelines good and evil		280	18.5	19.1
2 clear	guidelines but deviation sometimes justified		558	37.0	38.1
3 no cle	ear guidelines good and evil		627	41.5	42.8
Sum			1510	99.9	100.0
Valid	Cases		1466		

v105 - do you belong to a religious denomination (Q23)

Q23

Do you belong to a religious denomination?

<IF CODE 2, -1, -2: GO TO Q24>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v105, weighted by weight

	Ū	,	•			
Value	Label		Missing	Count	Percent	Valid Percent
-2	na		М	1	0.1	
-1	dk		М	2	0.1	
1	yes			1248	82.6	82.8
2	no			259	17.2	17.2
	Sum			1510	100.0	100.0
	Valid Ca	ses		1507		

v106 - which religious denomination do you belong to (Q23a) (standardized)

RELIGIOUS DENOMINATION - HARMONIZED VARIABLE

[Q23: Do you belong to a religious denomination?]

Q23a

<SHOW CARD 23a>

Which one?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 roman catholic
- 2 protestant
- 3 free church/ non-conformist/ evangelical
- 4 jew
- 5 muslim
- 6 hindu
- 7 buddhist
- 8 orthodox
- 9 other, please specifiy (Write in):

Note:

Harmonized variable: country-specific categories have been recoded according to EVS common coding frame. For country-specific variable, see V106_cs.

v106, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	262	17.4	
1	roman catholic		1097	72.6	87.8
2	protestant		78	5.2	6.2
3	free church/ non-conformist/ evangelical		6	0.4	0.5
5	muslim		27	1.8	2.2
7	buddhist		2	0.1	0.2
8	orthodox		16	1.1	1.3
9	other		23	1.5	1.8
	Sum		1510	100.1	100.0
	Valid Cases		1248		

v106_cs - which religious denomination do you belong to (Q23a)

[Q23: Do you belong to a religious denomination?]

Q23a

<SHOW CARD 23a>

Which one?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 40001 AT Roman catholic
- 40002 AT Protestant(without independent churches)
- 40003 AT Independent church
- 40004 AT Orthodox/greek-oriental orthodox
- 40005 AT Non-roman catholic(e.g. greek-catholic)
- 40006 AT Old catholic
- 40007 AT Other christian denomination
- 40008 AT Jewish
- 40009 AT Moslem
- 40010 AT Buddhist
- 40011 AT Hindu
- 40012 AT Jehovahs Witnesses
- 40096 AT Other religion

Note:

Standardized country-specific variable: combined variable containing country identifier (see COUNTRY) and country-specific categories (religious denominations). This variable has been harmonized into v106.

v106_cs, weighted by weight

Value Label		Missing	Count	Percent	Valid Percent
-3 nap		М	262	17.4	
40001 AT Roman catholic			1097	72.6	87.8
40002 AT Protestant(without inde	ependent churches)		78	5.2	6.2
40003 AT Independent church			6	0.4	0.5
40004 AT Orthodox/greek-orienta	al orthodox		16	1.1	1.3
40006 AT Old catholic			5	0.3	0.4
40007 AT Other christian denom	ination		7	0.5	0.6
40009 AT Moslem			27	1.8	2.2
40010 AT Buddhist			2	0.1	0.2
40096 AT Other religion			11	0.7	0.9
Sum			1510	100.1	100.0
Valid Cases			1248		

v107 - did you belong to a religious denomination (Q24)

Q24

<IF RESPONDENT IS CURRENTLY A MEMBER (Q23=1), PLEASE ASK:>

Were you ever a member of another religious denomination?

<IF RESPONDENT IS CURRENTLY NOT A MEMBER (Q23=2), PLEASE ASK:>

Were you ever a member of a religious denomination?

<IF CODE 2, -1, -2: GO TO Q25>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

Derivation:

Country specific filter was used that deviates from Master Questionnaire. Affected cases have been recoded from 7 'nap' into -5 'other missing'.

v107, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-5	other missing	М	3	0.2	
-2	na	М	1	0.1	
1	yes		212	14.0	14.1
2	no		1294	85.7	85.9
	Sum		1510	100.0	100.0
	Valid Cases		1506		

v108 - did you belong to a religious denomination (Q24a) (standardized)

(ANOTHER) RELIGIOUS DENOMINATION - HARMONIZED VARIABLE

[Q24: Were you ever a member of (another) religious denomination?]

Q24a

<SHOW CARD 24A - ONE ANSWER ONLY>

Which one?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 roman catholic
- 2 protestant
- 3 free church/ non-conformist/ evangelical
- 4 jew
- 5 muslim
- 6 hindu
- 7 buddhist
- 8 orthodox
- 9 other, please specifiy (Write in):

Note:

Harmonized variable: country-specific categories have been recoded according to EVS common coding frame.

For country-specific variable, see V108_cs.

v108, weighted by weight

Label	Missing	Count	Percent	Valid Percent
пар	М	1298	86.0	
roman catholic		180	11.9	84.9
protestant		22	1.5	10.4
free church/ non-conformist/ evangelical		1	0.1	0.5
jew		1	0.1	0.5
muslim		1	0.1	0.5
other		7	0.5	3.3
Sum		1510	100.0	100.0
Valid Cases		212		
	nap roman catholic protestant free church/ non-conformist/ evangelical jew muslim other Sum	nap M roman catholic protestant free church/ non-conformist/ evangelical jew muslim other Sum	nap M 1298 roman catholic 180 protestant 22 free church/ non-conformist/ evangelical 1 jew 1 muslim 1 other 7 Sum 1510	nap M 1298 86.0 roman catholic 180 11.9 protestant 22 1.5 free church/ non-conformist/ evangelical 1 0.1 jew 1 0.1 muslim 1 0.1 other 7 0.5 Sum 1510 100.0

v108_cs - did you belong to a religious denomination (Q24a)

[Q24: Were you ever a member of (another) religious denomination?]

<SHOW CARD 24A - ONE ANSWER ONLY>

Which one?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 40001 AT Roman catholic
- 40002 AT Protestant(without independent churches)
- 40003 AT Independent church
- 40004 AT Orthodox/greek-oriental orthodox
- 40005 AT Non-roman catholic(e.g. greek-catholic)
- 40006 AT Old catholic
- 40007 AT Other christian denomination
- 40008 AT Jewish
- 40009 AT Moslem
- 40010 AT Buddhist
- 40011 AT Hindu
- 40012 AT Jehovahs Witnesses
- 40096 AT Other religion

Note:

Standardized country-specific variable: combined variable containing country identifier (see COUNTRY) and country-specific categories (religious denominations). This variable has been harmonized into v108.

v108_cs, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	1298	86.0	
40001	AT Roman catholic		180	11.9	85.3
40002	AT Protestant(without independent churches)		22	1.5	10.4
40003	AT Independent church		1	0.1	0.5
40006	AT Old catholic		1	0.1	0.5
40007	AT Other christian denomination		2	0.1	0.9
40008	AT Jewish		1	0.1	0.5
40009	AT Moslem		1	0.1	0.5
40096	AT Other religion		3	0.2	1.4
	Sum		1510	99.9	100.0
	Valid Cases		212		

v109 - how often attend religious services (Q25)

Q25

<ASK ALL>

<SHOW CARD 25 - READ OUT - CODE ONE ANSWER ONLY>

Apart from weddings, funerals and christenings, about how often do you attend religious services these days?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 more than once week
- 2 once a week
- 3 once a month
- 4 only on specific holy days
- 5 once a year
- 6 less often
- 7 never, practically never

v109, weighted by weight

,	- 5 7 5 -				
Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	5	0.3	
1	more than once week		35	2.3	2.3
2	once a week		215	14.2	14.3
3	once a month		186	12.3	12.4
4	only on specific holy days		348	23.0	23.1
5	once a year		68	4.5	4.5
6	less often		169	11.2	11.2
7	never, practically never		483	32.0	32.1
	Sum		1510	100.0	100.0
	Valid Cases		1504		

v110 - how often attend religious services 12 years old (Q26)

Q26

<SHOW CARD 26 - READ OUT AND CODE ONE ANSWER ONLY>

Apart from weddings, funerals and christenings, about how often did you attend religious services when you were 12 years old?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 more than once week
- 2 once a week
- 3 once a month
- 4 only on specific holy days
- 5 once a year
- 6 less often
- 7 never, practically never

v110, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	33	2.2	
1	more than once week		117	7.7	7.9
2	once a week		576	38.1	39.0
3	once a month		185	12.3	12.5
4	only on specific holy days		284	18.8	19.2
5	once a year		34	2.3	2.3
6	less often		112	7.4	7.6
7	never, practically never		168	11.1	11.4
	Sum		1510	99.9	100.0
	Valid Cases		1477		

v111 - religious service important: birth (Q27A)

Q27

Do you personally think it is important to hold a religious service for any of the following events?

<READ OUT AND CODE ONE ANSWER PER LINE>

Q27.A Birth

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v111, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	38	2.5	
1	yes		1184	78.4	80.4
2	no		289	19.1	19.6
	Sum		1510	100.1	100.0
	Valid Cases		1472		

v112 - religious service important: marriage (Q27B)

Q27

Do you personally think it is important to hold a religious service for any of the following events?

<READ OUT AND CODE ONE ANSWER PER LINE>

Q27.B Marriage

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v112, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	38	2.5	
1	yes		1164	77.1	79.1
2	no		308	20.4	20.9
	Sum		1510	100.0	100.0
	Valid Cases		1472		

v113 - religious service important: death (Q27C)

Q27

Do you personally think it is important to hold a religious service for any of the following events?

<READ OUT AND CODE ONE ANSWER PER LINE>

Q27.C Death

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v113, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	39	2.6	
1	yes		1242	82.3	84.4
2	no		229	15.2	15.6
	Sum		1510	100.0	100.0
	Valid Cases		1471		

v114 - are you a religious person (Q28)

Q28

Independently of whether you go to church or not, would you say you are ...

<READ OUT EACH ITEM - CODE ONE ANSWER ONLY>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 a religious person
- 2 not a religious person
- 3 a convinced atheist

v114, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	6	0.4	
-1	dk	М	68	4.5	
1	religious person		920	60.9	64.1
2	not religious person		448	29.7	31.2
3	convinced atheist		68	4.5	4.7
	Sum		1510	100.0	100.0
	Valid Cases		1436		

v115 - church answers to: moral problems (Q29A)

 Ω 20

<FOR THOSE BELONGING TO A CHURCH OR A RELIGIOUS COMMUNITY; ASK YOUR CHURCH/ RELIGIOUS COMMUNITY (IF Q23=1)>

Generally speaking, do you think that your church is giving, in your country, adequate answers to ...

<FOR THOSE NOT BELONGING TO A CHURCH OR RELIGIOUS COMMUNITY ASK: THE CHURCHES (IF Q23=2)> Generally speaking, do you think that the churches are giving, in your country, adequate answers to ...

<READ OUT AND CODE ONE ANSWER PER LINE>

Q29.A The moral problems and needs of the individual

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v115, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	105	7.0	
1	yes		607	40.2	43.3
2	no		794	52.6	56.7
	Sum		1510	100.0	100.0
	Valid Cases		1401		

v116 - church answers to: family life problems (Q29B)

029

<FOR THOSE BELONGING TO A CHURCH OR A RELIGIOUS COMMUNITY; ASK YOUR CHURCH/ RELIGIOUS COMMUNITY (IF Q23=1)>

Generally speaking, do you think that your church is giving, in your country, adequate answers to ...

<FOR THOSE NOT BELONGING TO A CHURCH OR RELIGIOUS COMMUNITY ASK: THE CHURCHES (IF Q23=2)> Generally speaking, do you think that the churches are giving, in your country, adequate answers to ...

<READ OUT AND CODE ONE ANSWER PER LINE>

Q29.B The problems of family life

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v116, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	96	6.4	
1	yes		462	30.6	32.7
2	no		949	62.8	67.3
	Sum		1510	100.1	100.0
	Valid Cases		1411		

v117 - church answers to: spiritual needs (Q29C)

 Ω

<FOR THOSE BELONGING TO A CHURCH OR A RELIGIOUS COMMUNITY; ASK YOUR CHURCH/ RELIGIOUS COMMUNITY (IF Q23=1)>

Generally speaking, do you think that your church is giving, in your country, adequate answers to ...

<FOR THOSE NOT BELONGING TO A CHURCH OR RELIGIOUS COMMUNITY ASK: THE CHURCHES (IF Q23=2)>
Generally speaking, do you think that the churches are giving, in your country, adequate answers to ...

<READ OUT AND CODE ONE ANSWER PER LINE>

Q29.C People's spiritual needs

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v117, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	97	6.4	
1	yes		776	51.4	55.1
2	no		633	41.9	44.9
	Sum		1510	99.9	100.0
	Valid Cases		1409		

v118 - church answers to: social problems (Q29D)

 Ω

<FOR THOSE BELONGING TO A CHURCH OR A RELIGIOUS COMMUNITY; ASK YOUR CHURCH/ RELIGIOUS COMMUNITY (IF Q23=1)>

Generally speaking, do you think that your church is giving, in your country, adequate answers to ...

<FOR THOSE NOT BELONGING TO A CHURCH OR RELIGIOUS COMMUNITY ASK: THE CHURCHES (IF Q23=2)> Generally speaking, do you think that the churches are giving, in your country, adequate answers to ...

<READ OUT AND CODE ONE ANSWER PER LINE>

Q29.D The social problems facing our country today

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v118, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	100	6.6	
1	yes		447	29.6	31.8
2	no		959	63.5	68.2
	Sum		1510	99.9	100.0
	Valid Cases		1407		

v119 - do you believe in: God (Q30A)

Q30

Which, if any, of the following do you believe in?

<READ OUT AND CODE ONE ANSWER PER LINE>

Q30.A God

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v119, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	139	9.2	
1	yes		1090	72.2	79.6
2	no		279	18.5	20.4
	Sum		1510	100.1	100.0
	Valid Cases		1369		

v120 - do you believe in: life after death (Q30B)

Q30

Which, if any, of the following do you believe in?

<READ OUT AND CODE ONE ANSWER PER LINE>

Q30.B Life after death

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v120, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	219	14.5	
1	yes		779	51.6	60.5
2	no		508	33.6	39.5
	Sum		1510	99.9	100.0
	Valid Cases		1288		

v121 - do you believe in: hell (Q30C)

Q30

Which, if any, of the following do you believe in?

<READ OUT AND CODE ONE ANSWER PER LINE>

Q30.C Hell

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v121, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	178	11.8	
1	yes		360	23.8	27.0
2	no		971	64.3	73.0
	Sum		1510	100.0	100.0
	Valid Cases		1331		

v122 - do you believe in: heaven (Q30D)

Q30

Which, if any, of the following do you believe in?

<READ OUT AND CODE ONE ANSWER PER LINE>

Q30.D Heaven

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v122, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	174	11.5	
1	yes		543	36.0	40.7
2	no		792	52.5	59.3
	Sum		1510	100.0	100.0
	Valid Cases		1335		

v123 - do you believe in: sin (Q30E)

Q30

Which, if any, of the following do you believe in?

<READ OUT AND CODE ONE ANSWER PER LINE>

Q30.E Sin

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v123, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	132	8.7	
1	yes		778	51.5	56.5
2	no		598	39.6	43.5
	Sum		1510	99.9	100.0
	Valid Cases		1377		

v124 - do you believe in: re-incarnation (Q31)

0.31

Do you believe in re-incarnation, that is, that we are born into this world again?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v124, weighted by weight

Value	Labol	Miccina	Count	Dorcont	Valid Percent
value	Label	iviissiiig	Count	reiceili	Valid Percerit
-2	na	М	1	0.1	
-1	dk	М	211	14.0	
1	yes		356	23.6	27.4
2	no		942	62.4	72.6
	Sum		1510	100.0	100.0
	Valid Cases		1298		

v125 - which statement closest to your beliefs (Q32)

Q32

<SHOW CARD 32 - READ OUT - CODE ONE ANSWER ONLY>

Which of these statements comes closest to your beliefs?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 there is a personal God
- 2 there is some sort of spirit or life force
- 3 I don't really know what to think
- 4 I don't really think there is any sort of spirit, God or life force

v125, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	24	1.6	
1	personal God		388	25.7	26.1
2	spirit or life force		725	48.0	48.8
3	dont know what to think		204	13.5	13.7
4	no spirit, God or life force		169	11.2	11.4
	Sum		1510	100.0	100.0
	Valid Cases		1486		

v126 - own way of connecting with the divine (Q33)

Q33

<SHOW CARD 33>

Please choose a number on the card to describe how true this statement is for you?

I have my own way of connecting with the Divine without churches or religious services.

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 not at all
- 2
- 3
- 4
- 5 very much

v126, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	6	0.4	
-1	dk	М	32	2.1	
1	not at all		276	18.3	18.8
2			166	11.0	11.3
3			365	24.2	24.8
4			302	20.0	20.5
5	very much		362	24.0	24.6
	Sum		1510	99.9	100.0
	Valid Cases		1472		

v127 - how spiritual are you (Q34)

Q34

<SHOW CARD 34>

Whether or not you think of yourself as a religious person, how spiritual would you say you are, that is how strongly are you interested in the sacred or the supernatural?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very interested
- 2 somewhat interested
- 3 not very interested
- 4 not at all interested

v127, weighted by weight

,	. 5				
Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	7	0.5	
1	very interested		246	16.3	16.4
2	somewhat interested		613	40.6	40.8
3	not very interested		367	24.3	24.4
4	not at all interested		276	18.3	18.4
	Sum		1510	99.9	100.0
	Valid Cases		1503		

v128 - only one true religion or no religion offers any truths (Q35)

Q35

<SHOW CARD 35 - READ OUT AND CODE ONE ANSWER ONLY>

These are statements one sometimes hears. Please choose the statement that best describes your view?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 There is only one true religion
- 2 There is only one true religion, but other religions do contain some basic truths as well
- 3 There is not one true religion, but all great world religions contain some basic truths
- 4 None of the great religions have any truths to offer

v128, weighted by weight

Value Lab	pel	Missing	Count	Percent	Valid Percent
-1 dk		М	81	5.4	
1 only	y one true religion		133	8.8	9.3
2 only	y one true religion but others contain some truths too		354	23.4	24.8
3 no ti	true religion, all contain some basic truths		695	46.0	48.6
4 none	e of the religions offer any truths		247	16.4	17.3
Sum	n		1510	100.0	100.0
Valid	id Cases		1429		

v129 - how important is God in your life (Q36)

Q36

<SHOW CARD 36>

And how important is God in your life? Please use this card to indicate - 10 means very important and 1 means not at all important.

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 not at all important

2

3

4

5

6 7

8

9

10 very important

v129, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	25	1.7	
1	not at all		217	14.4	14.7
2			55	3.6	3.7
3			115	7.6	7.8
4			47	3.1	3.2
5			265	17.5	17.9
6			141	9.3	9.5
7			170	11.3	11.5
8			151	10.0	10.2
9			70	4.6	4.7
10	very		250	16.6	16.9
	Sum		1510	99.9	100.0
	Valid Cases		1482		

v130 - do you get comfort and strength from religion (Q37)

O37

Do you find that you get comfort and strength from religion or not?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v130, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	2	0.1	
-1	dk	М	92	6.1	
1	yes		813	53.8	57.4
2	no		603	39.9	42.6
	Sum		1510	100.0	100.0
	Valid Cases		1416		

v131 - do you take moments of prayer/meditation (Q38)

038

Do you take some moments of prayer, meditation or contemplation or something like that?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v131, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	2	0.1	
-1	dk	М	40	2.6	
1	yes		911	60.3	62.1
2	no		557	36.9	37.9
	Sum		1510	100.0	100.0
	Valid Cases		1468		

v132 - how often do you pray to God outside religious services (Q39)

Q39

<SHOW CARD 39 - READ OUT AND CODE ONE ANSWER ONLY>

How often do you pray to God outside of religious services? Would you say

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 every day
- 2 more than once week
- 3 once a week
- 4 at least once a month
- 5 several times a year
- 6 less often
- 7 never

v132, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	42	2.8	
1	every day		211	14.0	14.4
2	more than once week		184	12.2	12.6
3	once a week		137	9.1	9.4
4	at least once a month		185	12.3	12.6
5	several times a year		126	8.3	8.6
6	less often		252	16.7	17.2
7	never		369	24.4	25.2
	Sum		1510	100.0	100.0
	Valid Cases		1465		

v133 - do you believe that lucky charm protects (Q40)

Q40

<SHOW CARD 40>

Do you believe that a lucky charm such as a mascot or a talisman can protect or help you?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 definitely not

2

3

4

5

6

7 8

9

10 definitely yes

v133, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	49	3.2	
1	definitely not		570	37.7	39.0
2			52	3.4	3.6
3			149	9.9	10.2
4			38	2.5	2.6
5			223	14.8	15.3
6			70	4.6	4.8
7			101	6.7	6.9
8			83	5.5	5.7
9			45	3.0	3.1
10	definitely yes		129	8.5	8.8
	Sum		1510	100.0	100.0
	Valid Cases		1460		

v134 - politicians and God (Q41A)

Q41

<SHOW CARD 41 - READ OUT AND CODE ONE ANSWER PER LINE!>

How much do you agree or disagree with each of the following:

Q41.A Politicians who do not believe in God are unfit for public office

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree strongly

Remarks on original question in the field questionnaire:

Deviation in interviewer instruction (additional interviewer instruction "read out and code an answer for each")

v134, weighted by weight

Value Label	Missing	Count	Percent	Valid Percent
-2 na	М	4	0.3	
-1 dk	М	63	4.2	
1 agree strongly		116	7.7	8.0
2 agree		205	13.6	14.2
3 agree nor disagree		312	20.7	21.6
4 disagree		351	23.2	24.3
5 disagree strongly		459	30.4	31.8
Sum		1510	100.0	100.0
Valid Cases		1443		

v135 - religious leaders and influence government decisions (Q41B)

Q41

<SHOW CARD 41 - READ OUT AND CODE ONE ANSWER PER LINE!>

How much do you agree or disagree with each of the following:

Q41.B Religious leaders should not influence government decisions

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree strongly

Remarks on original question in the field questionnaire:

Deviation in interviewer instruction (additional interviewer instruction "read out and code an answer for each")

v135, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	44	2.9	
1	agree strongly		553	36.6	37.8
2	agree		443	29.3	30.3
3	agree nor disagree		273	18.1	18.7
4	disagree		130	8.6	8.9
5	disagree strongly		64	4.2	4.4
	Sum		1510	100.1	100.0
	Valid Cases		1463		

AT13 - more time for prayer, mediation or contemplation in all schools (Q41C)

Q41

<SHOW CARD 41 - READ OUT AND CODE ONE ANSWER PER LINE!>

How much do you agree or disagree with each of the following:

Q41C_at In my opinion, some time should be set aside for prayer, meditation or contemplation, in all schools

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree strongly

Remarks on original question in the field questionnaire:

AT13, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	54	3.6	
1	agree strongly		351	23.2	24.2
2	agree		481	31.9	33.1
3	agree nor disagree		331	21.9	22.8
4	disagree		181	12.0	12.5
5	disagree strongly		109	7.2	7.5
	Sum		1510	100.1	100.0
	Valid Cases		1453		

AT14 - how often do you think about the meaning of life? (Q41_1)

Q41.1_at

<SHOW CARD 41.1._at>

Do you ever think about the meaning of life? Would say you think about it...

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 often
- 2 sometimes
- 3 seldom
- 4 never

Remarks on original question in the field questionnaire:

AT14, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	2	0.1	
-1	dk	М	3	0.2	
1	often		361	23.9	24.0
2	sometimes		704	46.6	46.7
3	seldom		341	22.6	22.6
4	never		100	6.6	6.6
	Sum		1510	100.1	100.0
	Valid Cases		1505		

AT15 - did you receive a religious upbringing by your parents? (Q41_2)

Q41.2_at

Did you receive a religious upbringing from your parents?

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 very religious
- 2 rather religious
- 3 not very religious
- 4 not at all religious

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT15, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	2	0.1	
1	very religious		412	27.3	27.3
2	rather religious		606	40.1	40.2
3	not very religious		366	24.2	24.3
4	not at all religious		124	8.2	8.2
	Sum		1510	100.0	100.0
	Valid Cases		1508		

AT16 - agree with: there is a God who attends to every human personally (Q41_3A)

Q41.3_at

<SHOW CARD 41.3_at>

How much do you agree or disagree with each of the following opinions?

Q41.3A_at There is a God who who attends to every human being personally

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 disagree completely
- 2 tend to disagree
- 3 tend to agree
- 4 agree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT16, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	151	10.0	
1	disagree completely		340	22.5	25.1
2	tend to disagree		372	24.6	27.4
3	tend to agree		399	26.4	29.4
4	agree completely		245	16.2	18.1
	Sum		1510	100.0	100.0
	Valid Cases		1356		

AT17 - agree with: impossible to know if there is anything beyond this world (Q41_3B)

Q41.3_at

<SHOW CARD 41.3_at>

How much do you agree or disagree with each of the following opinions?

Q41.3B_at It's impossible to get an answer to the question whether there is anything beyond this world

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 disagree completely
- 2 tend to disagree
- 3 tend to agree
- 4 agree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT17, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	115	7.6	
1	disagree completely		139	9.2	10.0
2	tend to disagree		356	23.6	25.6
3	tend to agree		524	34.7	37.6
4	agree completely		373	24.7	26.8
	Sum		1510	100.1	100.0
	Valid Cases		1391		

AT18 - agree with: life carries its meaning within itself (Q41_3C)

Q41.3_at

<SHOW CARD 41.3_at>

How much do you agree or disagree with each of the following opinions?

Q41.3C_at For me, life carries its meaning within itself

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 disagree completely
- 2 tend to disagree
- 3 tend to agree
- 4 agree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT18, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	84	5.6	
1	disagree completely		79	5.2	5.6
2	tend to disagree		189	12.5	13.3
3	tend to agree		690	45.7	48.5
4	agree completely		465	30.8	32.7
	Sum		1510	100.1	100.0
	Valid Cases		1422		

AT19 - agree with: you dont feel much of Gods existence (Q41_3D)

Q41.3_at

<SHOW CARD 41.3_at>

How much do you agree or disagree with each of the following opinions?

Q41.3D_at If God exists at all, you don't feel much of his existence

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 disagree completely
- 2 tend to disagree
- 3 tend to agree
- 4 agree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT19, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	98	6.5	
1	disagree completely		197	13.0	14.0
2	tend to disagree		414	27.4	29.4
3	tend to agree		443	29.3	31.4
4	agree completely		355	23.5	25.2
	Sum		1510	100.0	100.0
	Valid Cases		1409		

AT20 - agree with: life is determined by the laws of nature (Q41_3E)

Q41.3_at

<SHOW CARD 41.3_at>

How much do you agree or disagree with each of the following opinions?

Q41.3E_at In the end, life is determined by the laws of nature

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 disagree completely
- 2 tend to disagree
- 3 tend to agree
- 4 agree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT20, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	2	0.1	
-1	dk	М	76	5.0	
1	disagree completely		80	5.3	5.6
2	tend to disagree		248	16.4	17.3
3	tend to agree		652	43.2	45.5
4	agree completely		452	29.9	31.6
	Sum		1510	100.0	100.0
	Valid Cases		1431		

AT21 - agree with: life only has a meaning if you give it one (Q41_3F)

Q41.3_at

<SHOW CARD 41.3_at>

How much do you agree or disagree with each of the following opinions?

Q41.3F_at Life is only meaningful if you provide the meaning yourself

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 disagree completely
- 2 tend to disagree
- 3 tend to agree
- 4 agree completely

Remarks on original question in the field questionnaire:

AT21, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	2	0.1	
-1	dk	М	41	2.7	
1	disagree completely		78	5.2	5.3
2	tend to disagree		134	8.9	9.1
3	tend to agree		545	36.1	37.2
4	agree completely		710	47.0	48.4
	Sum		1510	100.0	100.0
	Valid Cases		1467		

AT22 - agree with: there is something like a higher power (a higher being) (Q41_3G)

Q41.3_at

<SHOW CARD 41.3_at>

How much do you agree or disagree with each of the following opinions?

Q41.3G_at There is something like a higher power (a higher being)

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 disagree completely
- 2 tend to disagree
- 3 tend to agree
- 4 agree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT22, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	2	0.1	
-1	dk	М	123	8.1	
1	disagree completely		172	11.4	12.4
2	tend to disagree		165	10.9	11.9
3	tend to agree		522	34.6	37.7
4	agree completely		526	34.8	38.0
	Sum		1510	100.0	100.0
	Valid Cases		1385		

AT23 - agree with: there is a God who revealed Himself in Jesus (Q41_3H)

Q41.3_at

<SHOW CARD 41.3_at>

How much do you agree or disagree with each of the following opinions?

Q41.3H_at There is a God who revealed Himself in Jesus

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 disagree completely
- 2 tend to disagree
- 3 tend to agree
- 4 agree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT23, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	9	0.6	
-1	dk	М	196	13.0	
1	disagree completely		296	19.6	22.7
2	tend to disagree		276	18.3	21.1
3	tend to agree		393	26.0	30.1
4	agree completely		340	22.5	26.1
	Sum		1510	100.0	100.0
	Valid Cases		1305		

AT24 - agree with: there is not much meaning to life (Q41_3I)

Q41.3_at

<SHOW CARD 41.3_at>

How much do you agree or disagree with each of the following opinions?

Q41.3I_at In my opinion, there is not much meaning to life

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 disagree completely
- 2 tend to disagree
- 3 tend to agree
- 4 agree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT24, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	2	0.1	
-1	dk	М	57	3.8	
1	disagree completely		849	56.2	58.5
2	tend to disagree		376	24.9	25.9
3	tend to agree		113	7.5	7.8
4	agree completely		113	7.5	7.8
	Sum		1510	100.0	100.0
	Valid Cases		1451		

AT25 - agree with: life is just a part of the process of nature/part of the developments in nature (Q41_3J)

Q41.3_at

<SHOW CARD 41.3_at>

How much do you agree or disagree with each of the following opinions?

Q41.3J_at Life is just a part of the process in nature/part of the developments in nature

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 disagree completely
- 2 tend to disagree
- 3 tend to agree
- 4 agree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT25, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	2	0.1	
-1	dk	М	88	5.8	
1	disagree completely		131	8.7	9.2
2	tend to disagree		299	19.8	21.0
3	tend to agree		653	43.2	46.0
4	agree completely		338	22.4	23.8
	Sum		1510	100.1	100.0
	Valid Cases		1420		

AT26 - agree with: the resurrection of Jesus gives meaning to my death (Q41_3K)

Q41.3_at

<SHOW CARD 41.3_at>

How much do you agree or disagree with each of the following opinions?

Q41.3K_at The resurrection of Jesus gives meaning to my death

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 disagree completely
- 2 tend to disagree
- 3 tend to agree
- 4 agree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT26, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	10	0.7	
-1	dk	М	193	12.8	
1	disagree completely		434	28.7	33.2
2	tend to disagree		303	20.1	23.2
3	tend to agree		323	21.4	24.7
4	agree completely		247	16.4	18.9
	Sum		1510	100.0	100.0
	Valid Cases		1308		

AT27 - agree with: the future lies in the kingdom of God that has been promised by Jesus (Q41_3L)

Q41.3_at

<SHOW CARD 41.3_at>

How much do you agree or disagree with each of the following opinions?

Q41.3L_at The future lies in the Kingdom of God that has been promised by Jesus

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 disagree completely
- 2 tend to disagree
- 3 tend to agree
- 4 agree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT27, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	9	0.6	
-1	dk	М	215	14.2	
1	disagree completely		427	28.3	33.2
2	tend to disagree		319	21.1	24.8
3	tend to agree		314	20.8	24.4
4	agree completely		226	15.0	17.6
	Sum		1510	100.0	100.0
	Valid Cases		1286		

AT28 - agree with: death is a transition to another form of existence (Q41_3M)

Q41.3_at

<SHOW CARD 41.3_at>

How much do you agree or disagree with each of the following opinions?

Q41.3M_at Death is a transition to another form of existence

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 disagree completely
- 2 tend to disagree
- 3 tend to agree
- 4 agree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT28, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	5	0.3	
-1	dk	М	219	14.5	
1	disagree completely		285	18.9	22.2
2	tend to disagree		239	15.8	18.6
3	tend to agree		426	28.2	33.1
4	agree completely		336	22.3	26.1
	Sum		1510	100.0	100.0
	Valid Cases		1286		

AT29 - agree with: reincarnation of the soul (rebirth) of the soul in another life exists (Q41_3N)

Q41.3_at

<SHOW CARD 41.3_at>

How much do you agree or disagree with each of the following opinions?

Q41.3N_at Reincarnation of the soul (rebirth) of the soul in another life exists

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 disagree completely
- 2 tend to disagree
- 3 tend to agree
- 4 agree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT29, weighted by weight

Value	Label	Missing Cou		Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	252	16.7	
1	disagree completely		544	36.0	43.4
2	tend to disagree		286	18.9	22.8
3	tend to agree		239	15.8	19.1
4	agree completely		184	12.2	14.7
	Sum		1510	99.9	100.0
	Valid Cases		1254		

AT30 - agree with: after death, everything comes to a final end (Q41_30)

Q41.3_at

<SHOW CARD 41.3_at>

How much do you agree or disagree with each of the following opinions?

Q41.3O_at After death, everything comes to a final end

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 disagree completely
- 2 tend to disagree
- 3 tend to agree
- 4 agree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT30, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	236	15.6	
1	disagree completely		379	25.1	29.8
2	tend to disagree		335	22.2	26.4
3	tend to agree		252	16.7	19.8
4	agree completely		304	20.1	23.9
	Sum		1510	100.0	100.0
	Valid Cases		1270		

AT31 - agree with: the eternal cycle of man, nature and the cosmos is the higher power (Q41_3P_at)

Q41.3_at

<SHOW CARD 41.3_at>

How much do you agree or disagree with each of the following opinions?

Q41.3P_at The eternal cycle of man, nature and the cosmos is the higher power

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 disagree completely
- 2 tend to disagree
- 3 tend to agree
- 4 agree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT31, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	6	0.4	
-1	dk	М	223	14.8	
1	disagree completely		133	8.8	10.4
2	tend to disagree		253	16.8	19.8
3	tend to agree		588	38.9	45.9
4	agree completely		307	20.3	24.0
	Sum		1510	100.0	100.0
	Valid Cases		1280		

AT32 - how often do you think about religious topics? (Q41_4_at)

Q41.4_at

<SHOW CARD 41.4_AT>

How often do you think about religious topics?

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 never
- 2 seldom
- 3 sometimes
- 4 often
- 5 very often

Remarks on original question in the field questionnaire:

AT32, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	2	0.1	
1	never		226	15.0	15.0
2	seldom		433	28.7	28.7
3	sometimes		540	35.8	35.8
4	often		221	14.6	14.6
5	very often		89	5.9	5.9
	Sum		1510	100.1	100.0
	Valid Cases		1508		

AT33 - how often: situation where God or something divine intervenes in your life (Q41.5_at)

Q41.5_at

<SHOW CARD 41.5_at - FOR MEMBERS OF NON-CHRISTIAN DENOMINATION USE VERSION A) OR B)>

How often do you experience situations where you feel that God or something divine intervenes in your life?

- A) How often do you experience situations where you feel that God or something divine causes interventions in your life? (muslim)
- B) How often do you experience situations where you feel that God, other deities or something divine /one or more deities, or something divine intervenes in your life? (buddhist or hindu)
- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 never
- 2 seldom
- 3 sometimes
- 4 often
- 5 very often

Remarks on original question in the field questionnaire:

AT33, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	17	1.1	
1	never		403	26.7	27.0
2	seldom		402	26.6	26.9
3	sometimes		489	32.4	32.8
4	often		141	9.3	9.5
5	very often		57	3.8	3.8
	Sum		1510	99.9	100.0
	Valid Cases		1493		

AT34 - how often: a feeling of unity with everything (Q41.6_at)

Q41.6_at

<SHOW CARD 41.6_AT>

How often do you experience a feeling of unity with everything?

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 never
- 2 seldom
- 3 sometimes
- 4 often
- 5 very often

Remarks on original question in the field questionnaire:

AT34, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	46	3.0	
1	never		289	19.1	19.8
2	seldom		452	29.9	30.9
3	sometimes		531	35.2	36.3
4	often		156	10.3	10.7
5	very often		33	2.2	2.3
	Sum		1510	100.0	100.0
	Valid Cases		1461		

AT35 - how often do you meditate? (Q41.7_at)

Q41.7_at

<SHOW CARD 41.7_AT>

How often do you meditate?

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 serveral times a day
- 2 once a day
- 3 more than once a week
- 4 once a week
- 5 one to three times a month
- 6 several times a year
- 7 more seldom
- 8 never

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT35, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	2	0.1	
-1	dk	М	34	2.3	
1	serveral times a day		28	1.9	1.9
2	once a day		60	4.0	4.1
3	more than once a week		120	7.9	8.1
4	once a week		127	8.4	8.6
5	one to three times a month		108	7.2	7.3
6	several times a year		70	4.6	4.7
7	more seldom		224	14.8	15.2
8	never		737	48.8	50.0
	Sum		1510	100.0	100.0
	Valid Cases		1475		

AT36 - how strongly do you believe that God or something divine exists? (Q41.8_at)

Q41.8_at

<SHOW CARD 41.8_AT - FOR MEMBERS OF NON-CHRISTIAN DENOMINATION USE VERSION A)>

How strongly do you believe that God or something divine exists?

A) How strongly do you believe that God, other deities or something divine /one or more deities, or something divine exists? (buddhist, hindu)

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 not at all
- 2 a little
- 3 about average
- 4 fairly strongly
- 5 very strongly

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT36, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	25	1.7	
1	not at all		220	14.6	14.9
2	a little		254	16.8	17.2
3	about average		362	24.0	24.4
4	fairly strongly		304	20.1	20.5
5	very strongly		341	22.6	23.0
	Sum		1510	100.0	100.0
	Valid Cases		1482		

AT37 - how often: feel that God or something divine wants to tell or show you something\ wants to give you a sign (Q41.9_at)

Q41.9_at

<SHOW CARD 41.9_AT - FOR MEMBERS OF NON-CHRISTIAN DENOMINATION USE VERSION A) OR B)>

How often do you experience situations where you feel that God or something divine wants to tell or show you something / wants to give you a sign?

- A) How often do you experience situations where you feel that God or something divine causes signs or messages to you? (muslim)
- B) How often do you experience situations where you feel that God or something divine / one or more deities, or something divine want to tell or show you something / want to give you a sign? (buddhist, hindu)
- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 never
- 2 seldom
- 3 sometimes
- 4 often
- 5 very often

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT37, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	2	0.1	
-1	dk	М	28	1.9	
1	never		469	31.1	31.7
2	seldom		373	24.7	25.2
3	sometimes		447	29.6	30.2
4	often		130	8.6	8.8
5	very often		62	4.1	4.2
	Sum		1510	100.1	100.0
	Valid Cases		1480		

v136 - important in marriage: faithfulness (Q42A)

THE FOLLOWING QUESTIONS ARE ABOUT FAMILY LIFE AND MARRIAGE

Q42

<SHOW CARD 42 - READ OUT AND CODE ONE ANSWER PER LINE!>

Here is a list of things which some people think make for a successful marriage. Please tell me, for each one, whether you think it is very important, rather important or not very important for a successful marriage?

Q42.A Faithfulness

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very important
- 2 rather important
- 3 not very important

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v136, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	9	0.6	
1	very		1220	80.8	81.3
2	rather		265	17.5	17.7
3	not		16	1.1	1.1
	Sum		1510	100.1	100.0
	Valid Cases		1500		

v137 - important in marriage: adequate income (Q42B)

THE FOLLOWING QUESTIONS ARE ABOUT FAMILY LIFE AND MARRIAGE

Q42

<SHOW CARD 42 - READ OUT AND CODE ONE ANSWER PER LINE!>

Here is a list of things which some people think make for a successful marriage. Please tell me, for each one, whether you think it is very important, rather important or not very important for a successful marriage?

Q42.B An adequate income

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very important
- 2 rather important
- 3 not very important

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v137, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	12	0.8	
1	very		652	43.2	43.6
2	rather		682	45.2	45.6
3	not		163	10.8	10.9
	Sum		1510	99.9	100.0
	Valid Cases		1498		

v138 - important in marriage: same social background (Q42C)

THE FOLLOWING QUESTIONS ARE ABOUT FAMILY LIFE AND MARRIAGE

Q42

<SHOW CARD 42 - READ OUT AND CODE ONE ANSWER PER LINE!>

Here is a list of things which some people think make for a successful marriage. Please tell me, for each one, whether you think it is very important, rather important or not very important for a successful marriage?

Q42.C Being of the same social background

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very important
- 2 rather important
- 3 not very important

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v138, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	15	1.0	
1	very		322	21.3	21.6
2	rather		560	37.1	37.5
3	not		612	40.5	41.0
	Sum		1510	99.9	100.0
	Valid Cases		1495		

v139 - important in marriage: shared religious beliefs (Q42D)

THE FOLLOWING QUESTIONS ARE ABOUT FAMILY LIFE AND MARRIAGE

Q42

<SHOW CARD 42 - READ OUT AND CODE ONE ANSWER PER LINE!>

Here is a list of things which some people think make for a successful marriage. Please tell me, for each one, whether you think it is very important, rather important or not very important for a successful marriage?

Q42.D Shared religious beliefs

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very important
- 2 rather important
- 3 not very important

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v139, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	21	1.4	
1	very		282	18.7	19.0
2	rather		477	31.6	32.1
3	not		729	48.3	49.0
	Sum		1510	99.9	100.0
	Valid Cases		1489		

v140 - important in marriage: good housing (Q42E)

THE FOLLOWING QUESTIONS ARE ABOUT FAMILY LIFE AND MARRIAGE

Q42

<SHOW CARD 42 - READ OUT AND CODE ONE ANSWER PER LINE!>

Here is a list of things which some people think make for a successful marriage. Please tell me, for each one, whether you think it is very important, rather important or not very important for a successful marriage?

Q42.E Good housing

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very important
- 2 rather important
- 3 not very important

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v140, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	6	0.4	
1	very		625	41.4	41.6
2	rather		763	50.5	50.7
3	not		116	7.7	7.7
	Sum		1510	100.0	100.0
	Valid Cases		1504		

v141 - important in marriage: agreement on politics (Q42F)

THE FOLLOWING QUESTIONS ARE ABOUT FAMILY LIFE AND MARRIAGE

Q42

<SHOW CARD 42 - READ OUT AND CODE ONE ANSWER PER LINE!>

Here is a list of things which some people think make for a successful marriage. Please tell me, for each one, whether you think it is very important, rather important or not very important for a successful marriage?

Q42.F Agreement on politics

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very important
- 2 rather important
- 3 not very important

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v141, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	22	1.5	
1	very		175	11.6	11.8
2	rather		411	27.2	27.6
3	not		901	59.7	60.6
	Sum		1510	100.0	100.0
	Valid Cases		1487		

v142 - important in marriage: live apart from in-laws (Q42G)

THE FOLLOWING QUESTIONS ARE ABOUT FAMILY LIFE AND MARRIAGE

Q42

<SHOW CARD 42 - READ OUT AND CODE ONE ANSWER PER LINE!>

Here is a list of things which some people think make for a successful marriage. Please tell me, for each one, whether you think it is very important, rather important or not very important for a successful marriage?

Q42.G Living apart from your in-laws

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very important
- 2 rather important
- 3 not very important

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v142, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	41	2.7	
1	very		650	43.0	44.2
2	rather		498	33.0	33.9
3	not		321	21.3	21.9
	Sum		1510	100.0	100.0
	Valid Cases		1469		

v143 - important in marriage: happy sexual relationship (Q42H)

THE FOLLOWING QUESTIONS ARE ABOUT FAMILY LIFE AND MARRIAGE

Q42

<SHOW CARD 42 - READ OUT AND CODE ONE ANSWER PER LINE!>

Here is a list of things which some people think make for a successful marriage. Please tell me, for each one, whether you think it is very important, rather important or not very important for a successful marriage?

Q42.H Happy sexual relationship

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very important
- 2 rather important
- 3 not very important

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v143, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	10	0.7	
-1	dk	М	9	0.6	
1	very		838	55.5	56.2
2	rather		607	40.2	40.7
3	not		46	3.0	3.1
	Sum		1510	100.0	100.0
	Valid Cases		1491		

v144 - important in marriage: share household chores (Q42I)

THE FOLLOWING QUESTIONS ARE ABOUT FAMILY LIFE AND MARRIAGE

Q42

<SHOW CARD 42 - READ OUT AND CODE ONE ANSWER PER LINE!>

Here is a list of things which some people think make for a successful marriage. Please tell me, for each one, whether you think it is very important, rather important or not very important for a successful marriage?

Q42.I Sharing household chores

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very important
- 2 rather important
- 3 not very important

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v144, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	12	0.8	
1	very		538	35.6	35.9
2	rather		686	45.4	45.8
3	not		274	18.1	18.3
	Sum		1510	100.0	100.0
	Valid Cases		1498		

v145 - important in marriage: children (Q42J)

THE FOLLOWING QUESTIONS ARE ABOUT FAMILY LIFE AND MARRIAGE

Q42

<SHOW CARD 42 - READ OUT AND CODE ONE ANSWER PER LINE!>

Here is a list of things which some people think make for a successful marriage. Please tell me, for each one, whether you think it is very important, rather important or not very important for a successful marriage?

Q42.J Children

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very important
- 2 rather important
- 3 not very important

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v145, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	25	1.7	
1	very		790	52.3	53.2
2	rather		480	31.8	32.3
3	not		214	14.2	14.4
	Sum		1510	100.0	100.0
	Valid Cases		1485		

v146 - important in marriage: discuss problems (Q42K)

THE FOLLOWING QUESTIONS ARE ABOUT FAMILY LIFE AND MARRIAGE

Q42

<SHOW CARD 42 - READ OUT AND CODE ONE ANSWER PER LINE!>

Here is a list of things which some people think make for a successful marriage. Please tell me, for each one, whether you think it is very important, rather important or not very important for a successful marriage?

Q42.K Being willing to discuss the problems that come up between husband and wife

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very important
- 2 rather important
- 3 not very important

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v146, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	7	0.5	
1	very		960	63.6	63.9
2	rather		505	33.4	33.6
3	not		38	2.5	2.5
	Sum		1510	100.1	100.0
	Valid Cases		1502		

v147 - important in marriage: time for friends and personal hobbies (Q42L)

THE FOLLOWING QUESTIONS ARE ABOUT FAMILY LIFE AND MARRIAGE

Q42

<SHOW CARD 42 - READ OUT AND CODE ONE ANSWER PER LINE!>

Here is a list of things which some people think make for a successful marriage. Please tell me, for each one, whether you think it is very important, rather important or not very important for a successful marriage?

Q42.L Having some time for one's own friends and for personal hobbies/activities

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very important
- 2 rather important
- 3 not very important

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item).

v147, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	5	0.3	
1	very		761	50.4	50.5
2	rather		649	43.0	43.1
3	not		96	6.4	6.4
	Sum		1510	100.1	100.0
	Valid Cases		1505		

AT38 - important in marriage: mutual respect and appreciation (Q42M_at)

THE FOLLOWING QUESTIONS ARE ABOUT FAMILY LIFE AND MARRIAGE

Q42

<SHOW CARD 42 - READ OUT AND CODE ONE ANSWER PER LINE!>

Here is a list of things which some people think make for a successful marriage. Please tell me, for each one, whether you think it is very important, rather important or not very important for a successful marriage?

Q42M_at Mutual respect and appreciation

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very important
- 2 rather important
- 3 not very important

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT38, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	4	0.3	
1	very		1227	81.3	81.5
2	rather		266	17.6	17.7
3	not very important		13	0.9	0.9
	Sum		1510	100.1	100.0
	Valid Cases		1505		

AT39 - important in marriage: mutual understanding and tolerance (Q42N_at)

THE FOLLOWING QUESTIONS ARE ABOUT FAMILY LIFE AND MARRIAGE

Q42

<SHOW CARD 42 - READ OUT AND CODE ONE ANSWER PER LINE!>

Here is a list of things which some people think make for a successful marriage. Please tell me, for each one, whether you think it is very important, rather important or not very important for a successful marriage?

Q42N_at Mutual understanding and tolerance

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very important
- 2 rather important
- 3 not very important

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT39, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	6	0.4	
1	very		1180	78.1	78.5
2	rather		313	20.7	20.8
3	not very important		10	0.7	0.7
	Sum		1510	100.0	100.0
	Valid Cases		1503		

AT40 - important in marriage: spending as much time together as possible (Q42O_at)

THE FOLLOWING QUESTIONS ARE ABOUT FAMILY LIFE AND MARRIAGE

Q42

<SHOW CARD 42 - READ OUT AND CODE ONE ANSWER PER LINE!>

Here is a list of things which some people think make for a successful marriage. Please tell me, for each one, whether you think it is very important, rather important or not very important for a successful marriage?

Q42O_at Spending as much time together as possible

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very important
- 2 rather important
- 3 not very important

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT40, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	10	0.7	
1	very		509	33.7	34.0
2	rather		780	51.7	52.0
3	not very important		210	13.9	14.0
	Sum		1510	100.0	100.0
	Valid Cases		1499		

AT41 - important in marriage: talking a lot about mutual interests (Q42P_at)

THE FOLLOWING QUESTIONS ARE ABOUT FAMILY LIFE AND MARRIAGE

Q42

<SHOW CARD 42 - READ OUT AND CODE ONE ANSWER PER LINE!>

Here is a list of things which some people think make for a successful marriage. Please tell me, for each one, whether you think it is very important, rather important or not very important for a successful marriage?

Q42P_at Talking a lot about mutual interests

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very important
- 2 rather important
- 3 not very important

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT41, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	8	0.5	
1	very		678	44.9	45.1
2	rather		731	48.4	48.7
3	not very important		93	6.2	6.2
	Sum		1510	100.0	100.0
	Valid Cases		1502		

AT42 - important in marriage: same ethnic or national background (Q42Q_at)

THE FOLLOWING QUESTIONS ARE ABOUT FAMILY LIFE AND MARRIAGE

Q42

<SHOW CARD 42 - READ OUT AND CODE ONE ANSWER PER LINE!>

Here is a list of things which some people think make for a successful marriage. Please tell me, for each one, whether you think it is very important, rather important or not very important for a successful marriage?

Q42Q_at Same ethnic or national background

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very important
- 2 rather important
- 3 not very important

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT42, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	13	0.9	
1	very		290	19.2	19.4
2	rather		504	33.4	33.7
3	not very important		703	46.6	47.0
	Sum		1510	100.0	100.0
	Valid Cases		1497		

AT43 - important in marriage: shared goals in life (Q42R_at)

THE FOLLOWING QUESTIONS ARE ABOUT FAMILY LIFE AND MARRIAGE

Q42

<SHOW CARD 42 - READ OUT AND CODE ONE ANSWER PER LINE!>

Here is a list of things which some people think make for a successful marriage. Please tell me, for each one, whether you think it is very important, rather important or not very important for a successful marriage?

Q42R_at Shared goals in life

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very important
- 2 rather important
- 3 not very important

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT43, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	9	0.6	
1	very		773	51.2	51.5
2	rather		662	43.8	44.1
3	not very important		66	4.4	4.4
	Sum		1510	100.0	100.0
	Valid Cases		1501		

AT44 - important in marriage: the partner having a secure job (Q42S_at)

THE FOLLOWING QUESTIONS ARE ABOUT FAMILY LIFE AND MARRIAGE

Q42

<SHOW CARD 42 - READ OUT AND CODE ONE ANSWER PER LINE!>

Here is a list of things which some people think make for a successful marriage. Please tell me, for each one, whether you think it is very important, rather important or not very important for a successful marriage?

Q42S_at The partner having a secure job

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very important
- 2 rather important
- 3 not very important

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT44, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	16	1.1	
1	very		642	42.5	43.0
2	rather		625	41.4	41.8
3	not very important		227	15.0	15.2
	Sum		1510	100.1	100.0
	Valid Cases		1493		

v148 - children need both parents to grow up happily (Q43)

Q43

If someone says a child needs a home with both a father and a mother to grow up happily, would you tend to agree or disagree?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 tend to agree
- 2 tend to disagree

v148, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	7	0.5	
-1	dk	М	42	2.8	
1	tend to agree		1176	77.9	80.5
2	tend to disagree		285	18.9	19.5
	Sum		1510	100.0	100.0
	Valid Cases		1460		

v149 - women need children in order to be fulfilled (Q44)

Q44

Do you think that a woman has to have children in order to be fulfilled or is this not necessary?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 needs children
- 2 not necessary

v149, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	6	0.4	
-1	dk	М	156	10.3	
1	needs children		487	32.3	36.1
2	not necessary		861	57.0	63.9
	Sum		1510	100.0	100.0
	Valid Cases		1349		

v150 - marriage is outdated (Q45)

Q45

Do you agree or disagree with the following statement: Marriage is an outdated institution?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree
- 2 disagree

v150, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	85	5.6	
1	agree		429	28.4	30.2
2	disagree		993	65.8	69.8
	Sum		1510	100.0	100.0
	Valid Cases		1422		

v151 - woman single parent, no stable relationship with man (Q46)

Q46

If a woman wants to have a child as a single parent, but she doesn't want to have a stable relationship with a man, do you approve or disapprove?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 approve
- 2 disapprove
- 3 depends (spontaneous)

v151, weighted by weight

, .					
Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	2	0.1	
-1	dk	М	35	2.3	
1	approve		615	40.7	41.8
2	disapprove		492	32.6	33.4
3	depends		366	24.2	24.8
	Sum		1510	100.0	100.0
	Valid Cases		1473		

v152 - men need children in order to be fulfilled (Q47A)

Q47

<SHOW CARD 47 - READ OUT AND CODE ONE ANSWER PER LINE>

How would you feel about the following statements? Do you agree or disagree with them?

Q47.A A man has to have children in order to be fulfilled

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree strongly

v152, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	75	5.0	
1	agree strongly		156	10.3	10.9
2	agree		305	20.2	21.2
3	agree nor disagree		388	25.7	27.0
4	disagree		359	23.8	25.0
5	disagree strongly		228	15.1	15.9
	Sum		1510	100.1	100.0
	Valid Cases		1435		

v153 - long-term relationship necessary to be happy (Q47B)

Q47

<SHOW CARD 47 - READ OUT AND CODE ONE ANSWER PER LINE>

How would you feel about the following statements? Do you agree or disagree with them?

Q47.B A marriage or a long-term stable relationship is necessary to be happy

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree strongly

v153, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	29	1.9	
1	agree strongly		246	16.3	16.6
2	agree		474	31.4	32.0
3	agree nor disagree		382	25.3	25.8
4	disagree		256	17.0	17.3
5	disagree strongly		124	8.2	8.4
	Sum		1510	100.1	100.0
	Valid Cases		1481		

v154 - homosex couples - adopt children (Q47C)

Q47

<SHOW CARD 47 - READ OUT AND CODE ONE ANSWER PER LINE>

How would you feel about the following statements? Do you agree or disagree with them?

Q47.C Homosexual couples should be able to adopt children

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree strongly

v154, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	70	4.6	
1	agree strongly		138	9.1	9.6
2	agree		285	18.9	19.8
3	agree nor disagree		258	17.1	18.0
4	disagree		324	21.5	22.6
5	disagree strongly		431	28.5	30.0
	Sum		1510	100.0	100.0
	Valid Cases		1436		

v155 - It is alright to live together without getting married (Q47D)

Q47

<SHOW CARD 47 - READ OUT AND CODE ONE ANSWER PER LINE>

How would you feel about the following statements? Do you agree or disagree with them?

Q47.D It is alright for two people to live together without getting married

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree strongly

v155, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	2	0.1	
-1	dk	М	15	1.0	
1	agree strongly		777	51.5	52.0
2	agree		434	28.7	29.1
3	agree nor disagree		195	12.9	13.1
4	disagree		70	4.6	4.7
5	disagree strongly		17	1.1	1.1
	Sum		1510	100.0	100.0
	Valid Cases		1493		

v156 - duty towards society to have children (Q47E)

Q47

<SHOW CARD 47 - READ OUT AND CODE ONE ANSWER PER LINE>

How would you feel about the following statements? Do you agree or disagree with them?

Q47.E It is a duty towards society to have children

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree strongly

v156, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	25	1.7	
1	agree strongly		95	6.3	6.4
2	agree		270	17.9	18.2
3	agree nor disagree		372	24.6	25.1
4	disagree		412	27.3	27.8
5	disagree strongly		335	22.2	22.6
	Sum		1510	100.0	100.0
	Valid Cases		1484		

v157 - people should decide themselves to have children (Q47F)

Q47

<SHOW CARD 47 - READ OUT AND CODE ONE ANSWER PER LINE>

How would you feel about the following statements? Do you agree or disagree with them?

Q47.F People should decide for themselves whether to have children or not

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree strongly

v157, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	8	0.5	
1	agree strongly		997	66.0	66.5
2	agree		363	24.0	24.2
3	agree nor disagree		124	8.2	8.3
4	disagree		9	0.6	0.6
5	disagree strongly		7	0.5	0.5
	Sum		1510	99.9	100.0
	Valid Cases		1501		

v158 - It is childs duty to take care of ill parent (Q47G)

Q47

<SHOW CARD 47 - READ OUT AND CODE ONE ANSWER PER LINE>

How would you feel about the following statements? Do you agree or disagree with them?

Q47.G When a parent is seriously ill or fragile, it is mainly the adult child's duty to take care of him/her

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree strongly

v158, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	32	2.1	
1	agree strongly		296	19.6	20.0
2	agree		551	36.5	37.3
3	agree nor disagree		380	25.2	25.7
4	disagree		176	11.7	11.9
5	disagree strongly		74	4.9	5.0
	Sum		1510	100.0	100.0
	Valid Cases		1477		

v159 - working mother warm relationship with children (Q48A)

Q48

<SHOW CARD 48 - READ OUT AND CODE ONE ANSWER PER LINE>

People talk about the changing roles of men and women today. For each of the following statements I read out, can you tell me how much you agree with each. Please use the responses on this card.

Q48.A A working mother can establish just as warm and secure a relationship with her children as a mother who does not work

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 disagree
- 4 disagree strongly

v159, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	65	4.3	
1	agree strongly		475	31.5	32.9
2	agree		561	37.2	38.9
3	disagree		294	19.5	20.4
4	disagree strongly		112	7.4	7.8
	Sum		1510	100.0	100.0
	Valid Cases		1442		

v160 - pre-school child suffers with working mother (Q48B)

Q48

<SHOW CARD 48 - READ OUT AND CODE ONE ANSWER PER LINE>

People talk about the changing roles of men and women today. For each of the following statements I read out, can you tell me how much you agree with each. Please use the responses on this card.

Q48.B A pre-school child is likely to suffer if his or her mother works

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 disagree
- 4 disagree strongly

v160, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	65	4.3	
1	agree strongly		372	24.6	25.8
2	agree		560	37.1	38.8
3	disagree		387	25.6	26.8
4	disagree strongly		123	8.1	8.5
	Sum		1510	100.0	100.0
	Valid Cases		1442		

v161 - women really want home and children (Q48C)

Q48

<SHOW CARD 48 - READ OUT AND CODE ONE ANSWER PER LINE>

People talk about the changing roles of men and women today. For each of the following statements I read out, can you tell me how much you agree with each. Please use the responses on this card.

Q48.C A job is alright but what most women really want is a home and children

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 disagree
- 4 disagree strongly

v161, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	6	0.4	
-1	dk	М	167	11.1	
1	agree strongly		155	10.3	11.6
2	agree		500	33.1	37.4
3	disagree		478	31.7	35.8
4	disagree strongly		204	13.5	15.3
	Sum		1510	100.0	100.0
	Valid Cases		1337		

v162 - being housewife as fulfilling as paid job (Q48D)

Q48

<SHOW CARD 48 - READ OUT AND CODE ONE ANSWER PER LINE>

People talk about the changing roles of men and women today. For each of the following statements I read out, can you tell me how much you agree with each. Please use the responses on this card.

Q48.D Being a housewife is just as fulfilling as working for pay

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 disagree
- 4 disagree strongly

v162, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	8	0.5	
-1	dk	М	184	12.2	
1	agree strongly		267	17.7	20.3
2	agree		497	32.9	37.7
3	disagree		418	27.7	31.7
4	disagree strongly		136	9.0	10.3
	Sum		1510	100.0	100.0
	Valid Cases		1318		

v163 - job best way for independence women (Q48E)

Q48

<SHOW CARD 48 - READ OUT AND CODE ONE ANSWER PER LINE>

People talk about the changing roles of men and women today. For each of the following statements I read out, can you tell me how much you agree with each. Please use the responses on this card.

Q48.E Having a job is the best way for a woman to be an independent person

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 disagree
- 4 disagree strongly

v163, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	5	0.3	
-1	dk	М	85	5.6	
1	agree strongly		557	36.9	39.2
2	agree		646	42.8	45.5
3	disagree		182	12.1	12.8
4	disagree strongly		35	2.3	2.5
	Sum		1510	100.0	100.0
	Valid Cases		1420		

v164 - husband+wife contribute to household income (Q48F)

Q48

<SHOW CARD 48 - READ OUT AND CODE ONE ANSWER PER LINE>

People talk about the changing roles of men and women today. For each of the following statements I read out, can you tell me how much you agree with each. Please use the responses on this card.

Q48.F Both the husband and wife should contribute to household income

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 disagree
- 4 disagree strongly

v164, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	95	6.3	
1	agree strongly		483	32.0	34.2
2	agree		687	45.5	48.7
3	disagree		220	14.6	15.6
4	disagree strongly		22	1.5	1.6
	Sum		1510	100.0	100.0
	Valid Cases		1412		

v165 - fathers as well suited to look after children as mothers (Q48G)

Q48

<SHOW CARD 48 - READ OUT AND CODE ONE ANSWER PER LINE>

People talk about the changing roles of men and women today. For each of the following statements I read out, can you tell me how much you agree with each. Please use the responses on this card.

Q48.G In general, fathers are as well suited to look after their children as mothers

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 disagree
- 4 disagree strongly

v165, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	2	0.1	
-1	dk	М	82	5.4	
1	agree strongly		440	29.1	30.8
2	agree		684	45.3	47.9
3	disagree		264	17.5	18.5
4	disagree strongly		39	2.6	2.7
	Sum		1510	100.1	100.0
	Valid Cases		1426		

v166 - men should take the same responsibility for home and children (Q48H)

Q48

<SHOW CARD 48 - READ OUT AND CODE ONE ANSWER PER LINE>

People talk about the changing roles of men and women today. For each of the following statements I read out, can you tell me how much you agree with each. Please use the responses on this card.

Q48.H Men should take as much responsibility as women for the home and children

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 disagree
- 4 disagree strongly

v166, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	2	0.1	
-1	dk	М	50	3.3	
1	agree strongly		565	37.4	38.8
2	agree		668	44.2	45.8
3	disagree		197	13.0	13.5
4	disagree strongly		27	1.8	1.9
	Sum		1510	99.9	100.0
	Valid Cases		1458		

v167 - love and respect parents always/earned (Q49)

Q49

<SHOW CARD 49 - READ OUT AND CODE ONE ANSWER ONLY>

Which of these two statements do you tend to agree with?

- A: Regardless of what the qualities and faults of one's parents are, one must always love and respect them;
- B: One does not have the duty to respect and love parents who have not earned it by their behaviour and attitudes
- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 Tend to agree with statement A
- 2 Tend to agree with statement B

v167, weighted by weight

, , , ,				
Value Label	Missing	Count	Percent	Valid Percent
-2 na	M	6	0.4	
-1 dk	M	78	5.2	
1 always love parents		754	49.9	52.9
2 parents have to earn respec	t	672	44.5	47.1
Sum		1510	100.0	100.0
Valid Cases		1426		

v168 - parents responsibilities to their children at expense of/not sacrifice own well-being (Q50)

Q50

<SHOW CARD 50 - READ OUT AND CODE ONE ANSWER ONLY>

Which of the following statements best describes your views about parents', responsibilities to their children?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 Parents' duty is to do their best for their children even at the expense of their own well-being
- 2 Parents have a life of their own and should not be asked to sacrifice their own well-being for the sake of their children
- 3 neither (spontaneous)

v168, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	26	1.7	
1	do utmost best for children		862	57.1	58.1
2	no sacrifice of well-being		473	31.3	31.9
3	neither		148	9.8	10.0
	Sum		1510	100.0	100.0
	Valid Cases		1483		

v169 - children responsibilities to their parents in need at expense of/not sacrifice own well-being (Q51)

Q51

<SHOW CARD 51 - READ OUT AND CODE ONE ANSWER ONLY>

Which of the following statements best describes your views about responsibilities of adult children towards their parents when their parents are in need of long-term care?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 Adult children have the duty to provide long-term care for their parents even at the expense of their own well-being
- 2 Adult children have a life of their own and should not be asked to sacrifice their own well-being for the sake of their parents
- 3 neither (spontaneous)

v169, weighted by weight

Value Label	Missing	Count	Percent	Valid Percent
-2 na	М	2	0.1	
-1 dk	М	45	3.0	
1 duty to provide long-term care		537	35.6	36.7
2 no sacrifice of well-being		714	47.3	48.8
3 neither		212	14.0	14.5
Sum		1510	100.0	100.0
Valid Cases		1463		

v170 - learn children at home: good manners (Q52A)

Q52

<SHOW CARD 52>

Here is a list of qualities which children can be encouraged to learn at home. Which, if any, do you consider to be especially important?

Please choose up to five!

<CODE NOT MORE THAN FIVE!>

Q52.A Good manners

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

v170, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	10	0.7	
1	mentioned		1161	76.9	77.6
2	not mentioned		336	22.3	22.4
	Sum		1510	100.0	100.0
	Valid Cases		1497		

v171 - learn children at home: independence (Q52B)

Q52

<SHOW CARD 52>

Here is a list of qualities which children can be encouraged to learn at home. Which, if any, do you consider to be especially important?

Please choose up to five!

<CODE NOT MORE THAN FIVE!>

Q52.B Independence

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

v171, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	5	0.3	
-1	dk	М	13	0.9	
1	mentioned		968	64.1	64.9
2	not mentioned		524	34.7	35.1
	Sum		1510	100.0	100.0
	Valid Cases		1492		

v172 - learn children at home: hard work (Q52C)

Q52

<SHOW CARD 52>

Here is a list of qualities which children can be encouraged to learn at home. Which, if any, do you consider to be especially important?

Please choose up to five!

<CODE NOT MORE THAN FIVE!>

Q52.C Hard work

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

v172, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	13	0.9	
-1	dk	М	32	2.1	
1	mentioned		192	12.7	13.1
2	not mentioned		1273	84.3	86.9
	Sum		1510	100.0	100.0
	Valid Cases		1465		

v173 - learn children at home: feeling of responsibility (Q52D)

Q52

<SHOW CARD 52>

Here is a list of qualities which children can be encouraged to learn at home. Which, if any, do you consider to be especially important?

Please choose up to five!

<CODE NOT MORE THAN FIVE!>

Q52.D Feeling of responsibility

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

v173, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	7	0.5	
1	mentioned		1175	77.8	78.4
2	not mentioned		324	21.5	21.6
	Sum		1510	100.0	100.0
	Valid Cases		1499		

v174 - learn children at home: imagination (Q52E)

Q52

<SHOW CARD 52>

Here is a list of qualities which children can be encouraged to learn at home. Which, if any, do you consider to be especially important?

Please choose up to five!

<CODE NOT MORE THAN FIVE!>

Q52.E Imagination

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

v174, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	10	0.7	
-1	dk	М	30	2.0	
1	mentioned		308	20.4	21.0
2	not mentioned		1161	76.9	79.0
	Sum		1510	99.9	100.0
	Valid Cases		1469		

v175 - learn children at home: tolerance+respect (Q52F)

Q52

<SHOW CARD 52>

Here is a list of qualities which children can be encouraged to learn at home. Which, if any, do you consider to be especially important?

Please choose up to five!

<CODE NOT MORE THAN FIVE!>

Q52.F Tolerance and respect for other people

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

v175, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	7	0.5	
-1	dk	М	11	0.7	
1	mentioned		1018	67.4	68.2
2	not mentioned		474	31.4	31.8
	Sum		1510	100.0	100.0
	Valid Cases		1492		

v176 - learn children at home: thrift (Q52G)

Q52

<SHOW CARD 52>

Here is a list of qualities which children can be encouraged to learn at home. Which, if any, do you consider to be especially important?

Please choose up to five!

<CODE NOT MORE THAN FIVE!>

Q52.G Thrift, saving money and things

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

v176, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	12	0.8	
-1	dk	М	13	0.9	
1	mentioned		644	42.6	43.4
2	not mentioned		841	55.7	56.6
	Sum		1510	100.0	100.0
	Valid Cases		1485		

v177 - learn children at home: determination, perseverance (Q52H)

Q52

<SHOW CARD 52>

Here is a list of qualities which children can be encouraged to learn at home. Which, if any, do you consider to be especially important?

Please choose up to five!

<CODE NOT MORE THAN FIVE!>

Q52.H Determination, perseverance

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

v177, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	18	1.2	
-1	dk	М	25	1.7	
1	mentioned		450	29.8	30.7
2	not mentioned		1017	67.4	69.3
	Sum		1510	100.0	100.0
	Valid Cases		1467		

v178 - learn children at home: religious faith (Q52I)

Q52

<SHOW CARD 52>

Here is a list of qualities which children can be encouraged to learn at home. Which, if any, do you consider to be especially important?

Please choose up to five!

<CODE NOT MORE THAN FIVE!>

Q52.I Religious faith

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

v178, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	22	1.5	
-1	dk	М	29	1.9	
1	mentioned		256	17.0	17.5
2	not mentioned		1203	79.7	82.5
	Sum		1510	100.0	100.0
	Valid Cases		1459		

v179 - learn children at home: unselfishness (Q52J)

Q52

<SHOW CARD 52>

Here is a list of qualities which children can be encouraged to learn at home. Which, if any, do you consider to be especially important?

Please choose up to five!

<CODE NOT MORE THAN FIVE!>

Q52.J Unselfishness

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

v179, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	21	1.4	
-1	dk	М	30	2.0	
1	mentioned		144	9.5	9.9
2	not mentioned		1315	87.1	90.1
	Sum		1510	100.0	100.0
	Valid Cases		1459		

f180 - flag variable: learn children at home (Q52)

Flag variable:

Indicates inconsistencies for variables v170-v180.

If respondent named more than the allowed 5 qualities.

if (v170 to v180=1) > 5 then f180=1

- 0 Limitation followed
- 1 Limitation ignored

Note:

For detailed information, see overview flag variables.

f180

Value Label	Missing Count	Percent	Valid Percent
0 Limitation followed	1510	100.0	100.0
Sum	1510	100.0	100.0
Valid Cases	1510		

v180 - learn children at home: obedience (Q52K)

Q52

<SHOW CARD 52>

Here is a list of qualities which children can be encouraged to learn at home. Which, if any, do you consider to be especially important?

Please choose up to five!

<CODE NOT MORE THAN FIVE!>

Q52.K Obedience

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

v180, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	19	1.3	
-1	dk	М	27	1.8	
1	mentioned		205	13.6	14.0
2	not mentioned		1259	83.4	86.0
	Sum		1510	100.0	100.0
	Valid Cases		1464		

f181 - flag variable: learn children at home: none (Q52)

Flag variable:

Indicates inconsistencies for the variables v170-v180.

Inconsistent 1

If respondent mentiones at least one quality and "none".

if v181=1 and any of v170 to v180=1 then f181=1

Inconsistent 2

If respondent does not know for at least one quality whether it is important and mentiones "none".

if v181=1 and none of v170 to v180=1 and any of v170 to v180=8 then f181=2

Inconsistent 3

If respondent does not know for at least one quality whether it is important and does not mention "none".

if v181=2 and none of v170 to v180=1 and any of v170 to v180=8 then f181=3

Inconsistent 4

If respondent does not mention any quality and does not mention "none".

if v181=2 and all of v170 to v180=2 then f181=4

Inconsistent 5

If respondent mentions at least one quality and does not know whether "none" is important.

if v181=8 and any of v170 to v180=1 then f181=5

Inconsistent 6

If respondent does not mention any quality and does not know whether "none" is important.

if v181=8 and all of v170 to v180=2 then f181=6

Inconsistent 7

If respondent mentions at least one quality and does not answer whether "none" is important.

if v181=9 and any of v170 to v180=1 then f181=7

Inconsistent 8

If respondent does not mention any quality and does not answer whether "none" is important.

if v181=9 and all of v170 to v180=2 then f181=8

Note:

For detailed information, see overview flag variables.

f181

Value	Label	Missing Count	Percent	Valid Percent
0	consistent	1510	100.0	100.0
	Sum	1510	100.0	100.0
	Valid Cases	1510		

v181 - learn children at home: none (Q52)

Q52

<SHOW CARD 52>

Here is a list of qualities which children can be encouraged to learn at home. Which, if any, do you consider to be especially important?

Please choose up to five!

<CODE NOT MORE THAN FIVE!>

None (spontaneous)

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 mentioned
- 2 not mentioned

Note:

Multiple response questions are split into a set of dichotomized variables (1 mentioned, 2 not mentioned) which represent each of the original categories and are identified by the respective original category code as question number suffix.

v181, weighted by weight

Value Label	Missing Count	Percent	Valid Percent
1 mentioned	122	8.1	8.1
2 not mentioned	1388	91.9	91.9
Sum	1510	100.0	100.0
Valid Cases	1510		

v184 - abortion if woman not married approve/disapprove (Q53A)

Q53

<SHOW CARD 53 - READ OUT AND CODE ONE ANSWER PER LINE>

Do you approve or disapprove of abortion under the following circumstances?

Q53.A Where the woman is not married

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 approve
- 2 disapprove

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item)

v184, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	13	0.9	
-1	dk	М	159	10.5	
1	approve		567	37.5	42.4
2	disapprove		771	51.1	57.6
	Sum		1510	100.0	100.0
	Valid Cases		1338		

v185 - abortion if couple doesnt want more children approve/disapprove (Q53B)

Q53

<SHOW CARD 53 - READ OUT AND CODE ONE ANSWER PER LINE>

Do you approve or disapprove of abortion under the following circumstances?

Q53.B Where a married couple does not want to have any more children

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 approve
- 2 disapprove

Remarks on original question in the field questionnaire:

Deviation in show card (includes additional country-specific item)

v185, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	12	0.8	
-1	dk	М	129	8.5	
1	approve		589	39.0	43.0
2	disapprove		780	51.7	57.0
	Sum		1510	100.0	100.0
	Valid Cases		1369		

AT45 - abortion if the mothers life is endangered by the pregnancy (Q53C_at)

Q53

<SHOW CARD 53 - READ OUT AND CODE ONE ANSWER PER LINE>

Do you approve or disapprove of abortion under the following circumstances?

Q53C_at If the mother's life is endangered by the pregnancy

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 approve
- 2 disapprove

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT45, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	18	1.2	
-1	dk	М	89	5.9	
1	approve		1272	84.2	90.6
2	disapprove		132	8.7	9.4
	Sum		1510	100.1	100.0
	Valid Cases		1403		

AT46 - abortion if it is likely that the child will be handicapped (Q53D_at)

Q53

<SHOW CARD 53 - READ OUT AND CODE ONE ANSWER PER LINE>

Do you approve or disapprove of abortion under the following circumstances?

Q53D_at If it is likely that the child will be handicapped

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 approve
- 2 disapprove

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT46, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	18	1.2	
-1	dk	М	156	10.3	
1	approve		1056	69.9	79.0
2	disapprove		280	18.5	21.0
	Sum		1510	100.0	100.0
	Valid Cases		1336		

v186 - how interested are you in politics (Q54)

NOW WE ARE GOING TO ASK QUESTIONS ABOUT CONTEMPORARY SOCIAL ISSUES

Q54

<SHOW CARD 54>

How interested would you say you are in politics?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very interested
- 2 somewhat interested
- 3 not very interested
- 4 not at all interested

v186, weighted by weight

	• •				
Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	4	0.3	
1	very interested		305	20.2	20.3
2	somewhat interested		597	39.5	39.6
3	not very interested		377	25.0	25.0
4	not at all interested		227	15.0	15.1
	Sum		1510	100.0	100.0
	Valid Cases		1506		

v187 - signing a petition (Q55A)

Q55

<SHOW CARD 55 - READ OUT - CODE ONE ANSWER PER LINE>

Now I'd like you to look at this card. I'm going to read out some different forms of political action that people can take, and I'd like you to tell me, for each one, whether you have actually done any of these things, whether you might do it or would never, under any circumstances, do it

Q55.A Signing a petition

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 have done
- 2 might do
- 3 would never do

v187, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	48	3.2	
1	have done		716	47.4	49.0
2	might do		440	29.1	30.1
3	would never do		304	20.1	20.8
	Sum		1510	100.1	100.0
	Valid Cases		1459		

v188 - joining in boycotts (Q55B)

Q55

<SHOW CARD 55 - READ OUT - CODE ONE ANSWER PER LINE>

Now I'd like you to look at this card. I'm going to read out some different forms of political action that people can take, and I'd like you to tell me, for each one, whether you have actually done any of these things, whether you might do it or would never, under any circumstances, do it

Q55.B Joining in boycotts

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 have done
- 2 might do
- 3 would never do

v188, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	97	6.4	
1	have done		133	8.8	9.4
2	might do		528	35.0	37.5
3	would never do		748	49.5	53.1
	Sum		1510	100.0	100.0
	Valid Cases		1409		

v189 - attending lawful demonstrations (Q55C)

Q55

<SHOW CARD 55 - READ OUT - CODE ONE ANSWER PER LINE>

Now I'd like you to look at this card. I'm going to read out some different forms of political action that people can take, and I'd like you to tell me, for each one, whether you have actually done any of these things, whether you might do it or would never, under any circumstances, do it

Q55.C Attending lawful demonstrations

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 have done
- 2 might do
- 3 would never do

v189, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	67	4.4	
1	have done		233	15.4	16.2
2	might do		508	33.6	35.3
3	would never do		697	46.2	48.5
	Sum		1510	99.9	100.0
	Valid Cases		1438		

v190 - joining unofficial strikes (Q55D)

Q55

<SHOW CARD 55 - READ OUT - CODE ONE ANSWER PER LINE>

Now I'd like you to look at this card. I'm going to read out some different forms of political action that people can take, and I'd like you to tell me, for each one, whether you have actually done any of these things, whether you might do it or would never, under any circumstances, do it.

Q55.D Joining unofficial strikes

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 have done
- 2 might do
- 3 would never do

v190, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	5	0.3	
-1	dk	М	97	6.4	
1	have done		60	4.0	4.3
2	might do		333	22.1	23.6
3	would never do		1016	67.3	72.1
	Sum		1510	100.1	100.0
	Valid Cases		1408		

v191 - occupying buildings/factories (Q55E)

Q55

<SHOW CARD 55 - READ OUT - CODE ONE ANSWER PER LINE>

Now I'd like you to look at this card. I'm going to read out some different forms of political action that people can take, and I'd like you to tell me, for each one, whether you have actually done any of these things, whether you might do it or would never, under any circumstances, do it

Q55.E Occupying buildings or factories

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 have done
- 2 might do
- 3 would never do

v191, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	85	5.6	
1	have done		34	2.3	2.4
2	might do		206	13.6	14.5
3	would never do		1181	78.2	83.1
	Sum		1510	100.0	100.0
	Valid Cases		1421		

v192 - more important: freedom or equality (Q56)

Q56

<SHOW CARD 56 - READ OUT AND CODE ONE ANSWER ONLY>

Which of these two statements comes closest to your own opinion?

A: I find that both freedom and equality are important. But if I were to choose one or the other, I would consider personal freedom more important, that is, everyone can live in freedom and develop without hindrance

B: Certainly both freedom and equality are important. But if I were to choose one or the other, I would consider equality more important, that is, that nobody is underprivileged and that social class differences are not so strong

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 Agree with statement A
- 2 Agree with statement B
- 3 neither (spontaneous)

v192, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	67	4.4	
1	freedom above equality		725	48.0	50.4
2	equality above freedom		570	37.7	39.6
3	neither		143	9.5	9.9
	Sum		1510	99.9	100.0
	Valid Cases		1439		

v193 - political view: left-right (Q57)

Q57

<SHOW CARD 57>

In political matters, people talk of 'the left' and the 'the right'. How would you place your views on this scale, generally speaking?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 left
- 2
- 3
- 4
- 5
- 6
- 7
- 8
- 9
- 10 right

v193, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	78	5.2	
-1	dk	М	184	12.2	
1	left		42	2.8	3.4
2			39	2.6	3.1
3			161	10.7	12.9
4			120	7.9	9.6
5			464	30.7	37.1
6			144	9.5	11.5
7			108	7.2	8.6
8			88	5.8	7.0
9			23	1.5	1.8
10	right		61	4.0	4.9
	Sum		1510	100.1	100.0
	Valid Cases		1249		

v194 - individual-state responsibility for providing (Q58A)

Q58A

<SHOW CARD 58 - READ OUT>

On this card you see a number of opposite views on various issues. How would you place your views on this scale?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 Individuals should take more responsibility for providing for themselves

2

4

5

6

7

8 9

10 The state should take more responsibility to ensure that everyone is provided for

v194, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	2	0.1	
-1	dk	М	24	1.6	
1	individual responsibility		349	23.1	23.5
2			73	4.8	4.9
3			247	16.4	16.6
4			109	7.2	7.3
5			311	20.6	21.0
6			110	7.3	7.4
7			113	7.5	7.6
8			91	6.0	6.1
9			17	1.1	1.1
10	state responsibility		64	4.2	4.3
	Sum		1510	100.0	100.0
	Valid Cases		1484		

v195 - take any job-right to refuse job when unemployed (Q58B)

Q58B

<SHOW CARD 58 - READ OUT>

On this card you see a number of opposite views on various issues. How would you place your views on this scale?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 People who are unemployed should have to take any job available or lose their unemployment benefits

2 3 4

5 6

7

8

10 People who are unemployed should have the right to refuse a job they do not want

v195, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	12	0.8	
1	unemployed take any job		378	25.0	25.3
2			75	5.0	5.0
3			285	18.9	19.0
4			109	7.2	7.3
5			247	16.4	16.5
6			113	7.5	7.5
7			106	7.0	7.1
8			77	5.1	5.1
9			15	1.0	1.0
10	unemployed right to refuse a job		92	6.1	6.1
	Sum		1510	99.9	100.0
	Valid Cases		1498		

v196 - competition good-harmful for people (Q58C)

Q58C

<SHOW CARD 58 - READ OUT>

On this card you see a number of opposite views on various issues. How would you place your views on this scale?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 Competition is good. It stimulates people to work hard and develop new ideas

2

4

5

6 7

8

9

10 Competition is harmful, it brings out the worst in people

v196, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	36	2.4	
1	competition good		256	17.0	17.4
2			105	7.0	7.1
3			281	18.6	19.1
4			164	10.9	11.2
5			355	23.5	24.1
6			121	8.0	8.2
7			106	7.0	7.2
8			45	3.0	3.1
9			10	0.7	0.7
10	competition harmful		27	1.8	1.8
	Sum		1510	99.9	100.0
	Valid Cases		1471		

v197 - state give more freedom to-control firms more effectively (Q58D)

Q58D

<SHOW CARD 58 - READ OUT>

On this card you see a number of opposite views on various issues. How would you place your views on this scale?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 The state should give more freedom to firms

2

3

5

6

7

8 9

10 The state should control firms more effectively

v197, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	62	4.1	
1	state gives freedom to firms		248	16.4	17.1
2			72	4.8	5.0
3			208	13.8	14.4
4			104	6.9	7.2
5			311	20.6	21.5
6			125	8.3	8.6
7			150	9.9	10.4
8			110	7.3	7.6
9			35	2.3	2.4
10	state control firms		84	5.6	5.8
	Sum		1510	100.0	100.0
	Valid Cases		1447		

v198 - equalize incomes-incentives for individual effort (Q58E)

Q58E

<SHOW CARD 58 - READ OUT>

On this card you see a number of opposite views on various issues. How would you place your views on this scale?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 Incomes should be made more equal

2

3

4

5 6

7

8 9

10 There should be greater incentives for individual effort

v198, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	34	2.3	
1	incomes more equal		428	28.3	29.1
2			103	6.8	7.0
3			298	19.7	20.2
4			131	8.7	8.9
5			266	17.6	18.1
6			97	6.4	6.6
7			70	4.6	4.8
8			39	2.6	2.6
9			13	0.9	0.9
10	incentives to individual efforts		27	1.8	1.8
	Sum		1510	100.0	100.0
	Valid Cases		1473		

v199 - private-government ownership business (Q58F)

Q58F

<SHOW CARD 58 - READ OUT>

On this card you see a number of opposite views on various issues. How would you place your views on this scale?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 Private ownership of business and industry should be increased

2

4

5

6

7 8

9

10 Government ownership of business and industry should be increased

v199, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	149	9.9	
1	private ownership increased		213	14.1	15.7
2			75	5.0	5.5
3			156	10.3	11.5
4			91	6.0	6.7
5			449	29.7	33.1
6			128	8.5	9.4
7			110	7.3	8.1
8			72	4.8	5.3
9			12	0.8	0.9
10	government ownership increased		50	3.3	3.7
	Sum		1510	99.9	100.0
	Valid Cases		1357		

v200 - opinion on society (Q59)

Q59

<SHOW CARD 59 - READ OUT AND CODE ONE ANSWER ONLY>

On this card are three basic kinds of attitudes vis à vis the society we live in. Please choose the one which best describes your own opinion.

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 The entire way our society is organized must be radically changed by revolutionary action
- 2 Our society must be gradually changed by reforms
- 3 Our present society must be valiantly defended against all changes

v200, weighted by weight

Value Label	Missing	Count	Percent	Valid Percent
-2 na	М	8	0.5	
-1 dk	М	84	5.6	
1 radically changed by revolutionary action		68	4.5	4.8
2 gradually changed by reforms		1096	72.6	77.3
3 defended against all changes		253	16.8	17.9
Sum		1510	99.9	100.0
Valid Cases		1417		

f201 - flag variable: aims of this country - most/second important (Q60/Q61)

Flag variable:

Indicates inconsistencies for the variables v201/v202.

If most important reason is identical with second most important reason.

if v201 = v202 then f201=1

- 0 consistent
- 1 inconsistent

Note:

For detailed information, see overview flag variables.

f201

Value	Label	Missing Count	Percent	Valid Percent
0	consistent	1448	95.9	95.9
1	inconsistent	62	4.1	4.1
	Sum	1510	100.0	100.0
	Valid Cases	1510		

v201 - aims of this country-most important (Q60)

Q60

<SHOW CARD 60 - READ OUT AND CODE ONE ANSWER ONLY FOR Q60>

There is a lot of talk these days about what the aims of this country should be for the next ten years. On this card are listed some of the goals which different people would give top priority. If you had to choose, which of the things on this card would you say is most important?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 maintaining order in the nation
- 2 giving people more say in important government decisions
- 3 fighting rising prices
- 4 protecting freedom of speech

v201, weighted by weight

Value Lab	bel	Missing	Count	Percent	Valid Percent
-2 na		М	1	0.1	
-1 dk		М	18	1.2	
1 ma	aintain order in nation		400	26.5	26.8
2 mo	ore say important government decisions		418	27.7	28.0
3 figh	hting rising prices		487	32.3	32.7
4 pro	otect freedom of speech		186	12.3	12.5
Sui	ım		1510	100.0	100.0
Val	lid Cases		1491		

v202 - aims of this country-2nd most important (Q61)

[Q60: What would you say should be the most important aims of this country for the next ten years?]

Q61

<STILL KEEP SHOW CARD 60 - READ OUT AND CODE ONE ANSWER ONLY FOR Q61>

And which would be the next most important?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 maintaining order in nation
- 2 giving people more say in important government decisions
- 3 fighting rising prices
- 4 protecting freedom of speech

v202, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	21	1.4	
1	maintain order in nation		345	22.8	23.2
2	more say important government decisions		357	23.6	24.0
3	fighting rising prices		465	30.8	31.3
4	protect freedom of speech		319	21.1	21.5
	Sum		1510	100.0	100.0
	Valid Cases		1487		

v203 - good/bad: less emphasis on money (Q62A)

Q62

<SHOW CARD 62 – READ OUT AND CODE ONE ANSWER PER LINE!>

Here are a couple changes in our way of life that might take place in the near future. Please tell me for each one, if it were to happen whether you think it would be a good thing, a bad thing, or don't you mind?

Q62.A Less emphasis on money and material possessions

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 good
- 2 bad
- 3 don't mind

Remarks on original question in the field questionnaire:

Deviation in question text ("a couple changes" instead of "two changes") and in show card (includes additional country-specific item).

v203, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	57	3.8	
1	good		714	47.3	49.2
2	bad		427	28.3	29.4
3	dont mind		309	20.5	21.3
	Sum		1510	100.0	100.0
	Valid Cases		1450		

v204 - good/bad: more respect for authority (Q62B)

Q62

<SHOW CARD 62 – READ OUT AND CODE ONE ANSWER PER LINE!>

Here are a couple changes in our way of life that might take place in the near future. Please tell me for each one, if it were to happen whether you think it would be a good thing, a bad thing, or don't you mind?

Q62.B Greater respect for authority

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 good
- 2 bad
- 3 don't mind

Remarks on original question in the field questionnaire:

Deviation in question text ("a couple changes" instead of "two changes") and in show card (includes additional country-specific item).

v204, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	2	0.1	
-1	dk	М	68	4.5	
1	good		752	49.8	52.3
2	bad		421	27.9	29.3
3	dont mind		266	17.6	18.5
	Sum		1510	99.9	100.0
	Valid Cases		1440		

AT47 - good\bad: decrease the importance of work in our lives (Q62C_at)

Q62

<SHOW CARD 62 – READ OUT AND CODE ONE ANSWER PER LINE!>

Here are a couple changes in our way of life that might take place in the near future. Please tell me for each one, if it were to happen whether you think it would be a good thing, a bad thing, or don't you mind?

Q62C_at Decrease the importance of work in our lives

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 good
- 2 bad
- 3 don't mind

Remarks on original question in the field questionnaire:

AT47, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	6	0.4	
-1	dk	М	53	3.5	
1	good		550	36.4	37.9
2	bad		707	46.8	48.7
3	dont mind		194	12.8	13.4
	Sum		1510	100.0	100.0
	Valid Cases		1451		

AT48 - good\bad: more emphasis on the development of technology (Q62D_at)

Q62

<SHOW CARD 62 – READ OUT AND CODE ONE ANSWER PER LINE!>

Here are a couple changes in our way of life that might take place in the near future. Please tell me for each one, if it were to happen whether you think it would be a good thing, a bad thing, or don't you mind?

Q62D_at More emphasis on the development of technology

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 good
- 2 bad
- 3 don't mind

Remarks on original question in the field questionnaire:

AT48, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	5	0.3	
-1	dk	М	83	5.5	
1	good		727	48.1	51.1
2	bad		342	22.6	24.0
3	dont mind		354	23.4	24.9
	Sum		1510	100.1	100.0
	Valid Cases		1423		

AT49 - good\bad: greater emphasis on the development of the individual (Q62E_at)

Q62

<SHOW CARD 62 – READ OUT AND CODE ONE ANSWER PER LINE!>

Here are a couple changes in our way of life that might take place in the near future. Please tell me for each one, if it were to happen whether you think it would be a good thing, a bad thing, or don't you mind?

Q62E_at Greater emphasis on the development of the individual

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 good
- 2 bad
- 3 don't mind

Remarks on original question in the field questionnaire:

AT49, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	33	2.2	
1	good		1252	82.9	84.9
2	bad		79	5.2	5.4
3	dont mind		143	9.5	9.7
	Sum		1510	100.0	100.0
	Valid Cases		1474		

AT50 - good\bad: more emphasis on family life (Q62F_at)

Q62

<SHOW CARD 62 – READ OUT AND CODE ONE ANSWER PER LINE!>

Here are a couple changes in our way of life that might take place in the near future. Please tell me for each one, if it were to happen whether you think it would be a good thing, a bad thing, or don't you mind?

Q62F_at More emphasis on family life

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 good
- 2 bad
- 3 don't mind

Remarks on original question in the field questionnaire:

AT50, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	30	2.0	
1	good		1314	87.0	89.0
2	bad		42	2.8	2.8
3	dont mind		121	8.0	8.2
	Sum		1510	100.0	100.0
	Valid Cases		1477		

AT51 - good\bad: a simple and more natural lifestyle (Q62G_at)

Q62

<SHOW CARD 62 - READ OUT AND CODE ONE ANSWER PER LINE!>

Here are a couple changes in our way of life that might take place in the near future. Please tell me for each one, if it were to happen whether you think it would be a good thing, a bad thing, or don't you mind?

Q62G_at A simple and more natural lifestyle

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 good
- 2 bad
- 3 don't mind

Remarks on original question in the field questionnaire:

AT51, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	35	2.3	
1	good		1189	78.7	80.8
2	bad		106	7.0	7.2
3	dont mind		177	11.7	12.0
	Sum		1510	100.0	100.0
	Valid Cases		1472		

v205 - how much confidence in: church (Q63A)

Q63

<SHOW CARD 63 – READ OUT AND CODE ONE ANSWER PER LINE!>

Please look at this card and tell me, for each item listed, how much confidence you have in them, is it a great deal, quite a lot, not very much or none at all?

Q63.A The church

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 a great deal
- 2 quite a lot
- 3 not very much
- 4 none at all

v205, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	6	0.4	
-1	dk	М	44	2.9	
1	a great deal		119	7.9	8.2
2	quite a lot		410	27.2	28.1
3	not very much		612	40.5	41.9
4	none at all		319	21.1	21.8
	Sum		1510	100.0	100.0
	Valid Cases		1460		

v206 - how much confidence in: armed forces (Q63B)

Q63

<SHOW CARD 63 – READ OUT AND CODE ONE ANSWER PER LINE!>

Please look at this card and tell me, for each item listed, how much confidence you have in them, is it a great deal, quite a lot, not very much or none at all?

Q63.B The armed forces

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 a great deal
- 2 quite a lot
- 3 not very much
- 4 none at all

v206, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	62	4.1	
1	a great deal		102	6.8	7.1
2	quite a lot		520	34.4	36.0
3	not very much		612	40.5	42.3
4	none at all		212	14.0	14.7
	Sum		1510	100.1	100.0
	Valid Cases		1445		

v207 - how much confidence in: education system (Q63C)

Q63

<SHOW CARD 63 – READ OUT AND CODE ONE ANSWER PER LINE!>

Please look at this card and tell me, for each item listed, how much confidence you have in them, is it a great deal, quite a lot, not very much or none at all?

Q63.C The education system

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 a great deal
- 2 quite a lot
- 3 not very much
- 4 none at all

v207, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	37	2.5	
1	a great deal		161	10.7	10.9
2	quite a lot		807	53.4	54.9
3	not very much		440	29.1	29.9
4	none at all		63	4.2	4.3
	Sum		1510	99.9	100.0
	Valid Cases		1472		

v208 - how much confidence in: the press (Q63D)

Q63

<SHOW CARD 63 – READ OUT AND CODE ONE ANSWER PER LINE!>

Please look at this card and tell me, for each item listed, how much confidence you have in them, is it a great deal, quite a lot, not very much or none at all?

Q63.D The press

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 a great deal
- 2 quite a lot
- 3 not very much
- 4 none at all

v208, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	40	2.6	
1	a great deal		45	3.0	3.1
2	quite a lot		478	31.7	32.5
3	not very much		719	47.6	48.9
4	none at all		227	15.0	15.5
	Sum		1510	100.0	100.0
	Valid Cases		1468		

v209 - how much confidence in: trade unions (Q63E)

Q63

<SHOW CARD 63 – READ OUT AND CODE ONE ANSWER PER LINE!>

Please look at this card and tell me, for each item listed, how much confidence you have in them, is it a great deal, quite a lot, not very much or none at all?

Q63.E Trade unions

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 a great deal
- 2 quite a lot
- 3 not very much
- 4 none at all

v209, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	114	7.5	
1	a great deal		72	4.8	5.2
2	quite a lot		334	22.1	23.9
3	not very much		746	49.4	53.4
4	none at all		244	16.2	17.5
	Sum		1510	100.1	100.0
	Valid Cases		1395		

v210 - how much confidence in: the police (Q63F)

Q63

<SHOW CARD 63 – READ OUT AND CODE ONE ANSWER PER LINE!>

Please look at this card and tell me, for each item listed, how much confidence you have in them, is it a great deal, quite a lot, not very much or none at all?

Q63.F The police

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 a great deal
- 2 quite a lot
- 3 not very much
- 4 none at all

v210, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	27	1.8	
1	a great deal		223	14.8	15.0
2	quite a lot		797	52.8	53.7
3	not very much		383	25.4	25.8
4	none at all		80	5.3	5.4
	Sum		1510	100.0	100.0
	Valid Cases		1483		

v211 - how much confidence in: parliament (Q63G)

Q63

<SHOW CARD 63 – READ OUT AND CODE ONE ANSWER PER LINE!>

Please look at this card and tell me, for each item listed, how much confidence you have in them, is it a great deal, quite a lot, not very much or none at all?

Q63.G Parliament

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 a great deal
- 2 quite a lot
- 3 not very much
- 4 none at all

v211, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	61	4.0	
1	a great deal		64	4.2	4.4
2	quite a lot		377	25.0	26.0
3	not very much		726	48.1	50.1
4	none at all		281	18.6	19.4
	Sum		1510	99.9	100.0
	Valid Cases		1449		

v212 - how much confidence in: civil service (Q63H)

Q63

<SHOW CARD 63 – READ OUT AND CODE ONE ANSWER PER LINE!>

Please look at this card and tell me, for each item listed, how much confidence you have in them, is it a great deal, quite a lot, not very much or none at all?

Q63.H Civil service

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 a great deal
- 2 quite a lot
- 3 not very much
- 4 none at all

v212, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	2	0.1	
-1	dk	М	73	4.8	
1	a great deal		57	3.8	4.0
2	quite a lot		535	35.4	37.3
3	not very much		707	46.8	49.3
4	none at all		136	9.0	9.5
	Sum		1510	100.0	100.0
	Valid Cases		1435		

v213 - how much confidence in: social security system (Q63I)

Q63

<SHOW CARD 63 – READ OUT AND CODE ONE ANSWER PER LINE!>

Please look at this card and tell me, for each item listed, how much confidence you have in them, is it a great deal, quite a lot, not very much or none at all?

Q63.I The social security system

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 a great deal
- 2 quite a lot
- 3 not very much
- 4 none at all

v213, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	53	3.5	
1	a great deal		181	12.0	12.4
2	quite a lot		647	42.8	44.4
3	not very much		535	35.4	36.7
4	none at all		95	6.3	6.5
	Sum		1510	100.1	100.0
	Valid Cases		1457		

v214 - how much confidence in: european union (Q63J)

Q63

<SHOW CARD 63 – READ OUT AND CODE ONE ANSWER PER LINE!>

Please look at this card and tell me, for each item listed, how much confidence you have in them, is it a great deal, quite a lot, not very much or none at all?

Q63.J The European Union

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 a great deal
- 2 quite a lot
- 3 not very much
- 4 none at all

v214, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	64	4.2	
1	a great deal		63	4.2	4.4
2	quite a lot		371	24.6	25.7
3	not very much		676	44.8	46.7
4	none at all		336	22.3	23.2
	Sum		1510	100.0	100.0
	Valid Cases		1446		

v215 - how much confidence in: NATO (Q63K)

Q63

<SHOW CARD 63 – READ OUT AND CODE ONE ANSWER PER LINE!>

Please look at this card and tell me, for each item listed, how much confidence you have in them, is it a great deal, quite a lot, not very much or none at all?

Q63.K NATO

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 a great deal
- 2 quite a lot
- 3 not very much
- 4 none at all

v215, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	6	0.4	
-1	dk	М	147	9.7	
1	a great deal		29	1.9	2.1
2	quite a lot		266	17.6	19.6
3	not very much		670	44.4	49.4
4	none at all		392	26.0	28.9
	Sum		1510	100.0	100.0
	Valid Cases		1357		

v216 - how much confidence in: united nations organisation (Q63L)

Q63

<SHOW CARD 63 – READ OUT AND CODE ONE ANSWER PER LINE!>

Please look at this card and tell me, for each item listed, how much confidence you have in them, is it a great deal, quite a lot, not very much or none at all?

Q63.L United Nations Organization

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 a great deal
- 2 quite a lot
- 3 not very much
- 4 none at all

v216, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	142	9.4	
1	a great deal		57	3.8	4.2
2	quite a lot		391	25.9	28.7
3	not very much		608	40.3	44.6
4	none at all		308	20.4	22.6
	Sum		1510	100.0	100.0
	Valid Cases		1364		

v217 - how much confidence in: health care system (Q63M)

Q63

<SHOW CARD 63 – READ OUT AND CODE ONE ANSWER PER LINE!>

Please look at this card and tell me, for each item listed, how much confidence you have in them, is it a great deal, quite a lot, not very much or none at all?

Q63.M Health care system

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 a great deal
- 2 quite a lot
- 3 not very much
- 4 none at all

v217, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	18	1.2	
1	a great deal		305	20.2	20.4
2	quite a lot		775	51.3	51.9
3	not very much		344	22.8	23.0
4	none at all		69	4.6	4.6
	Sum		1510	100.1	100.0
	Valid Cases		1492		

v218 - how much confidence in: justice system (Q63N)

Q63

<SHOW CARD 63 – READ OUT AND CODE ONE ANSWER PER LINE!>

Please look at this card and tell me, for each item listed, how much confidence you have in them, is it a great deal, quite a lot, not very much or none at all?

Q63.N The justice system

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 a great deal
- 2 quite a lot
- 3 not very much
- 4 none at all

v218, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	41	2.7	
1	a great deal		243	16.1	16.5
2	quite a lot		706	46.8	48.0
3	not very much		426	28.2	29.0
4	none at all		95	6.3	6.5
	Sum		1510	100.1	100.0
	Valid Cases		1469		

v219 - how much confidence in: major companies (Q63O)

Q63

<SHOW CARD 63 – READ OUT AND CODE ONE ANSWER PER LINE!>

Please look at this card and tell me, for each item listed, how much confidence you have in them, is it a great deal, quite a lot, not very much or none at all?

Q63.O Major companies

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 a great deal
- 2 quite a lot
- 3 not very much
- 4 none at all

v219, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	2	0.1	
-1	dk	М	86	5.7	
1	a great deal		48	3.2	3.4
2	quite a lot		356	23.6	25.0
3	not very much		748	49.5	52.6
4	none at all		271	17.9	19.0
	Sum		1510	100.1	100.0
	Valid Cases		1422		

v220 - how much confidence in: environmental organizations (Q63P)

Q63

<SHOW CARD 63 – READ OUT AND CODE ONE ANSWER PER LINE!>

Please look at this card and tell me, for each item listed, how much confidence you have in them, is it a great deal, quite a lot, not very much or none at all?

Q63.P Environmental organizations

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 a great deal
- 2 quite a lot
- 3 not very much
- 4 none at all

v220, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	66	4.4	
1	a great deal		144	9.5	10.0
2	quite a lot		649	43.0	44.9
3	not very much		494	32.7	34.2
4	none at all		157	10.4	10.9
	Sum		1510	100.1	100.0
	Valid Cases		1443		

v221 - how much confidence in: political parties (Q63Q)

Q63

<SHOW CARD 63 – READ OUT AND CODE ONE ANSWER PER LINE!>

Please look at this card and tell me, for each item listed, how much confidence you have in them, is it a great deal, quite a lot, not very much or none at all?

Q63.Q Political parties

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 a great deal
- 2 quite a lot
- 3 not very much
- 4 none at all

v221, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	40	2.6	
1	a great deal		22	1.5	1.5
2	quite a lot		197	13.0	13.4
3	not very much		807	53.4	54.9
4	none at all		445	29.5	30.3
	Sum		1510	100.1	100.0
	Valid Cases		1470		

v222 - how much confidence in: government (Q63R)

Q63

<SHOW CARD 63 – READ OUT AND CODE ONE ANSWER PER LINE!>

Please look at this card and tell me, for each item listed, how much confidence you have in them, is it a great deal, quite a lot, not very much or none at all?

Q63.R Government

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 a great deal
- 2 quite a lot
- 3 not very much
- 4 none at all

v222, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	72	4.8	
1	a great deal		28	1.9	1.9
2	quite a lot		238	15.8	16.5
3	not very much		763	50.5	53.0
4	none at all		410	27.2	28.5
	Sum		1510	100.1	100.0
	Valid Cases		1438		

v223 - are you satisfied with democracy (Q64)

Q64

<SHOW CARD 64>

On the whole are you very satisfied, rather satisfied, not very satisfied or not at all satisfied with the way democracy is developing in our country?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very satisfied
- 2 rather satisfied
- 3 not very satisfied
- 4 not at all satisfied

v223, weighted by weight

	0 , 0				
Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	48	3.2	
1	very satisfied		62	4.1	4.2
2	rather satisfied		717	47.5	49.1
3	not very satisfied		521	34.5	35.7
4	not at all satisfied		159	10.5	10.9
	Sum		1510	100.0	100.0
	Valid Cases		1459		

v224 - view government: very bad-very good (Q65)

Q65

<SHOW CARD 65>

People have different views about the system for governing this country. Here is a scale for rating how well things are going: 1 means very bad; 10 means very good

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very bad
- 2
- 3
- 4
- 5
- 6
- 7
- 8
- 9
- 10 very good

v224, weighted by weight

		_			
Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	36	2.4	
1	very bad		174	11.5	11.8
2			67	4.4	4.6
3			259	17.2	17.6
4			143	9.5	9.7
5			371	24.6	25.2
6			117	7.7	8.0
7			168	11.1	11.4
8			128	8.5	8.7
9			28	1.9	1.9
10	very good		15	1.0	1.0
	Sum		1510	100.0	100.0
	Valid Cases		1470		

v225 - political system: strong leader (Q66A)

Q66

<SHOW CARD 66 - READ OUT AND CODE ONE ANSWER PER LINE>

I'm going to describe various types of political systems and ask what you think about each as a way of governing this country. For each one, would you say it is a very good, fairly good, fairly bad or very bad way of governing this country?

Q66.A Having a strong leader who does not have to bother with parliament and elections

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very good
- 2 fairly good
- 3 fairly bad
- 4 very bad

v225, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	14	0.9	
-1	dk	М	124	8.2	
1	very good		74	4.9	5.4
2	fairly good		237	15.7	17.3
3	fairly bad		282	18.7	20.6
4	very bad		779	51.6	56.8
	Sum		1510	100.0	100.0
	Valid Cases		1372		

v226 - political system: experts making decisions (Q66B)

Q66

<SHOW CARD 66 - READ OUT AND CODE ONE ANSWER PER LINE>

I'm going to describe various types of political systems and ask what you think about each as a way of governing this country. For each one, would you say it is a very good, fairly good, fairly bad or very bad way of governing this country?

Q66.B Having experts, not government, make decisions according to what they think is best for the country

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very good
- 2 fairly good
- 3 fairly bad
- 4 very bad

v226, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	12	0.8	
-1	dk	М	155	10.3	
1	very good		186	12.3	13.8
2	fairly good		590	39.1	43.9
3	fairly bad		344	22.8	25.6
4	very bad		224	14.8	16.7
	Sum		1510	100.1	100.0
	Valid Cases		1344		

v227 - political system: the army ruling (Q66C)

Q66

<SHOW CARD 66 - READ OUT AND CODE ONE ANSWER PER LINE>

I'm going to describe various types of political systems and ask what you think about each as a way of governing this country. For each one, would you say it is a very good, fairly good, fairly bad or very bad way of governing this country?

Q66.C Having the army rule the country

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very good
- 2 fairly good
- 3 fairly bad
- 4 very bad

v227, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	12	0.8	
-1	dk	М	95	6.3	
1	very good		19	1.3	1.4
2	fairly good		79	5.2	5.6
3	fairly bad		194	12.8	13.8
4	very bad		1110	73.5	79.2
	Sum		1510	99.9	100.0
	Valid Cases		1402		

v228 - political system: democratic (Q66D)

Q66

<SHOW CARD 66 - READ OUT AND CODE ONE ANSWER PER LINE>

I'm going to describe various types of political systems and ask what you think about each as a way of governing this country. For each one, would you say it is a very good, fairly good, fairly bad or very bad way of governing this country?

Q66.D Having a democratic political system

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very good
- 2 fairly good
- 3 fairly bad
- 4 very bad

v228, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	10	0.7	
-1	dk	М	121	8.0	
1	very good		757	50.1	54.9
2	fairly good		510	33.8	37.0
3	fairly bad		69	4.6	5.0
4	very bad		43	2.8	3.1
	Sum		1510	100.0	100.0
	Valid Cases		1379		

v229 - democracy: best political system (Q67A)

Q67

<SHOW CARD 67 - READ OUT AND CODE ONE ANSWER PER LINE>

I'm going to read off some things that people sometimes say about a democratic political system. Could you please tell me if you agree strongly, agree, disagree or disagree strongly, after I read each of them?

Q67.A Democracy may have problems but it's better than any other form of government

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 disagree
- 4 disagree strongly

v229, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	11	0.7	
-1	dk	М	108	7.2	
1	agree strongly		825	54.6	59.3
2	agree		513	34.0	36.9
3	disagree		41	2.7	2.9
4	disagree strongly		12	0.8	0.9
	Sum		1510	100.0	100.0
	Valid Cases		1391		

v230 - democracy: causes bad economy (Q67B)

Q67

<SHOW CARD 67 - READ OUT AND CODE ONE ANSWER PER LINE>

I'm going to read off some things that people sometimes say about a democratic political system. Could you please tell me if you agree strongly, agree, disagree or disagree strongly, after I read each of them?

Q67.B In democracy, the economic system runs badly

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 disagree
- 4 disagree strongly

v230, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	11	0.7	
-1	dk	М	216	14.3	
1	agree strongly		42	2.8	3.3
2	agree		203	13.4	15.8
3	disagree		668	44.2	52.1
4	disagree strongly		369	24.4	28.8
	Sum		1510	99.9	100.0
	Valid Cases		1283		

v231 - democracy: is indecisive (Q67C)

Q67

<SHOW CARD 67 - READ OUT AND CODE ONE ANSWER PER LINE>

I'm going to read off some things that people sometimes say about a democratic political system. Could you please tell me if you agree strongly, agree, disagree or disagree strongly, after I read each of them?

Q67.C Democracies are indecisive and have too much squabbling

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 disagree
- 4 disagree strongly

v231, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	11	0.7	
-1	dk	М	168	11.1	
1	agree strongly		117	7.7	8.8
2	agree		526	34.8	39.5
3	disagree		490	32.5	36.8
4	disagree strongly		198	13.1	14.9
	Sum		1510	100.0	100.0
	Valid Cases		1331		

v232 - democracy: cannot maintain order (Q67D)

Q67

<SHOW CARD 67 - READ OUT AND CODE ONE ANSWER PER LINE>

I'm going to read off some things that people sometimes say about a democratic political system. Could you please tell me if you agree strongly, agree, disagree or disagree strongly, after I read each of them?

Q67.D Democracies aren't good at maintaining order

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 disagree
- 4 disagree strongly

v232, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	11	0.7	
-1	dk	М	167	11.1	
1	agree strongly		53	3.5	4.0
2	agree		225	14.9	16.9
3	disagree		601	39.8	45.1
4	disagree strongly		453	30.0	34.0
	Sum		1510	100.0	100.0
	Valid Cases		1332		

v233 - do you justify: claiming state benefits (Q68A)

Q68

<SHOW CARD 68 - READ OUT AND CODE ONE ANSWER PER LINE>

Please tell me for each of the following whether you think it can always be justified, never be justified, or something in between, using this card.

Q68.A Claiming state benefits which you are not entitled to

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 never
- 2
- 3
- 4
- 5
- 6 7
- 8
- 9
- 10 always

v233, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	23	1.5	
1	never		844	55.9	56.8
2			103	6.8	6.9
3			193	12.8	13.0
4			43	2.8	2.9
5			174	11.5	11.7
6			42	2.8	2.8
7			31	2.1	2.1
8			13	0.9	0.9
9			1	0.1	0.1
10	always		41	2.7	2.8
	Sum		1510	99.9	100.0
	Valid Cases		1487		

v234 - do you justify: cheating on tax (Q68B)

Q68

<SHOW CARD 68 - READ OUT AND CODE ONE ANSWER PER LINE>

Please tell me for each of the following whether you think it can always be justified, never be justified, or something in between, using this card.

Q68.B Cheating on tax if you have the chance

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 never
- 2
- 3
- 4
- 5
- 6
- 7
- 8
- 10 always

v234, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	2	0.1	
-1	dk	М	16	1.1	
1	never		850	56.3	57.0
2			115	7.6	7.7
3			195	12.9	13.1
4			40	2.6	2.7
5			116	7.7	7.8
6			35	2.3	2.3
7			65	4.3	4.4
8			14	0.9	0.9
9			7	0.5	0.5
10	always		54	3.6	3.6
	Sum		1510	99.9	100.0
	Valid Cases		1492		

v235 - do you justify: joyriding (Q68C)

Q68

<SHOW CARD 68 - READ OUT AND CODE ONE ANSWER PER LINE>

Please tell me for each of the following whether you think it can always be justified, never be justified, or something in between, using this card.

Q68.C Taking and driving away a car belonging to someone else (joyriding)

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 never
- 2
- 3
- 4
- 5
- 6
- 7
- 8
- 10 always

v235, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	5	0.3	
1	never		1306	86.5	86.8
2			23	1.5	1.5
3			74	4.9	4.9
4			21	1.4	1.4
5			55	3.6	3.7
6			6	0.4	0.4
7			10	0.7	0.7
9			3	0.2	0.2
10	always		7	0.5	0.5
	Sum		1510	100.0	100.0
	Valid Cases		1505		

v236 - do you justify: taking soft drugs (Q68D)

Q68

<SHOW CARD 68 - READ OUT AND CODE ONE ANSWER PER LINE>

Please tell me for each of the following whether you think it can always be justified, never be justified, or something in between, using this card.

Q68.D Taking the drug marijuana or hashish

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 never
- 2
- 3
- 4
- 5
- 6
- 7
- 8
- 10 always

v236, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	22	1.5	
1	never		1049	69.5	70.5
2			64	4.2	4.3
3			114	7.5	7.7
4			37	2.5	2.5
5			101	6.7	6.8
6			31	2.1	2.1
7			31	2.1	2.1
8			20	1.3	1.3
9			13	0.9	0.9
10	always		28	1.9	1.9
	Sum		1510	100.0	100.0
	Valid Cases		1488		

v237 - do you justify: lying in own interest (Q68E)

Q68

<SHOW CARD 68 - READ OUT AND CODE ONE ANSWER PER LINE>

Please tell me for each of the following whether you think it can always be justified, never be justified, or something in between, using this card.

Q68.E Lying in your own interest

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 never
- 2
- 3
- 4
- 5
- 6
- 7
- 8
- 10 always

v237, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	19	1.3	
1	never		574	38.0	38.5
2			127	8.4	8.5
3			269	17.8	18.1
4			81	5.4	5.4
5			212	14.0	14.2
6			61	4.0	4.1
7			73	4.8	4.9
8			31	2.1	2.1
9			8	0.5	0.5
10	always		54	3.6	3.6
	Sum		1510	100.1	100.0
	Valid Cases		1489		

v238 - do you justify: adultery (Q68F)

Q68

<SHOW CARD 68 - READ OUT AND CODE ONE ANSWER PER LINE>

Please tell me for each of the following whether you think it can always be justified, never be justified, or something in between, using this card.

Q68.F Married men/women having an affair

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 never
- 2
- 3
- 4
- 5
- 6
- 7
- 8 9
- 10 always

v238, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	2	0.1	
-1	dk	М	39	2.6	
1	never		701	46.4	47.8
2			96	6.4	6.5
3			205	13.6	14.0
4			49	3.2	3.3
5			241	16.0	16.4
6			51	3.4	3.5
7			58	3.8	4.0
8			28	1.9	1.9
9			9	0.6	0.6
10	always		30	2.0	2.0
	Sum		1510	99.9	100.0
	Valid Cases		1469		

v239 - do you justify: accepting a bribe (Q68G)

Q68

<SHOW CARD 68 - READ OUT AND CODE ONE ANSWER PER LINE>

Please tell me for each of the following whether you think it can always be justified, never be justified, or something in between, using this card.

Q68.G Someone accepting a bribe in the course of their duties

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 never
- 2
- 3
- 4
- 5 6
- 7
- 8
- 9
- 10 always

v239, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	17	1.1	
1	never		1004	66.5	67.2
2			99	6.6	6.6
3			193	12.8	12.9
4			42	2.8	2.8
5			93	6.2	6.2
6			26	1.7	1.7
7			20	1.3	1.3
8			4	0.3	0.3
10	always		13	0.9	0.9
	Sum		1510	100.1	100.0
	Valid Cases		1493		

v240 - do you justify: homosexuality (Q68H)

Q68

<SHOW CARD 68 - READ OUT AND CODE ONE ANSWER PER LINE>

Please tell me for each of the following whether you think it can always be justified, never be justified, or something in between, using this card.

Q68.H Homosexuality

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 never
- 2
- 3
- 4
- 5
- 6 7
- 8
- 9
- 10 always

v240, weighted by weight

- ,	- 3 , -	5			
Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	8	0.5	
-1	dk	М	73	4.8	
1	never		360	23.8	25.2
2			26	1.7	1.8
3			99	6.6	6.9
4			46	3.0	3.2
5			300	19.9	21.0
6			82	5.4	5.7
7			99	6.6	6.9
8			67	4.4	4.7
9			44	2.9	3.1
10	always		306	20.3	21.4
	Sum		1510	100.0	100.0
	Valid Cases		1429		

v241 - do you justify: abortion (Q68I)

Q68

<SHOW CARD 68 - READ OUT AND CODE ONE ANSWER PER LINE>

Please tell me for each of the following whether you think it can always be justified, never be justified, or something in between, using this card.

Q68.I Abortion

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 never
- 2
- 3
- 4
- 5
- 6
- 7 8
- 9
- 10 always

v241, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	50	3.3	
1	never		332	22.0	22.8
2			80	5.3	5.5
3			172	11.4	11.8
4			66	4.4	4.5
5			358	23.7	24.6
6			84	5.6	5.8
7			90	6.0	6.2
8			83	5.5	5.7
9			51	3.4	3.5
10	always		140	9.3	9.6
	Sum		1510	100.0	100.0
	Valid Cases		1457		

v242 - do you justify: divorce (Q68J)

Q68

<SHOW CARD 68 - READ OUT AND CODE ONE ANSWER PER LINE>

Please tell me for each of the following whether you think it can always be justified, never be justified, or something in between, using this card.

Q68.J Divorce

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 never
- 2
- 3
- 4
- 5 6
- 7
- 8
- 9
- 10 always

v242, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	7	0.5	
-1	dk	М	38	2.5	
1	never		164	10.9	11.2
2			26	1.7	1.8
3			125	8.3	8.5
4			49	3.2	3.3
5			395	26.2	27.0
6			94	6.2	6.4
7			137	9.1	9.4
8			107	7.1	7.3
9			75	5.0	5.1
10	always		293	19.4	20.0
	Sum		1510	100.0	100.0
	Valid Cases		1466		

v243 - do you justify: euthanasia (Q68K)

Q68

<SHOW CARD 68 - READ OUT AND CODE ONE ANSWER PER LINE>

Please tell me for each of the following whether you think it can always be justified, never be justified, or something in between, using this card.

Q68.K Euthanasia (terminating the life of the incurably sick)

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 never
- 2
- 3
- 4
- 5
- 6
- 7
- 8
- 10 always

v243, weighted by weight

	,	-			
Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	9	0.6	
-1	dk	М	75	5.0	
1	never		424	28.1	29.7
2			68	4.5	4.8
3			136	9.0	9.5
4			59	3.9	4.1
5			248	16.4	17.4
6			92	6.1	6.4
7			98	6.5	6.9
8			120	7.9	8.4
9			35	2.3	2.5
10	always		148	9.8	10.4
	Sum		1510	100.1	100.0
	Valid Cases		1426		

v244 - do you justify: suicide (Q68L)

Q68

<SHOW CARD 68 - READ OUT AND CODE ONE ANSWER PER LINE>

Please tell me for each of the following whether you think it can always be justified, never be justified, or something in between, using this card.

Q68.L Suicide

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 never
- 2
- 3
- 4
- 5
- 6
- 7
- 8
- 10 always

v244, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	10	0.7	
-1	dk	М	88	5.8	
1	never		708	46.9	50.1
2			66	4.4	4.7
3			158	10.5	11.2
4			46	3.0	3.3
5			215	14.2	15.2
6			55	3.6	3.9
7			55	3.6	3.9
8			44	2.9	3.1
9			15	1.0	1.1
10	always		51	3.4	3.6
	Sum		1510	100.1	100.0
	Valid Cases		1413		

v245 - do you justify: paying cash to avoid taxes (Q68M)

Q68

<SHOW CARD 68 - READ OUT AND CODE ONE ANSWER PER LINE>

Please tell me for each of the following whether you think it can always be justified, never be justified, or something in between, using this card.

Q68.M Paying cash for services to avoid taxes

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 never
- 2
- 3
- 4
- 5 6
- 7
- 8
- 9
- 10 always

v245, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	35	2.3	
1	never		512	33.9	34.8
2			78	5.2	5.3
3			206	13.6	14.0
4			73	4.8	5.0
5			234	15.5	15.9
6			71	4.7	4.8
7			84	5.6	5.7
8			76	5.0	5.2
9			33	2.2	2.2
10	always		105	7.0	7.1
	Sum		1510	100.1	100.0
	Valid Cases		1471		

v246 - do you justify: having casual sex (Q68N)

Q68

<SHOW CARD 68 - READ OUT AND CODE ONE ANSWER PER LINE>

Please tell me for each of the following whether you think it can always be justified, never be justified, or something in between, using this card.

Q68.N Having casual sex

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 never
- 2
- 3
- 4
- 5 6
- 7
- 8
- 9
- 10 always

v246, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	10	0.7	
-1	dk	М	36	2.4	
1	never		560	37.1	38.3
2			88	5.8	6.0
3			220	14.6	15.0
4			58	3.8	4.0
5			264	17.5	18.0
6			48	3.2	3.3
7			67	4.4	4.6
8			51	3.4	3.5
9			21	1.4	1.4
10	always		87	5.8	5.9
	Sum		1510	100.0	100.0
	Valid Cases		1464		

v247 - do you justify: avoiding fare public transport (Q68O)

Q68

<SHOW CARD 68 - READ OUT AND CODE ONE ANSWER PER LINE>

Please tell me for each of the following whether you think it can always be justified, never be justified, or something in between, using this card.

Q68.O Avoiding a fare on public transport

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 never
- 2
- 3
- 4
- 5
- 6
- 7
- 8
- 10 always

v247, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	22	1.5	
1	never		783	51.9	52.6
2			78	5.2	5.2
3			191	12.6	12.8
4			84	5.6	5.6
5			153	10.1	10.3
6			44	2.9	3.0
7			48	3.2	3.2
8			37	2.5	2.5
9			14	0.9	0.9
10	always		56	3.7	3.8
	Sum		1510	100.0	100.0
	Valid Cases		1488		

v248 - do you justify: prostitution (Q68P)

Q68

<SHOW CARD 68 - READ OUT AND CODE ONE ANSWER PER LINE>

Please tell me for each of the following whether you think it can always be justified, never be justified, or something in between, using this card.

Q68.P Prostitution

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 never
- 2
- 3
- 4
- 5
- 6
- 7
- 8
- 10 always

v248, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	8	0.5	
-1	dk	М	29	1.9	
1	never		437	28.9	29.7
2			47	3.1	3.2
3			183	12.1	12.4
4			77	5.1	5.2
5			285	18.9	19.3
6			59	3.9	4.0
7			98	6.5	6.7
8			97	6.4	6.6
9			45	3.0	3.1
10	always		145	9.6	9.8
	Sum		1510	100.0	100.0
	Valid Cases		1473		

v249 - do you justify: experiments human embryos (Q68Q)

Q68

<SHOW CARD 68 - READ OUT AND CODE ONE ANSWER PER LINE>

Please tell me for each of the following whether you think it can always be justified, never be justified, or something in between, using this card.

Q68.Q Scientific experiments on human embryos

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 never
- 2
- 3
- 4
- 5
- 6 7
- 8
- 9
- 10 always

v249, weighted by weight

	,	0			
Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	43	2.8	
1	never		937	62.1	64.0
2			81	5.4	5.5
3			170	11.3	11.6
4			43	2.8	2.9
5			100	6.6	6.8
6			46	3.0	3.1
7			32	2.1	2.2
8			21	1.4	1.4
9			4	0.3	0.3
10	always		30	2.0	2.0
	Sum		1510	100.0	100.0
	Valid Cases	3	1464		

v250 - do you justify: manipulation food (Q68R)

Q68

<SHOW CARD 68 - READ OUT AND CODE ONE ANSWER PER LINE>

Please tell me for each of the following whether you think it can always be justified, never be justified, or something in between, using this card.

Q68.R Genetic manipulation of food stuffs

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 never
- 2
- 3
- 4
- 5
- 6 7
- 8
- 9
- 10 always

v250, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	34	2.3	
1	never		990	65.6	67.3
2			88	5.8	6.0
3			178	11.8	12.1
4			47	3.1	3.2
5			86	5.7	5.8
6			26	1.7	1.8
7			22	1.5	1.5
8			11	0.7	0.7
9			6	0.4	0.4
10	always		18	1.2	1.2
	Sum		1510	99.9	100.0
	Valid Cases		1473		

v251 - do you justify: invitro fertilization (Q68S)

Q68

<SHOW CARD 68 - READ OUT AND CODE ONE ANSWER PER LINE>

Please tell me for each of the following whether you think it can always be justified, never be justified, or something in between, using this card.

Q68.S Artificial insemination or in-vitro fertilization

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 never
- 2
- 3
- 4
- 5
- 6
- 7 8
- 9
- 10 always

v251, weighted by weight

	0 ,	·			
Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	62	4.1	
1	never		321	21.3	22.2
2			30	2.0	2.1
3			120	7.9	8.3
4			45	3.0	3.1
5			213	14.1	14.7
6			80	5.3	5.5
7			139	9.2	9.6
8			148	9.8	10.2
9			86	5.7	6.0
10	always		263	17.4	18.2
	Sum		1510	100.0	100.0
	Valid Cases		1445		

v252 - do you justify: death penalty (Q68T)

Q68

<SHOW CARD 68 - READ OUT AND CODE ONE ANSWER PER LINE>

Please tell me for each of the following whether you think it can always be justified, never be justified, or something in between, using this card.

Q68.T Death penalty

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 never
- 2
- 3
- 4
- 5 6
- 7
- 8
- 9
- 10 always

v252, weighted by weight

	• .		J			
Value	Label		Missing	Count	Percent	Valid Percent
-2	na		М	3	0.2	
-1	dk		М	43	2.8	
1	never			826	54.7	56.5
2				66	4.4	4.5
3				147	9.7	10.0
4				40	2.6	2.7
5				134	8.9	9.2
6				44	2.9	3.0
7				47	3.1	3.2
8				54	3.6	3.7
9				11	0.7	0.8
10	always			94	6.2	6.4
	Sum			1510	99.9	100.0
	Valid Case	es		1464		

AT52 - do you justify: causing damage to a parked car and not reporting it (Q68U_at)

Q68

<SHOW CARD 68 - READ OUT AND CODE ONE ANSWER PER LINE>

Please tell me for each of the following whether you think it can always be justified, never be justified, or something in between, using this card.

Q68U_at Causing damage to a parked car and not reporting it

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 never
- 2
- 3
- 4
- 5
- 6
- 7
- 8
- 9
- 10 always

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT52, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	9	0.6	
1	never		1153	76.4	77.0
2			69	4.6	4.6
3			137	9.1	9.1
4			41	2.7	2.7
5			59	3.9	3.9
6			21	1.4	1.4
7			8	0.5	0.5
8			3	0.2	0.2
9			1	0.1	0.1
10	always		6	0.4	0.4
	Sum		1510	99.9	100.0
	Valid Cases		1499		

AT53 - do you justify: buying stolen goods (Q68V_at)

Q68

<SHOW CARD 68 - READ OUT AND CODE ONE ANSWER PER LINE>

Please tell me for each of the following whether you think it can always be justified, never be justified, or something in between, using this card.

Q68V_at Buying stolen goods

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 never
- 2
- 3
- 4
- 5
- 6
- 7
- 8
- 10 always

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT53, weighted by weight

	,	•			
Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	5	0.3	
-1	dk	М	20	1.3	
1	never		1046	69.3	70.4
2			71	4.7	4.8
3			174	11.5	11.7
4			39	2.6	2.6
5			110	7.3	7.4
6			20	1.3	1.3
7			13	0.9	0.9
8			5	0.3	0.3
9			2	0.1	0.1
10	always		6	0.4	0.4
	Sum		1510	100.1	100.0
	Valid Cases		1485		

f253 - flag variable: belong to geographic group - most/second (Q69/Q70)

Flag variable:

Indicates inconsistencies for the variables v253/v254.

If geographical group that respondent states to belong to first of all is identical with secondly stated group.

if v253 = v254 then f253=1

- 0 consistent
- 1 inconsistent

Note:

For detailed information, see overview flag variables.

f253

Value	Label	Missing Count	Percent	Valid Percent
0	consistent	1474	97.6	97.6
1	inconsistent	36	2.4	2.4
	Sum	1510	100.0	100.0
	Valid Cases	1510		

v253 - belong to geographic group - most (Q69)

Q69

<SHOW CARD 69 - CODE ONE ANSWER ONLY FOR Q69>

Which of these geographical groups would you say you belong to first of all?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 locality or town where you live
- 2 region of country where you live
- 3 [COUNTRY]
- 4 Europe
- 5 the world as a whole

v253, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	16	1.1	
1	locality or town		376	24.9	25.2
2	region country		534	35.4	35.7
3	country as a whole		410	27.2	27.4
4	Europe		94	6.2	6.3
5	world as a whole		81	5.4	5.4
	Sum		1510	100.1	100.0
	Valid Cases		1494		

v254 - belong to geographic group - next (Q70)

[Q69: Which of these geographical groups would you say you belong to first of all?]

Q70

<STILL USE CARD 69 - CODE ONE ANSWER ONLY FOR Q70>

And secondly?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 locality or town where you live
- 2 region of country where you live
- 3 [COUNTRY]
- 4 Europe
- 5 the world as a whole

v254, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	34	2.3	
1	locality or town		277	18.3	18.8
2	region country		421	27.9	28.5
3	country as a whole		568	37.6	38.5
4	Europe		154	10.2	10.4
5	world as a whole		55	3.6	3.7
	Sum		1510	100.0	100.0
	Valid Cases		1475		

v255 - are you a citizen of ... (country) (Q71)

Q71

Are you a citizen of [COUNTRY]?

<IF CODE 2, -1, -2: GO TO Q73>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v255, weighted by weight

Value	Label	Missing Count	Percent	Valid Percent
1	yes	1477	97.8	97.8
2	no	33	2.2	2.2
	Sum	1510	100.0	100.0
	Valid Cases	1510		

v256 - how proud are you to be a ... (country) citizen (Q72)

Q72

How proud are you to be a [COUNTRY] citizen?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very proud
- 2 quite proud
- 3 not very proud
- 4 not at all proud

v256, weighted by weight

	0 , 0				
Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	33	2.2	
-2	na	М	4	0.3	
-1	dk	М	52	3.4	
1	very proud		689	45.6	48.5
2	quite proud		582	38.5	41.0
3	not very proud		97	6.4	6.8
4	not at all proud		53	3.5	3.7
	Sum		1510	100.0	100.0
	Valid Cases		1421		

v257 - EU fears: loss of social security (Q73A)

Q73

<SHOW CARD 73 - READ OUT AND CODE ONE ANSWER PER LINE>

Some people may have fears about the building of the European Union. I am going to read a number of things which people say they are afraid of. For each tell me if you - personally - are currently afraid of:

Q73.A The loss of social security

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very much afraid

2

3

4 5

6

7

8 9

10 not afraid at all

v257, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	2	0.1	
-1	dk	М	16	1.1	
1	very much afraid		321	21.3	21.5
2			74	4.9	5.0
3			290	19.2	19.4
4			81	5.4	5.4
5			272	18.0	18.2
6			64	4.2	4.3
7			69	4.6	4.6
8			120	7.9	8.0
9			27	1.8	1.8
10	not afraid at all		174	11.5	11.7
	Sum		1510	100.0	100.0
	Valid Cases		1492		

v258 - EU fears: lose national identity/culture (Q73B)

Q73

<SHOW CARD 73 - READ OUT AND CODE ONE ANSWER PER LINE>

Some people may have fears about the building of the European Union. I am going to read a number of things which people say they are afraid of. For each tell me if you - personally - are currently afraid of:

Q73.B The loss of national identity and culture

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very much afraid

2

3

4

5 6

7

8 9

10 not afraid at all

v258, weighted by weight

	0 , 0				
Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	17	1.1	
1	very much afraid		266	17.6	17.9
2			61	4.0	4.1
3			217	14.4	14.6
4			69	4.6	4.6
5			233	15.4	15.6
6			76	5.0	5.1
7			99	6.6	6.6
8			132	8.7	8.9
9			61	4.0	4.1
10	not afraid at all		276	18.3	18.5
	Sum		1510	100.0	100.0
	Valid Cases		1489		

v259 - EU fears: own country pays (Q73C)

Q73

<SHOW CARD 73 - READ OUT AND CODE ONE ANSWER PER LINE>

Some people may have fears about the building of the European Union. I am going to read a number of things which people say they are afraid of. For each tell me if you - personally - are currently afraid of:

Q73.C Our country paying more and more to the European Union

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very much afraid

2

3

4 5

6

7

8 9

10 not afraid at all

v259, weighted by weight

	0 , 0				
Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	5	0.3	
-1	dk	М	37	2.5	
1	very much afraid		431	28.5	29.4
2			84	5.6	5.7
3			242	16.0	16.5
4			80	5.3	5.4
5			197	13.0	13.4
6			75	5.0	5.1
7			69	4.6	4.7
8			93	6.2	6.3
9			25	1.7	1.7
10	not afraid at all		172	11.4	11.7
	Sum		1510	100.0	100.0
	Valid Cases		1468		

v260 - EU fears: loss of power (Q73D)

Q73

<SHOW CARD 73 - READ OUT AND CODE ONE ANSWER PER LINE>

Some people may have fears about the building of the European Union. I am going to read a number of things which people say they are afraid of. For each tell me if you - personally - are currently afraid of:

Q73.D A loss of power in the world for [COUNTRY]

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very much afraid

2

3

4

5 6

7

8

9

10 not afraid at all

v260, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	43	2.8	
1	very much afraid		225	14.9	15.4
2			50	3.3	3.4
3			181	12.0	12.4
4			63	4.2	4.3
5			265	17.5	18.1
6			73	4.8	5.0
7			103	6.8	7.0
8			117	7.7	8.0
9			75	5.0	5.1
10	not afraid at all		310	20.5	21.2
	Sum		1510	99.9	100.0
	Valid Cases		1463		

v261 - EU fears: loss of jobs (Q73E)

Q73

<SHOW CARD 73 - READ OUT AND CODE ONE ANSWER PER LINE>

Some people may have fears about the building of the European Union. I am going to read a number of things which people say they are afraid of. For each tell me if you - personally - are currently afraid of:

Q73.E The loss of jobs in [COUNTRY]

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very much afraid

2

3

4

5 6

7

8

9

10 not afraid at all

v261, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	13	0.9	
1	very much afraid		387	25.6	25.9
2			99	6.6	6.6
3			299	19.8	20.0
4			115	7.6	7.7
5			244	16.2	16.3
6			60	4.0	4.0
7			85	5.6	5.7
8			69	4.6	4.6
9			24	1.6	1.6
10	not afraid at all		112	7.4	7.5
	Sum		1510	100.0	100.0
	Valid Cases		1494		

v262 - European Union enlargement (Q74)

Q74

<SHOW CARD 74>

Some say that the European Union enlargement should go further. Others say it has already gone too far. Using this card, which number best describes your position, where '1' means "should go further", and '10' means "has already gone too far"?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 should go further

2

3

4

5

6

7

8

10 has gone too far

v262, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	64	4.2	
1	should go further		114	7.5	7.9
2			18	1.2	1.2
3			111	7.4	7.7
4			69	4.6	4.8
5			193	12.8	13.4
6			96	6.4	6.7
7			101	6.7	7.0
8			184	12.2	12.8
9			68	4.5	4.7
10	has gone too far		489	32.4	33.9
	Sum		1510	100.0	100.0
	Valid Cases		1443		

v263 - would you vote at a general election tomorrow (Q75)

Q75

If there was a general election tomorrow, can you tell me if you would vote?

<IF CODE 1: GO TO Q75a;

IF CODE 2: GO TO Q75b;

IF CODE -1, -2, -3: GO TO Q76>

- -5 other missing
- -4 question not asked
- -3 does not have voting rights
- -2 na
- -1 dk
- 1 Yes, I would vote
- 2 No, I would not vote

v263, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	does not have voting rights	М	23	1.5	
-2	na	М	1	0.1	
-1	dk	М	110	7.3	
1	yes		1167	77.3	84.7
2	no		210	13.9	15.3
	Sum		1510	100.1	100.0
	Valid Cases		1377		

v264 - which political party would you vote for (Q75a)

[Q75: If there was a general election tomorrow, can you tell me if you would vote?]

Q75a

<SHOW CARD 75a>

IF YES: which party would you vote for?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 40001 AT Social Democratic Party of Austria SPÖ
- 40002 AT Austrian People's Party ÖVP
- 40003 AT Austrian Freedom Party FPÖ
- 40004 AT The Austrian Greens\Die Grünen
- 40005 AT Liberal Forum\Liberales Forum
- 40006 AT Communist Party of Austria KPÖ
- 40007 AT Alliance for the Future of Austria BZÖ
- 40008 AT Dr. Martin's List For Democracy, Control, Justice\Liste Hans Peter Martin
- 40096 AT Other

Note:

Standardized country-specific variable: combined variable containing country identifier (see COUNTRY) and country-specific categories (political parties). This variable has been harmonized into V264_LR (left-right scale variable).

v264, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	343	22.7	
-2	na	М	66	4.4	
-1	dk	М	246	16.3	
40001	AT Social Democratic Party of Austria SPÖ		296	19.6	34.6
40002	AT Austrian People's Party ÖVP		208	13.8	24.3
40003	AT Austrian Freedom Party FPÖ		112	7.4	13.1
40004	AT The Austrian Greens\Die Grünen		125	8.3	14.6
40005	AT Liberal Forum\Liberales Forum		29	1.9	3.4
40006	AT Communist Party of Austria KPÖ		10	0.7	1.2
40007	AT Alliance for the Future of Austria BZÖ		49	3.2	5.7
40008	AT Dr. Martin's List - For Democracy, Control, Justice\Liste Hans Peter Martin		8	0.5	0.9
40096	AT Other		19	1.3	2.2
	Sum		1510	100.1	100.0
	Valid Cases		855		

v264_LR - which political party would you vote for - left/right scale (Q75a)

POLITICAL PARTY: LEFT-RIGHT SCALE - HARMONIZED VARIABLE

[Q75: If there was a general election tomorrow, can you tell me if you would vote?]

Q75a

<SHOW CARD 75a>

IF YES: which party would you vote for?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 left
- 2
- 3
- 4
- 5
- 6
- 7
- 8 9
- 10 right

Note:

Harmonized variable: country-specific categories have been recoded according to the placement of national political parties on a left-right scale as delivered by country.

For more detailed information, see Method Report.

For country-specific variable, see V264.

v264_LR, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	674	44.6	
2			10	0.7	1.2
3			125	8.3	15.0
4			296	19.6	35.4
5			36	2.4	4.3
7			208	13.8	24.9
8			49	3.2	5.9
9			112	7.4	13.4
	Sum		1510	100.0	100.0
	Valid Cases		836		

v265 - which political party appeals to you most (Q75b)

[Q75: If there was a general election tomorrow, can you tell me if you would vote?]

Q75b

<SHOW CARD 75b>

IF NO: Which party appeals to you most?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 40001 AT Social Democratic Party of Austria SPÖ
- 40002 AT Austrian People's Party ÖVP
- 40003 AT Austrian Freedom Party FPÖ
- 40004 AT The Austrian Greens\Die Grünen
- 40005 AT Liberal Forum\Liberales Forum
- 40006 AT Communist Party of Austria KPÖ
- 40007 AT Alliance for the Future of Austria BZÖ
- 40008 AT Dr. Martin's List For Democracy, Control, Justice\Liste Hans Peter Martin
- 40096 AT Other

Note:

Standardized country-specific variable: combined variable containing country identifier (see COUNTRY) and country-specific categories (political parties). This variable has been harmonized into V265_LR (left-right scale variable).

v265, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	1300	86.1	
-2	na	М	67	4.4	
-1	dk	М	79	5.2	
40001	AT Social Democratic Party of Austria SPÖ		16	1.1	25.0
40002	AT Austrian People´s Party ÖVP		8	0.5	12.5
40003	AT Austrian Freedom Party FPÖ		12	0.8	18.8
40004	AT The Austrian Greens\Die Grünen		13	0.9	20.3
40005	AT Liberal Forum\Liberales Forum		2	0.1	3.1
40006	AT Communist Party of Austria KPÖ		3	0.2	4.7
40007	AT Alliance for the Future of Austria BZÖ		9	0.6	14.1
40096	AT Other		1	0.1	1.6
	Sum		1510	100.0	100.0
	Valid Cases		64		

v265_LR - which political party appeals to you most - left/right scale (Q75b)

POLITICAL PARTY: LEFT-RIGHT SCALE - HARMONIZED VARIABLE

[Q75: If there was a general election tomorrow, can you tell me if you would vote?]

Q75b

<SHOW CARD 75b>

IF NO: Which party appeals to you most?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 left
- 2
- 3
- 4
- 5
- 6
- 7
- 8 9
- •

10 right

Note:

Harmonized variable: country-specific categories have been recoded according to the placement of national political parties on a left-right scale as delivered by country.

For more detailed information, see Method Report.

For country-specific variable, see V265.

v265_LR, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	1448	95.9	
2			3	0.2	4.8
3			13	0.9	20.6
4			16	1.1	25.4
5			2	0.1	3.2
7			8	0.5	12.7
8			9	0.6	14.3
9			12	0.8	19.0
	Sum		1510	100.1	100.0
	Valid Cases		62		

v266 - work: people from less developed countries (Q76)

Q76

<ASK ALL>

<SHOW CARD 76 - READ OUT AND CODE ONE ANSWER ONLY>

How about people from less developed countries coming here to work? Which one of the following do you think the government should do?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 let anyone come who wants to
- 2 let people come as long as there are jobs available
- 3 put strict limits on the number of foreigners who can come here
- 4 prohibit people coming here from other countries

v266, weighted by weight

Value Label	Missing Count	Percent	Valid Percent
-1 dk	M 52	3.4	
1 anyone come who wants to	95	6.3	6.5
2 come when jobs available	659	43.6	45.2
3 strict limits on the number of foreigners	607	40.2	41.6
4 prohibit people coming here from other countries	97	6.4	6.7
Sum	1510	100.0	100.0
Valid Cases	1458		

v267 - terrorism: can be justified/must be condemned (Q77)

Q77

<SHOW CARD 77 - READ OUT AND CODE ONE ANSWER ONLY>

Terrorism is everyday news. In principle, most people are against it, but there is still room for differences of opinion. Which of these two statements do you tend to agree with?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 There may be certain circumstances where terrorism is justified
- 2 Terrorism for whatever motive must always be condemned
- 3 neither (spontaneous)

v267, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	21	1.4	
1	under certain circumstances justified		111	7.4	7.5
2	must always be condemned		1323	87.6	88.9
3	neither		54	3.6	3.6
	Sum		1510	100.0	100.0
	Valid Cases		1488		

v268 - immigrants take away jobs from [nationality] (Q78A)

Q78A

<SHOW CARD 78 - READ OUT>

Please look at the following statements and indicate where you would place your views on this scale?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 Immigrants take jobs away from natives in a country

2

4

5

6

7 8

9

10 Immigrants do not take jobs away from natives in a country

v268, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	14	0.9	
1	take away		354	23.4	23.7
2			61	4.0	4.1
3			216	14.3	14.4
4			91	6.0	6.1
5			316	20.9	21.1
6			68	4.5	4.5
7			98	6.5	6.6
8			115	7.6	7.7
9			42	2.8	2.8
10	do not take away		135	8.9	9.0
	Sum		1510	100.0	100.0
	Valid Cases		1496		

v269 - immigrants undermine countrys cultural life (Q78B)

Q78B

<SHOW CARD 78 - READ OUT>

Please look at the following statements and indicate where you would place your views on this scale?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 A country's cultural life is undermined by immigrants

2

3

4 5

6

7

8 9

10 A country's cultural life is not undermined by immigrants

v269, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	23	1.5	
1	undermine cultural life		331	21.9	22.3
2			71	4.7	4.8
3			247	16.4	16.6
4			90	6.0	6.1
5			260	17.2	17.5
6			78	5.2	5.2
7			124	8.2	8.3
8			117	7.7	7.9
9			38	2.5	2.6
10	do not undermine cultural life		130	8.6	8.7
	Sum		1510	100.0	100.0
	Valid Cases		1487		

v270 - immigrants increase crime problems (Q78C)

Q78C

<SHOW CARD 78 - READ OUT>

Please look at the following statements and indicate where you would place your views on this scale?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 Immigrants make crime problems worse

2

4

5

6

7

8 9

10 Immigrants do not make crime problems worse

v270, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	15	1.0	
1	make it worse		519	34.4	34.7
2			135	8.9	9.0
3			248	16.4	16.6
4			122	8.1	8.2
5			207	13.7	13.8
6			68	4.5	4.5
7			53	3.5	3.5
8			62	4.1	4.1
9			28	1.9	1.9
10	do not make it worse		53	3.5	3.5
	Sum		1510	100.0	100.0
	Valid Cases		1495		

v271 - immigrants are a strain on welfare system (Q78D)

Q78D

<SHOW CARD 78 - READ OUT>

Please look at the following statements and indicate where you would place your views on this scale?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 Immigrants are a strain on a country's welfare system

2

4

5

6

7 8

9

10 Immigrants are not a strain on a country's welfare system

v271, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	28	1.9	
1	are a strain		536	35.5	36.1
2			112	7.4	7.6
3			255	16.9	17.2
4			96	6.4	6.5
5			200	13.2	13.5
6			81	5.4	5.5
7			57	3.8	3.8
8			57	3.8	3.8
9			26	1.7	1.8
10	are not a strain		63	4.2	4.2
	Sum		1510	100.1	100.0
	Valid Cases		1481		

v272 - immigrants will become a threat to society (Q78E)

Q78E

<SHOW CARD 78 - READ OUT>

Please look at the following statements and indicate where you would place your views on this scale?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 In the future the proportion of immigrants will become a threat to society

2

4

5

6

7 8

9

10 In the future the proportion of immigrants will not become a threat to society

v272, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	18	1.2	
1	will become a threat		395	26.2	26.5
2			91	6.0	6.1
3			221	14.6	14.8
4			122	8.1	8.2
5			238	15.8	15.9
6			82	5.4	5.5
7			99	6.6	6.6
8			106	7.0	7.1
9			35	2.3	2.3
10	will not become a threat		104	6.9	7.0
	Sum		1510	100.1	100.0
	Valid Cases		1492		

v273 - immigrants maintain own/take over customs (Q78F)

Q78F

<SHOW CARD 78 - READ OUT>

Please look at the following statements and indicate where you would place your views on this scale?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 For the greater good of society it is better if immigrants maintain their distinct customs and traditions

2 3 4

5 6

7

8

10 For the greater good of society it is better if immigrants do not maintain their distinct customs and traditions but adopt the customs of the country

v273, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	2	0.1	
-1	dk	М	29	1.9	
1	maintain distinct customs and traditions		226	15.0	15.3
2			39	2.6	2.6
3			138	9.1	9.3
4			83	5.5	5.6
5			283	18.7	19.1
6			83	5.5	5.6
7			89	5.9	6.0
8			181	12.0	12.2
9			60	4.0	4.1
10	take over the customs of the country		296	19.6	20.0
	Sum		1510	99.9	100.0
	Valid Cases		1479		

v274 - immigrants living in your country: feels like a stranger (Q79A)

Q79

<SHOW CARD 79 - READ OUT AND CODE ONE ANSWER PER LINE>

Please indicate to what extent you agree or disagree with each of the following statements regarding immigrants living in your country:

Q79.A Because of the number of immigrants in [COUNTRY], I sometimes feel like a stranger

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree strongly

v274, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	2	0.1	
-1	dk	М	13	0.9	
1	agree strongly		283	18.7	18.9
2	agree		479	31.7	32.0
3	neither agree/nor disagree		265	17.5	17.7
4	disagree		266	17.6	17.8
5	disagree strongly		202	13.4	13.5
	Sum		1510	100.0	100.0
	Valid Cases		1495		

v275 - immigrants living in your country: there are too many (Q79B)

Q79

<SHOW CARD 79 - READ OUT AND CODE ONE ANSWER PER LINE>

Please indicate to what extent you agree or disagree with each of the following statements regarding immigrants living in your country:

Q79.B Today in [COUNTRY], there are too many immigrants

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree strongly

v275, weighted by weight

V/-1 -	Labat	Mennen	^	D	Maril Daniel
Value	Label	iviissing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	27	1.8	
1	agree strongly		419	27.7	28.3
2	agree		543	36.0	36.6
3	neither agree/nor disagree		288	19.1	19.4
4	disagree		151	10.0	10.2
5	disagree strongly		81	5.4	5.5
	Sum		1510	100.0	100.0
	Valid Cases		1482		

v276 - important: to have been born in [country] (Q80A)

Q80

<SHOW CARD 80>

Some people say the following things are important for being truly [NATIONALITY]. Others say they are not important. How important do you think each of the following is?

Q80.A To have been born in [COUNTRY]

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very important
- 2 quite important
- 3 not important
- 4 not important at all

v276, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	13	0.9	
1	very important		552	36.6	36.9
2	quite important		536	35.5	35.8
3	not important		289	19.1	19.3
4	not important at all		120	7.9	8.0
	Sum		1510	100.0	100.0
	Valid Cases		1497		

v277 - important: to respect [country nationality] political institutions and laws (Q80B)

Q80

<SHOW CARD 80>

Some people say the following things are important for being truly [NATIONALITY]. Others say they are not important. How important do you think each of the following is?

Q80.B To respect [COUNTRY]'s political institutions and laws

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very important
- 2 quite important
- 3 not important
- 4 not important at all

v277, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	2	0.1	
-1	dk	М	11	0.7	
1	very important		975	64.6	65.1
2	quite important		438	29.0	29.3
3	not important		63	4.2	4.2
4	not important at all		21	1.4	1.4
	Sum		1510	100.0	100.0
	Valid Cases		1497		

v278 - important: to have [country nationality] ancestry (Q80C)

Q80

<SHOW CARD 80>

Some people say the following things are important for being truly [NATIONALITY]. Others say they are not important. How important do you think each of the following is?

Q80.C To have [COUNTRY]'s ancestry

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very important
- 2 quite important
- 3 not important
- 4 not important at all

v278, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	11	0.7	
1	very important		365	24.2	24.3
2	quite important		395	26.2	26.3
3	not important		515	34.1	34.3
4	not important at all		225	14.9	15.0
	Sum		1510	100.1	100.0
	Valid Cases		1499		

v279 - important: to be able to speak [country language] (Q80D)

Q80

<SHOW CARD 80>

Some people say the following things are important for being truly [NATIONALITY]. Others say they are not important. How important do you think each of the following is?

Q80.D To be able to speak [THE NATIONAL LANGUAGE] [NOTE: if more than one national languages, ask the national languages]

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very important
- 2 quite important
- 3 not important
- 4 not important at all

v279, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	4	0.3	
1	very important		1007	66.7	66.9
2	quite important		420	27.8	27.9
3	not important		68	4.5	4.5
4	not important at all		11	0.7	0.7
	Sum		1510	100.0	100.0
	Valid Cases		1506		

v280 - important: to have lived in [country] for a long time (Q80E)

Q80

<SHOW CARD 80>

Some people say the following things are important for being truly [NATIONALITY]. Others say they are not important. How important do you think each of the following is?

Q80.E To have lived for a long time in [COUNTRY]

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very important
- 2 quite important
- 3 not important
- 4 not important at all

v280, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	15	1.0	
1	very important		611	40.5	40.9
2	quite important		572	37.9	38.3
3	not important		257	17.0	17.2
4	not important at all		54	3.6	3.6
	Sum		1510	100.0	100.0
	Valid Cases		1494		

v281 - how often do you follow politics in media (Q81)

Q81

<SHOW CARD 81>

How often do you follow politics in the news on television or on the radio or in the daily papers?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 every day
- 2 several times a week
- 3 once or twice a week
- 4 less often
- 5 never

v281, weighted by weight

, , , ,			
Value Label	Missing Count	Percent	Valid Percent
-1 dk	M 1	0.1	
1 every day	807	53.4	53.4
2 several times a week	304	20.1	20.1
3 once or twice a week	199	13.2	13.2
4 less often	151	10.0	10.0
5 never	49	3.2	3.2
Sum	1510	100.1	100.0
Valid Cases	1509		

v282 - give information to help justice (Q82A)

Q82

<SHOW CARD 82 – READ OUT AND CODE ONE ANSWER PER LINE!>

Can you tell me your opinion on each of the following statements?

Q82.A If someone has information that may help justice be done, generally he or she should give it to the authorities

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree strongly

v282, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	68	4.5	
1	agree strongly		423	28.0	29.4
2	agree		648	42.9	45.0
3	agree nor disagree		264	17.5	18.3
4	disagree		72	4.8	5.0
5	disagree strongly		33	2.2	2.3
	Sum		1510	100.1	100.0
	Valid Cases		1438		

v283 - people should stick to own affairs (Q82B)

Q82

<SHOW CARD 82 – READ OUT AND CODE ONE ANSWER PER LINE!>

Can you tell me your opinion on each of the following statements?

Q82.B People should stick to their own affairs and not show too much interest in what others say or do

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree strongly

v283, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	44	2.9	
1	agree strongly		343	22.7	23.5
2	agree		472	31.3	32.3
3	agree nor disagree		371	24.6	25.4
4	disagree		227	15.0	15.5
5	disagree strongly		49	3.2	3.4
	Sum		1510	100.0	100.0
	Valid Cases		1462		

v284 - are you concerned with: immediate family (Q83A)

Q83

<SHOW CARD 83 – READ OUT AND CODE ONE ANSWER PER LINE!>

To what extent do you feel concerned about the living conditions of:

Q83.A Your immediate family

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very much
- 2 much
- 3 to a certain extent
- 4 not so much
- 5 not at all

v284, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	4	0.3	
1	very much		477	31.6	31.7
2	much		454	30.1	30.2
3	to a certain extent		366	24.2	24.3
4	not so much		141	9.3	9.4
5	not at all		67	4.4	4.5
	Sum		1510	100.0	100.0
	Valid Cases		1505		

v285 - are you concerned with: people neighbourhood (Q83B)

Q83

<SHOW CARD 83 – READ OUT AND CODE ONE ANSWER PER LINE!>

To what extent do you feel concerned about the living conditions of:

Q83.B People in your neighbourhood

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very much
- 2 much
- 3 to a certain extent
- 4 not so much
- 5 not at all

v285, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	4	0.3	
1	very much		105	7.0	7.0
2	much		272	18.0	18.0
3	to a certain extent		628	41.6	41.7
4	not so much		330	21.9	21.9
5	not at all		172	11.4	11.4
	Sum		1510	100.1	100.0
	Valid Cases		1506		

v286 - are you concerned with: people own region (Q83C)

Q83

<SHOW CARD 83 – READ OUT AND CODE ONE ANSWER PER LINE!>

To what extent do you feel concerned about the living conditions of:

Q83.C The people of the region you live in

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very much
- 2 much
- 3 to a certain extent
- 4 not so much
- 5 not at all

v286, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	4	0.3	
1	very much		62	4.1	4.1
2	much		200	13.2	13.3
3	to a certain extent		578	38.3	38.4
4	not so much		436	28.9	28.9
5	not at all		231	15.3	15.3
	Sum		1510	100.1	100.0
	Valid Cases		1506		

v287 - are you concerned with: fellow countrymen (Q83D)

Q83

<SHOW CARD 83 – READ OUT AND CODE ONE ANSWER PER LINE!>

To what extent do you feel concerned about the living conditions of:

Q83.D Your fellow countrymen

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very much
- 2 much
- 3 to a certain extent
- 4 not so much
- 5 not at all

v287, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	5	0.3	
1	very much		64	4.2	4.3
2	much		191	12.6	12.7
3	to a certain extent		571	37.8	37.9
4	not so much		413	27.4	27.4
5	not at all		266	17.6	17.7
	Sum		1510	100.1	100.0
	Valid Cases		1505		

v288 - are you concerned with: europeans (Q83E)

Q83

<SHOW CARD 83 – READ OUT AND CODE ONE ANSWER PER LINE!>

To what extent do you feel concerned about the living conditions of:

Q83.E Europeans

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very much
- 2 much
- 3 to a certain extent
- 4 not so much
- 5 not at all

v288, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	9	0.6	
1	very much		39	2.6	2.6
2	much		139	9.2	9.3
3	to a certain extent		443	29.3	29.5
4	not so much		514	34.0	34.2
5	not at all		367	24.3	24.4
	Sum		1510	100.1	100.0
	Valid Cases		1501		

v289 - are you concerned with: humankind (Q83F)

Q83

<SHOW CARD 83 – READ OUT AND CODE ONE ANSWER PER LINE!>

To what extent do you feel concerned about the living conditions of:

Q83.F All humans all over the world

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very much
- 2 much
- 3 to a certain extent
- 4 not so much
- 5 not at all

v289, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	12	0.8	
1	very much		69	4.6	4.6
2	much		165	10.9	11.0
3	to a certain extent		491	32.5	32.8
4	not so much		427	28.3	28.5
5	not at all		346	22.9	23.1
	Sum		1510	100.1	100.0
	Valid Cases		1498		

v290 - are you concerned with: elderly people (Q84A)

Q84

<SHOW CARD 84 - READ OUT EACH ITEM AND CODE AN ANSWER FOR EACH>

To what extent do you feel concerned about the living conditions of:

Q84.A elderly people in [COUNTRY]

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very much
- 2 much
- 3 to a certain extent
- 4 not so much
- 5 not at all

v290, weighted by weight

Value	Label	Missing (Count	Percent	Valid Percent
-1	dk	М	7	0.5	
1	very much		206	13.6	13.7
2	much		510	33.8	34.0
3	to a certain extent		513	34.0	34.2
4	not so much		164	10.9	10.9
5	not at all		109	7.2	7.3
	Sum		1510	99.9	100.0
	Valid Cases		1503		

v291 - are you concerned with: unemployed people (Q84B)

Q84

<SHOW CARD 84 - READ OUT EACH ITEM AND CODE AN ANSWER FOR EACH>

To what extent do you feel concerned about the living conditions of:

Q84.B unemployed people in [COUNTRY]

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very much
- 2 much
- 3 to a certain extent
- 4 not so much
- 5 not at all

v291, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	7	0.5	
1	very much		88	5.8	5.9
2	much		272	18.0	18.1
3	to a certain extent		602	39.9	40.1
4	not so much		365	24.2	24.3
5	not at all		176	11.7	11.7
	Sum		1510	100.0	100.0
	Valid Cases		1503		

v292 - are you concerned with: immigrants (Q84C)

Q84

<SHOW CARD 84 - READ OUT EACH ITEM AND CODE AN ANSWER FOR EACH>

To what extent do you feel concerned about the living conditions of:

Q84.C immigrants in [COUNTRY]

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very much
- 2 much
- 3 to a certain extent
- 4 not so much
- 5 not at all

v292, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	9	0.6	
1	very much		40	2.6	2.7
2	much		179	11.9	11.9
3	to a certain extent		416	27.5	27.7
4	not so much		496	32.8	33.0
5	not at all		370	24.5	24.7
	Sum		1510	100.0	100.0
	Valid Cases		1501		

v293 - are you concerned with: sick and disabled (Q84D)

Q84

<SHOW CARD 84 - READ OUT EACH ITEM AND CODE AN ANSWER FOR EACH>

To what extent do you feel concerned about the living conditions of:

Q84.D sick and disabled people in [COUNTRY]

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very much
- 2 much
- 3 to a certain extent
- 4 not so much
- 5 not at all

v293, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	6	0.4	
1	very much		151	10.0	10.0
2	much		477	31.6	31.7
3	to a certain extent		580	38.4	38.6
4	not so much		181	12.0	12.0
5	not at all		115	7.6	7.6
	Sum		1510	100.0	100.0
	Valid Cases		1504		

v294 - are you concerned with: poor children (Q84E)

Q84

<SHOW CARD 84 - READ OUT EACH ITEM AND CODE AN ANSWER FOR EACH>

To what extent do you feel concerned about the living conditions of:

Q84.E Children in poor families in [COUNTRY]

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very much
- 2 much
- 3 to a certain extent
- 4 not so much
- 5 not at all

v294, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	7	0.5	
1	very much		195	12.9	13.0
2	much		529	35.0	35.2
3	to a certain extent		540	35.8	35.9
4	not so much		135	8.9	9.0
5	not at all		105	7.0	7.0
	Sum		1510	100.1	100.0
	Valid Cases		1503		

v295 - environment: giving part of income (Q85A)

Q85

<SHOW CARD 85 – READ OUT AND CODE ONE ANSWER PER LINE!>

I am now going to read out some statements about the environment. For each one read out, can you tell me whether you agree strongly, agree, disagree or strongly disagree?

Q85.A I would give part of my income if I were certain that the money would be used to prevent environmental pollution

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 disagree
- 4 disagree strongly

v295, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	7	0.5	
-1	dk	М	87	5.8	
1	agree strongly		149	9.9	10.5
2	agree		464	30.7	32.8
3	disagree		434	28.7	30.6
4	disagree strongly		369	24.4	26.1
	Sum		1510	100.0	100.0
	Valid Cases		1416		

v296 - environment: approaching the limit of people (Q85B)

Q85

<SHOW CARD 85 - READ OUT AND CODE ONE ANSWER PER LINE!>

I am now going to read out some statements about the environment. For each one read out, can you tell me whether you agree strongly, agree, disagree or strongly disagree?

Q85.B We are approaching the limit of the number of people the earth can support

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 disagree
- 4 disagree strongly

v296, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	5	0.3	
-1	dk	М	131	8.7	
1	agree strongly		229	15.2	16.7
2	agree		636	42.1	46.3
3	disagree		373	24.7	27.1
4	disagree strongly		136	9.0	9.9
	Sum		1510	100.0	100.0
	Valid Cases		1374		

v297 - environment: interference produces disastrous consequences (Q85C)

Q85

<SHOW CARD 85 - READ OUT AND CODE ONE ANSWER PER LINE!>

I am now going to read out some statements about the environment. For each one read out, can you tell me whether you agree strongly, agree, disagree or strongly disagree?

Q85.C When humans interfere with nature it often produces disastrous consequences

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 disagree
- 4 disagree strongly

v297, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	55	3.6	
1	agree strongly		621	41.1	42.7
2	agree		669	44.3	46.0
3	disagree		148	9.8	10.2
4	disagree strongly		16	1.1	1.1
	Sum		1510	100.0	100.0
	Valid Cases		1454		

v298 - environment: human ingenuity insures earth remaining fit (Q85D)

Q85

<SHOW CARD 85 - READ OUT AND CODE ONE ANSWER PER LINE!>

I am now going to read out some statements about the environment. For each one read out, can you tell me whether you agree strongly, agree, disagree or strongly disagree?

Q85.D Human ingenuity will insure that the earth remains fit to live in

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 disagree
- 4 disagree strongly

v298, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	206	13.6	
1	agree strongly		133	8.8	10.2
2	agree		607	40.2	46.7
3	disagree		399	26.4	30.7
4	disagree strongly		162	10.7	12.5
	Sum		1510	100.1	100.0
	Valid Cases		1300		

v299 - environment: nature is strong enough to cope with impacts of industry (Q85E)

Q85

<SHOW CARD 85 – READ OUT AND CODE ONE ANSWER PER LINE!>

I am now going to read out some statements about the environment. For each one read out, can you tell me whether you agree strongly, agree, disagree or strongly disagree?

Q85.E The balance of nature is strong enough to cope with the impacts of modern industrial nations

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 disagree
- 4 disagree strongly

v299, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	174	11.5	
1	agree strongly		71	4.7	5.3
2	agree		334	22.1	25.1
3	disagree		618	40.9	46.4
4	disagree strongly		310	20.5	23.3
	Sum		1510	100.0	100.0
	Valid Cases		1333		

v300 - environment: humans were meant to rule over nature (Q85F)

Q85

<SHOW CARD 85 - READ OUT AND CODE ONE ANSWER PER LINE!>

I am now going to read out some statements about the environment. For each one read out, can you tell me whether you agree strongly, agree, disagree or strongly disagree?

Q85.F Humans were meant to rule over the rest of nature

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 disagree
- 4 disagree strongly

v300, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
-1	dk	М	80	5.3	
1	agree strongly		69	4.6	4.8
2	agree		255	16.9	17.9
3	disagree		518	34.3	36.3
4	disagree strongly		585	38.7	41.0
	Sum		1510	100.0	100.0
	Valid Cases		1427		

v301 - environment: if things continue we will experience a catastrophe (Q85G)

Q85

<SHOW CARD 85 – READ OUT AND CODE ONE ANSWER PER LINE!>

I am now going to read out some statements about the environment. For each one read out, can you tell me whether you agree strongly, agree, disagree or strongly disagree?

Q85.G If things continue on their present course, we will soon experience a major ecological catastrophe

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 agree strongly
- 2 agree
- 3 disagree
- 4 disagree strongly

v301, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	123	8.1	
1	agree strongly		455	30.1	32.9
2	agree		699	46.3	50.5
3	disagree		190	12.6	13.7
4	disagree strongly		41	2.7	3.0
	Sum		1510	99.9	100.0
	Valid Cases		1386		

AT54 - agree with: where there is strict authority, there is also justice (Q85_at)

Q85-at

<SHOW CARD 85 - READ OUT AND CODE ONE PER LINE>

How much do you agree or disagree with each of the following statements. You can choose between 5 levels.

1 = I agree completely with the statement; 5 = I disagree completely with the statement; the others lie in between these.

Q85A_at Where there is strict authority, there is also justice

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 agree completely
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT54, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	77	5.1	
1	agree completely		88	5.8	6.1
2	agree		296	19.6	20.7
3	neither agree nor disagree		319	21.1	22.3
4	disagree		413	27.4	28.8
5	disagree completely		317	21.0	22.1
	Sum		1510	100.0	100.0
	Valid Cases		1433		

AT55 - agree with: the most important thing that children have to learn is obedience (Q85B_at)

Q85-at

<SHOW CARD 85 - READ OUT AND CODE ONE PER LINE>

How much do you agree or disagree with each of the following statements. You can choose between 5 levels.

1 = I agree completely with the statement; 5 = I disagree completely with the statement; the others lie in between these.

Q85B_at The most important thing that children have to learn, is obedience

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 agree completely
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT55, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	2	0.1	
-1	dk	М	45	3.0	
1	agree completely		147	9.7	10.0
2	agree		425	28.1	29.0
3	neither agree nor disagree		379	25.1	25.9
4	disagree		343	22.7	23.4
5	disagree completely		170	11.3	11.6
	Sum		1510	100.1	100.0
	Valid Cases		1464		

AT56 - agree with: only those with sufficient status should take part in discussions/ decisions (Q85C_at)

Q85-at

<SHOW CARD 85 - READ OUT AND CODE ONE PER LINE>

How much do you agree or disagree with each of the following statements. You can choose between 5 levels.

1 = I agree completely with the statement; 5 = I disagree completely with the statement; the others lie in between these.

Q85C_at Only those who have earned sufficient status through hard work should have a right to take part in discussions and decisions

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 agree completely
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT56, weighted by weight

Value Label	Missing	Count	Percent	Valid Percent
-2 na	М	1	0.1	
-1 dk	М	59	3.9	
1 agree completely		89	5.9	6.1
2 agree		333	22.1	23.0
3 neither agree nor disagree		342	22.6	23.6
4 disagree		460	30.5	31.7
5 disagree completely		226	15.0	15.6
Sum		1510	100.0	100.0
Valid Cases		1450		

AT57 - agree with: young people have too much freedom (Q85D_at)

Q85-at

<SHOW CARD 85 - READ OUT AND CODE ONE PER LINE>

How much do you agree or disagree with each of the following statements. You can choose between 5 levels.

1 = I agree completely with the statement; 5 = I disagree completely with the statement; the others lie in between these.

Q85D_at All that freedom that young people have nowadays can't be good for them

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 agree completely
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT57, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	48	3.2	
1	agree completely		232	15.4	15.9
2	agree		517	34.2	35.4
3	neither agree nor disagree		358	23.7	24.5
4	disagree		259	17.2	17.7
5	disagree completely		96	6.4	6.6
	Sum		1510	100.0	100.0
	Valid Cases		1462		

AT58 - agree with: the meaning of life consists of attaining a respectable position (Q85E_at)

Q85-at

<SHOW CARD 85 - READ OUT AND CODE ONE PER LINE>

How much do you agree or disagree with each of the following statements. You can choose between 5 levels.

1 = I agree completely with the statement; 5 = I disagree completely with the statement; the others lie in between these.

Q85E_at The meaning of life consists of attaining a respectable position

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 agree completely
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT58, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	2	0.1	
-1	dk	М	46	3.0	
1	agree completely		80	5.3	5.5
2	agree		353	23.4	24.1
3	neither agree nor disagree		450	29.8	30.8
4	disagree		383	25.4	26.2
5	disagree completely		196	13.0	13.4
	Sum		1510	100.0	100.0
	Valid Cases		1462		

AT59 - agree with: security and prosperity are more important than freedom (Q85F_at)

Q85-at

<SHOW CARD 85 - READ OUT AND CODE ONE PER LINE>

How much do you agree or disagree with each of the following statements. You can choose between 5 levels.

1 = I agree completely with the statement; 5 = I disagree completely with the statement; the others lie in between these.

Q85F_at Security and prosperity are more important than freedom

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 agree completely
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT59, weighted by weight

Value Label	Missing	Count	Percent	Valid Percent
-2 na	М	2	0.1	
-1 dk	М	52	3.4	
1 agree completely		76	5.0	5.2
2 agree		333	22.1	22.9
3 neither agree nor disagree		442	29.3	30.4
4 disagree		372	24.6	25.6
5 disagree completely		232	15.4	15.9
Sum		1510	99.9	100.0
Valid Cases		1456		

AT60 - agree with: purpose of a job should first and foremost consist in guaranteeing a secure income (Q85G_at)

Q85-at

<SHOW CARD 85 - READ OUT AND CODE ONE PER LINE>

How much do you agree or disagree with each of the following statements. You can choose between 5 levels.

1 = I agree completely with the statement; 5 = I disagree completely with the statement; the others lie in between these.

Q85G_at The purpose of a job should first and foremost consist in guaranteeing a secure income

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 agree completely
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT60, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	35	2.3	
1	agree completely		377	25.0	25.6
2	agree		728	48.2	49.4
3	neither agree nor disagree		256	17.0	17.4
4	disagree		90	6.0	6.1
5	disagree completely		22	1.5	1.5
	Sum		1510	99.9	100.0
	Valid Cases		1474		

AT61 - agree with: its important to live life as comfortably\pleasant\convenient as possible (Q85H_at)

Q85-at

<SHOW CARD 85 - READ OUT AND CODE ONE PER LINE>

How much do you agree or disagree with each of the following statements. You can choose between 5 levels.

1 = I agree completely with the statement; 5 = I disagree completely with the statement; the others lie in between these.

Q85H_at It's important to live life as comfortably/ pleasant/ convenient as possible

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 agree completely
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT61, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	2	0.1	
-1	dk	М	31	2.1	
1	agree completely		359	23.8	24.3
2	agree		733	48.5	49.6
3	neither agree nor disagree		297	19.7	20.1
4	disagree		73	4.8	4.9
5	disagree completely		16	1.1	1.1
	Sum		1510	100.1	100.0
	Valid Cases		1477		

AT62 - agree with: its important for a person to be happy. How they do that is everyones own business (Q85I_at)

Q85-at

<SHOW CARD 85 - READ OUT AND CODE ONE PER LINE>

How much do you agree or disagree with each of the following statements. You can choose between 5 levels.

1 = I agree completely with the statement; 5 = I disagree completely with the statement; the others lie in between these.

Q85I_at It's important for a person to be happy. How they do that is everyone's own business

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 agree completely
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT62, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	21	1.4	
1	agree completely		402	26.6	27.0
2	agree		734	48.6	49.3
3	neither agree nor disagree		217	14.4	14.6
4	disagree		96	6.4	6.5
5	disagree completely		39	2.6	2.6
	Sum		1510	100.0	100.0
	Valid Cases		1488		

AT63 - agree with: everyone has to solve their problems themselves\ on their own (Q85J_at)

Q85-at

<SHOW CARD 85 - READ OUT AND CODE ONE PER LINE>

How much do you agree or disagree with each of the following statements. You can choose between 5 levels.

1 = I agree completely with the statement; 5 = I disagree completely with the statement; the others lie in between these.

Q85J_at Everyone has to solve their problems themselves/on their own

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 agree completely
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT63, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	17	1.1	
1	agree completely		365	24.2	24.4
2	agree		696	46.1	46.6
3	neither agree nor disagree		275	18.2	18.4
4	disagree		139	9.2	9.3
5	disagree completely		18	1.2	1.2
	Sum		1510	100.0	100.0
	Valid Cases		1493		

AT64 - agree with: foreigners should adapt their lifestyle more to that of the local people (Q85K_at)

Q85-at

<SHOW CARD 85 - READ OUT AND CODE ONE PER LINE>

How much do you agree or disagree with each of the following statements. You can choose between 5 levels.

1 = I agree completely with the statement; 5 = I disagree completely with the statement; the others lie in between these.

Q85K_at Foreigners should adapt their lifestyle more to that of the local people

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 agree completely
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT64, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	23	1.5	
1	agree completely		613	40.6	41.3
2	agree		577	38.2	38.8
3	neither agree nor disagree		199	13.2	13.4
4	disagree		77	5.1	5.2
5	disagree completely		20	1.3	1.3
	Sum		1510	100.0	100.0
	Valid Cases		1486		

AT65 - agree with: when jobs become scarce, foreigners should be sent back to their home countries (Q85L_at)

Q85-at

<SHOW CARD 85 - READ OUT AND CODE ONE PER LINE>

How much do you agree or disagree with each of the following statements. You can choose between 5 levels.

1 = I agree completely with the statement; 5 = I disagree completely with the statement; the others lie in between these.

Q85L_at When jobs become scarce, foreigners should be sent back to their home countries

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 agree completely
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT65, weighted by weight

Value Label	Missing	Count	Percent	Valid Percent
-2 na	М	1	0.1	
-1 dk	М	40	2.6	
1 agree completely		307	20.3	20.9
2 agree		407	27.0	27.7
3 neither agree nor disagree		324	21.5	22.1
4 disagree		306	20.3	20.8
5 disagree completely		125	8.3	8.5
Sum		1510	100.0	100.0
Valid Cases		1469		

AT66 - agree with: foreigners should be banned from all political participation in our country (Q85M_at)

Q85-at

<SHOW CARD 85 - READ OUT AND CODE ONE PER LINE>

How much do you agree or disagree with each of the following statements. You can choose between 5 levels.

1 = I agree completely with the statement; 5 = I disagree completely with the statement; the others lie in between these.

Q85M_at Foreigners should be banned from all political participation in our country

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 agree completely
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT66, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	51	3.4	
1	agree completely		295	19.5	20.2
2	agree		336	22.3	23.0
3	neither agree nor disagree		289	19.1	19.8
4	disagree		362	24.0	24.8
5	disagree completely		177	11.7	12.1
	Sum		1510	100.0	100.0
	Valid Cases		1459		

AT67 - agree with: foreigners should choose their spouses among their kind (Q85N_at)

Q85-at

<SHOW CARD 85 - READ OUT AND CODE ONE PER LINE>

How much do you agree or disagree with each of the following statements. You can choose between 5 levels.

1 = I agree completely with the statement; 5 = I disagree completely with the statement; the others lie in between these.

Q85N_at Foreigners should choose their spouses among their own kind

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 agree completely
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT67, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	50	3.3	
1	agree completely		177	11.7	12.1
2	agree		274	18.1	18.8
3	neither agree nor disagree		377	25.0	25.8
4	disagree		381	25.2	26.1
5	disagree completely		252	16.7	17.2
	Sum		1510	100.1	100.0
	Valid Cases		1460		

AT68 - agree with: it should be permitted to take away from the wealthy in order to give it to those in need (Q850_at)

Q85-at

<SHOW CARD 85 - READ OUT AND CODE ONE PER LINE>

How much do you agree or disagree with each of the following statements. You can choose between 5 levels.

1 = I agree completely with the statement; 5 = I disagree completely with the statement; the others lie in between these.

Q85O_at It should be permitted to take away from the wealthy in order to give it to those in need

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 agree completely
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT68, weighted by weight

Value Label	Missing	Count	Percent	Valid Percent
-2 na	М	2	0.1	
-1 dk	М	45	3.0	
1 agree completely		173	11.5	11.8
2 agree		385	25.5	26.3
3 neither agree nor disagree		351	23.2	24.0
4 disagree		354	23.4	24.2
5 disagree completely		200	13.2	13.7
Sum		1510	100.0	100.0
Valid Cases		1463		

AT69 - agree with: differences in income should be reduced (Q85P_at)

Q85-at

<SHOW CARD 85 - READ OUT AND CODE ONE PER LINE>

How much do you agree or disagree with each of the following statements. You can choose between 5 levels.

1 = I agree completely with the statement; 5 = I disagree completely with the statement; the others lie in between these.

Q85P_at Differences in income should be reduced

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 agree completely
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT69, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	35	2.3	
1	agree completely		471	31.2	31.9
2	agree		645	42.7	43.7
3	neither agree nor disagree		243	16.1	16.5
4	disagree		93	6.2	6.3
5	disagree completely		23	1.5	1.6
	Sum		1510	100.1	100.0
	Valid Cases		1474		

AT70 - agree with: if we could all get by on a little bit less, poverty would soon be gone (Q85Q_at)

Q85-at

<SHOW CARD 85 - READ OUT AND CODE ONE PER LINE>

How much do you agree or disagree with each of the following statements. You can choose between 5 levels.

1 = I agree completely with the statement; 5 = I disagree completely with the statement; the others lie in between these.

Q85Q_at If we could all get by on a little bit less, poverty would soon be gone

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 agree completely
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT70, weighted by weight

Value I	Label	Missing	Count	Percent	Valid Percent
-2 ו	na	М	2	0.1	
-1 (dk	М	61	4.0	
1 8	agree completely		191	12.6	13.2
2 8	agree		604	40.0	41.7
3 ו	neither agree nor disagree		319	21.1	22.0
4 (disagree		244	16.2	16.9
5 (disagree completely		89	5.9	6.2
	Sum		1510	100.0	100.0
,	Valid Cases		1446		

AT71 - agree with: the most important thing that children have to learn, is sharing with others (Q85R_at)

Q85-at

<SHOW CARD 85 - READ OUT AND CODE ONE PER LINE>

How much do you agree or disagree with each of the following statements. You can choose between 5 levels.

1 = I agree completely with the statement; 5 = I disagree completely with the statement; the others lie in between these.

Q85R_at The most important thing that children have to learn, is sharing with others

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 agree completely
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT71, weighted by weight

Value Label	Missing	Count	Percent	Valid Percent
-2 na	М	2	0.1	
-1 dk	М	31	2.1	
1 agree completely		316	20.9	21.4
2 agree		800	53.0	54.2
3 neither agree nor disagree		263	17.4	17.8
4 disagree		72	4.8	4.9
5 disagree completely		25	1.7	1.7
Sum		1510	99.9	100.0
Valid Cases		1476		

AT72 - agree with: sharing can be learned best from ones own family (Q85S_at)

Q85-at

<SHOW CARD 85 - READ OUT AND CODE ONE PER LINE>

How much do you agree or disagree with each of the following statements. You can choose between 5 levels.

1 = I agree completely with the statement; 5 = I disagree completely with the statement; the others lie in between these.

Q85S_at Sharing can be learned best from one's own family

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 agree completely
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT72, weighted by weight

Value Label	Missing	Count	Percent	Valid Percent
-2 na	М	2	0.1	
-1 dk	М	28	1.9	
1 agree completely		481	31.9	32.5
2 agree		775	51.3	52.4
3 neither agree nor disagree		171	11.3	11.6
4 disagree		39	2.6	2.6
5 disagree completely		14	0.9	0.9
Sum		1510	100.0	100.0
Valid Cases		1480		

AT73 - agree with: one cannot learn to solve conflicts without a family (Q85T_at)

Q85-at

<SHOW CARD 85 - READ OUT AND CODE ONE PER LINE>

How much do you agree or disagree with each of the following statements. You can choose between 5 levels.

1 = I agree completely with the statement; 5 = I disagree completely with the statement; the others lie in between these.

Q85T_at One cannot learn to solve conflicts without a family

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 agree completely
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT73, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	37	2.5	
1	agree completely		328	21.7	22.4
2	agree		588	38.9	40.1
3	neither agree nor disagree		288	19.1	19.6
4	disagree		202	13.4	13.8
5	disagree completely		61	4.0	4.2
	Sum		1510	99.9	100.0
	Valid Cases		1469		

AT74 - agree with: everything ends with death (Q85U_at)

Q85-at

<SHOW CARD 85 - READ OUT AND CODE ONE PER LINE>

How much do you agree or disagree with each of the following statements. You can choose between 5 levels.

1 = I agree completely with the statement; 5 = I disagree completely with the statement; the others lie in between these.

Q85U_at Everything ends with death

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 agree completely
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT74, weighted by weight

Value Label	Missing	Count	Percent	Valid Percent
-2 na	М	2	0.1	
-1 dk	М	211	14.0	
1 agree completely		244	16.2	18.8
2 agree		258	17.1	19.9
3 neither agree nor disagree		212	14.0	16.3
4 disagree		330	21.9	25.4
5 disagree completely		254	16.8	19.6
Sum		1510	100.1	100.0
Valid Cases		1298		

AT75 - agree with: I hope there is a life after death (Q85V_at)

Q85-at

<SHOW CARD 85 - READ OUT AND CODE ONE PER LINE>

How much do you agree or disagree with each of the following statements. You can choose between 5 levels.

1 = I agree completely with the statement; 5 = I disagree completely with the statement; the others lie in between these.

Q85V_at I hope there is a life after death

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 agree completely
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT75, weighted by weight

Value Label	Missing	Count	Percent	Valid Percent
-2 na	М	5	0.3	
-1 dk	М	154	10.2	
1 agree completely		314	20.8	23.2
2 agree		456	30.2	33.8
3 neither agree nor disagree		260	17.2	19.2
4 disagree		155	10.3	11.5
5 disagree completely		166	11.0	12.3
Sum		1510	100.0	100.0
Valid Cases		1352		

AT76 - agree with: the dead will be resurrected in body and soul (Q85W_at)

Q85-at

<SHOW CARD 85 - READ OUT AND CODE ONE PER LINE>

How much do you agree or disagree with each of the following statements. You can choose between 5 levels.

1 = I agree completely with the statement; 5 = I disagree completely with the statement; the others lie in between these.

Q85W_at The dead will be resurrected in body and soul

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 agree completely
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT76, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	7	0.5	
-1	dk	М	287	19.0	
1	agree completely		114	7.5	9.4
2	agree		214	14.2	17.6
3	neither agree nor disagree		275	18.2	22.6
4	disagree		280	18.5	23.0
5	disagree completely		333	22.1	27.4
	Sum		1510	100.0	100.0
	Valid Cases		1216		

AT77 - agree with: life meaningful only because God exists (Q85X_at)

Q85-at

<SHOW CARD 85 - READ OUT AND CODE ONE PER LINE>

How much do you agree or disagree with each of the following statements. You can choose between 5 levels.

1 = I agree completely with the statement; 5 = I disagree completely with the statement; the others lie in between these.

Q85X_at Life is meaningful only because God exists

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 agree completely
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT77, weighted by weight

Value Label	Missing	Count	Percent	Valid Percent
-2 na	М	3	0.2	
-1 dk	М	143	9.5	
1 agree completely		138	9.1	10.1
2 agree		370	24.5	27.1
3 neither agree nor disagree		318	21.1	23.3
4 disagree		257	17.0	18.8
5 disagree completely		281	18.6	20.6
Sum		1510	100.0	100.0
Valid Cases		1364		

AT78 - agree with: the meaning of life is to make the most of it (Q85Y_at)

Q85-at

<SHOW CARD 85 - READ OUT AND CODE ONE PER LINE>

How much do you agree or disagree with each of the following statements. You can choose between 5 levels.

1 = I agree completely with the statement; 5 = I disagree completely with the statement; the others lie in between these.

Q85Y_at The meaning of life is to make the most of it

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 agree completely
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT78, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	36	2.4	
1	agree completely		407	27.0	27.6
2	agree		698	46.2	47.4
3	neither agree nor disagree		223	14.8	15.1
4	disagree		114	7.5	7.7
5	disagree completely		32	2.1	2.2
	Sum		1510	100.1	100.0
	Valid Cases		1473		

AT79 - agree with: i dont know what humans exist\live for (Q85Z_at)

Q85-at

<SHOW CARD 85 - READ OUT AND CODE ONE PER LINE>

How much do you agree or disagree with each of the following statements. You can choose between 5 levels.

1 = I agree completely with the statement; 5 = I disagree completely with the statement; the others lie in between these.

Q85Z_at I don't know what humans exist/live for

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 agree completely
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT79, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	7	0.5	
-1	dk	М	78	5.2	
1	agree completely		46	3.0	3.2
2	agree		129	8.5	9.0
3	neither agree nor disagree		285	18.9	20.0
4	disagree		509	33.7	35.7
5	disagree completely		457	30.3	32.0
	Sum		1510	100.1	100.0
	Valid Cases		1425		

AT80 - agree with: decisions should be made by those who are successful in life (Q85AA_at)

Q85-at

<SHOW CARD 85 - READ OUT AND CODE ONE PER LINE>

How much do you agree or disagree with each of the following statements. You can choose between 5 levels.

1 = I agree completely with the statement; 5 = I disagree completely with the statement; the others lie in between these.

Q85AA_at Decisions should be made by those who are successful in life

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 agree completely
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT80, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	2	0.1	
-1	dk	М	38	2.5	
1	agree completely		66	4.4	4.5
2	agree		273	18.1	18.6
3	neither agree nor disagree		407	27.0	27.7
4	disagree		448	29.7	30.5
5	disagree completely		275	18.2	18.7
	Sum		1510	99.9	100.0
	Valid Cases		1470		

AT81 - agree with: the weaker members of society should be subordinate to the stronger ones (Q85BB_at)

Q85-at

<SHOW CARD 85 - READ OUT AND CODE ONE PER LINE>

How much do you agree or disagree with each of the following statements. You can choose between 5 levels.

1 = I agree completely with the statement; 5 = I disagree completely with the statement; the others lie in between these.

Q85BB_at The weaker members of society should be subordinate to the stronger ones

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 agree completely
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT81, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	2	0.1	
-1	dk	М	39	2.6	
1	agree completely		40	2.6	2.7
2	agree		176	11.7	12.0
3	neither agree nor disagree		329	21.8	22.4
4	disagree		496	32.8	33.8
5	disagree completely		428	28.3	29.1
	Sum		1510	100.0	100.0
	Valid Cases		1469		

AT82 - agree with: everyone is the architect of their own fortune (Q85CC_at)

Q85-at

<SHOW CARD 85 - READ OUT AND CODE ONE PER LINE>

How much do you agree or disagree with each of the following statements. You can choose between 5 levels.

1 = I agree completely with the statement; 5 = I disagree completely with the statement; the others lie in between these.

Q85CC_at Everyone is the architect of their own fortune

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 agree completely
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT82, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	32	2.1	
1	agree completely		376	24.9	25.5
2	agree		694	46.0	47.0
3	neither agree nor disagree		306	20.3	20.7
4	disagree		76	5.0	5.1
5	disagree completely		25	1.7	1.7
	Sum		1510	100.0	100.0
	Valid Cases		1477		

AT83 - agree with: everyone is responsible for their own lack of success (Q85DD_at)

Q85-at

<SHOW CARD 85 - READ OUT AND CODE ONE PER LINE>

How much do you agree or disagree with each of the following statements. You can choose between 5 levels.

1 = I agree completely with the statement; 5 = I disagree completely with the statement; the others lie in between these.

Q85DD_at Everyone is responsible for their own lack of success

- -5 other missing
- -3 nap
- -2 na
- -1 dk
- 1 agree completely
- 2 agree
- 3 neither agree nor disagree
- 4 disagree
- 5 disagree completely

Remarks on original question in the field questionnaire:

Additional country-specific variable. English translation as provided by country.

AT83, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	2	0.1	
-1	dk	М	39	2.6	
1	agree completely		104	6.9	7.1
2	agree		370	24.5	25.2
3	neither agree nor disagree		527	34.9	35.9
4	disagree		345	22.8	23.5
5	disagree completely		123	8.1	8.4
	Sum		1510	100.0	100.0
	Valid Cases		1469		

v302 - sex respondent (Q86)

Q86

INCLUDE COUNTRY-SPECIFIC QUESTIONS HERE/ <ASK ALL> / Demographics

Sex of respondent

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 male
- 2 female

v302, weighted by weight

Value	Label	Missing Count	Percent	Valid Percent
1	male	726	48.1	48.1
2	female	784	51.9	51.9
	Sum	1510	100.0	100.0
	Valid Cases	1510		

v303 - year of birth respondent (Q87)

Q87

Can you tell me your year of birth, please

19..

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk

v304 - having [countrys] nationality (Q88)

Q88

Do you have [COUNTRY'S] nationality?

<NATIONALITY IS PASSPORT!>

<IF CODE 1, -1, -2: GO TO Q90>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v304, weighted by weight

		-		
Value	Label	Missing Count	Percent	Valid Percent
1	yes	1477	97.8	97.8
2	? no	33	2.2	2.2
	Sum	1510	100.0	100.0
	Valid Cases	1510		

152000000

156000000

Chile

China

v305b - respondents nationality - ISO 3166-1 code/Staatsbürgerschaft des Befragten ISO 3166-1 Code (Q89) Q89 What is your nationality? <NATIONALITY IS PASSPORT!> <CODING INSTRUCTION: TO BE CODED INTO PRE-SPECIFIED ISO 3166-1 (2-CHARACTER)> -5 other missing -4 question not asked -3 nap -2 na -1 dk 4000000 Afghanistan 8000000 Albania 12000000 Algeria 20000000 Andorra 24000000 Angola 28000000 Antigua and Barbuda 31000000 Azerbaijan 32000000 Argentina 36000000 Australia 40000000 Austria 44000000 Bahamas 48000000 Bahrain 50000000 Bangladesh 51000000 Armenia 52000000 Barbados 56000000 Belgium 64000000 Bhutan 68000000 Bolivia Bosnia-Herzegovina 70000000 72000000 Botswana 76000000 Brazil 84000000 Belize 90000000 Solomon Islands 96000000 Brunei 100000000 Bulgaria 104000000 Burma (Myanmar) 108000000 Burundi 112000000 Belarus 116000000 Cambodia 120000000 Cameroon 124000000 Canada 132000000 Cape Verde 140000000 Central African Republic 144000000 Sri Lanka 148000000 Chad

158000000	Taiwan
170000000	Colombia
174000000	Comoros
178000000	Congo, Rep. of
180000000	Congo, Dem. Rep. Of
188000000	Costa Rica
191000000	Croatia
192000000	Cuba
196100000	Cyprus
196200000	Northern Cyprus
200000000	Czechoslovakia
203000000	Czech Republic
204000000	Benin
208000000	Denmark
212000000	Dominica
214000000	Dominican Republic
218000000	Ecuador
222000000	El Salvador
226000000	Equatorial Guinea
231000000	Ethiopia
232000000	Eritrea
233000000	Estonia
242000000	Fiji
246000000	Finland
248000000	Ålands Islands
250000000	France
254000000	French Guiana
262000000	Djibouti
266000000	Gabon
268000000	Georgia
270000000	Gambia
275000000	Palestinian Territory, occupied
276000000	Germany
288000000	Ghana
292000000	Gibraltar
296000000	Kiribati
300000000	Greece
308000000	Grenada
312000000	Guadeloupe
320000000	Guatemala
324000000	Guinea
328000000	Guyana
332000000	Haiti
340000000	Honduras
344000000	Hong Kong
348000000	Hungary
352000000	Iceland
356000000	India
360000000	Indonesia

Iraq

Ireland

Israel

Italy

Côte d'Ivoire

Jamaica

Japan

364000000 368000000

372000000

376000000

380000000

384000000

388000000

392000000

GESIS StudyNo 4754, v1.1.0, http://dx.doi.org/doi:10.4232/1.10152

398000000 Kazakhstan 400000000 Jordan 40400000 Kenya 408000000 Korea, North 410000000 Korea, South 414000000 Kuwait 417000000 Kyrgyzstan 418000000 Laos 422000000 Lebanon 426000000 Lesotho 428000000 Latvia 430000000 Liberia 434000000 Libya 438000000 Liechtenstein 440000000 Lithuania 442000000 Luxembourg 446000000 Macau 450000000 Madagascar 454000000 Malawi 458000000 Malaysia 462000000 Maldives 466000000 Mali 470000000 Malta 474000000 Martinique 478000000 Mauritania 480000000 Mauritius 484000000 Mexico 492000000 Monaco 496000000 Mongolia 498000000 Moldava 504000000 Morocco 508000000 Mozambique 512000000 Oman 516000000 Namibia 520000000 Nauru 524000000 Nepal 528000000 Netherlands 530000000 **Netherlands Antilles** 533000000 Aruba 548000000 Vanuatu 554000000 New Zealand

558000000	Nicaragua
562000000	Niger
566000000	Nigeria
578000000	Norway
583000000	Micronesia
584000000	Marshall Islands
585000000	Palau
586000000	Pakistan
591000000	Panama
598000000	Papua New Guinea
600000000	Paraguay
604000000	Peru
608000000	Philippines
616000000	Poland
620000000	Portugal
624000000	Guinea-Bissau
626000000	East Timor
630000000	Puerto Rico
634000000	Qatar
638000000	Réunion
642000000	Romania
643000000	Russia
646000000	Rwanda
659000000	St. Kitts and Nevis
662000000	St. Lucia
663000000	Saint Martin
670000000	St. Vincent and the Grenadines
674000000	San Marino
678000000	São Tomé and Príncipe
682000000	Saudi Arabia
686000000	Senegal
690000000	Seychelles
694000000	Sierra Leone
702000000	Singapore
703000000	Slovakia
704000000	Vietnam
705000000	Slovenia
706000000	Somalia
710000000	South Africa
716000000	Zimbabwe
724000000	Spain
736000000	Sudan
740000000	Suriname
748000000	Swaziland
752000000	Sweden
756000000	Switzerland
760000000	Syria
762000000	Tajikistan
764000000	Thailand
, 0-000000	manum

```
768000000
           Togo
776000000
           Tonga
780000000
           Trinidad and Tobago
784000000
           United Arab Emirates
788000000
           Tunisia
792000000
           Turkey
795000000
           Turkmenistan
798000000
           Tuvalu
800000000
           Uganda
804000000
           Ukraine
807000000
           Macedonia
810000000
           USSR
818000000
           Egypt
826100000
           Great Britain
826200000
           Northern Ireland
834000000
           Tanzania
840000000
           United States
854000000
           Burkina Faso
858000000
           Uruguay
860000000
           Uzbekistan
862000000
           Venezuela
882000000
           Samoa
887000000
890000000
           Yugoslavia, Socialist Federal Republic of
891000000
           Serbia and Montenegro
894000000
           Zambia
900000000
           West-Germany
901000000 East-Germany
911000000
           Serbia
912000000 Montenegro
915000000
           Kosovo
916000000 Austria-Hungary
917000000
           Alien's passport
```

Note:

Verbatim recorded answers standardized according to ISO 3166-1 alpha-2 code.

Data for original string variable is not available in this dataset.

v306 - respondent born in [country] (Q90)

Q90

Were you born in [COUNTRY]?

<IF CODE 1, -1, -2: GO TO Q93>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v306, weighted by weight

Value	Label	Missing Count	Percent	Valid Percent
1	yes	1412	93.5	93.5
2	no	98	6.5	6.5
	Sum	1510	100.0	100.0
	Valid Cases	1510		

v307b - respondents country of birth - ISO 3166-1 code/Geburtsland des Befragten ISO 3166-1 Code (Q91) Q91

In which country were you born?

Write in: ...

<CODING INSTRUCTION: TO BE CODED INTO PRE-SPECIFIED ISO 3166-1 (2-CHARACTER)>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 4000000 Afghanistan
- 8000000 Albania
- 12000000 Algeria
- 20000000 Andorra
- 24000000 Angola
- 28000000 Antigua and Barbuda
- 31000000 Azerbaijan
- 32000000 Argentina
- 36000000 Australia
- 40000000 Austria
- 44000000 Bahamas
- 48000000 Bahrain
- 50000000 Bangladesh
- 51000000 Armenia
- 52000000 Barbados
- 56000000 Belgium
- 64000000 Bhutan
- 68000000 Bolivia
- 7000000 Bosnia-Herzegovina
- 72000000 Botswana
- 76000000 Brazil
- 84000000 Belize
- 90000000 Solomon Islands
- 96000000 Brunei
- 100000000 Bulgaria
- 104000000 Burma (Myanmar)
- 108000000 Burundi
- 112000000 Belarus
- 116000000 Cambodia
- 120000000 Cameroon
- 124000000 Canada
- 132000000 Cape Verde
- 14000000 Central African Republic
- 144000000 Sri Lanka
- 148000000 Chad
- 152000000 Chile
- 156000000 China

158000000	Taiwan
170000000	Colombia
174000000	Comoros
178000000	Congo, Rep. of
180000000	Congo, Dem. Rep. Of
188000000	Costa Rica
191000000	Croatia
192000000	Cuba
196100000	Cyprus
196200000	Northern Cyprus
200000000	Czechoslovakia
203000000	Czech Republic
204000000	Benin
208000000	Denmark
212000000	Dominica
214000000	
218000000	Ecuador
222000000	El Salvador
226000000	Equatorial Guinea
231000000	Ethiopia
232000000	Eritrea
233000000	Estonia
242000000	Fiji
246000000	Finland
248000000	Ålands Islands
250000000	France
254000000	French Guiana
262000000	Djibouti
266000000	Gabon
268000000	Georgia
270000000	Gambia
275000000	Palestinian Territory, occupied
276000000	Germany
288000000	Ghana
292000000	Gibraltar
296000000	Kiribati
300000000	Greece
308000000	Grenada
312000000	Guadeloupe
320000000	Guatemala
324000000	Guinea
328000000	Guyana
332000000	Haiti
340000000	Honduras
344000000	Hong Kong
348000000	Hungary
352000000	Iceland
356000000	India

360000000 Indonesia

Iraq

Ireland

Israel

Italy

364000000 368000000

372000000

376000000

380000000

GESIS StudyNo 4754, v1.1.0, http://dx.doi.org/doi:10.4232/1.10152

384000000 Côte d'Ivoire 388000000 Jamaica 392000000 Japan 398000000 Kazakhstan 400000000 Jordan 40400000 Kenya 408000000 Korea, North 410000000 Korea, South 414000000 Kuwait 417000000 Kyrgyzstan 418000000 Laos 422000000 Lebanon 426000000 Lesotho 428000000 Latvia 430000000 Liberia 434000000 Libya 438000000 Liechtenstein 440000000 Lithuania 442000000 Luxembourg 446000000 Macau 450000000 Madagascar 454000000 Malawi 458000000 Malaysia 462000000 Maldives 466000000 Mali 470000000 Malta 474000000 Martinique 478000000 Mauritania 480000000 Mauritius 484000000 Mexico 492000000 Monaco 496000000 Mongolia 498000000 Moldava 504000000 Morocco 508000000 Mozambique 512000000 Oman 516000000 Namibia 520000000 Nauru 524000000 Nepal 528000000 Netherlands 530000000 **Netherlands Antilles** 533000000 Aruba 548000000 Vanuatu 554000000 New Zealand

558000000	Nicaragua
562000000	Niger
566000000	Nigeria
578000000	Norway
583000000	Micronesia
584000000	Marshall Islands
585000000	Palau
586000000	Pakistan
591000000	Panama
598000000	Papua New Guinea
600000000	Paraguay
604000000	Peru
608000000	Philippines
616000000	Poland
620000000	Portugal
624000000	Guinea-Bissau
626000000	East Timor
630000000	Puerto Rico
634000000	Qatar
638000000	Réunion
642000000	Romania
643000000	Russia
646000000	Rwanda
659000000	St. Kitts and Nevis
662000000	St. Lucia
663000000	Saint Martin
670000000	St. Vincent and the Grenadines
674000000	San Marino
678000000	São Tomé and Príncipe
682000000	Saudi Arabia
686000000	Senegal
690000000	Seychelles
694000000	Sierra Leone
702000000	Singapore
703000000	Slovakia
704000000	Vietnam
705000000	Slovenia
706000000	Somalia
710000000	South Africa
716000000	Zimbabwe
724000000	Spain
736000000	Sudan
740000000	Suriname
748000000	Swaziland
752000000	Sweden
756000000	Switzerland
760000000	Syria
762000000	Tajikistan
764000000	Thailand

```
768000000
           Togo
776000000
           Tonga
780000000
           Trinidad and Tobago
784000000
           United Arab Emirates
788000000
           Tunisia
792000000
           Turkey
795000000
           Turkmenistan
798000000
           Tuvalu
800000000
           Uganda
804000000
           Ukraine
807000000
           Macedonia
810000000
           USSR
818000000
           Egypt
826100000
           Great Britain
826200000
           Northern Ireland
834000000
           Tanzania
840000000
           United States
854000000
           Burkina Faso
858000000
           Uruguay
860000000
           Uzbekistan
862000000
           Venezuela
882000000
           Samoa
887000000
890000000
           Yugoslavia, Socialist Federal Republic of
891000000
           Serbia and Montenegro
894000000
           Zambia
900000000
           West-Germany
901000000 East-Germany
911000000
           Serbia
912000000 Montenegro
915000000
           Kosovo
916000000 Austria-Hungary
917000000
           Alien's passport
```

Note:

Verbatim recorded answers standardized according to ISO 3166-1 alpha-2 code.

Data for original string variable is not available in this dataset.

v308 - year in which respondent came to live in [country] (Q92)

Q92

Can you tell me in which year you first came to live in [COUNTRY]

Write in year: ...

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk

v309 - father born in [country] (Q93)

Q93

<ASK ALL>

Was your father born in [COUNTRY]?

<IF CODE 1, -1, -2: GO TO Q95>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v309, weighted by weight

	-	•	-			
Value	Label		Missing	Count	Percent	Valid Percent
-1	dk		М	12	0.8	
1	yes			1326	87.8	88.5
2	no			172	11.4	11.5
	Sum			1510	100.0	100.0
	Valid Cas	ses		1498		

v310b - fathers country of birth - ISO 3166-1 code/Geburtsland des Vaters ISO 3166-1 Code (Q94)

Q94

In which country was your father born?

Write in: ...

<CODING INSTRUCTION: TO BE CODED INTO PRE-SPECIFIED ISO 3166-1 (2-CHARACTER)>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 4000000 Afghanistan
- 8000000 Albania
- 12000000 Algeria
- 20000000 Andorra
- 24000000 Angola
- 28000000 Antigua and Barbuda
- 31000000 Azerbaijan
- 32000000 Argentina
- 36000000 Australia
- 40000000 Austria
- 44000000 Bahamas
- 48000000 Bahrain
- 50000000 Bangladesh
- 51000000 Armenia
- 52000000 Barbados
- 56000000 Belgium
- 64000000 Bhutan
- 68000000 Bolivia
- 7000000 Bosnia-Herzegovina
- 72000000 Botswana
- 76000000 Brazil
- 84000000 Belize
- 90000000 Solomon Islands
- 96000000 Brunei
- 100000000 Bulgaria
- 104000000 Burma (Myanmar)
- 108000000 Burundi
- 112000000 Belarus
- 116000000 Cambodia
- 120000000 Cameroon
- 124000000 Canada
- 132000000 Cape Verde
- 140000000 Central African Republic
- 144000000 Sri Lanka
- 148000000 Chad
- 152000000 Chile
- 156000000 China

158000000	Taiwan
170000000	Colombia
174000000	Comoros
178000000	Congo, Rep. of
180000000	Congo, Dem. Rep. Of
188000000	Costa Rica
191000000	Croatia
192000000	Cuba
196100000	Cyprus
196200000	Northern Cyprus
200000000	Czechoslovakia
203000000	Czech Republic
204000000	Benin
208000000	Denmark
212000000	Dominica
214000000	Dominican Republic
218000000	Ecuador
222000000	El Salvador
226000000	Equatorial Guinea
231000000	Ethiopia
232000000	Eritrea
233000000	Estonia
242000000	Fiji
246000000	Finland
248000000	Ålands Islands
250000000	France
254000000	French Guiana
262000000	Djibouti
266000000	Gabon
268000000	Georgia
270000000	Gambia
275000000	Palestinian Territory, occupied
276000000	Germany
288000000	Ghana
292000000	Gibraltar
296000000	Kiribati
300000000	Greece
308000000	Grenada
312000000	Guadeloupe
320000000	Guatemala
324000000	Guinea
328000000	Guyana
332000000	Haiti
340000000	Honduras
344000000	Hong Kong
348000000	Hungary
352000000	Iceland
356000000	India

360000000 Indonesia

Iraq

Ireland

Israel

Italy

364000000 368000000

372000000

376000000

380000000

GESIS StudyNo 4754, v1.1.0, http://dx.doi.org/doi:10.4232/1.10152

384000000 Côte d'Ivoire 388000000 Jamaica 392000000 Japan 398000000 Kazakhstan 400000000 Jordan 40400000 Kenya 408000000 Korea, North 410000000 Korea, South 414000000 Kuwait 417000000 Kyrgyzstan 418000000 Laos 422000000 Lebanon 426000000 Lesotho 428000000 Latvia 430000000 Liberia 434000000 Libya 438000000 Liechtenstein 440000000 Lithuania 442000000 Luxembourg 446000000 Macau 450000000 Madagascar 454000000 Malawi 458000000 Malaysia 462000000 Maldives 466000000 Mali 470000000 Malta 474000000 Martinique 478000000 Mauritania 480000000 Mauritius 484000000 Mexico 492000000 Monaco 496000000 Mongolia 498000000 Moldava 504000000 Morocco 508000000 Mozambique 512000000 Oman 516000000 Namibia 520000000 Nauru 524000000 Nepal 528000000 Netherlands 530000000 **Netherlands Antilles** 533000000 Aruba 548000000 Vanuatu 554000000 New Zealand

558000000	Nicaragua
562000000	Niger
566000000	Nigeria
578000000	Norway
583000000	Micronesia
584000000	Marshall Islands
585000000	Palau
586000000	Pakistan
591000000	Panama
598000000	Papua New Guinea
600000000	Paraguay
604000000	Peru
608000000	Philippines
616000000	Poland
620000000	Portugal
624000000	Guinea-Bissau
626000000	East Timor
630000000	Puerto Rico
634000000	Qatar
638000000	Réunion
642000000	Romania
643000000	Russia
646000000	Rwanda
659000000	St. Kitts and Nevis
662000000	St. Lucia
663000000	Saint Martin
670000000	St. Vincent and the Grenadines
674000000	San Marino
678000000	São Tomé and Príncipe
682000000	Saudi Arabia
686000000	Senegal
690000000	Seychelles
694000000	Sierra Leone
702000000	Singapore
703000000	Slovakia
704000000	Vietnam
705000000	Slovenia
706000000	Somalia
710000000	South Africa
716000000	Zimbabwe
724000000	Spain
736000000	Sudan
740000000	Suriname
748000000	Swaziland
752000000	Sweden
756000000	Switzerland
760000000	Syria
762000000	Tajikistan
70400000	The all are al

764000000 Thailand

```
768000000
           Togo
776000000
           Tonga
780000000
           Trinidad and Tobago
784000000
           United Arab Emirates
788000000
           Tunisia
792000000
           Turkey
795000000
           Turkmenistan
798000000
           Tuvalu
800000000
           Uganda
804000000
           Ukraine
807000000
           Macedonia
810000000
           USSR
818000000
           Egypt
826100000
           Great Britain
826200000
           Northern Ireland
834000000
           Tanzania
840000000
           United States
854000000
           Burkina Faso
858000000
           Uruguay
860000000
           Uzbekistan
862000000
           Venezuela
882000000
           Samoa
887000000
890000000
           Yugoslavia, Socialist Federal Republic of
891000000
           Serbia and Montenegro
894000000
           Zambia
900000000
           West-Germany
901000000 East-Germany
911000000
           Serbia
912000000 Montenegro
915000000
           Kosovo
916000000 Austria-Hungary
917000000
           Alien's passport
```

Note:

Verbatim recorded answers standardized according to ISO 3166-1 alpha-2 code.

Data for original string variable is not available in this dataset.

v311 - mother born in [country] (Q95)

Q95

Was your mother born in [COUNTRY]?

<IF CODE 1, -1, -2: GO TO Q97>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v311, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	1	0.1	
1	yes		1355	89.7	89.8
2	no		154	10.2	10.2
	Sum		1510	100.0	100.0
	Valid Cases		1509		

v312b - mothers country of birth - ISO 3166-1 code/Geburtsland der Mutter ISO 3166-1 Code (Q96)

Q96

In which country was your mother born?

Write in: ...

<CODING INSTRUCTION: TO BE CODED INTO PRE-SPECIFIED ISO 3166-1 (2-CHARACTER)>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 4000000 Afghanistan
- 8000000 Albania
- 12000000 Algeria
- 20000000 Andorra
- 24000000 Angola
- 28000000 Antigua and Barbuda
- 31000000 Azerbaijan
- 32000000 Argentina
- 36000000 Australia
- 40000000 Austria
- 44000000 Bahamas
- 48000000 Bahrain
- 50000000 Bangladesh
- 51000000 Armenia
- 52000000 Barbados
- 56000000 Belgium
- 64000000 Bhutan
- 68000000 Bolivia
- 7000000 Bosnia-Herzegovina
- 72000000 Botswana
- 76000000 Brazil
- 84000000 Belize
- 90000000 Solomon Islands
- 96000000 Brunei
- 100000000 Bulgaria
- 104000000 Burma (Myanmar)
- 108000000 Burundi
- 112000000 Belarus
- 116000000 Cambodia
- 120000000 Cameroon
- 124000000 Canada
- 132000000 Cape Verde
- 140000000 Central African Republic
- 144000000 Sri Lanka
- 148000000 Chad
- 152000000 Chile
- 156000000 China

158000000	Taiwan
170000000	Colombia
174000000	Comoros
178000000	Congo, Rep. of
180000000	Congo, Dem. Rep. Of
188000000	Costa Rica
191000000	Croatia
192000000	Cuba
196100000	Cyprus
196200000	Northern Cyprus
200000000	Czechoslovakia
203000000	Czech Republic
204000000	Benin
208000000	Denmark
212000000	Dominica
214000000	Dominican Republic
218000000	Ecuador
222000000	El Salvador
226000000	Equatorial Guinea
231000000	Ethiopia
232000000	Eritrea
233000000	Estonia
242000000	Fiji
246000000	Finland
248000000	Ålands Islands
250000000	France
254000000	French Guiana
262000000	Djibouti
266000000	Gabon
268000000	Georgia
270000000	Gambia
275000000	Palestinian Territory, occupied
276000000	Germany
288000000	Ghana
292000000	Gibraltar
296000000	Kiribati
300000000	Greece
308000000	Grenada
312000000	Guadeloupe
320000000	Guatemala
324000000	Guinea
328000000	Guyana
332000000	Haiti
340000000	Honduras
344000000	Hong Kong
348000000	Hungary
352000000	Iceland
356000000	India

360000000 Indonesia

364000000 Iran

GESIS StudyNo 4754, v1.1.0, http://dx.doi.org/doi:10.4232/1.10152

364000000	ıran
368000000	Iraq
372000000	Ireland
376000000	Israel
380000000	Italy
384000000	Côte d'Ivoire
388000000	Jamaica
392000000	Japan
398000000	Kazakhstan
400000000	Jordan
404000000	Kenya
408000000	Korea, North
410000000	Korea, South
414000000	Kuwait
417000000	Kyrgyzstan
418000000	Laos
422000000	Lebanon
426000000	Lesotho
428000000	Latvia
430000000	Liberia
434000000	Libya
438000000	Liechtenstein
440000000	Lithuania
442000000	Luxembourg
446000000	Macau
450000000	Madagascar
454000000	Malawi
458000000	Malaysia
462000000	Maldives
466000000	Mali
470000000	Malta
474000000	Martinique
478000000	Mauritania
480000000	Mauritius
484000000	Mexico
492000000	Monaco
496000000	Mongolia
498000000	Moldava
504000000	Morocco
508000000	Mozambique
512000000	Oman
516000000	Namibia
520000000	Nauru
524000000	Nepal
528000000	Netherlands
530000000	Netherlands Antilles
533000000	Aruba
548000000	Vanuatu
554000000	New Zealand

558000000	Nicaragua
562000000	Niger
566000000	Nigeria
578000000	Norway
583000000	Micronesia
584000000	Marshall Islands
585000000	Palau
586000000	Pakistan
591000000	Panama
598000000	Papua New Guinea
600000000	Paraguay
604000000	Peru
608000000	Philippines
616000000	Poland
620000000	Portugal
624000000	Guinea-Bissau
626000000	East Timor
630000000	Puerto Rico
634000000	Qatar
638000000	Réunion
642000000	Romania
643000000	Russia
646000000	Rwanda
659000000	St. Kitts and Nevis
662000000	St. Lucia
663000000	Saint Martin
670000000	St. Vincent and the Grenadines
674000000	San Marino
678000000	São Tomé and Príncipe
682000000	Saudi Arabia
686000000	Senegal
690000000	Seychelles
694000000	Sierra Leone
702000000	Singapore
703000000	Slovakia
704000000	Vietnam
705000000	Slovenia
706000000	Somalia
710000000	South Africa
716000000	Zimbabwe
724000000	Spain
736000000	Sudan
740000000	Suriname
748000000	Swaziland
752000000	Sweden
756000000	Switzerland
760000000	Syria
762000000	Tajikistan
764000000	Thailand

```
768000000
           Togo
776000000
           Tonga
780000000
           Trinidad and Tobago
784000000
           United Arab Emirates
788000000
           Tunisia
792000000
           Turkey
795000000
           Turkmenistan
798000000
           Tuvalu
800000000
           Uganda
804000000
           Ukraine
807000000
           Macedonia
810000000
           USSR
818000000
           Egypt
826100000
           Great Britain
826200000
           Northern Ireland
834000000
           Tanzania
840000000
           United States
854000000
           Burkina Faso
858000000
           Uruguay
860000000
           Uzbekistan
862000000
           Venezuela
882000000
           Samoa
887000000
890000000
           Yugoslavia, Socialist Federal Republic of
891000000
           Serbia and Montenegro
894000000
           Zambia
900000000
           West-Germany
901000000 East-Germany
911000000
           Serbia
912000000 Montenegro
915000000
           Kosovo
916000000 Austria-Hungary
917000000
           Alien's passport
```

Note:

Verbatim recorded answers standardized according to ISO 3166-1 alpha-2 code.

Data for original string variable is not available in this dataset.

v313 - current legal marital status respondent (Q97)

 $\Omega97$

<WIDOWED AND DIVORCED FROM A MARRIAGE OR REGISTERED PARTNERSHIP ONLY, NOT FROM COHABITATION>

What is your current legal marital status?

<IF CODE 6, -1, -2: GO TO Q99>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 married
- 2 registered partnership
- 3 widowed
- 4 divorced
- 5 separated
- 6 never married and never registered partnership

Remarks on original question in the field questionnaire:

Deviation in interviewer instruction and response categories ("registered partnership" has been omitted, since this status did not exist in Austria in 2008).

Derivation:

Category 2 'registered partnership' was not implemented in Field Questionnaire.

v313, weighted by weight

Value	Label	Missing Count	Percent	Valid Percent
1	married	722	47.8	47.8
3	widowed	159	10.5	10.5
4	divorced	153	10.1	10.1
5	separated	17	1.1	1.1
6	never married or registered partnership	460	30.5	30.4
	Sum	1510	100.1	100.0
	Valid Cases	1510		

v314 - lived with partner before marriage (Q98)

Q98

Did you live together with your partner before your marriage or before the registration of your partnership?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

Remarks on original question in the field questionnaire:

Deviation in question text ("registered partnership" has been omitted, since this status did not exist in Austria in 2008).

v314, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	460	30.5	
-2	na	М	5	0.3	
-1	dk	М	5	0.3	
1	yes		604	40.0	58.0
2	no		437	28.9	42.0
	Sum		1510	100.1	100.0
	Valid Cases		1041		

v315 - living with partner (Q99)

099

<IF MARRIED OR REGISTERED PARTNERSHIP (Q97 IS 1 OR 2) GO TO Q101>

Do you live with a partner?

<IF CODE 1: GO TO Q101>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

Remarks on original question in the field questionnaire:

Deviation in interviewer instruction ("registered partnership" has been omitted, since this status did not exist in Austria in 2008).

v315, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	722	47.8	
1	yes		161	10.7	20.4
2	no		627	41.5	79.6
	Sum		1510	100.0	100.0
	Valid Cases		788		

v316 - having steady relationship (Q100)

Q100

Do you have a steady relationship?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v316, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	883	58.5	
-1	dk	М	3	0.2	
1	yes		101	6.7	16.2
2	no		523	34.6	83.8
	Sum		1510	100.0	100.0
	Valid Cases		624		

v317 - did you live with a partner before (Q101)

Q101

<FOR THOSE DIVORCED, SEPARATED, WIDOWED (Q97 IS 3, 4 OR 5) GO TO Q105 / FOR THOSE MARRIED OR IN A REGISTERED PARTNERSHIP (Q97 IS 1 OR 2) ASK:>

Before your current relationship, were you married or did you live with a partner?

<FOR THOSE NEVER MARRIED OR NEVER IN A REGISTERED PARTNERSHIP (Q97 IS 6) ASK:>

Did you live with a partner before?

<IF CODE 2, -1, -2, -3: GO TO Q105>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

Remarks on original question in the field questionnaire:

Deviation in interviewer instruction ("registered partnership" has been omitted, since this status did not exist in Austria in 2008).

v317, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-5	other missing	М	269	17.8	
-3	nap	М	328	21.7	
-2	na	М	2	0.1	
-1	dk	М	2	0.1	
1	yes		298	19.7	32.8
2	no		611	40.5	67.2
	Sum		1510	100.0	100.0
	Valid Cases		909		

v318 - married to this partner or in registered partnership (Q102)

Q102

<FOR THOSE NEVER MARRIED OR NEVER IN A REGISTERED PARTNERSHIP (Q97 IS 6) GO TO Q104 / ASK FROM THOSE WHO ARE MARRIED OR IN REGISTERED PARTNERSHIP Q97=1 OR 2>

Were you married to this partner or did you have a registered partnership with this partner?

<IF CODE 3, -1, -2, -3: GO TO Q104>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes, married
- 2 yes registered partnership
- 3 no

Remarks on original question in the field questionnaire:

Deviation in question text, response categories and interviewer instruction ("registered partnership" has been omitted, since this status did not exist in Austria in 2008).

Derivation:

Category 2 'registered partnership' was not implemented in Field Questionnaire.

v318, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	1292	85.6	
1	yes, married		64	4.2	29.4
3	no		154	10.2	70.6
	Sum		1510	100.0	100.0
	Valid Cases		218		

v319 - lived with this partner before marriage/registration of partnership (Q103)

0103

Did you live together with this partner before your marriage or before the registration of your partnership?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

Remarks on original question in the field questionnaire:

Deviation in question text ("registered partnership" has been omitted, since this status did not exist in Austria in 2008).

v319, weighted by weight

Value Label	Missing	Count	Percent	Valid Percent
-3 nap	М	1446	95.8	
1 yes		47	3.1	73.4
2 no		17	1.1	26.6
Sum		1510	100.0	100.0
Valid Cases		64		

v320 - end of relationship because of separation or partners death

Q104

Did this relationship end because you separated or because your partner died?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 separated
- 2 partner died

v320, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-5	other missing	М	503	33.3	
-3	nap	М	943	62.5	
1	separated		57	3.8	89.1
2	partner died		7	0.5	10.9
	Sum		1510	100.0	100.0
	Valid Cases		65		

v321 - how many children do you have (Q105)

Q105

<ASK ALL>

How many children do you have?

Write in: ...

<IF CODE 0: GO TO Q107>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 0 no children

v321, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
0	no children		540	35.8	35.8
1			262	17.4	17.4
2			410	27.2	27.2
3			190	12.6	12.6
4			67	4.4	4.4
5			21	1.4	1.4
6			7	0.5	0.5
7			7	0.5	0.5
8			3	0.2	0.2
9			1	0.1	0.1
	Sum		1510	99.9	100.0
	Valid Cases		1509		

v322 - year in which firstborn child was born (Q106)

Q106

Of all your children, in which year was your first child born? Please, include deceased children.

Write in year: ...

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk

v323 - living in household: partner, husband or wife (Q107)

Q107a

<SHOW CARD 107>

Who, apart from you, is living in this household? Please select from the card all those that are relevant, and tell me how many of each live in the household.

<DEFINITION OF LIVING IN HOUSEHOLD: SPENDING ON AVERAGE FOUR OR MORE NIGHTS PER WEEK IN THE SAME HOUSE.</p>
CHILDREN INCLUDE PARTNER'S, ADOPTION AND FOSTER CHILDREN.>

v323 partner, husband or wife

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v323, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	2	0.1	
1	yes		862	57.1	57.2
2	no		646	42.8	42.8
	Sum		1510	100.0	100.0
	Valid Cases		1508		

v324a - living in household: children (Q107)

Q107a

<SHOW CARD 107>

Who, apart from you, is living in this household? Please select from the card all those that are relevant, and tell me how many of each live in the household.

<DEFINITION OF LIVING IN HOUSEHOLD: SPENDING ON AVERAGE FOUR OR MORE NIGHTS PER WEEK IN THE SAME HOUSE.</p>
CHILDREN INCLUDE PARTNER'S, ADOPTION AND FOSTER CHILDREN.>

v324a children

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v324a, weighted by weight

Value	Label	Missing Count	Percent	Valid Percent
1	yes	512	33.9	33.9
2	no	998	66.1	66.1
	Sum	1510	100.0	100.0
	Valid Cases	1510		

v324b - living in household: number of children (Q107)

Q107b

<SHOW CARD 107>

Who, apart from you, is living in this household? Please select from the card all those that are relevant, and tell me how many of each live in the household.

IF YES, number: ...

v324b number of children

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk

v324b, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	998	66.1	
1			241	16.0	47.1
2			178	11.8	34.8
3			72	4.8	14.1
4			17	1.1	3.3
5			2	0.1	0.4
6			1	0.1	0.2
7			1	0.1	0.2
	Sum		1510	100.0	100.0
	Valid Cases		512		

v325a - living in household: parents (Q107)

Q107a

<SHOW CARD 107>

Who, apart from you, is living in this household? Please select from the card all those that are relevant, and tell me how many of each live in the household.

<DEFINITION OF LIVING IN HOUSEHOLD: SPENDING ON AVERAGE FOUR OR MORE NIGHTS PER WEEK IN THE SAME HOUSE.</p>
CHILDREN INCLUDE PARTNER'S, ADOPTION AND FOSTER CHILDREN.>

v325a parents

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v325a, weighted by weight

Value	Label	Missing Count	Percent	Valid Percent
1	yes	152	10.1	10.1
2	no	1358	89.9	89.9
	Sum	1510	100.0	100.0
	Valid Cases	1510		

v325b - living in household: number of parents (Q107)

Q107b

<SHOW CARD 107>

Who, apart from you, is living in this household? Please select from the card all those that are relevant, and tell me how many of each live in the household.

IF YES, number: ...

v325b number of parents

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk

v325b, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	1358	89.9	
1			52	3.4	34.2
2			100	6.6	65.8
	Sum		1510	100.0	100.0
	Valid Cases		152		

v326a - living in household: grandparents (Q107)

Q107a

<SHOW CARD 107>

Who, apart from you, is living in this household? Please select from the card all those that are relevant, and tell me how many of each live in the household.

<DEFINITION OF LIVING IN HOUSEHOLD: SPENDING ON AVERAGE FOUR OR MORE NIGHTS PER WEEK IN THE SAME HOUSE.</p>
CHILDREN INCLUDE PARTNER'S, ADOPTION AND FOSTER CHILDREN.>

v326a grandparents

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v326a, weighted by weight

Value	Label	Missing Count	Percent	Valid Percent
1	yes	8	0.5	0.5
2	no	1502	99.5	99.5
	Sum	1510	100.0	100.0
	Valid Cases	1510		

v326b - living in household: number of grandparents (Q107)

Q107b

<SHOW CARD 107>

Who, apart from you, is living in this household? Please select from the card all those that are relevant, and tell me how many of each live in the household.

IF YES, number: ...

v326b number of grandparents

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk

v326b, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	1502	99.5	
1			7	0.5	77.8
2			2	0.1	22.2
	Sum		1510	100.1	100.0
	Valid Cases		8		

v327a - living in household: relatives (Q107)

Q107a

<SHOW CARD 107>

Who, apart from you, is living in this household? Please select from the card all those that are relevant, and tell me how many of each live in the household.

<DEFINITION OF LIVING IN HOUSEHOLD: SPENDING ON AVERAGE FOUR OR MORE NIGHTS PER WEEK IN THE SAME HOUSE.</p>
CHILDREN INCLUDE PARTNER'S, ADOPTION AND FOSTER CHILDREN.>

v327a other relatives (brothers, sisters etc)

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v327a, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
1	yes		70	4.6	4.6
2	no		1440	95.4	95.4
	Sum		1510	100.0	100.0
	Valid Cases		1510		

v327b - living in household: number of relatives (Q107)

Q107b

<SHOW CARD 107>

Who, apart from you, is living in this household? Please select from the card all those that are relevant, and tell me how many of each live in the household.

IF YES, number: ...

v327b number of other relatives (brothers, sisters etc)

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk

v327b, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	1440	95.4	
1			48	3.2	68.6
2			15	1.0	21.4
3			5	0.3	7.1
5			2	0.1	2.9
	Sum		1510	100.0	100.0
	Valid Cases		70		

v328a - living in household: non relatives (Q107)

Q107a

<SHOW CARD 107>

Who, apart from you, is living in this household? Please select from the card all those that are relevant, and tell me how many of each live in the household.

<DEFINITION OF LIVING IN HOUSEHOLD: SPENDING ON AVERAGE FOUR OR MORE NIGHTS PER WEEK IN THE SAME HOUSE.</p>
CHILDREN INCLUDE PARTNER'S, ADOPTION AND FOSTER CHILDREN.>

v328a Other non relatives

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v328a, weighted by weight

Value	Label	Missing Count	Percent	Valid Percent
1	yes	29	1.9	1.9
2	no	1481	98.1	98.1
	Sum	1510	100.0	100.0
	Valid Cases	1510		

v328b - living in household: number of non relatives (Q107)

Q107b

<SHOW CARD 107>

Who, apart from you, is living in this household? Please select from the card all those that are relevant, and tell me how many of each live in the household.

IF YES, number: ...

v328b number of other non relatives

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk

v328b, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	1481	98.1	
-2	na	М	2	0.1	
1			22	1.5	84.6
2			3	0.2	11.5
3			1	0.1	3.8
	Sum		1510	99.9	100.0
	Valid Cases		26		

v329a - experienced: death of own children (Q108)

Q108a

Have you ever experienced any of the following events?

[If yes, at what age have you experienced this for the first time?]

v329a The death of one of your children?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v329a, weighted by weight

Value	Label	Missing Count	Percent	Valid Percent
1	yes	58	3.8	3.8
2	no	1452	96.2	96.2
	Sum	1510	100.0	100.0
	Valid Cases	1510		

v329b - age experienced: death of own children (Q108)

Q108b

[Have you ever experienced any of the following events?]

If yes, at what age have you experienced this for the first time?

IF YES, age: ...

v329b The death of one of your children? (age experienced)

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk

v330a - experienced: divorce of own children (Q108)

Q108a

Have you ever experienced any of the following events?

[If yes, at what age have you experienced this for the first time?]

v330a The divorce of one of your children?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v330a, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
1	yes		112	7.4	7.4
2	no		1397	92.5	92.6
	Sum		1510	100.0	100.0
	Valid Cases		1509		

v330b - age experienced: divorce of own children (Q108)

Q108b

[Have you ever experienced any of the following events?]

If yes, at what age have you experienced this for the first time?

IF YES, age: ...

v330b The divorce of one of your children? (age experienced)

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk

v331a - experienced: divorce of parents (Q108)

Q108a

Have you ever experienced any of the following events?

[If yes, at what age have you experienced this for the first time?]

v331a The divorce of your parents

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v331a, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	3	0.2	
1	yes		151	10.0	10.0
2	no		1355	89.7	90.0
	Sum		1510	100.0	100.0
	Valid Cases		1506		

v331b - age experienced: divorce of parents (Q108)

Q108b

[Have you ever experienced any of the following events?]

If yes, at what age have you experienced this for the first time?

IF YES, age: ...

v331b The divorce of your parents (age experienced)

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk

v332a - experienced: divorce of relative (Q108)

Q108a

Have you ever experienced any of the following events?

[If yes, at what age have you experienced this for the first time?]

v332a The divorce of another relative

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v332a, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	1	0.1	
-1	dk	М	17	1.1	
1	yes		349	23.1	23.4
2	no		1142	75.6	76.6
	Sum		1510	99.9	100.0
	Valid Cases		1492		

v332b - age experienced: divorce of relative (Q108)

Q108b

[Have you ever experienced any of the following events?]

If yes, at what age have you experienced this for the first time?

IF YES, age: ...

v332b The divorce of another relative (age experienced)

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk

v333a - experienced: death of father (Q108)

Q108a

Have you ever experienced any of the following events?

[If yes, at what age have you experienced this for the first time?]

v333a The death of your father

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v333a, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	2	0.1	
-1	dk	М	11	0.7	
1	yes		677	44.8	45.3
2	no		819	54.2	54.7
	Sum		1510	99.9	100.0
	Valid Cases		1497		

v333b - age experienced: death of father (Q108)

Q108b

[Have you ever experienced any of the following events?]

If yes, at what age have you experienced this for the first time?

IF YES, age: ...

v333b The death of your father (age experienced)

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk

v334a - experienced: death of mother (Q108)

Q108a

Have you ever experienced any of the following events?

[If yes, at what age have you experienced this for the first time?]

v334a The death of your mother

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v334a, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	3	0.2	
1	yes		524	34.7	34.9
2	no		979	64.8	65.1
	Sum		1510	100.0	100.0
	Valid Cases		1503		

v334b - age experienced: death of mother (Q108)

Q108b

[Have you ever experienced any of the following events?]

If yes, at what age have you experienced this for the first time?

IF YES, age: ...

v334b The death of your mother (age experienced)

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk

v335 - age completed education respondent (Q109)

Q109

At what age did you complete your full time education, either at school or at an institution of higher education? Please exclude apprenticeships.

<IF RESPONDENT IS STILL AT SCHOOL, ASK:>

At what age do you expect you will have completed your education?

Write in age: ...

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 0 no formal education

v336 - Educational level respondent: ISCED-code one digit (Q110)

EDUCATIONAL LEVEL OF RESPONDENT: ISCED (ONE-DIGIT CODE) - HARMONIZED VARIABLE

Q110

<SHOW CARD 110>

What is the highest level you have completed in your education?

<'COMPLETED' MEANS DIPLOMA/CERTIFICATE>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- $0\quad 0: \ Pre-primary \ education \ or \ none \ education$
- 1 1: Primary education or first stage of basic education
- 2 2: Lower secondary or second stage of basic education
- 3 3: (Upper) secondary education
- 4 4: Post-secondary non-tertiary education
- 5 5: First stage of tertiary education
- 6 6: Second stage of tertiary education

Note:

Harmonized variable: national educational categories have been recoded according to ISCED97. For country-specific variable, see V336_cs. For detailed correspondence with country-specific categories, see Method Report.

v336, weighted by weight

Value Label	Missing	Count	Percent	Valid Percent
-1 dk	М	1	0.1	
1 1: Primary education or first stage of basic education		43	2.8	2.8
2 2: Lower secondary or second stage of basic education		241	16.0	16.0
3 3: (Upper) secondary education		923	61.1	61.2
4 4: Post-secondary non-tertiary education		156	10.3	10.3
5 5: First stage of tertiary education		121	8.0	8.0
6 6: Second stage of tertiary education		25	1.7	1.7
Sum		1510	100.0	100.0
Valid Cases		1509		

v336_2 - Educational level respondent: ISCED-code two digits (Q110)

EDUCATIONAL LEVEL OF RESPONDENT: ISCED (TWO-DIGIT CODE) - HARMONIZED VARIABLE

Q110

<SHOW CARD 110>

What is the highest level you have completed in your education?

<'COMPLETED' MEANS DIPLOMA/CERTIFICATE>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 0 0: Pre-primary education or none education
- 10 1: Primary education or first stage of basic education
- 11 1: Primary education or first stage of basic education
- 21 2A: General lower secondary or second stage of basic education
- 22 2B: Pre-vocational or pre-technical lower secondary or second stage of basic education
- 23 2C: Vocational or technical lower secondary or second stage of basic education
- 31 3A: General (upper) secondary education
- 32 3B: Pre-vocational or pre-technical (upper) secondary education
- 33 3C: Vocational or technical (upper) secondary education
- 40 4: Post-secondary non-tertiary education
- 41 4A: Post-secondary non-tertiary education
- 42 4B: Post-secondary non-tertiary education
- 43 4C: Post-secondary non-tertiary education
- 51 5A: First stage of tertiary education (general)
- 52 5B: First stage of tertiary education (vocational)
- 60 6: Second stage of tertiary education

Note:

Harmonized variable: national educational categories have been recoded according to ISCED97. For country-specific variable, see V336_cs. For detailed correspondence with country-specific categories, see Method Report.

v336_2, weighted by weight

Value Label	Missing Count	Percent	Valid Percent
-1 dk	M 1	0.1	
10 1: Primary education or first stage of basic education	43	2.8	2.8
21 2A: General lower secondary or second stage of basic education	241	16.0	16.0
31 3A: General (upper) secondary education	239	15.8	15.8
32 3B: Pre-vocational or pre-technical (upper) secondary education	685	45.4	45.4
41 4A: Post-secondary non-tertiary education	129	8.5	8.5
42 4B: Post-secondary non-tertiary education	27	1.8	1.8
51 5A: First stage of tertiary education (general)	93	6.2	6.2
52 5B: First stage of tertiary education (vocational)	28	1.9	1.9
60 6: Second stage of tertiary education	25	1.7	1.7
Sum	1510	100.1	100.0
Valid Cases	1509		

v336_3 - Educational level respondent: ISCED-code three digits (Q110)

EDUCATIONAL LEVEL OF RESPONDENT: ISCED (THREE-DIGIT CODE) - HARMONIZED VARIABLE

Q110

<SHOW CARD 110>

What is the highest level you have completed in your education?

<'COMPLETED' MEANS DIPLOMA/CERTIFICATE>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 0 0: Pre-primary education or none education
- 100 1: Primary education or first stage of basic education
- 110 1: Primary education or first stage of basic education
- 210 2A: General lower secondary or second stage of basic education
- 220 2B: Pre-vocational or pre-technical lower secondary or second stage of basic education
- 230 2C: Vocational or technical lower secondary or second stage of basic education
- 310 3A: General (upper) secondary education
- 320 3B: Pre-vocational or pre-technical (upper) secondary education
- 330 3C: Vocational or technical (upper) secondary education
- 400 4: Post-secondary non-tertiary education
- 410 4A: Post-secondary non-tertiary education
- 420 4B: Post-secondary non-tertiary education
- 430 4C: Post-secondary non-tertiary education
- 510 5A: First stage of tertiary education (general)
- 511 5A (first degree): First stage of tertiary education (general)
- 512 5A (second degree): First stage of tertiary education (general)
- 513 5A (third degree): First stage of tertiary education (general)
- 520 5B: First stage of tertiary education (vocational)
- 600 6: Second stage of tertiary education
- 601 6 (first degree): Second stage of tertiary education
- 602 6 (second degree): Second stage of tertiary education

Note:

Harmonized variable: national educational categories have been recoded according to ISCED97. For country-specific variable, see V336_cs. For detailed correspondence with country-specific categories, see Method Report.

v336_3, weighted by weight

Value Label		Missing	Count	Percent	Valid Percent
-1 dk		M	1	0.1	
100 1 : Primary education or first	stage of basic education		43	2.8	2.8
210 2A: General lower secondary	y or second stage of basic education		241	16.0	16.0
310 3A: General (upper) seconda	ary education		239	15.8	15.8
320 3B : Pre-vocational or pre-ted	chnical (upper) secondary education		685	45.4	45.4
410 4A: Post-secondary non-tert	iary education		129	8.5	8.5
420 4B : Post-secondary non-tert	iary education		27	1.8	1.8
511 5A (first degree): First stage	of tertiary education (general)		17	1.1	1.1
512 5A (second degree): First sta	age of tertiary education (general)		75	5.0	5.0
520 5B : First stage of tertiary edu	ucation (vocational)		28	1.9	1.9
600 6: Second stage of tertiary e	ducation		25	1.7	1.7
Sum			1510	100.0	100.0
Valid Cases			1509		

v336_cs - Country specific: ISCED code education respondent (Q110)

Q110

<SHOW CARD 110>

What is the highest level you have completed in your education?

<'COMPLETED' MEANS DIPLOMA/CERTIFICATE>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 400101 1: AT Noch Schüler/in (ohne ersten allgemeinbildenden Schulabschluss)
- 400102 1: AT Schule beendet ohne ersten allgemeinbildenden Abschluss
- 400210 2A: AT Hauptschulabschluss, Volksschulabschluss 8. Klasse (auch bei Besuch einer mittleren/höheren Schule ohne ents
- 400310 3A: AT AHS oder Berufsbildende Höhere Schule
- 400321 3B: AT Berufsbildende mittlere Schule (z.B. Handelsschule) oder Polytechnikum (9. Klasse)
- 400322 3B: AT Berufsschule und Lehre
- 400410 4A: AT Matura oder Hochschulreife im zweiten Bildungsweg
- 400420 4B: AT Gesundheits- und Krankenpflegeschule, Schulen für medizinisch-technischen Fachdienst
- 400511 5A (First degree): AT Bachalaureat (Hochschule, Universität, Fachhochschule)
- 400512 5A (Second degree): AT Magisterium/Diplom (Hochschule, Universität, Fachhochschule)
- 400520 5B: AT Berufsakademie oder Kolleg (mit Diplomabschluss)
- 400600 6: AT Promotion od. Habilitation (Hochschule, Universität, Fachhochschule)

Note:

Standardized country-specific variable: combined variable containing country identifier (see COUNTRY) and country-specific educational categories. This variable has been harmonized according to ISCED97 into V336 (1-digit), V336_2 (2-digit), and v336_3 (3-digit).

v336_cs

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	1	0.1	
400101	1 : AT - Noch Schüler/in (ohne ersten allgemeinbildenden Schulabschluss)		6	0.4	0.4
400102	1 : AT - Schule beendet ohne ersten allgemeinbildenden Abschluss		37	2.5	2.5
400210	2A : AT - Hauptschulabschluss, Volksschulabschluss 8. Klasse (auch bei Besuch einer mittleren/höhe		241	16.0	16.0
400310	3A : AT - AHS oder Berufsbildende Höhere Schule		239	15.8	15.8
400321	3B : AT - Berufsbildende mittlere Schule (z.B. Handelsschule) oder Polytechnikum (9. Klasse)		126	8.3	8.4
400322	3B : AT - Berufsschule und Lehre		558	37.0	37.0
400410	4A : AT - Matura oder Hochschulreife im zweiten Bildungsweg		129	8.5	8.6
400420	4B : AT - Gesundheits- und Krankenpflegeschule, Schulen für medizinisch-technischen Fachdienst		27	1.8	1.8
400511	5A (First degree): AT - Bachalaureat (Hochschule, Universität, Fachhochschule)		17	1.1	1.1
400512	5A (Second degree): AT - Magisterium/Diplom (Hochschule, Universität, Fachhochschule)		75	5.0	5.0
400520	5B : AT - Berufsakademie oder Kolleg (mit Diplomabschluss)		28	1.9	1.9
400600	6 : AT - Promotion od. Habilitation (Hochschule, Universität, Fachhochschule)		25	1.7	1.7
	Sum		1510	99.9	100.0
	Valid Cases		1509		

v337 - paid employment/no paid employment (Q111)

Q111

<SHOW CARD 111>

Are you yourself gainfully employed at the moment or not? Please select from the card the employment status that applies to you.

<IF MORE THAN ONE JOB: ONLY FOR THE MAIN JOB>

<IF PAID EMPLOYMENT (CODE 1-3) GO TO Q112;</p>

IF NO PAID EMPLOYMENT (CODE 4-10) and -1, -2: GO TO Q111A>

Paid employment

- 1 30 hours a week or more
- 2 Less than 30 hours a week
- 3 Self employed

No paid employment

- 4 Military Service
- 5 Retired/pensioned
- 6 Housewife not otherwise employed
- 7 Student
- 8 Unemployed
- 9 Disabled <ONLY IF RESPONDENT DOES NOT WORK BECAUSE OF DISABILITY!>
- 10 Other, please specify (WRITE IN): ...
- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk

v337, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-1	dk	М	1	0.1	
1	30h a week or more		596	39.5	39.5
2	less then 30h a week		129	8.5	8.5
3	self employed		95	6.3	6.3
4	military service		3	0.2	0.2
5	retired/pensioned		401	26.6	26.6
6	housewife not otherwise employed		112	7.4	7.4
7	student		117	7.7	7.8
8	unemployed		33	2.2	2.2
9	disabled		7	0.5	0.5
10	other		16	1.1	1.1
	Sum		1510	100.0	100.0
	Valid Cases		1509		

v338 - employment/self-employment: last job (Q111A)

Q111A

In your LAST job were you employed (either full time or part time) or were you self-employed?

<IF CODE 3: GO TO Q115>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 employed
- 2 self-employed
- 3 never had a paid job

Derivation:

Response category 7 'never had a paid job' was recoded into 3 'never had a paid job' and response category -3 'nap' was added.

v338, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	820	54.3	
-2	na	М	1	0.1	
-1	dk	М	3	0.2	
1	employed		483	32.0	70.4
2	self-employed		72	4.8	10.5
3	never had apaid job		131	8.7	19.1
	Sum		1510	100.0	100.0
	Valid Cases		686		

v339ISCO - kind of job respondent - ISCO88 code (Q112)

Q112

<IF RESPONDENT HAS PAID EMPLOYMENT NOW, Q111 = 01-03, ASK:>

What is the name or title of your main job?

<IF RESPONDENT HAD PAID EMPLOYMENT BEFORE, Q111A =1,2 ASK:>

What was the name or title of your main job?

<IF RESPONDENT HAS/HAD MORE THAN ONE JOB. ASK HIS/HER OCCUPATION IN THE MAIN JOB!>

Write in: ...

Q112a

In your main job, what kind of work do/did you do most of the time?

Write in: ...

<CODING INSTRUCTION: CODE ACCORDING TO ISCO88 (4 DIGITS) AFTER INTERVIEW ON THE BASIS OF Q112 AND Q112A>

Code: ...

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 110 [armed forces]
- 1000 legislators, senior officials & managers
- 1100 legislators & senior officials
- 1110 legislators
- 1120 senior government officials
- 1130 traditional chiefs & heads of villages
- 1140 senior officials special-interest organisations
- 1141 senior officials political-party organisations
- 1142 senior officials economic-interest organisations
- 1143 senior officials special-interest organisations
- 1200 corporate managers [large enterprise]
- 1210 directors & chief executives [large enterprise]
- 1220 department managers [large enterprise][production-operation]
- 1221 department managers agriculture, hunting, forestry & fishIng
- 1222 department managers manufacturing
- 1223 department managers construction
- 1224 department managers wholesale & retail trade
- 1225 department managers restaurants & hotels
- 1226 department managers transport, storage & communications
- 1227 department managers business services
- 1228 department managers personal care, cleaning etc
- 1229 operations department managers nec
- 1230 other department managers [large enterprise]
- 1231 finance & administration department managers
- 1232 personnel & industrial relations department managers
- 1233 sales & marketing department managers
- 1234 advertising & public relations department managers

1235	supply & distribution department managers
1236	computing services department managers
1237	research & development department managers
1239	other department managers nec
1240	[office manager]
1250	[military officers]
1251	[high military officers]
1252	[lower military officers]
1300	general managers [small enterprise]
1310	general managers [small enterprise]
1311	general managers agriculture, hunting forestry & fishing
1312	general managers manufacturing
1313	general managers construction
1314	general managers wholesale & retail trade
1315	general managers restaurants & hotels
1316	general managers transport, storage & communications
1317	general managers business services
1318	general managers personal care, cleaning etc services
1319	general managers nec
2000	professionals
2100	physical, mathematical & engineering science professionals
2110	physicists, chemists & related professionals
2111	physicists & astronomers
2112	meteorologists
2113	chemists
2114	geologists & geophysicists
2120	mathematicians, statisticians etc professionals
2121	mathematicians etc professionals
2122	statisticians
2130	computing professionals
2131	computer systems designers & analysts
2132	computer programmers
2139	computing professionals nec
2140	architects, engineers etc professionals
2141	architects town & traffic planners
2142	civil engineers
2143	electrical engineers
2144	electronics & telecommunications engineers
2145	mechanical engineers
2146	chemical engineers
2147	mining engineers metallurgists etc professionals
2148	cartographers & surveyors
2149	architects engineers etc professionals nec
2200	life science & health professionals
2210	life science professionals
2211	biologists, botanists zoologists etc professionals
2212	pharmacologists, pathologists etc professionals

2213 agronomists etc professionals2220 health professionals (except nursing)

2221	medical doctors
2222	dentists
2223	veterinarians
2224	pharmacists
2229	health professionals except nursing nec
2230	nursing & midwifery professionals
2300	teaching professionals
2310	higher education teaching professionals
2320	secondary education teaching professionals
2321	[second academ teacher]
2322	[second voc teacher]
2323	[middle school teacher]
2330	primary & pre-primary education teaching professionals
2331	primary education teaching professionals
2332	pre-primary education teaching professionals
2340	special education teaching professionals
2350	other teaching professionals
2351	education methods specialists
2352	school inspectors
2359	other teaching professionals nec
2400	other professionals
2410	business professionals
2411	accountants
2412	personnel & careers professionals
2419	business professionals nec
2420	legal professionals
2421	lawyers
2422	judges
2429	legal professionals nec
2430	archivists, librarians etc information professionals
2431	archivists & curators
2432	librarians etc information professionals
2440	social science etc professionals
2441	economists
2442	sociologists anthropologists etc professionals
2443	philosophers, historians & political scientists
2444	philologists translators & interpreters
2445	psychologists
2446	social work professionals
2450	writers & creative or performing artists
2451	authors journalists & other writers
2452	sculptors, painters etc artists
2453	composers musicians & singers
2454	choreographers & dancers
2455	film, stage etc actors & directors
2460	religious professionals
3000	technicians and associated professionals
3100	physical & engineering science associate professionals

3110 physical & engineering science technicians

3111	chemical & physical science technicians
3112	civil engineering technicians
3113	electrical engineering technicians
3114	electronics & telecommunications engineering technicians
3115	mechanical engineering technicians
3116	chemical engineering technicians
3117	mining & metallurgical technicians
3118	draughtspersons
3119	physical & engineering science technicians nec
3120	computer associate professionals
3121	computer assistants
3122	computer equipment operators
3123	industrial robot controllers
3130	optical & electronic equipment operators
3131	photographers & elctr equipment operators
3132	broadcasting & telecommunications equipment operators
3133	medical equipment operators
3139	optical & electronic equipment operators nec
3140	ship & aircraft controllers & technicians
3141	ships engineers
3142	ships deck officers & pilots
3143	aircraft pilots etc associate professionals
3144	air traffic controllers
3145	air traffic safety technicians
3150	safety & ouality inspectors
3151	building & fire inspectors
3152	safety, health & quality inspectors
3200	life science & health associate professionals
3210	life science technicians etc associate professionals
3211	life science technicians
3212	agronomy & forestry technicians
3213	farming & forestry advisers
3220	modern health associate professionals except nursing
3221	medical assistants
3222	sanitarians
3223	dieticians & nutritionists
3224	optometrists & opticians
3225	dental assistants
3226	physiotherapists etc associate professionals
3227	veterinary assistants
3228	pharmaceutical assistants
3229	modern health associate professionals except nursing nec
3230	nursing & midwifery associate professionals
3231	nursing associate professionals
3232	midwifery associate professionals
3240	traditional medicine practitioners & faith healers
3241	traditional medicine practitioners
3242	faith healers

3300 teaching associate professionals

3310	primary education teaching associate professionals
3320	pre-primary education teaching associate professionals
3330	special education teaching associate professionals
3340	other teaching associate professionals
3400	other associate professionals
3410	finance & sales associate professionals
3411	securities & finance dealers & brokers
3412	insurance representatives
3413	estate agents
3414	travel consultants & organisers
3415	technical & commercial sales representatives
3416	buyers
3417	appraisers, valuers & auctioneers
3419	finance & sales associate professionals nec
3420	business services agents and trade brokers
3421	trade brokers
3422	clearing & forwarding agents
3423	employment agents & labour contractors
3429	business services agents & trade brokers nec
3430	administrative associate professionals
3431	administrative secretaries etc associate professionals
3432	legal etc business associate professionals
3433	bookkeepers
3434	statistical, mathematical etc associate professionals
3439	administrative associate professionals nec
3440	customs, tax etc government associate professionals
3441	customs & border inspectors
3442	government tax & excise officials
3443	government social benefits officials
3444	government licensing officials
3449	customs tax etc government associate professionals nec
3450	[police and army officers]
3451	[police inspectors-detectives]
3452	[armed forces non-commissioned officers + army nfs]
3460	social work associate professionals
3470	artistic, entertainment & sports associate professionals
3471	decorators & commercial designers
3472	radio, television & other announcers
3473	street night-club etc musicians singers & dancers
3474	clowns magicians acrobats etc associate professionals
3475	athletes sportspersons etc associate professionals
3480	religious associate professionals
4000	clerks
4100	office clerks
4110	secretaries & keyboard-operating clerks
4111	stenographers & typists
4112	word-processor etc operators
4113	data entry operators

4114 calculating-machine operators

4115 secretaries 4120 numerical clerks 4121 accounting & bookkeeping clerks 4122 statistical & finance clerks 4130 material-recording & transport clerks 4131 stock clerks 4132 production clerks 4133 transport clerks 4140 library, mail etc clerks 4141 library & filing clerks 4142 mail carriers & sorting clerks 4143 coding proof-reading etc clerks 4144 scribes etc workers 4190 other office clerks 4200 customer services clerks 4210 cashiers, tellers etc clerks 4211 cashiers & ticket clerks 4212 tellers & other counter clerks 4213 bookmakers & croupiers 4214 pawnbrokers & money-lenders 4215 debt-collectors etc workers 4220 client information clerks 4221 travel agency etc clerks 4222 receptionists & information clerks 4223 telephone switchboard operators 5000 service workers & shop & market sales workers 5100 personal & protective services workers 5110 travel attendants etc 5111 travel attendants & travel stewards 5112 transport conductors 5113 travel, museum guides 5120 housekeeping & restaurant services workers 5121 housekeepers etc workers 5122 cooks 5123 waiters waitresses & bartenders 5130 personal care etc work 5131 child-care workers 5132 institution-based personal care workers 5133 home based personal care workers 5139 personal care etc workers nec 5140 other personal services workers 5141 hairdressers barbers beauticians etc workers 5142 companions & valets 5143 undertakers & embalmers 5149 other personal services workers nec

5150 astrologers, fortune-tellers etc workers

5152 fortune-tellers, palmists etc workers

5151 astrologers etc workers

5160 protective services workers

page 477

5161	fire-fighters
5162	police officers
5163	prison guards
5164	[army: enlisted men - soldiers]
5169	protective services workers nec
5200	models, salespersons & demonstrators
5210	fashion & other models
5220	shop salespersons & demonstrators
5230	stall & market salespersons
6000	skilled agricultural & fishery workers
6100	market-oriented skilled agricultural & fishery workers
6110	market gardeners & cropgrowers
6111	field crop & vegetable growers
6112	tree & shrub crop growers
6113	gardeners, horticultural & nursery growers
6114	mixed-crop growers
6120	market-oriented animal producers etc workers
6121	dairy & livestock producers
6122	poultry producers
6123	apiarists & sericulturists
6124	mixed -animal producers
6129	market-oriented animal producers etc workers nec
6130	market-oriented crop & animal producers
6131	[mixed farmer]
6132	[farm foreman]
6133	[farmer nfs]
6134	[skilled farm worker nfs]
6140	forestry etc workers
6141	forestry workers & loggers
6142	charcoal burners etc workers
6150	fishery workers, hunters & trappers
6151	aquatic-life cultivation workers
6152	inland & coastal waters fishery workers
6153	deep-sea fishery workers
6154	hunters & trappers
6200	subsistence agricultural & fishery workers
6210	subsistence agricultural & fishery workers
7000	craft etc trades workers
7100	extraction & building trades workers
7110	miners, shotfirers, stone cutters & carvers
7111	miners & quarry workers
7112	shotfirers & blasters
7113	stone splitters cutters & carvers
7120	[builders] & ar trades workers
7121	builders traditional materials
7122	bricklayers & stonemasons
7123	concrete placers concrete finishers etc workers

7124 carpenters & joiners

7129 [builders] ar trades workers nec

- 7130 building finishers etc trades workers
- 7131 roofers
- 7132 floor layers & tile setters
- 7133 plasterers
- 7134 insulation workers
- 7135 glaziers
- 7136 plumbers & pipe fitters
- 7137 building etc electricians
- 7140 painters, building structure cleaners etc trades workers
- 7141 painters etc workers
- 7142 varnishers etc painters
- 7143 building structure cleaners
- 7200 metal, machinery etc trades workers
- 7210 metal moulders, welders, sheetmetal workers structural metal
- 7211 metal moulders & coremakers
- 7212 welders & flamecutters
- 7213 sheet-metal workers
- 7214 structural-metal preparers & erectors
- 7215 riggers & cable splicers
- 7216 underwater workers
- 7220 blacksmiths, tool-makers etc trades workers
- 7221 blacksmiths hammer-smiths & forgingpress workers
- 7222 tool-makers etc workers
- 7223 machine-tool setters & setter-operators
- 7224 metal wheel-grinders polishers & tool sharpeners
- 7230 machinery mechanics & fitters
- 7231 motor vehicle mechanics & fitters
- 7232 aircraft engine mechanics & fitters
- 7233 agricultural- or industrial-machinery mechanics & fitters
- 7234 [unskl garage worker, oiler]
- 7240 electrical & electronic equipment mechanics & fitters
- 7241 electrical mechanics & fitters
- 7242 electronics fitters
- 7243 electronics mechanics & servicers
- 7244 telegraph & telephone installers & servicers
- 7245 electrical line installers repairers & cable jointers
- 7300 precision, handicraft, printing etc trades workers
- 7310 precision workers in metal etc materials
- 7311 precision-instrument makers & repairers
- 7312 musical-instrument makers & tuners
- 7313 jewellery & precious-metal workers
- 7320 potters, glass-makers etc trades workers
- 7321 abrasive wheel formers potters etc workers
- 7322 glass-makers cutters grinders & finishers
- 7323 glass engravers & etchers
- 7324 glass ceramics etc decorative painters
- 7330 handicraft workers in wood, textile, leather etc
- 7331 handicraft workers in wood etc materials
- 7332 handicraft workers in textile leather etc materials

- 7340 printing etc trades workers
- 7341 compositors typesetters etc workers
- 7342 stereotypers & electrotypers
- 7343 printing engravers & etchers
- 7344 photographic etc workers
- 7345 bookbinders etc workers
- 7346 silk-screen, block & textile printers
- 7400 other craft etc trades workers
- 7410 food processing etc trades workers
- 7411 butchers fishmongers etc food preparers
- 7412 bakers-pastry cooks & confectionery makers
- 7413 dairy-products makers
- 7414 fruit, vegetable etc preservers
- 7415 food & beverage tasters & graders
- 7416 tobacco preparers & tobacco products makers
- 7420 wood treaters, cabinet-makers etc trades workers
- 7421 wood treaters
- 7422 cabinet-makers etc workers
- 7423 woodworking-machine setters & setteroperators
- 7424 basketry weavers brush makers etc workers
- 7430 textile, garment etc trades workers
- 7431 fibre preparers
- 7432 weavers knitters etc workers
- 7433 tailors dressmakers & hatters
- 7434 furriers etc workers
- 7435 textile leather etc pattern-makers & cutters
- 7436 sewers embroiderers etc workers
- 7437 upholsterers etc workers
- 7440 pelt, leather & shoemaking trades workers
- 7441 pelt dressers tanners & fellmongers
- 7442 shoe-makers etc workers
- 7500 [skilled manual worker nfs]
- 7510 [non farm foremen nfs]
- 7520 [skilled manual nfs]
- 7530 [apprentice skilled work nfs]
- 8000 plant & machine operators & assemblers
- 8100 stationary-plant etc operators
- 8110 mining- & mineral-processing plant operators
- 8111 mining-plant operators
- 8112 mineral-ore- & stone-processing-plant operators
- 8113 well drillers & borers etc workers
- 8120 metal-processing-plant operators
- 8121 ore & metal furnace operators
- 8122 metal melters casters & rolling-mill operators
- 8123 metal-heat-treating-plant operators
- 8124 metal drawers & extruders
- 8130 glass, ceramics etc plant operators
- 8131 glass & ceramics kiln etc machine operators
- 8139 glass, ceramics etc plant operators nec

8140	wood-processing- & papermaking-plant operators
8141	wood-processing-plant operators
8142	paper-pulp plant operators
8143	papermaking-plant operators
8150	chemical-processing-plant operators
8151	crushing- grinding- & chemical-mixing machinery operators
8152	chemical-heat-treating-plant operators
8153	chemical-filtering- & separating-equipment operators
8154	chemical-still & reactor operators
8155	petroleum- & natural-gas-refining-plant operators
8159	chemical-processing-plant operators nec
8160	power-production etc plant operators
8161	power-production plant operators
8162	steam-engine & boiler operators
8163	incinerator water-treatment etc plant operators
8170	automated-assembly-line & industrial-robot operators
8171	automated-assembly- line operators
8172	industrial-robot operators
8200	machine operators & assemblers
8210	metal- & mineral-products machine operators
8211	machine-tool operators
8212	cement & other mineral products machine operators
8220	chemical-products machine operators
8221	pharmaceutical- & toiletry-products machine operators
8222	ammunition- & explosive-products machine operators
8223	metal finishing- plating- & coating-machine operators
8224	photographic-products machine operators
8229	chemical-products machine operators nec
8230	rubber- & plastic-products machine operators
8231	rubber-products machine operators
8232	plastic-products machine operators
8240	wood-products machine operators
8250	printing- binding- & paper-products machine operators
8251	printing-machine operators
8252	bookbinding-machine operators
8253	paper-products machine operators
8260	textile- fur- & leather-products machine operators
8261	fibre-preparing-, spinning- & windingmachine operators
8262	weaving- & knitting-machine operators
8263	sewing-machine operators
8264	bleaching- dyeing- & cleaning-machine operators
8265	fur- & leather-preparing-machine operators
8266	shoemaking- etc machine operators
8269	textile- fur- & leather-products machine operators nec
8270	food etc products machine operators
8271	meat- & fish-processing-machine operators
8272	dairy-products machine operators
8273	grain- & spice-milling-machine operators

8274 baked-goods cereal & chocolate-products machine operators

8275	fruit- vegetable- & nut-processing-machine operators
8276	sugar production machine operators
8277	tea- coffee- & cocoa-processing-machine operators
8278	brewers- wine & other beverage machine operators
8279	tobacco production machine operators
8280	assemblers
8281	mechanical-machinery assemblers
8282	electrical-equipment assemblers
8283	electronic-equipment assemblers
8284	• •
8285	
8286	paperboard textile etc products assemblers
8290	other machine operators & assemblers
8300	drivers & mobile-plant operators
8310	locomotive-engine drivers etc workers
8311	locomotive-engine drivers
8312	railway brakers signallers & shunters
8320	motor-vehicle drivers
8321	motor-cycle drivers
8322	car taxi & van drivers
8323	bus & tram drivers
8324	heavy truck & lorry drivers
8330	agricultural & other mobile-plant operators
8331	motorised farm & forestry plant operators
8332	earth-moving- etc plant operators
8333	crane hoist etc plant operators
8334	lifting-truck operators
8340	ships deck crews etc workers
8400	[semiskld worker nfs]
9000	elementary occupations
9100	sales & services elementary occupations
9110	street vendors etc workers
9111	street food vendors
9112	street vendors non-food products
9113	door-to-door & telephone salespersons
9120	street services elementary occupations
9130	domestic etc helpers cleaners & launderers
9131	domestic helpers & cleaners
9132	helpers & cleaners in establishments
9133	hand-launderers & pressers
9140	building caretakers, window etc cleaners
9141	building caretakers
9142	vehicle, window etc cleaners
9150	messengers, porters, doorkeepers etc workers
9151	messengers package & luggage porters & deliverers
9152	doorkeepers watchpersons etc workers
9153	vending-machine money collectors meter readers etc workers

9160 garbage collectors etc labourers

9161 garbage collectors

- 9162 sweepers etc labourers
- 9200 agricultural, fishery etc labourers
- 9210 agricultural, fishery etc labourers
- 9211 farm-hands & labourers
- 9212 forestry labourers
- 9213 fishery hunting & trapping labourers
- 9300 labourers in mining, construction, manufacturing & transport
- 9310 mining & construction labourers
- 9311 mining & quarrying labourers
- 9312 construction & maintenance labourers: roads dams etc
- 9313 building construction labourers
- 9320 manufacturing labourers
- 9321 assembling labourers
- 9322 hand packers & other manufacturing labourers
- 9330 transport labourers & freight handlers
- 9331 hand or pedal vehicle drivers
- 9332 drivers of animal-drawn vehicles & machinery
- 9333 freight handlers

Note:

Standardized variable on the basis of two open-ended questions on occupation and kind of activity (Q112 and Q112a) post-coded according to 4-digit ISCO88.

Data for original string variable is not available in this dataset.

v339SIOPS - occupational status respondent - SIOPS (Q112)

COMPUTED VARIABLE - SIOPS: Standard International Occupational Prestige Scale (Treiman, 1977)

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk

Note:

Source classification: Ganzeboom, H. B. G. & Treiman, D. J. (1996), Internationally Comparable Measures of Occupational Status for the 1988 International Standard Classification of Occupations. Social Science Research, 25, 201-239.

Source syntax to compute classification: http://home.fsw.vu.nl/hbg.ganzeboom/pisa/index.htm

v339ISEI - occupational status respondent - ISEI (Q112)

COMPUTED VARIABLE - ISEI: International Socio-Economic Index of Occupational Status (Ganzeboom et al., 1992)

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk

Note:

Source classification: Ganzeboom, H. B. G. & Treiman, D. J. (1996), Internationally Comparable Measures of Occupational Status for the 1988 International Standard Classification of Occupations. Social Science Research, 25, 201-239.

Source syntax to compute classification: http://home.fsw.vu.nl/hbg.ganzeboom/pisa/index.htm

v339ESeC - occupational status respondent - European ESeC (Q112)

COMPUTED VARIABLE - European Socio-economic Classification (ESeC)

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 Large employers, higher managers/professionals
- 2 Lower managers/professionals, higher supervisory/technicians
- 3 Intermediate occupations
- 4 Small employers and self-employed (non-agriculture)
- 5 Small employers and self-employed (agriculture)
- 6 Lower supervisors and technicians
- 7 Lower sales and service
- 8 Lower technical
- 9 Routine

Note:

Source: http://www.iser.essex.ac.uk/research/esec/matrices-and-syntax

Date: 27-09-2006 (last update syntax).

v339ESeC, weighted by weight

Value Label	Missing	Count	Percent	Valid Percent
-5 other missing	M	2	0.1	
-3 nap	M	131	8.7	
-2 na	М	20	1.3	
1 Large employers, higher mgrs/professionals		91	6.0	6.7
2 Lower mgrs/professionals, higher supervisory/technicians		290	19.2	21.4
3 Intermediate occupations		190	12.6	14.0
4 Small employers and self-employed (non-agriculture)		85	5.6	6.3
5 Small employers and self-employed (agriculture)		46	3.0	3.4
6 Lower supervisors and technicians		94	6.2	6.9
7 Lower sales and service		202	13.4	14.9
8 Lower technical		138	9.1	10.2
9 Routine		222	14.7	16.3
Sum		1510	100.1	100.0
Valid Cases		1357		

v339egp - occupational status respondent - egp11 (Q112)

COMPUTED VARIABLE - EGP: EGP class scheme (Erikson/Goldthorpe/Portocarero, 1979)

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 I: Higher Controllers
- 2 II: Lower Controllers
- 3 IIIa: Routine Non-manual
- 4 IIIb: Lower Sales-Service
- 5 IVa: Self-employed with employees
- 6 IVb: Self-employed with no employees
- 7 V: Manual Supervisors
- 8 VI: Skilled Worker
- 9 VIIa: Unskilled Worker
- 10 VIIb: Farm Labor
- 11 IVc: Self-employed Farmer

Note:

Source classification: Ganzeboom, H. B. G. & Treiman, D. J. (1996), Internationally Comparable Measures of Occupational Status for the 1988 International Standard Classification of Occupations. Social Science Research, 25, 201-239.

Source syntax to compute classification: http://home.fsw.vu.nl/hbg.ganzeboom/pisa/index.htm

v339egp, weighted by weight

	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,				
Value	Label	Missing	Count	Percent	Valid Percent
-5	other missing	М	2	0.1	
-3	nap	М	131	8.7	
-2	na	М	20	1.3	
1	I :Higher Controllers		162	10.7	11.9
2	II :Lower Controllers		265	17.5	19.5
3	Illa:Routine Nonmanual		164	10.9	12.1
4	IIIb:Lower Sales-Service		219	14.5	16.1
5	IVa:Selfempl with empl		47	3.1	3.5
6	IVb:Selfempl no empl		34	2.3	2.5
7	V :Manual Supervisors		46	3.0	3.4
8	VI :Skilled Worker		139	9.2	10.2
9	VIIa:Unskilled Worker		214	14.2	15.7
10	VIIb:Farm Labor		21	1.4	1.5
11	IVc:Selfempl Farmer		48	3.2	3.5
	Sum		1510	100.1	100.0
	Valid Cases		1357		

v340 - respondent had/has how many employees (Q113)

Q113

<ASK FROM CURRENLTLY SELF-EMPLOYED (Q111=3) OR SELF-EMPLOYED IN THEIR LAST JOB (Q111A=2)>

How many employees do/did you have?

<SKIP TO Q115>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 none
- 2 1-9
- 3 10-24
- 4 25 or more

v340, weighted by weight

	•	•	•			
Value	Label		Missing	Count	Percent	Valid Percent
-3	nap		М	1343	88.9	
-1	dk		М	1	0.1	
1	none			85	5.6	51.2
2	1-9			67	4.4	40.4
3	10-24			8	0.5	4.8
4	25 or mo	re		6	0.4	3.6
	Sum			1510	100.0	100.0
	Valid Cas	ses		166		

v341 - do you supervise someone (Q114)

Q114

<ASK FROM CURRENTLY EMPLOYED (Q111=1 OR 2) OR PEOPLE NOT SELF-EMPLOYED IN THEIR LAST JOB (Q111A=1)> Do/did you have any responsibility for supervising the work of other employees?

<IF CODE 2, -1, -2: GO TO Q115>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v341, weighted by weight

- ,	- 3 , -	5			
Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	302	20.0	
-2	na	М	4	0.3	
-1	dk	М	3	0.2	
1	yes		344	22.8	28.6
2	no		857	56.8	71.4
	Sum		1510	100.0	100.0
	Valid Cases		1201		

v341a - how many people do you supervise (Q114a)

Q114a

How many other employees do/did you supervise?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 1-9
- 2 10-24
- 3 25 or more

v341a, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	1166	77.2	
-1	dk	М	1	0.1	
1	1-9		219	14.5	63.8
2	10-24		61	4.0	17.8
3	25 or more		63	4.2	18.4
	Sum		1510	100.0	100.0
	Valid Cases		342		

v342 - partner/spouse born in [country] (Q115)

Q115

<FOR RESPONDENTS WHO LIVE WITH A SPOUSE OR PARTNER (Q97=1 OR 2 OR Q99=1); OTHERWISE GO TO Q121> Was your partner/spouse born in ... [COUNTRY]?

<IF CODE 1, -1, -2: GO TO Q116>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v342, weighted by weight

- ,	- 3 ,	5			
Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	627	41.5	
-2	na	М	2	0.1	
-1	dk	М	1	0.1	
1	yes		803	53.2	91.1
2	no		78	5.2	8.9
	Sum		1510	100.1	100.0
	Valid Cases		881		

v343b - spouse/partners country of birth - ISO 3166-1 code/Geburtsland des (Ehe-)Partners/der (Ehe-)Partnerin ISO 3166-1 Code (Q115a) Q115a In which country was your spouse/partner born? <ASK AND WRITE IN FULL DETAIL> Write in: <CODING INSTRUCTION: TO BE CODED INTO PRE-SPECIFIED ISO 3166-1 (2-CHARACTER)> Code:

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 4000000 Afghanistan
- 8000000 Albania
- 12000000 Algeria
- 20000000 Andorra
- 24000000 Angola
- 28000000 Antigua and Barbuda
- 31000000 Azerbaijan
- 32000000 Argentina
- 36000000 Australia
- 40000000 Austria
- 44000000 Bahamas
- 48000000 Bahrain
- 50000000 Bangladesh
- 51000000 Armenia
- 52000000 Barbados
- 56000000 Belgium
- 64000000 Bhutan
- 68000000 Bolivia
- 7000000 Bosnia-Herzegovina
- 72000000 Botswana
- 76000000 Brazil
- 84000000 Belize
- 90000000 Solomon Islands
- 96000000 Brunei
- 100000000 Bulgaria
- 104000000 Burma (Myanmar)
- 108000000 Burundi
- 112000000 Belarus
- 116000000 Cambodia
- 120000000 Cameroon
- 124000000 Canada
- 132000000 Cape Verde
- 140000000 Central African Republic
- 144000000 Sri Lanka
- 148000000 Chad

152000000	Chile
156000000	China
158000000	Taiwan
170000000	Colombia
174000000	Comoros
178000000	Congo, Rep. of
180000000	Congo, Dem. Rep. Of
188000000	Costa Rica
191000000	Croatia
192000000	Cuba
196100000	Cyprus
196200000	Northern Cyprus
200000000	Czechoslovakia
203000000	Czech Republic
204000000	Benin
208000000	Denmark
212000000	Dominica
214000000	Dominican Republic
218000000	Ecuador
222000000	El Salvador
226000000	Equatorial Guinea
231000000	Ethiopia
232000000	Eritrea
233000000	Estonia
242000000	Fiji
246000000	Finland
248000000	Ålands Islands
250000000	France
254000000	French Guiana
262000000	Djibouti
266000000	Gabon
268000000	Georgia
270000000	Gambia
275000000	Palestinian Territory, occupied
276000000	Germany
288000000	Ghana
292000000	Gibraltar
296000000	Kiribati
300000000	Greece
308000000	Grenada
312000000	Guadeloupe
320000000	Guatemala
324000000	Guinea
328000000	Guyana
332000000	Haiti
340000000	Honduras
344000000	Hong Kong
348000000	Hungary
05000000	Landard Communication of the C

352000000 Iceland

356000000	India
360000000	Indonesia
364000000	Iran
368000000	Iraq
372000000	Ireland
376000000	Israel
380000000	Italy
384000000	Côte d'Ivoire
388000000	Jamaica
392000000	Japan
398000000	Kazakhstan
400000000	Jordan
404000000	Kenya
408000000	Korea, North
410000000	Korea, South
414000000	Kuwait
417000000	Kyrgyzstan
418000000	Laos
422000000	Lebanon
426000000	Lesotho
428000000	Latvia
430000000	Liberia
434000000	Libya
438000000	Liechtenstein
440000000	Lithuania
442000000	Luxembourg
446000000	Macau
450000000	Madagascar
454000000	Malawi
458000000	Malaysia
462000000	Maldives
466000000	Mali
470000000	Malta
474000000	Martinique
478000000	Mauritania
480000000	Mauritius
484000000	Mexico
492000000	Monaco
496000000	Mongolia
498000000	Moldava
504000000	Morocco
508000000	Mozambique
512000000	Oman
516000000	Namibia
520000000	Nauru
524000000	Nepal
528000000	Netherlands
530000000	Netherlands Antilles
533000000	Aruba

548000000	Vanuatu
554000000	New Zealand
558000000	Nicaragua
562000000	Niger
566000000	Nigeria
578000000	Norway
583000000	Micronesia
584000000	Marshall Islands
585000000	Palau
586000000	Pakistan
591000000	Panama
598000000	Papua New Guinea
600000000	Paraguay
604000000	Peru
608000000	Philippines
616000000	Poland
620000000	Portugal
624000000	Guinea-Bissau
626000000	East Timor
630000000	Puerto Rico
634000000	Qatar
638000000	Réunion
642000000	Romania
643000000	Russia
646000000	Rwanda
659000000	St. Kitts and Nevis
662000000	St. Lucia
663000000	Saint Martin
670000000	St. Vincent and the Grenadines
674000000	San Marino
678000000	São Tomé and Príncipe
682000000	Saudi Arabia
686000000	Senegal
690000000	Seychelles
694000000	Sierra Leone
702000000	Singapore
703000000	Slovakia
704000000	Vietnam
705000000	Slovenia
706000000	Somalia
710000000	South Africa
716000000	Zimbabwe
724000000	Spain
736000000	Sudan
740000000	Suriname
748000000	Swaziland
752000000	Sweden
756000000	Switzerland

760000000 Syria

762000000 Tajikistan 764000000 Thailand 768000000 Togo 776000000 Tonga 780000000 Trinidad and Tobago 784000000 **United Arab Emirates** 788000000 Tunisia 792000000 Turkey 795000000 Turkmenistan 798000000 Tuvalu 800000000 Uganda 804000000 Ukraine 807000000 Macedonia USSR 810000000 818000000 Egypt 826100000 **Great Britain** 826200000 Northern Ireland 834000000 Tanzania 840000000 **United States** 854000000 Burkina Faso 858000000 Uruguay 860000000 Uzbekistan 862000000 Venezuela 882000000 Samoa 887000000 Yemen 890000000 Yugoslavia, Socialist Federal Republic of 891000000 Serbia and Montenegro 894000000 Zambia 900000000 West-Germany 901000000 East-Germany 911000000 Serbia 912000000 Montenegro 915000000 Kosovo 916000000 Austria-Hungary 917000000 Alien's passport

Note:

 $\label{thm:condensate} Verbatim\ recorded\ answers\ standardized\ according\ to\ ISO\ 3166-1\ alpha-2\ code.$

Data for original string variable is not available in this dataset.

v344 - Educational level partner: ISCED-code one digit (Q116)

EDUCATIONAL LEVEL OF PARTNER: ISCED (ONE-DIGIT CODE) - HARMONIZED VARIABLE

Q116

<SHOW CARD 116>

What is the highest level your spouse or partner has completed in his or her education?

<"COMPLETED" MEANS DIPLOMA/CERTIFICATE>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- $0\quad 0: \ Pre-primary \ education \ or \ none \ education$
- 1 1: Primary education or first stage of basic education
- 2 2: Lower secondary or second stage of basic education
- 3 3: (Upper) secondary education
- 4 4: Post-secondary non-tertiary education
- 5 5: First stage of tertiary education
- 6 6: Second stage of tertiary education

Note:

Harmonized variable: national educational categories have been recoded according to ISCED97. For country-specific variable see V344_cs. For detailed correspondence with country-specific categories see Method Report.

v344, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	627	41.5	
-2	na	М	5	0.3	
-1	dk	М	1	0.1	
1	1 : Primary education or first stage of basic education		25	1.7	2.9
2	2: Lower secondary or second stage of basic education		127	8.4	14.5
3	3: (Upper) secondary education		563	37.3	64.2
4	4 : Post-secondary non-tertiary education		81	5.4	9.2
5	5 : First stage of tertiary education		67	4.4	7.6
6	6 : Second stage of tertiary education		14	0.9	1.6
	Sum		1510	100.0	100.0
	Valid Cases		877		

v344_2 - Educational level partner: ISCED-code two digits (Q116)

EDUCATIONAL LEVEL OF PARTNER: ISCED (TWO-DIGIT CODE) - HARMONIZED VARIABLE

Q116

<SHOW CARD 116>

What is the highest level your spouse or partner has completed in his or her education?

<"COMPLETED" MEANS DIPLOMA/CERTIFICATE>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 0 0: Pre-primary education or none education
- 10 1: Primary education or first stage of basic education
- 11 1: Primary education or first stage of basic education
- 21 2A: General lower secondary or second stage of basic education
- 22 2B: Pre-vocational or pre-technical lower secondary or second stage of basic education
- 23 2C: Vocational or technical lower secondary or second stage of basic education
- 31 3A: General (upper) secondary education
- 32 3B: Pre-vocational or pre-technical (upper) secondary education
- 33 3C: Vocational or technical (upper) secondary education
- 40 4: Post-secondary non-tertiary education
- 41 4A: Post-secondary non-tertiary education
- 42 4B: Post-secondary non-tertiary education
- 43 4C: Post-secondary non-tertiary education
- 51 5A: First stage of tertiary education (general)
- 52 5B: First stage of tertiary education (vocational)
- 60 6: Second stage of tertiary education

Note:

Harmonized variable: national educational categories have been recoded according to ISCED97. For country-specific variable, see V336_cs. For detailed correspondence with country-specific categories, see Method Report.

v344_2, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	627	41.5	
-2	na	М	5	0.3	
-1	dk	М	1	0.1	
10	1 : Primary education or first stage of basic education		25	1.7	2.8
21	2A: General lower secondary or second stage of basic education		127	8.4	14.5
31	3A: General (upper) secondary education		98	6.5	11.2
32	3B: Pre-vocational or pre-technical (upper) secondary education		465	30.8	53.0
41	4A: Post-secondary non-tertiary education		68	4.5	7.7
42	4B : Post-secondary non-tertiary education		14	0.9	1.6
51	5A: First stage of tertiary education (general)		53	3.5	6.0
52	5B: First stage of tertiary education (vocational)		14	0.9	1.6
60	6 : Second stage of tertiary education		14	0.9	1.6
	Sum		1510	100.1	100.0
	Valid Cases		877		

v344_3 - Educational level partner: ISCED-code three digits (Q116)

EDUCATIONAL LEVEL OF PARTNER: ISCED (THREE-DIGIT CODE) - HARMONIZED VARIABLE

Q116

<SHOW CARD 116>

What is the highest level your spouse or partner has completed in his or her education?

<"COMPLETED" MEANS DIPLOMA/CERTIFICATE>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 0 0: Pre-primary education or none education
- 100 1: Primary education or first stage of basic education
- 110 1: Primary education or first stage of basic education
- 210 2A: General lower secondary or second stage of basic education
- 220 2B: Pre-vocational or pre-technical lower secondary or second stage of basic education
- 230 2C: Vocational or technical lower secondary or second stage of basic education
- 310 3A: General (upper) secondary education
- 320 3B: Pre-vocational or pre-technical (upper) secondary education
- 330 3C: Vocational or technical (upper) secondary education
- 400 4: Post-secondary non-tertiary education
- 410 4A: Post-secondary non-tertiary education
- 420 4B: Post-secondary non-tertiary education
- 430 4C: Post-secondary non-tertiary education
- 510 5A: First stage of tertiary education (general)
- 511 5A (first degree): First stage of tertiary education (general)
- 512 5A (second degree): First stage of tertiary education (general)
- 513 5A (third degree): First stage of tertiary education (general)
- 520 5B: First stage of tertiary education (vocational)
- 600 6: Second stage of tertiary education
- 601 6 (first degree): Second stage of tertiary education
- 602 6 (second degree): Second stage of tertiary education

Note:

Harmonized variable: national educational categories have been recoded according to ISCED97. For country-specific variable see V344_cs. For detailed correspondence with country-specific categories see Method Report.

v344_3, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	627	41.5	
-2	na	М	5	0.3	
-1	dk	М	1	0.1	
100	1 : Primary education or first stage of basic education		25	1.7	2.8
210	2A: General lower secondary or second stage of basic education		127	8.4	14.5
310	3A: General (upper) secondary education		98	6.5	11.2
320	3B: Pre-vocational or pre-technical (upper) secondary education		465	30.8	53.0
410	4A: Post-secondary non-tertiary education		68	4.5	7.7
420	4B : Post-secondary non-tertiary education		14	0.9	1.6
511	5A (first degree): First stage of tertiary education (general)		9	0.6	1.0
512	5A (second degree): First stage of tertiary education (general)		44	2.9	5.0
520	5B: First stage of tertiary education (vocational)		14	0.9	1.6
600	6 : Second stage of tertiary education		14	0.9	1.6
	Sum		1510	100.1	100.0
	Valid Cases		877		

v344_cs - Country specific:ISCED code education partner (Q116)

Q116

<SHOW CARD 116>

What is the highest level your spouse or partner has completed in his or her education?

<"COMPLETED" MEANS DIPLOMA/CERTIFICATE>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 400101 1: AT Noch Schüler/in (ohne ersten allgemeinbildenden Schulabschluss)
- 400102 1: AT Schule beendet ohne ersten allgemeinbildenden Abschluss
- 400210 2A: AT Hauptschulabschluss, Volksschulabschluss 8. Klasse (auch bei Besuch einer mittleren/höheren Schule ohne en
- 400310 3A: AT AHS oder Berufsbildende Höhere Schule
- 400321 3B: AT Berufsbildende mittlere Schule (z.B. Handelsschule) oder Polytechnikum (9. Klasse)
- 400322 3B: AT Berufsschule und Lehre
- 400410 4A: AT Matura oder Hochschulreife im zweiten Bildungsweg
- 400420 4B: AT Gesundheits- und Krankenpflegeschule, Schulen für medizinisch-technischen Fachdienst
- 400511 5A (First degree): AT Bachalaureat (Hochschule, Universität, Fachhochschule)
- 400512 5A (Second degree): AT Magisterium/Diplom (Hochschule, Universität, Fachhochschule)
- 400520 5B: AT Berufsakademie oder Kolleg (mit Diplomabschluss)
- 400600 6: AT Promotion od. Habilitation (Hochschule, Universität, Fachhochschule)

Note:

Standardized country-specific variable: combined variable containing country identifier (see COUNTRY) and country-specific educational categories. This variable has been harmonized according to ISCED97 into V344 (1-digit), V344_2 (2-digit), and v344_3 (3-digit).

v344_cs

Value Lat	bel	Missing	Count	Percent	Valid Percent
-3 nar	p	М	627	41.5	
-2 na		М	5	0.3	
-1 dk		М	1	0.1	
400101 1:	: AT - Noch Schüler/in (ohne ersten allgemeinbildenden Schulabschluss)		3	0.2	0.3
400102 1:	: AT - Schule beendet ohne ersten allgemeinbildenden Abschluss		22	1.5	2.5
400210 2A	A : AT - Hauptschulabschluss, Volksschulabschluss 8. Klasse (auch bei Besuch einer mittleren/höhei		127	8.4	14.5
400310 3A	A : AT - AHS oder Berufsbildende Höhere Schule		98	6.5	11.2
400321 3B	3 : AT - Berufsbildende mittlere Schule (z.B. Handelsschule) oder Polytechnikum (9. Klasse)		75	5.0	8.5
400322 3B	3 : AT - Berufsschule und Lehre		390	25.8	44.4
400410 4A	A : AT - Matura oder Hochschulreife im zweiten Bildungsweg		68	4.5	7.7
400420 4B	3 : AT - Gesundheits- und Krankenpflegeschule, Schulen für medizinisch-technischen Fachdienst		14	0.9	1.6
400511 5A	A (First degree) : AT - Bachalaureat (Hochschule, Universität, Fachhochschule)		9	0.6	1.0
400512 5A	A (Second degree) : AT - Magisterium/Diplom (Hochschule, Universität, Fachhochschule)		44	2.9	5.0
400520 5B	3 : AT - Berufsakademie oder Kolleg (mit Diplomabschluss)		14	0.9	1.6
400600 6:	: AT - Promotion od. Habilitation (Hochschule, Universität, Fachhochschule)		14	0.9	1.6
Sui	m		1510	100.1	100.0
Val	lid Cases		877		

v345 - paid employment/no paid employment spouse/partner (Q117)

Q117

<SHOW CARD 117>

Is your spouse/partner gainfully employed at the moment or not? Please select from the card the employment status that applies to your spouse/partner.

<IF MORE THAN ONE JOB: ONLY FOR THE MAIN JOB>

<IF PAID EMPLOYMENT (CODE 1-3): GO TO Q118;</p>

IF NO PAID EMPLOYMENT (CODE 4-10) and -1, -2, -3: GO TO Q117A>

Paid employment

- 1 30 hours a week or more
- 2 Less than 30 hours a week
- 3 Self employed

No paid employment

- 4 Military Service
- 5 Retired/pensioned
- 6 Housewife not otherwise employed
- 7 Student
- 8 Unemployed
- 9 Disabled <ONLY IF SPOUSE/PARTNER DOES NOT WORK BECAUSE OF DISABILITY!>
- 10 Other, please specify (WRITE IN): ...
- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk

v345, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	627	41.5	
-2	na	М	4	0.3	
1	30h a week or more		442	29.3	50.3
2	less then 30h a week		63	4.2	7.2
3	self employed		38	2.5	4.3
5	retired/pensioned		204	13.5	23.2
6	housewife not otherwise employed		92	6.1	10.5
7	student		15	1.0	1.7
8	unemployed		17	1.1	1.9
9	disabled		2	0.1	0.2
10	other		6	0.4	0.7
	Sum		1510	100.0	100.0
	Valid Cases		879		

v345a - employment/self-employment: last job (Q117a)

Q117A

In his/her LAST job was he/she employed (either full time or part time) or was he/she self-employed?

<IF CODE 3: GO TO Q121>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 employed
- 2 self-employed
- 3 never had a paid job

Derivation:

Response category 7 'never had a paid job' was recoded into 3 'never had a paid job' and response category -3 'nap' was added.

v345a, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	1170	77.5	
-2	na	М	4	0.3	
-1	dk	М	1	0.1	
1	employed		259	17.2	77.3
2	self-employed		43	2.8	12.8
3	never had a paid job		33	2.2	9.9
	Sum		1510	100.0	100.0
	Valid Cases		335		

v346ISCO - kind of job spouse/partner - ISCO88 code (Q118)

Q118

<IF PARTNER HAS PAID EMPLOYMENT NOW, Q117 = 01-03 ASK:>

What is the name or title of your partner's main job?

<IF PARTNER HAD PAID EMPLOYMENT BEFORE Q117A = 1-2, ASK:>

What was the name or title of your partner's main job?

<IF PARTNER HAS/HAD MORE THAN ONE JOB. ASK HIS/HER OCCUPATION IN THE MAIN JOB! WRITE IN AS DETAILED AS POSSIBLE!>

Write in: ...

Q118a

In his or her main job, what kind of work do/did your partner most of the time?

<CODING INSTRUCTION: CODE ACCORDING TO ISCO88 (4 DIGITS) AFTER INTERVIEW ON THE BASIS OF Q118 AND Q118A> Code:

<IF NEVER HAD A JOB: GO TO Q121>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 110 [armed forces]
- 1000 legislators, senior officials & managers
- 1100 legislators & senior officials
- 1110 legislators
- 1120 senior government officials
- 1130 traditional chiefs & heads of villages
- 1140 senior officials special-interest organisations
- 1141 senior officials political-party organisations
- 1142 senior officials economic-interest organisations
- 1143 senior officials special-interest organisations
- 1200 corporate managers [large enterprise]
- 1210 directors & chief executives [large enterprise]
- 1220 department managers [large enterprise][production-operation]
- 1221 department managers agriculture, hunting, forestry & fishIng
- 1222 department managers manufacturing
- 1223 department managers construction
- 1224 department managers wholesale & retail trade
- 1225 department managers restaurants & hotels
- 1226 department managers transport, storage & communications
- 1227 department managers business services
- 1228 department managers personal care, cleaning etc
- 1229 operations department managers nec
- 1230 other department managers [large enterprise]
- 1231 finance & administration department managers
- 1232 personnel & industrial relations department managers
- 1233 sales & marketing department managers

1234	advertising & public relations department managers
1235	supply & distribution department managers
1236	computing services department managers
1237	research & development department managers
1239	other department managers nec
1240	[office manager]
1250	[military officers]
1251	[high military officers]
1252	[lower military officers]
1300	general managers [small enterprise]
1310	general managers [small enterprise]
1311	general managers agriculture, hunting forestry & fishing
1312	general managers manufacturing
1313	general managers construction
1314	general managers wholesale & retail trade
1315	general managers restaurants & hotels
1316	general managers transport, storage & communications
1317	general managers business services
1318	general managers personal care, cleaning etc services
1319	general managers nec
2000	professionals
2100	physical, mathematical & engineering science professionals
2110	physicists, chemists & related professionals
2111	physicists & astronomers
2112	meteorologists
2113	chemists
2114	geologists & geophysicists
2120	mathematicians, statisticians etc professionals
2121	mathematicians etc professionals
2122	statisticians
2130	computing professionals
2131	computer systems designers & analysts
2132	computer programmers
2139	computing professionals nec
2140	architects, engineers etc professionals
2141	architects town & traffic planners
2142	civil engineers
2143	electrical engineers
2144	electronics & telecommunications engineers
2145	mechanical engineers
2146	chemical engineers
2147	mining engineers metallurgists etc professionals
2148	cartographers & surveyors
2149	architects engineers etc professionals nec
2200	life science & health professionals
2210	life science professionals

2211 biologists, botanists zoologists etc professionals2212 pharmacologists, pathologists etc professionals

2213 agronomists etc professionals

2220 health professionals (except nursing) 2221 medical doctors 2222 dentists 2223 veterinarians 2224 pharmacists 2229 health professionals except nursing nec 2230 nursing & midwifery professionals 2300 teaching professionals 2310 higher education teaching professionals 2320 secondary education teaching professionals 2321 [second academ teacher] 2322 [second voc teacher] 2323 [middle school teacher] 2330 primary & pre-primary education teaching professionals 2331 primary education teaching professionals 2332 pre-primary education teaching professionals 2340 special education teaching professionals 2350 other teaching professionals 2351 education methods specialists 2352 school inspectors 2359 other teaching professionals nec 2400 other professionals 2410 business professionals 2411 accountants 2412 personnel & careers professionals 2419 business professionals nec 2420 legal professionals 2421 lawyers 2422 judges 2429 legal professionals nec 2430 archivists, librarians etc information professionals 2431 archivists & curators 2432 librarians etc information professionals 2440 social science etc professionals 2441 economists 2442 sociologists anthropologists etc professionals 2443 philosophers, historians & political scientists 2444 philologists translators & interpreters 2445 psychologists 2446 social work professionals 2450 writers & creative or performing artists 2451 authors journalists & other writers 2452 sculptors, painters etc artists 2453 composers musicians & singers 2454 choreographers & dancers 2455 film, stage etc actors & directors

2460 religious professionals

3000 technicians and associated professionals

3100 physical & engineering science associate professionals

3242 faith healers

GESIS StudyNo 4754, v1.1.0, http://dx.doi.org/doi:10.4232/1.10152

3110	physical & engineering science technicians
3111	chemical & physical science technicians
3112	civil engineering technicians
3113	electrical engineering technicians
3114	electronics & telecommunications engineering technicians
3115	mechanical engineering technicians
3116	chemical engineering technicians
3117	mining & metallurgical technicians
3118	draughtspersons
3119	physical & engineering science technicians nec
3120	computer associate professionals
3121	computer assistants
3122	computer equipment operators
3123	industrial robot controllers
3130	optical & electronic equipment operators
3131	photographers & elctr equipment operators
3132	broadcasting & telecommunications equipment operators
3133	medical equipment operators
3139	optical & electronic equipment operators nec
3140	ship & aircraft controllers & technicians
3141	ships engineers
3142	ships deck officers & pilots
3143	aircraft pilots etc associate professionals
3144	air traffic controllers
3145	air traffic safety technicians
3150	safety & ouality inspectors
3151	building & fire inspectors
3152	safety, health & quality inspectors
3200	life science & health associate professionals
3210	life science technicians etc associate professionals
3211	life science technicians
3212	agronomy & forestry technicians
3213	farming & forestry advisers
3220	modern health associate professionals except nursing
3221	medical assistants
3222	sanitarians
3223	dieticians & nutritionists
3224	optometrists & opticians
3225	dental assistants
3226	physiotherapists etc associate professionals
3227	veterinary assistants
3228	pharmaceutical assistants
3229	modern health associate professionals except nursing nec
3230	nursing & midwifery associate professionals
3231	nursing associate professionals
3232	midwifery associate professionals
3240	traditional medicine practitioners & faith healers
3241	traditional medicine practitioners

3300	teaching associate professionals
3310	primary education teaching associate professionals
3320	pre-primary education teaching associate professionals
3330	special education teaching associate professionals
3340	other teaching associate professionals
3400	other associate professionals
3410	finance & sales associate professionals
3411	securities & finance dealers & brokers
3412	insurance representatives
3413	estate agents
3414	travel consultants & organisers
3415	technical & commercial sales representatives
3416	buyers
3417	appraisers, valuers & auctioneers
3419	finance & sales associate professionals nec
3420	business services agents and trade brokers
3421	trade brokers
3422	clearing & forwarding agents
3423	employment agents & labour contractors
3429	business services agents & trade brokers nec
3430	administrative associate professionals
3431	administrative secretaries etc associate professionals
3432	legal etc business associate professionals
3433	bookkeepers
3434	statistical, mathematical etc associate professionals
3439	administrative associate professionals nec
3440	customs, tax etc government associate professionals
3441	customs & border inspectors
3442	government tax & excise officials
3443	government social benefits officials
3444	government licensing officials
3449	customs tax etc government associate professionals nec
3450	[police and army officers]
3451	[police inspectors-detectives]
3452	[armed forces non-commissioned officers + army nfs]
3460	social work associate professionals
3470	artistic, entertainment & sports associate professionals
3471	decorators & commercial designers
3472	radio, television & other announcers
3473	street night-club etc musicians singers & dancers
3474	clowns magicians acrobats etc associate professionals
3475	athletes sportspersons etc associate professionals
3480	religious associate professionals
4000	clerks
4100	office clerks
4110	secretaries & keyboard-operating clerks
4111	stenographers & typists
4112	word-processor etc operators

4113 data entry operators

- 4114 calculating-machine operators
- 4115 secretaries
- 4120 numerical clerks
- 4121 accounting & bookkeeping clerks
- 4122 statistical & finance clerks
- 4130 material-recording & transport clerks
- 4131 stock clerks
- 4132 production clerks
- 4133 transport clerks
- 4140 library, mail etc clerks
- 4141 library & filing clerks
- 4142 mail carriers & sorting clerks
- 4143 coding proof-reading etc clerks
- 4144 scribes etc workers
- 4190 other office clerks
- 4200 customer services clerks
- 4210 cashiers, tellers etc clerks
- 4211 cashiers & ticket clerks
- 4212 tellers & other counter clerks
- 4213 bookmakers & croupiers
- 4214 pawnbrokers & money-lenders
- 4215 debt-collectors etc workers
- 4220 client information clerks
- 4221 travel agency etc clerks
- 4222 receptionists & information clerks
- 4223 telephone switchboard operators
- 5000 service workers & shop & market sales workers
- 5100 personal & protective services workers
- 5110 travel attendants etc
- 5111 travel attendants & travel stewards
- 5112 transport conductors
- 5113 travel, museum guides
- 5120 housekeeping & restaurant services workers
- 5121 housekeepers etc workers
- 5122 cooks
- 5123 waiters waitresses & bartenders
- 5130 personal care etc work
- 5131 child-care workers
- 5132 institution-based personal care workers
- 5133 home based personal care workers
- 5139 personal care etc workers nec
- 5140 other personal services workers
- 5141 hairdressers barbers beauticians etc workers
- 5142 companions & valets
- 5143 undertakers & embalmers
- 5149 other personal services workers nec
- 5150 astrologers, fortune-tellers etc workers
- 5151 astrologers etc workers
- 5152 fortune-tellers, palmists etc workers

5160	protective services workers
5161	fire-fighters
5162	police officers
5163	prison guards
5164	[army: enlisted men - soldiers]
5169	protective services workers nec
5200	models, salespersons & demonstrators
5210	fashion & other models
5220	shop salespersons & demonstrators
5230	stall & market salespersons
6000	skilled agricultural & fishery workers
6100	market-oriented skilled agricultural & fishery workers
6110	market gardeners & cropgrowers
6111	field crop & vegetable growers
6112	tree & shrub crop growers
6113	gardeners, horticultural & nursery growers
6114	mixed-crop growers
6120	market-oriented animal producers etc workers
6121	dairy & livestock producers
6122	poultry producers
6123	apiarists & sericulturists
6124	mixed -animal producers
6129	market-oriented animal producers etc workers nec
6130	market-oriented crop & animal producers
6131	[mixed farmer]
6132	[farm foreman]
6133	[farmer nfs]
6134	[skilled farm worker nfs]
6140	forestry etc workers
6141	forestry workers & loggers
6142	charcoal burners etc workers
6150	fishery workers, hunters & trappers
6151	aquatic-life cultivation workers
6152	inland & coastal waters fishery workers
6153	deep-sea fishery workers
6154	hunters & trappers
6200	subsistence agricultural & fishery workers
6210	subsistence agricultural & fishery workers
7000	craft etc trades workers
7100	extraction & building trades workers
7110	miners, shotfirers, stone cutters & carvers
7111	miners & quarry workers
7112	shotfirers & blasters
7113	stone splitters cutters & carvers
7120	[builders] & ar trades workers
7121	builders traditional materials
7122	bricklayers & stonemasons

7123 concrete placers concrete finishers etc workers

7124 carpenters & joiners

- 7129 [builders] ar trades workers nec
- 7130 building finishers etc trades workers
- 7131 roofers
- 7132 floor layers & tile setters
- 7133 plasterers
- 7134 insulation workers
- 7135 glaziers
- 7136 plumbers & pipe fitters
- 7137 building etc electricians
- 7140 painters, building structure cleaners etc trades workers
- 7141 painters etc workers
- 7142 varnishers etc painters
- 7143 building structure cleaners
- 7200 metal, machinery etc trades workers
- 7210 metal moulders, welders, sheetmetal workers structural metal
- 7211 metal moulders & coremakers
- 7212 welders & flamecutters
- 7213 sheet-metal workers
- 7214 structural-metal preparers & erectors
- 7215 riggers & cable splicers
- 7216 underwater workers
- 7220 blacksmiths, tool-makers etc trades workers
- 7221 blacksmiths hammer-smiths & forgingpress workers
- 7222 tool-makers etc workers
- 7223 machine-tool setters & setter-operators
- 7224 metal wheel-grinders polishers & tool sharpeners
- 7230 machinery mechanics & fitters
- 7231 motor vehicle mechanics & fitters
- 7232 aircraft engine mechanics & fitters
- 7233 agricultural- or industrial-machinery mechanics & fitters
- 7234 [unskl garage worker, oiler]
- 7240 electrical & electronic equipment mechanics & fitters
- 7241 electrical mechanics & fitters
- 7242 electronics fitters
- 7243 electronics mechanics & servicers
- 7244 telegraph & telephone installers & servicers
- 7245 electrical line installers repairers & cable jointers
- 7300 precision, handicraft, printing etc trades workers
- 7310 precision workers in metal etc materials
- 7311 precision-instrument makers & repairers
- 7312 musical-instrument makers & tuners
- 7313 jewellery & precious-metal workers
- 7320 potters, glass-makers etc trades workers
- 7321 abrasive wheel formers potters etc workers
- 7322 glass-makers cutters grinders & finishers
- 7323 glass engravers & etchers
- 7324 glass ceramics etc decorative painters
- 7330 handicraft workers in wood, textile, leather etc
- 7331 handicraft workers in wood etc materials

- 7332 handicraft workers in textile leather etc materials
- 7340 printing etc trades workers
- 7341 compositors typesetters etc workers
- 7342 stereotypers & electrotypers
- 7343 printing engravers & etchers
- 7344 photographic etc workers
- 7345 bookbinders etc workers
- 7346 silk-screen, block & textile printers
- 7400 other craft etc trades workers
- 7410 food processing etc trades workers
- 7411 butchers fishmongers etc food preparers
- 7412 bakers-pastry cooks & confectionery makers
- 7413 dairy-products makers
- 7414 fruit, vegetable etc preservers
- 7415 food & beverage tasters & graders
- 7416 tobacco preparers & tobacco products makers
- 7420 wood treaters, cabinet-makers etc trades workers
- 7421 wood treaters
- 7422 cabinet-makers etc workers
- 7423 woodworking-machine setters & setteroperators
- 7424 basketry weavers brush makers etc workers
- 7430 textile, garment etc trades workers
- 7431 fibre preparers
- 7432 weavers knitters etc workers
- 7433 tailors dressmakers & hatters
- 7434 furriers etc workers
- 7435 textile leather etc pattern-makers & cutters
- 7436 sewers embroiderers etc workers
- 7437 upholsterers etc workers
- 7440 pelt, leather & shoemaking trades workers
- 7441 pelt dressers tanners & fellmongers
- 7442 shoe-makers etc workers
- 7500 [skilled manual worker nfs]
- 7510 [non farm foremen nfs]
- 7520 [skilled manual nfs]
- 7530 [apprentice skilled work nfs]
- 8000 plant & machine operators & assemblers
- 8100 stationary-plant etc operators
- 8110 mining- & mineral-processing plant operators
- 8111 mining-plant operators
- 8112 mineral-ore- & stone-processing-plant operators
- 8113 well drillers & borers etc workers
- 8120 metal-processing-plant operators
- 8121 ore & metal furnace operators
- 8122 metal melters casters & rolling-mill operators
- 8123 metal-heat-treating-plant operators
- 8124 metal drawers & extruders
- 8130 glass, ceramics etc plant operators
- 8131 glass & ceramics kiln etc machine operators

8139	glass, ceramics etc plant operators nec
8140	wood-processing- & papermaking-plant operators
8141	wood-processing-plant operators
8142	paper-pulp plant operators
8143	papermaking-plant operators
8150	chemical-processing-plant operators
8151	crushing- grinding- & chemical-mixing machinery operators
8152	chemical-heat-treating-plant operators
8153	chemical-filtering- & separating-equipment operators
8154	chemical-still & reactor operators
8155	petroleum- & natural-gas-refining-plant operators
8159	chemical-processing-plant operators nec
8160	power-production etc plant operators
8161	power-production plant operators
8162	steam-engine & boiler operators
8163	incinerator water-treatment etc plant operators
8170	automated-assembly-line & industrial-robot operators
8171	automated-assembly- line operators
8172	industrial-robot operators
8200	machine operators & assemblers
8210	metal- & mineral-products machine operators
8211	machine-tool operators
8212	cement & other mineral products machine operators
8220	chemical-products machine operators
8221	pharmaceutical- & toiletry-products machine operators
8222	ammunition- & explosive-products machine operators
8223	metal finishing- plating- & coating-machine operators
8224	photographic-products machine operators
8229	chemical-products machine operators nec
8230	rubber- & plastic-products machine operators
8231	rubber-products machine operators
8232	plastic-products machine operators
8240	wood-products machine operators
8250	printing- binding- & paper-products machine operators
8251	printing-machine operators
8252	bookbinding-machine operators
8253	paper-products machine operators
8260	textile- fur- & leather-products machine operators
8261	fibre-preparing-, spinning- & windingmachine operators
8262	weaving- & knitting-machine operators
8263	sewing-machine operators
8264	bleaching- dyeing- & cleaning-machine operators
8265	fur- & leather-preparing-machine operators
8266	shoemaking- etc machine operators
8269	textile- fur- & leather-products machine operators nec
8270	food etc products machine operators
8271	meat- & fish-processing-machine operators
8272	dairy-products machine operators

8273 grain- & spice-milling-machine operators

8274 baked-goods cereal & chocolate-products machine operators 8275 fruit- vegetable- & nut-processing-machine operators 8276 sugar production machine operators tea- coffee- & cocoa-processing-machine operators 8277 8278 brewers- wine & other beverage machine operators 8279 tobacco production machine operators 8280 assemblers 8281 mechanical-machinery assemblers 8282 electrical-equipment assemblers 8283 electronic-equipment assemblers 8284 metal- rubber- & plastic-products assemblers 8285 wood etc products assemblers 8286 paperboard textile etc products assemblers 8290 other machine operators & assemblers 8300 drivers & mobile-plant operators 8310 locomotive-engine drivers etc workers 8311 locomotive-engine drivers 8312 railway brakers signallers & shunters 8320 motor-vehicle drivers 8321 motor-cycle drivers 8322 car taxi & van drivers 8323 bus & tram drivers 8324 heavy truck & lorry drivers 8330 agricultural & other mobile-plant operators 8331 motorised farm & forestry plant operators 8332 earth-moving- etc plant operators 8333 crane hoist etc plant operators 8334 lifting-truck operators 8340 ships deck crews etc workers 8400 [semiskld worker nfs] 9000 elementary occupations 9100 sales & services elementary occupations 9110 street vendors etc workers 9111 street food vendors 9112 street vendors non-food products 9113 door-to-door & telephone salespersons 9120 street services elementary occupations 9130 domestic etc helpers cleaners & launderers 9131 domestic helpers & cleaners 9132 helpers & cleaners in establishments 9133 hand-launderers & pressers 9140 building caretakers, window etc cleaners 9141 building caretakers 9142 vehicle, window etc cleaners 9150 messengers, porters, doorkeepers etc workers 9151 messengers package & luggage porters & deliverers

9152 doorkeepers watchpersons etc workers

9160 garbage collectors etc labourers

9153 vending-machine money collectors meter readers etc workers

- 9161 garbage collectors
- 9162 sweepers etc labourers
- 9200 agricultural, fishery etc labourers
- 9210 agricultural, fishery etc labourers
- 9211 farm-hands & labourers
- 9212 forestry labourers
- 9213 fishery hunting & trapping labourers
- 9300 labourers in mining, construction, manufacturing & transport
- 9310 mining & construction labourers
- 9311 mining & quarrying labourers
- 9312 construction & maintenance labourers: roads dams etc
- 9313 building construction labourers
- 9320 manufacturing labourers
- 9321 assembling labourers
- 9322 hand packers & other manufacturing labourers
- 9330 transport labourers & freight handlers
- 9331 hand or pedal vehicle drivers
- 9332 drivers of animal-drawn vehicles & machinery
- 9333 freight handlers

Note:

Standardized variable on the basis of two open-ended questions on occupation and kind of activity (Q118 and Q118a) post-coded according to 4-digit ISCO88.

Data for original string variable is not available in this dataset.

v346SIOPS - occupational status spouse/partner - SIOPS (Q118)

COMPUTED VARIABLE - SIOPS: Standard International Occupational Prestige Scale (Treiman, 1977)

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk

Note:

Source classification: Ganzeboom, H. B. G. & Treiman, D. J. (1996), Internationally Comparable Measures of Occupational Status for the 1988 International Standard Classification of Occupations. Social Science Research, 25, 201-239.

Source syntax to compute classification: http://home.fsw.vu.nl/hbg.ganzeboom/pisa/index.htm

v346ISEI - occupational status spouse/partner - ISEI (Q118)

COMPUTED VARIABLE - ISEI: International Socio-Economic Index of Occupational Status (Ganzeboom et al., 1992)

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk

Note:

Source classification: Ganzeboom, H. B. G. & Treiman, D. J. (1996), Internationally Comparable Measures of Occupational Status for the 1988 International Standard Classification of Occupations. Social Science Research, 25, 201-239.

Source syntax to compute classification: http://home.fsw.vu.nl/hbg.ganzeboom/pisa/index.htm

v346ESeC - occupational status spouse/partner - European ESeC (Q118)

COMPUTED VARIABLE - European Socio-economic Classification (ESeC)

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 Large employers, higher managers/professionals
- 2 Lower managers/professionals, higher supervisory/technicians
- 3 Intermediate occupations
- 4 Small employers and self-employed (non-agriculture)
- 5 Small employers and self-employed (agriculture)
- 6 Lower supervisors and technicians
- 7 Lower sales and service
- 8 Lower technical
- 9 Routine

Note:

Source: http://www.iser.essex.ac.uk/research/esec/matrices-and-syntax

Date: 27-09-2006 (last update syntax).

v346ESeC, weighted by weight

Value Label	Missing	Count	Percent	Valid Percent
-5 other missing	М	5	0.3	
-3 nap	М	660	43.7	
-2 na	М	18	1.2	
1 Large employers, higher mgrs/professionals		50	3.3	6.0
2 Lower mgrs/professionals, higher supervisory/technicians		141	9.3	17.0
3 Intermediate occupations		136	9.0	16.4
4 Small employers and self-employed (non-agriculture)		43	2.8	5.2
5 Small employers and self-employed (agriculture)		24	1.6	2.9
6 Lower supervisors and technicians		72	4.8	8.7
7 Lower sales and service		112	7.4	13.5
8 Lower technical		95	6.3	11.5
9 Routine		154	10.2	18.6
Sum		1510	100.0	100.0
Valid Cases		827		

v346egp - occupational status spouse/partner - egp11 (Q118)

COMPUTED VARIABLE - EGP: EGP class scheme (Erikson/Goldthorpe/Portocarero, 1979)

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 I: Higher Controllers
- 2 II: Lower Controllers
- 3 IIIa: Routine Non-manual
- 4 IIIb: Lower Sales-Service
- 5 IVa: Self-employed with employees
- 6 IVb: Self-employed with no employees
- 7 V: Manual Supervisors
- 8 VI: Skilled Worker
- 9 VIIa: Unskilled Worker
- 10 VIIb: Farm Labor
- 11 IVc: Self-employed Farmer

Note:

Source classification: Ganzeboom, H. B. G. & Treiman, D. J. (1996), Internationally Comparable Measures of Occupational Status for the 1988 International Standard Classification of Occupations. Social Science Research, 25, 201-239.

 $Source\ syntax\ to\ compute\ classification:\ http://home.fsw.vu.nl/hbg.ganzeboom/pisa/index.htm$

v346egp, weighted by weight

Label	Missing	Count	Percent	Valid Percent
other missing	М	5	0.3	
nap	М	660	43.7	
na	М	18	1.2	
I :Higher Controllers		81	5.4	9.8
II :Lower Controllers		151	10.0	18.3
Illa:Routine Nonmanual		93	6.2	11.2
IIIb:Lower Sales-Service		145	9.6	17.5
IVa:Selfempl with empl		25	1.7	3.0
IVb:Selfempl no empl		15	1.0	1.8
V :Manual Supervisors		27	1.8	3.3
VI :Skilled Worker		96	6.4	11.6
VIIa:Unskilled Worker		156	10.3	18.9
VIIb:Farm Labor		10	0.7	1.2
IVc:Selfempl Farmer		28	1.9	3.4
Sum		1510	100.0	100.0
Valid Cases		827		
	other missing nap na I:Higher Controllers II:Lower Controllers IIIa:Routine Nonmanual IIIb:Lower Sales-Service IVa:Selfempl with empl IVb:Selfempl no empl V:Manual Supervisors VI:Skilled Worker VIIa:Unskilled Worker VIIb:Farm Labor IVc:Selfempl Farmer Sum	other missing M nap M na M I:Higher Controllers II:Lower Controllers IIIa:Routine Nonmanual IIIb:Lower Sales-Service IVa:Selfempl with empl IVb:Selfempl no empl V:Manual Supervisors VI:Skilled Worker VIIa:Unskilled Worker VIIb:Farm Labor IVc:Selfempl Farmer Sum	other missing M 5 nap M 660 na M 18 I : Higher Controllers 81 II : Lower Controllers 151 Illa: Routine Nonmanual 93 Illb: Lower Sales-Service 145 IVa: Selfempl with empl 25 IVb: Selfempl no empl 15 V : Manual Supervisors 27 VI : Skilled Worker 96 VIIa: Unskilled Worker 156 VIIb: Farm Labor 10 IVc: Selfempl Farmer 28 Sum 1510	other missing M 5 0.3 nap M 660 43.7 na M 18 1.2 I :Higher Controllers 81 5.4 II :Lower Controllers 151 10.0 Illa:Routine Nonmanual 93 6.2 Illb:Lower Sales-Service 145 9.6 IVa:Selfempl with empl 25 1.7 IVb:Selfempl no empl 15 1.0 V :Manual Supervisors 27 1.8 VI :Skilled Worker 96 6.4 VIIa:Unskilled Worker 156 10.3 VIIb:Farm Labor 10 0.7 IVc:Selfempl Farmer 28 1.9 Sum 1510 100.0

v347 - spouse/partner had/has how many employees (Q119)

0110

<ASK IF SPOUSE/PARTNER IS CURRENTLY SELF-EMPLOYED (Q117=3) OR IF PARTNER WAS SELF-EMPLOYED IN THE LAST JOB (Q117A=2)>

How many employees does/did he or she have?

<SKIP TO Q121>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 none
- 2 1-9
- 3 10-24
- 4 25 or more

v347, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	1430	94.7	
1	none		39	2.6	48.8
2	1-9		36	2.4	45.0
3	10-24		3	0.2	3.8
4	25 or more		2	0.1	2.5
	Sum		1510	100.0	100.0
	Valid Cases		80		

v348 - does spouse/partner supervise someone (Q120)

Q120

<ASK IF SPOUSE/PARTNER IS CURRENTLY EMPLOYED (Q117=1 OR 2), OR IF SPOUSE/PARTNER WAS NOT SELF-EMPLOYED IN THE LAST JOB (Q117A=1)>

Does/did he or she have any responsibility for supervising the work of other employees?

<IF CODE 2, -1, -2; GO TO Q121>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v348, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	745	49.3	
-2	na	М	5	0.3	
-1	dk	М	19	1.3	
1	yes		190	12.6	25.6
2	no		551	36.5	74.4
	Sum		1510	100.0	100.0
	Valid Cases		741		

v348a - how many people does she/he supervise (Q120a)

Q120a

How many other employees does/did he or she supervise?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 1-9
- 2 10-24
- 3 25 or more

v348a, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	1320	87.4	
-1	dk	М	10	0.7	
1	1-9		130	8.6	72.2
2	10-24		22	1.5	12.2
3	25 or more		28	1.9	15.6
	Sum		1510	100.0	100.0
	Valid Cases		180		

v349 - respondent experienced unemployment longer than 3 months (Q121)

Q121

<ASK ALL>

During the last five years, have you experienced a continuous period of unemployment longer than 3 months?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v349, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	131	8.7	
-2	na	М	5	0.3	
-1	dk	М	6	0.4	
1	yes		155	10.3	11.3
2	no		1214	80.4	88.7
	Sum		1510	100.1	100.0
	Valid Cases		1368		

v350 - spouse/partner experienced unemployment longer than 3 months (Q122)

Q122

< ONLY IF LIVING WITH A PARTNER (Q97=1 OR 2 OR Q99=1); OTHERWISE GO TO Q123>

And your spouse/partner, has he/she experienced a continuous period of unemployment longer than 3 months during the last five years?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v350, weighted by weight

, .	roiginou by troi	9			
Value	Label	Missing	Count	Percent	Valid Percent
-5	other missing	М	33	2.2	
-3	nap	М	627	41.5	
-2	na	М	3	0.2	
-1	dk	М	3	0.2	
1	yes		65	4.3	7.7
2	no		780	51.7	92.3
	Sum		1510	100.1	100.0
	Valid Cases		844		

v351 - dependency on social security during last 5 years respondent (Q123)

Q123

During the last five years, have you been dependent on social security at any time?

<SOCIAL SECURITY REFERS TO SOCIAL WELFARE BENEFITS (MEANS TESTED). IMPORTANT: THIS DOES NOT INCLUDE ENTITLEMENTS TO UNEMPLOYMENT OR DISABILITY BENEFITS OR TO PENSIONS>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v351, weighted by weight

,	- 5 ,				
Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	4	0.3	
-1	dk	М	2	0.1	
1	yes		101	6.7	6.7
2	no		1404	93.0	93.3
	Sum		1510	100.1	100.0
	Valid Cases		1504		

v352 - dependency on social security during last 5 years spouse/partner (Q124)

Q124

<ONLY IF LIVING WITH A PARTNER (Q97=1 OR 2, OR Q99=1); OTHERWISE GO TO Q125>

And your spouse/partner, has he/she been dependent on social security during the last five years at any time?

 $<\!\!\mathsf{SOCIAL}\ \mathsf{SECURITY}\ \mathsf{REFERS}\ \mathsf{TO}\ \mathsf{SOCIAL}\ \mathsf{WELFARE}\ \mathsf{BENEFITS}\ (\mathsf{MEANS}\ \mathsf{TESTED}).\ \mathsf{IMPORTANT}\ \mathsf{THIS}\ \mathsf{DOES}\ \mathsf{NOT}\ \mathsf{INCLUDE}$

ENTITLEMENTS TO UNEMPLOYMENT OR DISABILITY BENEFITS OR TO PENSIONS>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v352, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	627	41.5	
-2	na	М	2	0.1	
1	yes		30	2.0	3.4
2	no		852	56.4	96.6
	Sum		1510	100.1	100.0
	Valid Cases		882		

v353WK - Weekly household income (Q125)

WEEKLY HOUSEHOLD INCOME IN EUROS- HARMONIZED VARIABLE

Q125

<SHOW CARD 125>

Here is a list of incomes and we would like to know in what group your household is, counting all wages, salaries, pensions and other incomes that come in. Just give the letter of the group your household falls into, after taxes and other deductions.

Approximate WEEKLY

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 Less than € 40
- 2 €40 to under €70
- 3 €70 to under €120
- 4 €120 to under €230
- 5 €230 to under €350
- 6 €350 to under €460
- 7 €460 to under €580
- 8 €580 to under €690
- 9 €690 to under €1150
- 10 €1150 to under €1730
- 11 €1730 to under €2310
- 12 €2310 or more

Note:

Harmonized variable recoded in nominal euros. For country-specific variable in national currency, see V353W_cs. For detailed conversion rates, see Method Report.

v353WK, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	M	144	9.5	
-1	dk	М	115	7.6	
1	Less than € 40		3	0.2	0.2
2	€40 to under €70		16	1.1	1.3
3	€70 to under €120		38	2.5	3.0
4	€120 to under €230		143	9.5	11.4
5	€230 to under €350		243	16.1	19.4
6	€350 to under €460		194	12.8	15.5
7	€460 to under €580		263	17.4	21.0
8	€580 to under €690		139	9.2	11.1
9	€690 to under €1150		174	11.5	13.9
10	€1150 to under €1730		26	1.7	2.1
11	€1730 to under €2310		8	0.5	0.6
12	€2310 or more		3	0.2	0.2
	Sum		1510	99.9	100.0
	Valid Cases		1251		

v353MM - monthly household income (Q125)

MONTHLY HOUSEHOLD INCOME IN EUROS - HARMONIZED VARIABLE

Q125

<SHOW CARD 125>

Here is a list of incomes and we would like to know in what group your household is, counting all wages, salaries, pensions and other incomes that come in. Just give the letter of the group your household falls into, after taxes and other deductions.

Approximate MONTHLY

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 Less than €150
- 2 €150 to under €300
- 3 €300 to under €500
- 4 €500 to under €1000
- 5 €1000 to under €1500
- 6 €1500 to under €2000
- 7 €2000 to under €2500
- 8 €2500 to under €3000
- 9 €3000 to under €5000
- 10 €5000 to under €7500
- 11 €7500 to under €10000
- 12 €10000 or more

Note:

Harmonized variable recoded in nominal euros. For country-specific variable in national currency, see V353M_cs. For detailed conversion rates, see Method Report.

v353MM, weighted by weight

	, , , ,				
Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	144	9.5	
-1	dk	М	115	7.6	
1	Less than €150		3	0.2	0.2
2	€150 to under €300		16	1.1	1.3
3	€300 to under €500		38	2.5	3.0
4	€500 to under €1000		143	9.5	11.4
5	€1000 to under €1500		243	16.1	19.4
6	€1500 to under €2000		194	12.8	15.5
7	€2000 to under €2500		263	17.4	21.0
8	€2500 to under €3000		139	9.2	11.1
9	€3000 to under €5000		174	11.5	13.9
10	€5000 to under €7500		26	1.7	2.1
11	€7500 to under €10000		8	0.5	0.6
12	€10000 or more		3	0.2	0.2
	Sum		1510	99.9	100.0
	Valid Cases		1251		

v353YR - annual household income (Q125)

ANNUAL HOUSEHOLD INCOME IN EUROS - HARMONIZED VARIABLE

Q125

<SHOW CARD 125>

Here is a list of incomes and we would like to know in what group your household is, counting all wages, salaries, pensions and other incomes that come in. Just give the letter of the group your household falls into, after taxes and other deductions.

Approximate ANNUAL

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 Less than €1800
- 2 €1800 to under €3600
- 3 €3600 to under €6000
- 4 €6000 to under €12000
- 5 €12000 to under €18000
- 6 €18000 to under €24000
- 7 €24000 to under €30000
- 8 €30000 to under €36000
- 9 €36000 to under €60000
- 10 €60000 to under €9000011 €90000 to under €120000
- 12 €120000 or more

Note:

Harmonized variable recoded in nominal euros. For country-specific variable in national currency, see V353Y_cs. For detailed conversion rates, see Method Report.

v353YR, weighted by weight

	, , , ,				
Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	144	9.5	
-1	dk	М	115	7.6	
1	Less than €1800		3	0.2	0.2
2	€1800 to under €3600		16	1.1	1.3
3	€3600 to under €6000		38	2.5	3.0
4	€6000 to under €12000		143	9.5	11.4
5	€12000 to under €18000		243	16.1	19.4
6	€18000 to under €24000		194	12.8	15.5
7	€24000 to under €30000		263	17.4	21.0
8	€30000 to under €36000		139	9.2	11.1
9	€36000 to under €60000		174	11.5	13.9
10	€60000 to under €90000		26	1.7	2.1
11	€90000 to under €120000		8	0.5	0.6
12	€120000 or more		3	0.2	0.2
	Sum		1510	99.9	100.0
	Valid Cases		1251		

v353W_cs - Country specific: Weekly household income (Q125)

Q125_1

<ASK ALL>

<SHOW CARD 125>

Here is a list of incomes and we would like to know in what group your household is, counting all wages, salaries, pensions and other incomes that come in. Just give the letter of the group your household falls into, after taxes and other deductions.

Approximate WEEKLY

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 40001 AT Less than € 40
- 40002 AT €40 to under €70
- 40003 AT €70 to under €120
- 40004 AT €120 to under €230
- 40005 AT €230 to under €350
- 40006 AT €350 to under €460
- 40007 AT €460 to under €580
- 40008 AT €580 to under €690
- 40009 AT €690 to under €1150
- 40010 AT €1150 to under €1730
- 40011 AT €1730 to under €2310
- 40012 AT €2310 or more

Note:

Standardized country-specific variable: combined variable containing country identifier (see COUNTRY) and country-specific income categories. This variable has been harmonized in nominal euros into v353WK. For detailed conversion rates, see Method Report.

v353W_cs, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	144	9.5	
-1	dk	М	115	7.6	
40001	AT - Less than € 40		3	0.2	0.2
40002	AT - €40 to under €70		16	1.1	1.3
40003	AT - €70 to under €120		38	2.5	3.0
40004	AT - €120 to under €230		143	9.5	11.4
40005	AT - €230 to under €350		243	16.1	19.4
40006	AT - €350 to under €460		194	12.8	15.5
40007	AT - €460 to under €580		263	17.4	21.0
40008	AT - €580 to under €690		139	9.2	11.1
40009	AT - €690 to under €1150		174	11.5	13.9
40010	AT - €1150 to under €1730		26	1.7	2.1
40011	AT - €1730 to under €2310		8	0.5	0.6
40012	AT - €2310 or more		3	0.2	0.2
	Sum		1510	99.9	100.0
	Valid Cases		1251		

v353M_cs - Country specific: Monthly household income (Q125)

Q125_2

<ASK ALL>

<SHOW CARD 125>

Here is a list of incomes and we would like to know in what group your household is, counting all wages, salaries, pensions and other incomes that come in. Just give the letter of the group your household falls into, after taxes and other deductions.

Approximate MONTHLY

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 40001 AT Less than €150
- 40002 AT €150 to under €300
- 40003 AT €300 to under €500
- 40004 AT €500 to under €1000
- 40005 AT €1000 to under €1500
- 40006 AT €1500 to under €2000
- 40007 AT €2000 to under €2500
- 40008 AT €2500 to under €3000
- 40009 AT €3000 to under €5000
- 40010 AT €5000 to under €7500
- 40011 AT €7500 to under €10000
- 40012 AT €10000 or more

Note:

Standardized country-specific variable: combined variable containing country identifier (see COUNTRY) and country-specific income categories. This variable has been harmonized in nominal euros into v353MM. For detailed conversion rates, see Method Report.

v353M_cs, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	144	9.5	
-1	dk	М	115	7.6	
40001	AT - Less than €150		3	0.2	0.2
40002	AT - €150 to under €300		16	1.1	1.3
40003	AT - €300 to under €500		38	2.5	3.0
40004	AT - €500 to under €1000		143	9.5	11.4
40005	AT - €1000 to under €1500		243	16.1	19.4
40006	AT - €1500 to under €2000		194	12.8	15.5
40007	AT - €2000 to under €2500		263	17.4	21.0
40008	AT - €2500 to under €3000		139	9.2	11.1
40009	AT - €3000 to under €5000		174	11.5	13.9
40010	AT - €5000 to under €7500		26	1.7	2.1
40011	AT - €7500 to under €10000		8	0.5	0.6
40012	AT - €10000 or more		3	0.2	0.2
	Sum		1510	99.9	100.0
	Valid Cases		1251		

v353Y_cs - Country specific: Annual household income (Q125)

Q125_3

<ASK ALL>

<SHOW CARD 125>

Here is a list of incomes and we would like to know in what group your household is, counting all wages, salaries, pensions and other incomes that come in. Just give the letter of the group your household falls into, after taxes and other deductions.

Approximate ANNUAL

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 40001 AT Less than €1800
- 40002 AT €1800 to under €3600
- 40003 AT €3600 to under €6000
- 40004 AT €6000 to under €12000
- 40005 AT €12000 to under €18000
- 40006 AT €18000 to under €24000
- 40007 AT €24000 to under €30000
- 40008 AT €30000 to under €36000
- 40009 AT €36000 to under €60000
- 40010 AT €60000 to under €90000
- 40011 AT €90000 to under €120000
- 40012 AT €120000 or more

Note:

Standardized country-specific variable: combined variable containing country identifier (see COUNTRY) and country-specific income categories. This variable has been harmonized in nominal euros into v353YR. For detailed conversion rates, see Method Report.

v353Y_cs, weighted by weight

					\/ E15
Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	144	9.5	
-1	dk	М	115	7.6	
40001	AT - Less than €1800		3	0.2	0.2
40002	AT - €1800 to under €3600		16	1.1	1.3
40003	AT - €3600 to under €6000		38	2.5	3.0
40004	AT - €6000 to under €12000		143	9.5	11.4
40005	AT - €12000 to under €18000		243	16.1	19.4
40006	AT - €18000 to under €24000		194	12.8	15.5
40007	AT - €24000 to under €30000		263	17.4	21.0
40008	AT - €30000 to under €36000		139	9.2	11.1
40009	AT - €36000 to under €60000		174	11.5	13.9
40010	AT - €60000 to under €90000		26	1.7	2.1
40011	AT - €90000 to under €120000		8	0.5	0.6
40012	AT - €120000 or more		3	0.2	0.2
	Sum		1510	99.9	100.0
	Valid Cases		1251		

v354 - lived with parents at the age of 14 (Q126)

Q126

When you were 14, did you live with your parents?

<REPLACE FATHER BY MOTHER AND HIS BY HER IN QUESTIONS 127 TO 131 IF Q126=3>

<IF CODE 4: GO TO Q133>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 Yes, with both parents
- 2 Only with my father
- 3 Only with my mother
- 4 No, I did not live with my parents

v354, weighted by weight

,	. 5 , 5 .				
Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
1	yes with both parents		1290	85.4	85.7
2	only with father		20	1.3	1.3
3	only with mother		155	10.3	10.3
4	no did not live with parents		41	2.7	2.7
	Sum		1510	99.9	100.0
	Valid Cases		1507		

v355 - Educational level father [mother]: ISCED-code one digit (Q127)

EDUCATIONAL LEVEL OF FATHER [MOTHER]: ISCED (ONE-DIGIT CODE) - HARMONIZED VARIABLE

[Q126. <Replace father by mother in Q127-Q131 if respondent lived only with mother at age of 14>]

Q127

<SHOW CARD 127>

What is the highest level your father [mother] has completed in education?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 0 0: Pre-primary education or none education
- 1 1: Primary education or first stage of basic education
- 2 2: Lower secondary or second stage of basic education
- 3 3: (Upper) secondary education
- 4 4: Post-secondary non-tertiary education
- 5 5: First stage of tertiary education
- 6 6: Second stage of tertiary education

Note:

Harmonized variable: national educational categories have been recoded according to ISCED97. For country-specific variable, see V355_cs. For detailed correspondence with country-specific categories, see Method Report.

v355_2 - Educational level father [mother]: ISCED-code two digits (Q127)

EDUCATIONAL LEVEL OF FATHER [MOTHER]: ISCED (TWO-DIGIT CODE) - HARMONIZED VARIABLE

[Q126. <Replace father by mother in Q127-Q131 if respondent lived only with mother at age of 14>]

Q127

<SHOW CARD 127>

What is the highest level your father [mother] has completed in education?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 0 0: Pre-primary education or none education
- 10 1: Primary education or first stage of basic education
- 11 1: Primary education or first stage of basic education
- 21 2A: General lower secondary or second stage of basic education
- 22 2B: Pre-vocational or pre-technical lower secondary or second stage of basic education
- 23 2C: Vocational or technical lower secondary or second stage of basic education
- 31 3A: General (upper) secondary education
- 32 3B: Pre-vocational or pre-technical (upper) secondary education
- 33 3C: Vocational or technical (upper) secondary education
- 40 4: Post-secondary non-tertiary education
- 41 4A: Post-secondary non-tertiary education
- 42 4B: Post-secondary non-tertiary education
- 43 4C: Post-secondary non-tertiary education
- 51 5A: First stage of tertiary education (general)
- 52 5B: First stage of tertiary education (vocational)
- 60 6: Second stage of tertiary education

Note:

Harmonized variable: national educational categories have been recoded according to ISCED97. For country-specific variable, see V355_cs. For detailed correspondence with country-specific categories, see Method Report.

v355_2, weighted by weight

Value Label		Missing	Count	Percent	Valid Percent
-5 other missing		M	3	0.2	
-3 nap		M	41	2.7	
-2 na		M	40	2.6	
-1 dk		M	38	2.5	
10 1: Primary ed	ucation or first stage of basic education		67	4.4	4.8
21 2A : General lo	ower secondary or second stage of basic education		438	29.0	31.6
31 3A: General (upper) secondary education		82	5.4	5.9
32 3B : Pre-vocat	tional or pre-technical (upper) secondary education		645	42.7	46.5
41 4A : Post-seco	ondary non-tertiary education		61	4.0	4.4
42 4B : Post-seco	ondary non-tertiary education		5	0.3	0.4
51 5A : First stage	e of tertiary education (general)		48	3.2	3.5
52 5B : First stage	e of tertiary education (vocational)		13	0.9	0.9
60 6: Second sta	age of tertiary education		28	1.9	2.0
Sum			1510	99.9	100.0
Valid Cases			1388		

v355_3 - Educational level father [mother]: ISCED-code three digits (Q127)

EDUCATIONAL LEVEL OF FATHER [MOTHER]: ISCED (THREE-DIGIT CODE) - HARMONIZED VARIABLE

[Q126. < Replace father by mother in Q127-Q131 if respondent lived only with mother at age of 14>]

Q127

<SHOW CARD 127>

What is the highest level your father [mother] has completed in education?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 0 0: Pre-primary education or none education
- 100 1: Primary education or first stage of basic education
- 110 1: Primary education or first stage of basic education
- 210 2A: General lower secondary or second stage of basic education
- 220 2B: Pre-vocational or pre-technical lower secondary or second stage of basic education
- 230 2C: Vocational or technical lower secondary or second stage of basic education
- 310 3A: General (upper) secondary education
- 320 3B: Pre-vocational or pre-technical (upper) secondary education
- 330 3C: Vocational or technical (upper) secondary education
- 400 4: Post-secondary non-tertiary education
- 410 4A: Post-secondary non-tertiary education
- 420 4B: Post-secondary non-tertiary education
- 430 4C: Post-secondary non-tertiary education
- 510 5A: First stage of tertiary education (general)
- 511 5A (first degree): First stage of tertiary education (general)
- 512 5A (second degree): First stage of tertiary education (general)
- 513 5A (third degree): First stage of tertiary education (general)
- 520 5B: First stage of tertiary education (vocational)
- 600 6: Second stage of tertiary education
- 601 6 (first degree): Second stage of tertiary education
- 602 6 (second degree): Second stage of tertiary education

Note:

Harmonized variable: national educational categories have been recoded according to ISCED97. For country-specific variable, see V355_cs. For detailed correspondence with country-specific categories, see Method Report.

v355_3, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-5	other missing	М	3	0.2	
-3	пар	М	41	2.7	
-2	na	М	40	2.6	
-1	dk	М	38	2.5	
100	1 : Primary education or first stage of basic education		67	4.4	4.8
210	2A: General lower secondary or second stage of basic education		438	29.0	31.6
310	3A: General (upper) secondary education		82	5.4	5.9
320	3B: Pre-vocational or pre-technical (upper) secondary education		645	42.7	46.5
410	4A: Post-secondary non-tertiary education		61	4.0	4.4
420	4B: Post-secondary non-tertiary education		5	0.3	0.4
511	5A (first degree): First stage of tertiary education (general)		7	0.5	0.5
512	5A (second degree): First stage of tertiary education (general)		41	2.7	3.0
520	5B: First stage of tertiary education (vocational)		13	0.9	0.9
600	6 : Second stage of tertiary education		28	1.9	2.0
	Sum		1510	99.9	100.0
	Valid Cases		1388		

v355_cs - Country specific:ISCED code education father (Q127)

[Q126 < REPLACE FATHER BY MOTHER AND HIS BY HER IN QUESTIONS 127 TO 131 IF Q126=3>]

Q127

<SHOW CARD 127>

What is the highest level your father [mother] has completed in education?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 400101 1: AT Noch Schüler/in (ohne ersten allgemeinbildenden Schulabschluss)
- 400102 1: AT Schule beendet ohne ersten allgemeinbildenden Abschluss
- 400210 2A: AT Hauptschulabschluss, Volksschulabschluss 8. Klasse (auch bei Besuch einer mittleren/höheren Schule ohne ents
- 400310 3A: AT AHS oder Berufsbildende Höhere Schule
- 400321 3B: AT Berufsbildende mittlere Schule (z.B. Handelsschule) oder Polytechnikum (9. Klasse)
- 400322 3B: AT Berufsschule und Lehre
- 400410 4A: AT Matura oder Hochschulreife im zweiten Bildungsweg
- 400420 4B: AT Gesundheits- und Krankenpflegeschule, Schulen für medizinisch-technischen Fachdienst
- 400511 5A (First degree): AT Bachalaureat (Hochschule, Universität, Fachhochschule)
- 400512 5A (Second degree): AT Magisterium/Diplom (Hochschule, Universität, Fachhochschule)
- 400520 5B: AT Berufsakademie oder Kolleg (mit Diplomabschluss)
- 400600 6: AT Promotion od. Habilitation (Hochschule, Universität, Fachhochschule)

Note:

Standardized country-specific variable: combined variable containing country identifier (see COUNTRY) and country-specific educational categories. This variable has been harmonized according to ISCED97 into V355 (1-digit), V355_2 (2-digit), and v355_3 (3-digit).

v355_cs

Value	Label	Missing	Count	Percent	Valid Percent
-5	other missing	М	3	0.2	
-3	пар	М	41	2.7	
-2	na	М	40	2.6	
-1	dk	М	38	2.5	
400102	1 : AT - Schule beendet ohne ersten allgemeinbildenden Abschluss		67	4.4	4.8
400210	2A : AT - Hauptschulabschluss, Volksschulabschluss 8. Klasse (auch bei Besuch einer mittleren/höhei		438	29.0	31.6
400310	3A : AT - AHS oder Berufsbildende Höhere Schule		82	5.4	5.9
400321	3B : AT - Berufsbildende mittlere Schule (z.B. Handelsschule) oder Polytechnikum (9. Klasse)		58	3.8	4.2
400322	3B : AT - Berufsschule und Lehre		587	38.9	42.3
400410	4A : AT - Matura oder Hochschulreife im zweiten Bildungsweg		61	4.0	4.4
400420	4B : AT - Gesundheits- und Krankenpflegeschule, Schulen für medizinisch-technischen Fachdienst		5	0.3	0.4
400511	5A (First degree): AT - Bachalaureat (Hochschule, Universität, Fachhochschule)		7	0.5	0.5
400512	5A (Second degree): AT - Magisterium/Diplom (Hochschule, Universität, Fachhochschule)		41	2.7	3.0
400520	5B : AT - Berufsakademie oder Kolleg (mit Diplomabschluss)		13	0.9	0.9
400600	6 : AT - Promotion od. Habilitation (Hochschule, Universität, Fachhochschule)		28	1.9	2.0
	Sum		1510	99.9	100.0
	Valid Cases		1388		

v356 - father/mother employed at respondents age of 14 (Q128)

Q128

When you were 14, was your father [mother] employed, self employed or not?

<IF CODE 3, -1, -2, -3: GO TO Q132>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes, employed
- 2 yes, self employed
- 3 without employment

v356, weighted by weight

	0 , 0				
Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	44	2.9	
-2	na	М	20	1.3	
-1	dk	М	30	2.0	
1	yes employed		1060	70.2	74.9
2	yes self employed	ı	286	18.9	20.2
3	no		69	4.6	4.9
	Sum		1510	99.9	100.0
	Valid Cases		1416		

v357ISCO - kind of job father/mother - ISCO88 code (Q129a)

Q129

What was the name or title of your father's [mother's] job?

Write in: ...

Q129a

In his [her] job, what kind of work did your father [mother] most of the time?

<CODING INSTRUCTION: CODE ACCORDING TO ISCO88 (4 DIGITS) AFTER INTERVIEW ON THE BASIS OF Q129 AND Q129A> Code:

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 110 [armed forces]
- 1000 legislators, senior officials & managers
- 1100 legislators & senior officials
- 1110 legislators
- 1120 senior government officials
- 1130 traditional chiefs & heads of villages
- 1140 senior officials special-interest organisations
- 1141 senior officials political-party organisations
- 1142 senior officials economic-interest organisations
- 1143 senior officials special-interest organisations
- 1200 corporate managers [large enterprise]
- 1210 directors & chief executives [large enterprise]
- 1220 department managers [large enterprise][production-operation]
- 1221 department managers agriculture, hunting, forestry & fishIng
- 1222 department managers manufacturing
- 1223 department managers construction
- 1224 department managers wholesale & retail trade
- 1225 department managers restaurants & hotels
- 1226 department managers transport, storage & communications
- 1227 department managers business services
- 1228 department managers personal care, cleaning etc
- 1229 operations department managers nec
- 1230 other department managers [large enterprise]
- 1231 finance & administration department managers
- 1232 personnel & industrial relations department managers
- 1233 sales & marketing department managers
- 1234 advertising & public relations department managers
- 1235 supply & distribution department managers
- 1236 computing services department managers
- 1237 research & development department managers
- 1239 other department managers nec
- 1240 [office manager]
- 1250 [military officers]

1251	[high military officers]
1252	[lower military officers]
1300	general managers [small enterprise]
1310	general managers [small enterprise]
1311	general managers agriculture, hunting forestry & fishing
1312	general managers manufacturing
1313	general managers construction
1314	general managers wholesale & retail trade
1315	general managers restaurants & hotels
1316	general managers transport, storage & communications
1317	general managers business services
1318	general managers personal care, cleaning etc services
1319	general managers nec
2000	professionals
2100	physical, mathematical & engineering science professionals
2110	physicists, chemists & related professionals
2111	physicists & astronomers
2112	meteorologists
2113	chemists
2114	geologists & geophysicists
2120	mathematicians, statisticians etc professionals
2121	mathematicians etc professionals
2122	statisticians
2130	computing professionals
2131	computer systems designers & analysts
2132	computer programmers
2139	computing professionals nec
2140	architects, engineers etc professionals
2141	architects town & traffic planners
2142	civil engineers
2143	electrical engineers
2144	electronics & telecommunications engineers
2145	mechanical engineers
2146	chemical engineers
2147	mining engineers metallurgists etc professionals
2148	cartographers & surveyors
2149	architects engineers etc professionals nec
2200	life science & health professionals
2210	life science professionals
2211	biologists, botanists zoologists etc professionals
2212	pharmacologists, pathologists etc professionals
2213	agronomists etc professionals
2220	health professionals (except nursing)
2221	medical doctors
2222	dentists
2223	veterinarians
2224	pharmacists

health professionals except nursing necnursing & midwifery professionals

2300	teaching professionals
2310	higher education teaching professionals
2320	secondary education teaching professionals
2321	[second academ teacher]
2322	[second voc teacher]
2323	[middle school teacher]
2330	primary & pre-primary education teaching professionals
2331	primary education teaching professionals
2332	pre-primary education teaching professionals
2340	special education teaching professionals
2350	other teaching professionals
2351	education methods specialists
2352	school inspectors
2359	other teaching professionals nec
2400	other professionals
2410	business professionals
2411	accountants
2412	personnel & careers professionals
2419	business professionals nec
2420	legal professionals
2421	lawyers
2422	judges
2429	legal professionals nec
2430	archivists, librarians etc information professionals
2431	archivists & curators
2432	librarians etc information professionals
2440	social science etc professionals
2441	economists
2442	sociologists anthropologists etc professionals
2443	philosophers, historians & political scientists
2444	philologists translators & interpreters
2445	psychologists
2446	social work professionals
2450	writers & creative or performing artists
2451	authors journalists & other writers
2452	sculptors, painters etc artists
2453	composers musicians & singers
2454	choreographers & dancers
2455	film, stage etc actors & directors
2460	religious professionals
3000	technicians and associated professionals
3100	physical & engineering science associate professionals
3110	physical & engineering science technicians
3111	chemical & physical science technicians
3112	civil engineering technicians
3113	electrical engineering technicians
3114	electronics & telecommunications engineering technicians
3115	mechanical engineering technicians

3116 chemical engineering technicians

3117	mining & metallurgical technicians
3118	draughtspersons
3119	physical & engineering science technicians nec
3120	computer associate professionals
3121	computer assistants
3122	computer equipment operators
3123	industrial robot controllers
3130	optical & electronic equipment operators
3131	photographers & elctr equipment operators
3132	broadcasting & telecommunications equipment operators
3133	medical equipment operators
3139	optical & electronic equipment operators nec
3140	ship & aircraft controllers & technicians
3141	ships engineers
3142	ships deck officers & pilots
3143	aircraft pilots etc associate professionals
3144	air traffic controllers
3145	air traffic safety technicians
3150	safety & ouality inspectors
3151	building & fire inspectors
3152	safety, health & quality inspectors
3200	life science & health associate professionals
3210	life science technicians etc associate professionals
3211	life science technicians
3212	agronomy & forestry technicians
3213	farming & forestry advisers
3220	modern health associate professionals except nursing
3221	medical assistants
3222	sanitarians
3223	dieticians & nutritionists
3224	optometrists & opticians
3225	dental assistants
3226	physiotherapists etc associate professionals
3227	veterinary assistants
3228	pharmaceutical assistants
3229	modern health associate professionals except nursing nec
3230	nursing & midwifery associate professionals
3231	nursing associate professionals
3232	midwifery associate professionals
3240	traditional medicine practitioners & faith healers
3241	traditional medicine practitioners
3242	faith healers
3300	teaching associate professionals
3310	primary education teaching associate professionals
3320	pre-primary education teaching associate professionals
3330	special education teaching associate professionals
3340	other teaching associate professionals
3400	other associate professionals

3410 finance & sales associate professionals

3411	securities & finance dealers & brokers
3412	insurance representatives
3413	estate agents
3414	travel consultants & organisers
3415	technical & commercial sales representatives
3416	buyers
3417	appraisers, valuers & auctioneers
3419	finance & sales associate professionals nec
3420	business services agents and trade brokers
3421	trade brokers
3422	clearing & forwarding agents
3423	employment agents & labour contractors
3429	business services agents & trade brokers nec
3430	administrative associate professionals
3431	administrative secretaries etc associate professionals
3432	legal etc business associate professionals
3433	bookkeepers
3434	statistical, mathematical etc associate professionals
3439	administrative associate professionals nec
3440	customs, tax etc government associate professionals
3441	customs & border inspectors
3442	government tax & excise officials
3443	government social benefits officials
3444	government licensing officials
3449	customs tax etc government associate professionals nec
3449 3450	customs tax etc government associate professionals nec [police and army officers]
	· ·
3450	[police and army officers]
3450 3451	[police and army officers] [police inspectors-detectives]
3450 3451 3452	[police and army officers] [police inspectors-detectives] [armed forces non-commissioned officers + army nfs]
3450 3451 3452 3460	[police and army officers] [police inspectors-detectives] [armed forces non-commissioned officers + army nfs] social work associate professionals
3450 3451 3452 3460 3470 3471	[police and army officers] [police inspectors-detectives] [armed forces non-commissioned officers + army nfs] social work associate professionals artistic, entertainment & sports associate professionals
3450 3451 3452 3460 3470 3471	[police and army officers] [police inspectors-detectives] [armed forces non-commissioned officers + army nfs] social work associate professionals artistic, entertainment & sports associate professionals decorators & commercial designers
3450 3451 3452 3460 3470 3471 3472	[police and army officers] [police inspectors-detectives] [armed forces non-commissioned officers + army nfs] social work associate professionals artistic, entertainment & sports associate professionals decorators & commercial designers radio, television & other announcers
3450 3451 3452 3460 3470 3471 3472 3473	[police and army officers] [police inspectors-detectives] [armed forces non-commissioned officers + army nfs] social work associate professionals artistic, entertainment & sports associate professionals decorators & commercial designers radio, television & other announcers street night-club etc musicians singers & dancers
3450 3451 3452 3460 3470 3471 3472 3473 3474	[police and army officers] [police inspectors-detectives] [armed forces non-commissioned officers + army nfs] social work associate professionals artistic, entertainment & sports associate professionals decorators & commercial designers radio, television & other announcers street night-club etc musicians singers & dancers clowns magicians acrobats etc associate professionals
3450 3451 3452 3460 3470 3471 3472 3473 3474 3475	[police and army officers] [police inspectors-detectives] [armed forces non-commissioned officers + army nfs] social work associate professionals artistic, entertainment & sports associate professionals decorators & commercial designers radio, television & other announcers street night-club etc musicians singers & dancers clowns magicians acrobats etc associate professionals athletes sportspersons etc associate professionals
3450 3451 3452 3460 3470 3471 3472 3473 3474 3475 3480	[police and army officers] [police inspectors-detectives] [armed forces non-commissioned officers + army nfs] social work associate professionals artistic, entertainment & sports associate professionals decorators & commercial designers radio, television & other announcers street night-club etc musicians singers & dancers clowns magicians acrobats etc associate professionals athletes sportspersons etc associate professionals religious associate professionals
3450 3451 3452 3460 3470 3471 3472 3473 3474 3475 3480 4000	[police and army officers] [police inspectors-detectives] [armed forces non-commissioned officers + army nfs] social work associate professionals artistic, entertainment & sports associate professionals decorators & commercial designers radio, television & other announcers street night-club etc musicians singers & dancers clowns magicians acrobats etc associate professionals athletes sportspersons etc associate professionals religious associate professionals clerks
3450 3451 3452 3460 3470 3471 3472 3473 3474 3475 3480 4000 4100	[police and army officers] [police inspectors-detectives] [armed forces non-commissioned officers + army nfs] social work associate professionals artistic, entertainment & sports associate professionals decorators & commercial designers radio, television & other announcers street night-club etc musicians singers & dancers clowns magicians acrobats etc associate professionals athletes sportspersons etc associate professionals religious associate professionals clerks office clerks
3450 3451 3452 3460 3470 3471 3472 3473 3474 3475 3480 4000 4100 4110	[police and army officers] [police inspectors-detectives] [armed forces non-commissioned officers + army nfs] social work associate professionals artistic, entertainment & sports associate professionals decorators & commercial designers radio, television & other announcers street night-club etc musicians singers & dancers clowns magicians acrobats etc associate professionals athletes sportspersons etc associate professionals religious associate professionals clerks office clerks secretaries & keyboard-operating clerks
3450 3451 3452 3460 3470 3471 3472 3473 3474 3475 3480 4000 4100 4110 4111	[police and army officers] [police inspectors-detectives] [armed forces non-commissioned officers + army nfs] social work associate professionals artistic, entertainment & sports associate professionals decorators & commercial designers radio, television & other announcers street night-club etc musicians singers & dancers clowns magicians acrobats etc associate professionals athletes sportspersons etc associate professionals religious associate professionals clerks office clerks secretaries & keyboard-operating clerks stenographers & typists
3450 3451 3452 3460 3470 3471 3472 3473 3474 3475 3480 4000 4100 4110 4111 4112	[police and army officers] [police inspectors-detectives] [armed forces non-commissioned officers + army nfs] social work associate professionals artistic, entertainment & sports associate professionals decorators & commercial designers radio, television & other announcers street night-club etc musicians singers & dancers clowns magicians acrobats etc associate professionals athletes sportspersons etc associate professionals religious associate professionals clerks office clerks secretaries & keyboard-operating clerks stenographers & typists word-processor etc operators
3450 3451 3452 3460 3470 3471 3472 3473 3474 3475 3480 4000 4110 4111 4112 4113	[police and army officers] [police inspectors-detectives] [armed forces non-commissioned officers + army nfs] social work associate professionals artistic, entertainment & sports associate professionals decorators & commercial designers radio, television & other announcers street night-club etc musicians singers & dancers clowns magicians acrobats etc associate professionals athletes sportspersons etc associate professionals religious associate professionals clerks office clerks secretaries & keyboard-operating clerks stenographers & typists word-processor etc operators data entry operators
3450 3451 3452 3460 3470 3471 3472 3473 3474 3475 3480 4000 4110 4111 4111 4112 4113 4114	[police and army officers] [police inspectors-detectives] [armed forces non-commissioned officers + army nfs] social work associate professionals artistic, entertainment & sports associate professionals decorators & commercial designers radio, television & other announcers street night-club etc musicians singers & dancers clowns magicians acrobats etc associate professionals athletes sportspersons etc associate professionals religious associate professionals clerks office clerks secretaries & keyboard-operating clerks stenographers & typists word-processor etc operators data entry operators calculating-machine operators
3450 3451 3452 3460 3470 3471 3472 3473 3474 3475 3480 4000 4110 4111 4112 4113 4114 4115	[police and army officers] [police inspectors-detectives] [armed forces non-commissioned officers + army nfs] social work associate professionals artistic, entertainment & sports associate professionals decorators & commercial designers radio, television & other announcers street night-club etc musicians singers & dancers clowns magicians acrobats etc associate professionals athletes sportspersons etc associate professionals religious associate professionals clerks office clerks secretaries & keyboard-operating clerks stenographers & typists word-processor etc operators data entry operators calculating-machine operators secretaries
3450 3451 3452 3460 3470 3471 3472 3473 3474 3475 3480 4000 4110 4111 4112 4113 4114 4115 4120	[police and army officers] [police inspectors-detectives] [armed forces non-commissioned officers + army nfs] social work associate professionals artistic, entertainment & sports associate professionals decorators & commercial designers radio, television & other announcers street night-club etc musicians singers & dancers clowns magicians acrobats etc associate professionals athletes sportspersons etc associate professionals religious associate professionals clerks office clerks secretaries & keyboard-operating clerks stenographers & typists word-processor etc operators data entry operators calculating-machine operators secretaries numerical clerks

4131 stock clerks

- 4132 production clerks
- 4133 transport clerks
- 4140 library, mail etc clerks
- 4141 library & filing clerks
- 4142 mail carriers & sorting clerks
- 4143 coding proof-reading etc clerks
- 4144 scribes etc workers
- 4190 other office clerks
- 4200 customer services clerks
- 4210 cashiers, tellers etc clerks
- 4211 cashiers & ticket clerks
- 4212 tellers & other counter clerks
- 4213 bookmakers & croupiers
- 4214 pawnbrokers & money-lenders
- 4215 debt-collectors etc workers
- 4220 client information clerks
- 4221 travel agency etc clerks
- 4222 receptionists & information clerks
- 4223 telephone switchboard operators
- 5000 service workers & shop & market sales workers
- 5100 personal & protective services workers
- 5110 travel attendants etc
- 5111 travel attendants & travel stewards
- 5112 transport conductors
- 5113 travel, museum guides
- 5120 housekeeping & restaurant services workers
- 5121 housekeepers etc workers
- 5122 cooks
- 5123 waiters waitresses & bartenders
- 5130 personal care etc work
- 5131 child-care workers
- 5132 institution-based personal care workers
- 5133 home based personal care workers
- 5139 personal care etc workers nec
- 5140 other personal services workers
- 5141 hairdressers barbers beauticians etc workers
- 5142 companions & valets
- 5143 undertakers & embalmers
- 5149 other personal services workers nec
- 5150 astrologers, fortune-tellers etc workers
- 5151 astrologers etc workers
- 5152 fortune-tellers, palmists etc workers
- 5160 protective services workers
- 5161 fire-fighters
- 5162 police officers
- 5163 prison guards
- 5164 [army: enlisted men soldiers]
- 5169 protective services workers nec
- 5200 models, salespersons & demonstrators

EVS 2008 - Austria GESIS StudyNo 4754 v1 1 0 http://dx.doi.u g/doi:10.4232/1.10152

GES	IS StudyNo 4754, v1.1.0, http://dx.doi.org/d
5210	fashion & other models
5220	shop salespersons & demonstrators
5230	stall & market salespersons
6000	skilled agricultural & fishery workers
6100	market-oriented skilled agricultural & fishery workers
6110	market gardeners & cropgrowers
6111	field crop & vegetable growers
6112	tree & shrub crop growers
6113	gardeners, horticultural & nursery growers
6114	mixed-crop growers
6120	market-oriented animal producers etc workers
6121	dairy & livestock producers
6122	poultry producers
6123	apiarists & sericulturists
6124	mixed -animal producers
6129	market-oriented animal producers etc workers nec
6130	market-oriented crop & animal producers
6131	[mixed farmer]
6132	[farm foreman]
6133	[farmer nfs]
6134	[skilled farm worker nfs]
6140	forestry etc workers
6141	forestry workers & loggers
6142	charcoal burners etc workers
6150	fishery workers, hunters & trappers
6151	aquatic-life cultivation workers
6152	inland & coastal waters fishery workers
6153	deep-sea fishery workers
6154	hunters & trappers
6200	subsistence agricultural & fishery workers
6210	subsistence agricultural & fishery workers
7000	craft etc trades workers
7100	extraction & building trades workers
7110	miners, shotfirers, stone cutters & carvers
7111	miners & quarry workers
7112	shotfirers & blasters
7113	stone splitters cutters & carvers
7120	[builders] & ar trades workers
7121	builders traditional materials
7122	bricklayers & stonemasons
7123	concrete placers concrete finishers etc workers
7124	carpenters & joiners
7129	[builders] ar trades workers nec
7130	building finishers etc trades workers
7131	roofers
7132	floor layers & tile setters

7133 plasterers

7135 glaziers

7134 insulation workers

- 7136 plumbers & pipe fitters
- 7137 building etc electricians
- 7140 painters, building structure cleaners etc trades workers
- 7141 painters etc workers
- 7142 varnishers etc painters
- 7143 building structure cleaners
- 7200 metal, machinery etc trades workers
- 7210 metal moulders, welders, sheetmetal workers structural metal
- 7211 metal moulders & coremakers
- 7212 welders & flamecutters
- 7213 sheet-metal workers
- 7214 structural-metal preparers & erectors
- 7215 riggers & cable splicers
- 7216 underwater workers
- 7220 blacksmiths, tool-makers etc trades workers
- 7221 blacksmiths hammer-smiths & forgingpress workers
- 7222 tool-makers etc workers
- 7223 machine-tool setters & setter-operators
- 7224 metal wheel-grinders polishers & tool sharpeners
- 7230 machinery mechanics & fitters
- 7231 motor vehicle mechanics & fitters
- 7232 aircraft engine mechanics & fitters
- 7233 agricultural- or industrial-machinery mechanics & fitters
- 7234 [unskl garage worker, oiler]
- 7240 electrical & electronic equipment mechanics & fitters
- 7241 electrical mechanics & fitters
- 7242 electronics fitters
- 7243 electronics mechanics & servicers
- 7244 telegraph & telephone installers & servicers
- 7245 electrical line installers repairers & cable jointers
- 7300 precision, handicraft, printing etc trades workers
- 7310 precision workers in metal etc materials
- 7311 precision-instrument makers & repairers
- 7312 musical-instrument makers & tuners
- 7313 jewellery & precious-metal workers
- 7320 potters, glass-makers etc trades workers
- 7321 abrasive wheel formers potters etc workers
- 7322 glass-makers cutters grinders & finishers
- 7323 glass engravers & etchers
- 7324 glass ceramics etc decorative painters
- 7330 handicraft workers in wood, textile, leather etc
- 7331 handicraft workers in wood etc materials
- 7332 handicraft workers in textile leather etc materials
- 7340 printing etc trades workers
- 7341 compositors typesetters etc workers
- 7342 stereotypers & electrotypers
- 7343 printing engravers & etchers
- 7344 photographic etc workers
- 7345 bookbinders etc workers

- 7346 silk-screen, block & textile printers
- 7400 other craft etc trades workers
- 7410 food processing etc trades workers
- 7411 butchers fishmongers etc food preparers
- 7412 bakers-pastry cooks & confectionery makers
- 7413 dairy-products makers
- 7414 fruit, vegetable etc preservers
- 7415 food & beverage tasters & graders
- 7416 tobacco preparers & tobacco products makers
- 7420 wood treaters, cabinet-makers etc trades workers
- 7421 wood treaters
- 7422 cabinet-makers etc workers
- 7423 woodworking-machine setters & setteroperators
- 7424 basketry weavers brush makers etc workers
- 7430 textile, garment etc trades workers
- 7431 fibre preparers
- 7432 weavers knitters etc workers
- 7433 tailors dressmakers & hatters
- 7434 furriers etc workers
- 7435 textile leather etc pattern-makers & cutters
- 7436 sewers embroiderers etc workers
- 7437 upholsterers etc workers
- 7440 pelt, leather & shoemaking trades workers
- 7441 pelt dressers tanners & fellmongers
- 7442 shoe-makers etc workers
- 7500 [skilled manual worker nfs]
- 7510 [non farm foremen nfs]
- 7520 [skilled manual nfs]
- 7530 [apprentice skilled work nfs]
- 8000 plant & machine operators & assemblers
- 8100 stationary-plant etc operators
- 8110 mining- & mineral-processing plant operators
- 8111 mining-plant operators
- 8112 mineral-ore- & stone-processing-plant operators
- 8113 well drillers & borers etc workers
- 8120 metal-processing-plant operators
- 8121 ore & metal furnace operators
- 8122 metal melters casters & rolling-mill operators
- 8123 metal-heat-treating-plant operators
- 8124 metal drawers & extruders
- 8130 glass, ceramics etc plant operators
- 8131 glass & ceramics kiln etc machine operators
- 8139 glass, ceramics etc plant operators nec
- 8140 wood-processing- & papermaking-plant operators
- 8141 wood-processing-plant operators
- 8142 paper-pulp plant operators
- 8143 papermaking-plant operators
- 8150 chemical-processing-plant operators
- 8151 crushing- grinding- & chemical-mixing machinery operators

8152 chemical-heat-treating-plant operators 8153 chemical-filtering- & separating-equipment operators 8154 chemical-still & reactor operators petroleum- & natural-gas-refining-plant operators 8155 8159 chemical-processing-plant operators nec power-production etc plant operators 8161 power-production plant operators 8162 steam-engine & boiler operators 8163 incinerator water-treatment etc plant operators 8170 automated-assembly-line & industrial-robot operators 8171 automated-assembly-line operators 8172 industrial-robot operators 8200 machine operators & assemblers 8210 metal- & mineral-products machine operators 8211 machine-tool operators 8212 cement & other mineral products machine operators 8220 chemical-products machine operators 8221 pharmaceutical- & toiletry-products machine operators 8222 ammunition- & explosive-products machine operators 8223 metal finishing- plating- & coating-machine operators 8224 photographic-products machine operators 8229 chemical-products machine operators nec 8230 rubber- & plastic-products machine operators 8231 rubber-products machine operators plastic-products machine operators 8232 wood-products machine operators 8240 8250 printing- binding- & paper-products machine operators 8251 printing-machine operators 8252 bookbinding-machine operators 8253 paper-products machine operators 8260 textile- fur- & leather-products machine operators 8261 fibre-preparing-, spinning- & windingmachine operators 8262 weaving- & knitting-machine operators 8263 sewing-machine operators 8264 bleaching- dyeing- & cleaning-machine operators 8265 fur- & leather-preparing-machine operators 8266 shoemaking- etc machine operators 8269 textile- fur- & leather-products machine operators nec 8270 food etc products machine operators 8271 meat- & fish-processing-machine operators 8272 dairy-products machine operators 8273 grain- & spice-milling-machine operators 8274 baked-goods cereal & chocolate-products machine operators 8275 fruit- vegetable- & nut-processing-machine operators 8276 sugar production machine operators 8277 tea- coffee- & cocoa-processing-machine operators 8278 brewers- wine & other beverage machine operators

8279 tobacco production machine operators

8280 assemblers

8281	mechanical-machinery assemblers
8282	electrical-equipment assemblers
8283	electronic-equipment assemblers
8284	metal- rubber- & plastic-products assemblers
8285	wood etc products assemblers
8286	paperboard textile etc products assemblers
8290	other machine operators & assemblers
8300	drivers & mobile-plant operators
8310	locomotive-engine drivers etc workers
8311	locomotive-engine drivers
8312	railway brakers signallers & shunters
8320	motor-vehicle drivers
8321	motor-cycle drivers
8322	car taxi & van drivers
8323	bus & tram drivers
8324	heavy truck & lorry drivers
8330	agricultural & other mobile-plant operators
8331	motorised farm & forestry plant operators
8332	earth-moving- etc plant operators
8333	crane hoist etc plant operators
8334	lifting-truck operators
8340	ships deck crews etc workers
8400	[semiskld worker nfs]
9000	elementary occupations
9100	sales & services elementary occupations
9110	street vendors etc workers
9111	street food vendors
9112	street vendors non-food products
9113	door-to-door & telephone salespersons
9120	street services elementary occupations
9130	domestic etc helpers cleaners & launderers
9131	domestic helpers & cleaners
9132	helpers & cleaners in establishments
9133	hand-launderers & pressers
9140	building caretakers, window etc cleaners
9141	building caretakers
9142	vehicle, window etc cleaners
9150	messengers, porters, doorkeepers etc workers
9151	messengers package & luggage porters & deliverers
9152	doorkeepers watchpersons etc workers
9153	vending-machine money collectors meter readers etc worker
9160	garbage collectors etc labourers
9161	garbage collectors
9162	sweepers etc labourers
9200	agricultural, fishery etc labourers
9210	agricultural, fishery etc labourers
9211	farm-hands & labourers
9212	forestry labourers
	•

9213 fishery hunting & trapping labourers

- 9300 labourers in mining, construction, manufacturing & transport
- 9310 mining & construction labourers
- 9311 mining & quarrying labourers
- 9312 construction & maintenance labourers: roads dams etc
- 9313 building construction labourers
- 9320 manufacturing labourers
- 9321 assembling labourers
- 9322 hand packers & other manufacturing labourers
- 9330 transport labourers & freight handlers
- 9331 hand or pedal vehicle drivers
- 9332 drivers of animal-drawn vehicles & machinery
- 9333 freight handlers

Note:

Standardized variable on the basis of two open-ended questions on occupation and kind of activity (Q129 and Q129a) post-coded according to 4-digit ISCO88.

Data for original string variable is not available in this dataset.

v357SIOPS - occupational status father/mother - SIOPS (Q129a)

COMPUTED VARIABLE - SIOPS: Standard International Occupational Prestige Scale (Treiman, 1977)

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk

Note:

Source classification: Ganzeboom, H. B. G. & Treiman, D. J. (1996), Internationally Comparable Measures of Occupational Status for the 1988 International Standard Classification of Occupations. Social Science Research, 25, 201-239.

Source syntax to compute classification: http://home.fsw.vu.nl/hbg.ganzeboom/pisa/index.htm

v357ISEI - occupational status father/mother - ISEI (Q129a)

COMPUTED VARIABLE - ISEI: International Socio-Economic Index of Occupational Status (Ganzeboom et al., 1992)

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk

Note:

Source classification: Ganzeboom, H. B. G. & Treiman, D. J. (1996), Internationally Comparable Measures of Occupational Status for the 1988 International Standard Classification of Occupations. Social Science Research, 25, 201-239.

Source syntax to compute classification: http://home.fsw.vu.nl/hbg.ganzeboom/pisa/index.htm

v357ESeC - occupational status father/mother - European ESeC (Q129a)

COMPUTED VARIABLE - European Socio-economic Classification (ESeC)

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 Large employers, higher managers/professionals
- 2 Lower managers/professionals, higher supervisory/technicians
- 3 Intermediate occupations
- 4 Small employers and self-employed (non-agriculture)
- 5 Small employers and self-employed (agriculture)
- 6 Lower supervisors and technicians
- 7 Lower sales and service
- 8 Lower technical
- 9 Routine

Note:

Source: http://www.iser.essex.ac.uk/research/esec/matrices-and-syntax

Date: 27-09-2006 (last update syntax).

v357ESeC, weighted by weight

Value Label		Missing	Count	Percent	Valid Percent
-3 nap		M	164	10.9	
-2 na		M	34	2.3	
-1 dk		M	18	1.2	
1 Large employers	, higher mgrs/professionals		75	5.0	5.8
2 Lower mgrs/profe	essionals, higher supervisory/technicians		162	10.7	12.5
3 Intermediate occi	upations		90	6.0	7.0
4 Small employers	and self-employed (non-agriculture)		98	6.5	7.6
5 Small employers	and self-employed (agriculture)		139	9.2	10.7
6 Lower supervisor	rs and technicians		98	6.5	7.6
7 Lower sales and	service		78	5.2	6.0
8 Lower technical			312	20.7	24.1
9 Routine			242	16.0	18.7
Sum			1510	100.0	100.0
Valid Cases			1294		

v357egp - occupational status father/mother - egp11 (Q129a)

COMPUTED VARIABLE - EGP: EGP class scheme (Erikson/Goldthorpe/Portocarero, 1979)

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 I: Higher Controllers
- 2 II: Lower Controllers
- 3 IIIa: Routine Nonmanual
- 4 IIIb: Lower Sales-Service
- 5 IVa: Self-employed with employees
- 6 IVb: Self-employed with no employees
- 7 V: Manual Supervisors
- 8 VI: Skilled Worker
- 9 VIIa: Unskilled Worker
- 10 VIIb: Farm Labor
- 11 IVc: Self-employed Farmer

Note:

Source classification: Ganzeboom, H. B. G. & Treiman, D. J. (1996), Internationally Comparable Measures of Occupational Status for the 1988 International Standard Classification of Occupations. Social Science Research, 25, 201-239.

 $Source\ syntax\ to\ compute\ classification:\ http://home.fsw.vu.nl/hbg.ganzeboom/pisa/index.htm$

v357egp, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	164	10.9	
-2	na	М	34	2.3	
-1	dk	М	18	1.2	
1	I :Higher Controllers		122	8.1	9.4
2	II :Lower Controllers		133	8.8	10.3
3	Illa:Routine Nonmanual		63	4.2	4.9
4	IIIb:Lower Sales-Service		92	6.1	7.1
5	IVa:Selfempl with empl		78	5.2	6.0
6	IVb:Selfempl no empl		20	1.3	1.5
7	V :Manual Supervisors		50	3.3	3.9
8	VI :Skilled Worker		253	16.8	19.6
9	VIIa:Unskilled Worker		294	19.5	22.7
10	VIIb:Farm Labor		42	2.8	3.2
11	IVc:Selfempl Farmer		147	9.7	11.4
	Sum		1510	100.0	100.0
	Valid Cases		1294		

v358 - father/mother had how many employees (Q130)

Q130

<FOR SELF EMPLOYED FATHERS/MOTHERS (Q128=2)>

How many employees did your father [mother] have?

<SKIP TO 132>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 none
- 2 1-9
- 3 10-24
- 4 25 or more

v358, weighted by weight

,	- 3 ,	5			
Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	1224	81.1	
-2	na	М	2	0.1	
-1	dk	М	10	0.7	
1	none		122	8.1	44.5
2	1-9		125	8.3	45.6
3	10-24		15	1.0	5.5
4	25 or more		12	0.8	4.4
	Sum		1510	100.0	100.0
	Valid Cases		274		

v359 - did father/mother supervise someone (Q131)

Q131

<FOR EMPLOYED FATHERS/MOTHERS (Q128=1) >

Did your father [mother] have any responsibility for supervising the work of other employees?

<IF CODE 2, -1, -3: GO TO Q132>

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 yes
- 2 no

v359, weighted by weight

		_			
Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	450	29.8	
-2	na	М	8	0.5	
-1	dk	М	55	3.6	
1	yes		250	16.6	25.1
2	no		747	49.5	74.9
	Sum		1510	100.0	100.0
	Valid Cases		997		

v359a - how many people did she/he supervise (Q131a)

Q131a

How many other employees did your father [mother] supervise?

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 1-9
- 2 10-24
- 3 25 or more

v359a, weighted by weight

,	- 5 ,	- 0			
Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	1260	83.4	
-2	na	М	4	0.3	
-1	dk	М	20	1.3	
1	1-9		144	9.5	63.4
2	10-24		38	2.5	16.7
3	25 or more		45	3.0	19.8
	Sum		1510	100.1	100.0
	Valid Cases		227		

v360 - mother liked to read books (Q132A)

Q132

<SHOW CARD 132>

When you think about your parents when you were about 14 years old, could you say whether these statements correctly describe your parents?

<DO NOT ASK A, B, C IF RESPONDENT LIVED WITH FATHER ONLY (SEE Q126=2)>

<DO NOT ASK E, F, G IF RESPONDENT LIVED WITH MOTHER ONLY (SEE Q126=3)>

Q132.A My mother liked to read books

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 Yes
- 2 To some extent
- 3 A little bit
- 4 No

v360, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	64	4.2	
-2	na	М	8	0.5	
-1	dk	М	60	4.0	
1	yes		389	25.8	28.2
2	to some extent		380	25.2	27.6
3	a little bit		265	17.5	19.2
4	no		343	22.7	24.9
	Sum		1510	99.9	100.0
	Valid Cases		1377		

v361 - discussed politics with mother (Q132B)

Q132

<SHOW CARD 132>

When you think about your parents when you were about 14 years old, could you say whether these statements correctly describe your parents?

<DO NOT ASK A, B, C IF RESPONDENT LIVED WITH FATHER ONLY (SEE Q126=2)>

<DO NOT ASK E, F, G IF RESPONDENT LIVED WITH MOTHER ONLY (SEE Q126=3)>

Q132.B I discussed politics at home with my mother

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 Yes
- 2 To some extent
- 3 A little bit
- 4 No

v361, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	64	4.2	
-2	na	М	6	0.4	
-1	dk	М	41	2.7	
1	yes		65	4.3	4.6
2	to some extent		202	13.4	14.4
3	a little bit		292	19.3	20.9
4	no		839	55.6	60.0
	Sum		1510	99.9	100.0
	Valid Cases		1398		

v362 - mother liked to follow the news (Q132C)

Q132

<SHOW CARD 132>

When you think about your parents when you were about 14 years old, could you say whether these statements correctly describe your parents?

<DO NOT ASK A, B, C IF RESPONDENT LIVED WITH FATHER ONLY (SEE Q126=2)>

<DO NOT ASK E, F, G IF RESPONDENT LIVED WITH MOTHER ONLY (SEE Q126=3)>

Q132.C My mother liked to follow the news

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 Yes
- 2 To some extent
- 3 A little bit
- 4 No

v362, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	64	4.2	
-2	na	М	8	0.5	
-1	dk	М	48	3.2	
1	yes		390	25.8	28.1
2	to some extent		455	30.1	32.7
3	a little bit		346	22.9	24.9
4	no		199	13.2	14.3
	Sum		1510	100.0	100.0
	Valid Cases		1390		

v363 - parent(s) had problems making ends meet (Q132D)

Q132

<SHOW CARD 132>

When you think about your parents when you were about 14 years old, could you say whether these statements correctly describe your parents?

<DO NOT ASK A, B, C IF RESPONDENT LIVED WITH FATHER ONLY (SEE Q126=2)>

<DO NOT ASK E, F, G IF RESPONDENT LIVED WITH MOTHER ONLY (SEE Q126=3)>

Q132.D My parent(s) had problems making ends meet

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 Yes
- 2 To some extent
- 3 A little bit
- 4 No

v363, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-5	other missing	М	3	0.2	
-3	nap	М	41	2.7	
-2	na	М	6	0.4	
-1	dk	М	86	5.7	
1	yes		201	13.3	14.6
2	to some extent		342	22.6	24.9
3	a little bit		369	24.4	26.9
4	no		462	30.6	33.6
	Sum		1510	100.0	100.0
	Valid Cases		1374		

v364 - father liked to read books (Q132E)

Q132

<SHOW CARD 132>

When you think about your parents when you were about 14 years old, could you say whether these statements correctly describe your parents?

<DO NOT ASK A, B, C IF RESPONDENT LIVED WITH FATHER ONLY (SEE Q126=2)>

<DO NOT ASK E, F, G IF RESPONDENT LIVED WITH MOTHER ONLY (SEE Q126=3)>

Q132.E My father liked to read books

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 Yes
- 2 To some extent
- 3 A little bit
- 4 No

v364, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	199	13.2	
-2	na	М	8	0.5	
-1	dk	М	62	4.1	
1	yes		241	16.0	19.4
2	to some extent		232	15.4	18.7
3	a little bit		296	19.6	23.9
4	no		471	31.2	38.0
	Sum		1510	99.9	100.0
	Valid Cases		1240		

v365 - discussed politics with father (Q132F)

Q132

<SHOW CARD 132>

When you think about your parents when you were about 14 years old, could you say whether these statements correctly describe your parents?

<DO NOT ASK A, B, C IF RESPONDENT LIVED WITH FATHER ONLY (SEE Q126=2)>

<DO NOT ASK E, F, G IF RESPONDENT LIVED WITH MOTHER ONLY (SEE Q126=3)>

Q132.F I discussed politics at home with my father

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 Yes
- 2 To some extent
- 3 A little bit
- 4 No

v365, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	199	13.2	
-2	na	М	7	0.5	
-1	dk	М	45	3.0	
1	yes		124	8.2	9.9
2	to some extent		280	18.5	22.3
3	a little bit		293	19.4	23.3
4	no		561	37.2	44.6
	Sum		1510	99.9	100.0
	Valid Cases		1258		

v366 - father liked to follow the news (Q132G)

Q132

<SHOW CARD 132>

When you think about your parents when you were about 14 years old, could you say whether these statements correctly describe your parents?

<DO NOT ASK A, B, C IF RESPONDENT LIVED WITH FATHER ONLY (SEE Q126=2)>

<DO NOT ASK E, F, G IF RESPONDENT LIVED WITH MOTHER ONLY (SEE Q126=3)>

Q132.G My father liked to follow the news

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 Yes
- 2 To some extent
- 3 A little bit
- 4 No

v366, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-3	nap	М	199	13.2	
-2	na	М	8	0.5	
-1	dk	М	39	2.6	
1	yes		639	42.3	50.6
2	to some extent		425	28.1	33.6
3	a little bit		139	9.2	11.0
4	no		61	4.0	4.8
	Sum		1510	100.0	100.0
	Valid Cases		1264		

v367 - parent(s) had problems replacing broken things (Q132H)

Q132

<SHOW CARD 132>

When you think about your parents when you were about 14 years old, could you say whether these statements correctly describe your parents?

<DO NOT ASK A, B, C IF RESPONDENT LIVED WITH FATHER ONLY (SEE Q126=2)>

<DO NOT ASK E, F, G IF RESPONDENT LIVED WITH MOTHER ONLY (SEE Q126=3)>

Q132.H My parent(s) had problems replacing broken things

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 Yes
- 2 To some extent
- 3 A little bit
- 4 No

v367, weighted by weight

Value	Label	Missing	Count	Percent	Valid Percent
-5	other missing	М	3	0.2	
-3	nap	М	41	2.7	
-2	na	М	8	0.5	
-1	dk	М	116	7.7	
1	yes		126	8.3	9.4
2	to some extent		283	18.7	21.1
3	a little bit		329	21.8	24.5
4	no		604	40.0	45.0
	Sum		1510	100.0	100.0
	Valid Cases		1342		

v368b_CC - Region at age 14: country (Q133)

HARMONIZED VARIABLE - COUNTRY

Q133

Where did you live when you were about 14 years old?

Code:

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 4000000 Afghanistan
- 8000000 Albania
- 12000000 Algeria
- 20000000 Andorra
- 24000000 Angola
- 28000000 Antigua and Barbuda
- 31000000 Azerbaijan
- 32000000 Argentina
- 36000000 Australia
- 40000000 Austria
- 44000000 Bahamas
- 48000000 Bahrain
- 50000000 Bangladesh
- 51000000 Armenia
- 52000000 Barbados
- 56000000 Belgium
- 64000000 Bhutan
- 68000000 Bolivia
- 7000000 Bosnia-Herzegovina
- 72000000 Botswana
- 76000000 Brazil
- 84000000 Belize
- 90000000 Solomon Islands
- 96000000 Brunei
- 100000000 Bulgaria
- 104000000 Burma (Myanmar)
- 108000000 Burundi
- 112000000 Belarus
- 116000000 Cambodia
- 120000000 Cameroon
- 124000000 Canada
- 132000000 Cape Verde
- 14000000 Central African Republic
- 144000000 Sri Lanka
- 148000000 Chad

152000000 Chile 156000000 China 158000000 Taiwan 170000000 Colombia 174000000 Comoros 178000000 Congo, Rep. of 180000000 Congo, Dem. Rep. Of 188000000 Costa Rica 191000000 Croatia 192000000 Cuba 196100000 Cyprus 196200000 Northern Cyprus 200000000 Czechoslovakia 203000000 Czech Republic 204000000 Benin 208000000 Denmark 212000000 Dominica 214000000 Dominican Republic 218000000 Ecuador El Salvador 222000000 226000000 **Equatorial Guinea** 231000000 Ethiopia 232000000 Eritrea 233000000 Estonia 242000000 Fiji 246000000 Finland 248000000 Ålands Islands 250000000 France 254000000 French Guiana 262000000 Djibouti 266000000 Gabon 268000000 Georgia 270000000 Gambia 275000000 Palestinian Territory, occupied 276000000 Germany 288000000 Ghana 292000000 Gibraltar 296000000 Kiribati 300000000 Greece 308000000 Grenada 312000000 Guadeloupe 320000000 Guatemala 324000000 Guinea 328000000 Guyana 332000000 Haiti 340000000 Honduras 344000000 Hong Kong 348000000 Hungary 352000000

Iceland

356000000	India
360000000	Indonesia
364000000	Iran
368000000	Iraq
372000000	Ireland
376000000	Israel
380000000	Italy
384000000	Côte d'Ivoire
388000000	Jamaica
392000000	Japan
398000000	Kazakhstan
400000000	Jordan
404000000	Kenya
408000000	Korea, North
410000000	Korea, South
414000000	Kuwait
417000000	Kyrgyzstan
418000000	Laos
422000000	Lebanon
426000000	Lesotho
428000000	Latvia
430000000	Liberia
434000000	Libya
438000000	Liechtenstein
440000000	Lithuania
442000000	Luxembourg
446000000	Macau
450000000	Madagascar
454000000	Malawi
458000000	Malaysia
462000000	Maldives
466000000	Mali
470000000	Malta
474000000	Martinique
478000000	Mauritania
480000000	Mauritius
484000000	Mexico
492000000	Monaco
496000000	Mongolia
498000000	Moldava
504000000	Morocco
508000000	Mozambique
512000000	Oman
516000000	Namibia
520000000	Nauru
524000000	Nepal
528000000	Netherlands
530000000	Netherlands Antilles
533000000	Aruba

548000000	Vanuatu
554000000	New Zealand
558000000	Nicaragua
562000000	Niger
566000000	Nigeria
578000000	Norway
583000000	Micronesia
584000000	Marshall Islands
585000000	Palau
586000000	Pakistan
591000000	Panama
598000000	Papua New Guinea
600000000	Paraguay
604000000	Peru
608000000	Philippines
616000000	Poland
620000000	Portugal
624000000	Guinea-Bissau
626000000	East Timor
630000000	Puerto Rico
634000000	Qatar
638000000	Réunion
642000000	Romania
643000000	Russia
646000000	Rwanda
659000000	St. Kitts and Nevis
662000000	St. Lucia
663000000	Saint Martin
670000000	St. Vincent and the Grenadines
674000000	San Marino
678000000	São Tomé and Príncipe
682000000	Saudi Arabia
686000000	Senegal
690000000	Seychelles
694000000	Sierra Leone
702000000	Singapore
703000000	Slovakia
704000000	Vietnam
705000000	Slovenia
706000000	Somalia
710000000	South Africa
716000000	Zimbabwe
724000000	Spain
736000000	Sudan
740000000	Suriname
748000000	Swaziland
752000000	Sweden
756000000	Switzerland

760000000 Syria

762000000 Tajikistan 764000000 Thailand 768000000 Togo 776000000 Tonga 780000000 Trinidad and Tobago 784000000 **United Arab Emirates** 788000000 Tunisia 792000000 Turkey 795000000 Turkmenistan 798000000 Tuvalu 800000000 Uganda 804000000 Ukraine 807000000 Macedonia **USSR** 810000000 818000000 Egypt 826100000 Great Britain 826200000 Northern Ireland 834000000 Tanzania 840000000 **United States** 854000000 Burkina Faso 858000000 Uruguay 860000000 Uzbekistan 862000000 Venezuela 882000000 Samoa 887000000 Yemen 890000000 Yugoslavia, Socialist Federal Republic of 891000000 Serbia and Montenegro 894000000 Zambia 900000000 West-Germany 901000000 East-Germany 911000000 Serbia 912000000 Montenegro 915000000 Kosovo 916000000 Austria-Hungary 917000000 Alien's passport

Derivation:

Variable was computed from variable v368b_N3.

```
v368b_N3 - Region at age 14: NUTS-3 code (Q133)
Q133
Where did you live when you were about 14 years old?
<CODING INSTRUCTION: CODE REGION IN NUTS 3>
Code:
-5
   other missing
-4
   question not asked
   nap
-3
-2
   na
-1
   dk
8000101 AL - Albania - South-Albania - Berat
8000102 AL – Albania – South-Albania – Elbasan
8000103 AL - Albania - South-Albania - Fier
8000104 AL - Albania - South-Albania - Gjirokastër
8000105 AL - Albania - South-Albania - Korçë
8000106 AL - Albania - South-Albania - Vlore
8000201 AL - Albania - North-Albania - Dibër
8000202 AL - Albania - North-Albania - Durrës
8000203 AL - Albania - North-Albania - Kukës
8000204 AL - Albania - North-Albania - Lezhë
8000205 AL - Albania - North-Albania - Shkodër
8000206 AL - Albania - North-Albania - Tiranë
31000001 AZ - Azerbaijan - Baku-Absheron
31000002 AZ – Azerbaijan – Nakhchivan
31000003 AZ - Azerbaijan - Ganja-Gazakh
31000004 AZ - Azerbaijan - Sheki-Zagatala
31000005 AZ - Azerbaijan - Lankaran-Astara
31000006 AZ - Azerbaijan - Garabakh-Mil
31000007 AZ - Azerbaijan - Guba-Khachmaz
31000008 AZ - Azerbaijan - Mughan-Salyan
31000009 AZ - Azerbaijan - Shirvan
40010101 AT - Ostösterreich - Burgenland (A) - Mittelburgenland
40010102 AT - Ostösterreich - Burgenland (A) - Nordburgenland
40010103 AT - Ostösterreich - Burgenland (A) - Südburgenland
40010201 AT – Ostösterreich – Niederösterreich – Mostviertel-Eisenwurzen
40010202 AT - Ostösterreich - Niederösterreich - Niederösterreich-Süd
40010203 AT - Ostösterreich - Niederösterreich - Sankt Pölten
40010204
           AT – Ostösterreich – Niederösterreich – Wadviertel
40010205 AT - Ostösterreich - Niederösterreich - Weinviertel
40010206
          AT - Ostösterreich - Niederösterreich Wiener Umland/Nordteil
40010207
           AT - Ostösterreich - Niederösterreich Wiener Umland/Südteil
40010300 AT - Ostösterreich - Wien - Wien
40020101 AT - Südösterreich - Kärnten - Klagenfurt-Villach
40020102 AT - Südösterreich - Kärnten - Oberkärnten
40020103 AT - Südösterreich - Kärnten - Unterkärnten
40020201 AT - Südösterreich - Steiermark - Graz
```

40020202 AT - Südösterreich - Steiermark - Liezen

```
40020203
           AT - Südösterreich - Steiermark - Östliche Obersteiermark
40020204
           AT - Südösterreich - Steiermark - Oststeiermark
           AT - Südösterreich - Steiermark - West-und Südsteiermark
40020205
           AT - Südösterreich - Steiermark - Westliche Obersteiermark
40020206
40030101
          AT – Westöstereich – Oberösterreich – Innviertel
40030102
           AT - Westöstereich - Oberösterreich - Linz-Wels
40030103 AT - Westöstereich - Oberösterreich - Mühlviertel
40030104 AT - Westöstereich - Oberösterreich - Steyr-Kirchdorf
40030105 AT – Westöstereich – Oberösterreich – Traunviertel
40030201 AT – Westöstereich – Salzburg – Lungau
40030202 AT - Westöstereich - Salzburg - Pinzgau-Pongau
40030203 AT - Westöstereich - Salzburg - Salzburg und Umgebung
40030301 AT – Westöstereich – Tirol – Außerfern
40030302 AT – Westöstereich – Tirol – Inssbruck
40030303 AT - Westöstereich - Tirol - Osttirol
40030304 AT - Westöstereich - Tirol - Tiroler Oberland
40030305 AT - Westöstereich - Tirol - Tiroler Unterland
40030401 AT - Westöstereich - Vorarlberg - Bludens-Bregenzer Wald
40030402 AT - Westöstereich - Vorarlberg - Rheintal-Bodenseegebiet
51000101 AM - Armenia - Yerevan - Yerevan
51000201 AM - Armenia - Rest of Armenia - Shirak
51000202 AM - Armenia - Rest of Armenia - Lori
51000203
           AM - Armenia - Rest of Armenia - Tavush
51000204
          AM – Armenia – Rest of Armenia – Aragatsotn
51000205
           AM - Armenia - Rest of Armenia - Kotayk
51000206 AM - Armenia - Rest of Armenia - Gegharkunik
51000207 AM - Armenia - Rest of Armenia - Armavir
51000208 AM - Armenia - Rest of Armenia - Ararat
51000209 AM - Armenia - Rest of Armenia - Vavots Dzor
51000210 AM - Armenia - Rest of Armenia - Syunik
56010000 BE - Région de Bruxelles-capitale/Brussels hoofdstedelijk gewest - Arr. de Bruxelles-Capitale/ Arr. van Brussel-Ho
56020101
           BE - Vlaams gewest - Prov. Antwerpen - Arr. Antwerpen
56020102
           BE - Vlaams gewest - Prov. Antwerpen - Arr. Mechelen
           BE - Vlaams gewest - Prov. Antwerpen - Arr. Turnhout
56020103
56020201
           BE - Vlaams gewest - Prov. Limburg (B) - Arr. Hasselt
56020202
           BE - Vlaams gewest - Prov. Limburg (B) - Arr. Maaseik
56020203 BE – Vlaams gewest – Prov. Limburg (B) – Arr. Tongeren
56020301 BE - Vlaams gewest - Prov. Oost-Vlaanderen - Arr. Aalst
56020302 BE - Vlaams gewest - Prov. Oost-Vlaanderen - Arr. Dendermonde
56020303 BE - Vlaams gewest - Prov. Oost-Vlaanderen - Arr. Eeklo
56020304
           BE - Vlaams gewest - Prov. Oost-Vlaanderen - Arr. Gent
56020305
           BE - Vlaams gewest - Prov. Oost-Vlaanderen - Arr. Oudenaarde
56020306
           BE - Vlaams gewest - Prov. Oost-Vlaanderen - Arr. Sint-Niklaas
           BE - Vlaams gewest - Prov. Vlaams-Brabant - Arr. Halle- Vilvoorde
56020401
56020402
           BE - Vlaams gewest - Prov. Vlaams-Brabant - Arr. Leuven
56020501
           BE - Vlaams gewest - Prov. West-Vlaanderen - Arr. Brugge
56020502
          BE - Vlaams gewest - Prov. West-Vlaanderen - Arr. Diksmuide
56020503
           BE - Vlaams gewest - Prov. West-Vlaanderen - Arr. leper
56020504 BE - Vlaams gewest - Prov. West-Vlaanderen - Arr. Kortrijk
```

```
56020505
           BE - Vlaams gewest - Prov. West-Vlaanderen - Arr. Oostende
56020506
           BE - Vlaams gewest - Prov. West-Vlaanderen - Arr. Roeselare
56020507
           BE - Vlaams gewest - Prov. West-Vlaanderen - Arr. Tielt
           BE - Vlaams gewest - Prov. West-Vlaanderen - Arr. Veurne
56020508
56030100
           BE - Région Wallonne - Prov. Brabant Wallon - Arr. Nivelles
56030201
           BE - Région Wallonne - Prov. Hainaut - Arr. Ath
56030202 BE - Région Wallonne - Prov. Hainaut - Arr. Charleroi
56030203 BE - Région Wallonne - Prov. Hainaut - Arr. Mons
56030204 BE - Région Wallonne - Prov. Hainaut - Arr. Mouscron
56030205
           BE - Région Wallonne - Prov. Hainaut - Arr. Soignies
56030206
          BE - Région Wallonne - Prov. Hainaut - Arr. Thuin
56030207
           BE - Région Wallonne - Prov. Hainaut - Arr. Tournai
56030301
           BE - Région Wallonne - Prov. Liège - Arr. Huy
56030302
           BE - Région Wallonne - Prov. Liège - Arr. Liège
56030304
           BE - Région Wallonne - Prov. Liège - Arr. Waremme
56030305
           BE - Région Wallonne - Prov. Liège - Arr. Verviers-commmunes francophonees
56030306 BE – Région Wallonne – Prov. Liège – Bezirk Verviers-Deutschsprachige Gemeinschaft
56030401 BE - Région Wallonne - Prov. Luxembourg (B) - Arr. Arlon
56030402 BE - Région Wallonne - Prov. Luxembourg (B) - Arr. Bastogne
56030403
           BE - Région Wallonne - Prov. Luxembourg (B) - Arr. Marche-en-Famenne
56030404
           BE - Région Wallonne - Prov. Luxembourg (B) - Arr. Neufchâteau
56030405
           BE - Région Wallonne - Prov. Luxembourg (B) - Arr. Virton
56030501
           BE - Région Wallonne - Prov. Namur - Arr. Dinant
56030502
           BE - Région Wallonne - Prov. Namur - Arr. Namur
56030503
           BE - Région Wallonne - Prov. Namur - Arr. Philippeville
           BA – Bosna i Hercegovina – Federacija Bosne i Hercegovine – Unsko-sanksi
70000101
70000102
           BA - Bosna i Hercegovina - Federacija Bosne i Hercegovine - Povaski
70000103 BA – Bosna i Hercegovina – Federacija Bosne i Hercegovine – Tuzlanski
70000104
           BA – Bosna i Hercegovina – Federacija Bosne i Hercegovine – Zenicko-dobjski
70000105 BA – Bosna i Hercegovina – Federacija Bosne i Hercegovine – Bosansko-podrinjski
70000106
           BA - Bosna i Hercegovina - Federacija Bosne i Hercegovine - Srednjobosanski
70000107
           BA - Bosna i Hercegovina - Federacija Bosne i Hercegovine - Hercegovacko-neretvanski
70000108
           BA - Bosna i Hercegovina - Federacija Bosne i Hercegovine - Zapadnhercegovacki
70000109
           BA - Bosna i Hercegovina - Federacija Bosne i Hercegovine - Sarajevo
70000110
           BA - Bosna i Hercegovina - Federacija Bosne i Hercegovine - Livno (Kanton br. 10)
70000201
           BA - Bosna i Hercegovina - Republika Srpska - Banja Luka
70000202 BA - Bosna i Hercegovina - Republika Srpska - Doboj
70000203 BA - Bosna i Hercegovina - Republika Srpska - Bijeljina
70000204 BA - Bosna i Hercegovina - Republika Srpska - Vlasenica
70000205 BA - Bosna i Hercegovina - Republika Srpska - Sarajevo -Romanija
70000206 BA - Bosna i Hercegovina - Republika Srpska - Foca
70000207
           BA - Bosna i Hercegovina - Republika Srpska - Trebinje
70000301
           BA - Bosna i Hercegovina - Brcko - Brcko Disrikt
            BG - Severna i iztochna Bulgaria - Severozapaden - Vidin
100030101
100030102
            BG - Severna i iztochna Bulgaria - Severozapaden - Montana
100030103
            BG - Severna i iztochna Bulgaria - Severozapaden - Vratsa
100030104
            BG – Severna i iztochna Bulgaria – Severozapaden – Pleven
100030105 BG - Severna i iztochna Bulgaria - Severozapaden - Lovech
100030201 BG - Severna i iztochna Bulgaria - Severen tsentralen - Veliko Tarnovo
```

```
100030202
            BG - Severna i iztochna Bulgaria - Severen tsentralen - Gabrovo
100030203
            BG - Severna i iztochna Bulgaria - Severen tsentralen - Ruse
100030204
            BG - Severna i iztochna Bulgaria - Severen tsentralen - Razgrad
100030205
            BG - Severna i iztochna Bulgaria - Severen tsentralen - Silistra
100030301
            BG - Severna i iztochna Bulgaria - Severoiztochen - Varna
100030302
            BG - Severna i iztochna Bulgaria - Severoiztochen - Dobrich
100030303
            BG - Severna i iztochna Bulgaria - Severoiztochen - Shumen
100030304
            BG - Severna i iztochna Bulgaria - Severoiztochen - Targovishte
100030401
            BG - Severna i iztochna Bulgaria - Yugoiztochen - Burgas
100030402
            BG - Severna i iztochna Bulgaria - Yugoiztochen - Sliven
100030403
            BG - Severna i iztochna Bulgaria - Yugoiztochen - Yambol
100030404
            BG - Severna i iztochna Bulgaria - Yugoiztochen - Stara Zagora
100040101
            BG – Yugozapadna i yuzhna tsentralna Bulgaria – Yugozapaden – Sofia (stolitsa)
            BG - Yugozapadna i yuzhna tsentralna Bulgaria - Yugozapaden - Sofia
100040102
100040103
            BG - Yugozapadna i yuzhna tsentralna Bulgaria - Yugozapaden - Blagoevgrad
100040104
            BG - Yugozapadna i yuzhna tsentralna Bulgaria - Yugozapaden - Pernik
100040105
            BG - Yugozapadna i yuzhna tsentralna Bulgaria - Yugozapaden - Kyustendil
100040201
            BG - Yugozapadna i yuzhna tsentralna Bulgaria - Yuzhen tsentralen - Plovdiv
100040202
            BG - Yugozapadna i yuzhna tsentralna Bulgaria - Yuzhen tsentralen - Haskovo
100040203
            BG - Yugozapadna i yuzhna tsentralna Bulgaria - Yuzhen tsentralen - Pazardzhik
100040204
            BG - Yugozapadna i yuzhna tsentralna Bulgaria - Yuzhen tsentralen - Smolyan
100040205
            BG - Yugozapadna i yuzhna tsentralna Bulgaria - Yuzhen tsentralen - Kardzhali
112000001
            BY - Belarus - Brest
112000002
            BY - Belarus - Vitebsk
112000003
            BY - Belarus - Gomel
112000004 BY - Belarus - Grodno
112000005 BY - Belarus - Minsk (capital city)
112000006 BY - Belarus - Minsk
112000007 BY - Belarus - Mogilev
191000101 HR - Hrvatska - Sjeverozapadna Hrvatska - Grad Zagreb
191000102 HR - Hrvatska - Sjeverozapadna Hrvatska - Zagrebacka zupanija
191000103 HR - Hrvatska - Sjeverozapadna Hrvatska - Krapinsko-zagorska zupanija
191000104 HR - Hrvatska - Sjeverozapadna Hrvatska - Varazdinska zupanija
191000105 HR – Hrvatska – Sjeverozapadna Hrvatska – Koprivnicko-krizevacka zupanija
191000106 HR – Hrvatska – Sjeverozapadna Hrvatska – Medimurska zupanija
191000201
            HR – Hrvatska – Sredisnja i Istocna (Panonska) Hrvatska – Bjelovarsko-bilogorska zupanija
191000202 HR - Hrvatska - Sredisnja i Istocna (Panonska) Hrvatska - Viroviticko-podravska zupanija
191000203 HR - Hrvatska - Sredisnja i Istocna (Panonska) Hrvatska - Pozesko-slavonska zupanija
191000204
            HR – Hrvatska – Sredisnja i Istocna (Panonska) Hrvatska – Brodsko-posavska zupanija
191000205 HR – Hrvatska – Sredisnja i Istocna (Panonska) Hrvatska – Osjecko-baranjska zupanija
191000206 HR - Hrvatska - Sredisnja i Istocna (Panonska) Hrvatska - Vukovarsko-srijemska zupanija
191000207
            HR - Hrvatska - Sredisnja i Istocna (Panonska) Hrvatska - Karlovacka zupanija
191000208
            HR - Hrvatska - Sredisnja i Istocna (Panonska) Hrvatska - Sisacko-moslavacka zupanija
            HR - Hrvatska - Jadranska Hrvatska - Primorsko-goranska zupanija
191000301
191000302 HR - Hrvatska - Jadranska Hrvatska - Licko-senjska zupanija
191000303
            HR - Hrvatska - Jadranska Hrvatska - Zadarska zupanija
191000304 HR – Hrvatska – Jadranska Hrvatska – Sibensko-kninska zupanija
191000305 HR – Hrvatska – Jadranska Hrvatska – Splitsko-dalmatinska zupanija
191000306 HR – Hrvatska – Jadranska Hrvatska – Istarska zupanija
```

```
191000307
           HR - Hrvatska - Jadranska Hrvatska - Dubrovacko-neretvanska zupanija
203000100
           CZ - Ceska Republika - Praha - Hlavni mesto Praha
203000200
            CZ - Ceska Republika - Stredni Cechy - Stredocesky kraj
203000301
            CZ - Ceska Republika - Jihozapad - Jihocesky kraj
203000302 CZ - Ceska Republika - Jihozapad - Plzensky kraj
           CZ - Ceska Republika - Severozapad Karlovarsky kraj
203000401
203000402 CZ - Ceska Republika - Severozapad Ustecky kraj
203000501
            CZ - Ceska Republika - Severovychod Liberecky kraj
203000502 CZ - Ceska Republika - Severovychod Kraloveehradecky kraj
203000503
           CZ - Ceska Republika - Severovychod Pardubicky kraj
203000603
           CZ - Ceska Republika - Jihovychod - Vysocina
203000604
            CZ - Ceska Republika - Jihovychod - Jihomoravsky kraj
203000701 CZ – Ceska Republika – Stredni Morava – Olomoucky kraj
203000702 CZ - Ceska Republika - Stredni Morava - Zlinsky kraj
203000800 CZ - Ceska Republika - Moravskoslezsko - Moravskoslezsky kraj
           DK - Danmark - Hovedstaden - Byen København
208000101
208000102 DK - Danmark - Hovedstaden - Københavns omegn
208000103 DK - Danmark - Hovedstaden Nordsjælland
208000104
           DK - Danmark - Hovedstaden - Bornholm
208000201
            DK - Danmark- Sjælland - Østsjælland
208000202
            DK - Danmark- Sjælland - Vest - og Sydsjælland
208000301
            DK - Danmark - Syddanmark - Fyn
208000302
            DK - Danmark - Syddanmark - Sydjylland
208000401
            DK - Danmark - Midtjylland - Vestjylland
            DK - Danmark - Midtjylland - Østjylland
208000402
208000500
           DK - Danmark - Nordjylland - Nordjylland
233000001
            EE - Eesti - Põhja-Eesti
233000004 EE - Eesti - Lääne-Eesti
233000006 EE - Eesti - Kesk-Eesti
233000007 EE - Eesti - Kirde-Eesti
233000008
           EE - Eesti - Lõuna-Eesti
246010301 FI - Manner-Suomi - Itä-Suomi - Etalä-Savo
246010302 FI - Manner-Suomi - Itä-Suomi - Pohjois-Savo
246010303 FI - Manner-Suomi - Itä-Suomi - Pohjois-Karjala
246010304
           FI - Manner-Suomi - Itä-Suomi - Kainuu
246010801
           FI - Manner-Suomi - Etalä-Suomi - Uusimaa
246010802 FI - Manner-Suomi - Etalä-Suomi - Itä-Uusimaa
246010803 FI - Manner-Suomi - Etalä-Suomi - Varsinais-Suomi
246010804
           FI - Manner-Suomi - Etalä-Suomi - Kanta-Häme
246010805 FI - Manner-Suomi - Etalä-Suomi - Päijät-Häme
246010806 FI - Manner-Suomi - Etalä-Suomi - Kymenlaakso
246010807
           FI - Manner-Suomi - Etalä-Suomi - Etelä-Karjala
246010903 FI - Manner-Suomi - Länsi-Suomi - Keski-Suomi
           FI - Manner-Suomi - Länsi-Suomi - Etelä-Pohjanmaa
246010904
246010905 FI - Manner-Suomi - Länsi-Suomi - Pohjanmaa
246010906
           FI - Manner-Suomi - Länsi-Suomi - Satakunta
246010907 FI - Manner-Suomi - Länsi-Suomi - Pirkanmaa
246011001 FI - Manner-Suomi - Pohjois-Suomi - Keski-Pohjanmaa
246011002 FI - Manner-Suomi - Pohjois-Suomi - Pohjois-Pohjanmaa
```

```
246011003 FI - Manner-Suomi - Pohjois-Suomi - Lappi
246020000 FI - Åland
250010001 FR - Île de France - Paris
250010002
           FR - Île de France - Seine-et-Marne
250010003 FR - Île de France - Yvelines
250010004 FR - Île de France - Essonne
250010005 FR - Île de France - Hauts-de-Seine
250010006 FR - Île de France - Seine-Saint-Denis
250010007 FR – Île de France – Val-de-Marne
250010008 FR - Île de France - Val-d'Oise
250020101 FR - Bassin Parisien - Champagne-Ardenne - Ardennes
250020102 FR – Bassin Parisien – Champagne-Ardenne – Aube
250020103 FR - Bassin Parisien - Champagne-Ardenne - Marne
250020104 FR – Bassin Parisien – Champagne-Ardenne – Haute-Marne
250020201 FR - Bassin Parisien - Champagne-Ardenne - Picardie - Aisne
250020202 FR - Bassin Parisien - Champagne-Ardenne - Picardie - Oise
250020203 FR - Bassin Parisien - Champagne-Ardenne - Picardie - Somme
250020301 FR - Bassin Parisien - Champagne-Ardenne - Haute-Normandie - Eure
250020302 FR – Bassin Parisien – Champagne-Ardenne – Haute-Normandie – Seine-Martime
250020401 FR - Bassin Parisien - Champagne-Ardenne - Centre - Cher
250020402 FR - Bassin Parisien - Champagne-Ardenne - Centre - Eure-et-Loir
250020403 FR - Bassin Parisien - Champagne-Ardenne - Centre - Indre
250020404 FR - Bassin Parisien - Champagne-Ardenne - Centre - Indre-et-Loire
250020405 FR - Bassin Parisien - Champagne-Ardenne - Centre - Loir-et-Cher
250020406 FR - Bassin Parisien - Champagne-Ardenne - Centre - Loiret
250020501 FR - Bassin Parisien - Champagne-Ardenne - Basse-Normandie - Calvados
250020502 FR - Bassin Parisien - Champagne-Ardenne - Basse-Normandie - Manche
250020503 FR - Bassin Parisien - Champagne-Ardenne - Basse-Normandie - Orne
250020601 FR - Bassin Parisien - Champagne-Ardenne - Bourgogne - Côte-d'Or
250020602 FR - Bassin Parisien - Champagne-Ardenne - Bourgogne - Nièvre
250020603 FR - Bassin Parisien - Champagne-Ardenne - Bourgogne - Saône-et-Loire
250020604 FR - Bassin Parisien - Champagne-Ardenne - Bourgogne - Yonne
250030001 FR - Nord-pas-de-Calais - Nord
250030002 FR - Nord-pas-de-Calais - Pas-de-Calais
250040101 FR - Est - Lorraine - Meurthe-et-Moselle
250040102 FR - Est - Lorraine - Meuse
250040103 FR - Est - Lorraine - Moselle
250040104 FR - Est - Lorraine - Vosges
250040201 FR - Est - Alsace - Bas-Rhin
250040202 FR - Est - Alsace - Haut-Rhin
250040301 FR - Est - Franche-Comté - Doubs
250040302 FR - Est - Franche-Comté - Jura
250040303 FR - Est - Franche-Comté - Haute-Saône
250040304 FR - Est - Franche-Comté - Territoire de Belfort
250050101 FR - Ouest - Pays de la Loire - Loire-Atlantique
250050102 FR - Ouest - Pays de la Loire - Maine-et-Loire
250050103 FR - Ouest - Pays de la Loire - Mayenne
250050104 FR - Ouest - Pays de la Loire - Sarthe
250050105 FR - Ouest - Pays de la Loire - Vendée
```

```
250050201 FR - Ouest - Bretagne - Côtes-d'Armor
250050202
           FR - Ouest - Bretagne - Finistère
250050203 FR - Ouest - Bretagne - Ille-et-Vilaine
250050204
           FR - Ouest - Bretagne - Morbihan
250050301 FR - Ouest - Poitou-Charentes - Charente
250050302 FR - Ouest - Poitou-Charentes - Charente-Maritime
250050303 FR - Ouest - Poitou-Charentes - Deux-Sèvres
250050304 FR - Ouest - Poitou-Charentes - Vienne
250060101 FR - Sud-Ouest - Aquitaine - Dordogne
250060102 FR - Sud-Ouest - Aquitaine - Gironde
250060103 FR - Sud-Ouest - Aquitaine - Landes
250060104 FR - Sud-Ouest - Aquitaine - Lot-et-Garonne
250060105 FR - Sud-Ouest - Aquitaine - Pyrénées-Atlantiques
250060201 FR - Sud-Ouest - Midi-Pyrénées - Ariège
250060202 FR - Sud-Ouest - Midi-Pyrénées - Aveyron
250060203 FR - Sud-Ouest - Midi-Pyrénées - Haute-Garonne
250060204 FR - Sud-Ouest - Midi-Pyrénées - Gerts
250060205 FR - Sud-Ouest - Midi-Pyrénées - Lot
250060206 FR - Sud-Ouest - Midi-Pyrénées - Hautes-Pyrénées
250060207 FR - Sud-Ouest - Midi-Pyrénées - Tarn
250060208 FR - Sud-Ouest - Midi-Pyrénées - Tarn-et-Garonne
250060301 FR - Sud-Ouest - Limousin - Corrèze
250060302 FR - Sud-Ouest - Limousin - Creuse
250060303 FR - Sud-Ouest - Limousin - Haute-Vienne
250070101 FR - Centre-Est - Rhône-Alpes - Ain
250070102 FR - Centre-Est - Rhône-Alpes - Ardèche
250070103 FR - Centre-Est - Rhône-Alpes - Drôme
250070104 FR - Centre-Est - Rhône-Alpes - Isère
250070105 FR - Centre-Est - Rhône-Alpes - Loire
250070106 FR - Centre-Est - Rhône-Alpes - Rhône
250070107 FR - Centre-Est - Rhône-Alpes - Savoie
250070108 FR - Centre-Est - Rhône-Alpes - Haute-Savoie
250070201 FR - Centre-Est - Auvergne - Allier
250070202 FR - Centre-Est - Auvergne - Cantal
250070203 FR - Centre-Est - Auvergne - Haute-Loire
250070204 FR - Centre-Est - Auvergne - Puy-de-Dôme
250080101 FR - Méditerranée - Languedoc-Roussillon - Aude
250080102 FR - Méditerranée - Languedoc-Roussillon - Gard
250080103 FR - Méditerranée - Languedoc-Roussillon - Hérault
250080104 FR - Méditerranée - Languedoc-Roussillon - Lozère
250080105 FR - Méditerranée - Languedoc-Roussillon - Pyrénées-Orientales
250080201 FR - Méditerranée - Provence-Alpes-Côte d'Azur - Alpes-de-Haute-Provence
250080202 FR - Méditerranée - Provence-Alpes-Côte d'Azur - Hautes-Alpes
250080203
           FR - Méditerranée - Provence-Alpes-Côte d'Azur - Alpes-Maritimes
250080204 FR - Méditerranée - Provence-Alpes-Côte d'Azur - Bouches-du-Rhône
           FR - Méditerranée - Provence-Alpes-Côte d'Azur - Var
250080205
250080206 FR - Méditerranée - Provence-Alpes-Côte d'Azur - Vaucluse
250080301 FR - Méditerranée - Corse - Corse-du-Sud
250080302 FR - Méditerranée - Corse - Haute-Corse
```

```
250090100 FR - Départements d'outre-mer - Guadeloupe
250090200
           FR - Départements d'outre-mer - Martinique
250090300
           FR - Départements d'outre-mer - Guyane
250090400
           FR - Départements d'outre-mer - Réunion
268000001
           GE - Georgia - Tbilisi
268000002 GE - Georgia - Kakheti
268000003 GE - Georgia - Shida Kartli
268000004
           GE - Georgia - Kvemo Kartli
268000005 GE - Georgia - Samtskhe-Djavakheti
268000006 GE - Georgia - Adjara
268000007 GE - Georgia - Guria
268000008
            GE - Georgia - Samegrelo/ SvaneTi
           GE - Georgia - Imereti
268000009
           GE - Georgia - Mtskheta/TianeTi
268000010
268000011
           GE - Georgia - Racha-Lechkhumi/kvemo Svaneti
268000012 GE - Georgia - Abkhazia
276010101 DE - Baden-Württemberg - Stuttgart - Stuttgart, Stadtkreis
276010102 DE – Baden-Württemberg – Stuttgart – Böblingen
276010103 DE - Baden-Württemberg - Stuttgart - Esslingen
276010104 DE – Baden-Württemberg – Stuttgart – Göppingen
276010105 DE - Baden-Württemberg - Stuttgart - Ludwigsburg
276010106 DE – Baden-Württemberg – Stuttgart – Rems-Murr-Kreis
276010107
           DE - Baden-Württemberg - Stuttgart - Heilbronn, Stadtkreis
276010108
           DE - Baden-Württemberg - Stuttgart - Heilbronn, Landkreis
276010109
           DE - Baden-Württemberg - Stuttgart - Hohenlohekreis
276010110 DE - Baden-Württemberg - Stuttgart - Schwäbisch Hall
276010111 DE - Baden-Württemberg - Stuttgart - Main-Tauber-Kreis
276010112 DE - Baden-Württemberg - Stuttgart - Heidenheim
276010113 DE - Baden-Württemberg - Stuttgart - Ostalbkreis
276010201 DE - Baden-Württemberg - Karlsruhe - Baden-Baden, Stadtkreis
276010202 DE – Baden-Württemberg – Karlsruhe – Karlsruhe, Stadtkreis
276010203 DE - Baden-Württemberg - Karlsruhe - Karlsruhe, Landkreis
276010204
           DE - Baden-Württemberg - Karlsruhe - Rastatt
276010205 DE - Baden-Württemberg - Karlsruhe - Heidelberg, Stadtkreis
276010206 DE - Baden-Württemberg - Karlsruhe - Mannheim, Stadtkreis
276010207 DE – Baden-Württemberg – Karlsruhe – Neckar-Odenwald-Kreis
276010208 DE – Baden-Württemberg – Karlsruhe – Rhein-Neckar-Kreis
276010209
           DE - Baden-Württemberg - Karlsruhe - Pforzheim, Stadtkreis
276010210 DE - Baden-Württemberg - Karlsruhe - Calw
276010211 DE - Baden-Württemberg - Karlsruhe - Enzkreis
276010212 DE - Baden-Württemberg - Karlsruhe - Freudenstadt
276010301
            DE - Baden-Württemberg - Freiburg - Freiburg im Breisgau, Stadtkreis
276010302
           DE - Baden-Württemberg - Freiburg - Breisgau-Hochschwarzwald
276010303
            DE - Baden-Württemberg - Freiburg - Emmendingen
276010304
            DE - Baden-Württemberg - Freiburg - Ortenaukreis
276010305
            DE - Baden-Württemberg - Freiburg - Rottweil
276010306 DE - Baden-Württemberg - Freiburg - Schwarzwald-Baar-Kreis
276010307 DE - Baden-Württemberg - Freiburg - Tuttlingen
276010308 DE - Baden-Württemberg - Freiburg - Konstanz
```

```
276010309
           DE - Baden-Württemberg - Freiburg - Lörrach
276010310
           DE - Baden-Württemberg - Freiburg - Waldshut
276010401
            DE - Baden-Württemberg - Tübingen - Reutlingen
276010402
            DE - Baden-Württemberg - Tübingen - Tübingen, Landkreis
276010403
           DE - Baden-Württemberg - Tübingen - Zollernalbkreis
276010404
           DE - Baden-Württemberg - Tübingen - Ulm, Stadtkreis
276010405 DE – Baden-Württemberg – Tübingen – Alb-Donau-Kreis
276010406
           DE - Baden-Württemberg - Tübingen - Biberach
276010407 DE - Baden-Württemberg - Tübingen - Bodenseekreis
276010408
           DE - Baden-Württemberg - Tübingen - Ravensburg
276010409
           DE - Baden-Württemberg - Tübingen - Sigmaringen
276020101
           DE - Bayern - Oberbayern - Ingolstadt, Kreisfreie Stadt
276020102 DE - Bayern - Oberbayern - München, Kreisfreie Stadt
276020103 DE - Bayern - Oberbayern - Rosenheim, Kreisfreie Stadt
276020104 DE - Bayern - Oberbayern - Altötting
276020105 DE - Bayern - Oberbayern - Berchtesgadener Land
276020106 DE – Bayern – Oberbayern – Bad Tölz-Wolfratshausen
276020107 DE - Bayern - Oberbayern - Dachau
276020108 DE - Bayern - Oberbayern - Ebersberg
276020109
           DE - Bayern - Oberbayern - Eichstätt
276020110 DE - Bayern - Oberbayern - Erding
276020111
           DE - Bayern - Oberbayern - Freising
276020112
           DE - Bayern - Oberbayern - Fürstenfeldbruck
276020113
           DE - Bayern - Oberbayern - Garmisch-Partenkirchen
276020114
           DE - Bayern - Oberbayern - Landsberg a. Lech
276020115 DE - Bayern - Oberbayern - Miesbach
276020116 DE - Bayern - Oberbayern - Mühldorf a. Inn
276020117 DE - Bayern - Oberbayern - München, Landkreis
276020118 DE – Bavern – Oberbavern – Neuburg-Schrobenhausen
276020119 DE - Bayern - Oberbayern - Pfaffenhofen a. d. Ilm
276020120
           DE - Bayern - Oberbayern - Rosenheim, Landkreis
276020121 DE - Bayern - Oberbayern - Starnberg
276020122 DE - Bayern - Oberbayern - Traunstein
276020123 DE - Bayern - Oberbayern - Weilheim-Schongau
276020201
           DE - Bayern - Niederbayern - Landshut, Kreisfreie Stadt
276020202 DE - Bayern - Niederbayern - Passau, Kreisfreie Stadt
276020203 DE – Bayern – Niederbayern – Straubing, Kreisfreie Stadt
276020204
           DE - Bayern - Niederbayern - Deggendorf
276020205 DE - Bayern - Niederbayern - Freyung-Grafenau
276020206 DE - Bayern - Niederbayern - Kelheim
276020207
           DE - Bayern - Niederbayern - Landshut, Landkreis
276020208
           DE - Bayern - Niederbayern - Passau, Landkreis
276020209
            DE - Bayern - Niederbayern - Regen
276020210
           DE - Bayern - Niederbayern - Rottal-Inn
276020211
           DE - Bayern - Niederbayern - Straubing-Bogen
276020212
           DE - Bayern - Niederbayern - Dingolfing-Landau
276020301 DE - Bayern - Oberpfalz - Amberg, Kreisfreie Stadt
276020302 DE - Bayern - Oberpfalz - Regensburg, Kreisfreie Stadt
276020303 DE - Bayern - Oberpfalz - Weiden i. d. Opf, Kreisfreie Stadt
```

```
276020304 DE - Bayern - Oberpfalz - Amberg-Sulzbach
276020305 DE - Bayern - Oberpfalz - Cham
276020306
            DE - Bayern - Oberpfalz - Neumarkt i. d. OPf.
276020307
            DE - Bayern - Oberpfalz - Neustadt a. d. Waldnaab
276020308 DE - Bayern - Oberpfalz - Regensburg, Landkreis
276020309 DE - Bayern - Oberpfalz - Schwandorf
276020310 DE - Bayern - Oberpfalz - Tirschenreuth
276020401 DE - Bayern - Oberfranken - Bamberg, Kreisfreie Stadt
276020402 DE - Bayern - Oberfranken - Bayreuth, Kreisfreie Stadt
276020403 DE - Bayern - Oberfranken - Coburg, Kreisfreie Stadt
276020404 DE - Bayern - Oberfranken - Hof, Kreisfreie Stadt
276020405 DE - Bayern - Oberfranken - Bamberg, Landkreis
276020406 DE - Bayern - Oberfranken - Bayreuth, Landkreis
276020407 DE - Bayern - Oberfranken - Coburg, Landkreis
276020408 DE - Bayern - Oberfranken - Forchheim
276020409 DE - Bayern - Oberfranken - Hof, Landkreis
276020410 DE – Bayern – Oberfranken – Kronach
276020411 DE - Bayern - Oberfranken - Kulmbach
276020412 DE – Bayern – Oberfranken – Lichtenfels
276020413 DE - Bayern - Oberfranken - Wunsiedel i. Fichtelgebirge
276020501 DE - Bayern - Mittelfranken - Ansbach, Kreisfreie Stadt
276020502 DE - Bayern - Mittelfranken - Erlangen, Kreisfreie Stadt
276020503
            DE - Bayern - Mittelfranken - Fürth, Kreisfreie Stadt
276020504
            DE - Bayern - Mittelfranken - Nürnberg, Kreisfreie Stadt
            DE - Bayern - Mittelfranken - Schwabach, Kreisfreie Stadt
276020505
276020506 DE - Bayern - Mittelfranken - Ansbach, Landkreis
276020507
            DE - Bayern - Mittelfranken - Erlangen-Höchstadt
276020508 DE - Bayern - Mittelfranken - Fürth, Landkreis
276020509 DE – Bayern – Mittelfranken – Nürnberger Land
276020510 DE - Bayern - Mittelfranken - Neustadt a. d. Aisch-Bad Windsheim
276020511 DE - Bayern - Mittelfranken - Roth
276020512 DE – Bayern – Mittelfranken – Weißenburg-Gunzenhausen
276020601
           DE - Bayern - Unterfranken - Aschaffenburg, Kreisfreie Stadt
276020602 DE - Bayern - Unterfranken - Schweinfurt, Kreisfreie Stadt
276020603
            DE - Bayern - Unterfranken - Würzburg, Kreisfreie Stadt
276020604 DE - Bayern - Unterfranken - Aschaffenburg, Landkreis
276020605 DE – Bayern – Unterfranken – Bad Kissingen
276020606 DE – Bayern – Unterfranken – Rhön-Grabfeld
276020607 DE - Bayern - Unterfranken - Haßberge
276020608 DE - Bayern - Unterfranken - Kitzingen
276020609 DE - Bayern - Unterfranken - Miltenberg
276020610 DE - Bayern - Unterfranken - Main-Spessart
276020611
            DE - Bayern - Unterfranken - Schweinfurt, Landkreis
276020612
            DE - Bayern - Unterfranken - Würzburg, Landkreis
276020613 DE - Bayern - Unterfranken - Donau-Ries
276020614
            DE - Bayern - Unterfranken - Oberallgäu
276020701 DE - Bayern - Schwaben - Augsburg, Kreisfreie Stadt
276020702 DE - Bayern - Schwaben - Kaufbeuren, Kreisfreie Stadt
276020703 DE - Bayern - Schwaben - Kempten (Allgäu), Kreisfreie Stadt
```

```
276020704 DE - Bayern - Schwaben - Memmingen, Kreisfreie Stadt
276020705 DE - Bayern - Schwaben - Aichach-Friedberg
276020706 DE - Bayern - Schwaben - Augsburg, Landkreis
276020707
           DE - Bayern - Schwaben - Dillingen a.d. Donau
276020708 DE - Bayern - Schwaben - Günzburg
276020709 DE - Bayern - Schwaben - Neu-Ulm
276020710 DE - Bayern - Schwaben - Lindau (Bodensee)
276020711 DE - Bayern - Schwaben - Ostallgäu
276020712 DE - Bayern - Schwaben - Unterallgäu
276020713 DE - Bayern - Schwaben - Donau-Ries
276020714 DE - Bayern - Schwaben - Oberallgäu
276030000 DE - Berlin
276040101 DE - Brandenburg - Brandenburg - Nordost - Frankfurt (Oder), Kreisfreie Stadt
276040102 DE – Brandenburg – Brandenburg – Nordost – Barnim
276040103 DE - Brandenburg - Brandenburg - Nordost - Märkisch-Oderland
276040104 DE – Brandenburg – Brandenburg – Nordost – Oberhavel
276040105 DE – Brandenburg – Brandenburg – Nordost – Oder-Spree
276040106 DE – Brandenburg – Brandenburg – Nordost – Ostprignitz-Ruppin
276040107 DE – Brandenburg – Brandenburg – Nordost – Prignitz
276040108 DE – Brandenburg – Brandenburg – Nordost – Uckermark
276040201 DE - Brandenburg - Brandenburg Südwest - Brandenburg an der Havel, Kreisfreie Stadt
276040202 DE - Brandenburg - Brandenburg Südwest - Cottbus, Kreisfreie Stadt
276040203
           DE - Brandenburg - Brandenburg Südwest - Potsdam, Kreisfreie Stadt
276040204 DE - Brandenburg - Brandenburg Südwest - Dahme-Spreewald
276040205 DE - Brandenburg - Brandenburg Südwest - Elbe-Elster
276040206 DE - Brandenburg - Brandenburg Südwest - Havelland
276040207
           DE - Brandenburg - Brandenburg Südwest - Oberspreewald-Lausitz
276040208 DE – Brandenburg – Brandenburg Südwest – Potsdam-Mittelmark
276040209 DE – Brandenburg – Brandenburg Südwest – Spree-Neiße
276040210 DE – Brandenburg – Brandenburg Südwest – Teltow-Fläming
276050001 DE - Bremen - Bremen, Kreisfreie Stadt
276050002 DE - Bremen - Bremerhaven, Kreisfreie Stadt
276060000 DE - Hamburg
276070101 DE - Hessen - Darmstadt - Darmstadt, Kreisfreie Stadt
276070102 DE - Hessen - Darmstadt - Frankfurt am Main, Kreisfreie Stadt
276070103 DE - Hessen - Darmstadt - Offenbach am Main, Kreisfreie Stadt
276070104 DE - Hessen - Darmstadt - Wiesbaden, Kreisfreie Stadt
276070105 DE - Hessen - Darmstadt - Bergstraße
276070106 DE - Hessen - Darmstadt - Darmstadt-Dieburg
276070107 DE - Hessen - Darmstadt - Groß-Gerau
276070108 DE – Hessen – Darmstadt – Hochtaunuskreis
276070109 DE - Hessen - Darmstadt - Main-Kinzig-Kreis
276070110 DE - Hessen - Darmstadt - Main-Taunus-Kreis
276070111 DE - Hessen - Darmstadt - Odenwaldkreis
276070112 DE - Hessen - Darmstadt - Offenbach, Landkreis
276070113 DE - Hessen - Darmstadt - Rheingau-Taunus-Kreis
276070114 DE - Hessen - Darmstadt - Wetteraukreis
276070201 DE - Hessen - Gießen - Gießen, Landkreis
276070202 DE - Hessen - Gießen - Lahn-Dill-Kreis
```

```
276070203
           DE - Hessen - Gießen - Limburg-Weilburg
276070204
           DE - Hessen - Gießen - Marburg-Biedenkopf
276070205
           DE - Hessen - Gießen - Vogelsbergkreis
276070301
           DE - Hessen - Kassel - Kassel, Kreisfreie Stadt
276070302 DE - Hessen - Kassel - Fulda
276070303 DE - Hessen - Kassel - Hersfeld-Rotenburg
276070304 DE - Hessen - Kassel - Kassel, Landkreis
276070305
           DE - Hessen - Kassel - Schwalm-Eder-Kreis
276070306 DE – Hessen – Kassel – Waldeck-Frankenberg
276070307
           DE - Hessen - Kassel - Werra-Meißner-Kreis
276080001 DE - Mecklenburg-Vorpommern - Greifswald, Kreisfreie Stadt
276080002
           DE - Mecklenburg-Vorpommern - Neubrandenburg, Kreisfreie Stadt
276080003 DE - Mecklenburg-Vorpommern - Rostock, Kreisfreie Stadt
           DE - Mecklenburg-Vorpommern - Schwerin, Kreisfreie Stadt
276080004
276080005 DE - Mecklenburg-Vorpommern - Stralsund, Kreisfreie Stadt
276080006
           DE - Mecklenburg-Vorpommern - Wismar, Kreisfreie Stadt
276080007
           DE - Mecklenburg-Vorpommern - Bad Doberan
276080008
           DE - Mecklenburg-Vorpommern - Demmin
276080009
           DE - Mecklenburg-Vorpommern - Güstrow
276080010
           DE - Mecklenburg-Vorpommern - Ludwigslust
276080011
           DE - Mecklenburg-Vorpommern - Mecklenburg-Strelitz
276080012
           DE - Mecklenburg-Vorpommern - Müritz
276080013
           DE - Mecklenburg-Vorpommern - Nordvorpommern
276080014
           DE - Mecklenburg-Vorpommern - Nordwestmecklenburg
276080015
           DE - Mecklenburg-Vorpommern - Ostvorpommern
276080016 DE – Mecklenburg-Vorpommern – Parchim
276080017
           DE - Mecklenburg-Vorpommern - Rügen
276080018 DE - Mecklenburg-Vorpommern - Uecker-Randow
276090101
           DE - Niedersachsen - Braunschweig - Braunschweig, Kreisfreie Stadt
276090102 DE - Niedersachsen - Braunschweig - Salzgitter, Kreisfreie Stadt
276090103
           DE - Niedersachsen - Braunschweig - Wolfsburg, Kreisfreie Stadt
276090104 DE - Niedersachsen - Braunschweig - Gifhorn
276090105 DE - Niedersachsen - Braunschweig - Göttingen
276090106 DE - Niedersachsen - Braunschweig - Goslar
276090107 DE - Niedersachsen - Braunschweig - Helmstedt
276090108 DE - Niedersachsen - Braunschweig - Northeim
276090109 DE - Niedersachsen - Braunschweig - Osterode am Harz
276090110 DE - Niedersachsen - Braunschweig - Peine
276090111 DE - Niedersachsen - Braunschweig - Wolfenbüttel
276090202 DE - Niedersachsen - Hannover - Diepholz
276090203
           DE - Niedersachsen - Hannover - Hameln-Pyrmont
276090205
           DE - Niedersachsen - Hannover - Hildesheim
276090206
           DE - Niedersachsen - Hannover - Holzminden
276090207
           DE - Niedersachsen - Hannover - Nienburg (Weser)
276090208
           DE - Niedersachsen - Hannover - Schaumburg
276090209
           DE - Niedersachsen - Hannover - Region Hannover
276090301
           DE - Niedersachsen - Lüneburg - Celle
276090302 DE – Niedersachsen – Lüneburg – Cuxhaven
276090303 DE - Niedersachsen - Lüneburg - Harburg
```

```
276090304
            DE - Niedersachsen - Lüneburg - Lüchow-Dannenberg
276090305
            DE - Niedersachsen - Lüneburg - Lüneburg, Landkreis
276090306
            DE - Niedersachsen - Lüneburg - Osterholz
276090307
            DE - Niedersachsen - Lüneburg - Rotenburg (Wümme)
276090308
            DE - Niedersachsen - Lüneburg - Soltau-Fallingbostel
276090309
            DE - Niedersachsen - Lüneburg - Stade
276090310 DE - Niedersachsen - Lüneburg - Uelzen
276090311
            DE - Niedersachsen - Lüneburg - Verden
276090401
            DE - Niedersachsen - Weser-Ems - Delmenhorst, Kreisfreie Stadt
276090402 DE - Niedersachsen - Weser-Ems - Emden, Kreisfreie Stadt
276090403
            DE - Niedersachsen - Weser-Ems - Oldenburg (Oldenburg), Kreisfreie Stadt
            DE - Niedersachsen - Weser-Ems - Osnabrück, Kreisfreie Stadt
276090404
276090405 DE - Niedersachsen - Weser-Ems - Wilhelmshaven, Kreisfreie Stadt
276090406 DE - Niedersachsen - Weser-Ems - Ammerland
276090407 DE - Niedersachsen - Weser-Ems - Aurich
276090408 DE - Niedersachsen - Weser-Ems - Cloppenburg
276090409 DE - Niedersachsen - Weser-Ems - Emsland
276090410 DE - Niedersachsen - Weser-Ems - Friesland (D)
276090411 DE - Niedersachsen - Weser-Ems - Grafschaft Bentheim
276090412 DE - Niedersachsen - Weser-Ems - Leer
276090413 DE - Niedersachsen - Weser-Ems - Oldenburg, Landkreis
276090414
            DE - Niedersachsen - Weser-Ems - Osnabrück, Landkreis
276090415 DE - Niedersachsen - Weser-Ems - Vechta
276090416 DE - Niedersachsen - Weser-Ems - Wesermarsch
            DE - Niedersachsen - Weser-Ems - Wittmund
276090417
276100101 DE - Nordrhein-Westfalen - Düsseldorf - Düsseldorf, Kreisfreie Stadt
276100102 DE - Nordrhein-Westfalen - Düsseldorf - Duisburg, Kreisfreie Stadt
276100103 DE - Nordrhein-Westfalen - Düsseldorf - Essen, Kreisfreie Stadt
276100104
            DE - Nordrhein-Westfalen - Düsseldorf - Krefeld, Kreisfreie Stadt
276100105 DE – Nordrhein-Westfalen – Düsseldorf – Mönchengladbach, Kreisfreie Stadt
276100106
            DE - Nordrhein-Westfalen - Düsseldorf - Mülheim an der Ruhr. Kreisfreie Stadt
276100107 DE - Nordrhein-Westfalen - Düsseldorf - Oberhausen, Kreisfreie Stadt
276100108
            DE - Nordrhein-Westfalen - Düsseldorf - Remscheid, Kreisfreie Stadt
276100109 DE - Nordrhein-Westfalen - Düsseldorf - Solingen, Kreisfreie Stadt
276100110 DE - Nordrhein-Westfalen - Düsseldorf - Wuppertal, Kreisfreie Stadt
276100111 DE - Nordrhein-Westfalen - Düsseldorf - Kleve
276100112 DE - Nordrhein-Westfalen - Düsseldorf - Mettmann
276100113 DE - Nordrhein-Westfalen - Düsseldorf - Rhein-Kreis Neuss
276100114 DE – Nordrhein-Westfalen – Düsseldorf – Viersen
276100115 DE - Nordrhein-Westfalen - Düsseldorf - Wesel
276100201
            DE - Nordrhein-Westfalen - Köln - Aachen, Kreisfreie Stadt
276100202
            DE - Nordrhein-Westfalen - Köln - Bonn, Kreisfreie Stadt
276100203
            DE - Nordrhein-Westfalen - Köln - Köln, Kreisfreie Stadt
276100204
            DE - Nordrhein-Westfalen - Köln - Leverkusen, Kreisfreie Stadt
276100205 DE - Nordrhein-Westfalen - Köln - Aachen, Kreis
            DE - Nordrhein-Westfalen - Köln - Düren
276100206
276100207 DE - Nordrhein-Westfalen - Köln - Rhein-Erft-Kreis
276100208 DE - Nordrhein-Westfalen - Köln - Euskirchen
276100209 DE - Nordrhein-Westfalen - Köln - Heinsberg
```

```
276100210 DE - Nordrhein-Westfalen - Köln - Oberbergischer Kreis
276100211 DE - Nordrhein-Westfalen - Köln - Rheinisch-Bergischer Kreis
276100212
            DE - Nordrhein-Westfalen - Köln - Rhein-Sieg-Kreis
276100301
            DE - Nordrhein-Westfalen - Münster - Bottrop, Kreisfreie Stadt
276100302 DE - Nordrhein-Westfalen - Münster - Gelsenkirchen, Kreisfreie Stadt
276100303
            DE - Nordrhein-Westfalen - Münster - Münster, Kreisfreie Stadt
276100304 DE – Nordrhein-Westfalen – Münster – Borken
276100305
            DE - Nordrhein-Westfalen - Münster - Coesfeld
276100306 DE – Nordrhein-Westfalen – Münster – Recklinghausen
276100307
            DE - Nordrhein-Westfalen - Münster - Steinfurt
276100308 DE - Nordrhein-Westfalen - Münster - Warendorf
            DE - Nordrhein-Westfalen - Detmold - Bielefeld, Kreisfreie Stadt
276100401
276100402 DE - Nordrhein-Westfalen - Detmold - Gütersloh
276100403 DE - Nordrhein-Westfalen - Detmold - Herford
276100404 DE - Nordrhein-Westfalen - Detmold - Höxter
276100405 DE - Nordrhein-Westfalen - Detmold - Lippe
276100406 DE - Nordrhein-Westfalen - Detmold - Minden-Lübbecke
276100407 DE - Nordrhein-Westfalen - Detmold - Paderborn
276100501 DE - Nordrhein-Westfalen - Arnsberg - Bochum, Kreisfreie Stadt
276100502 DE - Nordrhein-Westfalen - Arnsberg - Dortmund, Kreisfreie Stadt
276100503 DE - Nordrhein-Westfalen - Arnsberg - Hagen, Kreisfreie Stadt
276100504
            DE - Nordrhein-Westfalen - Arnsberg - Hamm, Kreisfreie Stadt
276100505
            DE - Nordrhein-Westfalen - Arnsberg - Herne, Kreisfreie Stadt
276100506
            DE - Nordrhein-Westfalen - Arnsberg - Ennepe-Ruhr-Kreis
            DE - Nordrhein-Westfalen - Arnsberg - Hochsauerlandkreis
276100507
276100508 DE - Nordrhein-Westfalen - Arnsberg - Märkischer Kreis
276100509
            DE - Nordrhein-Westfalen - Arnsberg - Olpe
276100510 DE - Nordrhein-Westfalen - Arnsberg - Siegen-Wittgenstein
276100511 DE - Nordrhein-Westfalen - Arnsberg - Soest
276100512 DE - Nordrhein-Westfalen - Arnsberg - Unna
276110101 DE - Rheinland-Pfalz - Koblenz - Koblenz, Kreisfreie Stadt
276110102 DE - Rheinland-Pfalz - Koblenz - Ahrweiler
276110103 DE - Rheinland-Pfalz - Koblenz - Altenkirchen (Westerwald)
276110104 DE - Rheinland-Pfalz - Koblenz - Bad Kreuznach
276110105 DE - Rheinland-Pfalz - Koblenz - Birkenfeld
276110106 DE - Rheinland-Pfalz - Koblenz - Cochem-Zell
276110107 DE - Rheinland-Pfalz - Koblenz - Mayen-Koblenz
276110108 DE - Rheinland-Pfalz - Koblenz - Neuwied
276110109 DE - Rheinland-Pfalz - Koblenz - Rhein-Hunsrück-Kreis
276110110 DE - Rheinland-Pfalz - Koblenz - Rhein-Lahn-Kreis
276110111 DE - Rheinland-Pfalz - Koblenz - Westerwaldkreis
276110201 DE - Rheinland-Pfalz - Trier - Trier, Kreisfreie Stadt
276110202 DE - Rheinland-Pfalz - Trier - Bernkastel-Wittlich
276110203
            DE - Rheinland-Pfalz - Trier - Bitburg-Prüm
276110204 DE - Rheinland-Pfalz - Trier - Daun
276110205 DE - Rheinland-Pfalz - Trier - Trier-Saarburg
276110301 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Frankenthal (Pfalz), Kreisfreie Stadt
276110302 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Kaiserslautern, Kreisfreie Stadt
276110303 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Landau in der Pfalz, Kreisfreie Stadt
```

```
276110304 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Ludwigshafen am Rhein, Kreisfreie Stadt
276110305 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Mainz, Kreisfreie Stadt
276110306 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Neustadt an der Weinstraße, Kreisfreie Stadt
276110307
            DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Pirmasens, Kreisfreie Stadt
276110308 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Speyer, Kreisfreie Stadt
276110309 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Worms, Kreisfreie Stadt
276110310 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Zweibrücken, Kreisfreie Stadt
276110311 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Alzey-Worms
276110312 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Bad Dürkheim
276110313 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Donnersbergkreis
276110314 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Germersheim
276110315 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Kaiserslautern, Landkreis
276110316 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Kusel
276110317 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Südliche Weinstraße
276110318 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Rhein-Pfalz-Kreis
276110319 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Mainz-Bingen
276110320 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Südwestpfalz
276120001 DE - Saarland - Stadtverband Saarbrücken
276120002 DE - Saarland - Merzig-Wadern
276120003 DE - Saarland - Neunkirchen
276120004 DE - Saarland - Saarlouis
276120005 DE - Saarland - Saarpfalz-Kreis
276120006 DE - Saarland - St. Wendel
276130101 DE - Sachsen - Chemnitz - Chemnitz, Kreisfreie Stadt
276130102 DE - Sachsen - Chemnitz - Plauen, Kreisfreie Stadt
276130103 DE - Sachsen - Chemnitz - Zwickau, Kreisfreie Stadt
276130104 DE - Sachsen - Chemnitz - Annaberg
276130105 DE - Sachsen - Chemnitz - Chemnitzer Land
276130106 DE - Sachsen - Chemnitz - Freiberg
276130107 DE - Sachsen - Chemnitz - Vogtlandkreis
276130108 DE - Sachsen - Chemnitz - Mittlerer Erzgebirgskreis
276130109 DE - Sachsen - Chemnitz - Mittweida
276130110 DE - Sachsen - Chemnitz - Stollberg
276130111 DE - Sachsen - Chemnitz - Aue-Schwarzenberg
276130112 DE - Sachsen - Chemnitz - Zwickauer Land
276130201 DE - Sachsen - Dresden - Dresden, Kreisfreie Stadt
276130202 DE - Sachsen - Dresden - Görlitz, Kreisfreie Stadt
276130203 DE - Sachsen - Dresden - Hoyerswerda, Kreisfreie Stadt
276130204 DE - Sachsen - Dresden - Bautzen
276130205 DE - Sachsen - Dresden - Meißen
276130206 DE - Sachsen - Dresden - Niederschlesischer Oberlausitzkreis
276130207 DE - Sachsen - Dresden - Riesa-Großenhain
276130208 DE - Sachsen - Dresden - Löbau-Zittau
276130209 DE - Sachsen - Dresden - Sächsische Schweiz
276130210 DE - Sachsen - Dresden - Weißeritzkreis
276130211 DE - Sachsen - Dresden - Kamenz
276130301 DE - Sachsen - Leipzig - Leipzig, Kreisfreie Stadt
276130302 DE - Sachsen - Leipzig - Delitzsch
276130303 DE - Sachsen - Leipzig - Döbeln
```

```
276130304 DE - Sachsen - Leipzig - Leipziger Land
276130305 DE - Sachsen - Leipzig - Muldentalkreis
276130306
            DE - Sachsen - Leipzig - Torgau-Oschatz
276140001
            DE - Sachsen-Anhalt - Dessau-Roßlau, Kreisfreie Stadt
276140002 DE - Sachsen-Anhalt - Halle (Saale), Kreisfreie Stadt
276140003 DE - Sachsen-Anhalt - Magdeburg, Kreisfreie Stadt
276140004 DE - Sachsen-Anhalt - Altmarkkreis Salzwedel
276140005 DE - Sachsen-Anhalt - Anhalt-Bitterfeld
276140006 DE - Sachsen-Anhalt - Jerichower Land
276140007 DE - Sachsen-Anhalt - Börde
276140008 DE - Sachsen-Anhalt - Burgenland (D)
276140009 DE - Sachsen-Anhalt - Harz
276140010 DE - Sachsen-Anhalt - Mansfeld-Südharz
276140011 DE - Sachsen-Anhalt - Saalekreis
276140012 DE - Sachsen-Anhalt -Salzland
276140013 DE - Sachsen-Anhalt - Stendal
276140014 DE - Sachsen-Anhalt - Wittenberg
276150001 DE - Schleswig-Holstein - Flensburg, Kreisfreie Stadt
276150002 DE - Schleswig-Holstein - Kiel, Kreisfreie Stadt
276150003 DE - Schleswig-Holstein - Lübeck, Kreisfreie Stadt
276150004 DE - Schleswig-Holstein - Neumünster, Kreisfreie Stadt
276150005 DE - Schleswig-Holstein - Dithmarschen
276150006 DE - Schleswig-Holstein - Herzogtum Lauenburg
276150007 DE - Schleswig-Holstein - Nordfriesland
276150008 DE - Schleswig-Holstein - Ostholstein
276150009 DE - Schleswig-Holstein - Pinneberg
276150010 DE - Schleswig-Holstein - Plön
276150011 DE - Schleswig-Holstein - Rendsburg-Eckernförde
276150012 DE - Schleswig-Holstein - Schleswig-Flensburg
276150013 DE - Schleswig-Holstein - Segeberg
276150014 DE - Schleswig-Holstein - Steinburg
276150015 DE - Schleswig-Holstein - Stormarn
276160001 DE - Thüringen - Erfurt, Kreisfreie Stadt
276160002 DE - Thüringen - Gera, Kreisfreie Stadt
276160003 DE - Thüringen - Jena, Kreisfreie Stadt
276160004 DE - Thüringen - Suhl, Kreisfreie Stadt
276160005 DE - Thüringen - Weimar, Kreisfreie Stadt
276160006 DE - Thüringen - Eichsfeld
276160007 DE - Thüringen - Nordhausen
276160009 DE - Thüringen - Unstrut-Hainich-Kreis
276160010 DE - Thüringen - Kyffhäuserkreis
276160011
           DE - Thüringen - Schmalkalden-Meiningen
276160012 DE - Thüringen - Gotha
276160013 DE - Thüringen - Sömmerda
276160014 DE - Thüringen - Hildburghausen
276160015 DE - Thüringen - Ilm-Kreis
276160016 DE - Thüringen - Weimarer Land
276160017 DE - Thüringen - Sonneberg
276160018 DE - Thüringen - Saalfeld-Rudolstadt
```

```
276160019
            DE - Thüringen - Saale-Holzland-Kreis
276160020
            DE - Thüringen - Saale-Orla-Kreis
276160021
            DE - Thüringen - Greiz
276160022
            DE - Thüringen - Altenburger Land
276160023
            DE - Thüringen - Eisenach, Kreisfreie Stadt
276160025 DE - Thüringen - Wartburgkreis
300010101 GR - Voreia Ellada - Anatoliki Makedonia, Thraki - Evros
300010102 GR - Voreia Ellada - Anatoliki Makedonia, Thraki - Xanthi
300010103 GR - Voreia Ellada - Anatoliki Makedonia, Thraki - Rodopi
300010104
            GR - Voreia Ellada - Anatoliki Makedonia, Thraki - Drama
300010105
            GR - Voreia Ellada - Anatoliki Makedonia, Thraki - Kavala
300010201
            GR - Voreia Ellada - Kentriki Makedonia - Imathia
300010202 GR - Voreia Ellada - Kentriki Makedonia - Thessaloniki
300010203
            GR - Voreia Ellada - Kentriki Makedonia - Kilkis
300010204
            GR - Voreia Ellada - Kentriki Makedonia - Pella
300010205
            GR - Voreia Ellada - Kentriki Makedonia - Pieria
300010206 GR - Voreia Ellada - Kentriki Makedonia - Serres
300010207
            GR - Voreia Ellada - Kentriki Makedonia - Chalkidiki
300010301
            GR - Voreia Ellada - Dytiki Makedonia - Grevena
300010302
            GR - Voreia Ellada - Dytiki Makedonia - Kastoria
300010303
            GR - Voreia Ellada - Dytiki Makedonia - Kozani
300010304
            GR - Voreia Ellada - Dytiki Makedonia - Florina
300010401
            GR - Voreia Ellada - Thessalia - Karditsa
300010402
            GR - Voreia Ellada - Thessalia - Larisa
300010403
            GR - Voreia Ellada - Thessalia - Magnisia
300010404
            GR - Voreia Ellada - Thessalia - Trikala
300020101
            GR - Kentriki Ellada - Ipeiros - Arta
300020102 GR - Kentriki Ellada - Ipeiros - Thesprotia
300020103
            GR - Kentriki Ellada - Ipeiros - Ioannina
300020104
            GR - Kentriki Ellada - Ipeiros - Preveza
300020201
            GR - Kentriki Ellada - Ionia Nisia - Zakynthos
300020202
            GR - Kentriki Ellada - Ionia Nisia - Kerkyra
300020203
            GR - Kentriki Ellada - Ionia Nisia - Kefallinia
            GR - Kentriki Ellada - Ionia Nisia - Lefkada
300020204
            GR - Kentriki Ellada - Dytiki Ellada - Aitoloakarnania
300020301
300020302
            GR - Kentriki Ellada - Dytiki Ellada - Achaia
300020303 GR - Kentriki Ellada - Dytiki Ellada - Ileia
300020401
            GR - Kentriki Ellada - Sterea Ellada - Voiotia
300020402 GR - Kentriki Ellada - Sterea Ellada - Evvoia
300020403
            GR - Kentriki Ellada - Sterea Ellada - Evrytania
300020404
            GR - Kentriki Ellada - Sterea Ellada - Fthiotida
300020405
            GR - Kentriki Ellada - Sterea Ellada - Fokida
300020501
            GR - Kentriki Ellada - Peloponnisos - Argolida
300020502
            GR - Kentriki Ellada - Peloponnisos - Arkadia
300020503
            GR - Kentriki Ellada - Peloponnisos - Korinthia
300020504
            GR - Kentriki Ellada - Peloponnisos - Lakonia
300020505
            GR - Kentriki Ellada - Peloponnisos - Messinia
300030000
            GR - Attiki
300040101
            GR - Nisia Aigaiou, Kriti - Voreio Aigaio - Lesvos
```

```
300040102 GR - Nisia Aigaiou, Kriti - Voreio Aigaio - Samos
300040103
            GR - Nisia Aigaiou, Kriti - Voreio Aigaio - Chios
300040201
            GR - Nisia Aigaiou, Kriti - Notio Aigaio - Dodekanisos
300040202
            GR - Nisia Aigaiou, Kriti - Notio Aigaio - Kyklades
300040301
            GR - Nisia Aigaiou, Kriti - Kriti - Irakleio
300040302
            GR - Nisia Aigaiou, Kriti - Kriti - Lasithi
300040303 GR - Nisia Aigaiou, Kriti - Kriti - Rethymni
300040304
            GR - Nisia Aigaiou, Kriti - Kriti - Chania
348010001 HU - Közép-Magyarország - Budapest
348010002 HU - Közép-Magyarország - Pest
348020101 HU - Dunántúl - Közép-Dunántúl - Fejér
348020102 HU – Dunántúl – Közép-Dunántúl – Komárom-Esztergom
348020103 HU - Dunántúl - Közép-Dunántúl - Veszprém
348020201 HU – Dunántúl – Nyugat-Dunántúl – Gyor-Moson-Sopron
348020202 HU - Dunántúl - Nyugat-Dunántúl - Vas
348020203 HU - Dunántúl - Nyugat-Dunántúl - Zala
348020301 HU - Dunántúl - Dél-Dunántúl - Baranya
348020302 HU - Dunántúl - Dél-Dunántúl - Somogy
348020303 HU - Dunántúl - Dél-Dunántúl - Tolna
348030101 HU – Alföld és Észak – Észak-Magyarország – Borsod-Abaúj-Zemplén
348030102 HU - Alföld és Észak - Észak-Magyarország - Heves
348030103 HU - Alföld és Észak - Észak-Magyarország - Nógrád
348030201
            HU - Alföld és Észak - Észak-Alföld - Hajdú-Bihar
348030202 HU - Alföld és Észak - Észak-Alföld - Jász-Nagykun-Szolnok
            HU - Alföld és Észak - Észak-Alföld - Szabolcs-Szatmár-Bereg
348030203
348030301 HU - Alföld és Észak - Dél-Alföld - Bács-Kiskun
348030302 HU - Alföld és Észak - Dél-Alföld - Békés
348030303 HU - Alföld és Észak - Dél-Alföld - Csongrád
352000001
            IS – Ísland – Höfuðborgarsvæði
352000002 IS - Ísland - Landsbyggð
372000101
            IE - Ireland - Border, Midland and Western - Border
372000102 IE - Ireland - Border, Midland and Western - Midland
372000103 IE - Ireland - Border, Midland and Western - West
372000201 IE - Ireland - Southern and Eastern - Dublin
372000202 IE - Ireland - Southern and Eastern - Mid-East
372000203 IE - Ireland - Southern and Eastern - Mid-West
372000204 IE - Ireland - Southern and Eastern - South-East (IRL)
372000205 IE - Ireland - Southern and Eastern - South-West (IRL)
380120101 IT - Nord-Ovest - Piemonte - Torino
380120102 IT - Nord-Ovest - Piemonte - Vercelli
380120103 IT - Nord-Ovest - Piemonte - Biella
380120104 IT - Nord-Ovest - Piemonte - Verbano-Cusio-Ossola
380120105 IT - Nord-Ovest - Piemonte - Novara
380120106 IT - Nord-Ovest - Piemonte - Cuneo
380120107 IT - Nord-Ovest - Piemonte - Asti
380120108 IT - Nord-Ovest - Piemonte - Alessandria
380120200 IT - Nord-Ovest - Valle d'Aosta/Vallée d'Aoste
380120301 IT - Nord-Ovest - Liguria - Imperia
380120302 IT - Nord-Ovest - Liguria - Savona
```

```
380120303 IT - Nord-Ovest - Liguria - Genova
380120304 IT - Nord-Ovest - Liguria - La Spezia
380120401 IT - Nord-Ovest - Lombardia - Varese
380120402 IT - Nord-Ovest - Lombardia - Como
380120403 IT - Nord-Ovest - Lombardia - Lecco
380120404 IT - Nord-Ovest - Lombardia - Sondrio
380120405 IT - Nord-Ovest - Lombardia - Milano
380120406 IT - Nord-Ovest - Lombardia - Bergamo
380120407 IT - Nord-Ovest - Lombardia - Brescia
380120408 IT - Nord-Ovest - Lombardia - Pavia
380120409 IT - Nord-Ovest - Lombardia - Lodi
380120410 IT - Nord-Ovest - Lombardia - Cremona
380120411 IT - Nord-Ovest - Lombardia - Mantova
380130100 IT - Nord-Est - Provincia Autonoma Bolzano/Bozen - Bolzano-Bozen
380130200 IT - Nord-Est - Provincia Autonoma Trento - Trento
380130301 IT - Nord-Est - Veneto - Verona
380130302 IT - Nord-Est - Veneto - Vicenza
380130303 IT - Nord-Est - Veneto - Belluno
380130304 IT - Nord-Est - Veneto - Treviso
380130305 IT - Nord-Est - Veneto - Venezia
380130306 IT - Nord-Est - Veneto - Padova
380130307 IT - Nord-Est - Veneto - Rovigo
380130401 IT - Nord-Est - Friuli-Venezia Giulia - Pordenone
380130402 IT - Nord-Est - Friuli-Venezia Giulia - Udine
380130403 IT - Nord-Est - Friuli-Venezia Giulia - Gorizia
380130404 IT - Nord-Est - Friuli-Venezia Giulia - Trieste
380130501 IT - Nord-Est - Emilia-Romagna - Piacenza
380130502 IT - Nord-Est - Emilia-Romagna - Parma
380130503 IT - Nord-Est - Emilia-Romagna - Reggio nell'Emilia
380130504 IT - Nord-Est - Emilia-Romagna - Modena
380130505 IT - Nord-Est - Emilia-Romagna - Bologna
380130506 IT - Nord-Est - Emilia-Romagna - Ferrara
380130507 IT - Nord-Est - Emilia-Romagna - Ravenna
380130508 IT - Nord-Est - Emilia-Romagna - Forlì-Cesena
380130509 IT - Nord-Est - Emilia-Romagna - Rimini
380140101 IT - Centro (I) - Toscana - Massa-Carrara
380140102 IT - Centro (I) - Toscana - Lucca
380140103 IT - Centro (I) - Toscana - Pistoia
380140104 IT - Centro (I) - Toscana - Firenze
380140105 IT - Centro (I) - Toscana - Prato
380140106 IT - Centro (I) - Toscana - Livorno
380140107 IT - Centro (I) - Toscana - Pisa
380140108 IT - Centro (I) - Toscana - Arezzo
380140109 IT - Centro (I) - Toscana - Siena
380140110 IT - Centro (I) - Toscana - Grosseto
380140201 IT - Centro (I) - Umbria - Perugia
380140202 IT - Centro (I) - Umbria - Terni
380140301 IT - Centro (I) - Marche - Pesaro e Urbino
380140302 IT - Centro (I) - Marche - Ancona
```

```
380140303 IT - Centro (I) - Marche - Macerata
380140304 IT - Centro (I) - Marche - Ascoli Piceno
380140401 IT - Centro (I) - Lazio - Viterbo
380140402 IT - Centro (I) - Lazio - Rieti
380140403 IT - Centro (I) - Lazio - Roma
380140404 IT - Centro (I) - Lazio - Latina
380140405 IT - Centro (I) - Lazio - Frosinone
380150001 IT - Sud - Calabria - Cosenza
380150002 IT - Sud - Calabria - Crotone
380150003 IT - Sud - Calabria - Catanzaro
380150004 IT - Sud - Calabria - Vibo Valentia
380150005 IT - Sud - Calabria - Reggio di Calabria
380150101 IT - Sud - Abruzzo - L'Aquila
380150102 IT - Sud - Abruzzo - Teramo
380150103 IT - Sud - Abruzzo - Pescara
380150104 IT - Sud - Abruzzo - Chieti
380150201 IT - Sud - Molise - Isernia
380150202 IT - Sud - Molise - Campobasso
380150301 IT - Sud - Campania - Caserta
380150302 IT - Sud - Campania - Benevento
380150303 IT - Sud - Campania - Napoli
380150304 IT - Sud - Campania - Avellino
380150305 IT - Sud - Campania - Salerno
380150401 IT - Sud - Puglia - Foggia
380150402 IT - Sud - Puglia - Bari
380150403 IT - Sud - Puglia - Taranto
380150404 IT - Sud - Puglia - Brindisi
380150405 IT - Sud - Puglia - Lecce
380150501 IT - Sud - Basilicata - Potenza
380150502 IT - Sud - Basilicata - Matera
380160101 IT - Isole - Sicilia - Trapani
380160102 IT - Isole - Sicilia - Palermo
380160103 IT - Isole - Sicilia - Messina
380160104 IT - Isole - Sicilia - Agrigento
380160105 IT - Isole - Sicilia - Caltanissetta
380160106 IT - Isole - Sicilia - Enna
380160107 IT - Isole - Sicilia - Catania
380160108 IT - Isole - Sicilia - Ragusa
380160109 IT - Isole - Sicilia Siracusa
380160205 IT - Isole - Sardegna - Sassari
380160206 IT - Isole - Sardegna - Nuoro
380160207 IT - Isole - Sardegna - Cagliari
380160208 IT - Isole - Sardegna - Oristano
380160209 IT - Isole - Sardegna - Olbia-Tempio
380160210 IT - Isole - Sardegna - Ogliastra
380160211 IT - Isole - Sardegna - Medio Campidano
380160212 IT - Isole - Sardegna - Carbonia-Iglesias
428000003 LV - Latvija - Kurzeme
428000005 LV - Latvija - Latgale
```

```
428000006 LV - Latvija - Riga
428000007
            LV - Latvija - Pieriga
428000008
            LV - Latvija - Vidzeme
            LV - Latvija - Zemgale
428000009
440000001 LT - Lietuva - Alytaus apskritis
440000002 LT - Lietuva - Kauno apskritis
440000003 LT - Lietuva - Klaipedos apskritis
440000004 LT - Lietuva - Marijampoles apskritis
44000005 LT - Lietuva - Panevežio apskritis
44000006 LT - Lietuva - Šiauliu apskritis
440000007 LT - Lietuva - Taurages apskritis
440000008 LT - Lietuva - Telšiu apskritis
440000009 LT - Lietuva - Utenos apskritis
440000010 LT - Lietuva - Vilniaus apskritis
442000000 LU - Luxembourg (Grand-Duché)
470000001
            MT- Malta - Malta
470000002 MT- Malta - Gozo and Comino / Ghawdex u Kemmuna
498000001
            MD - Moldova - Cahul and Cantemir
498000002 MD - Moldova - Taraclia
498000003
            MD - Moldova - Autonomous Territorial Unit of Gagauzia and Comrat
498000004
            MD - Moldova - Basarabeasca, Hincesti, Leova and Cimislia
498000005
            MD - Moldova - Causeni, Stefan Voda
498000006
            MD - Moldova - Anenii Noi, Criuleni, Dubasari, Ialoveni and Straseni
498000007
            MD - Moldova - Chisinau
            MD - Moldova - Orhei, Rezina, Soldanesti, Telenesti
498000008
498000009
            MD - Moldova - Ungheni, Calarasi, and Nisporeni
498000010
            MD - Moldova - Balti, Falesti, Glodeni, Riscani, Singerei
498000011
            MD - Moldova - Soroca, Drochia and Floresti
498000012 MD - Moldova - Briceni, Edinet, Ocnita and Donduseni
498000013 MD - Moldova - Autonomous territorial unit of Transnistria and Bender
            ME - Montenegro - North - Beransko-Rozajski
499000101
499000102 ME - Montenegro - North - Bijelopoljsko-Pljevaljski
499000103
           ME - Montenegro - North - Niksicko-Zabljacki
            ME - Montenegro - Center - Podgoricko-Cetinjski
499000201
499000301
            ME - Montenegro - South - Barsko-Ulcinjski
           ME - Montenegro - South - Boka Kotorska
499000302
528010101
           NL - Noord-Nederland - Groningen - Oost-Groningen
528010102 NL - Noord-Nederland - Groningen - Delfzijl en omgeving
528010103 NL - Noord-Nederland - Groningen - Overig Groningen
528010201
            NL - Noord-Nederland - Friesland (NL) - Noord-Friesland
528010202 NL - Noord-Nederland - Friesland (NL) - Zuidwest-Friesland
528010203
            NL - Noord-Nederland - Friesland (NL) - Zuidoost-Friesland
528010301
            NL - Noord-Nederland - Drenthe - Noord-Drenthe
528010302
            NL - Noord-Nederland - Drenthe - Zuidsoost-Drenthe
528010303
            NL - Noord-Nederland - Drenthe - Zuidwest-Drenthe
528020101
            NL - Oost-Nederland - Overijssel - Noord-Overijssel
528020102 NL - Oost-Nederland - Overijssel - Zuidwest-Overijssel
528020103
            NL - Oost-Nederland - Overijssel - Twente
528020201 NL - Oost-Nederland - Gelderland - Veluwe
```

```
528020204
            NL - Oost-Nederland - Gelderland - Zuidwest-Gelderland
528020205
            NL - Oost-Nederland - Gelderland - Achterhoek
528020206
            NL - Oost-Nederland - Gelderland - Arnhem/Nijmegen
            NL - Oost-Nederland - Flevoland
528020300
528030100
           NL - West-Nederland - Utrecht
            NL - West-Nederland - Noord-Holland - Kop van Noord-Holland
528030201
528030202 NL - West-Nederland - Noord-Holland - Alkmaar en omgeving
528030203
           NL - West-Nederland - Noord-Holland - Ijmond
528030204
           NL - West-Nederland - Noord-Holland - Agglomeratie Haarlem
528030205
           NL - West-Nederland - Noord-Holland - Zaanstreek
528030206 NL - West-Nederland - Noord-Holland - Groot-Amsterdam
           NL - West-Nederland - Noord-Holland - Het Gooi en Vechtstreek
528030207
528030301 NL - West-Nederland - Zuid-Holland - Agglomeratie Leiden en Bollenstreek
528030302 NL – West-Nederland – Zuid-Holland – Agglomeratie 's-Gravenhage
528030303 NL - West-Nederland - Zuid-Holland - Delft en Westlijn
528030304
           NL - West-Nederland - Zuid-Holland - Oost-Zuid-Holland
528030305 NL - West-Nederland - Zuid-Holland - Groot-Rijnmond
528030306 NL - West-Nederland - Zuid-Holland - Zuidoost-Zuid-Holland
528030401 NL - West-Nederland - Zeeland - Zeeuwsch-Vlaanderen
528030402
           NL - West-Nederland - Zeeland - Overig Zeeland
528040101
           NL - Zuid-Nederland - Noord-Brabant - West-Noord-Brabant
528040102 NL - Zuid-Nederland - Noord-Brabant - Midden-Noord-Brabant
528040103
            NL - Zuid-Nederland - Noord-Brabant - Noordoost-Noord-Brabant
           NL - Zuid-Nederland - Noord-Brabant - Zuidoost-Noord-Brabant
528040104
            NL - Zuid-Nederland - Limburg (NL) - Noord-Limburg
528040201
528040202 NL - Zuid-Nederland - Limburg (NL) - Midden-Limburg
528040203 NL - Zuid-Nederland - Limburg (NL) - Zuid-Limburg
578000001 NO - Norge - Trøndelag - Sør-Trøndelag
578000002 NO - Norge - Trøndelag - Nord-Trøndelag
578000101 NO - Norge - Oslo og Akershus - Oslo
578000102 NO - Norge - Oslo og Akershus - Akershus
578000201
           NO - Norge - Hedmark og Oppland - Hedmark
578000202 NO - Norge - Hedmark og Oppland - Oplland
578000301 NO - Norge - Sør-Østlandet - Østfold
578000302
           NO - Norge - Sør-Østlandet - Buskerud
578000303 NO - Norge - Sør-Østlandet - Vestfold
578000304
           NO - Norge - Sør-Østlandet - Telemark
578000401 NO - Norge - Agder og Rogaland - Aust-Agder
578000402
           NO - Norge - Agder og Rogaland - Vest-Agder
578000403 NO - Norge - Agder og Rogaland - Rogaland
578000501
           NO - Norge - Vestlandet - Hordaland
578000502
           NO - Norge - Vestlandet - Sogn og Fjordane
578000503
           NO - Norge - Vestlandet Møre og Romsdal
578000701
           NO - Norge - Nord-Norge - Nordland
578000702 NO - Norge - Nord-Norge - Troms
578000703
           NO - Norge - Nord-Norge - Finnmark
616010103 PL - Region centralny - Lódzkie - Miasto Lódz
616010104 PL - Region centralny - Lódzkie - Lódzki
616010105 PL - Region centralny - Lódzkie - Piotrkowski
```

```
616010106 PL - Region centralny - Lódzkie - Sieradzki
616010107 PL - Region centralny - Lódzkie - Skierniewicki
616010201 PL - Region centralny - Mazowieckie - Ciechanowsko-plocki
616010202 PL - Region centralny - Mazowieckie - Ostrolecko-siedlecki
616010207 PL - Region centralny - Mazowieckie - Miasto Warszawa
616010208 PL - Region centralny - Mazowieckie - Radomski
616010209 PL - Region centralny - Mazowieckie - Warszawski-wschodni
616010210 PL - Region centralny - Mazowieckie - Warszawski-zachodni
616020103 PL - Region Poludinowy - Malopolskie - Miasto Kraków
616020104 PL - Region Poludinowy - Malopolskie - Krakowski
616020105 PL - Region Poludinowy - Malopolskie - Nowosadecki
616020106 PL - Region Poludinowy - Malopolskie - Oswiecimski
616020107 PL - Region Poludinowy - Malopolskie - Tarnowski
616020204 PL - Region Poludinowy - Slaskie - Czestochowski
616020205 PL - Region Poludinowy - Slaskie - Bielski
616020207 PL - Region Poludinowy - Slaskie - Rybnicki
616020208 PL – Region Poludinowy – Slaskie – Bytomski
616020209 PL - Region Poludinowy - Slaskie - Gliwicki
616020210 PL - Region Poludinowy - Slaskie - Katowicki
616020211 PL - Region Poludinowy - Slaskie - Sosnowiecki
616020212 PL - Region Poludinowy - Slaskie - Tyski
616030101 PL - Region Wschodni - Lubelskie - Bialski
616030102 PL - Region Wschodni - Lubelskie - Chelmsko-zamojski
616030104 PL - Region Wschodni - Lubelskie - Lubelski
616030105 PL - Region Wschodni - Lubelskie - Pulawski
616030203 PL - Region Wschodni - Podkarpackie - Krosnienski
616030204 PL - Region Wschodni - Podkarpackie - Przemyski
616030205 PL - Region Wschodni - Podkarpackie - Rzeszowski
616030206 PL – Region Wschodni – Podkarpackie – Tarnobrzeski
616030301 PL - Region Wschodni - Swietokrzyskie - Kielecki
616030302 PL - Region Wschodni - Swietokrzyskie - Sandomiersko-jedrzejowski
616030403 PL - Region Wschodni - Podlaskie - Bialostocki
616030404 PL - Region Wschodni - Podlaskie - Lomzynski
616030405 PL - Region Wschodni - Podlaskie - Suwalski
616040101 PL – Region Pólnocno-Zachodni – Wielkopolskie – Pilski
616040104 PL - Region Pólnocno-Zachodni - Wielkopolskie - Koninski
616040105 PL – Region Pólnocno-Zachodni – Wielkopolskie – Miasto Poznan
616040106 PL - Region Pólnocno-Zachodni - Wielkopolskie - Kaliski
616040107 PL – Region Pólnocno-Zachodni – Wielkopolskie – Leszczynski
616040108 PL - Region Pólnocno-Zachodni - Wielkopolskie - Poznanski
616040202 PL – Region Pólnocno-Zachodni – Zachodniopomorskie – Koszalinski
616040203 PL – Region Pólnocno-Zachodni – Zachodniopomorskie – Stargardzki
616040204 PL - Region Pólnocno-Zachodni - Zachodniopomorskie - Miasto Szczecin
616040205 PL - Region Pólnocno-Zachodni - Zachodniopomorskie - Szczecinski
616040301 PL - Region Pólnocno-Zachodni - Lubuskie - Gorzowski
616040302 PL – Region Pólnocno-Zachodni – Lubuskie – Zielonogórski
616050104 PL - Region Póludniowo-Zachodni - Dolnoslaskie - Miasto Wroclaw
616050105 PL – Region Póludniowo-Zachodni – Dolnoslaskie – Jeleniogórski
616050106 PL - Region Póludniowo-Zachodni - Dolnoslaskie - Legnicko-Glogowski
```

```
616050107 PL – Region Póludniowo-Zachodni – Dolnoslaskie – Walbrzyski
616050108 PL - Region Póludniowo-Zachodni - Dolnoslaskie - Wrocławski
616050201 PL - Region Póludniowo-Zachodni - Opolskie - Nyski
616050202 PL - Region Póludniowo-Zachodni - Opolskie - Opolski
616060103 PL - Region Pólnocny - Kujawsko-Pomorskie - Bydgosko-Torunski
616060104 PL - Region Pólnocny - Kujawsko-Pomorskie - Grudziadzki
616060105 PL – Region Pólnocny – Kujawsko-Pomorskie – Wlocławski
616060201 PL - Region Pólnocny - Warminsko-Mazurskie - Elblaski
616060202 PL – Region Pólnocny – Warminsko-Mazurskie – Olsztynski
616060203 PL – Region Pólnocny – Warminsko-Mazurskie – Elcki
616060301 PL - Region Pólnocny - Pomorskie - Slupski
616060303 PL – Region Pólnocny – Pomorskie – Trójmiejski
616060304 PL - Region Pólnocny - Pomorskie - Gdanski
616060305 PL – Region Pólnocny – Pomorskie – Starogardzki
620010101 PT - Continente - Northe - Minho-Lima
620010102 PT - Continente - Northe - Cávado
620010103 PT - Continente - Northe - Ave
620010104 PT - Continente - Northe - Grande Porto
620010105 PT - Continente - Northe - Tâmega
620010106 PT - Continente - Northe - Entre Douro e Vouga
620010107 PT - Continente - Northe - Douro
620010108 PT - Continente - Northe - Alto Trás-os-Montes
620010500 PT - Continente - Algarve
620010601 PT - Continente - Centro (P) - Baixo Vouga
620010602 PT - Continente - Centro (P) - Baixo Mondego
620010603 PT - Continente - Centro (P) - Pinhal Litoral
620010604 PT - Continente - Centro (P) - Pinhal Interior Norte
620010605 PT - Continente - Centro (P) - Dâo-Lafôes
620010606 PT - Continente - Centro (P) - Pinhal Interior Sul
620010607 PT - Continente - Centro (P) - Serra da Estrela
620010608 PT - Continente - Centro (P) - Beira Interior Norte
620010609 PT - Continente - Centro (P) - Beira Interior Sul
620010610 PT - Continente - Centro (P) - Cova de Beira
620010611 PT - Continente - Centro (P) - Oeste
620010612 PT - Continente - Centro (P) - Médio Tejo
620010701 PT - Continente - Lisboa - Grande Lisboa
620010702 PT - Continente - Lisboa - Península de Setúbal
620010801 PT - Continente - Alentejo - Alentejo Litoral
620010802 PT - Continente - Alentejo - Alto Alentejo
620010803 PT - Continente - Alentejo - Alentejo Central
620010804 PT - Continente - Alentejo - Baixo Alentejo
620010805 PT - Continente - Alentejo - Lezíra do Tejo
620020000 PT - Região Autónoma dos Açores
620030000 PT - Região Autónoma da Madeira
642010101 RO - Macroregiunea unu - Nord-Vest - Bihor
642010102 RO - Macroregiunea unu - Nord-Vest - Bistrita-Nasaud
642010103 RO - Macroregiunea unu - Nord-Vest - Cluj
642010104 RO - Macroregiunea unu - Nord-Vest - Maramures
642010105 RO - Macroregiunea unu - Nord-Vest - Satu Mare
```

```
642010106
           RO - Macroregiunea unu - Nord-Vest - Salaj
642010201
            RO - Macroregiunea unu - Centru - Alba
642010202
            RO - Macroregiunea unu - Centru - Brasov
642010203
            RO - Macroregiunea unu - Centru - Covasna
642010204
            RO - Macroregiunea unu - Centru - Harghita
642010205
            RO - Macroregiunea unu - Centru - Mures
642010206 RO - Macroregiunea unu - Centru - Sibiu
642020101
            RO - Macroregiunea doi - Nord-Est - Bacau
642020102 RO – Macroregiunea doi – Nord-Est – Botosani
642020103
            RO - Macroregiunea doi - Nord-Est - Iasi
642020104 RO - Macroregiunea doi - Nord-Est - Neamt
642020105
            RO - Macroregiunea doi - Nord-Est - Suceava
642020106 RO - Macroregiunea doi - Nord-Est - Vaslui
642020201
            RO - Macroregiunea doi - Sud-Est - Braila
642020202
           RO - Macroregiunea doi - Sud-Est - Buzau
642020203
            RO - Macroregiunea doi - Sud-Est - Constanta
642020204
            RO - Macroregiunea doi - Sud-Est - Galati
642020205 RO - Macroregiunea doi - Sud-Est - Tulcea
642020206 RO - Macroregiunea doi - Sud-Est - Vrancea
642030101
            RO - Macroregiunea trei - Sud-Muntenia - Arges
642030102
            RO – Macroregiunea trei – Sud-Muntenia – Calarasi
642030103
            RO - Macroregiunea trei - Sud-Muntenia - Dâmbovita
642030104
            RO - Macroregiunea trei - Sud-Muntenia - Giurgiu
642030105
            RO - Macroregiunea trei - Sud-Muntenia - Ialomita
642030106
            RO - Macroregiunea trei - Sud-Muntenia - Prahova
642030107
            RO – Macroregiunea trei – Sud-Muntenia – Teleorman
642030201
            RO - Macroregiunea trei - Bucuresti-Ilfov - Bucuresti
642030202 RO - Macroregiunea trei - Bucuresti-Ilfov - Ilfov
642040101
            RO - Macroregiunea patru - Sud-Vest Oltenia - Dolj
642040102 RO - Macroregiunea patru - Sud-Vest Oltenia - Gorj
642040103
            RO - Macroregiunea patru - Sud-Vest Oltenia - Mehedinti
642040104
            RO - Macroregiunea patru - Sud-Vest Oltenia - Olt
642040105 RO - Macroregiunea patru - Sud-Vest Oltenia - Vâlcea
642040201
            RO – Macroregiunea patru – Vest – Arad
642040202
            RO - Macroregiunea patru - Vest - Caras-Severin
642040203 RO - Macroregiunea patru - Vest - Hunedoara
642040204
            RO - Macroregiunea patru - Vest - Timis
643010101 RU - Central Federal District - Belgorodskaya oblast
643010102 RU - Central Federal District - Bryanskaya oblast
643010103 RU - Central Federal District - Vladimirskaya oblast
643010104 RU - Central Federal District - Voronezhskaya oblast
643010105 RU - Central Federal District - Ivanovskaya oblast
643010106 RU - Central Federal District - Kaluzhskaya oblast
643010107
            RU - Central Federal District - Kostromskaya oblast
643010108
            RU - Central Federal District - Kurskaya oblast
643010109
            RU - Central Federal District - Lipetskaya oblast
643010110 RU - Central Federal District - Moscow
643010111 RU - Central Federal District - Moscowskaya oblast
643010112 RU - Central Federal District - Orlovskaya oblast
```

```
643010113 RU - Central Federal District - Ryazanskaya oblast
643010114
            RU - Central Federal District - Smolenskaya oblast
643010115
            RU - Central Federal District - Tambovskaya oblast
643010116
            RU - Central Federal District - Tverskaya oblast
643010117
            RU - Central Federal District - Tulskaya oblast
643010118 RU - Central Federal District - Yaroslavskaya oblast
643020101 RU - North West federal district - Republic of Kareliya
643020102 RU - North West federal district - Republic of Komi
643020103 RU - North West federal district - Arkhangelskaya oblast
643020104 RU - North West federal district - Vologodskaya oblast
643020105 RU - North West federal district - Kaliningradskaya oblast
643020106
            RU - North West federal district - Leningradskaya oblast
643020107 RU - North West federal district - Murmanskaya oblast
643020108
            RU - North West federal district - Novgorodskaya oblast
643020109
            RU - North West federal district - Pskovskaya oblast
643020110 RU - North West federal district - St. Petersburg
643020111 RU - North West federal district - Nenetsky autonomous okrug
643030101 RU - South Federal district - Republic of Adygeya
643030102 RU - South Federal district - Republic of Dagestan
643030103 RU - South Federal district - Republic of Ingushetiya
643030104 RU - South Federal district - Republic of Kabardino-Balkariya
643030105 RU - South Federal district - Republic of Kalmykiya
643030106
            RU - South Federal district - Republic of Karachaevo-Cherkessia
643030107
            RU - South Federal district - Republic of North Ossetia
643030108
            RU - South Federal district - Chechen Republic
643030109 RU - South Federal district - Krasnodarsky krai
643030110 RU - South Federal district - Stavropolsky krai
643030111 RU - South Federal district - Astrakhanskaya oblast
643030112 RU - South Federal district - Volgogradskaya oblast
643030113 RU - South Federal district - Rostovskaya oblast
643040101
            RU - Privolzhsky federal district - Republic of Bashkortostan
643040102 RU - Privolzhsky federal district - Republic of Mary-El
643040103 RU - Privolzhsky federal district - Republic of Mordoviya
643040104 RU – Privolzhsky federal district – Republic of Tatarstan
643040105 RU - Privolzhsky federal district - Republic of Udmurtiya
643040106 RU - Privolzhsky federal district - Republic of Chuvashiya
643040107 RU - Privolzhsky federal district - Permsky krai
643040108 RU - Privolzhsky federal district - Kirovskaya oblast
643040109
            RU - Privolzhsky federal district - Nizhegorodskaya oblast
643040110 RU - Privolzhsky federal district - Orenburgskaya oblast
643040111
            RU - Privolzhsky federal district - Penzenskaya oblast
643040112 RU - Privolzhsky federal district - Samarskaya oblast
643040113 RU - Privolzhsky federal district - Saratovskaya oblast
643040114
            RU - Privolzhsky federal district - Ulyanovskaya oblast
643050101 RU - Urals federal district - Sverdlovskaya oblast
643050102
            RU - Urals federal district - Kurganskaya oblast
643050103 RU - Urals federal district - Tyumenskaya oblast
643050104 RU - Urals federal district - Chelyabinskaya oblast
643050105 RU - Urals federal district - Yamalo-Nenetsky autonomous okrug
```

```
643050106 RU - Urals federal district - Khanty-Mansiysky autonomous okrug
643060101
            RU - Siberian federal district - Altai Republic
643060102
            RU - Siberian federal district - Republic of Khakassia
643060103
            RU - Siberian federal district - Republic of Buryatia
643060104 RU - Siberian federal district - Tyva Republic
643060105 RU - Siberian federal district - Altai krai
643060106 RU - Siberian federal district - Krasnoyarsky krai
643060107
            RU - Siberian federal district - Irkutskaya oblast
643060108 RU - Siberian federal district - Kemerovskava oblast
643060109
            RU - Siberian federal district - Novosibirskaya oblast
643060110 RU - Siberian federal district - Omskaya oblast
643060111
            RU - Siberian federal district - Tomskaya oblast
643060112 RU - Siberian federal district - Chitinskaya oblast
643060113 RU – Siberian federal district – Aginsky Buryatsky autonomous okrug
643070101 RU - Far East federal district - Kamchatsky krai
643070102 RU - Far East federal district - Primorsky krai
643070103 RU - Far East federal district - Khabarovsky krai
643070104 RU - Far East federal district - Amur oblast
643070105 RU - Far East federal district - Magadanskaya oblast
643070106 RU - Far East federal district - Sakhalinskaya oblast
643070107 RU - Far East federal district - Yakutia
643070108
            RU - Far East federal district - Chukotsky autonomous okrug
643070109
            RU - Far East federal district - Evreyskaya autonomous okrug
            RS - Centralna Srbija - Grad Beograd - Grad Beograd
688010100
            RS - Centralna Srbija - Other Centralna Srbija - Macvanski okrug
688010201
688010202
            RS - Centralna Srbija - Other Centralna Srbija - Kolubarski okrug
688010203
            RS - Centralna Srbija - Other Centralna Srbija - Podunavski okrug
688010204
            RS - Centralna Srbija - Other Centralna Srbija - Branicevski okrug
688010205
            RS - Centralna Srbija - Other Centralna Srbija - Šumadijski okrug
688010206
            RS - Centralna Srbija - Other Centralna Srbija - Pomoravski okrug
            RS - Centralna Srbija - Other Centralna Srbija - Borski okrug
688010207
            RS - Centralna Srbija - Other Centralna Srbija - Zajecarski okrug
688010208
688010209
            RS - Centralna Srbija - Other Centralna Srbija - Zlatiborski ogrug
            RS - Centralna Srbija - Other Centralna Srbija - Moravicki okrug
688010210
            RS - Centralna Srbija - Other Centralna Srbija - Raški okrug
688010211
688010212
            RS - Centralna Srbija - Other Centralna Srbija - Rasinski okrug
            RS - Centralna Srbija - Other Centralna Srbija - Nišavski okrug
688010213
688010214
            RS - Centralna Srbija - Other Centralna Srbija - Tobolicki okrug
688010215
            RS - Centralna Srbija - Other Centralna Srbija - Pirotski okrug
688010216
            RS - Centralna Srbija - Other Centralna Srbija - Jablanicki okrug
688010217
            RS - Centralna Srbija - Other Centralna Srbija - Pcinjski okrug
688020101
            RS - Vojvodina - Severno-backi okrug
688020102
            RS - Vojvodina - Srednje-banatski okrug
688020103
            RS - Vojvodina - Severno-banatski okrug
688020104
            RS - Vojvodina - Južno-banatski okrug
688020105
            RS - Vojvodina - Zapadno-backi okrug
688020106
            RS - Vojvodina - Južno-backi okrug
688020107
            RS - Vojvodina - Sremski okrug
703000100 SK – Slovenská Republika – Bratislavský kraj
```

```
703000201
            SK - Slovenská Republika - Západné Slovensko - Trnavský kraj
703000202
            SK - Slovenská Republika - Západné Slovensko - Trenciansky kraj
703000203
            SK - Slovenská Republika - Západné Slovensko - Nitriansky kraj
703000301
            SK - Slovenská Republika - Stredné Slovensko - Žilinský kraj
            SK - Slovenská Republika - Stredné Slovensko - Banskobystrický kraj
703000302
            SK - Slovenská Republika - Východné Slovensko - Prešovský kraj
703000401
703000402 SK – Slovenská Republika – Východné Slovensko – Košický kraj
705000101
            SI - Slovenija - Vzhodna Slovenija - Pomurska
705000102 SI - Slovenija - Vzhodna Slovenija - Podrvaska
705000103 SI - Slovenija - Vzhodna Slovenija - Koroska
705000104 SI - Slovenija - Vzhodna Slovenija - Savinjska
705000105 SI - Slovenija - Vzhodna Slovenija - Zasavska
705000106 SI - Slovenija - Vzhodna Slovenija - Spodnjeposavska
            SI – Slovenija – Vzhodna Slovenija – Jugovzhodna Slovenija
705000107
705000108 SI – Slovenija – Vzhodna Slovenija – Notranjsko-kraska
705000201
            SI - Slovenija - Zahodna Slovenija - Osrednjeslovenska
705000202 SI – Slovenija – Zahodna Slovenija – Gorenjska
705000203 SI - Slovenija - Zahodna Slovenija - Goriska
705000204 SI - Slovenija - Zahodna Slovenija - Obalno-kraska
724010101 ES - Noroeste - Galicia - A Coruña
724010102 ES - Noroeste - Galicia - Lugo
724010103 ES - Noroeste - Galicia - Ourense
724010104 ES - Noroeste - Galicia - Pontevedra
724010200 ES - Noroeste - Principado de Asturias - Asturias
724010300 ES - Noroeste - Cantabria
724020101 ES - Noreste - País Vasco - Álava
724020102 ES - Noreste - País Vasco - Guipúzcoa
724020103 ES - Noreste - País Vasco - Vizcaya
724020200 ES - Noreste - Comunidad Foral de Navarra - Navarra
724020300 ES - Noreste - La Rioia
724020401 ES - Noreste - Aragón - Huesca
724020402 ES - Noreste - Aragón - Teruel
724020403 ES - Noreste - Aragón - Zaragoza
724030000 ES - Comunidad de Madrid - Madrid
724040101 ES - Centro (E) - Castilla y León - Ávilla
724040102 ES - Centro (E) - Castilla y León - Burgos
724040103 ES - Centro (E) - Castilla y León - León
724040104 ES - Centro (E) - Castilla y León - Palencia
724040105 ES - Centro (E) - Castilla y León - Salamanca
724040106 ES - Centro (E) - Castilla y León - Segovia
724040107 ES - Centro (E) - Castilla y León - Soria
724040108 ES - Centro (E) - Castilla y León - Vallodolid
724040109 ES - Centro (E) - Castilla y León - Zamora
724040201 ES - Centro (E) - Castilla-La Mancha - Albacete
724040202 ES - Centro (E) - Castilla-La Mancha - Ciudad Real
724040203 ES - Centro (E) - Castilla-La Mancha - Cuenca
724040204 ES - Centro (E) - Castilla-La Mancha - Guadalajara
724040205 ES - Centro (E) - Castilla-La Mancha - Toledo
724040301 ES - Centro (E) - Extremadura - Badajoz
```

```
724040302 ES - Centro (E) - Extremadura - Cáceres
724050101 ES - Este - Cataluña - Barcelona
724050102
            ES - Este - Cataluña - Girona
724050103
            ES - Este - Cataluña - Lleida
724050104 ES - Este - Cataluña - Tarragona
724050201 ES - Este - Comunidad Valenciana - Alicante/Alacant
724050202 ES - Este - Comunidad Valenciana - Castellón/Castelló
724050203 ES - Este - Comunidad Valenciana - Valencia/València
724050301 ES - Este - Illes Balears - Eivissa y Formentera
724050302 ES - Este - Illes Balears - Mallorca
724050303 ES - Este - Illes Balears - Menorca
724060101 ES - Sur - Andalucía - Almería
724060102 ES - Sur - Andalucía - Cádiz
724060103 ES - Sur - Andalucía - Córdoba
724060104 ES - Sur - Andalucía - Granada
724060105 ES - Sur - Andalucía - Huelva
724060106 ES - Sur - Andalucía - Jaén
724060107 ES - Sur - Andalucía - Málaga
724060108 ES - Sur - Andalucía - Sevilla
724060200 ES - Sur - Región de Murcia - Murcia
724060300 ES - Sur - Ciudad Autónoma de Ceuta - Ceuta
724060400 ES - Sur - Ciudad Autónoma de Melilla - Melilla
724070003
            ES - Canarias - El Hierro
724070004 ES - Canarias - Fuerteventura
724070005 ES - Canarias - Gran Canaria
724070006 ES - Canarias - La Gomera
724070007 ES - Canarias - La Palma
724070008 ES - Canarias - Lanzarote
724070009 ES - Canarias - Tenerife
752010100 SE - Östra Sverige - Stockholm - Stockholms län
752010201
            SE - Östra Sverige - Östra Mellansverige - Uppsala län
752010202 SE – Östra Sverige – Östra Mellansverige – Södermanlands län
            SE - Östra Sverige - Östra Mellansverige - Östergötlands län
752010203
752010204
            SE - Östra Sverige - Östra Mellansverige - Örebro län
            SE – Östra Sverige – Östra Mellansverige – Västmanlands län
752010205
752020101
            SE - Södra Sverige - Småland med öarna - Jönköpings län
752020102 SE – Södra Sverige – Småland med öarna – Kronobergs län
752020103 SE – Södra Sverige – Småland med öarna – Kalmar län
752020104
            SE - Södra Sverige - Småland med öarna - Gotlands län
752020201
            SE - Södra Sverige - Sydsverige - Blekinge län
752020204
            SE - Södra Sverige - Sydsverige - Skåne län
752020301
            SE - Södra Sverige - Västsverige - Hallands län
752020302
            SE - Södra Sverige - Västsverige - Västra Götalands län
752030101
            SE - Norra Sverige - Norra Mellansverige - Värmlands län
752030102 SE - Norra Sverige - Norra Mellansverige - Dalarnas län
752030103
            SE - Norra Sverige - Norra Mellansverige - Gävleborgs län
752030201
            SE - Norra Sverige - Mellersta Norrland - Västernorrlands län
752030202 SE - Norra Sverige - Mellersta Norrland - Jämtlands län
752030301 SE – Norra Sverige – Övre Norrland – Västerbottens län
```

```
752030302
           SE - Norra Sverige - Övre Norrland - Norrbottens län
756000101
            CH - Schweiz/Suisse/Svizzera - Région lémanique - Vaud
756000102
            CH - Schweiz/Suisse/Svizzera - Région lémanique - Valais
756000103
            CH - Schweiz/Suisse/Svizzera - Région lémanique - Genève
756000201
            CH - Schweiz/Suisse/Svizzera - Espace Mittelland - Bern
756000202 CH - Schweiz/Suisse/Svizzera - Espace Mittelland - Freiburg
756000203 CH - Schweiz/Suisse/Svizzera - Espace Mittelland - Solothurn
756000204
            CH - Schweiz/Suisse/Svizzera - Espace Mittelland - Neuchâtel
756000205 CH - Schweiz/Suisse/Svizzera - Espace Mittelland - Jura
756000301
           CH - Schweiz/Suisse/Svizzera - Nordwestschweiz - Basel-Stadt
756000302 CH - Schweiz/Suisse/Svizzera - Nordwestschweiz - Basel-Landschaft
756000303
            CH - Schweiz/Suisse/Svizzera - Nordwestschweiz - Aargau
756000400 CH - Schweiz/Suisse/Svizzera - Zürich
756000501
           CH - Schweiz/Suisse/Svizzera - Ostschweiz - Glarus
756000502 CH - Schweiz/Suisse/Svizzera - Ostschweiz - Schaffhausen
756000503 CH - Schweiz/Suisse/Svizzera - Ostschweiz - Appenzell Ausserrhoden
756000504 CH – Schweiz/Suisse/Svizzera – Ostschweiz – Appenzell Innerrhoden
756000505 CH - Schweiz/Suisse/Svizzera - Ostschweiz - St. Gallen
756000506 CH – Schweiz/Suisse/Svizzera – Ostschweiz – Graubünden
756000507
            CH - Schweiz/Suisse/Svizzera - Ostschweiz - Thurgau
756000601 CH - Schweiz/Suisse/Svizzera - Zentralschweiz - Luzern
756000602 CH - Schweiz/Suisse/Svizzera - Zentralschweiz - Uri
756000603
            CH - Schweiz/Suisse/Svizzera - Zentralschweiz - Schwyz
            CH - Schweiz/Suisse/Svizzera - Zentralschweiz - Obwalden
756000604
756000605
            CH - Schweiz/Suisse/Svizzera - Zentralschweiz - Nidwalden
756000606 CH - Schweiz/Suisse/Svizzera - Zentralschweiz - Zug
756000700 CH - Schweiz/Suisse/Svizzera - Ticino
792010000 TR - Istanbul
792020101 TR -Turkiye - Bati Marmara - Tekirdag - Tekirdag
792020102 TR -Turkiye - Bati Marmara - Tekirdag - Edirne
792020103 TR -Turkiye - Bati Marmara - Tekirdag - Kirklareli
792020201 TR -Turkiye - Bati Marmara - Balikesir - Balikesir
792020202 TR -Turkiye - Bati Marmara - Balikesir - Canakkale
792030100 TR - EGE - Izmir
792030201 TR - EGE - Aydin - Aydin
792030202 TR - EGE - Aydin - Denizli
792030203 TR - EGE - Aydin - Mugla
792030301 TR - EGE - Manisa - Manisa
792030302 TR - EGE - Manisa - Afyon
792030303 TR - EGE - Manisa - Kutahya
792030304 TR - EGE - Manisa - Usak
792040101 TR - Dogu Marmara - Bursa - Bursa
792040102 TR - Dogu Marmara - Bursa - Eskisehir
792040103 TR - Dogu Marmara - Bursa - Bilecik
792040201 TR - Dogu Marmara - Kocaeli - Kocaeli
792040202 TR - Dogu Marmara - Kocaeli - Sakarya
792040203 TR - Dogu Marmara - Kocaeli - Duzce
792040204 TR - Dogu Marmara - Kocaeli - Bolu
792040205 TR - Dogu Marmara - Kocaeli - Yalova
```

```
792050100 TR - Bati Anadolu - Ankara - Ankara
792050201
           TR - Bati Anadolu - Konya - Konya
792050202 TR – Bati Anadolu – Konya – Karaman
792060101
           TR - Akdeniz - Antalya - Antalya
792060102 TR - Akdeniz - Antalya - Isparta
792060103 TR - Akdeniz - Antalya - Burdur
792060201 TR - Akdeniz - Adana - Adana
792060202 TR - Akdeniz - Adana - Icel
792060301 TR - Akdeniz - Hatay - Hatay
792060302 TR - Akdeniz - Hatay - Kahramanmaras
792060303 TR - Akdeniz - Hatay - Osmaniye
792070101 TR - Orta Anadolu - Kirikkale - Kirikkale
792070102 TR - Orta Anadolu - Kirikkale - Aksaray
792070103 TR - Orta Anadolu - Kirikkale - Nigde
792070104 TR - Orta Anadolu - Kirikkale - Nevsehir
792070105 TR - Orta Anadolu - Kirikkale - Kirsehir
792070201 TR – Orta Anadolu – Kayseri – Kayseri
792070202 TR - Orta Anadolu - Kayseri - Sivas
792070203 TR - Orta Anadolu - Kayseri - Yozgat
792080101 TR - Bati Karadeniz - Zonguldak - Zonguldak
792080102 TR - Bati Karadeniz - Zonguldak - Karabuk
792080103 TR - Bati Karadeniz - Zonguldak - Batin
792080201
           TR - Bati Karadeniz - Kastamonu - Kastamonu
792080202 TR - Bati Karadeniz - Kastamonu - Cankiri
792080203 TR - Bati Karadeniz - Kastamonu - Sinop
792080301 TR - Bati Karadeniz - Samsun - Samsun
792080302 TR - Bati Karadeniz - Samsun - Tokat
792080303 TR - Bati Karadeniz - Samsun - Corum
792080304 TR - Bati Karadeniz - Samsun - Amasva
792090001 TR – Dogu Karadeniz – Trabzon – Trabzon
792090002 TR - Dogu Karadeniz - Trabzon - Ordu
792090003 TR – Dogu Karadeniz – Trabzon – Giresun
792090004 TR - Dogu Karadeniz - Trabzon - Rize
792090005 TR - Dogu Karadeniz - Trabzon - Artvin
792090006 TR - Dogu Karadeniz - Trabzon - Gumushane
792100101 TR - Kuzeydogu Anadolu - Erzurum - Erzurum
792100102 TR - Kuzeydogu Anadolu - Erzurum - Erzincan
792100103 TR - Kuzeydogu Anadolu - Erzurum - Bayburt
792100201 TR - Kuzeydogu Anadolu - Agri - Agri
792100202 TR - Kuzeydogu Anadolu - Agri - Kars
792100203 TR - Kuzeydogu Anadolu - Agri - Igdir
792100204 TR - Kuzeydogu Anadolu - Agri - Ardahan
792110101 TR - Ortadogu Anadolu - Malatya - Malatya
792110102 TR - Ortadogu Anadolu - Malatya - Elazig
792110103 TR - Ortadogu Anadolu - Malatya - Bingöl
792110104 TR - Ortadogu Anadolu - Malatya - Tunceli
792110201 TR - Ortadogu Anadolu - Van - Van
792110202 TR - Ortadogu Anadolu - Van - Mus
792110203 TR - Ortadogu Anadolu - Van - Bitlis
```

```
792110204 TR - Ortadogu Anadolu - Van - Hakkari
792120101 TR - Guneydogu Anadolu - Gaziantep - Gaziantep
792120102 TR - Guneydogu Anadolu - Gaziantep - Adiyaman
792120103 TR - Guneydogu Anadolu - Gaziantep - Kilis
792120201 TR - Guneydogu Anadolu - Sanliurfa - Sanliurfa
792120202 TR - Guneydogu Anadolu - Sanliurfa - Diyarbakir
792120301 TR – Guneydogu Anadolu – Mardin – Mardin
792120302 TR - Guneydogu Anadolu - Mardin - Batman
792120303 TR - Guneydogu Anadolu - Mardin - Sirnak
792120304 TR - Guneydogu Anadolu - Mardin - Siirt
804010100 UA - West - Volyn Oblast
804010200 UA - West - Zakarpattia Oblast
804010300 UA - West - Ivano-Frankovsk Oblast
804010400 UA - West - Lviv Oblast
804010500 UA - West - Rivne Oblast
804010600 UA - West - Ternopil Oblast
804010700 UA - West - Chernivtsi Oblast
804020100 UA - Centre - Vinnytsia Oblast
804020200 UA - Centre - Kirovohrad Oblast
804020300 UA - Centre - Poltava Oblast
804020400 UA - Centre - Khmelnytskyi Oblast
804020500 UA - Centre - Cherkasy Oblast
804030100 UA - North - Zhytomyr Oblast
804030200 UA - North - Kyiv
804030300 UA - North - Kyiv Oblast
804030400 UA - North - Sumy Oblast
804030500 UA - North - Chernihiv Oblast
804040100 UA - East - Dnipropetrovsk
804040200 UA - East - Dnipropetrovsk Oblast
804040300 UA - East - Zaporizhia
804040400 UA - East - Zaporizhia Oblast
804040500 UA - East - Donetsk
804040600 UA - East - Donetsk Oblast
804040700 UA - East - Luhansk Oblast
804040800 UA - East - Kharkiv
804040900 UA - East - Kharkiv Oblast
804050100 UA - South - Sevastopol
804050200 UA - South - Autonomous Republic of Crimea
804050300 UA - South - Mykolaiv Oblast
804050400 UA - South - Odesa
804050500 UA - South - Odesa Oblast
804050600
           UA - South - Kherson Oblast
807000001
           MK - Poranesnata jugoslovenska Republika Makedonija - Vardarski
807000002
           MK - Poranesnata jugoslovenska Republika Makedonija - Istocen
807000003
           MK - Poranesnata jugoslovenska Republika Makedonija - Jugozapaden
807000004
           MK - Poranesnata jugoslovenska Republika Makedonija - Jugoistocen
           MK - Poranesnata jugoslovenska Republika Makedonija - Pelagoniski
807000005
807000006
           MK - Poranesnata jugoslovenska Republika Makedonija - Polski
807000007
           MK - Poranesnata jugoslovenska Republika Makedonija - Severoistocen
```

```
807000008
            MK - Poranesnata jugoslovenska Republika Makedonija - Skopski
826120101
            GB - North East (England) - Tees Valley and Durham - Hartlepool and Stockton-on-Tees
826120102
            GB - North East (England) - Tees Valley and Durham - South Teesside
826120103
            GB - North East (England) - Tees Valley and Durham - Darlington
826120104
            GB - North East (England) - Tees Valley and Durham - Durham CC
826120201
            GB - North East (England) - Northumberland and Tyne and Wear - Northumberland
826120202
            GB - North East (England) - Northumberland and Tyne and Wear - Tyneside
826120203
            GB - North East (England) - Northumberland and Tyne and Wear - Sunderland
826130101
            GB - North West (England) - Cumbria - West Cumbria
826130102
            GB - North West (England) - Cumbria - East Cubria
826130201
            GB - North West (England) - Cheshire - Halton and Warrington
826130202
            GB - North West (England) - Cheshire - Cheshire CC
826130301
            GB - North West (England) - Greater Manchester - Greater Manchester South
826130302
            GB - North West (England) - Greater Manchester - Greater Manchester North
826130401
            GB - North West (England) - Lancashire - Blackburn with Darwen
826130402 GB - North West (England) - Lancashire - Blackpool
826130403 GB - North West (England) - Lancashire - Lancashire CC
826130501
            GB - North West (England) - Merseyside - East Merseyside
826130502
            GB - North West (England) - Merseyside - Liverpool
826130503
            GB - North West (England) - Merseyside - Sefton
826130504
            GB - North West (England) - Merseyside - Wirral
826140101
            GB - Yorkshire and the Humber - East Yorkshire and Northern Lincolnshire - Kingston upon Hull, City of
826140102
            GB - Yorkshire and the Humber - East Yorkshire and Northern Lincolnshire - East Riding of Yorkshire
826140103
            GB - Yorkshire and the Humber - East Yorkshire and Northern Lincolnshire - North and North East Lincolnshire
826140201
            GB - Yorkshire and the Humber - North Yorkshire - York
826140202
            GB - Yorkshire and the Humber - North Yorkshire - North Yorkshire CC
826140301
            GB - Yorkshire and the Humber - South Yorkshire - Barnsley, Doncaster and Rotherham
826140302
            GB - Yorkshire and the Humber - South Yorkshire - Sheffield
826140401
            GB - Yorkshire and the Humber - West Yorkshire - Bradford
826140402
            GB - Yorkshire and the Humber - West Yorkshire - Leeds
826140403
            GB - Yorkshire and the Humber - West Yorkshire - Calderdale, Kirklees and Wakefield
826150101
            GB - East Midlands (England) - Derbyshire and Nottinghamshire - Derby
826150102 GB - East Midlands (England) - Derbyshire and Nottinghamshire - East Derbyshire
826150103
            GB - East Midlands (England) - Derbyshire and Nottinghamshire - South and West Derbyshire
826150104
            GB - East Midlands (England) - Derbyshire and Nottinghamshire - Nottingham
826150105
            GB - East Midlands (England) - Derbyshire and Nottinghamshire - North Nottinghamshire
826150106
            GB - East Midlands (England) - Derbyshire and Nottinghamshire - South Nottinghamshire
826150201
            GB - East Midlands (England) - Leicestershire, Rutland and Northamptonshire - Leicester
826150202
            GB - East Midlands (England) - Leicestershire, Rutland and Northamptonshire - Leicestershire CC and Rutland
826150203
            GB - East Midlands (England) - Leicestershire, Rutland and Northamptonshire - Northamptonshire
826150300
            GB - East Midlands (England) - Lincolnshire
826160101
            GB - West Midlands (England) - Herefordshire, Worcestershire and Warwickshire - Herefordshire, County of
826160102
            GB - West Midlands (England) - Herefordshire, Worcestershire and Warwickshire - Worcestershire
826160103
            GB - West Midlands (England) - Herefordshire, Worcestershire and Warwickshire - Warwickshire
826160201
            GB - West Midlands (England) - Shropshire and Staffordshire - Telford and Wrekin
826160202
            GB - West Midlands (England) - Shropshire and Staffordshire - Shropshire CC
826160203
            GB - West Midlands (England) - Shropshire and Staffordshire - Stoke-on-Trent
826160204
            GB - West Midlands (England) - Shropshire and Staffordshire - Staffordshire CC
826160301 GB - West Midlands (England) - West Midlands - Birmingham
```

```
826160302 GB - West Midlands (England) - West Midlands - Solihull
826160303
            GB - West Midlands (England) - West Midlands - Coventry
826160304
            GB - West Midlands (England) - West Midlands - Dudley and Sandwell
826160305
            GB - West Midlands (England) - West Midlands - Walsall and Wolverhampton
826170101
            GB - East of England - East Anglia - Peterborough
826170102 GB - East of England - East Anglia - Cambridgeshire CC
826170103 GB - East of England - East Anglia - Norfolk
826170104
            GB - East of England - East Anglia - Suffolk
826170201 GB - East of England - Bedfordshire and Hertfordshire - Luton
826170202 GB - East of England - Bedfordshire and Hertfordshire - Bedfordshire CC
826170203
            GB - East of England - Bedfordshire and Hertfordshire - Hertfordshire
826170301
            GB - East of England - Essex - Southend-on-Sea
826170302 GB - East of England - Essex - Thurrock
826170303 GB - East of England - Essex - Essex CC
826180101
            GB - London - Inner London - West
826180102 GB - London - Inner London - East
826180201
            GB - London - Outer London - East and North East
826180202 GB - London - Outer London - South
826180203 GB - London - Outer London - West and North West
826190101
            GB - South East (England) - Berkshire, Buckinghamshire and Oxfordshire - Berkshire
826190102
            GB - South East (England) - Berkshire, Buckinghamshire and Oxfordshire - Milton Keynes
826190103
            GB - South East (England) - Berkshire, Buckinghamshire and Oxfordshire - Buckinghamshire CC
826190104
            GB - South East (England) - Berkshire, Buckinghamshire and Oxfordshire - Oxfordshire
826190201
            GB - South East (England) - Surrey, East and West Sussex - Brighton and Hove
826190202
            GB - South East (England) - Surrey, East and West Sussex - East Sussex CC
826190203
            GB - South East (England) - Surrey, East and West Sussex - Surrey
826190204
            GB - South East (England) - Surrey, East and West Sussex - West Sussex
826190301
            GB - South East (England) - Hampshire and Isle of Wight - Portsmouth
826190302
            GB - South East (England) - Hampshire and Isle of Wight - Southampton
826190303
            GB - South East (England) - Hampshire and Isle of Wight - Hampshire CC
826190304
            GB - South East (England) - Hampshire and Isle of Wight - Isle of Wight
826190401
            GB - South East (England) - Kent - Medway
826190402
            GB - South East (England) - Kent - Kent CC
            GB - South West (England) - Gloucestershire, Wiltshire and Bristol/Bath area - Bristol, City of
826200101
826200102
            GB - South West (England) - Gloucestershire, Wiltshire and Bristol/Bath area - Bath and North East Somerset, Nort
826200103
            GB - South West (England) - Gloucestershire, Wiltshire and Bristol/Bath area - Gloucestershire
826200104
            GB - South West (England) - Gloucestershire, Wiltshire and Bristol/Bath area - Swindon
826200105
            GB - South West (England) - Gloucestershire, Wiltshire and Bristol/Bath area - Wiltshire CC
826200201
            GB - South West (England) - Dorset and Somerset - Bournemouth and Poole
826200202
            GB - South West (England) - Dorset and Somerset - Dorset CC
826200203
            GB - South West (England) - Dorset and Somerset - Somerset
826200300
            GB - South West (England) - Cornwall and Isles of Scilly
826200401
            GB - South West (England) - Devon - Plymouth
            GB - South West (England) - Devon - Torbay
826200402
826200403
            GB - South West (England) - Devon - Devon CC
826210101
            GB - Wales - West Wales and The Valleys - Isle of Anglesey
826210102 GB - Wales - West Wales and The Valleys - Gwynedd
826210103 GB - Wales - West Wales and The Valleys - Conwy and Denbighshire
826210104 GB - Wales - West Wales and The Valleys - South West Wales
```

```
826210105 GB - Wales - West Wales and The Valleys - Central Valleys
826210106 GB - Wales - West Wales and The Valleys - Gwent Valleys
826210107
            GB - Wales - West Wales and The Valleys - Bridgend and Neath Port Talbot
            GB - Wales - West Wales and The Valleys - Swansea
826210108
826210201 GB - Wales - East Wales - Monmouthshire and Newport
826210202 GB - Wales - East Wales - Cardiff and Vale of Glamorgan
826210203 GB - Wales - East Wales - Flintshire and Wrexham
826210204 GB - Wales - East Wales - Powys
826220201 GB - Scotland - Eastern Scotland - Angus and Dundee City
826220202 GB - Scotland - Eastern Scotland - Clackmannanshire and Fife
826220203 GB - Scotland - Eastern Scotland - East Lothian and Midlothian
826220204
            GB - Scotland - Eastern Scotland - Scottish Borders
826220205 GB - Scotland - Eastern Scotland - Edinburgh, City of
826220206 GB - Scotland - Eastern Scotland - Falkirk
826220207
            GB - Scotland - Eastern Scotland - Perth & Kinross and Stirling
826220208 GB - Scotland - Eastern Scotland - West Lothian
826220301 GB - Scotland - South Western Scotland - East Dunbartonshire, West Dunbartonshire and Helensburgh & Lomond
826220302 GB - Scotland - South Western Scotland - Dumfries & Galloway
826220303 GB - Scotland - South Western Scotland - East Ayrshire and North Ayrshire mainland
826220304
            GB - Scotland - South Western Scotland - Glasgow City
826220305
            GB - Scotland - South Western Scotland - Inverclyde, East Renfrewshire and Renfrewshire
826220306
            GB - Scotland - South Western Scotland - North Lanarkshire
826220307
            GB - Scotland - South Western Scotland - South Ayrshire
            GB - Scotland - South Western Scotland - South Lanarkshire
826220308
826220500
            GB - Scotland - North Eastern Scotland - Aberdeen City and Aberdeenshire
826220601
            GB - Scotland - Highlands and Islands - Caithness & Sutherland and Ross & Cromarty
826220602
            GB - Scotland - Highlands and Islands - Inverness & Nairn and Moray, Badenoch & Strathspey
826220603
            GB - Scotland - Highlands and Islands - Lochaber, Skye & Lochalsh, Arran & Cumbrae and Argyll & Bute
826220604
            GB - Scotland - Highlands and Islands - Eilean Siar (Western Isles)
826220605
            GB - Scotland - Highlands and Islands - Orkney Islands
826220606
            GB - Scotland - Highlands and Islands - Shetland Islands
826230001
            NIR - Northern Ireland - Belfast
826230002 NIR - Northern Ireland - Outer Belfast
826230003 NIR - Northern Ireland - East of Northern Ireland
826230004 NIR - Northern Ireland - North of Northern Ireland
826230005 NIR - Northern Ireland - West and South of Northern Ireland
1001000001 KOS - Kosovo - Komuna e Gjakovës/Đakovicki okrug
1001000002 KOS - Kosovo - Komuna e Gjilanit/Gnjilanski okrug
1001000003 KOS - Kosovo - Komuna e Mitrovicës/Kosovskomitrovacki okrug
1001000004 KOS - Kosovo - Komuna e Pejës/Pecki okrug
1001000005 KOS - Kosovo - Komuna e Prishtinës/Prištinski okrug
1001000006 KOS - Kosovo - Komuna e Prizrenit/Prizrenski okrug
1001000007 KOS - Kosovo - Komuna e Ferizajt/Uroševacki okrug
1961000000 CY - Kypros / Kibris
```

Note:

Verbatim recorded variable post-coded according to NUTS-3 code. Data for original string variable is not available in this dataset. For detailed information about country-specific classification and correspondence with NUTS, see Method Report.

```
v368b_N2 - Region at age 14: NUTS-2 code (Q133)
HARMONIZED VARIABLE - NUTS 2
```

Q133

Where did you live when you were about 14 years old?

Code:

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 800001 AL Albania South-Albania
- 800002 AL Albania North-Albania
- 3100000 AZ Azerbaijan
- 4000101 AT Ostösterreich Burgenland (A)
- 4000102 AT Ostösterreich Niederösterreich
- 4000103 AT Ostösterreich Wien
- 4000201 AT Südösterreich Kärnten
- 4000202 AT Südösterreich Steiermark
- 4000301 AT Westöstereich Oberösterreich
- 4000302 AT Westöstereich Salzburg
- 4000303 AT Westöstereich Tirol
- 4000304 AT Westöstereich Vorarlberg
- 5100001 AM Armenia Yerevan
- 5100002 AM Armenia Rest of Armenia
- 5600100 BE Région de Bruxelles-capitale/Brussels hoofdstedelijk gewest
- 5600201 BE Vlaams gewest Prov. Antwerpen
- 5600202 BE Vlaams gewest Prov. Limburg (B)
- 5600203 BE Vlaams gewest Prov. Oost-Vlaanderen
- 5600204 BE Vlaams gewest Prov. Vlaams-Brabant
- 5600205 BE Vlaams gewest Prov. West-Vlaanderen
- 5600301 BE Région Wallonne Prov. Brabant Wallon
- 5600302 BE Région Wallonne Prov. Hainaut
- 5600303 BE Région Wallonne Prov. Liège
- 5600304 BE Région Wallonne Prov. Luxembourg (B)
- 5600305 BE Région Wallonne Prov. Namur
- 7000001 BA Bosna i Hercegovina Federacija Bosne i Hercegovine
- 7000002 BA Bosna i Hercegovina Republika Srpska
- 7000003 BA Bosna i Hercegovina Brcko
- 10000301 BG Severna i iztochna Bulgaria Severozapaden
- 10000302 BG Severna i iztochna Bulgaria Severen tsentralen
- 10000303 BG Severna i iztochna Bulgaria Severoiztochen
- 10000304 BG Severna i iztochna Bulgaria Yugoiztochen
- 10000401 BG Yugozapadna i yuzhna tsentralna Bulgaria Yugozapaden
- 10000402 BG Yugozapadna i yuzhna tsentralna Bulgaria Yuzhen tsentralen
- 11200000 BY Belarus

```
19100001
          HR - Hrvatska - Sjeverozapadna Hrvatska
19100002
          HR - Hrvatska - Sredisnja i Istocna (Panonska) Hrvatska
19100003
         HR - Hrvatska - Jadranska Hrvatska
19610000
          CY - Kypros / Kibris
20300001 CZ - Ceska Republika - Praha
20300002 CZ - Ceska Republika - Stredni Cechy
20300003 CZ - Ceska Republika - Jihozapad
20300004 CZ - Ceska Republika - Severozapad
20300005 CZ - Ceska Republika - Severovychod
20300006 CZ - Ceska Republika - Jihovychod
20300007 CZ - Ceska Republika - Stredni Morava
20300008 CZ - Ceska Republika - Moravskoslezsko
20800001
          DK - Danmark - Hovedstaden
20800002 DK - Danmark- Sjælland
20800003 DK – Danmark – Syddanmark
20800004 DK - Danmark - Midtjylland
20800005 DK - Danmark - Nordjylland
23300000 EE - Eesti
24600103 FI - Manner-Suomi - Itä-Suomi
24600108 FI - Manner-Suomi - Etalä-Suomi
24600109 FI - Manner-Suomi - Länsi-Suomi
24600110 FI - Manner-Suomi - Pohjois-Suomi
24600200 FI - Åland
25000100 FR - Île de France
25000201 FR - Bassin Parisien - Champagne-Ardenne
25000202 FR - Bassin Parisien - Champagne-Ardenne - Picardie
25000203 FR - Bassin Parisien - Champagne-Ardenne - Haute-Normandie
25000204 FR - Bassin Parisien - Champagne-Ardenne - Centre
25000205 FR - Bassin Parisien - Champagne-Ardenne - Basse-Normandie
25000206 FR - Bassin Parisien - Champagne-Ardenne - Bourgogne
25000300 FR - Nord-pas-de-Calais
25000401 FR - Est - Lorraine
25000402 FR - Est - Alsace
25000403 FR - Est - Franche-Comté
25000501 FR - Ouest - Pays de la Loire
25000502 FR - Ouest - Bretagne
25000503 FR - Ouest - Poitou-Charentes
25000601 FR - Sud-Ouest - Aquitaine
25000602 FR - Sud-Ouest - Midi-Pyrénées
25000603 FR - Sud-Ouest - Limousin
25000701 FR - Centre-Est - Rhône-Alpes
25000702 FR - Centre-Est - Auvergne
25000801
          FR - Méditerranée - Languedoc-Roussillon
25000802 FR - Méditerranée - Provence-Alpes-Côte d'Azur
25000803 FR - Méditerranée - Corse
25000901
          FR - Départements d'outre-mer - Guadeloupe
25000902 FR – Départements d'outre-mer – Martinique
25000903
          FR - Départements d'outre-mer - Guyane
25000904 FR - Départements d'outre-mer - Réunion
```

```
26800000
          GE - Georgia
27600101
          DE - Baden-Württemberg - Stuttgart
27600102
          DE - Baden-Württemberg - Karlsruhe
27600103
          DE - Baden-Württemberg - Freiburg
27600104 DE - Baden-Württemberg - Tübingen
27600201 DE - Bayern - Oberbayern
27600202 DE - Bayern - Niederbayern
27600203 DE - Bayern - Oberpfalz
27600204 DE - Bayern - Oberfranken
27600205 DE - Bayern - Mittelfranken
27600206 DE - Bayern - Unterfranken
27600207 DE - Bayern - Schwaben
27600300 DE - Berlin
27600401
          DE - Brandenburg - Brandenburg - Nordost
27600402 DE - Brandenburg - Brandenburg Südwest
27600500 DE - Bremen
27600600 DE - Hamburg
27600701 DE - Hessen - Darmstadt
27600702 DE - Hessen - Gießen
27600703 DE - Hessen - Kassel
27600800 DE - Mecklenburg-Vorpommern
27600901
          DE - Niedersachsen - Braunschweig
27600902
          DE - Niedersachsen - Hannover
27600903
         DE - Niedersachsen - Lüneburg
27600904
          DE - Niedersachsen - Weser-Ems
27601001 DE - Nordrhein-Westfalen - Düsseldorf
27601002 DE - Nordrhein-Westfalen - Köln
27601003 DE - Nordrhein-Westfalen - Münster
27601004 DE - Nordrhein-Westfalen - Detmold
27601005 DE - Nordrhein-Westfalen - Arnsberg
27601101 DE - Rheinland-Pfalz - Koblenz
27601102 DE - Rheinland-Pfalz - Trier
27601103 DE - Rheinland-Pfalz - Rheinhessen-Pfalz
27601200 DE - Saarland
27601301 DE - Sachsen - Chemnitz
27601302 DE - Sachsen - Dresden
27601303 DE - Sachsen - Leipzig
27601400 DE - Sachsen-Anhalt
27601500 DE - Schleswig-Holstein
27601600 DE - Thüringen
30000101 GR - Voreia Ellada - Anatoliki Makedonia, Thraki
30000102 GR - Voreia Ellada - Kentriki Makedonia
30000103
         GR - Voreia Ellada - Dytiki Makedonia
30000104
          GR - Voreia Ellada - Thessalia
30000201 GR - Kentriki Ellada - Ipeiros
30000202 GR - Kentriki Ellada - Ionia Nisia
30000203 GR - Kentriki Ellada - Dytiki Ellada
          GR - Kentriki Ellada - Sterea Ellada
30000204
30000205 GR - Kentriki Ellada - Peloponnisos
```

```
30000300
           GR - Attiki
30000401
           GR - Nisia Aigaiou, Kriti - Voreio Aigaio
30000402
           GR - Nisia Aigaiou, Kriti - Notio Aigaio
30000403
           GR - Nisia Aigaiou, Kriti - Kriti
34800100 HU - Közép-Magyarország
34800201 HU - Dunántúl - Közép-Dunántúl
34800202 HU - Dunántúl - Nyugat-Dunántúl
34800203 HU - Dunántúl - Dél-Dunántúl
34800301 HU – Alföld és Észak – Észak-Magyarország
34800302 HU – Alföld és Észak – Észak-Alföld
34800303 HU - Alföld és Észak - Dél-Alföld
35200000
          IS - Ísland
37200001
          IE - Ireland - Border, Midland and Western
37200002
          IE - Ireland - Southern and Eastern
38001201 IT - Nord-Ovest - Piemonte
38001202 IT - Nord-Ovest - Valle d'Aosta/Vallée d'Aoste
38001203 IT - Nord-Ovest - Liguria
38001204 IT - Nord-Ovest - Lombardia
38001301 IT – Nord-Est – Provincia Autonoma Bolzano/Bozen
38001302 IT - Nord-Est - Provincia Autonoma Trento
38001303 IT - Nord-Est - Veneto
38001304 IT - Nord-Est - Friuli-Venezia Giulia
38001305 IT - Nord-Est - Emilia-Romagna
38001401 IT - Centro (I) - Toscana
38001402 IT - Centro (I) - Umbria
38001403 IT - Centro (I) - Marche
38001404 IT - Centro (I) - Lazio
38001501 IT - Sud - Abruzzo
38001502 IT - Sud - Molise
38001503 IT - Sud - Campania
38001504 IT - Sud - Puglia
38001505 IT - Sud - Basilicata
38001506 IT - Sud - Calabria
38001601 IT - Isole - Sicilia
38001602 IT - Isole - Sardegna
42800000 LV - Latvija
44000000 LT - Lietuva
44200000 LU - Luxembourg (Grand-Duché)
47000000 MT- Malta
49800000 MD - Moldova
49900001 ME - Montenegro - North
49900002 ME - Montenegro - Center
49900003 ME - Montenegroc - South
52800101 NL - Noord-Nederland - Groningen
52800102 NL – Noord-Nederland – Friesland (NL)
52800103 NL - Noord-Nederland - Drenthe
52800201 NL - Oost-Nederland - Overijssel
52800202 NL - Oost-Nederland - Gelderland
52800203 NL - Oost-Nederland - Flevoland
```

52800301 NL - West-Nederland - Utrecht 52800302 NL - West-Nederland - Noord-Holland 52800303 NL - West-Nederland - Zuid-Holland 52800304 NL - West-Nederland - Zeeland 52800401 NL - Zuid-Nederland - Noord-Brabant 52800402 NL – Zuid-Nederland – Limburg (NL) 57800001 NO - Norge - Oslo og Akershus 57800002 NO - Norge - Hedmark og Oppland 57800003 NO - Norge - Sør-Østlandet 57800004 NO - Norge - Agder og Rogaland 57800005 NO - Norge - Vestlandet 57800006 NO - Norge - Trøndelag 57800007 NO - Norge - Nord-Norge 61600101 PL - Region centralny - Lódzkie 61600102 PL - Region centralny - Mazowieckie 61600201 PL - Region Poludinowy - Malopolskie 61600202 PL - Region Poludinowy - Slaskie 61600301 PL - Region Wschodni - Lubelskie 61600302 PL - Region Wschodni - Podkarpackie 61600303 PL - Region Wschodni - Swietokrzyskie 61600304 PL - Region Wschodni - Podlaskie 61600401 PL - Region Pólnocno-Zachodni - Wielkopolskie 61600402 PL - Region Pólnocno-Zachodni - Zachodniopomorskie 61600403 PL - Region Pólnocno-Zachodni - Lubuskie 61600501 PL - Region Póludniowo-Zachodni - Dolnoslaskie 61600502 PL – Region Póludniowo-Zachodni – Opolskie 61600601 PL - Region Pólnocny - Kujawsko-Pomorskie 61600602 PL - Region Pólnocny - Warminsko-Mazurskie 61600603 PL – Region Pólnocny – Pomorskie 62000101 PT - Continente - Northe 62000105 PT - Continente - Algarve 62000106 PT - Continente - Centro (P) 62000107 PT - Continente - Lisboa 62000108 PT - Continente - Alentejo 62000200 PT - Região Autónoma dos Açores 62000300 PT - Região Autónoma da Madeira 64200101 RO - Macroregiunea unu - Nord-Vest 64200102 RO - Macroregiunea unu - Centru 64200201 RO - Macroregiunea doi - Nord-Est 64200202 RO - Macroregiunea doi - Sud-Est 64200301 RO - Macroregiunea trei - Sud-Muntenia 64200302 RO - Macroregiunea trei - Bucuresti-Ilfov 64200401 RO - Macroregiunea patru - Sud-Vest Oltenia 64200402 RO - Macroregiunea patru - Vest 64300101 RU - Central Federal District 64300201 RU - North West federal district 64300301 RU - South Federal district 64300401 RU - Privolzhsky federal district 64300501 RU - Urals federal district

64300601 RU - Siberian federal district 64300701 RU - Far East federal district 68800101 RS - Centralna Srbija - Grad Beograd 68800102 RS - Centralna Srbija - Other Centralna Srbija 68800201 RS - Vojvodina 70300001 SK – Slovenská Republika – Bratislavský kraj 70300002 SK – Slovenská Republika – Západné Slovensko 70300003 SK - Slovenská Republika - Stredné Slovensko 70300004 SK - Slovenská Republika - Východné Slovensko 70500001 SI - Slovenija - Vzhodna Slovenija 70500002 SI - Slovenija - Zahodna Slovenija 72400101 ES - Noroeste - Galicia 72400102 ES - Noroeste - Principado de Asturias 72400103 ES - Noroeste - Cantabria 72400201 ES - Noreste - País Vasco 72400202 ES - Noreste - Comunidad Foral de Navarra 72400203 ES - Noreste - La Rioja 72400204 ES - Noreste - Aragón 72400300 ES - Comunidad de Madrid 72400401 ES - Centro (E) - Castilla y León 72400402 ES - Centro (E) - Castilla-La Mancha 72400403 ES - Centro (E) - Extremadura 72400501 ES - Este - Cataluña 72400502 ES - Este - Comunidad Valenciana 72400503 ES - Este - Illes Balears 72400601 ES - Sur - Andalucía 72400602 ES - Sur - Región de Murcia 72400603 ES - Sur - Ciudad Autónoma de Ceuta 72400604 ES - Sur - Ciudad Autónoma de Melilla 72400700 ES - Canarias 75200101 SE - Östra Sverige - Stockholm 75200102 SE - Östra Sverige - Östra Mellansverige 75200201 SE - Södra Sverige - Småland med öarna 75200202 SE - Södra Sverige - Sydsverige 75200203 SE – Södra Sverige – Västsverige 75200301 SE - Norra Sverige - Norra Mellansverige 75200302 SE - Norra Sverige - Mellersta Norrland 75200303 SE - Norra Sverige - Övre Norrland 75600001 CH - Schweiz/Suisse/Svizzera - Région lémanique 75600002 CH - Schweiz/Suisse/Svizzera - Espace Mittelland 75600003 CH - Schweiz/Suisse/Svizzera - Nordwestschweiz 75600004 CH - Schweiz/Suisse/Svizzera - Zürich 75600005 CH - Schweiz/Suisse/Svizzera - Ostschweiz 75600006 CH - Schweiz/Suisse/Svizzera - Zentralschweiz 75600007 CH - Schweiz/Suisse/Svizzera - Ticino 79200100 TR - Istanbul 79200201 TR - Turkiye - Bati Marmara - Tekirdag 79200202 TR - Turkiye - Bati Marmara - Balikesir 79200301 TR - EGE - Izmir

79200302 TR - EGE - Aydin 79200303 TR - EGE - Manisa 79200401 TR - Dogu Marmara - Bursa 79200402 TR - Dogu Marmara - Kocaeli 79200501 TR - Bati Anadolu - Ankara 79200502 TR - Bati Anadolu - Konya 79200601 TR - Akdeniz - Antalya 79200602 TR - Akdeniz - Adana 79200603 TR - Akdeniz - Hatay 79200701 TR - Orta Anadolu - Kirikkale 79200702 TR - Orta Anadolu - Kayseri 79200801 TR - Bati Karadeniz - Zonguldak 79200802 TR - Bati Karadeniz - Kastamonu 79200803 TR - Bati Karadeniz - Samsun 79200900 TR - Dogu Karadeniz - Trabzon 79201001 TR - Kuzeydogu Anadolu - Erzurum 79201002 TR - Kuzeydogu Anadolu - Agri 79201101 TR - Ortadogu Anadolu - Malatya 79201102 TR - Ortadogu Anadolu - Van 79201201 TR - Guneydogu Anadolu - Gaziantep 79201202 TR - Guneydogu Anadolu - Sanliurfa 79201203 TR - Guneydogu Anadolu - Mardin 80400101 UA - West - Volyn Oblast 80400102 UA - West - Zakarpattia Oblast 80400103 UA - West - Ivano-Frankovsk Oblast 80400104 UA - West - Lviv Oblast 80400105 UA - West - Rivne Oblast 80400106 UA - West - Ternopil Oblast 80400107 UA - West - Chernivtsi Oblast 80400201 UA - Centre - Vinnytsia Oblast 80400202 UA - Centre - Kirovohrad Oblast 80400203 UA - Centre - Poltava Oblast 80400204 UA - Centre - Khmelnytskyi Oblast 80400205 UA - Centre - Cherkasy Oblast 80400301 UA - North - Zhytomyr Oblast 80400302 UA - North - Kyiv 80400303 UA - North - Kyiv Oblast 80400304 UA - North - Sumy Oblast 80400305 UA - North - Chernihiv Oblast 80400401 UA - East - Dnipropetrovsk 80400402 UA - East - Dnipropetrovsk Oblast 80400403 UA - East - Zaporizhia 80400404 UA - East - Zaporizhia Oblast 80400406 UA - East - Donetsk Oblast 80400407 UA - East - Luhansk Oblast 80400409 UA - East - Kharkiv Oblast 80400501 UA - South - Sevastopol

```
80400502
           UA - South - Autonomous Republic of Crimea
80400503
           UA - South - Mykolaiv Oblast
80400504
           UA - South - Odesa
80400505
           UA - South - Odesa Oblast
80400506 UA - South - Kherson Oblast
80700000 MK - Poranesnata jugoslovenska Republika Makedonija
82611201 GB - North East (England) - Tees Valley and Durham
82611202 GB - North East (England) - Northumberland and Tyne and Wear
82611301 GB - North West (England) - Cumbria
82611302 GB - North West (England) - Cheshire
82611303 GB - North West (England) - Greater Manchester
82611304 GB - North West (England) - Lancashire
82611305 GB - North West (England) - Merseyside
82611401
          GB - Yorkshire and the Humber - East Yorkshire and Northern Lincolnshire
82611402 GB - Yorkshire and the Humber - North Yorkshire
82611403 GB - Yorkshire and the Humber - South Yorkshire
82611404 GB - Yorkshire and the Humber - West Yorkshire
82611501 GB - East Midlands (England) - Derbyshire and Nottinghamshire
82611502 GB - East Midlands (England) - Leicestershire, Rutland and Northamptonshire
82611503 GB - East Midlands (England) - Lincolnshire
82611601 GB - West Midlands (England) - Herefordshire, Worcestershire and Warwickshire
82611602 GB - West Midlands (England) - Shropshire and Staffordshire
82611603
           GB - West Midlands (England) - West Midlands
82611701
           GB - East of England - East Anglia
82611702
           GB - East of England - Bedfordshire and Hertfordshire
82611703 GB - East of England - Essex
82611801 GB - London - Inner London
82611802 GB - London - Outer London
82611901 GB - South East (England) - Berkshire, Buckinghamshire and Oxfordshire
82611902 GB - South East (England) - Surrey, East and West Sussex
82611903 GB - South East (England) - Hampshire and Isle of Wight
82611904 GB - South East (England) - Kent
82612001
          GB - South West (England) - Gloucestershire, Wiltshire and Bristol/Bath area
82612002 GB - South West (England) - Dorset and Somerset
82612003 GB - South West (England) - Cornwall and Isles of Scilly
82612004 GB - South West (England) - Devon
82612101 GB - Wales - West Wales and The Valleys
82612102 GB - Wales - East Wales
82612202 GB - Scotland - Eastern Scotland
82612203 GB - Scotland - South Western Scotland
82612205 GB - Scotland - North Eastern Scotland
82622206 GB - Scotland - Highlands and Islands
82622300 NIR - Northern Ireland
91500000 KOS - Kosovo
```

Derivation:

Variable was computed from variable v368b_N3.

276004 DE – Brandenburg 276005 DE – Bremen

GESIS StudyNo 4754, v1.1.0, http://dx.doi.org/doi:10.4232/1.10152

v368b_N1 - Region at age 14: NUTS-1 code (Q133) HARMONIZED VARIABLE - NUTS 1 Q133 Where did you live when you were about 14 years old? Code: -5 other missing question not asked -4 -3 nap -2 na -1 dk 8000 AL - Albania 31000 AZ – Azerbaijan 40001 AT - Ostösterreich 40002 AT - Südösterreich 40003 AT - Westöstereich 51000 AM – Armenia 56001 BE - Région de Bruxelles-capitale/Brussels hoofdstedelijk gewest 56002 BE - Vlaams gewest 56003 BE - Région Wallonne 70000 BA - Bosna i Hercegovina 100003 BG - Severna i iztochna Bulgaria 100004 BG - Yugozapadna i yuzhna tsentralna Bulgaria 112000 BY - Belarus 191000 HR - Hrvatska 196100 CY - Kypros / Kibris 203000 CZ - Ceska Republika 208000 DK - Danmark 233000 EE - Eesti 246001 FI - Manner-Suomi 246002 FI - Åland 250001 FR - Île de France 250002 FR - Bassin Parisien 250003 FR - Nord-pas-de-Calais 250004 FR - Est 250005 FR - Ouest 250006 FR - Sud-Ouest 250007 FR - Centre-Est 250008 FR - Méditerranée 250009 FR - Départements d'outre-mer 268000 GE - Georgia 276001 DE - Baden-Württemberg 276002 DE - Bayern 276003 DE - Berlin

276006 DE - Hamburg 276007 DE - Hessen 276008 DE - Mecklenburg-Vorpommern 276009 DE - Niedersachsen 276010 DE - Nordrhein-Westfalen 276011 DE - Rheinland-Pfalz 276012 DE - Saarland 276013 DE - Sachsen 276014 DE - Sachsen-Anhalt 276015 DE - Schleswig-Holstein 276016 DE - Thüringen 300001 GR - Voreia Ellada 300002 GR - Kentriki Ellada 300003 GR - Attiki 300004 GR - Nisia Aigaiou, Kriti 348001 HU - Közép-Magyarország 348002 HU - Dunántúl 348003 HU - Alföld és Észak 352000 IS - Ísland 372000 IE - Ireland 380012 IT - Nord-Ovest 380013 IT - Nord-Est 380014 IT - Centro (I) 380015 IT - Sud 380016 IT - Isole 428000 LV - Latvija 440000 LT - Lietuva 442000 LU - Luxembourg (Grand-Duché) 470000 MT - Malta 498000 MD - Moldova 499000 ME - Montenegro 528001 NL - Noord-Nederland 528002 NL - Oost-Nederland 528003 NL - West-Nederland 528004 NL - Zuid-Nederland 578000 NO - Norge 616001 PL - Region centralny 616002 PL - Region Poludinowy 616003 PL - Region Wschodni 616004 PL - Region Pólnocno-Zachodni 616005 PL - Region Póludniowo-Zachodni 616006 PL - Region Pólnocny 620001 PT - Continente 620002 PT - Região Autónoma dos Açores 620003 PT - Região Autónoma da Madeira 642001 RO - Macroregiunea unu 642002 RO - Macroregiunea doi 642003 RO - Macroregiunea trei

642004 RO - Macroregiunea patru

643001 RU - Central Federal District 643002 RU - North West federal district 643003 RU - South Federal district 643004 RU - Privolzhsky federal district 643005 RU - Urals federal district 643006 RU - Siberian federal district 643007 RU - Far East federal district 688001 RS - Centralna Srbija 688002 RS - Vojvodina 703000 SK - Slovenská Republika 705000 SI - Slovenija 724001 ES - Noroeste 724002 ES - Noreste 724003 ES – Comunidad de Madrid 724004 ES - Centro (E) 724005 ES - Este 724006 ES - Sur 724007 ES - Canarias 752001 SE - Östra Sverige 752002 SE - Södra Sverige 752003 SE – Norra Sverige 756000 CH - Schweiz/Suisse/Svizzera 792001 TR - Istanbul 792002 TR - Turkiye - Bati Marmara 792003 TR - Turkiye - EGE 792004 TR - Dogu Marmara 792005 TR - Bati Anadolu 792006 TR - Akdeniz 792007 TR - Orta Anadolu 792008 TR - Bati Karadeniz 792009 TR - Dogu Karadeniz 792010 TR – Kuzeydogu Anadolu 792011 TR – Ortadogu Anadolu 792012 TR - Guneydogu Anadolu 804001 UA - West 804002 UA - Centre 804003 UA - North 804004 UA - East 804005 UA - South 807000 MK – Poranesnata jugoslovenska Republika Makedonija 826112 GB - North East (England) 826113 GB - North West (England) 826114 GB - Yorkshire and the Humber 826115 GB - East Midlands (England) 826116 GB - West Midlands (England) 826117 GB - East of England 826118 GB - London 826119 GB - South East (England)

826120 GB - South West (England)

826121 GB - Wales

826122 GB - Scotland

826223 NIR - Northern Ireland

915000 KOS - Kosovo

Derivation:

Variable was computed from variable v368b_N3.

v370 - size of town where interview was conducted (Q135)

Q135

Size of town

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 under 2.000
- 2 2 5.000
- 3 5 10.000
- 4 10 20.000
- 5 20 50.000
- 6 50 100.000
- 7 100 500.000
- 8 500.000 and more

v370

Value	Label	Missing Count	Percent	Valid Percent
1	under 2000	295	19.5	19.5
2	2-5000	328	21.7	21.7
3	5-10000	175	11.6	11.6
4	10-20000	126	8.3	8.3
5	20-50000	87	5.8	5.8
6	50-100000	81	5.4	5.4
7	100-500000	114	7.5	7.5
8	500000 and more	304	20.1	20.1
	Sum	1510	100.0	100.0
	Valid Cases	1510		

```
v371b_N3 - Region: NUTS-3 code (Q136)
Q136
Region:
Write in:
<CODING INSTRUCTION: CODE REGION IN NUTS 3>
-5
   other missing
-4
   question not asked
   nap
-3
-2
   na
-1
   dk
8000101 AL - Albania - South-Albania - Berat
8000102 AL – Albania – South-Albania – Elbasan
8000103 AL - Albania - South-Albania - Fier
8000104 AL - Albania - South-Albania - Gjirokastër
8000105 AL - Albania - South-Albania - Korçë
8000106 AL - Albania - South-Albania - Vlore
8000201 AL - Albania - North-Albania - Dibër
8000202 AL - Albania - North-Albania - Durrës
8000203 AL - Albania - North-Albania - Kukës
8000204 AL - Albania - North-Albania - Lezhë
8000205 AL - Albania - North-Albania - Shkodër
8000206 AL - Albania - North-Albania - Tiranë
31000001 AZ - Azerbaijan - Baku-Absheron
31000002 AZ – Azerbaijan – Nakhchivan
31000003 AZ - Azerbaijan - Ganja-Gazakh
31000004 AZ - Azerbaijan - Sheki-Zagatala
31000005 AZ - Azerbaijan - Lankaran-Astara
31000006 AZ - Azerbaijan - Garabakh-Mil
31000007 AZ - Azerbaijan - Guba-Khachmaz
31000008 AZ - Azerbaijan - Mughan-Salyan
31000009 AZ – Azerbaijan – Shirvan
40010101 AT - Ostösterreich - Burgenland (A) - Mittelburgenland
40010102 AT - Ostösterreich - Burgenland (A) - Nordburgenland
40010103 AT - Ostösterreich - Burgenland (A) - Südburgenland
40010201 AT – Ostösterreich – Niederösterreich – Mostviertel-Eisenwurzen
40010202 AT - Ostösterreich - Niederösterreich - Niederösterreich-Süd
40010203 AT - Ostösterreich - Niederösterreich - Sankt Pölten
40010204
           AT – Ostösterreich – Niederösterreich – Wadviertel
40010205 AT - Ostösterreich - Niederösterreich - Weinviertel
40010206
           AT - Ostösterreich - Niederösterreich Wiener Umland/Nordteil
40010207
           AT - Ostösterreich - Niederösterreich Wiener Umland/Südteil
40010300 AT - Ostösterreich - Wien - Wien
40020101 AT - Südösterreich - Kärnten - Klagenfurt-Villach
40020102 AT - Südösterreich - Kärnten - Oberkärnten
40020103 AT - Südösterreich - Kärnten - Unterkärnten
40020201 AT - Südösterreich - Steiermark - Graz
```

40020202 AT - Südösterreich - Steiermark - Liezen

```
40020203
           AT - Südösterreich - Steiermark - Östliche Obersteiermark
40020204
           AT - Südösterreich - Steiermark - Oststeiermark
           AT - Südösterreich - Steiermark - West-und Südsteiermark
40020205
           AT - Südösterreich - Steiermark - Westliche Obersteiermark
40020206
40030101
          AT – Westöstereich – Oberösterreich – Innviertel
40030102 AT – Westöstereich – Oberösterreich – Linz-Wels
40030103 AT - Westöstereich - Oberösterreich - Mühlviertel
40030104 AT - Westöstereich - Oberösterreich - Steyr-Kirchdorf
40030105 AT – Westöstereich – Oberösterreich – Traunviertel
40030201 AT – Westöstereich – Salzburg – Lungau
40030202 AT - Westöstereich - Salzburg - Pinzgau-Pongau
40030203 AT - Westöstereich - Salzburg - Salzburg und Umgebung
40030301 AT – Westöstereich – Tirol – Außerfern
40030302 AT – Westöstereich – Tirol – Inssbruck
40030303 AT - Westöstereich - Tirol - Osttirol
40030304 AT - Westöstereich - Tirol - Tiroler Oberland
40030305 AT - Westöstereich - Tirol - Tiroler Unterland
40030401 AT - Westöstereich - Vorarlberg - Bludens-Bregenzer Wald
40030402 AT - Westöstereich - Vorarlberg - Rheintal-Bodenseegebiet
51000101 AM - Armenia - Yerevan - Yerevan
51000201 AM - Armenia - Rest of Armenia - Shirak
51000202 AM - Armenia - Rest of Armenia - Lori
51000203
           AM - Armenia - Rest of Armenia - Tavush
51000204
          AM – Armenia – Rest of Armenia – Aragatsotn
51000205
           AM - Armenia - Rest of Armenia - Kotayk
51000206 AM - Armenia - Rest of Armenia - Gegharkunik
51000207 AM - Armenia - Rest of Armenia - Armavir
51000208 AM - Armenia - Rest of Armenia - Ararat
51000209 AM - Armenia - Rest of Armenia - Vavots Dzor
51000210 AM - Armenia - Rest of Armenia - Syunik
56010000 BE - Région de Bruxelles-capitale/Brussels hoofdstedelijk gewest - Arr. de Bruxelles-Capitale/ Arr. van Brussel-Ho
56020101
           BE - Vlaams gewest - Prov. Antwerpen - Arr. Antwerpen
56020102
           BE - Vlaams gewest - Prov. Antwerpen - Arr. Mechelen
          BE - Vlaams gewest - Prov. Antwerpen - Arr. Turnhout
56020103
56020201
           BE - Vlaams gewest - Prov. Limburg (B) - Arr. Hasselt
56020202
           BE - Vlaams gewest - Prov. Limburg (B) - Arr. Maaseik
56020203 BE – Vlaams gewest – Prov. Limburg (B) – Arr. Tongeren
56020301 BE - Vlaams gewest - Prov. Oost-Vlaanderen - Arr. Aalst
56020302 BE - Vlaams gewest - Prov. Oost-Vlaanderen - Arr. Dendermonde
56020303 BE - Vlaams gewest - Prov. Oost-Vlaanderen - Arr. Eeklo
56020304
           BE - Vlaams gewest - Prov. Oost-Vlaanderen - Arr. Gent
56020305
           BE - Vlaams gewest - Prov. Oost-Vlaanderen - Arr. Oudenaarde
56020306
           BE - Vlaams gewest - Prov. Oost-Vlaanderen - Arr. Sint-Niklaas
           BE - Vlaams gewest - Prov. Vlaams-Brabant - Arr. Halle- Vilvoorde
56020401
56020402
           BE - Vlaams gewest - Prov. Vlaams-Brabant - Arr. Leuven
56020501
           BE - Vlaams gewest - Prov. West-Vlaanderen - Arr. Brugge
56020502
          BE - Vlaams gewest - Prov. West-Vlaanderen - Arr. Diksmuide
56020503
           BE - Vlaams gewest - Prov. West-Vlaanderen - Arr. leper
56020504 BE - Vlaams gewest - Prov. West-Vlaanderen - Arr. Kortrijk
```

```
56020505
           BE - Vlaams gewest - Prov. West-Vlaanderen - Arr. Oostende
56020506
           BE - Vlaams gewest - Prov. West-Vlaanderen - Arr. Roeselare
56020507
           BE - Vlaams gewest - Prov. West-Vlaanderen - Arr. Tielt
           BE - Vlaams gewest - Prov. West-Vlaanderen - Arr. Veurne
56020508
56030100
           BE - Région Wallonne - Prov. Brabant Wallon - Arr. Nivelles
56030201
           BE - Région Wallonne - Prov. Hainaut - Arr. Ath
56030202 BE - Région Wallonne - Prov. Hainaut - Arr. Charleroi
56030203 BE - Région Wallonne - Prov. Hainaut - Arr. Mons
56030204 BE - Région Wallonne - Prov. Hainaut - Arr. Mouscron
56030205
           BE - Région Wallonne - Prov. Hainaut - Arr. Soignies
56030206 BE - Région Wallonne - Prov. Hainaut - Arr. Thuin
56030207
           BE - Région Wallonne - Prov. Hainaut - Arr. Tournai
56030301
           BE - Région Wallonne - Prov. Liège - Arr. Huy
56030302
           BE - Région Wallonne - Prov. Liège - Arr. Liège
56030304
           BE - Région Wallonne - Prov. Liège - Arr. Waremme
56030305
           BE - Région Wallonne - Prov. Liège - Arr. Verviers-commmunes francophonees
56030306 BE – Région Wallonne – Prov. Liège – Bezirk Verviers-Deutschsprachige Gemeinschaft
56030401 BE - Région Wallonne - Prov. Luxembourg (B) - Arr. Arlon
56030402 BE - Région Wallonne - Prov. Luxembourg (B) - Arr. Bastogne
56030403
           BE - Région Wallonne - Prov. Luxembourg (B) - Arr. Marche-en-Famenne
56030404
           BE - Région Wallonne - Prov. Luxembourg (B) - Arr. Neufchâteau
56030405
           BE - Région Wallonne - Prov. Luxembourg (B) - Arr. Virton
56030501
           BE - Région Wallonne - Prov. Namur - Arr. Dinant
56030502
           BE - Région Wallonne - Prov. Namur - Arr. Namur
56030503
           BE - Région Wallonne - Prov. Namur - Arr. Philippeville
           BA – Bosna i Hercegovina – Federacija Bosne i Hercegovine – Unsko-sanksi
70000101
70000102 BA - Bosna i Hercegovina - Federacija Bosne i Hercegovine - Povaski
70000103 BA – Bosna i Hercegovina – Federacija Bosne i Hercegovine – Tuzlanski
70000104
           BA - Bosna i Hercegovina - Federacija Bosne i Hercegovine - Zenicko-dobjski
70000105 BA – Bosna i Hercegovina – Federacija Bosne i Hercegovine – Bosansko-podrinjski
70000106
           BA - Bosna i Hercegovina - Federacija Bosne i Hercegovine - Srednjobosanski
70000107
           BA - Bosna i Hercegovina - Federacija Bosne i Hercegovine - Hercegovacko-neretvanski
70000108
           BA - Bosna i Hercegovina - Federacija Bosne i Hercegovine - Zapadnhercegovacki
70000109
           BA - Bosna i Hercegovina - Federacija Bosne i Hercegovine - Sarajevo
70000110
           BA - Bosna i Hercegovina - Federacija Bosne i Hercegovine - Livno (Kanton br. 10)
70000201
           BA - Bosna i Hercegovina - Republika Srpska - Banja Luka
70000202 BA - Bosna i Hercegovina - Republika Srpska - Doboj
70000203 BA - Bosna i Hercegovina - Republika Srpska - Bijeljina
70000204 BA - Bosna i Hercegovina - Republika Srpska - Vlasenica
70000205 BA - Bosna i Hercegovina - Republika Srpska - Sarajevo -Romanija
70000206 BA - Bosna i Hercegovina - Republika Srpska - Foca
70000207
           BA - Bosna i Hercegovina - Republika Srpska - Trebinje
70000301
           BA - Bosna i Hercegovina - Brcko - Brcko Disrikt
            BG - Severna i iztochna Bulgaria - Severozapaden - Vidin
100030101
100030102
            BG - Severna i iztochna Bulgaria - Severozapaden - Montana
100030103
            BG - Severna i iztochna Bulgaria - Severozapaden - Vratsa
100030104
            BG – Severna i iztochna Bulgaria – Severozapaden – Pleven
100030105 BG - Severna i iztochna Bulgaria - Severozapaden - Lovech
100030201 BG - Severna i iztochna Bulgaria - Severen tsentralen - Veliko Tarnovo
```

```
100030202
            BG - Severna i iztochna Bulgaria - Severen tsentralen - Gabrovo
100030203
            BG - Severna i iztochna Bulgaria - Severen tsentralen - Ruse
100030204
            BG - Severna i iztochna Bulgaria - Severen tsentralen - Razgrad
100030205
            BG - Severna i iztochna Bulgaria - Severen tsentralen - Silistra
100030301
            BG - Severna i iztochna Bulgaria - Severoiztochen - Varna
100030302
            BG - Severna i iztochna Bulgaria - Severoiztochen - Dobrich
100030303
            BG - Severna i iztochna Bulgaria - Severoiztochen - Shumen
100030304
            BG - Severna i iztochna Bulgaria - Severoiztochen - Targovishte
100030401
            BG - Severna i iztochna Bulgaria - Yugoiztochen - Burgas
100030402
            BG - Severna i iztochna Bulgaria - Yugoiztochen - Sliven
100030403
            BG - Severna i iztochna Bulgaria - Yugoiztochen - Yambol
100030404
            BG - Severna i iztochna Bulgaria - Yugoiztochen - Stara Zagora
100040101
            BG – Yugozapadna i yuzhna tsentralna Bulgaria – Yugozapaden – Sofia (stolitsa)
            BG - Yugozapadna i yuzhna tsentralna Bulgaria - Yugozapaden - Sofia
100040102
100040103
            BG - Yugozapadna i yuzhna tsentralna Bulgaria - Yugozapaden - Blagoevgrad
100040104
            BG - Yugozapadna i yuzhna tsentralna Bulgaria - Yugozapaden - Pernik
100040105
            BG - Yugozapadna i yuzhna tsentralna Bulgaria - Yugozapaden - Kyustendil
100040201
            BG - Yugozapadna i yuzhna tsentralna Bulgaria - Yuzhen tsentralen - Plovdiv
100040202
            BG - Yugozapadna i yuzhna tsentralna Bulgaria - Yuzhen tsentralen - Haskovo
100040203
            BG - Yugozapadna i yuzhna tsentralna Bulgaria - Yuzhen tsentralen - Pazardzhik
100040204
            BG - Yugozapadna i yuzhna tsentralna Bulgaria - Yuzhen tsentralen - Smolyan
100040205
            BG - Yugozapadna i yuzhna tsentralna Bulgaria - Yuzhen tsentralen - Kardzhali
112000001
            BY - Belarus - Brest
112000002
            BY - Belarus - Vitebsk
112000003
            BY - Belarus - Gomel
112000004 BY - Belarus - Grodno
112000005 BY - Belarus - Minsk (capital city)
112000006 BY - Belarus - Minsk
112000007 BY - Belarus - Mogilev
191000101 HR - Hrvatska - Sjeverozapadna Hrvatska - Grad Zagreb
191000102 HR - Hrvatska - Sjeverozapadna Hrvatska - Zagrebacka zupanija
191000103 HR - Hrvatska - Sjeverozapadna Hrvatska - Krapinsko-zagorska zupanija
191000104 HR - Hrvatska - Sjeverozapadna Hrvatska - Varazdinska zupanija
191000105 HR – Hrvatska – Sjeverozapadna Hrvatska – Koprivnicko-krizevacka zupanija
191000106 HR – Hrvatska – Sjeverozapadna Hrvatska – Medimurska zupanija
191000201
            HR – Hrvatska – Sredisnja i Istocna (Panonska) Hrvatska – Bjelovarsko-bilogorska zupanija
191000202 HR - Hrvatska - Sredisnja i Istocna (Panonska) Hrvatska - Viroviticko-podravska zupanija
191000203 HR - Hrvatska - Sredisnja i Istocna (Panonska) Hrvatska - Pozesko-slavonska zupanija
191000204
            HR – Hrvatska – Sredisnja i Istocna (Panonska) Hrvatska – Brodsko-posavska zupanija
191000205 HR – Hrvatska – Sredisnja i Istocna (Panonska) Hrvatska – Osjecko-baranjska zupanija
191000206 HR - Hrvatska - Sredisnja i Istocna (Panonska) Hrvatska - Vukovarsko-srijemska zupanija
191000207
            HR - Hrvatska - Sredisnja i Istocna (Panonska) Hrvatska - Karlovacka zupanija
191000208
            HR - Hrvatska - Sredisnja i Istocna (Panonska) Hrvatska - Sisacko-moslavacka zupanija
            HR - Hrvatska - Jadranska Hrvatska - Primorsko-goranska zupanija
191000301
191000302 HR - Hrvatska - Jadranska Hrvatska - Licko-senjska zupanija
191000303
            HR - Hrvatska - Jadranska Hrvatska - Zadarska zupanija
191000304 HR – Hrvatska – Jadranska Hrvatska – Sibensko-kninska zupanija
191000305 HR – Hrvatska – Jadranska Hrvatska – Splitsko-dalmatinska zupanija
191000306 HR – Hrvatska – Jadranska Hrvatska – Istarska zupanija
```

```
191000307
           HR - Hrvatska - Jadranska Hrvatska - Dubrovacko-neretvanska zupanija
203000100
           CZ - Ceska Republika - Praha - Hlavni mesto Praha
203000200
            CZ - Ceska Republika - Stredni Cechy - Stredocesky kraj
203000301
            CZ - Ceska Republika - Jihozapad - Jihocesky kraj
203000302 CZ - Ceska Republika - Jihozapad - Plzensky kraj
           CZ - Ceska Republika - Severozapad Karlovarsky kraj
203000401
203000402 CZ - Ceska Republika - Severozapad Ustecky kraj
203000501
            CZ - Ceska Republika - Severovychod Liberecky kraj
203000502 CZ - Ceska Republika - Severovychod Kraloveehradecky kraj
203000503
           CZ - Ceska Republika - Severovychod Pardubicky kraj
203000603 CZ - Ceska Republika - Jihovychod - Vysocina
203000604
            CZ - Ceska Republika - Jihovychod - Jihomoravsky kraj
203000701 CZ – Ceska Republika – Stredni Morava – Olomoucky kraj
203000702 CZ - Ceska Republika - Stredni Morava - Zlinsky kraj
203000800 CZ - Ceska Republika - Moravskoslezsko - Moravskoslezsky kraj
208000101
           DK - Danmark - Hovedstaden - Byen København
208000102 DK - Danmark - Hovedstaden - Københavns omegn
208000103 DK - Danmark - Hovedstaden Nordsjælland
208000104
           DK - Danmark - Hovedstaden - Bornholm
208000201
            DK - Danmark- Sjælland - Østsjælland
208000202
            DK - Danmark- Sjælland - Vest - og Sydsjælland
208000301
            DK - Danmark - Syddanmark - Fyn
208000302
            DK - Danmark - Syddanmark - Sydjylland
208000401
            DK - Danmark - Midtjylland - Vestjylland
            DK - Danmark - Midtjylland - Østjylland
208000402
208000500
           DK - Danmark - Nordjylland - Nordjylland
233000001
            EE - Eesti - Põhja-Eesti
233000004 EE - Eesti - Lääne-Eesti
233000006 EE - Eesti - Kesk-Eesti
233000007 EE - Eesti - Kirde-Eesti
233000008
           EE - Eesti - Lõuna-Eesti
246010301 FI - Manner-Suomi - Itä-Suomi - Etalä-Savo
246010302 FI - Manner-Suomi - Itä-Suomi - Pohjois-Savo
246010303 FI - Manner-Suomi - Itä-Suomi - Pohjois-Karjala
246010304
           FI - Manner-Suomi - Itä-Suomi - Kainuu
246010801
           FI - Manner-Suomi - Etalä-Suomi - Uusimaa
246010802 FI - Manner-Suomi - Etalä-Suomi - Itä-Uusimaa
246010803 FI - Manner-Suomi - Etalä-Suomi - Varsinais-Suomi
246010804
           FI - Manner-Suomi - Etalä-Suomi - Kanta-Häme
246010805 FI - Manner-Suomi - Etalä-Suomi - Päijät-Häme
246010806 FI - Manner-Suomi - Etalä-Suomi - Kymenlaakso
246010807
           FI - Manner-Suomi - Etalä-Suomi - Etelä-Karjala
246010903 FI - Manner-Suomi - Länsi-Suomi - Keski-Suomi
           FI - Manner-Suomi - Länsi-Suomi - Etelä-Pohjanmaa
246010904
246010905 FI - Manner-Suomi - Länsi-Suomi - Pohjanmaa
246010906
           FI - Manner-Suomi - Länsi-Suomi - Satakunta
246010907 FI - Manner-Suomi - Länsi-Suomi - Pirkanmaa
246011001 FI - Manner-Suomi - Pohjois-Suomi - Keski-Pohjanmaa
246011002 FI - Manner-Suomi - Pohjois-Suomi - Pohjois-Pohjanmaa
```

```
246011003 FI - Manner-Suomi - Pohjois-Suomi - Lappi
246020000 FI - Åland
250010001 FR - Île de France - Paris
250010002 FR - Île de France - Seine-et-Marne
250010003 FR - Île de France - Yvelines
250010004 FR - Île de France - Essonne
250010005 FR - Île de France - Hauts-de-Seine
250010006 FR - Île de France - Seine-Saint-Denis
250010007 FR – Île de France – Val-de-Marne
250010008 FR - Île de France - Val-d'Oise
250020101 FR - Bassin Parisien - Champagne-Ardenne - Ardennes
250020102 FR – Bassin Parisien – Champagne-Ardenne – Aube
250020103 FR - Bassin Parisien - Champagne-Ardenne - Marne
250020104 FR – Bassin Parisien – Champagne-Ardenne – Haute-Marne
250020201 FR - Bassin Parisien - Champagne-Ardenne - Picardie - Aisne
250020202 FR - Bassin Parisien - Champagne-Ardenne - Picardie - Oise
250020203 FR - Bassin Parisien - Champagne-Ardenne - Picardie - Somme
250020301 FR - Bassin Parisien - Champagne-Ardenne - Haute-Normandie - Eure
250020302 FR – Bassin Parisien – Champagne-Ardenne – Haute-Normandie – Seine-Martime
250020401 FR - Bassin Parisien - Champagne-Ardenne - Centre - Cher
250020402 FR - Bassin Parisien - Champagne-Ardenne - Centre - Eure-et-Loir
250020403 FR - Bassin Parisien - Champagne-Ardenne - Centre - Indre
250020404 FR - Bassin Parisien - Champagne-Ardenne - Centre - Indre-et-Loire
250020405 FR - Bassin Parisien - Champagne-Ardenne - Centre - Loir-et-Cher
250020406 FR - Bassin Parisien - Champagne-Ardenne - Centre - Loiret
250020501 FR - Bassin Parisien - Champagne-Ardenne - Basse-Normandie - Calvados
250020502 FR - Bassin Parisien - Champagne-Ardenne - Basse-Normandie - Manche
250020503 FR - Bassin Parisien - Champagne-Ardenne - Basse-Normandie - Orne
250020601 FR - Bassin Parisien - Champagne-Ardenne - Bourgogne - Côte-d'Or
250020602 FR - Bassin Parisien - Champagne-Ardenne - Bourgogne - Nièvre
250020603 FR - Bassin Parisien - Champagne-Ardenne - Bourgogne - Saône-et-Loire
250020604 FR - Bassin Parisien - Champagne-Ardenne - Bourgogne - Yonne
250030001 FR - Nord-pas-de-Calais - Nord
250030002 FR - Nord-pas-de-Calais - Pas-de-Calais
250040101 FR - Est - Lorraine - Meurthe-et-Moselle
250040102 FR - Est - Lorraine - Meuse
250040103 FR - Est - Lorraine - Moselle
250040104 FR - Est - Lorraine - Vosges
250040201 FR - Est - Alsace - Bas-Rhin
250040202 FR - Est - Alsace - Haut-Rhin
250040301 FR - Est - Franche-Comté - Doubs
250040302 FR - Est - Franche-Comté - Jura
250040303 FR - Est - Franche-Comté - Haute-Saône
250040304 FR - Est - Franche-Comté - Territoire de Belfort
250050101 FR - Ouest - Pays de la Loire - Loire-Atlantique
250050102 FR - Ouest - Pays de la Loire - Maine-et-Loire
250050103 FR - Ouest - Pays de la Loire - Mayenne
250050104 FR - Ouest - Pays de la Loire - Sarthe
250050105 FR - Ouest - Pays de la Loire - Vendée
```

```
250050201
           FR - Ouest - Bretagne - Côtes-d'Armor
250050202
           FR - Ouest - Bretagne - Finistère
250050203
           FR - Ouest - Bretagne - Ille-et-Vilaine
250050204
           FR - Ouest - Bretagne - Morbihan
250050301 FR - Ouest - Poitou-Charentes - Charente
250050302 FR - Ouest - Poitou-Charentes - Charente-Maritime
250050303 FR - Ouest - Poitou-Charentes - Deux-Sèvres
250050304 FR - Ouest - Poitou-Charentes - Vienne
250060101 FR - Sud-Ouest - Aquitaine - Dordogne
250060102 FR - Sud-Ouest - Aquitaine - Gironde
250060103 FR - Sud-Ouest - Aquitaine - Landes
250060104 FR - Sud-Ouest - Aquitaine - Lot-et-Garonne
250060105 FR - Sud-Ouest - Aquitaine - Pyrénées-Atlantiques
250060201 FR - Sud-Ouest - Midi-Pyrénées - Ariège
250060202 FR - Sud-Ouest - Midi-Pyrénées - Aveyron
250060203 FR - Sud-Ouest - Midi-Pyrénées - Haute-Garonne
250060204 FR - Sud-Ouest - Midi-Pyrénées - Gerts
250060205 FR - Sud-Ouest - Midi-Pyrénées - Lot
250060206 FR - Sud-Ouest - Midi-Pyrénées - Hautes-Pyrénées
250060207 FR - Sud-Ouest - Midi-Pyrénées - Tarn
250060208 FR - Sud-Ouest - Midi-Pyrénées - Tarn-et-Garonne
250060301 FR - Sud-Ouest - Limousin - Corrèze
250060302 FR - Sud-Ouest - Limousin - Creuse
250060303 FR - Sud-Ouest - Limousin - Haute-Vienne
250070101 FR - Centre-Est - Rhône-Alpes - Ain
250070102 FR - Centre-Est - Rhône-Alpes - Ardèche
250070103 FR - Centre-Est - Rhône-Alpes - Drôme
250070104 FR - Centre-Est - Rhône-Alpes - Isère
250070105 FR - Centre-Est - Rhône-Alpes - Loire
250070106 FR - Centre-Est - Rhône-Alpes - Rhône
250070107 FR - Centre-Est - Rhône-Alpes - Savoie
250070108 FR - Centre-Est - Rhône-Alpes - Haute-Savoie
250070201 FR - Centre-Est - Auvergne - Allier
250070202 FR - Centre-Est - Auvergne - Cantal
250070203 FR - Centre-Est - Auvergne - Haute-Loire
250070204 FR - Centre-Est - Auvergne - Puy-de-Dôme
250080101 FR - Méditerranée - Languedoc-Roussillon - Aude
250080102 FR - Méditerranée - Languedoc-Roussillon - Gard
250080103 FR - Méditerranée - Languedoc-Roussillon - Hérault
250080104 FR - Méditerranée - Languedoc-Roussillon - Lozère
250080105 FR - Méditerranée - Languedoc-Roussillon - Pyrénées-Orientales
250080201 FR - Méditerranée - Provence-Alpes-Côte d'Azur - Alpes-de-Haute-Provence
250080202 FR - Méditerranée - Provence-Alpes-Côte d'Azur - Hautes-Alpes
250080203
           FR - Méditerranée - Provence-Alpes-Côte d'Azur - Alpes-Maritimes
250080204 FR - Méditerranée - Provence-Alpes-Côte d'Azur - Bouches-du-Rhône
           FR - Méditerranée - Provence-Alpes-Côte d'Azur - Var
250080205
250080206 FR - Méditerranée - Provence-Alpes-Côte d'Azur - Vaucluse
250080301 FR - Méditerranée - Corse - Corse-du-Sud
250080302 FR - Méditerranée - Corse - Haute-Corse
```

```
250090100 FR - Départements d'outre-mer - Guadeloupe
250090200
           FR - Départements d'outre-mer - Martinique
250090300
           FR - Départements d'outre-mer - Guyane
250090400
           FR - Départements d'outre-mer - Réunion
268000001
           GE - Georgia - Tbilisi
268000002 GE - Georgia - Kakheti
268000003 GE - Georgia - Shida Kartli
268000004
           GE - Georgia - Kvemo Kartli
268000005 GE - Georgia - Samtskhe-Djavakheti
268000006 GE - Georgia - Adjara
268000007 GE - Georgia - Guria
268000008
            GE - Georgia - Samegrelo/ SvaneTi
           GE - Georgia - Imereti
268000009
268000010
           GE - Georgia - Mtskheta/TianeTi
268000011
           GE - Georgia - Racha-Lechkhumi/kvemo Svaneti
268000012 GE - Georgia - Abkhazia
276010101 DE - Baden-Württemberg - Stuttgart - Stuttgart, Stadtkreis
276010102 DE – Baden-Württemberg – Stuttgart – Böblingen
276010103 DE - Baden-Württemberg - Stuttgart - Esslingen
276010104 DE – Baden-Württemberg – Stuttgart – Göppingen
276010105 DE - Baden-Württemberg - Stuttgart - Ludwigsburg
276010106 DE - Baden-Württemberg - Stuttgart - Rems-Murr-Kreis
276010107
           DE - Baden-Württemberg - Stuttgart - Heilbronn, Stadtkreis
276010108 DE - Baden-Württemberg - Stuttgart - Heilbronn, Landkreis
276010109
           DE - Baden-Württemberg - Stuttgart - Hohenlohekreis
276010110 DE - Baden-Württemberg - Stuttgart - Schwäbisch Hall
276010111 DE - Baden-Württemberg - Stuttgart - Main-Tauber-Kreis
276010112 DE - Baden-Württemberg - Stuttgart - Heidenheim
276010113 DE - Baden-Württemberg - Stuttgart - Ostalbkreis
276010201 DE - Baden-Württemberg - Karlsruhe - Baden-Baden, Stadtkreis
276010202 DE – Baden-Württemberg – Karlsruhe – Karlsruhe, Stadtkreis
276010203 DE - Baden-Württemberg - Karlsruhe - Karlsruhe, Landkreis
276010204
           DE - Baden-Württemberg - Karlsruhe - Rastatt
276010205 DE - Baden-Württemberg - Karlsruhe - Heidelberg, Stadtkreis
276010206 DE - Baden-Württemberg - Karlsruhe - Mannheim, Stadtkreis
276010207 DE – Baden-Württemberg – Karlsruhe – Neckar-Odenwald-Kreis
276010208 DE – Baden-Württemberg – Karlsruhe – Rhein-Neckar-Kreis
276010209
           DE - Baden-Württemberg - Karlsruhe - Pforzheim, Stadtkreis
276010210 DE - Baden-Württemberg - Karlsruhe - Calw
276010211 DE - Baden-Württemberg - Karlsruhe - Enzkreis
276010212 DE - Baden-Württemberg - Karlsruhe - Freudenstadt
276010301
            DE - Baden-Württemberg - Freiburg - Freiburg im Breisgau, Stadtkreis
276010302
           DE - Baden-Württemberg - Freiburg - Breisgau-Hochschwarzwald
276010303
            DE - Baden-Württemberg - Freiburg - Emmendingen
276010304
            DE - Baden-Württemberg - Freiburg - Ortenaukreis
276010305
            DE - Baden-Württemberg - Freiburg - Rottweil
276010306 DE - Baden-Württemberg - Freiburg - Schwarzwald-Baar-Kreis
276010307 DE - Baden-Württemberg - Freiburg - Tuttlingen
276010308 DE - Baden-Württemberg - Freiburg - Konstanz
```

```
276010309
           DE - Baden-Württemberg - Freiburg - Lörrach
276010310
           DE - Baden-Württemberg - Freiburg - Waldshut
276010401
            DE - Baden-Württemberg - Tübingen - Reutlingen
276010402
            DE - Baden-Württemberg - Tübingen - Tübingen, Landkreis
276010403
           DE - Baden-Württemberg - Tübingen - Zollernalbkreis
276010404
           DE - Baden-Württemberg - Tübingen - Ulm, Stadtkreis
276010405 DE – Baden-Württemberg – Tübingen – Alb-Donau-Kreis
276010406 DE - Baden-Württemberg - Tübingen - Biberach
276010407 DE - Baden-Württemberg - Tübingen - Bodenseekreis
276010408
           DE - Baden-Württemberg - Tübingen - Ravensburg
276010409
           DE - Baden-Württemberg - Tübingen - Sigmaringen
276020101
           DE - Bayern - Oberbayern - Ingolstadt, Kreisfreie Stadt
276020102 DE - Bayern - Oberbayern - München, Kreisfreie Stadt
276020103 DE - Bayern - Oberbayern - Rosenheim, Kreisfreie Stadt
276020104 DE - Bayern - Oberbayern - Altötting
276020105 DE - Bayern - Oberbayern - Berchtesgadener Land
276020106 DE – Bayern – Oberbayern – Bad Tölz-Wolfratshausen
276020107 DE - Bayern - Oberbayern - Dachau
276020108 DE - Bayern - Oberbayern - Ebersberg
276020109
           DE - Bayern - Oberbayern - Eichstätt
276020110 DE - Bayern - Oberbayern - Erding
276020111
           DE - Bayern - Oberbayern - Freising
276020112
           DE - Bayern - Oberbayern - Fürstenfeldbruck
276020113
           DE - Bayern - Oberbayern - Garmisch-Partenkirchen
276020114
           DE - Bayern - Oberbayern - Landsberg a. Lech
276020115 DE - Bayern - Oberbayern - Miesbach
276020116 DE - Bayern - Oberbayern - Mühldorf a. Inn
276020117 DE - Bayern - Oberbayern - München, Landkreis
276020118 DE – Bavern – Oberbavern – Neuburg-Schrobenhausen
276020119 DE - Bayern - Oberbayern - Pfaffenhofen a. d. Ilm
276020120
           DE - Bayern - Oberbayern - Rosenheim, Landkreis
276020121 DE - Bayern - Oberbayern - Starnberg
276020122 DE - Bayern - Oberbayern - Traunstein
276020123 DE - Bayern - Oberbayern - Weilheim-Schongau
276020201
           DE - Bayern - Niederbayern - Landshut, Kreisfreie Stadt
276020202 DE - Bayern - Niederbayern - Passau, Kreisfreie Stadt
276020203 DE – Bayern – Niederbayern – Straubing, Kreisfreie Stadt
276020204
           DE - Bayern - Niederbayern - Deggendorf
276020205 DE - Bayern - Niederbayern - Freyung-Grafenau
276020206 DE - Bayern - Niederbayern - Kelheim
276020207
           DE - Bayern - Niederbayern - Landshut, Landkreis
276020208
           DE - Bayern - Niederbayern - Passau, Landkreis
276020209
            DE - Bayern - Niederbayern - Regen
276020210
           DE - Bayern - Niederbayern - Rottal-Inn
276020211
           DE - Bayern - Niederbayern - Straubing-Bogen
276020212
           DE - Bayern - Niederbayern - Dingolfing-Landau
276020301 DE - Bayern - Oberpfalz - Amberg, Kreisfreie Stadt
276020302 DE - Bayern - Oberpfalz - Regensburg, Kreisfreie Stadt
276020303 DE - Bayern - Oberpfalz - Weiden i. d. Opf, Kreisfreie Stadt
```

```
276020304 DE - Bayern - Oberpfalz - Amberg-Sulzbach
276020305 DE - Bayern - Oberpfalz - Cham
276020306
            DE - Bayern - Oberpfalz - Neumarkt i. d. OPf.
276020307
            DE - Bayern - Oberpfalz - Neustadt a. d. Waldnaab
276020308 DE - Bayern - Oberpfalz - Regensburg, Landkreis
276020309 DE - Bayern - Oberpfalz - Schwandorf
276020310 DE - Bayern - Oberpfalz - Tirschenreuth
276020401 DE - Bayern - Oberfranken - Bamberg, Kreisfreie Stadt
276020402 DE - Bayern - Oberfranken - Bayreuth, Kreisfreie Stadt
276020403 DE - Bayern - Oberfranken - Coburg, Kreisfreie Stadt
276020404 DE - Bayern - Oberfranken - Hof, Kreisfreie Stadt
276020405 DE - Bayern - Oberfranken - Bamberg, Landkreis
276020406 DE - Bayern - Oberfranken - Bayreuth, Landkreis
276020407 DE – Bayern – Oberfranken – Coburg, Landkreis
276020408 DE - Bayern - Oberfranken - Forchheim
276020409 DE - Bayern - Oberfranken - Hof, Landkreis
276020410 DE – Bayern – Oberfranken – Kronach
276020411 DE - Bayern - Oberfranken - Kulmbach
276020412 DE – Bayern – Oberfranken – Lichtenfels
276020413 DE - Bayern - Oberfranken - Wunsiedel i. Fichtelgebirge
276020501 DE - Bayern - Mittelfranken - Ansbach, Kreisfreie Stadt
276020502 DE - Bayern - Mittelfranken - Erlangen, Kreisfreie Stadt
276020503
            DE - Bayern - Mittelfranken - Fürth, Kreisfreie Stadt
276020504
            DE - Bayern - Mittelfranken - Nürnberg, Kreisfreie Stadt
            DE - Bayern - Mittelfranken - Schwabach, Kreisfreie Stadt
276020505
276020506 DE - Bayern - Mittelfranken - Ansbach, Landkreis
276020507
            DE - Bayern - Mittelfranken - Erlangen-Höchstadt
276020508 DE - Bayern - Mittelfranken - Fürth, Landkreis
276020509 DE – Bayern – Mittelfranken – Nürnberger Land
276020510 DE - Bayern - Mittelfranken - Neustadt a. d. Aisch-Bad Windsheim
276020511 DE - Bayern - Mittelfranken - Roth
276020512 DE – Bayern – Mittelfranken – Weißenburg-Gunzenhausen
276020601
           DE - Bayern - Unterfranken - Aschaffenburg, Kreisfreie Stadt
276020602 DE - Bayern - Unterfranken - Schweinfurt, Kreisfreie Stadt
276020603
            DE - Bayern - Unterfranken - Würzburg, Kreisfreie Stadt
276020604 DE - Bayern - Unterfranken - Aschaffenburg, Landkreis
276020605 DE – Bayern – Unterfranken – Bad Kissingen
276020606 DE - Bayern - Unterfranken - Rhön-Grabfeld
276020607 DE - Bayern - Unterfranken - Haßberge
276020608 DE - Bayern - Unterfranken - Kitzingen
276020609 DE - Bayern - Unterfranken - Miltenberg
276020610 DE - Bayern - Unterfranken - Main-Spessart
276020611
            DE - Bayern - Unterfranken - Schweinfurt, Landkreis
276020612
            DE - Bayern - Unterfranken - Würzburg, Landkreis
276020613 DE - Bayern - Unterfranken - Donau-Ries
276020614
            DE - Bayern - Unterfranken - Oberallgäu
276020701 DE - Bayern - Schwaben - Augsburg, Kreisfreie Stadt
276020702 DE - Bayern - Schwaben - Kaufbeuren, Kreisfreie Stadt
276020703 DE - Bayern - Schwaben - Kempten (Allgäu), Kreisfreie Stadt
```

```
276020704 DE - Bayern - Schwaben - Memmingen, Kreisfreie Stadt
276020705 DE - Bayern - Schwaben - Aichach-Friedberg
276020706 DE - Bayern - Schwaben - Augsburg, Landkreis
276020707
           DE - Bayern - Schwaben - Dillingen a.d. Donau
276020708 DE - Bayern - Schwaben - Günzburg
276020709 DE - Bayern - Schwaben - Neu-Ulm
276020710 DE - Bayern - Schwaben - Lindau (Bodensee)
276020711 DE - Bayern - Schwaben - Ostallgäu
276020712 DE - Bayern - Schwaben - Unterallgäu
276020713 DE - Bayern - Schwaben - Donau-Ries
276020714 DE - Bayern - Schwaben - Oberallgäu
276030000 DE - Berlin
276040101 DE - Brandenburg - Brandenburg - Nordost - Frankfurt (Oder), Kreisfreie Stadt
276040102 DE – Brandenburg – Brandenburg – Nordost – Barnim
276040103 DE - Brandenburg - Brandenburg - Nordost - Märkisch-Oderland
276040104 DE – Brandenburg – Brandenburg – Nordost – Oberhavel
276040105 DE – Brandenburg – Brandenburg – Nordost – Oder-Spree
276040106 DE – Brandenburg – Brandenburg – Nordost – Ostprignitz-Ruppin
276040107 DE – Brandenburg – Brandenburg – Nordost – Prignitz
276040108 DE – Brandenburg – Brandenburg – Nordost – Uckermark
276040201 DE - Brandenburg - Brandenburg Südwest - Brandenburg an der Havel, Kreisfreie Stadt
276040202 DE - Brandenburg - Brandenburg Südwest - Cottbus, Kreisfreie Stadt
276040203
           DE - Brandenburg - Brandenburg Südwest - Potsdam, Kreisfreie Stadt
276040204 DE - Brandenburg - Brandenburg Südwest - Dahme-Spreewald
276040205 DE - Brandenburg - Brandenburg Südwest - Elbe-Elster
276040206 DE - Brandenburg - Brandenburg Südwest - Havelland
276040207
           DE - Brandenburg - Brandenburg Südwest - Oberspreewald-Lausitz
276040208 DE – Brandenburg – Brandenburg Südwest – Potsdam-Mittelmark
276040209 DE – Brandenburg – Brandenburg Südwest – Spree-Neiße
276040210 DE – Brandenburg – Brandenburg Südwest – Teltow-Fläming
276050001 DE - Bremen - Bremen, Kreisfreie Stadt
276050002 DE - Bremen - Bremerhaven, Kreisfreie Stadt
276060000 DE - Hamburg
276070101 DE - Hessen - Darmstadt - Darmstadt, Kreisfreie Stadt
276070102 DE - Hessen - Darmstadt - Frankfurt am Main, Kreisfreie Stadt
276070103 DE - Hessen - Darmstadt - Offenbach am Main, Kreisfreie Stadt
276070104 DE - Hessen - Darmstadt - Wiesbaden, Kreisfreie Stadt
276070105 DE - Hessen - Darmstadt - Bergstraße
276070106 DE - Hessen - Darmstadt - Darmstadt-Dieburg
276070107 DE - Hessen - Darmstadt - Groß-Gerau
276070108 DE – Hessen – Darmstadt – Hochtaunuskreis
276070109 DE - Hessen - Darmstadt - Main-Kinzig-Kreis
276070110 DE - Hessen - Darmstadt - Main-Taunus-Kreis
276070111 DE - Hessen - Darmstadt - Odenwaldkreis
276070112 DE - Hessen - Darmstadt - Offenbach, Landkreis
276070113 DE - Hessen - Darmstadt - Rheingau-Taunus-Kreis
276070114 DE - Hessen - Darmstadt - Wetteraukreis
276070201 DE - Hessen - Gießen - Gießen, Landkreis
276070202 DE - Hessen - Gießen - Lahn-Dill-Kreis
```

```
276070203
           DE - Hessen - Gießen - Limburg-Weilburg
276070204
           DE - Hessen - Gießen - Marburg-Biedenkopf
276070205
           DE - Hessen - Gießen - Vogelsbergkreis
276070301
           DE - Hessen - Kassel - Kassel, Kreisfreie Stadt
276070302 DE - Hessen - Kassel - Fulda
276070303 DE - Hessen - Kassel - Hersfeld-Rotenburg
276070304 DE - Hessen - Kassel - Kassel, Landkreis
276070305
           DE - Hessen - Kassel - Schwalm-Eder-Kreis
276070306 DE – Hessen – Kassel – Waldeck-Frankenberg
276070307
           DE - Hessen - Kassel - Werra-Meißner-Kreis
276080001 DE - Mecklenburg-Vorpommern - Greifswald, Kreisfreie Stadt
276080002
           DE - Mecklenburg-Vorpommern - Neubrandenburg, Kreisfreie Stadt
276080003 DE - Mecklenburg-Vorpommern - Rostock, Kreisfreie Stadt
           DE - Mecklenburg-Vorpommern - Schwerin, Kreisfreie Stadt
276080004
276080005 DE - Mecklenburg-Vorpommern - Stralsund, Kreisfreie Stadt
276080006
           DE - Mecklenburg-Vorpommern - Wismar, Kreisfreie Stadt
276080007
           DE - Mecklenburg-Vorpommern - Bad Doberan
276080008
           DE - Mecklenburg-Vorpommern - Demmin
276080009
           DE - Mecklenburg-Vorpommern - Güstrow
276080010
           DE - Mecklenburg-Vorpommern - Ludwigslust
276080011
           DE - Mecklenburg-Vorpommern - Mecklenburg-Strelitz
276080012
           DE - Mecklenburg-Vorpommern - Müritz
276080013
           DE - Mecklenburg-Vorpommern - Nordvorpommern
276080014
           DE - Mecklenburg-Vorpommern - Nordwestmecklenburg
276080015
           DE - Mecklenburg-Vorpommern - Ostvorpommern
276080016 DE – Mecklenburg-Vorpommern – Parchim
276080017
           DE - Mecklenburg-Vorpommern - Rügen
276080018 DE - Mecklenburg-Vorpommern - Uecker-Randow
276090101
           DE - Niedersachsen - Braunschweig - Braunschweig, Kreisfreie Stadt
276090102 DE - Niedersachsen - Braunschweig - Salzgitter, Kreisfreie Stadt
276090103
           DE - Niedersachsen - Braunschweig - Wolfsburg, Kreisfreie Stadt
276090104 DE - Niedersachsen - Braunschweig - Gifhorn
276090105 DE - Niedersachsen - Braunschweig - Göttingen
276090106 DE - Niedersachsen - Braunschweig - Goslar
276090107 DE - Niedersachsen - Braunschweig - Helmstedt
276090108 DE - Niedersachsen - Braunschweig - Northeim
276090109 DE - Niedersachsen - Braunschweig - Osterode am Harz
276090110 DE - Niedersachsen - Braunschweig - Peine
276090111 DE - Niedersachsen - Braunschweig - Wolfenbüttel
276090202 DE - Niedersachsen - Hannover - Diepholz
276090203 DE - Niedersachsen - Hannover - Hameln-Pyrmont
276090205 DE - Niedersachsen - Hannover - Hildesheim
276090206
           DE - Niedersachsen - Hannover - Holzminden
276090207
           DE - Niedersachsen - Hannover - Nienburg (Weser)
276090208
           DE - Niedersachsen - Hannover - Schaumburg
276090209
           DE - Niedersachsen - Hannover - Region Hannover
276090301
           DE - Niedersachsen - Lüneburg - Celle
276090302 DE – Niedersachsen – Lüneburg – Cuxhaven
276090303 DE - Niedersachsen - Lüneburg - Harburg
```

```
276090304
            DE - Niedersachsen - Lüneburg - Lüchow-Dannenberg
276090305
            DE - Niedersachsen - Lüneburg - Lüneburg, Landkreis
276090306
            DE - Niedersachsen - Lüneburg - Osterholz
276090307
            DE - Niedersachsen - Lüneburg - Rotenburg (Wümme)
276090308
            DE - Niedersachsen - Lüneburg - Soltau-Fallingbostel
276090309
            DE - Niedersachsen - Lüneburg - Stade
276090310 DE - Niedersachsen - Lüneburg - Uelzen
276090311
            DE - Niedersachsen - Lüneburg - Verden
276090401
            DE - Niedersachsen - Weser-Ems - Delmenhorst, Kreisfreie Stadt
276090402 DE - Niedersachsen - Weser-Ems - Emden, Kreisfreie Stadt
276090403
            DE - Niedersachsen - Weser-Ems - Oldenburg (Oldenburg), Kreisfreie Stadt
            DE - Niedersachsen - Weser-Ems - Osnabrück, Kreisfreie Stadt
276090404
276090405 DE - Niedersachsen - Weser-Ems - Wilhelmshaven, Kreisfreie Stadt
276090406 DE - Niedersachsen - Weser-Ems - Ammerland
276090407 DE - Niedersachsen - Weser-Ems - Aurich
276090408 DE - Niedersachsen - Weser-Ems - Cloppenburg
276090409 DE - Niedersachsen - Weser-Ems - Emsland
276090410 DE - Niedersachsen - Weser-Ems - Friesland (D)
276090411 DE - Niedersachsen - Weser-Ems - Grafschaft Bentheim
276090412 DE - Niedersachsen - Weser-Ems - Leer
276090413 DE - Niedersachsen - Weser-Ems - Oldenburg, Landkreis
276090414
            DE - Niedersachsen - Weser-Ems - Osnabrück, Landkreis
276090415 DE - Niedersachsen - Weser-Ems - Vechta
276090416 DE - Niedersachsen - Weser-Ems - Wesermarsch
            DE - Niedersachsen - Weser-Ems - Wittmund
276090417
276100101 DE - Nordrhein-Westfalen - Düsseldorf - Düsseldorf, Kreisfreie Stadt
276100102 DE - Nordrhein-Westfalen - Düsseldorf - Duisburg, Kreisfreie Stadt
276100103 DE - Nordrhein-Westfalen - Düsseldorf - Essen, Kreisfreie Stadt
276100104 DE - Nordrhein-Westfalen - Düsseldorf - Krefeld, Kreisfreie Stadt
276100105 DE – Nordrhein-Westfalen – Düsseldorf – Mönchengladbach, Kreisfreie Stadt
276100106
            DE - Nordrhein-Westfalen - Düsseldorf - Mülheim an der Ruhr. Kreisfreie Stadt
276100107 DE - Nordrhein-Westfalen - Düsseldorf - Oberhausen, Kreisfreie Stadt
276100108
            DE - Nordrhein-Westfalen - Düsseldorf - Remscheid, Kreisfreie Stadt
276100109 DE - Nordrhein-Westfalen - Düsseldorf - Solingen, Kreisfreie Stadt
276100110 DE - Nordrhein-Westfalen - Düsseldorf - Wuppertal, Kreisfreie Stadt
276100111 DE - Nordrhein-Westfalen - Düsseldorf - Kleve
276100112 DE - Nordrhein-Westfalen - Düsseldorf - Mettmann
276100113 DE - Nordrhein-Westfalen - Düsseldorf - Rhein-Kreis Neuss
276100114 DE – Nordrhein-Westfalen – Düsseldorf – Viersen
276100115 DE - Nordrhein-Westfalen - Düsseldorf - Wesel
276100201
            DE - Nordrhein-Westfalen - Köln - Aachen, Kreisfreie Stadt
276100202
            DE - Nordrhein-Westfalen - Köln - Bonn, Kreisfreie Stadt
276100203
            DE - Nordrhein-Westfalen - Köln - Köln, Kreisfreie Stadt
276100204
            DE - Nordrhein-Westfalen - Köln - Leverkusen, Kreisfreie Stadt
276100205 DE - Nordrhein-Westfalen - Köln - Aachen, Kreis
            DE - Nordrhein-Westfalen - Köln - Düren
276100206
276100207 DE - Nordrhein-Westfalen - Köln - Rhein-Erft-Kreis
276100208 DE - Nordrhein-Westfalen - Köln - Euskirchen
276100209 DE - Nordrhein-Westfalen - Köln - Heinsberg
```

```
276100210 DE - Nordrhein-Westfalen - Köln - Oberbergischer Kreis
276100211 DE - Nordrhein-Westfalen - Köln - Rheinisch-Bergischer Kreis
276100212
            DE - Nordrhein-Westfalen - Köln - Rhein-Sieg-Kreis
276100301
            DE - Nordrhein-Westfalen - Münster - Bottrop, Kreisfreie Stadt
276100302 DE - Nordrhein-Westfalen - Münster - Gelsenkirchen, Kreisfreie Stadt
276100303
            DE - Nordrhein-Westfalen - Münster - Münster, Kreisfreie Stadt
276100304 DE – Nordrhein-Westfalen – Münster – Borken
276100305
            DE - Nordrhein-Westfalen - Münster - Coesfeld
276100306 DE – Nordrhein-Westfalen – Münster – Recklinghausen
276100307
            DE - Nordrhein-Westfalen - Münster - Steinfurt
276100308 DE - Nordrhein-Westfalen - Münster - Warendorf
            DE - Nordrhein-Westfalen - Detmold - Bielefeld, Kreisfreie Stadt
276100401
276100402 DE - Nordrhein-Westfalen - Detmold - Gütersloh
276100403 DE - Nordrhein-Westfalen - Detmold - Herford
276100404 DE - Nordrhein-Westfalen - Detmold - Höxter
276100405 DE - Nordrhein-Westfalen - Detmold - Lippe
276100406 DE - Nordrhein-Westfalen - Detmold - Minden-Lübbecke
276100407 DE - Nordrhein-Westfalen - Detmold - Paderborn
276100501 DE - Nordrhein-Westfalen - Arnsberg - Bochum, Kreisfreie Stadt
276100502 DE - Nordrhein-Westfalen - Arnsberg - Dortmund, Kreisfreie Stadt
276100503 DE - Nordrhein-Westfalen - Arnsberg - Hagen, Kreisfreie Stadt
276100504
            DE - Nordrhein-Westfalen - Arnsberg - Hamm, Kreisfreie Stadt
276100505
            DE - Nordrhein-Westfalen - Arnsberg - Herne, Kreisfreie Stadt
276100506
            DE - Nordrhein-Westfalen - Arnsberg - Ennepe-Ruhr-Kreis
            DE - Nordrhein-Westfalen - Arnsberg - Hochsauerlandkreis
276100507
276100508 DE - Nordrhein-Westfalen - Arnsberg - Märkischer Kreis
276100509
            DE - Nordrhein-Westfalen - Arnsberg - Olpe
276100510 DE - Nordrhein-Westfalen - Arnsberg - Siegen-Wittgenstein
276100511 DE - Nordrhein-Westfalen - Arnsberg - Soest
276100512 DE - Nordrhein-Westfalen - Arnsberg - Unna
276110101 DE - Rheinland-Pfalz - Koblenz - Koblenz, Kreisfreie Stadt
276110102 DE - Rheinland-Pfalz - Koblenz - Ahrweiler
276110103 DE - Rheinland-Pfalz - Koblenz - Altenkirchen (Westerwald)
276110104 DE - Rheinland-Pfalz - Koblenz - Bad Kreuznach
276110105 DE - Rheinland-Pfalz - Koblenz - Birkenfeld
276110106 DE - Rheinland-Pfalz - Koblenz - Cochem-Zell
276110107 DE - Rheinland-Pfalz - Koblenz - Mayen-Koblenz
276110108 DE - Rheinland-Pfalz - Koblenz - Neuwied
276110109 DE - Rheinland-Pfalz - Koblenz - Rhein-Hunsrück-Kreis
276110110 DE - Rheinland-Pfalz - Koblenz - Rhein-Lahn-Kreis
276110111 DE - Rheinland-Pfalz - Koblenz - Westerwaldkreis
276110201 DE - Rheinland-Pfalz - Trier - Trier, Kreisfreie Stadt
276110202 DE - Rheinland-Pfalz - Trier - Bernkastel-Wittlich
276110203
            DE - Rheinland-Pfalz - Trier - Bitburg-Prüm
276110204 DE - Rheinland-Pfalz - Trier - Daun
276110205 DE - Rheinland-Pfalz - Trier - Trier-Saarburg
276110301 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Frankenthal (Pfalz), Kreisfreie Stadt
276110302 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Kaiserslautern, Kreisfreie Stadt
276110303 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Landau in der Pfalz, Kreisfreie Stadt
```

```
276110304 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Ludwigshafen am Rhein, Kreisfreie Stadt
276110305 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Mainz, Kreisfreie Stadt
276110306 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Neustadt an der Weinstraße, Kreisfreie Stadt
276110307
            DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Pirmasens, Kreisfreie Stadt
276110308 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Speyer, Kreisfreie Stadt
276110309 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Worms, Kreisfreie Stadt
276110310 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Zweibrücken, Kreisfreie Stadt
276110311 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Alzey-Worms
276110312 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Bad Dürkheim
276110313 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Donnersbergkreis
276110314 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Germersheim
276110315 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Kaiserslautern, Landkreis
276110316 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Kusel
276110317 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Südliche Weinstraße
276110318 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Rhein-Pfalz-Kreis
276110319 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Mainz-Bingen
276110320 DE - Rheinland-Pfalz - Rheinhessen-Pfalz - Südwestpfalz
276120001 DE - Saarland - Stadtverband Saarbrücken
276120002 DE - Saarland - Merzig-Wadern
276120003 DE - Saarland - Neunkirchen
276120004 DE - Saarland - Saarlouis
276120005 DE - Saarland - Saarpfalz-Kreis
276120006 DE - Saarland - St. Wendel
276130101 DE - Sachsen - Chemnitz - Chemnitz, Kreisfreie Stadt
276130102 DE - Sachsen - Chemnitz - Plauen, Kreisfreie Stadt
276130103 DE - Sachsen - Chemnitz - Zwickau, Kreisfreie Stadt
276130104 DE - Sachsen - Chemnitz - Annaberg
276130105 DE - Sachsen - Chemnitz - Chemnitzer Land
276130106 DE - Sachsen - Chemnitz - Freiberg
276130107 DE - Sachsen - Chemnitz - Vogtlandkreis
276130108 DE - Sachsen - Chemnitz - Mittlerer Erzgebirgskreis
276130109 DE - Sachsen - Chemnitz - Mittweida
276130110 DE - Sachsen - Chemnitz - Stollberg
276130111 DE - Sachsen - Chemnitz - Aue-Schwarzenberg
276130112 DE - Sachsen - Chemnitz - Zwickauer Land
276130201 DE - Sachsen - Dresden - Dresden, Kreisfreie Stadt
276130202 DE - Sachsen - Dresden - Görlitz, Kreisfreie Stadt
276130203 DE - Sachsen - Dresden - Hoyerswerda, Kreisfreie Stadt
276130204 DE - Sachsen - Dresden - Bautzen
276130205 DE - Sachsen - Dresden - Meißen
276130206 DE - Sachsen - Dresden - Niederschlesischer Oberlausitzkreis
276130207 DE - Sachsen - Dresden - Riesa-Großenhain
276130208 DE - Sachsen - Dresden - Löbau-Zittau
276130209 DE - Sachsen - Dresden - Sächsische Schweiz
276130210 DE - Sachsen - Dresden - Weißeritzkreis
276130211 DE - Sachsen - Dresden - Kamenz
276130301 DE - Sachsen - Leipzig - Leipzig, Kreisfreie Stadt
276130302 DE - Sachsen - Leipzig - Delitzsch
276130303 DE - Sachsen - Leipzig - Döbeln
```

```
276130304 DE - Sachsen - Leipzig - Leipziger Land
276130305 DE - Sachsen - Leipzig - Muldentalkreis
276130306
            DE - Sachsen - Leipzig - Torgau-Oschatz
276140001
            DE - Sachsen-Anhalt - Dessau-Roßlau, Kreisfreie Stadt
276140002 DE - Sachsen-Anhalt - Halle (Saale), Kreisfreie Stadt
276140003 DE - Sachsen-Anhalt - Magdeburg, Kreisfreie Stadt
276140004 DE - Sachsen-Anhalt - Altmarkkreis Salzwedel
276140005 DE - Sachsen-Anhalt - Anhalt-Bitterfeld
276140006 DE - Sachsen-Anhalt - Jerichower Land
276140007 DE - Sachsen-Anhalt - Börde
276140008 DE - Sachsen-Anhalt - Burgenland (D)
276140009 DE - Sachsen-Anhalt - Harz
276140010 DE - Sachsen-Anhalt - Mansfeld-Südharz
276140011 DE - Sachsen-Anhalt - Saalekreis
276140012 DE - Sachsen-Anhalt -Salzland
276140013 DE - Sachsen-Anhalt - Stendal
276140014 DE - Sachsen-Anhalt - Wittenberg
276150001 DE - Schleswig-Holstein - Flensburg, Kreisfreie Stadt
276150002 DE - Schleswig-Holstein - Kiel, Kreisfreie Stadt
276150003 DE - Schleswig-Holstein - Lübeck, Kreisfreie Stadt
276150004 DE - Schleswig-Holstein - Neumünster, Kreisfreie Stadt
276150005 DE - Schleswig-Holstein - Dithmarschen
276150006 DE - Schleswig-Holstein - Herzogtum Lauenburg
276150007 DE - Schleswig-Holstein - Nordfriesland
276150008 DE - Schleswig-Holstein - Ostholstein
276150009 DE - Schleswig-Holstein - Pinneberg
276150010 DE - Schleswig-Holstein - Plön
276150011 DE - Schleswig-Holstein - Rendsburg-Eckernförde
276150012 DE - Schleswig-Holstein - Schleswig-Flensburg
276150013 DE - Schleswig-Holstein - Segeberg
276150014 DE - Schleswig-Holstein - Steinburg
276150015 DE - Schleswig-Holstein - Stormarn
276160001 DE - Thüringen - Erfurt, Kreisfreie Stadt
276160002 DE - Thüringen - Gera, Kreisfreie Stadt
276160003 DE - Thüringen - Jena, Kreisfreie Stadt
276160004 DE - Thüringen - Suhl, Kreisfreie Stadt
276160005 DE - Thüringen - Weimar, Kreisfreie Stadt
276160006 DE - Thüringen - Eichsfeld
276160007 DE - Thüringen - Nordhausen
276160009 DE - Thüringen - Unstrut-Hainich-Kreis
276160010 DE - Thüringen - Kyffhäuserkreis
276160011
           DE - Thüringen - Schmalkalden-Meiningen
276160012 DE - Thüringen - Gotha
276160013 DE - Thüringen - Sömmerda
276160014 DE - Thüringen - Hildburghausen
276160015 DE - Thüringen - Ilm-Kreis
276160016 DE - Thüringen - Weimarer Land
276160017 DE - Thüringen - Sonneberg
276160018 DE - Thüringen - Saalfeld-Rudolstadt
```

```
276160019
            DE - Thüringen - Saale-Holzland-Kreis
276160020
            DE - Thüringen - Saale-Orla-Kreis
276160021
            DE - Thüringen - Greiz
276160022
            DE - Thüringen - Altenburger Land
276160023
            DE - Thüringen - Eisenach, Kreisfreie Stadt
276160025 DE - Thüringen - Wartburgkreis
300010101 GR - Voreia Ellada - Anatoliki Makedonia, Thraki - Evros
300010102 GR - Voreia Ellada - Anatoliki Makedonia, Thraki - Xanthi
300010103 GR - Voreia Ellada - Anatoliki Makedonia, Thraki - Rodopi
300010104
            GR - Voreia Ellada - Anatoliki Makedonia, Thraki - Drama
300010105
            GR - Voreia Ellada - Anatoliki Makedonia, Thraki - Kavala
300010201
            GR - Voreia Ellada - Kentriki Makedonia - Imathia
300010202 GR - Voreia Ellada - Kentriki Makedonia - Thessaloniki
300010203
            GR - Voreia Ellada - Kentriki Makedonia - Kilkis
300010204
            GR - Voreia Ellada - Kentriki Makedonia - Pella
300010205
            GR - Voreia Ellada - Kentriki Makedonia - Pieria
300010206 GR - Voreia Ellada - Kentriki Makedonia - Serres
300010207
            GR - Voreia Ellada - Kentriki Makedonia - Chalkidiki
300010301
            GR - Voreia Ellada - Dytiki Makedonia - Grevena
300010302
            GR - Voreia Ellada - Dytiki Makedonia - Kastoria
300010303
            GR - Voreia Ellada - Dytiki Makedonia - Kozani
300010304
            GR - Voreia Ellada - Dytiki Makedonia - Florina
300010401
            GR - Voreia Ellada - Thessalia - Karditsa
300010402
            GR - Voreia Ellada - Thessalia - Larisa
300010403
            GR - Voreia Ellada - Thessalia - Magnisia
300010404
            GR - Voreia Ellada - Thessalia - Trikala
300020101
            GR - Kentriki Ellada - Ipeiros - Arta
300020102 GR - Kentriki Ellada - Ipeiros - Thesprotia
300020103
            GR - Kentriki Ellada - Ipeiros - Ioannina
300020104
            GR - Kentriki Ellada - Ipeiros - Preveza
300020201
            GR - Kentriki Ellada - Ionia Nisia - Zakynthos
300020202
            GR - Kentriki Ellada - Ionia Nisia - Kerkyra
300020203
            GR - Kentriki Ellada - Ionia Nisia - Kefallinia
            GR - Kentriki Ellada - Ionia Nisia - Lefkada
300020204
            GR - Kentriki Ellada - Dytiki Ellada - Aitoloakarnania
300020301
300020302
            GR - Kentriki Ellada - Dytiki Ellada - Achaia
300020303 GR - Kentriki Ellada - Dytiki Ellada - Ileia
300020401
            GR - Kentriki Ellada - Sterea Ellada - Voiotia
300020402 GR - Kentriki Ellada - Sterea Ellada - Evvoia
300020403
            GR - Kentriki Ellada - Sterea Ellada - Evrytania
300020404
            GR - Kentriki Ellada - Sterea Ellada - Fthiotida
300020405
            GR - Kentriki Ellada - Sterea Ellada - Fokida
300020501
            GR - Kentriki Ellada - Peloponnisos - Argolida
300020502
            GR - Kentriki Ellada - Peloponnisos - Arkadia
300020503
            GR - Kentriki Ellada - Peloponnisos - Korinthia
300020504
            GR - Kentriki Ellada - Peloponnisos - Lakonia
300020505
            GR - Kentriki Ellada - Peloponnisos - Messinia
300030000
            GR - Attiki
300040101
            GR - Nisia Aigaiou, Kriti - Voreio Aigaio - Lesvos
```

```
300040102 GR - Nisia Aigaiou, Kriti - Voreio Aigaio - Samos
300040103
            GR - Nisia Aigaiou, Kriti - Voreio Aigaio - Chios
300040201
            GR - Nisia Aigaiou, Kriti - Notio Aigaio - Dodekanisos
300040202
            GR - Nisia Aigaiou, Kriti - Notio Aigaio - Kyklades
300040301
            GR - Nisia Aigaiou, Kriti - Kriti - Irakleio
300040302
            GR - Nisia Aigaiou, Kriti - Kriti - Lasithi
300040303 GR - Nisia Aigaiou, Kriti - Kriti - Rethymni
300040304
            GR - Nisia Aigaiou, Kriti - Kriti - Chania
348010001 HU - Közép-Magyarország - Budapest
348010002 HU - Közép-Magyarország - Pest
348020101 HU - Dunántúl - Közép-Dunántúl - Fejér
348020102 HU – Dunántúl – Közép-Dunántúl – Komárom-Esztergom
348020103 HU – Dunántúl – Közép-Dunántúl – Veszprém
348020201 HU – Dunántúl – Nyugat-Dunántúl – Gyor-Moson-Sopron
348020202 HU - Dunántúl - Nyugat-Dunántúl - Vas
348020203 HU - Dunántúl - Nyugat-Dunántúl - Zala
348020301 HU - Dunántúl - Dél-Dunántúl - Baranya
348020302 HU - Dunántúl - Dél-Dunántúl - Somogy
348020303 HU - Dunántúl - Dél-Dunántúl - Tolna
348030101 HU – Alföld és Észak – Észak-Magyarország – Borsod-Abaúj-Zemplén
348030102 HU - Alföld és Észak - Észak-Magyarország - Heves
348030103 HU - Alföld és Észak - Észak-Magyarország - Nógrád
348030201
            HU - Alföld és Észak - Észak-Alföld - Hajdú-Bihar
348030202 HU - Alföld és Észak - Észak-Alföld - Jász-Nagykun-Szolnok
            HU - Alföld és Észak - Észak-Alföld - Szabolcs-Szatmár-Bereg
348030203
348030301 HU - Alföld és Észak - Dél-Alföld - Bács-Kiskun
348030302 HU - Alföld és Észak - Dél-Alföld - Békés
348030303 HU - Alföld és Észak - Dél-Alföld - Csongrád
352000001
            IS – Ísland – Höfuðborgarsvæði
352000002 IS - Ísland - Landsbyggð
372000101
            IE - Ireland - Border, Midland and Western - Border
372000102 IE - Ireland - Border, Midland and Western - Midland
372000103 IE - Ireland - Border, Midland and Western - West
372000201 IE - Ireland - Southern and Eastern - Dublin
372000202 IE - Ireland - Southern and Eastern - Mid-East
372000203 IE - Ireland - Southern and Eastern - Mid-West
372000204 IE - Ireland - Southern and Eastern - South-East (IRL)
372000205 IE - Ireland - Southern and Eastern - South-West (IRL)
380120101 IT - Nord-Ovest - Piemonte - Torino
380120102 IT - Nord-Ovest - Piemonte - Vercelli
380120103 IT - Nord-Ovest - Piemonte - Biella
380120104 IT - Nord-Ovest - Piemonte - Verbano-Cusio-Ossola
380120105 IT - Nord-Ovest - Piemonte - Novara
380120106 IT - Nord-Ovest - Piemonte - Cuneo
380120107 IT - Nord-Ovest - Piemonte - Asti
380120108 IT - Nord-Ovest - Piemonte - Alessandria
380120200 IT - Nord-Ovest - Valle d'Aosta/Vallée d'Aoste
380120301 IT - Nord-Ovest - Liguria - Imperia
380120302 IT - Nord-Ovest - Liguria - Savona
```

```
380120303 IT - Nord-Ovest - Liguria - Genova
380120304 IT - Nord-Ovest - Liguria - La Spezia
380120401 IT - Nord-Ovest - Lombardia - Varese
380120402 IT - Nord-Ovest - Lombardia - Como
380120403 IT - Nord-Ovest - Lombardia - Lecco
380120404 IT - Nord-Ovest - Lombardia - Sondrio
380120405 IT - Nord-Ovest - Lombardia - Milano
380120406 IT - Nord-Ovest - Lombardia - Bergamo
380120407 IT - Nord-Ovest - Lombardia - Brescia
380120408 IT - Nord-Ovest - Lombardia - Pavia
380120409 IT - Nord-Ovest - Lombardia - Lodi
380120410 IT - Nord-Ovest - Lombardia - Cremona
380120411 IT - Nord-Ovest - Lombardia - Mantova
380130100 IT - Nord-Est - Provincia Autonoma Bolzano/Bozen - Bolzano-Bozen
380130200 IT - Nord-Est - Provincia Autonoma Trento - Trento
380130301 IT - Nord-Est - Veneto - Verona
380130302 IT - Nord-Est - Veneto - Vicenza
380130303 IT - Nord-Est - Veneto - Belluno
380130304 IT - Nord-Est - Veneto - Treviso
380130305 IT - Nord-Est - Veneto - Venezia
380130306 IT - Nord-Est - Veneto - Padova
380130307 IT - Nord-Est - Veneto - Rovigo
380130401 IT - Nord-Est - Friuli-Venezia Giulia - Pordenone
380130402 IT - Nord-Est - Friuli-Venezia Giulia - Udine
380130403 IT - Nord-Est - Friuli-Venezia Giulia - Gorizia
380130404 IT - Nord-Est - Friuli-Venezia Giulia - Trieste
380130501 IT - Nord-Est - Emilia-Romagna - Piacenza
380130502 IT - Nord-Est - Emilia-Romagna - Parma
380130503 IT - Nord-Est - Emilia-Romagna - Reggio nell'Emilia
380130504 IT - Nord-Est - Emilia-Romagna - Modena
380130505 IT - Nord-Est - Emilia-Romagna - Bologna
380130506 IT - Nord-Est - Emilia-Romagna - Ferrara
380130507 IT - Nord-Est - Emilia-Romagna - Ravenna
380130508 IT - Nord-Est - Emilia-Romagna - Forlì-Cesena
380130509 IT - Nord-Est - Emilia-Romagna - Rimini
380140101 IT - Centro (I) - Toscana - Massa-Carrara
380140102 IT - Centro (I) - Toscana - Lucca
380140103 IT - Centro (I) - Toscana - Pistoia
380140104 IT - Centro (I) - Toscana - Firenze
380140105 IT - Centro (I) - Toscana - Prato
380140106 IT - Centro (I) - Toscana - Livorno
380140107 IT - Centro (I) - Toscana - Pisa
380140108 IT - Centro (I) - Toscana - Arezzo
380140109 IT - Centro (I) - Toscana - Siena
380140110 IT - Centro (I) - Toscana - Grosseto
380140201 IT - Centro (I) - Umbria - Perugia
380140202 IT - Centro (I) - Umbria - Terni
380140301 IT - Centro (I) - Marche - Pesaro e Urbino
380140302 IT - Centro (I) - Marche - Ancona
```

```
380140303 IT - Centro (I) - Marche - Macerata
380140304 IT - Centro (I) - Marche - Ascoli Piceno
380140401 IT - Centro (I) - Lazio - Viterbo
380140402 IT - Centro (I) - Lazio - Rieti
380140403 IT - Centro (I) - Lazio - Roma
380140404 IT - Centro (I) - Lazio - Latina
380140405 IT - Centro (I) - Lazio - Frosinone
380150001 IT - Sud - Calabria - Cosenza
380150002 IT - Sud - Calabria - Crotone
380150003 IT - Sud - Calabria - Catanzaro
380150004 IT - Sud - Calabria - Vibo Valentia
380150005 IT - Sud - Calabria - Reggio di Calabria
380150101 IT - Sud - Abruzzo - L'Aquila
380150102 IT - Sud - Abruzzo - Teramo
380150103 IT - Sud - Abruzzo - Pescara
380150104 IT - Sud - Abruzzo - Chieti
380150201 IT - Sud - Molise - Isernia
380150202 IT - Sud - Molise - Campobasso
380150301 IT - Sud - Campania - Caserta
380150302 IT - Sud - Campania - Benevento
380150303 IT - Sud - Campania - Napoli
380150304 IT - Sud - Campania - Avellino
380150305 IT - Sud - Campania - Salerno
380150401 IT - Sud - Puglia - Foggia
380150402 IT - Sud - Puglia - Bari
380150403 IT - Sud - Puglia - Taranto
380150404 IT - Sud - Puglia - Brindisi
380150405 IT - Sud - Puglia - Lecce
380150501 IT - Sud - Basilicata - Potenza
380150502 IT - Sud - Basilicata - Matera
380160101 IT - Isole - Sicilia - Trapani
380160102 IT - Isole - Sicilia - Palermo
380160103 IT - Isole - Sicilia - Messina
380160104 IT - Isole - Sicilia - Agrigento
380160105 IT - Isole - Sicilia - Caltanissetta
380160106 IT - Isole - Sicilia - Enna
380160107 IT - Isole - Sicilia - Catania
380160108 IT - Isole - Sicilia - Ragusa
380160109 IT - Isole - Sicilia Siracusa
380160205 IT - Isole - Sardegna - Sassari
380160206 IT - Isole - Sardegna - Nuoro
380160207 IT - Isole - Sardegna - Cagliari
380160208 IT - Isole - Sardegna - Oristano
380160209 IT - Isole - Sardegna - Olbia-Tempio
380160210 IT - Isole - Sardegna - Ogliastra
380160211 IT - Isole - Sardegna - Medio Campidano
380160212 IT - Isole - Sardegna - Carbonia-Iglesias
428000003 LV - Latvija - Kurzeme
428000005 LV - Latvija - Latgale
```

```
428000006 LV - Latvija - Riga
428000007
            LV - Latvija - Pieriga
428000008
            LV - Latvija - Vidzeme
428000009
            LV - Latvija - Zemgale
440000001 LT - Lietuva - Alytaus apskritis
440000002 LT - Lietuva - Kauno apskritis
440000003 LT - Lietuva - Klaipedos apskritis
440000004 LT - Lietuva - Marijampoles apskritis
44000005 LT - Lietuva - Panevežio apskritis
44000006 LT - Lietuva - Šiauliu apskritis
440000007 LT - Lietuva - Taurages apskritis
440000008 LT - Lietuva - Telšiu apskritis
440000009 LT - Lietuva - Utenos apskritis
440000010 LT - Lietuva - Vilniaus apskritis
442000000 LU - Luxembourg (Grand-Duché)
470000001
            MT- Malta - Malta
470000002 MT- Malta - Gozo and Comino / Ghawdex u Kemmuna
498000001
            MD - Moldova - Cahul and Cantemir
498000002 MD - Moldova - Taraclia
498000003
            MD - Moldova - Autonomous Territorial Unit of Gagauzia and Comrat
498000004
            MD - Moldova - Basarabeasca, Hincesti, Leova and Cimislia
498000005
            MD - Moldova - Causeni, Stefan Voda
498000006
            MD - Moldova - Anenii Noi, Criuleni, Dubasari, Ialoveni and Straseni
498000007
            MD - Moldova - Chisinau
            MD - Moldova - Orhei, Rezina, Soldanesti, Telenesti
498000008
498000009
            MD - Moldova - Ungheni, Calarasi, and Nisporeni
498000010
            MD - Moldova - Balti, Falesti, Glodeni, Riscani, Singerei
498000011
            MD - Moldova - Soroca, Drochia and Floresti
498000012 MD - Moldova - Briceni, Edinet, Ocnita and Donduseni
498000013 MD - Moldova - Autonomous territorial unit of Transnistria and Bender
            ME - Montenegro - North - Beransko-Rozajski
499000101
499000102 ME - Montenegro - North - Bijelopoljsko-Pljevaljski
499000103
           ME - Montenegro - North - Niksicko-Zabljacki
            ME - Montenegro - Center - Podgoricko-Cetinjski
499000201
499000301
            ME - Montenegro - South - Barsko-Ulcinjski
           ME - Montenegro - South - Boka Kotorska
499000302
528010101
            NL - Noord-Nederland - Groningen - Oost-Groningen
528010102 NL - Noord-Nederland - Groningen - Delfzijl en omgeving
528010103 NL - Noord-Nederland - Groningen - Overig Groningen
528010201
            NL - Noord-Nederland - Friesland (NL) - Noord-Friesland
528010202 NL - Noord-Nederland - Friesland (NL) - Zuidwest-Friesland
528010203
            NL - Noord-Nederland - Friesland (NL) - Zuidoost-Friesland
528010301
            NL - Noord-Nederland - Drenthe - Noord-Drenthe
528010302
            NL - Noord-Nederland - Drenthe - Zuidsoost-Drenthe
528010303
            NL - Noord-Nederland - Drenthe - Zuidwest-Drenthe
528020101
            NL - Oost-Nederland - Overijssel - Noord-Overijssel
528020102 NL - Oost-Nederland - Overijssel - Zuidwest-Overijssel
528020103
            NL - Oost-Nederland - Overijssel - Twente
528020201 NL - Oost-Nederland - Gelderland - Veluwe
```

```
528020204
            NL - Oost-Nederland - Gelderland - Zuidwest-Gelderland
528020205
            NL - Oost-Nederland - Gelderland - Achterhoek
528020206
            NL - Oost-Nederland - Gelderland - Arnhem/Nijmegen
528020300
            NL - Oost-Nederland - Flevoland
528030100
           NL - West-Nederland - Utrecht
            NL - West-Nederland - Noord-Holland - Kop van Noord-Holland
528030201
528030202 NL - West-Nederland - Noord-Holland - Alkmaar en omgeving
528030203
           NL - West-Nederland - Noord-Holland - Ijmond
528030204
           NL - West-Nederland - Noord-Holland - Agglomeratie Haarlem
528030205
           NL - West-Nederland - Noord-Holland - Zaanstreek
528030206 NL - West-Nederland - Noord-Holland - Groot-Amsterdam
           NL - West-Nederland - Noord-Holland - Het Gooi en Vechtstreek
528030207
528030301 NL - West-Nederland - Zuid-Holland - Agglomeratie Leiden en Bollenstreek
528030302 NL – West-Nederland – Zuid-Holland – Agglomeratie 's-Gravenhage
528030303 NL - West-Nederland - Zuid-Holland - Delft en Westlijn
528030304
           NL - West-Nederland - Zuid-Holland - Oost-Zuid-Holland
528030305 NL - West-Nederland - Zuid-Holland - Groot-Rijnmond
528030306 NL - West-Nederland - Zuid-Holland - Zuidoost-Zuid-Holland
528030401 NL - West-Nederland - Zeeland - Zeeuwsch-Vlaanderen
528030402
           NL - West-Nederland - Zeeland - Overig Zeeland
528040101
           NL - Zuid-Nederland - Noord-Brabant - West-Noord-Brabant
528040102 NL - Zuid-Nederland - Noord-Brabant - Midden-Noord-Brabant
528040103
            NL - Zuid-Nederland - Noord-Brabant - Noordoost-Noord-Brabant
           NL - Zuid-Nederland - Noord-Brabant - Zuidoost-Noord-Brabant
528040104
            NL - Zuid-Nederland - Limburg (NL) - Noord-Limburg
528040201
528040202 NL - Zuid-Nederland - Limburg (NL) - Midden-Limburg
528040203 NL - Zuid-Nederland - Limburg (NL) - Zuid-Limburg
578000001 NO - Norge - Trøndelag - Sør-Trøndelag
578000002 NO - Norge - Trøndelag - Nord-Trøndelag
578000101 NO - Norge - Oslo og Akershus - Oslo
578000102 NO - Norge - Oslo og Akershus - Akershus
578000201
           NO - Norge - Hedmark og Oppland - Hedmark
578000202 NO - Norge - Hedmark og Oppland - Oplland
578000301 NO - Norge - Sør-Østlandet - Østfold
578000302
           NO - Norge - Sør-Østlandet - Buskerud
578000303 NO - Norge - Sør-Østlandet - Vestfold
578000304
           NO - Norge - Sør-Østlandet - Telemark
578000401 NO - Norge - Agder og Rogaland - Aust-Agder
578000402
           NO - Norge - Agder og Rogaland - Vest-Agder
578000403 NO - Norge - Agder og Rogaland - Rogaland
578000501
           NO - Norge - Vestlandet - Hordaland
578000502
           NO - Norge - Vestlandet - Sogn og Fjordane
578000503
           NO - Norge - Vestlandet Møre og Romsdal
578000701
           NO - Norge - Nord-Norge - Nordland
578000702 NO - Norge - Nord-Norge - Troms
578000703
           NO - Norge - Nord-Norge - Finnmark
616010103 PL - Region centralny - Lódzkie - Miasto Lódz
616010104 PL - Region centralny - Lódzkie - Lódzki
616010105 PL - Region centralny - Lódzkie - Piotrkowski
```

```
616010106 PL - Region centralny - Lódzkie - Sieradzki
616010107 PL - Region centralny - Lódzkie - Skierniewicki
616010201 PL - Region centralny - Mazowieckie - Ciechanowsko-plocki
616010202 PL - Region centralny - Mazowieckie - Ostrolecko-siedlecki
616010207 PL - Region centralny - Mazowieckie - Miasto Warszawa
616010208 PL - Region centralny - Mazowieckie - Radomski
616010209 PL - Region centralny - Mazowieckie - Warszawski-wschodni
616010210 PL - Region centralny - Mazowieckie - Warszawski-zachodni
616020103 PL - Region Poludinowy - Malopolskie - Miasto Kraków
616020104 PL - Region Poludinowy - Malopolskie - Krakowski
616020105 PL - Region Poludinowy - Malopolskie - Nowosadecki
616020106 PL - Region Poludinowy - Malopolskie - Oswiecimski
616020107 PL - Region Poludinowy - Malopolskie - Tarnowski
616020204 PL - Region Poludinowy - Slaskie - Czestochowski
616020205 PL - Region Poludinowy - Slaskie - Bielski
616020207 PL - Region Poludinowy - Slaskie - Rybnicki
616020208 PL – Region Poludinowy – Slaskie – Bytomski
616020209 PL - Region Poludinowy - Slaskie - Gliwicki
616020210 PL - Region Poludinowy - Slaskie - Katowicki
616020211 PL - Region Poludinowy - Slaskie - Sosnowiecki
616020212 PL - Region Poludinowy - Slaskie - Tyski
616030101 PL - Region Wschodni - Lubelskie - Bialski
616030102 PL - Region Wschodni - Lubelskie - Chelmsko-zamojski
616030104 PL - Region Wschodni - Lubelskie - Lubelski
616030105 PL - Region Wschodni - Lubelskie - Pulawski
616030203 PL - Region Wschodni - Podkarpackie - Krosnienski
616030204 PL - Region Wschodni - Podkarpackie - Przemyski
616030205 PL - Region Wschodni - Podkarpackie - Rzeszowski
616030206 PL – Region Wschodni – Podkarpackie – Tarnobrzeski
616030301 PL - Region Wschodni - Swietokrzyskie - Kielecki
616030302 PL - Region Wschodni - Swietokrzyskie - Sandomiersko-jedrzejowski
616030403 PL - Region Wschodni - Podlaskie - Bialostocki
616030404 PL - Region Wschodni - Podlaskie - Lomzynski
616030405 PL - Region Wschodni - Podlaskie - Suwalski
616040101 PL – Region Pólnocno-Zachodni – Wielkopolskie – Pilski
616040104 PL - Region Pólnocno-Zachodni - Wielkopolskie - Koninski
616040105 PL – Region Pólnocno-Zachodni – Wielkopolskie – Miasto Poznan
616040106 PL - Region Pólnocno-Zachodni - Wielkopolskie - Kaliski
616040107 PL – Region Pólnocno-Zachodni – Wielkopolskie – Leszczynski
616040108 PL - Region Pólnocno-Zachodni - Wielkopolskie - Poznanski
616040202 PL – Region Pólnocno-Zachodni – Zachodniopomorskie – Koszalinski
616040203 PL – Region Pólnocno-Zachodni – Zachodniopomorskie – Stargardzki
616040204 PL - Region Pólnocno-Zachodni - Zachodniopomorskie - Miasto Szczecin
616040205 PL - Region Pólnocno-Zachodni - Zachodniopomorskie - Szczecinski
616040301 PL - Region Pólnocno-Zachodni - Lubuskie - Gorzowski
616040302 PL – Region Pólnocno-Zachodni – Lubuskie – Zielonogórski
616050104 PL - Region Póludniowo-Zachodni - Dolnoslaskie - Miasto Wroclaw
616050105 PL – Region Póludniowo-Zachodni – Dolnoslaskie – Jeleniogórski
616050106 PL - Region Póludniowo-Zachodni - Dolnoslaskie - Legnicko-Glogowski
```

```
616050107 PL – Region Póludniowo-Zachodni – Dolnoslaskie – Walbrzyski
616050108 PL - Region Póludniowo-Zachodni - Dolnoslaskie - Wroclawski
616050201 PL - Region Póludniowo-Zachodni - Opolskie - Nyski
616050202 PL - Region Póludniowo-Zachodni - Opolskie - Opolski
616060103 PL - Region Pólnocny - Kujawsko-Pomorskie - Bydgosko-Torunski
616060104 PL - Region Pólnocny - Kujawsko-Pomorskie - Grudziadzki
616060105 PL – Region Pólnocny – Kujawsko-Pomorskie – Wlocławski
616060201 PL – Region Pólnocny – Warminsko-Mazurskie – Elblaski
616060202 PL – Region Pólnocny – Warminsko-Mazurskie – Olsztynski
616060203 PL – Region Pólnocny – Warminsko-Mazurskie – Elcki
616060301 PL - Region Pólnocny - Pomorskie - Slupski
616060303 PL - Region Pólnocny - Pomorskie - Trójmiejski
616060304 PL - Region Pólnocny - Pomorskie - Gdanski
616060305 PL - Region Pólnocny - Pomorskie - Starogardzki
620010101 PT - Continente - Northe - Minho-Lima
620010102 PT - Continente - Northe - Cávado
620010103 PT - Continente - Northe - Ave
620010104 PT - Continente - Northe - Grande Porto
620010105 PT - Continente - Northe - Tâmega
620010106 PT - Continente - Northe - Entre Douro e Vouga
620010107 PT - Continente - Northe - Douro
620010108 PT - Continente - Northe - Alto Trás-os-Montes
620010500 PT - Continente - Algarve
620010601 PT - Continente - Centro (P) - Baixo Vouga
620010602 PT - Continente - Centro (P) - Baixo Mondego
620010603 PT - Continente - Centro (P) - Pinhal Litoral
620010604 PT - Continente - Centro (P) - Pinhal Interior Norte
620010605 PT - Continente - Centro (P) - Dâo-Lafôes
620010606 PT - Continente - Centro (P) - Pinhal Interior Sul
620010607 PT - Continente - Centro (P) - Serra da Estrela
620010608 PT - Continente - Centro (P) - Beira Interior Norte
620010609 PT - Continente - Centro (P) - Beira Interior Sul
620010610 PT - Continente - Centro (P) - Cova de Beira
620010611 PT - Continente - Centro (P) - Oeste
620010612 PT - Continente - Centro (P) - Médio Tejo
620010701 PT - Continente - Lisboa - Grande Lisboa
620010702 PT - Continente - Lisboa - Península de Setúbal
620010801 PT - Continente - Alentejo - Alentejo Litoral
620010802 PT - Continente - Alentejo - Alto Alentejo
620010803 PT - Continente - Alentejo - Alentejo Central
620010804 PT - Continente - Alentejo - Baixo Alentejo
620010805 PT - Continente - Alentejo - Lezíra do Tejo
620020000 PT - Região Autónoma dos Açores
620030000 PT - Região Autónoma da Madeira
642010101 RO - Macroregiunea unu - Nord-Vest - Bihor
642010102 RO - Macroregiunea unu - Nord-Vest - Bistrita-Nasaud
642010103 RO – Macroregiunea unu – Nord-Vest – Cluj
642010104 RO - Macroregiunea unu - Nord-Vest - Maramures
642010105 RO - Macroregiunea unu - Nord-Vest - Satu Mare
```

```
642010106
           RO - Macroregiunea unu - Nord-Vest - Salaj
642010201
            RO - Macroregiunea unu - Centru - Alba
642010202
            RO - Macroregiunea unu - Centru - Brasov
642010203
            RO - Macroregiunea unu - Centru - Covasna
642010204
            RO - Macroregiunea unu - Centru - Harghita
642010205
            RO - Macroregiunea unu - Centru - Mures
642010206 RO - Macroregiunea unu - Centru - Sibiu
642020101
            RO - Macroregiunea doi - Nord-Est - Bacau
642020102 RO – Macroregiunea doi – Nord-Est – Botosani
642020103
            RO - Macroregiunea doi - Nord-Est - Iasi
642020104 RO - Macroregiunea doi - Nord-Est - Neamt
642020105
            RO - Macroregiunea doi - Nord-Est - Suceava
642020106 RO - Macroregiunea doi - Nord-Est - Vaslui
642020201
            RO - Macroregiunea doi - Sud-Est - Braila
642020202
           RO - Macroregiunea doi - Sud-Est - Buzau
642020203
            RO - Macroregiunea doi - Sud-Est - Constanta
642020204
            RO - Macroregiunea doi - Sud-Est - Galati
642020205 RO - Macroregiunea doi - Sud-Est - Tulcea
642020206 RO - Macroregiunea doi - Sud-Est - Vrancea
642030101
            RO - Macroregiunea trei - Sud-Muntenia - Arges
642030102
            RO – Macroregiunea trei – Sud-Muntenia – Calarasi
642030103
            RO - Macroregiunea trei - Sud-Muntenia - Dâmbovita
642030104
            RO - Macroregiunea trei - Sud-Muntenia - Giurgiu
642030105
            RO - Macroregiunea trei - Sud-Muntenia - Ialomita
642030106
            RO - Macroregiunea trei - Sud-Muntenia - Prahova
642030107
            RO – Macroregiunea trei – Sud-Muntenia – Teleorman
642030201
            RO - Macroregiunea trei - Bucuresti-Ilfov - Bucuresti
642030202 RO - Macroregiunea trei - Bucuresti-Ilfov - Ilfov
642040101
            RO - Macroregiunea patru - Sud-Vest Oltenia - Dolj
642040102 RO - Macroregiunea patru - Sud-Vest Oltenia - Gorj
642040103
            RO - Macroregiunea patru - Sud-Vest Oltenia - Mehedinti
642040104
            RO - Macroregiunea patru - Sud-Vest Oltenia - Olt
642040105
           RO - Macroregiunea patru - Sud-Vest Oltenia - Vâlcea
642040201
            RO – Macroregiunea patru – Vest – Arad
642040202
            RO - Macroregiunea patru - Vest - Caras-Severin
642040203 RO - Macroregiunea patru - Vest - Hunedoara
642040204
            RO - Macroregiunea patru - Vest - Timis
643010101 RU - Central Federal District - Belgorodskaya oblast
643010102 RU - Central Federal District - Bryanskaya oblast
643010103 RU - Central Federal District - Vladimirskaya oblast
643010104 RU - Central Federal District - Voronezhskaya oblast
643010105 RU - Central Federal District - Ivanovskaya oblast
643010106 RU - Central Federal District - Kaluzhskaya oblast
643010107
            RU - Central Federal District - Kostromskaya oblast
643010108
            RU - Central Federal District - Kurskaya oblast
643010109
            RU - Central Federal District - Lipetskaya oblast
643010110 RU - Central Federal District - Moscow
643010111 RU - Central Federal District - Moscowskaya oblast
643010112 RU - Central Federal District - Orlovskaya oblast
```

```
643010113 RU - Central Federal District - Ryazanskaya oblast
643010114
            RU - Central Federal District - Smolenskaya oblast
643010115
            RU - Central Federal District - Tambovskaya oblast
643010116
            RU - Central Federal District - Tverskaya oblast
643010117
            RU - Central Federal District - Tulskaya oblast
643010118 RU - Central Federal District - Yaroslavskaya oblast
643020101 RU - North West federal district - Republic of Kareliya
643020102 RU - North West federal district - Republic of Komi
643020103 RU - North West federal district - Arkhangelskaya oblast
643020104 RU - North West federal district - Vologodskaya oblast
643020105 RU - North West federal district - Kaliningradskaya oblast
643020106
            RU - North West federal district - Leningradskaya oblast
643020107 RU - North West federal district - Murmanskaya oblast
643020108 RU - North West federal district - Novgorodskaya oblast
643020109
            RU - North West federal district - Pskovskaya oblast
643020110 RU - North West federal district - St. Petersburg
643020111 RU - North West federal district - Nenetsky autonomous okrug
643030101 RU - South Federal district - Republic of Adygeya
643030102 RU - South Federal district - Republic of Dagestan
643030103 RU - South Federal district - Republic of Ingushetiya
643030104 RU - South Federal district - Republic of Kabardino-Balkariya
643030105 RU - South Federal district - Republic of Kalmykiya
643030106
            RU - South Federal district - Republic of Karachaevo-Cherkessia
643030107
            RU - South Federal district - Republic of North Ossetia
643030108
            RU - South Federal district - Chechen Republic
643030109 RU - South Federal district - Krasnodarsky krai
643030110 RU - South Federal district - Stavropolsky krai
643030111 RU - South Federal district - Astrakhanskaya oblast
643030112 RU - South Federal district - Volgogradskaya oblast
643030113 RU - South Federal district - Rostovskaya oblast
643040101
            RU - Privolzhsky federal district - Republic of Bashkortostan
643040102 RU - Privolzhsky federal district - Republic of Mary-El
643040103 RU - Privolzhsky federal district - Republic of Mordoviya
643040104 RU – Privolzhsky federal district – Republic of Tatarstan
643040105 RU - Privolzhsky federal district - Republic of Udmurtiya
643040106 RU - Privolzhsky federal district - Republic of Chuvashiya
643040107 RU - Privolzhsky federal district - Permsky krai
643040108 RU - Privolzhsky federal district - Kirovskaya oblast
643040109
            RU - Privolzhsky federal district - Nizhegorodskaya oblast
643040110 RU - Privolzhsky federal district - Orenburgskaya oblast
643040111
            RU - Privolzhsky federal district - Penzenskaya oblast
643040112 RU - Privolzhsky federal district - Samarskaya oblast
643040113 RU - Privolzhsky federal district - Saratovskaya oblast
643040114
            RU - Privolzhsky federal district - Ulyanovskaya oblast
643050101 RU - Urals federal district - Sverdlovskaya oblast
643050102
            RU - Urals federal district - Kurganskaya oblast
643050103 RU - Urals federal district - Tyumenskaya oblast
643050104 RU - Urals federal district - Chelyabinskaya oblast
643050105 RU - Urals federal district - Yamalo-Nenetsky autonomous okrug
```

```
643050106 RU - Urals federal district - Khanty-Mansiysky autonomous okrug
643060101
            RU - Siberian federal district - Altai Republic
643060102
            RU - Siberian federal district - Republic of Khakassia
643060103
            RU - Siberian federal district - Republic of Buryatia
643060104
            RU - Siberian federal district - Tyva Republic
643060105 RU - Siberian federal district - Altai krai
643060106 RU - Siberian federal district - Krasnoyarsky krai
643060107
            RU - Siberian federal district - Irkutskaya oblast
643060108 RU - Siberian federal district - Kemerovskava oblast
643060109
            RU - Siberian federal district - Novosibirskaya oblast
643060110 RU - Siberian federal district - Omskaya oblast
643060111
            RU - Siberian federal district - Tomskaya oblast
643060112 RU - Siberian federal district - Chitinskaya oblast
643060113 RU - Siberian federal district - Aginsky Buryatsky autonomous okrug
643070101 RU - Far East federal district - Kamchatsky krai
643070102 RU - Far East federal district - Primorsky krai
643070103 RU - Far East federal district - Khabarovsky krai
643070104 RU - Far East federal district - Amur oblast
643070105 RU - Far East federal district - Magadanskaya oblast
643070106 RU - Far East federal district - Sakhalinskaya oblast
643070107 RU - Far East federal district - Yakutia
643070108
            RU - Far East federal district - Chukotsky autonomous okrug
643070109
            RU - Far East federal district - Evreyskaya autonomous okrug
            RS - Centralna Srbija - Grad Beograd - Grad Beograd
688010100
            RS - Centralna Srbija - Other Centralna Srbija - Macvanski okrug
688010201
688010202
            RS - Centralna Srbija - Other Centralna Srbija - Kolubarski okrug
688010203
            RS - Centralna Srbija - Other Centralna Srbija - Podunavski okrug
688010204
            RS - Centralna Srbija - Other Centralna Srbija - Branicevski okrug
688010205
            RS - Centralna Srbija - Other Centralna Srbija - Šumadijski okrug
688010206
            RS - Centralna Srbija - Other Centralna Srbija - Pomoravski okrug
            RS - Centralna Srbija - Other Centralna Srbija - Borski okrug
688010207
            RS - Centralna Srbija - Other Centralna Srbija - Zajecarski okrug
688010208
688010209
            RS - Centralna Srbija - Other Centralna Srbija - Zlatiborski ogrug
            RS - Centralna Srbija - Other Centralna Srbija - Moravicki okrug
688010210
            RS - Centralna Srbija - Other Centralna Srbija - Raški okrug
688010211
688010212
            RS - Centralna Srbija - Other Centralna Srbija - Rasinski okrug
            RS - Centralna Srbija - Other Centralna Srbija - Nišavski okrug
688010213
688010214
            RS - Centralna Srbija - Other Centralna Srbija - Tobolicki okrug
688010215
            RS - Centralna Srbija - Other Centralna Srbija - Pirotski okrug
688010216
            RS - Centralna Srbija - Other Centralna Srbija - Jablanicki okrug
688010217
            RS - Centralna Srbija - Other Centralna Srbija - Pcinjski okrug
688020101
            RS - Vojvodina - Severno-backi okrug
688020102
            RS - Vojvodina - Srednje-banatski okrug
688020103
            RS - Vojvodina - Severno-banatski okrug
688020104
            RS - Vojvodina - Južno-banatski okrug
688020105
            RS - Vojvodina - Zapadno-backi okrug
688020106
            RS - Vojvodina - Južno-backi okrug
688020107
            RS - Vojvodina - Sremski okrug
703000100 SK – Slovenská Republika – Bratislavský kraj
```

```
703000201
            SK - Slovenská Republika - Západné Slovensko - Trnavský kraj
703000202
            SK - Slovenská Republika - Západné Slovensko - Trenciansky kraj
703000203
            SK - Slovenská Republika - Západné Slovensko - Nitriansky kraj
703000301
            SK - Slovenská Republika - Stredné Slovensko - Žilinský kraj
            SK - Slovenská Republika - Stredné Slovensko - Banskobystrický kraj
703000302
            SK - Slovenská Republika - Východné Slovensko - Prešovský kraj
703000401
703000402 SK – Slovenská Republika – Východné Slovensko – Košický kraj
705000101
            SI - Slovenija - Vzhodna Slovenija - Pomurska
705000102 SI - Slovenija - Vzhodna Slovenija - Podrvaska
705000103 SI - Slovenija - Vzhodna Slovenija - Koroska
705000104 SI - Slovenija - Vzhodna Slovenija - Savinjska
705000105 SI - Slovenija - Vzhodna Slovenija - Zasavska
705000106 SI - Slovenija - Vzhodna Slovenija - Spodnjeposavska
705000107
            SI - Slovenija - Vzhodna Slovenija - Jugovzhodna Slovenija
705000108 SI – Slovenija – Vzhodna Slovenija – Notranjsko-kraska
705000201
            SI - Slovenija - Zahodna Slovenija - Osrednjeslovenska
705000202 SI – Slovenija – Zahodna Slovenija – Gorenjska
705000203 SI - Slovenija - Zahodna Slovenija - Goriska
705000204 SI - Slovenija - Zahodna Slovenija - Obalno-kraska
724010101 ES - Noroeste - Galicia - A Coruña
724010102 ES - Noroeste - Galicia - Lugo
724010103 ES - Noroeste - Galicia - Ourense
724010104 ES - Noroeste - Galicia - Pontevedra
724010200 ES - Noroeste - Principado de Asturias - Asturias
724010300 ES - Noroeste - Cantabria
724020101 ES - Noreste - País Vasco - Álava
724020102 ES - Noreste - País Vasco - Guipúzcoa
724020103 ES - Noreste - País Vasco - Vizcaya
724020200 ES - Noreste - Comunidad Foral de Navarra - Navarra
724020300 ES - Noreste - La Rioia
724020401 ES - Noreste - Aragón - Huesca
724020402 ES - Noreste - Aragón - Teruel
724020403 ES - Noreste - Aragón - Zaragoza
724030000 ES - Comunidad de Madrid - Madrid
724040101 ES - Centro (E) - Castilla y León - Ávilla
724040102 ES - Centro (E) - Castilla y León - Burgos
724040103 ES - Centro (E) - Castilla y León - León
724040104 ES - Centro (E) - Castilla y León - Palencia
724040105 ES - Centro (E) - Castilla y León - Salamanca
724040106 ES - Centro (E) - Castilla y León - Segovia
724040107 ES - Centro (E) - Castilla y León - Soria
724040108 ES - Centro (E) - Castilla y León - Vallodolid
724040109 ES - Centro (E) - Castilla y León - Zamora
724040201 ES - Centro (E) - Castilla-La Mancha - Albacete
724040202 ES - Centro (E) - Castilla-La Mancha - Ciudad Real
724040203
            ES - Centro (E) - Castilla-La Mancha - Cuenca
724040204 ES - Centro (E) - Castilla-La Mancha - Guadalajara
724040205 ES - Centro (E) - Castilla-La Mancha - Toledo
724040301 ES - Centro (E) - Extremadura - Badajoz
```

```
724040302 ES - Centro (E) - Extremadura - Cáceres
724050101 ES - Este - Cataluña - Barcelona
724050102
            ES - Este - Cataluña - Girona
724050103
            ES - Este - Cataluña - Lleida
724050104 ES - Este - Cataluña - Tarragona
724050201 ES - Este - Comunidad Valenciana - Alicante/Alacant
724050202 ES - Este - Comunidad Valenciana - Castellón/Castelló
724050203 ES - Este - Comunidad Valenciana - Valencia/València
724050301 ES - Este - Illes Balears - Eivissa y Formentera
724050302 ES - Este - Illes Balears - Mallorca
724050303 ES - Este - Illes Balears - Menorca
724060101 ES – Sur – Andalucía – Almería
724060102 ES - Sur - Andalucía - Cádiz
724060103 ES - Sur - Andalucía - Córdoba
724060104 ES - Sur - Andalucía - Granada
724060105 ES - Sur - Andalucía - Huelva
724060106 ES - Sur - Andalucía - Jaén
724060107 ES - Sur - Andalucía - Málaga
724060108 ES - Sur - Andalucía - Sevilla
724060200 ES - Sur - Región de Murcia - Murcia
724060300 ES - Sur - Ciudad Autónoma de Ceuta - Ceuta
724060400 ES - Sur - Ciudad Autónoma de Melilla - Melilla
724070003
            ES - Canarias - El Hierro
724070004 ES - Canarias - Fuerteventura
724070005 ES - Canarias - Gran Canaria
724070006 ES - Canarias - La Gomera
724070007 ES - Canarias - La Palma
724070008 ES - Canarias - Lanzarote
724070009 ES - Canarias - Tenerife
752010100 SE - Östra Sverige - Stockholm - Stockholms län
752010201
            SE - Östra Sverige - Östra Mellansverige - Uppsala län
752010202 SE – Östra Sverige – Östra Mellansverige – Södermanlands län
            SE - Östra Sverige - Östra Mellansverige - Östergötlands län
752010203
752010204
            SE - Östra Sverige - Östra Mellansverige - Örebro län
            SE – Östra Sverige – Östra Mellansverige – Västmanlands län
752010205
752020101
            SE - Södra Sverige - Småland med öarna - Jönköpings län
752020102 SE – Södra Sverige – Småland med öarna – Kronobergs län
752020103 SE – Södra Sverige – Småland med öarna – Kalmar län
752020104
            SE - Södra Sverige - Småland med öarna - Gotlands län
752020201
            SE - Södra Sverige - Sydsverige - Blekinge län
752020204
            SE - Södra Sverige - Sydsverige - Skåne län
752020301
            SE - Södra Sverige - Västsverige - Hallands län
752020302
            SE - Södra Sverige - Västsverige - Västra Götalands län
752030101
            SE - Norra Sverige - Norra Mellansverige - Värmlands län
752030102 SE - Norra Sverige - Norra Mellansverige - Dalarnas län
752030103
            SE - Norra Sverige - Norra Mellansverige - Gävleborgs län
752030201
            SE - Norra Sverige - Mellersta Norrland - Västernorrlands län
752030202 SE - Norra Sverige - Mellersta Norrland - Jämtlands län
752030301 SE – Norra Sverige – Övre Norrland – Västerbottens län
```

```
752030302
           SE - Norra Sverige - Övre Norrland - Norrbottens län
756000101
            CH - Schweiz/Suisse/Svizzera - Région lémanique - Vaud
756000102
            CH - Schweiz/Suisse/Svizzera - Région lémanique - Valais
756000103
            CH - Schweiz/Suisse/Svizzera - Région lémanique - Genève
756000201
            CH - Schweiz/Suisse/Svizzera - Espace Mittelland - Bern
756000202 CH - Schweiz/Suisse/Svizzera - Espace Mittelland - Freiburg
756000203 CH - Schweiz/Suisse/Svizzera - Espace Mittelland - Solothurn
756000204
            CH - Schweiz/Suisse/Svizzera - Espace Mittelland - Neuchâtel
756000205 CH - Schweiz/Suisse/Svizzera - Espace Mittelland - Jura
756000301
           CH - Schweiz/Suisse/Svizzera - Nordwestschweiz - Basel-Stadt
756000302 CH - Schweiz/Suisse/Svizzera - Nordwestschweiz - Basel-Landschaft
756000303
            CH - Schweiz/Suisse/Svizzera - Nordwestschweiz - Aargau
756000400 CH - Schweiz/Suisse/Svizzera - Zürich
756000501
           CH - Schweiz/Suisse/Svizzera - Ostschweiz - Glarus
756000502 CH - Schweiz/Suisse/Svizzera - Ostschweiz - Schaffhausen
756000503 CH - Schweiz/Suisse/Svizzera - Ostschweiz - Appenzell Ausserrhoden
756000504 CH – Schweiz/Suisse/Svizzera – Ostschweiz – Appenzell Innerrhoden
756000505 CH - Schweiz/Suisse/Svizzera - Ostschweiz - St. Gallen
756000506 CH – Schweiz/Suisse/Svizzera – Ostschweiz – Graubünden
756000507
            CH - Schweiz/Suisse/Svizzera - Ostschweiz - Thurgau
756000601 CH - Schweiz/Suisse/Svizzera - Zentralschweiz - Luzern
756000602 CH - Schweiz/Suisse/Svizzera - Zentralschweiz - Uri
756000603
            CH - Schweiz/Suisse/Svizzera - Zentralschweiz - Schwyz
            CH - Schweiz/Suisse/Svizzera - Zentralschweiz - Obwalden
756000604
756000605
            CH - Schweiz/Suisse/Svizzera - Zentralschweiz - Nidwalden
756000606 CH - Schweiz/Suisse/Svizzera - Zentralschweiz - Zug
756000700 CH - Schweiz/Suisse/Svizzera - Ticino
792010000 TR - Istanbul
792020101 TR -Turkiye - Bati Marmara - Tekirdag - Tekirdag
792020102 TR -Turkiye - Bati Marmara - Tekirdag - Edirne
792020103 TR -Turkiye - Bati Marmara - Tekirdag - Kirklareli
792020201 TR -Turkiye - Bati Marmara - Balikesir - Balikesir
792020202 TR -Turkiye - Bati Marmara - Balikesir - Canakkale
792030100 TR - EGE - Izmir
792030201 TR - EGE - Aydin - Aydin
792030202 TR - EGE - Aydin - Denizli
792030203 TR - EGE - Aydin - Mugla
792030301 TR - EGE - Manisa - Manisa
792030302 TR - EGE - Manisa - Afyon
792030303 TR - EGE - Manisa - Kutahya
792030304 TR - EGE - Manisa - Usak
792040101 TR - Dogu Marmara - Bursa - Bursa
792040102 TR - Dogu Marmara - Bursa - Eskisehir
792040103 TR - Dogu Marmara - Bursa - Bilecik
792040201 TR - Dogu Marmara - Kocaeli - Kocaeli
792040202 TR - Dogu Marmara - Kocaeli - Sakarya
792040203 TR - Dogu Marmara - Kocaeli - Duzce
792040204 TR - Dogu Marmara - Kocaeli - Bolu
792040205 TR - Dogu Marmara - Kocaeli - Yalova
```

```
792050100 TR - Bati Anadolu - Ankara - Ankara
792050201
           TR - Bati Anadolu - Konya - Konya
792050202 TR – Bati Anadolu – Konya – Karaman
792060101
           TR - Akdeniz - Antalya - Antalya
792060102 TR - Akdeniz - Antalya - Isparta
792060103 TR - Akdeniz - Antalya - Burdur
792060201 TR - Akdeniz - Adana - Adana
792060202 TR - Akdeniz - Adana - Icel
792060301 TR - Akdeniz - Hatay - Hatay
792060302 TR - Akdeniz - Hatay - Kahramanmaras
792060303 TR - Akdeniz - Hatay - Osmaniye
792070101 TR - Orta Anadolu - Kirikkale - Kirikkale
792070102 TR - Orta Anadolu - Kirikkale - Aksaray
792070103 TR - Orta Anadolu - Kirikkale - Nigde
792070104 TR - Orta Anadolu - Kirikkale - Nevsehir
792070105 TR - Orta Anadolu - Kirikkale - Kirsehir
792070201 TR – Orta Anadolu – Kayseri – Kayseri
792070202 TR - Orta Anadolu - Kayseri - Sivas
792070203 TR - Orta Anadolu - Kayseri - Yozgat
792080101 TR - Bati Karadeniz - Zonguldak - Zonguldak
792080102 TR - Bati Karadeniz - Zonguldak - Karabuk
792080103 TR - Bati Karadeniz - Zonguldak - Batin
792080201
           TR - Bati Karadeniz - Kastamonu - Kastamonu
792080202 TR - Bati Karadeniz - Kastamonu - Cankiri
792080203 TR - Bati Karadeniz - Kastamonu - Sinop
792080301 TR - Bati Karadeniz - Samsun - Samsun
792080302 TR - Bati Karadeniz - Samsun - Tokat
792080303 TR - Bati Karadeniz - Samsun - Corum
792080304 TR - Bati Karadeniz - Samsun - Amasva
792090001 TR – Dogu Karadeniz – Trabzon – Trabzon
792090002 TR - Dogu Karadeniz - Trabzon - Ordu
792090003 TR – Dogu Karadeniz – Trabzon – Giresun
792090004 TR - Dogu Karadeniz - Trabzon - Rize
792090005 TR - Dogu Karadeniz - Trabzon - Artvin
792090006 TR - Dogu Karadeniz - Trabzon - Gumushane
792100101 TR - Kuzeydogu Anadolu - Erzurum - Erzurum
792100102 TR - Kuzeydogu Anadolu - Erzurum - Erzincan
792100103 TR - Kuzeydogu Anadolu - Erzurum - Bayburt
792100201 TR - Kuzeydogu Anadolu - Agri - Agri
792100202 TR - Kuzeydogu Anadolu - Agri - Kars
792100203 TR - Kuzeydogu Anadolu - Agri - Igdir
792100204 TR - Kuzeydogu Anadolu - Agri - Ardahan
792110101 TR - Ortadogu Anadolu - Malatya - Malatya
792110102 TR - Ortadogu Anadolu - Malatya - Elazig
792110103 TR - Ortadogu Anadolu - Malatya - Bingöl
792110104 TR - Ortadogu Anadolu - Malatya - Tunceli
792110201 TR - Ortadogu Anadolu - Van - Van
792110202 TR - Ortadogu Anadolu - Van - Mus
792110203 TR - Ortadogu Anadolu - Van - Bitlis
```

792110204 TR - Ortadogu Anadolu - Van - Hakkari 792120101 TR - Guneydogu Anadolu - Gaziantep - Gaziantep 792120102 TR - Guneydogu Anadolu - Gaziantep - Adiyaman 792120103 TR - Guneydogu Anadolu - Gaziantep - Kilis 792120201 TR - Guneydogu Anadolu - Sanliurfa - Sanliurfa 792120202 TR - Guneydogu Anadolu - Sanliurfa - Diyarbakir 792120301 TR – Guneydogu Anadolu – Mardin – Mardin 792120302 TR - Guneydogu Anadolu - Mardin - Batman 792120303 TR - Guneydogu Anadolu - Mardin - Sirnak 792120304 TR - Guneydogu Anadolu - Mardin - Siirt 804010100 UA - West - Volyn Oblast 804010200 UA - West - Zakarpattia Oblast 804010300 UA - West - Ivano-Frankovsk Oblast 804010400 UA - West - Lviv Oblast 804010500 UA - West - Rivne Oblast 804010600 UA - West - Ternopil Oblast 804010700 UA - West - Chernivtsi Oblast 804020100 UA - Centre - Vinnytsia Oblast 804020200 UA - Centre - Kirovohrad Oblast 804020300 UA - Centre - Poltava Oblast 804020400 UA - Centre - Khmelnytskyi Oblast 804020500 UA - Centre - Cherkasy Oblast 804030100 UA - North - Zhytomyr Oblast 804030200 UA - North - Kyiv 804030300 UA - North - Kyiv Oblast 804030400 UA - North - Sumy Oblast 804030500 UA - North - Chernihiv Oblast 804040100 UA - East - Dnipropetrovsk 804040200 UA - East - Dnipropetrovsk Oblast 804040300 UA - East - Zaporizhia 804040400 UA - East - Zaporizhia Oblast 804040500 UA - East - Donetsk 804040600 UA - East - Donetsk Oblast 804040700 UA - East - Luhansk Oblast 804040800 UA - East - Kharkiv 804040900 UA - East - Kharkiv Oblast 804050100 UA - South - Sevastopol 804050200 UA - South - Autonomous Republic of Crimea 804050300 UA - South - Mykolaiv Oblast 804050400 UA - South - Odesa 804050500 UA - South - Odesa Oblast 804050600 UA - South - Kherson Oblast 807000001 MK - Poranesnata jugoslovenska Republika Makedonija - Vardarski 807000002 MK - Poranesnata jugoslovenska Republika Makedonija - Istocen 807000003 MK - Poranesnata jugoslovenska Republika Makedonija - Jugozapaden 807000004 MK - Poranesnata jugoslovenska Republika Makedonija - Jugoistocen MK - Poranesnata jugoslovenska Republika Makedonija - Pelagoniski 807000005 807000006 MK - Poranesnata jugoslovenska Republika Makedonija - Polski 807000007 MK - Poranesnata jugoslovenska Republika Makedonija - Severoistocen

```
807000008
            MK - Poranesnata jugoslovenska Republika Makedonija - Skopski
826120101
            GB - North East (England) - Tees Valley and Durham - Hartlepool and Stockton-on-Tees
826120102
            GB - North East (England) - Tees Valley and Durham - South Teesside
826120103
            GB - North East (England) - Tees Valley and Durham - Darlington
826120104
            GB - North East (England) - Tees Valley and Durham - Durham CC
826120201
            GB - North East (England) - Northumberland and Tyne and Wear - Northumberland
826120202
            GB - North East (England) - Northumberland and Tyne and Wear - Tyneside
826120203
            GB - North East (England) - Northumberland and Tyne and Wear - Sunderland
826130101
            GB - North West (England) - Cumbria - West Cumbria
826130102
            GB - North West (England) - Cumbria - East Cubria
826130201
            GB - North West (England) - Cheshire - Halton and Warrington
826130202
            GB - North West (England) - Cheshire - Cheshire CC
826130301
            GB - North West (England) - Greater Manchester - Greater Manchester South
826130302
            GB - North West (England) - Greater Manchester - Greater Manchester North
826130401
            GB - North West (England) - Lancashire - Blackburn with Darwen
826130402 GB - North West (England) - Lancashire - Blackpool
826130403 GB - North West (England) - Lancashire - Lancashire CC
826130501
            GB - North West (England) - Merseyside - East Merseyside
826130502
            GB - North West (England) - Merseyside - Liverpool
826130503
            GB - North West (England) - Merseyside - Sefton
826130504
            GB - North West (England) - Merseyside - Wirral
826140101
            GB - Yorkshire and the Humber - East Yorkshire and Northern Lincolnshire - Kingston upon Hull, City of
826140102
            GB - Yorkshire and the Humber - East Yorkshire and Northern Lincolnshire - East Riding of Yorkshire
826140103
            GB - Yorkshire and the Humber - East Yorkshire and Northern Lincolnshire - North and North East Lincolnshire
826140201
            GB - Yorkshire and the Humber - North Yorkshire - York
826140202
            GB - Yorkshire and the Humber - North Yorkshire - North Yorkshire CC
826140301
            GB - Yorkshire and the Humber - South Yorkshire - Barnsley, Doncaster and Rotherham
826140302
            GB - Yorkshire and the Humber - South Yorkshire - Sheffield
826140401
            GB - Yorkshire and the Humber - West Yorkshire - Bradford
826140402
            GB - Yorkshire and the Humber - West Yorkshire - Leeds
826140403
            GB - Yorkshire and the Humber - West Yorkshire - Calderdale, Kirklees and Wakefield
826150101
            GB - East Midlands (England) - Derbyshire and Nottinghamshire - Derby
826150102 GB - East Midlands (England) - Derbyshire and Nottinghamshire - East Derbyshire
826150103
            GB - East Midlands (England) - Derbyshire and Nottinghamshire - South and West Derbyshire
826150104
            GB - East Midlands (England) - Derbyshire and Nottinghamshire - Nottingham
            GB - East Midlands (England) - Derbyshire and Nottinghamshire - North Nottinghamshire
826150105
826150106
            GB - East Midlands (England) - Derbyshire and Nottinghamshire - South Nottinghamshire
826150201
            GB - East Midlands (England) - Leicestershire, Rutland and Northamptonshire - Leicester
826150202
            GB - East Midlands (England) - Leicestershire, Rutland and Northamptonshire - Leicestershire CC and Rutland
826150203
            GB - East Midlands (England) - Leicestershire, Rutland and Northamptonshire - Northamptonshire
826150300
            GB - East Midlands (England) - Lincolnshire
826160101
            GB - West Midlands (England) - Herefordshire, Worcestershire and Warwickshire - Herefordshire, County of
826160102
            GB - West Midlands (England) - Herefordshire, Worcestershire and Warwickshire - Worcestershire
826160103
            GB - West Midlands (England) - Herefordshire, Worcestershire and Warwickshire - Warwickshire
826160201
            GB - West Midlands (England) - Shropshire and Staffordshire - Telford and Wrekin
826160202
            GB - West Midlands (England) - Shropshire and Staffordshire - Shropshire CC
826160203
            GB - West Midlands (England) - Shropshire and Staffordshire - Stoke-on-Trent
826160204
            GB - West Midlands (England) - Shropshire and Staffordshire - Staffordshire CC
826160301 GB - West Midlands (England) - West Midlands - Birmingham
```

```
826160302
            GB - West Midlands (England) - West Midlands - Solihull
826160303
            GB - West Midlands (England) - West Midlands - Coventry
826160304
            GB - West Midlands (England) - West Midlands - Dudley and Sandwell
826160305
            GB - West Midlands (England) - West Midlands - Walsall and Wolverhampton
826170101
            GB - East of England - East Anglia - Peterborough
826170102 GB - East of England - East Anglia - Cambridgeshire CC
826170103 GB - East of England - East Anglia - Norfolk
826170104
            GB - East of England - East Anglia - Suffolk
826170201 GB - East of England - Bedfordshire and Hertfordshire - Luton
826170202 GB - East of England - Bedfordshire and Hertfordshire - Bedfordshire CC
826170203
            GB - East of England - Bedfordshire and Hertfordshire - Hertfordshire
826170301
            GB - East of England - Essex - Southend-on-Sea
826170302 GB - East of England - Essex - Thurrock
826170303 GB - East of England - Essex - Essex CC
826180101
            GB - London - Inner London - West
826180102 GB - London - Inner London - East
826180201
            GB - London - Outer London - East and North East
826180202 GB - London - Outer London - South
826180203 GB - London - Outer London - West and North West
826190101
            GB - South East (England) - Berkshire, Buckinghamshire and Oxfordshire - Berkshire
826190102
            GB - South East (England) - Berkshire, Buckinghamshire and Oxfordshire - Milton Keynes
826190103
            GB - South East (England) - Berkshire, Buckinghamshire and Oxfordshire - Buckinghamshire CC
826190104
            GB - South East (England) - Berkshire, Buckinghamshire and Oxfordshire - Oxfordshire
826190201
            GB - South East (England) - Surrey, East and West Sussex - Brighton and Hove
826190202
            GB - South East (England) - Surrey, East and West Sussex - East Sussex CC
826190203
            GB - South East (England) - Surrey, East and West Sussex - Surrey
826190204
            GB - South East (England) - Surrey, East and West Sussex - West Sussex
826190301
            GB - South East (England) - Hampshire and Isle of Wight - Portsmouth
826190302
            GB - South East (England) - Hampshire and Isle of Wight - Southampton
826190303
            GB - South East (England) - Hampshire and Isle of Wight - Hampshire CC
826190304
            GB - South East (England) - Hampshire and Isle of Wight - Isle of Wight
826190401
            GB - South East (England) - Kent - Medway
826190402
            GB - South East (England) - Kent - Kent CC
            GB - South West (England) - Gloucestershire, Wiltshire and Bristol/Bath area - Bristol, City of
826200101
826200102
            GB - South West (England) - Gloucestershire, Wiltshire and Bristol/Bath area - Bath and North East Somerset, Nort
826200103
            GB - South West (England) - Gloucestershire, Wiltshire and Bristol/Bath area - Gloucestershire
826200104
            GB - South West (England) - Gloucestershire, Wiltshire and Bristol/Bath area - Swindon
826200105
            GB - South West (England) - Gloucestershire, Wiltshire and Bristol/Bath area - Wiltshire CC
826200201
            GB - South West (England) - Dorset and Somerset - Bournemouth and Poole
826200202
            GB - South West (England) - Dorset and Somerset - Dorset CC
826200203
            GB - South West (England) - Dorset and Somerset - Somerset
826200300
            GB - South West (England) - Cornwall and Isles of Scilly
826200401
            GB - South West (England) - Devon - Plymouth
            GB - South West (England) - Devon - Torbay
826200402
826200403
            GB - South West (England) - Devon - Devon CC
826210101
            GB - Wales - West Wales and The Valleys - Isle of Anglesey
826210102 GB - Wales - West Wales and The Valleys - Gwynedd
826210103 GB - Wales - West Wales and The Valleys - Conwy and Denbighshire
826210104 GB - Wales - West Wales and The Valleys - South West Wales
```

```
826210105 GB - Wales - West Wales and The Valleys - Central Valleys
826210106 GB - Wales - West Wales and The Valleys - Gwent Valleys
826210107
            GB - Wales - West Wales and The Valleys - Bridgend and Neath Port Talbot
            GB - Wales - West Wales and The Valleys - Swansea
826210108
826210201 GB - Wales - East Wales - Monmouthshire and Newport
826210202 GB - Wales - East Wales - Cardiff and Vale of Glamorgan
826210203 GB - Wales - East Wales - Flintshire and Wrexham
826210204 GB - Wales - East Wales - Powys
826220201 GB - Scotland - Eastern Scotland - Angus and Dundee City
826220202 GB - Scotland - Eastern Scotland - Clackmannanshire and Fife
826220203 GB - Scotland - Eastern Scotland - East Lothian and Midlothian
826220204
            GB - Scotland - Eastern Scotland - Scottish Borders
826220205 GB - Scotland - Eastern Scotland - Edinburgh, City of
826220206 GB - Scotland - Eastern Scotland - Falkirk
826220207
            GB - Scotland - Eastern Scotland - Perth & Kinross and Stirling
826220208 GB - Scotland - Eastern Scotland - West Lothian
826220301 GB - Scotland - South Western Scotland - East Dunbartonshire, West Dunbartonshire and Helensburgh & Lomond
826220302 GB - Scotland - South Western Scotland - Dumfries & Galloway
826220303 GB - Scotland - South Western Scotland - East Ayrshire and North Ayrshire mainland
826220304
            GB - Scotland - South Western Scotland - Glasgow City
826220305
            GB - Scotland - South Western Scotland - Inverclyde, East Renfrewshire and Renfrewshire
826220306
            GB - Scotland - South Western Scotland - North Lanarkshire
826220307
            GB - Scotland - South Western Scotland - South Ayrshire
            GB - Scotland - South Western Scotland - South Lanarkshire
826220308
826220500
            GB - Scotland - North Eastern Scotland - Aberdeen City and Aberdeenshire
826220601
            GB - Scotland - Highlands and Islands - Caithness & Sutherland and Ross & Cromarty
826220602
            GB - Scotland - Highlands and Islands - Inverness & Nairn and Moray, Badenoch & Strathspey
826220603
            GB - Scotland - Highlands and Islands - Lochaber, Skye & Lochalsh, Arran & Cumbrae and Argyll & Bute
826220604
            GB - Scotland - Highlands and Islands - Eilean Siar (Western Isles)
826220605
            GB - Scotland - Highlands and Islands - Orkney Islands
826220606
            GB - Scotland - Highlands and Islands - Shetland Islands
826230001
            NIR - Northern Ireland - Belfast
826230002 NIR - Northern Ireland - Outer Belfast
826230003 NIR - Northern Ireland - East of Northern Ireland
826230004 NIR - Northern Ireland - North of Northern Ireland
826230005 NIR - Northern Ireland - West and South of Northern Ireland
1001000001 KOS - Kosovo - Komuna e Gjakovës/Đakovicki okrug
1001000002 KOS - Kosovo - Komuna e Gjilanit/Gnjilanski okrug
1001000003 KOS - Kosovo - Komuna e Mitrovicës/Kosovskomitrovacki okrug
1001000004 KOS - Kosovo - Komuna e Pejës/Pecki okrug
1001000005 KOS - Kosovo - Komuna e Prishtinës/Prištinski okrug
1001000006 KOS - Kosovo - Komuna e Prizrenit/Prizrenski okrug
1001000007 KOS - Kosovo - Komuna e Ferizajt/Uroševacki okrug
1961000000 CY - Kypros / Kibris
```

Note:

Verbatim recorded variable post-coded according to NUTS-3 code. Data for original string variable is not available in this dataset. For detailed information about country-specific classification and correspondence with NUTS, see Method Report.

```
v371b_N2 - Region: NUTS-2 code (Q136)
```

HARMONIZED VARIABLE - NUTS 2

Q136 Region:

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 800001 AL Albania South-Albania
- 800002 AL Albania North-Albania
- 3100000 AZ Azerbaijan
- 4000101 AT Ostösterreich Burgenland (A)
- 4000102 AT Ostösterreich Niederösterreich
- 4000103 AT Ostösterreich Wien
- 4000201 AT Südösterreich Kärnten
- 4000202 AT Südösterreich Steiermark
- 4000301 AT Westöstereich Oberösterreich
- 4000302 AT Westöstereich Salzburg
- 4000303 AT Westöstereich Tirol
- 4000304 AT Westöstereich Vorarlberg
- 5100001 AM Armenia Yerevan
- 5100002 AM Armenia Rest of Armenia
- 5600100 BE Région de Bruxelles-capitale/Brussels hoofdstedelijk gewest
- 5600201 BE Vlaams gewest Prov. Antwerpen
- 5600202 BE Vlaams gewest Prov. Limburg (B)
- 5600203 BE Vlaams gewest Prov. Oost-Vlaanderen
- 5600204 BE Vlaams gewest Prov. Vlaams-Brabant
- 5600205 BE Vlaams gewest Prov. West-Vlaanderen
- 5600301 BE Région Wallonne Prov. Brabant Wallon
- 5600302 BE Région Wallonne Prov. Hainaut
- 5600303 BE Région Wallonne Prov. Liège
- 5600304 BE Région Wallonne Prov. Luxembourg (B)
- 5600305 BE Région Wallonne Prov. Namur
- 7000001 BA Bosna i Hercegovina Federacija Bosne i Hercegovine
- 7000002 BA Bosna i Hercegovina Republika Srpska
- 7000003 BA Bosna i Hercegovina Brcko
- 10000301 BG Severna i iztochna Bulgaria Severozapaden
- 10000302 BG Severna i iztochna Bulgaria Severen tsentralen
- 10000303 BG Severna i iztochna Bulgaria Severoiztochen
- 10000304 BG Severna i iztochna Bulgaria Yugoiztochen
- 10000401 BG Yugozapadna i yuzhna tsentralna Bulgaria Yugozapaden
- 10000402 BG Yugozapadna i yuzhna tsentralna Bulgaria Yuzhen tsentralen
- 11200000 BY Belarus
- 19100001 HR Hrvatska Sjeverozapadna Hrvatska
- 19100002 HR Hrvatska Sredisnja i Istocna (Panonska) Hrvatska

```
19100003
          HR - Hrvatska - Jadranska Hrvatska
19610000
          CY - Kypros / Kibris
20300001
          CZ - Ceska Republika - Praha
20300002
          CZ - Ceska Republika - Stredni Cechy
20300003 CZ - Ceska Republika - Jihozapad
20300004 CZ - Ceska Republika - Severozapad
20300005 CZ - Ceska Republika - Severovychod
20300006 CZ - Ceska Republika - Jihovychod
20300007 CZ - Ceska Republika - Stredni Morava
20300008 CZ - Ceska Republika - Moravskoslezsko
20800001 DK - Danmark - Hovedstaden
20800002 DK - Danmark- Sjælland
20800003 DK – Danmark – Syddanmark
20800004 DK - Danmark - Midtjylland
20800005 DK - Danmark - Nordjylland
23300000 EE - Eesti
24600103 FI - Manner-Suomi - Itä-Suomi
24600108 FI - Manner-Suomi - Etalä-Suomi
24600109 FI - Manner-Suomi - Länsi-Suomi
24600110 FI - Manner-Suomi - Pohjois-Suomi
24600200 FI - Åland
25000100 FR - Île de France
25000201 FR - Bassin Parisien - Champagne-Ardenne
25000202 FR - Bassin Parisien - Champagne-Ardenne - Picardie
25000203 FR - Bassin Parisien - Champagne-Ardenne - Haute-Normandie
25000204 FR - Bassin Parisien - Champagne-Ardenne - Centre
25000205 FR - Bassin Parisien - Champagne-Ardenne - Basse-Normandie
25000206 FR - Bassin Parisien - Champagne-Ardenne - Bourgogne
25000300 FR - Nord-pas-de-Calais
25000401 FR - Est - Lorraine
25000402 FR - Est - Alsace
25000403 FR - Est - Franche-Comté
25000501 FR - Ouest - Pays de la Loire
25000502 FR - Ouest - Bretagne
25000503 FR - Ouest - Poitou-Charentes
25000601 FR - Sud-Ouest - Aquitaine
25000602 FR - Sud-Ouest - Midi-Pyrénées
25000603 FR - Sud-Ouest - Limousin
25000701 FR - Centre-Est - Rhône-Alpes
25000702 FR - Centre-Est - Auvergne
25000801 FR - Méditerranée - Languedoc-Roussillon
25000802 FR - Méditerranée - Provence-Alpes-Côte d'Azur
25000803 FR - Méditerranée - Corse
25000901
          FR - Départements d'outre-mer - Guadeloupe
25000902 FR - Départements d'outre-mer - Martinique
          FR - Départements d'outre-mer - Guyane
25000903
         FR - Départements d'outre-mer - Réunion
25000904
26800000
          GE - Georgia
27600101 DE - Baden-Württemberg - Stuttgart
```

```
27600102 DE - Baden-Württemberg - Karlsruhe
27600103 DE - Baden-Württemberg - Freiburg
27600104
          DE - Baden-Württemberg - Tübingen
27600201
          DE - Bayern - Oberbayern
27600202 DE - Bayern - Niederbayern
27600203 DE - Bayern - Oberpfalz
27600204 DE - Bayern - Oberfranken
27600205 DE - Bayern - Mittelfranken
27600206 DE - Bayern - Unterfranken
27600207 DE - Bayern - Schwaben
27600300 DE - Berlin
27600401 DE - Brandenburg - Brandenburg - Nordost
27600402 DE - Brandenburg - Brandenburg Südwest
27600500
         DE – Bremen
27600600 DE - Hamburg
27600701 DE - Hessen - Darmstadt
27600702 DE - Hessen - Gießen
27600703 DE - Hessen - Kassel
27600800 DE – Mecklenburg-Vorpommern
27600901 DE - Niedersachsen - Braunschweig
27600902 DE - Niedersachsen - Hannover
27600903 DE - Niedersachsen - Lüneburg
27600904 DE - Niedersachsen - Weser-Ems
27601001
          DE - Nordrhein-Westfalen - Düsseldorf
27601002 DE - Nordrhein-Westfalen - Köln
27601003 DE - Nordrhein-Westfalen - Münster
27601004 DE - Nordrhein-Westfalen - Detmold
27601005 DE - Nordrhein-Westfalen - Arnsberg
27601101 DE - Rheinland-Pfalz - Koblenz
27601102 DE - Rheinland-Pfalz - Trier
27601103 DE - Rheinland-Pfalz - Rheinhessen-Pfalz
27601200 DE - Saarland
27601301 DE - Sachsen - Chemnitz
27601302 DE - Sachsen - Dresden
27601303 DE - Sachsen - Leipzig
27601400 DE - Sachsen-Anhalt
27601500 DE - Schleswig-Holstein
27601600 DE - Thüringen
30000101 GR - Voreia Ellada - Anatoliki Makedonia, Thraki
30000102 GR - Voreia Ellada - Kentriki Makedonia
30000103 GR - Voreia Ellada - Dytiki Makedonia
30000104 GR - Voreia Ellada - Thessalia
30000201
          GR - Kentriki Ellada - Ipeiros
30000202
          GR - Kentriki Ellada - Ionia Nisia
30000203 GR - Kentriki Ellada - Dytiki Ellada
30000204 GR - Kentriki Ellada - Sterea Ellada
30000205 GR - Kentriki Ellada - Peloponnisos
30000300
          GR - Attiki
30000401 GR - Nisia Aigaiou, Kriti - Voreio Aigaio
```

```
30000402
           GR - Nisia Aigaiou, Kriti - Notio Aigaio
30000403
           GR - Nisia Aigaiou, Kriti - Kriti
34800100
         HU – Közép-Magyarország
34800201
          HU - Dunántúl - Közép-Dunántúl
34800202 HU - Dunántúl - Nyugat-Dunántúl
34800203 HU - Dunántúl - Dél-Dunántúl
34800301 HU – Alföld és Észak – Észak-Magyarország
34800302 HU - Alföld és Észak - Észak-Alföld
34800303 HU – Alföld és Észak – Dél-Alföld
35200000 IS - Ísland
37200001
          IE - Ireland - Border, Midland and Western
37200002
          IE - Ireland - Southern and Eastern
38001201
          IT - Nord-Ovest - Piemonte
38001202 IT - Nord-Ovest - Valle d'Aosta/Vallée d'Aoste
38001203 IT - Nord-Ovest - Liguria
38001204 IT - Nord-Ovest - Lombardia
38001301 IT - Nord-Est - Provincia Autonoma Bolzano/Bozen
38001302 IT - Nord-Est - Provincia Autonoma Trento
38001303 IT - Nord-Est - Veneto
38001304 IT - Nord-Est - Friuli-Venezia Giulia
38001305 IT - Nord-Est - Emilia-Romagna
38001401 IT - Centro (I) - Toscana
38001402 IT - Centro (I) - Umbria
38001403 IT - Centro (I) - Marche
38001404 IT - Centro (I) - Lazio
38001501 IT - Sud - Abruzzo
38001502 IT - Sud - Molise
38001503 IT - Sud - Campania
38001504 IT - Sud - Puglia
38001505 IT - Sud - Basilicata
38001506 IT - Sud - Calabria
38001601 IT - Isole - Sicilia
38001602 IT - Isole - Sardegna
42800000 LV - Latvija
44000000 LT - Lietuva
44200000 LU - Luxembourg (Grand-Duché)
47000000 MT- Malta
49800000 MD - Moldova
49900001 ME - Montenegro - North
49900002 ME - Montenegro - Center
49900003 ME - Montenegroc - South
52800101 NL - Noord-Nederland - Groningen
52800102 NL - Noord-Nederland - Friesland (NL)
52800103 NL - Noord-Nederland - Drenthe
52800201 NL - Oost-Nederland - Overijssel
52800202 NL - Oost-Nederland - Gelderland
52800203 NL - Oost-Nederland - Flevoland
          NL - West-Nederland - Utrecht
52800301
52800302 NL - West-Nederland - Noord-Holland
```

```
52800303
          NL - West-Nederland - Zuid-Holland
52800304
          NL - West-Nederland - Zeeland
52800401
          NL - Zuid-Nederland - Noord-Brabant
52800402
          NL - Zuid-Nederland - Limburg (NL)
57800001 NO - Norge - Oslo og Akershus
57800002 NO - Norge - Hedmark og Oppland
57800003 NO - Norge - Sør-Østlandet
57800004 NO - Norge - Agder og Rogaland
57800005 NO - Norge - Vestlandet
57800006 NO - Norge - Trøndelag
57800007 NO - Norge - Nord-Norge
61600101 PL - Region centralny - Lódzkie
61600102 PL - Region centralny - Mazowieckie
61600201 PL - Region Poludinowy - Malopolskie
61600202 PL - Region Poludinowy - Slaskie
61600301 PL - Region Wschodni - Lubelskie
61600302 PL - Region Wschodni - Podkarpackie
61600303 PL - Region Wschodni - Swietokrzyskie
61600304 PL - Region Wschodni - Podlaskie
61600401 PL – Region Pólnocno-Zachodni – Wielkopolskie
61600402 PL - Region Pólnocno-Zachodni - Zachodniopomorskie
61600403 PL - Region Pólnocno-Zachodni - Lubuskie
61600501 PL – Region Póludniowo-Zachodni – Dolnoslaskie
61600502 PL - Region Póludniowo-Zachodni - Opolskie
61600601 PL - Region Pólnocny - Kujawsko-Pomorskie
61600602 PL - Region Pólnocny - Warminsko-Mazurskie
61600603 PL - Region Pólnocny - Pomorskie
62000101 PT - Continente - Northe
62000105 PT - Continente - Algarye
62000106 PT - Continente - Centro (P)
62000107 PT - Continente - Lisboa
62000108 PT - Continente - Alentejo
62000200 PT – Região Autónoma dos Açores
62000300 PT – Região Autónoma da Madeira
64200101 RO - Macroregiunea unu - Nord-Vest
64200102 RO - Macroregiunea unu - Centru
64200201 RO - Macroregiunea doi - Nord-Est
64200202 RO - Macroregiunea doi - Sud-Est
64200301 RO - Macroregiunea trei - Sud-Muntenia
64200302 RO - Macroregiunea trei - Bucuresti-Ilfov
64200401 RO - Macroregiunea patru - Sud-Vest Oltenia
64200402 RO - Macroregiunea patru - Vest
64300101
          RU - Central Federal District
64300201
          RU - North West federal district
64300301 RU - South Federal district
64300401 RU - Privolzhsky federal district
64300501 RU - Urals federal district
          RU - Siberian federal district
64300601
64300701 RU - Far East federal district
```

68800101 RS - Centralna Srbija - Grad Beograd 68800102 RS - Centralna Srbija - Other Centralna Srbija 68800201 RS - Vojvodina 70300001 SK - Slovenská Republika - Bratislavský kraj 70300002 SK - Slovenská Republika - Západné Slovensko 70300003 SK - Slovenská Republika - Stredné Slovensko 70300004 SK - Slovenská Republika - Východné Slovensko 70500001 SI - Slovenija - Vzhodna Slovenija 70500002 SI - Slovenija - Zahodna Slovenija 72400101 ES - Noroeste - Galicia 72400102 ES - Noroeste - Principado de Asturias 72400103 ES - Noroeste - Cantabria 72400201 ES - Noreste - País Vasco 72400202 ES - Noreste - Comunidad Foral de Navarra 72400203 ES - Noreste - La Rioja 72400204 ES - Noreste - Aragón 72400300 ES - Comunidad de Madrid 72400401 ES - Centro (E) - Castilla y León 72400402 ES - Centro (E) - Castilla-La Mancha 72400403 ES - Centro (E) - Extremadura 72400501 ES - Este - Cataluña 72400502 ES - Este - Comunidad Valenciana 72400503 ES - Este - Illes Balears 72400601 ES - Sur - Andalucía 72400602 ES - Sur - Región de Murcia 72400603 ES - Sur - Ciudad Autónoma de Ceuta 72400604 ES - Sur - Ciudad Autónoma de Melilla 72400700 ES - Canarias 75200101 SE - Östra Sverige - Stockholm 75200102 SE – Östra Sverige – Östra Mellansverige 75200201 SE - Södra Sverige - Småland med öarna 75200202 SE - Södra Sverige - Sydsverige 75200203 SE - Södra Sverige - Västsverige 75200301 SE - Norra Sverige - Norra Mellansverige 75200302 SE - Norra Sverige - Mellersta Norrland 75200303 SE - Norra Sverige - Övre Norrland 75600001 CH - Schweiz/Suisse/Svizzera - Région lémanique 75600002 CH - Schweiz/Suisse/Svizzera - Espace Mittelland 75600003 CH - Schweiz/Suisse/Svizzera - Nordwestschweiz 75600004 CH - Schweiz/Suisse/Svizzera - Zürich 75600005 CH - Schweiz/Suisse/Svizzera - Ostschweiz 75600006 CH - Schweiz/Suisse/Svizzera - Zentralschweiz 75600007 CH - Schweiz/Suisse/Svizzera - Ticino 79200100 TR - Istanbul 79200201 TR - Turkiye - Bati Marmara - Tekirdag 79200202 TR - Turkiye - Bati Marmara - Balikesir 79200301 TR - EGE - Izmir 79200302 TR - EGE - Aydin 79200303 TR - EGE - Manisa

79200401 TR - Dogu Marmara - Bursa 79200402 TR - Dogu Marmara - Kocaeli 79200501 TR - Bati Anadolu - Ankara 79200502 TR - Bati Anadolu - Konya 79200601 TR - Akdeniz - Antalya 79200602 TR - Akdeniz - Adana 79200603 TR - Akdeniz - Hatay 79200701 TR - Orta Anadolu - Kirikkale 79200702 TR - Orta Anadolu - Kayseri 79200801 TR - Bati Karadeniz - Zonguldak 79200802 TR - Bati Karadeniz - Kastamonu 79200803 TR - Bati Karadeniz - Samsun 79200900 TR - Dogu Karadeniz - Trabzon 79201001 TR - Kuzeydogu Anadolu - Erzurum 79201002 TR - Kuzeydogu Anadolu - Agri 79201101 TR - Ortadogu Anadolu - Malatya 79201102 TR – Ortadogu Anadolu – Van 79201201 TR - Guneydogu Anadolu - Gaziantep 79201202 TR - Guneydogu Anadolu - Sanliurfa 79201203 TR – Guneydogu Anadolu – Mardin 80400101 UA - West - Volyn Oblast 80400102 UA - West - Zakarpattia Oblast 80400103 UA - West - Ivano-Frankovsk Oblast 80400104 UA - West - Lviv Oblast 80400105 UA - West - Rivne Oblast 80400106 UA - West - Ternopil Oblast 80400107 UA - West - Chernivtsi Oblast 80400201 UA - Centre - Vinnytsia Oblast 80400202 UA - Centre - Kirovohrad Oblast 80400203 UA - Centre - Poltava Oblast 80400204 UA - Centre - Khmelnytskyi Oblast 80400205 UA - Centre - Cherkasy Oblast 80400301 UA - North - Zhytomyr Oblast 80400302 UA - North - Kyiv 80400303 UA - North - Kyiv Oblast 80400304 UA - North - Sumy Oblast 80400305 UA - North - Chernihiv Oblast 80400401 UA - East - Dnipropetrovsk 80400402 UA - East - Dnipropetrovsk Oblast 80400403 UA - East - Zaporizhia 80400404 UA - East - Zaporizhia Oblast 80400406 UA - East - Donetsk Oblast 80400407 UA - East - Luhansk Oblast 80400409 UA - East - Kharkiv Oblast 80400501 UA - South - Sevastopol 80400502 UA - South - Autonomous Republic of Crimea

80400503 UA - South - Mykolaiv Oblast

```
80400504
           UA - South - Odesa
80400505
           UA - South - Odesa Oblast
80400506
           UA - South - Kherson Oblast
80700000
           MK - Poranesnata jugoslovenska Republika Makedonija
82611201 GB - North East (England) - Tees Valley and Durham
82611202 GB - North East (England) - Northumberland and Tyne and Wear
82611301 GB - North West (England) - Cumbria
82611302 GB - North West (England) - Cheshire
82611303 GB - North West (England) - Greater Manchester
82611304 GB - North West (England) - Lancashire
82611305 GB - North West (England) - Merseyside
82611401 GB - Yorkshire and the Humber - East Yorkshire and Northern Lincolnshire
82611402 GB - Yorkshire and the Humber - North Yorkshire
82611403
          GB - Yorkshire and the Humber - South Yorkshire
82611404 GB – Yorkshire and the Humber – West Yorkshire
82611501 GB - East Midlands (England) - Derbyshire and Nottinghamshire
82611502 GB - East Midlands (England) - Leicestershire, Rutland and Northamptonshire
82611503 GB - East Midlands (England) - Lincolnshire
82611601 GB - West Midlands (England) - Herefordshire, Worcestershire and Warwickshire
82611602 GB - West Midlands (England) - Shropshire and Staffordshire
82611603 GB - West Midlands (England) - West Midlands
82611701 GB - East of England - East Anglia
82611702 GB - East of England - Bedfordshire and Hertfordshire
82611703 GB - East of England - Essex
82611801
           GB - London - Inner London
82611802 GB - London - Outer London
82611901 GB - South East (England) - Berkshire, Buckinghamshire and Oxfordshire
82611902 GB - South East (England) - Surrey, East and West Sussex
82611903 GB - South East (England) - Hampshire and Isle of Wight
82611904 GB - South East (England) - Kent
82612001 GB - South West (England) - Gloucestershire, Wiltshire and Bristol/Bath area
82612002 GB - South West (England) - Dorset and Somerset
82612003 GB - South West (England) - Cornwall and Isles of Scilly
82612004 GB - South West (England) - Devon
82612101 GB - Wales - West Wales and The Valleys
82612102 GB - Wales - East Wales
82612202 GB - Scotland - Eastern Scotland
82612203 GB - Scotland - South Western Scotland
82612205 GB - Scotland - North Eastern Scotland
82622206 GB - Scotland - Highlands and Islands
82622300 NIR - Northern Ireland
91500000 KOS - Kosovo
```

Derivation:

Variable was computed from variable v371b_N3.

276006 DE – Hamburg 276007 DE – Hessen

GESIS StudyNo 4754, v1.1.0, http://dx.doi.org/doi:10.4232/1.10152

v371b_N1 - Region: NUTS-1 code (Q136) HARMONIZED VARIABLE - NUTS 1 Q136 Region: -5 other missing -4 question not asked -3 nap -2 na -1 dk 8000 AL - Albania 31000 AZ – Azerbaijan 40001 AT - Ostösterreich 40002 AT - Südösterreich 40003 AT - Westöstereich 51000 AM - Armenia 56001 BE – Région de Bruxelles-capitale/Brussels hoofdstedelijk gewest 56002 BE - Vlaams gewest 56003 BE - Région Wallonne 70000 BA - Bosna i Hercegovina 100003 BG - Severna i iztochna Bulgaria 100004 BG - Yugozapadna i yuzhna tsentralna Bulgaria 112000 BY - Belarus 191000 HR - Hrvatska 196100 CY - Kypros / Kibris 203000 CZ - Ceska Republika 208000 DK - Danmark 233000 EE - Eesti 246001 FI - Manner-Suomi 246002 FI - Åland 250001 FR - Île de France 250002 FR - Bassin Parisien 250003 FR - Nord-pas-de-Calais 250004 FR - Est 250005 FR - Ouest 250006 FR - Sud-Ouest 250007 FR - Centre-Est 250008 FR - Méditerranée 250009 FR - Départements d'outre-mer 268000 GE - Georgia 276001 DE - Baden-Württemberg 276002 DE - Bayern 276003 DE - Berlin 276004 DE - Brandenburg 276005 DE - Bremen

276008 DE – Mecklenburg-Vorpommern 276009 DE - Niedersachsen 276010 DE - Nordrhein-Westfalen 276011 DE - Rheinland-Pfalz 276012 DE - Saarland 276013 DE - Sachsen 276014 DE - Sachsen-Anhalt 276015 DE - Schleswig-Holstein 276016 DE - Thüringen 300001 GR - Voreia Ellada 300002 GR - Kentriki Ellada 300003 GR - Attiki 300004 GR - Nisia Aigaiou, Kriti 348001 HU - Közép-Magyarország 348002 HU - Dunántúl 348003 HU – Alföld és Észak 352000 IS - Ísland 372000 IE - Ireland 380012 IT - Nord-Ovest 380013 IT - Nord-Est 380014 IT - Centro (I) 380015 IT - Sud 380016 IT - Isole 428000 LV - Latvija 440000 LT - Lietuva 442000 LU - Luxembourg (Grand-Duché) 470000 MT - Malta 498000 MD - Moldova 499000 ME - Montenegro 528001 NL - Noord-Nederland 528002 NL - Oost-Nederland 528003 NL - West-Nederland 528004 NL - Zuid-Nederland 578000 NO - Norge 616001 PL - Region centralny 616002 PL - Region Poludinowy 616003 PL - Region Wschodni 616004 PL - Region Pólnocno-Zachodni 616005 PL - Region Póludniowo-Zachodni 616006 PL - Region Pólnocny 620001 PT - Continente 620002 PT - Região Autónoma dos Açores 620003 PT - Região Autónoma da Madeira 642001 RO - Macroregiunea unu 642002 RO - Macroregiunea doi 642003 RO - Macroregiunea trei 642004 RO - Macroregiunea patru 643001 RU - Central Federal District

643002 RU - North West federal district

643003 RU - South Federal district 643004 RU - Privolzhsky federal district 643005 RU - Urals federal district 643006 RU - Siberian federal district 643007 RU - Far East federal district 688001 RS - Centralna Srbija 688002 RS - Vojvodina 703000 SK - Slovenská Republika 705000 SI - Slovenija 724001 ES - Noroeste 724002 ES - Noreste 724003 ES - Comunidad de Madrid 724004 ES - Centro (E) 724005 ES - Este 724006 ES - Sur 724007 ES – Canarias 752001 SE – Östra Sverige 752002 SE - Södra Sverige 752003 SE - Norra Sverige 756000 CH - Schweiz/Suisse/Svizzera 792001 TR - Istanbul 792002 TR - Turkiye - Bati Marmara 792003 TR - Turkiye - EGE 792004 TR - Dogu Marmara 792005 TR - Bati Anadolu 792006 TR - Akdeniz 792007 TR - Orta Anadolu 792008 TR - Bati Karadeniz 792009 TR - Dogu Karadeniz 792010 TR - Kuzeydogu Anadolu 792011 TR - Ortadogu Anadolu 792012 TR - Guneydogu Anadolu 804001 UA - West 804002 UA - Centre 804003 UA - North 804004 UA - East 804005 UA - South 807000 MK – Poranesnata jugoslovenska Republika Makedonija 826112 GB - North East (England) 826113 GB - North West (England) 826114 GB - Yorkshire and the Humber 826115 GB - East Midlands (England) 826116 GB - West Midlands (England) 826117 GB - East of England 826118 GB – London 826119 GB - South East (England) 826120 GB - South West (England) 826121 GB - Wales

826122 GB - Scotland

826223 NIR - Northern Ireland

915000 KOS - Kosovo

Derivation:

Variable was computed from variable v371b_N3.

v372 - date of interview/datum des Interviews (Q137)

Q137

Date of the interview:

day/month/year

v373a - time of interview: start hour/Dauer des Interviews: Beginn Stunde (Q138)

Q138

Time of the interview

START HOUR:

v373b - time of interview: start minute/Dauer des Interviews: Beginn Minute (Q138)

Q138

Time of the interview

START MINUTE:

v374a - time of interview: end hour/Dauer des Interviews: Ende Stunde (Q138)

Q138

Time of the interview

END HOUR:

v374b - time of interview: end minute/Dauer des Interviews: Ende Minute (Q138)

Q138

Time of the interview

END MINUTE:

v374 - Duration of interview in minutes (Q138)

COMPUTED VARIABLE: Duration of interview in minutes

Derivation:

Variable was computed from variables v373a, v373b, v374a, v374b.

v375 - interest of respondent during interview (Q139)

Q139

During the interview the respondent was ...

- -5 other missing
- -4 question not asked
- -3 nap
- -2 na
- -1 dk
- 1 very interested
- 2 somewhat interested
- 3 not very interested

v375

Value	Label	Missing	Count	Percent	Valid Percent
-2	na	М	3	0.2	
1	very interested		946	62.6	62.8
2	somewhat interested		462	30.6	30.7
3	not very interested		99	6.6	6.6
	Sum		1510	100.0	100.0
	Valid Cases		1507		

v376 - language of interview (Q140)

Q140

In which language was the interview conducted?

Write in:

<CODING INSTRUCTION: USE PRE-SPECIFIED ISO 639-1 CODES FOR ALL LANGUAGES THAT QUESTIONNAIRE IS TRANSLATED

INTO>

Code:

33 German

Note:

Verbatim recorded answers standardized according to ISO 639-1 alpha-2 code.

Data for original string variable is not available in this dataset.

v376

Value	Label	Missing Count	Percent	Valid Percent
33	German	1510	100.0	100.0
	Sum	1510	100.0	100.0
	Valid Cases	1510		

AT84 - Country specific weight

Country-specific weight variable as delivered by country.

Note:

For detailed information, see Method Report.

Contents		page
StudyNo	GESIS Study Number	1
version	GESIS Archive Version	2
id_cocas	unified respondent number	3
caseno	original respondent number	4
intno	interviewer number	5
wave	wave	6
year	survey year	7
country	country code	8
country1	country code [with split ups]	9
c_abrv	country abbreviation	10
c_abrv1	country abbreviation [with split ups]	11
weight	weight	12
cntry_y	country_year	13
cntry1_y	country_[with split ups]- year	14
v1	how important in your life: work (Q1A)	15
v2	how important in your life: family (Q1B)	16
v3	how important in your life: friends and acquaintances (Q1C)	17
v4	how important in your life: leisure time (Q1D)	18
v5	how important in your life: politics (Q1E)	19
v6	how important in your life: religion (Q1F)	20
v7	how often discuss politics with friends (Q2)	21
v8	taking all things together how happy are you (Q3)	22
v9	describe your state of health these days (Q4)	23
v10	do you belong to: welfare organisation (Q5aA)	24
v11	do you belong to: religious organisation (Q5aB)	25
v12	do you belong to: cultural activities (Q5aC)	26
v13	do you belong to: trade unions (Q5aD)	27
v14	do you belong to: political parties/groups (Q5aE)	28
v15	do you belong to: local community action (Q5aF)	29
v16	do you belong to: 3w-development/human rights (Q5aG)	30
v17	do you belong to: environment (Q5aH)	31
v18	do you belong to: professional associations (Q5aI)	32
v19	do you belong to: youth work (Q5aJ)	33
v20	do you belong to: sports/recreation (Q5aK)	34
v21	do you belong to: womens groups (Q5aL)	35
v22	do you belong to: peace movement (Q5aM)	36
v23	do you belong to: voluntary health organisations (Q5aN)	37
AT01	do you belong to: fire fighters (Q5a_O)	38
v24	do you belong to: other groups (Q5aO)	39
f25	flag variable: do you belong to: none (Q5a)	40
v25	do you belong to: none (Q5a)	42
v28	do you work unpaid for: welfare organisation (Q5bA)	43
v29	do you work unpaid for: religious organisation (Q5bB)	44
v30	do you work unpaid for: cultural activities (Q5bC)	45
v31	do you work unpaid for: trade unions (Q5bD)	46
v32	do you work unpaid for: political parties/groups (Q5bE)	47
v33	do you work unpaid for: local community action (Q5bF)	48
v34	do you work unpaid for: 3w-development/human rights (Q5bG)	49

v35	do you work unpaid for: environment (Q5bH)	50
v36	do you work unpaid for: professional associations (Q5bl)	51
v37	do you work unpaid for: youth work (Q5bJ)	52
v38	do you work unpaid for: sports/recreation (Q5bK)	53
v39	do you work unpaid for: womens groups (Q5bL)	54
v40	do you work unpaid for: peace movement (Q5bM)	55
v41	do you work unpaid for: voluntary health organisations (Q5bN)	56
AT02	do you work unpaid for: fire fighters (Q5b_O)	57
v42	do you work unpaid for: other groups (Q5bO)	58
f43	flag variable: do you work unpaid for: none (Q5b)	59
v43	do you work unpaid for: none (Q5b)	60
AT03	reasons to work as a volunteer: sense of duty, moral obligation (Q5_at)	61
AT04	reasons to work as a volunteer: religious conviction (Q5_at)	62
AT05	reasons to work as a volunteer: to cause social or political changes (Q5_at)	63
AT06	reasons to work as a volunteer: to learn something new and make useful experiences (Q5_at)	64
AT07	reasons to work as a volunteer: because I want to do something for others (Q5_at)	65
v46	dont like as neighbours: people with criminal record (Q6A)	66
v47	dont like as neighbours: people of different race (Q6B)	67
v48	dont like as neighbours: left wing extremists (Q6C)	68
v49	dont like as neighbours: heavy drinkers (Q6D)	69
v50	dont like as neighbours: right wing extremists (Q6E)	70
v51	dont like as neighbours: large families (Q6F)	71
v52	dont like as neighbours: emotionally unstable people (Q6G)	72
v53	dont like as neighbours: muslims (Q6H)	73
v54	dont like as neighbours: immigrants/foreign workers (Q6I)	74
v55	dont like as neighbours: people with AIDS (Q6J)	75
v56	dont like as neighbours: drug addicts (Q6K)	76
v57	dont like as neighbours: homosexuals (Q6L)	77
v58	dont like as neighbours: jews (Q6M)	78
v59	dont like as neighbours: gypsies (Q6N)	79
v60	dont like as neighbours: christians (Q6O)	80
v62	people can be trusted/cant be too careful (Q7)	81
v63	most people try to take advantage of you or try to be fair (Q8)	82
v64	most of the time people try to be helpful or mostly looking out for themselves (Q9)	83
v65	how much control over your life (Q10)	84
v66	how satisfied are you with your life (Q11)	85
f67	flag variable: why do people live in need - most/second important (Q12/Q13)	86
v67	why do people live in need-most important (Q12)	87
v68	why do people live in need-2nd most important (Q13)	88
v69	important in a job: good pay (Q14A)	89
v70	important in a job: pleasant people (Q14B)	90
v71	important in a job: not too much pressure (Q14C)	91
v72	important in a job: job security (Q14D)	92
v73	important in a job: good hours (Q14E)	93
v74	important in a job: use initiative (Q14F)	94
v75	important in a job: useful for society (Q14G)	95
v76	important in a job: generous holidays (Q14H)	96
v77	important in a job: meeting people (Q14I)	97
v78	important in a job: achieving something (Q14J)	98
v79	important in a job: responsible job (Q14K)	99

v80	important in a job: interesting job (Q14L)	100
v81	important in a job: meeting abilities (Q14M)	101
v82	important in a job: learning new skills (Q14N)	102
v83	important in a job: family friendly (Q140)	103
v84	important in a job: have a say (Q14P)	104
v85	important in a job: people treated equally (Q14Q)	105
AT08	important in a job: good chances for promotion (Q14R)	106
AT09	important in a job: to have time off at the weekends (Q14S)	107
AT10	important in a job: a job respected by people in general (Q14T)	108
f86	flag variable: important in a job: none of these (Q14)	109
v86	important in a job: none of these (Q14)	110
v89	are you employed yes/no (Q15)	111
v90	how satisfied are you with your job (Q16)	112
v91	how free are you to make decisions in job (Q17)	113
AT11	and how satisfied or dissatisfied are you with your job security? (Q17_at)	114
v92	job needed to develop talents (Q18A)	115
v93	humiliating receiving money without working (Q18B)	116
v94	people turn lazy not working (Q18C)	117
v95	work is a duty towards society (Q18D)	118
v96	work comes always first (Q18E)	119
v97	leisure time: meeting nice people (Q19A)	120
v98	leisure time: relaxing (Q19B)	121
v99	leisure time: doing as I want (Q19C)	122
v100	leisure time: learning something new (Q19D)	123
v101	follow instructions at work/must be convinced first (Q20)	124
v102	jobs are scarce: giving(nation) priority (Q21A)	125
v103	jobs are scarce: giving men priority (Q21B)	126
AT12	jobs are scarce: giving younger people priority (Q21C)	127
v104	good and evil: clear/no clear guidelines (Q22)	128
v105	do you belong to a religious denomination (Q23)	129
v106	which religious denomination do you belong to (Q23a) (standardized)	130
v106_cs	which religious denomination do you belong to (Q23a)	131
v107	did you belong to a religious denomination (Q24)	132
v108	did you belong to a religious denomination (Q24a) (standardized)	133
v108_cs	did you belong to a religious denomination (Q24a)	134
v109	how often attend religious services (Q25)	135
v110	how often attend religious services 12 years old (Q26)	136
v111	religious service important: birth (Q27A)	137
v112	religious service important: marriage (Q27B)	138
v113	religious service important: death (Q27C)	139
v114	are you a religious person (Q28)	140
v115	church answers to: moral problems (Q29A)	141
v116	church answers to: family life problems (Q29B)	142
v117	church answers to: spiritual needs (Q29C)	143
v118	church answers to: social problems (Q29D)	144
v119	do you believe in: God (Q30A)	145
v120	do you believe in: life after death (Q30B)	146
v121	do you believe in: hell (Q30C)	147
v122	do you believe in: heaven (Q30D)	148
v123	do you believe in: sin (Q30E)	149

v124	do you believe in: re-incarnation (Q31)	150
v125	which statement closest to your beliefs (Q32)	151
v126	own way of connecting with the divine (Q33)	152
v127	how spiritual are you (Q34)	153
v128	only one true religion or no religion offers any truths (Q35)	154
v129	how important is God in your life (Q36)	155
v130	do you get comfort and strength from religion (Q37)	156
v131	do you take moments of prayer/meditation (Q38)	157
v132	how often do you pray to God outside religious services (Q39)	158
v133	do you believe that lucky charm protects (Q40)	159
v134	politicians and God (Q41A)	160
v135	religious leaders and influence government decisions (Q41B)	161
AT13	more time for prayer, mediation or contemplation in all schools (Q41C)	162
AT14	how often do you think about the meaning of life? (Q41_1)	163
AT15	did you receive a religious upbringing by your parents? (Q41_2)	164
AT16	agree with: there is a God who attends to every human personally (Q41_3A)	165
AT17	agree with: impossible to know if there is anything beyond this world (Q41_3B)	166
AT18	agree with: life carries its meaning within itself (Q41_3C)	167
AT19	agree with: you dont feel much of Gods existence (Q41_3D)	168
AT20	agree with: life is determined by the laws of nature (Q41_3E)	169
AT21	agree with: life only has a meaning if you give it one (Q41_3F)	170
AT22	agree with: there is something like a higher power (a higher being) (Q41_3G)	171
AT23	agree with: there is a God who revealed Himself in Jesus (Q41_3H)	172
AT24	agree with: there is not much meaning to life (Q41_3I)	173
AT25	agree with: life is just a part of the process of nature/part of the developments in nature (Q41_3J)	174
AT26	agree with: the resurrection of Jesus gives meaning to my death (Q41_3K)	175
AT27	agree with: the future lies in the kingdom of God that has been promised by Jesus (Q41_3L)	176
AT28	agree with: death is a transition to another form of existence (Q41_3M)	177
AT29	agree with: reincarnation of the soul (rebirth) of the soul in another life exists (Q41_3N)	178
AT30	agree with: after death, everything comes to a final end (Q41_30)	179
AT31	agree with: the eternal cycle of man, nature and the cosmos is the higher power (Q41_3P_at)	180
AT32	how often do you think about religious topics? (Q41_4_at)	181
AT33	how often: situation where God or something divine intervenes in your life (Q41.5_at)	182
AT34	how often: a feeling of unity with everything (Q41.6_at)	183
AT35	how often do you meditate? (Q41.7_at)	184
AT36	how strongly do you believe that God or something divine exists? (Q41.8_at)	185
AT37	how often: feel that God or something divine wants to tell or show you something\ wants to give you a sign	186
	(Q41.9_at)	
v136	important in marriage: faithfulness (Q42A)	187
v137	important in marriage: adequate income (Q42B)	188
v138	important in marriage: same social background (Q42C)	189
v139	important in marriage: shared religious beliefs (Q42D)	190
v140	important in marriage: good housing (Q42E)	191
v141	important in marriage: agreement on politics (Q42F)	192
v142	important in marriage: live apart from in-laws (Q42G)	193
v143	important in marriage: happy sexual relationship (Q42H)	194
v144	important in marriage: share household chores (Q42I)	195
v145	important in marriage: children (Q42J)	196
v146	important in marriage: discuss problems (Q42K)	197
v147	important in marriage: time for friends and personal hobbies (Q42L)	198

AT38	important in marriage: mutual respect and appreciation (Q42M_at)	199
AT39	important in marriage: mutual understanding and tolerance (Q42N_at)	200
AT40	important in marriage: spending as much time together as possible (Q42O_at)	201
AT41	important in marriage: talking a lot about mutual interests (Q42P_at)	202
AT42	important in marriage: same ethnic or national background (Q42Q_at)	203
AT43	important in marriage: shared goals in life (Q42R_at)	204
AT44	important in marriage: the partner having a secure job (Q42S_at)	205
v148	children need both parents to grow up happily (Q43)	206
v149	women need children in order to be fulfilled (Q44)	207
v150	marriage is outdated (Q45)	208
v151	woman single parent, no stable relationship with man (Q46)	209
v152	men need children in order to be fulfilled (Q47A)	210
v153	long-term relationship necessary to be happy (Q47B)	211
v154	homosex couples - adopt children (Q47C)	212
v155	It is alright to live together without getting married (Q47D)	213
v156	duty towards society to have children (Q47E)	214
v157	people should decide themselves to have children (Q47F)	215
v158	It is childs duty to take care of ill parent (Q47G)	216
v159	working mother warm relationship with children (Q48A)	217
v160	pre-school child suffers with working mother (Q48B)	218
v161	women really want home and children (Q48C)	219
v162	being housewife as fulfilling as paid job (Q48D)	220
v163	job best way for independence women (Q48E)	221
v164	husband+wife contribute to household income (Q48F)	222
v165	fathers as well suited to look after children as mothers (Q48G)	223
v166	men should take the same responsibility for home and children (Q48H)	224
v167	love and respect parents always/earned (Q49)	225
v168	parents responsibilities to their children at expense of/not sacrifice own well-being (Q50)	226
v169	children responsibilities to their parents in need at expense of/not sacrifice own well-being (Q51)	227
v170	learn children at home: good manners (Q52A)	228
v171	learn children at home: independence (Q52B)	229
v172	learn children at home: hard work (Q52C)	230
v173	learn children at home: feeling of responsibility (Q52D)	231
v174	learn children at home: imagination (Q52E)	232
v175	learn children at home: tolerance+respect (Q52F)	233
v176	learn children at home: thrift (Q52G)	234
v177	learn children at home: determination, perseverance (Q52H)	235
v178	learn children at home: religious faith (Q52I)	236
v179	learn children at home: unselfishness (Q52J)	237
f180	flag variable: learn children at home (Q52)	238
v180	learn children at home: obedience (Q52K)	239
f181	flag variable: learn children at home: none (Q52)	240
v181	learn children at home: none (Q52)	241
v184	abortion if woman not married approve/disapprove (Q53A)	242
v185	abortion if couple doesnt want more children approve/disapprove (Q53B)	243
AT45	abortion if the mothers life is endangered by the pregnancy (Q53C_at)	244
AT46	abortion if it is likely that the child will be handicapped (Q53D_at)	245
v186	how interested are you in politics (Q54)	246
v187	signing a petition (Q55A)	247
v188	joining in boycotts (Q55B)	248

v189	attending lawful demonstrations (Q55C)	249
v190	joining unofficial strikes (Q55D)	250
v191	occupying buildings/factories (Q55E)	251
v192	more important: freedom or equality (Q56)	252
v193	political view: left-right (Q57)	253
v194	individual-state responsibility for providing (Q58A)	254
v195	take any job-right to refuse job when unemployed (Q58B)	255
v196	competition good-harmful for people (Q58C)	256
v197	state give more freedom to-control firms more effectively (Q58D)	257
v198	equalize incomes-incentives for individual effort (Q58E)	258
v199	private-government ownership business (Q58F)	259
v200	opinion on society (Q59)	260
f201	flag variable: aims of this country - most/second important (Q60/Q61)	261
v201	aims of this country-most important (Q60)	262
v202	aims of this country-2nd most important (Q61)	263
v203	good/bad: less emphasis on money (Q62A)	264
v204	good/bad: more respect for authority (Q62B)	265
AT47	good\bad: decrease the importance of work in our lives (Q62C_at)	266
AT48	good\bad: more emphasis on the development of technology (Q62D_at)	267
AT49	good\bad: greater emphasis on the development of the individual (Q62E_at)	268
AT50	good\bad: more emphasis on family life (Q62F_at)	269
AT51	good\bad: a simple and more natural lifestyle (Q62G_at)	270
v205	how much confidence in: church (Q63A)	271
v206	how much confidence in: armed forces (Q63B)	272
v207	how much confidence in: education system (Q63C)	273
v208	how much confidence in: the press (Q63D)	274
v209	how much confidence in: trade unions (Q63E)	275
v210	how much confidence in: the police (Q63F)	276
v211	how much confidence in: parliament (Q63G)	277
v212	how much confidence in: civil service (Q63H)	278
v213	how much confidence in: social security system (Q63I)	279
v214	how much confidence in: european union (Q63J)	280
v215	how much confidence in: NATO (Q63K)	281
v216	how much confidence in: united nations organisation (Q63L)	282
v217	how much confidence in: health care system (Q63M)	283
v218	how much confidence in: justice system (Q63N)	284
v219	how much confidence in: major companies (Q63O)	285
v220	how much confidence in: environmental organizations (Q63P)	286
v221	how much confidence in: political parties (Q63Q)	287
v222	how much confidence in: government (Q63R)	288
v223	are you satisfied with democracy (Q64)	289
v224	view government: very bad-very good (Q65)	290
v225	political system: strong leader (Q66A)	291
v226	political system: experts making decisions (Q66B)	292
v227	political system: the army ruling (Q66C)	293
v228	political system: democratic (Q66D)	294
v229	democracy: best political system (Q67A)	295
v230	democracy: causes bad economy (Q67B)	296
v231	democracy: is indecisive (Q67C)	297
v232	democracy: cannot maintain order (Q67D)	298

v233	do you justify: claiming state benefits (Q68A)	299
v234	do you justify: cheating on tax (Q68B)	300
v235	do you justify: joyriding (Q68C)	301
v236	do you justify: taking soft drugs (Q68D)	302
v237	do you justify: lying in own interest (Q68E)	303
v238	do you justify: adultery (Q68F)	304
v239	do you justify: accepting a bribe (Q68G)	305
v240	do you justify: homosexuality (Q68H)	306
v241	do you justify: abortion (Q68I)	307
v242	do you justify: divorce (Q68J)	308
v243	do you justify: euthanasia (Q68K)	309
v244	do you justify: suicide (Q68L)	310
v245	do you justify: paying cash to avoid taxes (Q68M)	311
v246	do you justify: having casual sex (Q68N)	312
v247	do you justify: avoiding fare public transport (Q68O)	313
v248	do you justify: prostitution (Q68P)	314
v249	do you justify: experiments human embryos (Q68Q)	315
v250	do you justify: manipulation food (Q68R)	316
v251	do you justify: invitro fertilization (Q68S)	317
v252	do you justify: death penalty (Q68T)	318
AT52	do you justify: causing damage to a parked car and not reporting it (Q68U_at)	319
AT53	do you justify: buying stolen goods (Q68V_at)	320
f253	flag variable: belong to geographic group - most/second (Q69/Q70)	321
v253	belong to geographic group - most (Q69)	322
v254	belong to geographic group - next (Q70)	323
v255	are you a citizen of (country) (Q71)	324
v256	how proud are you to be a (country) citizen (Q72)	325
v257	EU fears: loss of social security (Q73A)	326
v258	EU fears: lose national identity/culture (Q73B)	327
v259	EU fears: own country pays (Q73C)	328
v260	EU fears: loss of power (Q73D)	329
v261	EU fears: loss of jobs (Q73E)	330
v262	European Union enlargement (Q74)	331
v263	would you vote at a general election tomorrow (Q75)	332
v264	which political party would you vote for (Q75a)	333
v264_LR	which political party would you vote for - left/right scale (Q75a)	334
v265	which political party appeals to you most (Q75b)	335
v265_LR	which political party appeals to you most - left/right scale (Q75b)	336
v266	work: people from less developed countries (Q76)	337
v267	terrorism: can be justified/must be condemned (Q77)	338
v268	immigrants take away jobs from [nationality] (Q78A)	339
v269	immigrants undermine countrys cultural life (Q78B)	340
v270	immigrants increase crime problems (Q78C)	341
v271	immigrants are a strain on welfare system (Q78D)	342
v272	immigrants will become a threat to society (Q78E)	343
v273	immigrants maintain own/take over customs (Q78F)	344
v274	immigrants living in your country: feels like a stranger (Q79A)	345
v275	immigrants living in your country: there are too many (Q79B)	346
v276	important: to have been born in [country] (Q80A)	347
v277	important: to respect [country nationality] political institutions and laws (Q80B)	348

v278	important: to have [country nationality] ancestry (Q80C)	349
v279	important: to be able to speak [country language] (Q80D)	350
v280	important: to have lived in [country] for a long time (Q80E)	351
v281	how often do you follow politics in media (Q81)	352
v282	give information to help justice (Q82A)	353
v283	people should stick to own affairs (Q82B)	354
v284	are you concerned with: immediate family (Q83A)	355
v285	are you concerned with: people neighbourhood (Q83B)	356
v286	are you concerned with: people own region (Q83C)	357
v287	are you concerned with: fellow countrymen (Q83D)	358
v288	are you concerned with: europeans (Q83E)	359
v289	are you concerned with: humankind (Q83F)	360
v290	are you concerned with: elderly people (Q84A)	361
v291	are you concerned with: unemployed people (Q84B)	362
v292	are you concerned with: immigrants (Q84C)	363
v293	are you concerned with: sick and disabled (Q84D)	364
v294	are you concerned with: poor children (Q84E)	365
v295	environment: giving part of income (Q85A)	366
v296	environment: approaching the limit of people (Q85B)	367
v297	environment: interference produces disastrous consequences (Q85C)	368
v298	environment: human ingenuity insures earth remaining fit (Q85D)	369
v299	environment: nature is strong enough to cope with impacts of industry (Q85E)	370
v300	environment: humans were meant to rule over nature (Q85F)	371
v301	environment: if things continue we will experience a catastrophe (Q85G)	372
AT54	agree with: where there is strict authority, there is also justice (Q85_at)	373
AT55	agree with: the most important thing that children have to learn is obedience (Q85B_at)	374
AT56	agree with: only those with sufficient status should take part in discussions/ decisions (Q85C_at)	375
AT57	agree with: young people have too much freedom (Q85D_at)	376
AT58	agree with: the meaning of life consists of attaining a respectable position (Q85E_at)	377
AT59	agree with: security and prosperity are more important than freedom (Q85F_at)	378
AT60	agree with: purpose of a job should first and foremost consist in guaranteeing a secure income (Q85G_at)	379
AT61	agree with: its important to live life as comfortably\pleasant\convenient as possible (Q85H_at)	380
AT62	agree with: its important for a person to be happy. How they do that is everyones own business (Q85I_at)	381
AT63	agree with: everyone has to solve their problems themselves\ on their own (Q85J_at)	382
AT64	agree with: foreigners should adapt their lifestyle more to that of the local people (Q85K_at)	383
AT65	agree with: when jobs become scarce, foreigners should be sent back to their home countries (Q85L_at)	384
AT66	agree with: foreigners should be banned from all political participation in our country (Q85M_at)	385
AT67	agree with: foreigners should choose their spouses among their kind (Q85N_at)	386
AT68	agree with: it should be permitted to take away from the wealthy in order to give it to those in need (Q85O_at)	387
AT69	agree with: differences in income should be reduced (Q85P_at)	388
AT70	agree with: if we could all get by on a little bit less, poverty would soon be gone (Q85Q_at)	389
AT71	agree with: the most important thing that children have to learn, is sharing with others (Q85R_at)	390
AT72	agree with: sharing can be learned best from ones own family (Q85S_at)	391
AT73	agree with: one cannot learn to solve conflicts without a family (Q85T_at)	392
AT74	agree with: everything ends with death (Q85U_at)	393
AT75	agree with: I hope there is a life after death (Q85V_at)	394
AT76	agree with: the dead will be resurrected in body and soul (Q85W_at)	395
AT77	agree with: life meaningful only because God exists (Q85X_at)	396
AT78	agree with: the meaning of life is to make the most of it (Q85Y_at)	397
AT79	agree with: i dont know what humans exist\live for (Q85Z_at)	398

AT80	agree with: decisions should be made by those who are successful in life (Q85AA_at)	399
AT81	agree with: the weaker members of society should be subordinate to the stronger ones (Q85BB_at)	400
AT82	agree with: everyone is the architect of their own fortune (Q85CC_at)	401
AT83	agree with: everyone is responsible for their own lack of success (Q85DD_at)	402
v302	sex respondent (Q86)	403
v303	year of birth respondent (Q87)	404
v304	having [countrys] nationality (Q88)	405
v305b	respondents nationality - ISO 3166-1 code/Staatsbürgerschaft des Befragten ISO 3166-1 Code (Q89)	406
v306	respondent born in [country] (Q90)	411
v307b	respondents country of birth - ISO 3166-1 code/Geburtsland des Befragten ISO 3166-1 Code (Q91)	412
v308	year in which respondent came to live in [country] (Q92)	417
v309	father born in [country] (Q93)	418
v310b	fathers country of birth - ISO 3166-1 code/Geburtsland des Vaters ISO 3166-1 Code (Q94)	419
v311	mother born in [country] (Q95)	424
v312b	mothers country of birth - ISO 3166-1 code/Geburtsland der Mutter ISO 3166-1 Code (Q96)	425
v313	current legal marital status respondent (Q97)	430
v314	lived with partner before marriage (Q98)	431
v315	living with partner (Q99)	432
v316	having steady relationship (Q100)	433
v317	did you live with a partner before (Q101)	434
v318	married to this partner or in registered partnership (Q102)	435
v319	lived with this partner before marriage/registration of partnership (Q103)	436
v320	end of relationship because of separation or partners death	437
v321	how many children do you have (Q105)	438
v322	year in which firstborn child was born (Q106)	439
v323	living in household: partner, husband or wife (Q107)	440
v324a	living in household: children (Q107)	441
v324b	living in household: number of children (Q107)	442
v325a	living in household: parents (Q107)	443
v325b	living in household: number of parents (Q107)	444
v326a	living in household: grandparents (Q107)	445
v326b	living in household: number of grandparents (Q107)	446
v327a	living in household: relatives (Q107)	447
v327b	living in household: number of relatives (Q107)	448
v328a	living in household: non relatives (Q107)	449
v328b	living in household: number of non relatives (Q107)	450
v329a	experienced: death of own children (Q108)	451
v329b	age experienced: death of own children (Q108)	452
v330a	experienced: divorce of own children (Q108)	453
v330b	age experienced: divorce of own children (Q108)	454
v331a	experienced: divorce of parents (Q108)	455
v331b	age experienced: divorce of parents (Q108)	456
v332a	experienced: divorce of relative (Q108)	457
v332b	age experienced: divorce of relative (Q108)	458
v333a	experienced: death of father (Q108)	459
v333b	age experienced: death of father (Q108)	460
v334a	experienced: death of mother (Q108)	461
v334b	age experienced: death of mother (Q108)	462
v335	age completed education respondent (Q109)	463
v336	Educational level respondent: ISCED-code one digit (Q110)	464

v336_2	Educational level respondent: ISCED-code two digits (Q110)	465
v336_3	Educational level respondent: ISCED-code three digits (Q110)	467
v336_cs	Country specific: ISCED code education respondent (Q110)	469
v337	paid employment/no paid employment (Q111)	470
v338	employment/self-employment: last job (Q111A)	471
v339ISCO	kind of job respondent - ISCO88 code (Q112)	472
v339SIOPS	occupational status respondent - SIOPS (Q112)	484
v339ISEI	occupational status respondent - ISEI (Q112)	485
v339ESeC	occupational status respondent - European ESeC (Q112)	486
v339egp	occupational status respondent - egp11 (Q112)	487
v340	respondent had/has how many employees (Q113)	488
v341	do you supervise someone (Q114)	489
v341a	how many people do you supervise (Q114a)	490
v342	partner/spouse born in [country] (Q115)	491
v343b	spouse/partners country of birth - ISO 3166-1 code/Geburtsland des (Ehe-)Partners/der (Ehe-)Partnerin ISO	492
	3166-1 Code (Q115a)	
v344	Educational level partner: ISCED-code one digit (Q116)	497
v344_2	Educational level partner: ISCED-code two digits (Q116)	498
v344_3	Educational level partner: ISCED-code three digits (Q116)	500
v344_cs	Country specific:ISCED code education partner (Q116)	502
v345	paid employment/no paid employment spouse/partner (Q117)	503
v345a	employment/self-employment: last job (Q117a)	504
v346ISCO	kind of job spouse/partner - ISCO88 code (Q118)	505
v346SIOPS	occupational status spouse/partner - SIOPS (Q118)	517
v346ISEI	occupational status spouse/partner - ISEI (Q118)	518
v346ESeC	occupational status spouse/partner - European ESeC (Q118)	519
v346egp	occupational status spouse/partner - egp11 (Q118)	520
v347	spouse/partner had/has how many employees (Q119)	521
v348	does spouse/partner supervise someone (Q120)	522
v348a	how many people does she/he supervise (Q120a)	523
v349	respondent experienced unemployment longer than 3 months (Q121)	524
v350	spouse/partner experienced unemployment longer than 3 months (Q122)	525
v351	dependency on social security during last 5 years respondent (Q123)	526
v352	dependency on social security during last 5 years spouse/partner (Q124)	527
v353WK	Weekly household income (Q125)	528
v353MM	monthly household income (Q125)	530
v353YR	annual household income (Q125)	532
v353W_cs	Country specific: Weekly household income (Q125)	534
v353M_cs	Country specific:Monthly household income (Q125)	536
v353Y_cs	Country specific: Annual household income (Q125)	538
v354	lived with parents at the age of 14 (Q126)	540
v355	Educational level father [mother]: ISCED-code one digit (Q127)	541
v355_2	Educational level father [mother]: ISCED-code two digits (Q127)	542
v355_3	Educational level father [mother]: ISCED-code three digits (Q127)	544
v355_cs	Country specific:ISCED code education father (Q127)	546
v356	father/mother employed at respondents age of 14 (Q128)	548
v357ISCO	kind of job father/mother - ISCO88 code (Q129a)	549
v357SIOPS	occupational status father/mother - SIOPS (Q129a)	561
v357ISEI	occupational status father/mother - ISEI (Q129a)	562
v357ESeC	occupational status father/mother - European ESeC (Q129a)	563

EVS 2008 - Austria

057	and (O420a)	FC.
v357egp	occupational status father/mother - egp11 (Q129a)	564
v358	father/mother had how many employees (Q130)	565
v359	did father/mother supervise someone (Q131)	566
v359a	how many people did she/he supervise (Q131a)	567
v360	mother liked to read books (Q132A)	568
v361	discussed politics with mother (Q132B)	569
v362	mother liked to follow the news (Q132C)	570
v363	parent(s) had problems making ends meet (Q132D)	571
v364	father liked to read books (Q132E)	572
v365	discussed politics with father (Q132F)	573
v366	father liked to follow the news (Q132G)	574
v367	parent(s) had problems replacing broken things (Q132H)	575
v368b_CC	Region at age 14: country (Q133)	576
v368b_N3	Region at age 14: NUTS-3 code (Q133)	581
v368b_N2	Region at age 14: NUTS-2 code (Q133)	616
v368b_N1	Region at age 14: NUTS-1 code (Q133)	624
v370	size of town where interview was conducted (Q135)	628
v371b_N3	Region: NUTS-3 code (Q136)	629
v371b_N2	Region: NUTS-2 code (Q136)	664
v371b_N1	Region: NUTS-1 code (Q136)	672
v372	date of interview/datum des Interviews (Q137)	676
v373a	time of interview: start hour/Dauer des Interviews: Beginn Stunde (Q138)	677
v373b	time of interview: start minute/Dauer des Interviews: Beginn Minute (Q138)	678
v374a	time of interview: end hour/Dauer des Interviews: Ende Stunde (Q138)	679
v374b	time of interview: end minute/Dauer des Interviews: Ende Minute (Q138)	680
v374	Duration of interview in minutes (Q138)	681
v375	interest of respondent during interview (Q139)	682
v376	language of interview (Q140)	683
AT84	Country specific weight	684